Wellbeing: A Theoretical and Empirical study

Georgina Hobbs
PhD 2016

# Wellbeing: A Theoretical and Empirical Study

# **Georgina Hobbs**

A thesis submitted in partial fulfilment of the requirements of the

Manchester Metropolitan University for the degree of

Doctor of Philosophy

Department of Psychology

Manchester Metropolitan University

September 2016

#### Abstract

Key words - Wellbeing, theoretical/empirical enquiry, personal responsibility, historical/linguistic contextualisation.

There is mounting theoretical evidence that wellbeing has been mobilised within a personal responsibility agenda (PRA). However, there is little empirical evidence to support this. Additionally, little is understood about the role of historical/linguistic factors in this mobilisation and their influence on mainstream representations of wellbeing. Research indicates current representations of wellbeing assign primacy to academic/professional conceptualisations neglecting lay conceptualisations.

This thesis examines wellbeing through complementary theoretical and empirical investigations. This responds to a need to explore: the mobilisation of wellbeing through empirical investigation; how and why historical and linguistic factors contribute to this mobilisation; whether this mobilisation is assimilated into lay conceptualisations. This is important given wellbeing is a central focus of international and national healthcare policy, an influential concept in health discourse and widely utilised to underpin health and social care services.

Three research strategies were utilised: a synthesis of theoretical evidence; a historical/linguistic contextualisation of wellbeing and qualitative analysis of participant data. Data were collected from existing literature, historical/linguistic sources and semi-structured interviews.

Findings illustrated that historical/linguistic associations between wellbeing and personal responsibility have facilitated its mobilisation within a PRA. Findings also identified that participants rarely associated wellbeing with personal responsibility, which was primarily perceived as the responsibility of the state, despite widespread dissemination of agentic representations of wellbeing.

Findings suggest that wellbeing has been mobilised within a PRA for governance purposes. This has been achieved primarily through the promulgation of agentic representations of wellbeing. However, findings indicated a disjunction between participant and academic/professional conceptualisations of responsibility for wellbeing. This contributes to the revitalised debate about representations of wellbeing and the division of responsibility for wellbeing between the individual and the state.

#### **ACKNOWLEDGEMENTS**

Thanks to my supervisors, Professor Carolyn Kagan and Dr Martin King for their unfailing support and input throughout this process.

Thank you to all the participants who took part in this research especially the coresearchers, Jim Trotman and Alison Power. I am forever indebted to them.

I can not express strongly enough how much I value the love and support of Wayne Hardy and Thomas Reilly. They have always provided me with the love and support I needed to finish this thesis.

For funding this project I would like to thank the Research Institute for Health and Social Change at Manchester Metropolitan University.

Finally thanks to my family and friends for their love and support. My mum and sister are not here to witness my achievement but I know how proud they would have been.

Wellbeing: A Theoretical and Empirical study

Georgina Hobbs
PhD 2016

Table of Contents	
Chapter 1: Introduction	. 12
1.1: Thesis overview and précis of chapters	. 12
1.2: Conceptual clarifications	. 14
1.3: Background to theoretical investigation	. 16
1.4: Rationale for theoretical investigation	. 17
1.5: Background for empirical investigation	. 19
1.6: Rationale for empirical investigation	. 22
1.7: Theoretical framework	. 24
1.7.1: Study aims	. 24
1.7.2: Research questions	. 25
1.8: Scope of the empirical study	. 26
1.9: Rationale for study parameters	. 26
1.9.1: Increasing proportion of UK population with LTCs	. 26
1.9.2: Ageing populations	. 28
1.9.3: Age parameters	. 28
1.9.4: Local context	. 29
1.10: Summary	. 30
Chapter 2: Literature Review	. 32
2.1: Literature search methods	. 32
2.2: Domain approach to wellbeing	. 33
2.3: The ageing process and wellbeing	. 34
2.4: Health	. 36
2.4.1: Adaptation	. 37
2.4.2: Biographical disruption	. 38
2.4.3: Self-reported health	. 39
2.4.4: Disability and impairment	. 39
2.5: Psychological* wellbeing	. 41
2.6: Physical wellbeing	. 45
2.7: Spiritual, religious and emotional wellbeing	. 46
2.7.1: Spiritual wellbeing	. 46
2.7.2: Religious wellbeing	. 47
2.7.3: Emotional wellbeing	. 48
2.8: Economic wellbeing	. 50
2.8.1: The relationship between economic wellbeing and income	. 50
2.8.2: Unemployment	. 53

2.8.3: Employment	54
2.9: Social wellbeing*	57
2.10: Environmental wellbeing*	60
2.10.1: Natural environment*	60
2.10.2: The built environment	62
2.10.3: Housing	63
2.10.4: Public spaces	64
2.11: Temporality	66
2.12: Component review summary	68
2.13: Chronological approach to wellbeing	69
2.13.1: The impact of the Local Government Act 2000	70
2.13.2: The wellbeing agenda	71
2.13.3: The role of personal responsibility	72
2.13.4: Structure versus agency debate	73
2.13.5: Situating wellbeing within historical and linguistic context	s75
2.13.6: Chronological review summary	76
2.13.7: Integration of theoretical and empirical reviews	77
Chapter 3: Theoretical Critiques of Wellbeing	79
3.1: Theoretical framework	80
3.2: Sociological critiques	82
3.2.1: Sense making	82
3.2.2: Challenging the wellbeing agenda	83
3.2.3: The role of government	84
3.2.4: Sociological contributions to the wellbeing debate	85
3.3: Health critiques	87
3.3.1: Concerns about the role of wellbeing	87
3.3.2: Consumer culture	88
3.3.3: Wellbeing and its relationship to welfare	89
3.3.4: Blurring the boundaries between wellbeing and QoL resea	ırch 91
3.4: Where wellbeing fits within citizenry and governance	93
3.4.1: Governmentality	93
3.4.2: Mechanisms of governance	94
3.4.3: Governance, responsibilisation and the WBA	96
3.4.4: Alternative accounts of the role of wellbeing	98
3.4.5: Alternative theories for the mobilisation of wellbeing	99
3.4.6: Summary	101

C	Chapter 4: Methodology and Methods Chapter	102
	4.1: Background to study	102
	4.2: Methodological approach	102
	4.2.1: Epistemology	103
	4.2.2: Paradigm	104
	4.2.3: Theoretical perspective	105
	4.3: Historical analysis	106
	4.3.1: Historical comprehension	109
	4.3.2: Historical Postmodernism	109
	4.4: Methods for empirical study	110
	4.4.1: Qualitative approach	110
	4.4.2: Sampling framework	113
	4.4.3: Selection bias	114
	4.4.4: Sample size	115
	4.5: Recruitment of participants	116
	4.5.1: Participants' profile	118
	4.5.2: Table 1- Participants' profile	120
	4.6: Co-researcher involvement	122
	4.6.1: Lay versus professional dichotomy	123
	4.6.2: Co-researcher training	124
	4.6.3: Remuneration	124
	4.6.4: Collaboration	125
	4.6.5: Choice of methods	126
	4.6.6: Analysing and interpreting data	127
	4.6.7: Writing up findings	127
	4.6.8: Dissemination	128
	4.6.9: Ownership	128
	4.6.10: Areas of research beyond the co-researchers' jurisdiction	129
	4.7: Method of data collection	129
	4.7.1: Interview process	130
	4.7.2: Analytic method	131
	4.7.3: Analytic process	133
	4.8: Reflexivity	135
	4.8.1: Epistemological reflexivity	137
	4.8.2: Reflective process	138
	4.9: Ethical considerations	139

Chapter 5: Linguistic and Historical Contextualisation of Wellbeing	. 141
5.1: Rationale	. 141
5.2 Linguistic contextualisation of wellbeing	. 141
5.2.1 Ontology of wellbeing	. 141
5.2.2 Logomachy	. 143
5.2.3: Etymology – Wellbeing, health, welfare and happiness	. 143
5.2.4 The semantic extension of wellbeing	. 146
5.3: Historical contextualisation of wellbeing	. 150
5.3.1: Rationale for historical analysis	. 150
5.3.2 Trans-historical account of wellbeing	. 151
5.3.3: Changing conceptualisation of wellbeing and welfare	. 155
5.3.4: The period of Enlightenment	. 157
5.3.5: Wellbeing, self-improvement and shifting normative values	. 160
5.3.6: Welfare	. 162
5.3.7: Beginning of the shift from welfare to wellbeing	. 163
5.3.8: The influence of the welfare state on conceptualisations of wellbeing	. 164
5.3.9: From welfare to wellbeing	. 165
5.3.10: A 'welfare to wellbeing' strategy	. 166
5.3.11: The emergence of wellbeing into the collective consciousness	. 167
5.3.12: The wider application of wellbeing	. 168
5.3.13: Challenges of the 21st century	. 169
5.4: Summary	. 173
Chapter 6: Findings: Experiential Conceptualisations of Personal Wellbeing	. 175
6.1: The ageing process	. 175
6.1.1: Premature ageing	. 177
6.1.2: Intimacy	. 178
6.1.3: Transitioning	. 180
6.1.4: Summary	. 182
6.2: Health	. 183
6.2.1: Adaptation	. 183
6.2.2: Coping mechanisms	. 186
6.2.3: Social sequelae of health	. 187
6.2.4: Structural issues	. 188
6.2.5: Acclimatisation	. 191
6.2.6: Summary	. 194
6.3: Markers of personal identities	. 195

6.3.1: Life aspirations	195
6.3.2: Power and control	197
6.3.3: Prioritisation and allocation of services	201
6.3.4: Cultural and group identification	202
6.3.5: Respect/disrespect	204
6.3.6: Autonomy and independence	206
6.3.7: Financial considerations	208
6.3.8: Summary	211
6.4: Social interaction	212
6.4.1: Filial legacy	214
6.4.2: Participation in society	216
6.4.3: Relationships	217
6.4.4: Social activities	217
6.4.5: Isolation and loneliness	219
6.4.6: Summary	221
6.5: The natural environment	222
6.5.1: Communal green spaces	223
6.5.2: Gardens	224
6.5.3: Unregulated green spaces	226
6.5.4: Summary	227
6.6: Built environment*	227
6.6.1: Aesthetic appeal	228
6.6.2: Community spaces	229
6.6.3: Community resources	230
6.6.4: Structural concerns	230
6.6.5: Living space	232
6.6.6: Summary	235
6.7: Temporality	236
6.7.1: Temporal preferences	237
6.7.2: Temporal attitudes	238
6.7.3: Temporal perspectives	238
6.7.4: Summary	240
6.8: Summary of study findings	240
Chapter 7: Discussion	242
7.1 Introduction	242
7.2. How wellheing has been mobilised	242

7.3: Why wellbeing has been mobilised	244
7.4: Is there empirical evidence participants associated wellbeing with persor responsibility?	
7.5: Is wellbeing a meaningful concept for lay people?	249
7.6: The role of the natural environment on wellbeing	251
7.7: The role of temporality on wellbeing	254
7.8: Limitations	256
7.9: Strengths	258
7.10: Summary	259
Chapter 8: Conclusion	261
8.1: A consideration of whether the study aims have been addressed	261
8.2: Notable findings	262
8.3: Policy recommendations	263
8.4: Implications for practice	264
8.5: Research recommendations	266
8.6: Contributions to knowledge	267
8.7: Concluding comments	268
References	270
Appendices	373
Appendix A: Glossary	373
Appendix B: Conceptual Distinctions between Wellbeing and Quality of Life	381
Appendix C: Literature Review Search Methods	407
Appendix D: Lay Versus Traditional Researcher: strengths and weaknesses	417
Appendix E: Research Participant Information Sheet	436
Appendix F: Consent Form Version 6 - 23rd July 2007	438
Appendix G: Interview schedule	439
Appendix H: Co-researcher Training	441
Appendix I: Researcher's Reflexive Journal	444
Appendix J: Alison's Reflective Diary (2008–2012)	476
Appendix K: Jim's Reflective Diary (2008–2011)	490
Appendix L: Second Interview with Jim - 27 <sup>th</sup> November 2009	496
Appendix M: Third Interview with Alison - 23rd March 2010	501
Appendix N: Interview with Tim - 17 <sup>th</sup> August 2008	507
Appendix O: Interview with Ivy - January 9 <sup>th</sup> 2008	513
Appendix P: Well Being - Jim Trotman	520

# **Chapter 1: Introduction**

# 1.1: Thesis overview and précis of chapters

This thesis has a dual purpose of explicating understandings of the concept of wellbeing simultaneously from theoretical and empirical perspectives. This thesis is a report of two investigations, which are intended to complement each other.

The first is an investigation of the mobilisation of wellbeing within a personal responsibility agenda\* (PRA), reported in chapters three and four. The second is an investigation of lay conceptualisations of personal wellbeing, reported in chapter six. The use of the asterisk symbol denotes a term which is defined in the glossary (Appendix A).

These two investigations are connected primarily through the theoretical framework, which will be introduced in section 1.4. Chapters one, two, four, seven and eight also link the two investigations by synthesising the theoretical and empirical evidence. This synthesis is achieved in part by exploring issues such as wellbeing and its association with personal responsibility. The association between wellbeing and personal responsibility is addressed, for example, in debates about agentic versus structural representations of wellbeing. These are discussed in detail in chapters two, seven and eight.

The following paragraphs outline what will be covered in the subsequent chapters.

Chapter one outlines the need to establish wellbeing as a concept distinct from coterminous terms such as quality of life, before providing the background and rationale for the study. These help to establish the ideas and arguments, which this research will subsequently develop. The theoretical framework, research aims and research questions are then outlined. This is followed by sections which detail the national and local context within which the research is situated, and ends with a summary.

Chapter 2 presents a literature review which is structured into two sections. The first utilises a domain approach, which facilitates a greater appreciation of the diverse and complex areas of study involved in wellbeing research. The second

utilises a chronological approach, which facilitates an appreciation of the development of wellbeing research across academic and policy spheres from 2000 to the present day.

Chapter 3 presents the theoretical framework underpinning this study. Critiques of wellbeing are discussed with reference to the theoretical premises. Utilising concepts such as governmentality and the responsibilisation process, this chapter explores and critiques how mechanisms of governance and discourse can be orchestrated to influence the way citizens think, act and conduct themselves. This knowledge is used to help frame an appreciation of the mobilisation of wellbeing within a wider PRA.

Chapter 4 starts with a presentation of the methodological features underpinning this thesis. This incorporates discussion about the epistemological positioning of the researcher and the paradigm and theoretical perspective underpinning this study. It also documents the analytical procedures used to synthesise the historical documentary evidence and the methods used to collect and analyse the empirical data. The chapter then documents the methods for the empirical investigation, discusses whether co-researchers can legitimately be considered 'lay' and outlines the parameters of co-researcher involvement in the research process. It concludes by summarising the reflective process and ethical considerations.

Chapter 5 presents a linguistic/historical contextualisation of wellbeing. The linguistic contextualisation synthesises the conceptual development of wellbeing. The historical analysis documents changing conceptualisations of wellbeing with reference to historical associations between wellbeing and personal responsibility. Collectively they frame the linguistic/historical influences on modern conceptualisations of wellbeing.

Chapter 6 documents the empirical findings of a small qualitative study, which explored lay conceptualisations of personal wellbeing in a sample of older adults with long-term conditions (LTCs).

Chapter 7 provides an overview of the study, reiterating key findings. It contextualises key findings in relation to other theories and similar research, particularly those referenced in the literature review. It recaps some of the existing gaps in knowledge and considers how this study has contributed to addressing these knowledge deficits. In addition, this chapter synthesises the theoretical and empirical findings, highlighting how these contribute to the generation of new knowledge. It draws out some recommendations for both policy and research and ends with a discussion of the strengths and limitations of the study.

Chapter 8 considers the extent to which the aims and objectives of the study have been met, and synthesises the most notable findings. This is followed by a summary of the study's contribution to knowledge. The concluding paragraphs draw out some recommendations for policy and research, consider the wider implications for practice and end with some concluding reflective comments.

# 1.2: Conceptual clarifications

Wellbeing is often conflated with quality of life (QoL), subjective wellbeing (SWB), and happiness (Ryan and Deci, 2001; Camfield and Skevington, 2008; Gasper, 2010). The researcher appreciates their commonalities and draws upon research, which use these terms coterminously. However, within the context of this study the researcher is specifically concerned with investigating wellbeing, and given the propensity for wellbeing to be used interchangeably with 'QoL', 'SWB' and 'happiness' the following section offers an appreciation of the conceptualisations and theoretical similarities and differences between these terms.

QoL and wellbeing are commonly associated with each other because they both have two distinct and identifiable aspects, i.e. objective and subjective dimensions (Rapley, 2003). However, where QoL may be sub-divided into objective and subjective domains, wellbeing has more subdivisions which enables this term to be utilised with more specificity than QoL offers (Camfield and Skevington, 2008; Gasper, 2010). It may be apposite to use this as a basis for appreciating the difference between wellbeing and QoL.

QoL is often used to describe how people feel at a particular stage in life in relation to their current life situation (Bowling, 2001; Bowling et al., 2003). Wellbeing, however, has a strong philosophical aspect (Raz, 2004; Tiberius, 2004, 2011; Sumner, 1999) which helps to distinguish it from conceptualisations of QoL (Camfield and Skevington, 2008; Gasper, 2010; Rapley, 2003). Appendix B explores the conceptual differences between wellbeing and QoL.

Whilst wellbeing and SWB are frequently perceived as synonymous, notions of what constitutes SWB differ from that of wellbeing (Ryan and Deci, 2001).

Wellbeing is perceived as consisting of a number of inter-related domains (Kahneman et al., 1999; Ryff, 1989, 1995; Ryff and Singer, 2000a; Ward et al., 2012). SWB, however, is widely perceived to consist of three components: life satisfaction, the presence of positive mood, and the absence of negative mood (Bell, 2005; Fabiola, 2013; Ryan and Deci, 2001). Until recently, SWB was widely accepted as the primary index of wellbeing (Bell, 2005; King, 2007; Ryan and Deci, 2001). However, in the past decade researchers increasingly differentiate between SWB and personal wellbeing (Atkinson and Joyce, 2011; Fabiola, 2013; Scott, 2012a). The UK Office for National Statistics (ONS) now incorporate SWB within the wider umbrella concept of personal wellbeing to measure the extent to which citizens assess their overall life satisfaction (Oguz et al., 2011).

Wellbeing and happiness have also been perceived as being coterminous (Kingfisher, 2013; McMahon, 2006; Oishi et al., 2015). See chapter five for a discussion of the historical/linguistic differences between wellbeing and happiness.

Variations in conceptualisations of happiness across languages, cultures, and time suggest that happiness can conceptually and linguistically be distinguished from associated terms such as SWB and QoL. One can observe discipline-based distinctions between the concepts of SWB and happiness. Economists such as Lord Layard and Richard Easterlin tend to use SWB as a synonym for happiness (Gough, 2002; King, 2007; Ryan and Deci, 2001). Psychologists, however, perceive happiness as a narrower concept than SWB, which means few in psychology use SWB as a synonym for happiness (King, 2007; Seligman, 2011b; Tiberius, 2004).

The utilisation of wellbeing in research and public policy may be grounded in research in which wellbeing, QoL, SWB, and happiness are used interchangeably. However, they are distinct concepts in respect to their linguistic, historical and conceptual identities (Camfield and Skevington, 2008; Gasper, 2010; Langlois and Anderson, 2002; Ryff, and Singer, 2000a; Sarvimaki, 2006; Sayer, 2011; Scott, 2012c; Spiro and Bosse, 2000). These differences mean that whilst I include pertinent findings garnered from QoL, SWB, and happiness research, the overall focus of this study relates specifically to wellbeing.

The following sections briefly review the research undertaken in this study. This is contextualised in relation to the existing literature base, the gaps in knowledge and questions that remain unanswered. These provide the context and justification for this study.

The theoretical aspect of this study outlined below considers the mobilisation of wellbeing within a PRA and explores the role of historical/linguistic factors in this mobilisation.

# 1.3: Background to theoretical investigation

The influential writer Nikolas Rose was amongst the first in the UK to propose the mobilisation of wellbeing within a PRA in his seminal works (1996, 1999). There is now a well-established theoretical argument that wellbeing has been mobilised within a PRA (Ahmed, 2010; Ferguson, 2007; Graham et al., 2010; Hughes, 1988; Peck, 2013). However, further research is required to investigate how and why this mobilisation has been achieved (Ahmed, 2010; Kingfisher, 2013; Peck, 2013). The theoretical study underpinned by theoretical framework intends to explicate understandings of current efforts to mobilise wellbeing within a PRA.

This thesis argues that wellbeing has been mobilised within a PRA for governance purposes. This argument will be set out in chapter three.

The PRA is a defining feature of western neo-liberalism (Ferguson, 2007; Micklethwait and Woolridge, 2003; Peck, 2013). The PRA refers to efforts utilised

within neoliberalism to facilitate citizens' adoption of responsibility for aspects of living which may formerly have been perceived to be the responsibility of governments\* or the state\*(Ferguson, 2007; Micklethwait and Woolridge, 2003; Peck, 2013). It is important to note that "The terms the state and government can be confounded but whilst governments can be removed the state with its rights and obligations remain" (Robinson, 2013:557).

The PRA is part of a 'responsibilisation' process (Micklethwait and Woolridge, 2003; Shamir, 2008a). The responsibilisation process refers to a pervasive influence which encourages individuals to take personal responsibility for their actions and behaviours (Ahmed, 2010; Micklethwait and Woolridge, 2003). The PRA and responsibilisation process instil values in line with governing socio-political ethos and are rooted in the prevailing cultural and historical positioning of individualistic motivation for action (Micklethwait and Woolridge, 2003; Shamir, 2008b). These have been utilised in the mobilisation of wellbeing, particularly in its association with health (Fullager, 2002, 2009; Seedhouse, 1995; Sointu, 2005). A consideration of the mobilisation of wellbeing within a PRA, contextualised by the notion of governmentality and governance will be addressed in chapter three.

#### 1.4: Rationale for theoretical investigation

Having outlined the theoretical arguments for 'why' and 'how' wellbeing has been mobilised within a PRA the study also investigates the linguistic and historical factors, which facilitate this mobilisation. This thesis argues that historical associations between wellbeing and personal responsibility and the linguistic status and legacy of wellbeing have been utilised in order to facilitate the mobilisation of wellbeing within a wider PRA. This aspect of wellbeing research is important because "It is only by reflecting on historical and cultural contexts of our categories of analysis, investigation and action that we can gain insight into how our work relates to forms of governance and the interests that they serve" (Kingfisher, 2013:79). Historical factors play a powerful role in wellbeing (Inglehart and Klingemann, 1998; Kingfisher, 2013), but further research is required "To unpack the historical and political forces at work in positing wellbeing as valid, reliable and

universal" (Crawshaw, 2008:259). The widely cited study by Inglehart and Klingemann (1998) identified that historic factors receive insufficient attention in the majority of wellbeing research because they are complex and difficult to disentangle. To date, little research has investigated the role of historical and linguistic factors in the mobilisation of wellbeing (Hughes, 1988; Kingfisher, 2013; McMahon, 2006). There also remains an unanswered question of how historical and linguistic aspects of wellbeing have influenced conceptualisations of wellbeing (Kingfisher, 2013; McMahon, 2006; Oishi et al., 2015).

The limited research, which has explored wellbeing within historical contexts, suggests wellbeing is a high status word historically and linguistically aligned with health and happiness, which are perceived to be personally controlled by individuals (Bergdolt, 2008; McMahon, 2006). This makes it a useful concept for the maintenance of neo-liberalism (Hughes, 1988; Rose, 1999). It has been posited that the high linguistic status accorded to wellbeing has facilitated its appropriation within professional and academic discourse whilst simultaneously undermining its transfer to lay usage (Hughes, 1988; Mathews and Izquierdo, 2009; Seedhouse, 1995). Within the last decade we have begun to witness an upward trajectory of wellbeing within academic, media and political discourse (Bache and Reardon, 2013; Scott, 2012c). One may reflect whether this trajectory can be witnessed within lay narratives (Eraurt and Whiting, 2008; Mathews and Izquierdo, 2009; Scott, 2012c; Seedhouse, 1995). Further research is required to investigate the relationship between professional wellbeing discourse and lay wellbeing narratives. This, however, requires more attention than this thesis can accommodate and is worthy of a thesis in its own right.

Historical and linguistic changes in concepts are particularly important for research, which aims to examine subjective states such as wellbeing (Mogilner et al., 2011; Oishi, 2012; Oishi et al., 2015). Further research is required "To situate wellbeing/happiness in historical context as there is nothing in our current conceptualisations or practices that are transparently self evident" (Kingfisher (2013:79).

Chapter five will explore the role of historical and linguistic factors in the mobilisation of wellbeing and consider how linguistic and historical changes in conceptualisations of wellbeing have influenced modern conceptualisations of wellbeing. This is achieved by synthesising the linguistic development of wellbeing and analysing the historical documentary evidence of wellbeing conceptually.

Having established the background and rationale for the theoretical study, the following sections outline the context and justification for the empirical aspect of this study. The empirical aspect of this study investigates conceptualisations of wellbeing in older adults with LTCs.

## 1.5: Background for empirical investigation

The dawn of the 21<sup>st</sup> century heralded a surge of interest in conjoining the concepts of health and wellbeing (Scriven et al., 2005; Seedhouse, 1995, 2006; Knight and McNaught, 2011). The concept of health increasingly came to be understood in relation to the salutogenic model of health (Antonovsky, 1996). This focuses on the notions of positive health which incorporate notions of wellbeing (Knight and McNaught, 2011; Prilleltensky and Prilleltensky, 2007). This paradigmatic shift in conceptualisations of health encouraged the adoption of wellbeing into health policy and legislation (Knight and McNaught, 2011). Since then, wellbeing has become an important and influential element in the language of public health and health promotion (Carlisle and Hanlon, 2007, 2008; Chavez et al., 2005; Seedhouse, 1995, 2006). However, "Although wellbeing has seemingly effortlessly entered in to recent public health parlance little critical attention has been paid to its real meaning or purpose" (Crawshaw, 2008:259).

A growing number of critics challenged the increasingly dominant rhetoric and discourse which popularised the role of wellbeing (Edwards and Imrie, 2008; Furedi, 2006; Ferguson, 2007). This stimulated intellectual debates such as the Economic and Social Research Council (ESRC) series of academic wellbeing seminars (2001, 2002), covered in detail in chapter two. These seminars explored a range of contested points such as dominant conceptualisations and representations of

wellbeing and discursive practices, which promote particular conceptualisations, and representations of wellbeing.

Within the context of health and wellbeing, Fisher (2008) argued "The discourses of governance and of governmentality restrict definitions of authentic wellbeing and empowerment to those consistent with the neoliberalism of the dominant symbolic order" (p.583). Fisher (2008) also questioned the mobilisation of neoliberal representations of wellbeing, which mean lay conceptualisations of wellbeing "Are being undermined by their contact with the health and social care services where they are confronted with oppressive frameworks of meaning" (p.583).

Schwanen and Ziegler (2011) added to this, noting that academics are complicit in neoliberal practices of framing\* wellbeing "As personal responsibility which helps to produce "good" older people who look after themselves and their wellbeing in ways that place a rather limited burden on existing and future healthcare systems and national government budgets" (p.723).

Academic wellbeing research may also reinforce the language and practices which hold older adults responsible for the conditions of life, which should be the responsibility of the state (Schwanen and Ziegler 2011). It is important therefore for academics "To provide nuanced and context specific accounts of wellbeing …to open up alternative conceptualisations of wellbeing" (Schwanen and Ziegler, 2011:723).

In addition to theoretical debates, the introduction of wellbeing powers for local government in the Local Government Act (LGA, 2000) encouraged debates about the practical application of wellbeing (Seaford, 2011). This led to a growing appreciation that qualitative research has an important role to play in explicating understandings of how people conceptualise wellbeing (Cooper, 2014; Diener et al., 1999; Huppert, 2013).

The Local Government Act (2000) was also important for stimulating interest in wellbeing beyond academic remits. Increasingly wellbeing was perceived as a practical, usable tool by policymakers and planners (Scott, 2012c). Professionals

working in public health, regeneration, community and sustainability fields increasingly adopted wellbeing as a malleable concept, which could be utilised to modify behaviour and inculcate a sense of personal responsibility (Seedhouse, 1995; Woolrych and Sixsmith, 2009).

There was growing research interest in the practical application of wellbeing and the role of conceptualisations of wellbeing at a theoretical level. Conceptualising the wellbeing of older adults has been undertaken through theoretical approaches; however, these primarily prioritise academic conceptualisations of wellbeing (Schwanen and Ziegler, 2011; Zaidi, 2008). This trend has been countered in part by Diener (1984, 1994, 2009) and Diener et al. (2009), who studied wellbeing as a subjective experience in the United States (US) and quality of life (QoL) research in the United Kingdom (UK). Research by Bowling (2005), Bowling and Gabriel (2007), Gabriel and Bowling (2004), Gilroy (2007, 2008) and Wilhemson et al. (2005) focused on lay understandings of the good life or QoL. These studies contribute to cross-discipline insight into lay understandings of the good life (Gilroy, 2007, 2008). However, at an empirical level little research has been undertaken to explore lay conceptualisations of wellbeing and how these could be embedded within wellbeing policies and practices (Schwanen and Ziegler, 2011; Zaidi, 2008).

Schwanen has written widely around the subjective experiences of wellbeing and explored different understandings and conceptualisations of wellbeing (Nordbakke and Schwanen, 2014; Schwanen and Paez, 2010; Schwanen and Wang, 2014; Schwanen and Ziegler, 2011). This is important given that research suggests there is a disjunction between lay and professional/academic conceptualisations of wellbeing (Hobbs and Sixsmith, 2010; Woolryche and Sixsmith, 2008). Whilst multiple conceptualisations of wellbeing proliferate, little research has been undertaken to represent lay conceptualisations of wellbeing (Ball et al., 2004; Nordbakke and Schwanen, 2014; Schwanen and Ziegler, 2011). This research attempts to begin to redress this balance.

Wellbeing is a central focus of international and national policy (Carlisle et al., 2009; Schwanen and Ziegler, 2011; Ziegler and Schwanen, 2011). Indeed, the United

Nations General Assembly adopted a resolution that "Invites member states to pursue the elaboration of additional measures that better capture the importance of the pursuit of happiness and well-being in development with a view to guiding their public policies" (Wright, 2014:799).

It is important therefore that lay conceptualisations of wellbeing are represented. This will help ensure aspects of wellbeing identified in lay conceptualisations are included in wellbeing interventions and policies. In order to facilitate this, empirical research is required to investigate lay conceptualisations of wellbeing (Abrahamson and Rubin, 2012; Atkinson and Joyce, 2011; Cooper, 2014; Seaford, 2011).

# 1.6: Rationale for empirical investigation

This PhD originates from a two-year research project in Manchester, North-West England, in which the researcher was employed as a research associate. The project 'Changing services for people with long-term neurological conditions: Promoting well-being within service provision' (2007–2009) was jointly commissioned by Manchester Metropolitan University (MMU), Manchester City Council and Manchester Primary Care Trust. The project explored ways in which the transformation of service provision for those with long-term neurological conditions might enhance health and wellbeing.

Preliminary findings indicated that wellbeing was not assimilated into participants' narratives, primarily because participants struggled to conceptualise wellbeing. When participants did offer conceptualisations of wellbeing these appeared to differ from professional and academic conceptualisations of wellbeing. In the participants' conceptualisations, they rarely appeared to associate wellbeing with personal responsibility. Conversely, wellbeing was primarily conceptualised as the responsibility of local government and/or statutory bodies\*. This suggested further research was required to explore lay conceptualisations of wellbeing.

Given that the participants primarily conceptualised wellbeing as the responsibility of local government and/or statutory bodies, research was also required to explore

whether wellbeing was a meaningful concept and the extent to which personal responsibility was associated with wellbeing.

In addition to the rationale which the original project provided for this research, wider considerations supported the justification for this empirical investigation.

First, at a theoretical level research has also identified a need for a clearer understanding of the extent to which wellbeing is associated with personal responsibility and government responsibility. This has primarily been addressed through considerations of how structural\* and agentic\* factors impact on wellbeing (Bache and Reardon, 2013; Seaford, 2011; Scott, 2012a; Scott and Bell 2013). The balance between agentic versus structural level approaches to enhancing wellbeing has become a key issue in wellbeing research (Cooper, 2014; Seaford, 2011; Scott, 2012a, 2012c). This is discussed in detail in chapter two. This thesis therefore also seeks to provide an appreciation of the role of the individual versus structural determinants of wellbeing. This will be pursued in relation to associations between wellbeing and personal responsibility.

Second, as outlined in previous sections there is strong theoretical evidence that wellbeing had been mobilised within a PRA. However, to date there is little empirical research to validate this. This research attempts to address this gap in knowledge by investigating the extent to which study participants associate wellbeing within the context of a PRA. Research suggests where there is a divergence between lay and professional conceptualisations, professional conceptualisations are not easily assimilated into lay perspectives (Woolryche and Sixsmith, 2008; Zaidi, 2008). Given the theoretical interest in the mobilisation of wellbeing, empirical research is required to explore whether this mobilisation has been assimilated into lay conceptualisations of wellbeing (Seedhouse, 1995; Schwanen and Ziegler, 2011).

Third, little has been done to operationalise wellbeing (Allen, 2008; Barnes et al., 2013). Some have recognised the need for government leadership to prioritise

older adults' wellbeing in order to address the social and psychological implications of LTCs (Allen, 2008; Office of the Deputy Prime Minister, 2006a). These implications include issues such as social isolation, depression and loneliness (Barnes et al., 2013; Coulter et al., 2013; Office of the Deputy Prime Minister, 2006a). However, given that wellbeing has been enshrined in law as a duty of local authorities (LGA, 2000; Amended Act, 2007) it is important to explore those conditions, aspects and features of human existence which enhance or diminish personal wellbeing from the perspective of older adults with LTCs (Barnes et al., 2013; Ward et al., 2012).

This is therefore a timely and important piece of research, which intends to complement existing quantitative research in order to support the provision of wellbeing services for older adults with LTCs.

#### 1.7: Theoretical framework

The three premises in theoretical framework were constructed to investigate the mobilisation of wellbeing within a PRA. The premises propose wellbeing is mobilised to:

- To obligate/persuade citizens to accept responsibility for aspects of wellbeing which may be beyond their control.
- Facilitate the reduction of government expenditure on health and welfare bills to an ageing population.
- Act as a socio-political tool concerned with approving and validating particular behaviours and lifestyles choices.

### 1.7.1: Study aims

1- The theoretical component of this thesis aims to explicate understandings of wellbeing and its association with personal responsibility. The theoretical component of this thesis is comprised of two parts. The first part synthesises theoretical evidence that wellbeing has been appropriated within a PRA. The

second part contextualises the linguistic/historical legacy of wellbeing and its association with personal responsibility. Taken together, these seek to explicate understandings of current efforts to mobilise wellbeing within a PRA.

- 2- The empirical component of this thesis aims to explicate understandings of lay conceptualisations of personal wellbeing. This will contribute to advancing knowledge about how lay people conceptualise wellbeing and the degree to which they associate wellbeing with personal responsibility. The understandings generated by this research could be utilised to facilitate the promotion of wellbeing policies, which are inclusive of and reflect lay conceptualisations of what is important for personal wellbeing.
- 3- The overarching aim of this thesis is the synthesis of the theoretical evidence with the empirical evidence. The theoretical framework and the research questions link the theoretical and empirical parts of the thesis by exploring the associations between personal responsibility and conceptualisations of wellbeing. This is undertaken through the prism of a historical/linguistic contextualisation of wellbeing. This synthesis may help to explicate understandings of modern conceptualisations of wellbeing and its usage, particularly within the UK healthcare and welfare systems.

### 1.7.2: Research questions

The empirical study examined the following research questions to explore lay conceptualisations of personal wellbeing among a sample of 22 older adults with LTCs aged 50 plus living in the community, residential care and sheltered accommodation.

The four research questions specifically addressed in this thesis are:

- Q1 Have historical/linguistic factors influenced current conceptualisations of wellbeing?
- Q2 Do older adults with LTCs associate wellbeing with personal responsibility?
- Q3 How do older adults with LTCs conceptualise personal wellbeing?

Q4 - Is wellbeing considered a meaningful concept in older adults with LTCs?

# 1.8: Scope of the empirical study

The empirical investigation is a substantive qualitative account of personal wellbeing amongst older adults with LTCs, which perforce means it has limited scope for wider generalisability. The investigation consisted of a small sample of twenty individuals and two co-researchers. Semi-structured interviews elicited participants' thoughts and opinions and thematic analysis was used to unpack their data.

It is important to acknowledge this is not a cross-cultural or universal study, but rather an exploration of personal wellbeing within specific and defined parameters. The study parameters were restricted to older adults with LTCs because of the rise of ageing population in conjunction with a marked increase in numbers living with LTCs not only in the UK but also in many westernised developed countries. This places considerable strain on government expenditure, particularly health and welfare (Figueras and McKee, 2012). These considerations will be discussed in detail in the following sections.

#### 1.9: Rationale for study parameters

# 1.9.1: Increasing proportion of UK population with LTCs

The term long-term condition (LTCs) used extensively throughout this thesis is understood within the context of the definition provided by the Department of Health (DoH) (2012b). An LTC is defined as "A condition that cannot, at present, be cured but is controlled by medication and/or other treatment/therapies" (DoH/Long Term Conditions, 2012b:3).

The prevalence of LTCs rises with age, affecting about 50 percent of people aged 50, and 80 percent of those aged 65. Many older people have coterminous conditions. It is important to note that in absolute terms "There are more people with LTCs under the age of 65 than in older age groups" (Coulter et al., 2013:3). This

factor influenced the choice of age parameters for this thesis, which will be discussed further in the following pages.

Evidence indicates there has been a sharp rise in the numbers of UK citizens with LTCs (Coulter et al., 2013; Cracknel, 2010; NHS England, 2013). Many of these, such as diabetes, liver disease, and obesity are believed to originate from behaviour and lifestyle choices (Coulter et al., 2013; Figueras and McKee, 2012; NHS England, 2013). LTCs are the most common cause of death and disability in England (Coulter et al., 2013; DoH, 2012b; Hibbard and Greene, 2013). More than 15 million people in England have LTCs for which there is no cure (Coulter et al., 2013; DoH, 2012b). There are also increasing numbers of citizens with congenital LTCs such as cerebral palsy who are living longer as a result of improvements in managing and treating these conditions (Figueras and McKee, 2012; NHS England, 2013).

Nationally, the number of people living with LTCs is also forecast to rise sharply over the next two decades (Coulter et al., 2013; Cracknel, 2010; NHS England, 2013; Office for National Statistics (ONS), 2013). The increasing number of citizens with LTCs has been identified as a significant drain on national expenditure, adding considerable strain to health and welfare systems (Cracknel, 2010; Coulter et al., 2013; ONS, 2013a). The DoH (2012a) has tried to address this by promoting a WBA, which they suggest, "Is associated with positive health behaviours in adults and may ultimately reduce the healthcare burden" (p.3).

The National Service Framework for Long Term (Neurological) Conditions has a remit which sets local government targets for improving services for service users with any LTCs (DoH, 2005b). The National Service Framework launched a funding programme to explore improved services for people with LTCs. However, the framework predominantly funded quantitative research predicated upon the need to gather a strong, numerical evidence base (Cracknel, 2010; Hobbs and Sixsmith, 2010; Sixsmith et al., 2013). Much of this work was grounded in the medical model of disability, taking little account of the social and psychological implications of LTCs and the experience of health and social care services (Sixsmith, 2012; Sixsmith et al., 2013). There existed therefore a need for qualitative research, which considered

the wider implications of the social and psychological impact of living with LTCs. This research attempts to address this gap in knowledge.

# 1.9.2: Ageing populations

The life course characteristics of different age groups have become increasingly important considerations in the study of wellbeing (Cooper, 2014; King, 2007; Marks, 2004). The number of older adults in England is rising and predicted to rise further (ONS, 2013a). Cracknel's (2010) projections predicted that there would be 5½ million older adults in 20 years. A significant proportion of public expenditure relates to welfare benefits for older adults (Cracknel, 2010). This represents a significant economic cost to the NHS. Cracknel (2010) claimed, "The welfare system and the NHS accounted for nearly half of UK government expenditure in 2009/10" (p. 45). With the majority of welfare expenditure dedicated to older adults, the growing proportion of older adults represents a significant challenge to the UK government (Cracknel, 2010; DoH, 2014; ONS, 2009c, 2013b).

The rise of ageing populations, survival rates in people with life limiting conditions and long-term preventable health conditions indicates a clear and pressing need for research, which explores the wellbeing requirements of older adults with LTCs (Barnes et al., 2013; Cooper, 2014; Ward et al., 2012). It is to be hoped that this research can facilitate the adoption of wellbeing approaches, which enhance wellbeing amongst older adults with LTCs and simultaneously "Reduce the healthcare burden" (DoH, 2012a:3).

#### 1.9.3: Age parameters

There is no agreed definition of 'older adults' (Cai et al., 2008; Higgs, 2008; Katz 2012; Ononiwu, 2013; Public Health England, 2013). "Modern chronological and generational boundaries that had set apart childhood, middle age, and old age have become blurred and indeterminate" (Katz, 2012:6).

Categorising older adults is complex and relies on a range of factors such as the positionality of the researcher. What constitutes the start of old age remains contested (Higgs, 2008). The definition of older adults has traditionally been based

on the chronological age in which their contribution to economic activity ends (WHO, 2010). This enables them to proceed into retirement, and become eligible to receive benefits associated with pensioner status (WHO, 2010). Retirement, however, "Is no longer the clear entry point to 'old age' that it once was and, therefore, is anachronistic as a definition of who 'older people' are" (Walker, 2003:6).

Some argue that 50–65 years represents middle not old age (Cronin et al., 2010). Indeed, there is still a strong perception that 65 marks the start of old age (Age Concern, 2006; Craig and Mindell, 2007; Lawton and Salthouse, 1998; Wiggins et al., 2008).

Numerous organisations, however, use 50 years of age as the starting point when defining older adults. ONS research defines older people as aged 50+ unless otherwise declared (ONS, 2012f). Other examples include, The European Commission's 'Quality of life in old age in Europe' (Walker, 2003), The Health Development Agency (2004) and The English Longitudinal Study of Ageing. There has also been an increasing tendency to identify early old age as encompassing individuals aged between 50 and 65 years, i.e. the 'young—old' (Blane et al., 2007a; Siegrist and Wahrendorf, 2010; Vanhoutte, 2012; Whitbourne, 2016). Given the empirical support for 50 as the starting point when defining older adults and the theoretical support for older adults to encompasses individuals aged from 50 years up to 75 years (Blane et al., 2007a; Higgs et al., 2003; Vanhoutte, 2012; Whitbourne, 2016), these mark the lower and upper age limits parameters in this thesis.

#### 1.9.4: Local context

The ONS classified Manchester as the third largest city in England, with a population of 503,000 (ONS, 2009i). It has the third most culturally diverse population in the UK (ONS, 2009i) and high levels of deprivation alongside pockets of affluence (Department of Communities and Local Government, 2010). Manchester has a significantly higher than national average incidence of people with the most dangerous LTCs, i.e. cardiovascular disease, stroke and coronary obstructive

pulmonary disease (DoH, 2012b). It also has a higher than national average rate of people who die from preventable LTCs, linked to lifestyle choices and risky behaviour such as smoking, drinking and drug taking (Public Health England, 2013). Manchester also has a higher than national average rate of early mortality, i.e. under 75 years (ONS, 2009c).

These contextual factors provided the rationale for situating this research within the geographical boundaries of Manchester, England. Manchester is not, however, so different from the rest of UK that it should be considered a special case. However, the high number of deaths from preventable conditions and people with LTCs in Manchester means that statutory agencies in this region are likely to seek to undertake activities and/or interventions which address these considerations (Iddons, 2009). This research aims to explore conceptualisations of wellbeing in older adults with LTCs in order to identify what they consider is most important for wellbeing. This is intended to support the provision of evidence-based wellbeing services for older adults with LTCs in Manchester.

#### **1.10: Summary**

This chapter illustrated how this thesis is structured, gave a précis of the thesis chapters, outlined the background and rationale for this thesis, reviewed the key issues requiring research attention and summarised the importance of this research. Additionally, it introduced the theoretical framework along with the study aims and the research questions. Before moving on to chapter two it is important to briefly review how the following chapters will attempt to answer the gaps in knowledge and unanswered questions identified in the introduction. If we are to understand how and why wellbeing has been mobilised within a PRA we need to:

- a) Consider how mechanisms of governance and discourse can be orchestrated to influence how citizens think, act and conduct themselves. This will be discussed in chapter three.
- b) Explore the historical/linguistic associations between wellbeing and personal responsibility. This will be discussed in chapter five.

c) Consider how older adults with LTCs conceptualise wellbeing within lay frames of reference. This will be addressed in chapter six.

### **Chapter 2: Literature Review**

#### 2.1: Literature search methods

Appendix C contains a detailed account of the process and procedures underpinning the literature search which commenced in 2008 and terminated in 2015. This literature review adopted a selective approach which focuses on studies of particular relevance to the researcher's work (Gasper, 2010), drawing from purposively selected literature in keeping with the parameters of this thesis.

The literature search was undertaken through four stages. The first stage undertaken prior to data collection identified potential references using a range of search strategies and inclusion/exclusion criteria. These were detailed searches, which aimed to provide a general overview of the evidence base rather than a specific literature review. The second stage was undertaken throughout the writingup process in order to ensure the literature review held current currency (Polit and Beck, 2006). This stage was based on retrieving references which had been identified in stage one and reading the abstract to ascertain relevance to the study parameters. The third stage was undertaken in order to ensure the review was as comprehensive as possible. It involved using references retained in stage two as the basis for the literature review. Searches were conducted via electronic database searches, web searches, web sites of key organisations and research organisations, journal searches, bibliographies and grey material. A fourth and final additional stage was undertaken to ensure the review was current and was as comprehensive as possible. This contained the original search parameters but went beyond the specification of the original literature overview.

Searches included terms such as 'personal', 'subjective' and 'eudaimonic wellbeing', and components of wellbeing such as 'economic' and 'psychological wellbeing'.

Synonymous or closely related terms such as 'quality of life', 'life satisfaction', 'happiness', and 'welfare' were also included. Additional searches combined wellbeing with 'health' 'age' 'history' and 'linguistics'.

# 2.2: Domain approach to wellbeing

The first part of this literature review utilised a 'Domain Approach' to wellbeing, summarising domains widely perceived to be wellbeing's principal components. This approach facilitates researchers' ability to analyse specific components of wellbeing, providing a means of exploring one aspect of wellbeing in relation to another. The domain approach primarily explores the current state of knowledge for the empirical study. Some question the efficacy of utilising this approach because these tend to result in research being conducted in silo, failing to reflect the holistic nature of wellbeing/QoL (Atkinson 2013; Atkinson and Joyce, 2011; Atkinson et al., 2012). The second part of the literature review utilised a 'Chronological Approach' to explore the current state of knowledge for the theoretical study. This approach "Moves towards integrating the diverse domains and dimensions of wellbeing through a relational and situated account of wellbeing" (Atkinson 2013:138). The two approaches complement each other, providing a broad, holistic overview of the development of wellbeing research.

There are innumerable understandings of wellbeing; however, most agree wellbeing involves an experiential understanding of what is important in the domains of everyday life (Diener et al., 2003; DoH, 2014; Ryan and Deci, 2001). Concurrently, many argue that individuals structure their cognitive, affective experiences and memories by life domains (Clark, 2002b; Clark and Gough, 2005; Durayappah, 2010; King, 2007; Lewin, 1951; Sirgy, 2002). Wellbeing is increasingly envisaged as incorporating a number of components organised around central human needs (Abdallah and Shah, 2012; Diener et al., 2003; Ryan and Deci, 2001; Sirgy, 2002).

Most wellbeing researchers agree that the dimensions of wellbeing are not mutually exclusive domains but inter-related components of a holistic whole (Diener et al., 2003; Ryan and Deci, 2001; Ryff, 1989, 1993; Ward et al., 2012). The following domain approach facilitates an appreciation of the diverse and complex areas of study involved in wellbeing research.

# 2.3: The ageing process and wellbeing

The ageing process represents a significant challenge to the personal wellbeing of older adults (Demakakos et al., 2010; Seedhouse, 2001; Steptoe et al., 2012). There is a general consensus that psychological wellbeing and functioning are particularly affected by the ageing process (Barnes et al., 2013; Ryff, 1989b, 1995; Ryff and Keyes, 1995; Ward et al., 2012). Early empirical research undertaken by Ryff (1989b) and Ryff and Keyes (1995) indicated that psychological wellbeing has an age-differentiated profile. Ryff (1989b) and Ryff and Keyes (1995) reported that increasing age is associated with higher levels of depression, and a reduction of 'purpose in life' and 'personal growth'. These results have been validated by the work of others, revealing that "Psychological functioning has a highly differentiated profile across the adult life cycle" (Warr, 2007:64). Differential ageing\* has been identified as a key factor in the degree to which age may impact on wellbeing. Schuz et al. (2009) identified that ageing has a smaller impact on wellbeing for older adults who remain physically and mentally healthy than for those whose health declines.

There is increasing evidence to suggest the ageing process leads to concomitant health conditions (Bernard and Scharf, 2007; Clarke and Warren, 2007; ONS, 2012, 2013). The association between concomitant health conditions and the ageing process has been widely identified as detrimental to wellbeing (Fortin et al., 2006; King, 2007; Seedhouse, 2001; Sixsmith, 2012). Laux et al. (2008) argued that agerelated health clustering impacts significantly on the ability to manage life circumstances in relation to wellbeing. Coping mechanisms are an important way to offset the impact of ageing and concomitant health issues on wellbeing (Biesecker and Erby, 2008). Employing coping mechanisms to deal with the ageing process and health conditions is more strongly associated with wellbeing in people with concomitant or serious illnesses (Schuz et al., 2009). Older people with health issues are more likely to use coping mechanisms to cope with loss, disappointment and negative states of mind, thereby aiding emotional wellbeing (Carstensen, 2009).

Some argue ageing *per se* is not necessarily associated with negative wellbeing (Gasper, 2007b; King, 2007; Myers and Diener, 1995). Carstensen (2009) reported that greater awareness of mortality encouraged older people to actively inculcate wellbeing by living in the moment. The importance of living in the moment has led to suggestions that temporal perspectives may be important factors in lay perceptions of ageing and wellbeing (Clarke and Warren, 2007; Durayappah, 2010). Some argue that "Average happiness is remarkably stable over the lifespan" (Layard, 2005:33). However, over 40 years of data suggests age and wellbeing has a U-shaped curve (Allen, 2008; Diener et al., 2009; Dolan et al., 2008; Easterlin, 2000; King, 2007; Oguz et al., 2013). A number of theories have been proposed to account for this well observed phenomenon.

Some suggest differing levels of wellbeing over a life-course trajectory may be the result of internal physiological changes rather than external circumstances (Ebstein, 1996; Roe et al., 2010). Others argue age-related changes in wellbeing are more likely to be the result of psychological changes (Bell, 2005; Demakakos et al., 2010; Steptoe et al., 2012); whilst others claim the decline in wellbeing during the middle age and the subsequent rise later in life represents what has been called the "Death of ambition and birth of acceptance" (The Economist, 2010).

In recent years, 'active ageing' has been increasingly associated with wellbeing. Stenner et al. (2011) argue this is because it "Will enable the expanding older population to remain healthy (thus reducing the burden on health and social care systems) and to stay in employment longer (thus reducing pension costs)" (p.467). Bowling (2008) surveyed 300 people aged 65+ in the UK, reporting that active ageing was primarily associated with physical health and functioning (43%), leisure/social activities (34%), mental health (18%) and social relationships/connections (15%).

The active ageing agenda has been criticised in recent years for encouraging an idealisation of active and successful ageing which relies on normative standards that many older people cannot attain (Barnes et al., 2013; Holstein and Minkler, 2007; Stenner et al., 2011; Ward et al., 2012). Active ageing has been perceived as a concerted effort for "Increasing quality and quantity of life years, enhancing

autonomy and independence, and reducing health and care system costs" (Stenner et al., 2011:468).

The active ageing agenda has identified that physical, mental and social activities are all important aspects of the ageing process. Further research is required to explore interactions between temporality, ageing and wellbeing; however, the ageing process and resultant health conditions have been widely identified as detrimental to wellbeing. The following section explores the impact of health on wellbeing.

#### 2.4: Health

Extensive research has explored the relationship between health and wellbeing and documents a well-established association between health status and wellbeing (Ballas et al., 2007; Okun et al., 1984; DoH, 2012b; Prilleltensky and Prilleltensky, 2007; Sixsmith and Boneham, 2007). Health is often perceived as "One of the key determinants of SWB" (Schuz et al., 2009:23). Research reports that people frequently cite health as an important contributor to wellbeing (Barnes et al., 2012; Cooper, 2014; Diener et al., 2008; Layard, 2005; Shah 2005; Schuz et al., 2009; Ward et al., 2013). Adaptation theories for example have been used to investigate the degree to which changes in health impact on wellbeing.

Adaptation to changes within physical health and the subsequent impact on wellbeing is well documented (Cohen and Biesecker, 2010; Gasper, 2010; Schuz et al., 2009; Schwartz et al., 2006). Diener et al. (2008), for example, explored how the emergence of a health condition may affect SWB temporarily. They reported that levels of SWB tend to revert overall to pre-condition levels of SWB. This phenomenon known as 'Set-point Theory' has been widely reported and is discussed in more detail in chapter six.

### 2.4.1: Adaptation

Theories of adaptation are chiefly concerned with the degree to which people can cope with illness and can adjust or modify their aims and objectives to suit their changing life circumstances or new health status (Diener et al., 2008). In the context of ill health and/or disability, adaptation is "The process of coming to terms with the implications of a health threat and the observable outcomes of that process" (Biesecker and Erby, 2008:400). Diener et al. (2008) argued that an individual's ability to adapt has an important bearing on SWB. Little qualitative research has been undertaken in the UK to explore how adaptation to ill health and coping mechanisms affects personal wellbeing (Hobbs and Sixsmith, 2010; Kahneman and Deaton, 2010; Layard 2005; Lazarus, 1991).

Whilst adaptation to health issues and impact on personal wellbeing requires further research, much can be learned from theories of adaptation, which have been widely researched in QoL research. Cohen and Biesecker (2010) proposed that theories of adaptation are extremely useful for framing understandings of QoL as a quantifiable outcome of the adaptation process. Coping mechanisms play an important role within adaptation theory (Abbot et al., 2008). Some coping mechanisms, however, appear to be more effective than others are. Avoidance, distraction, and disengagement techniques were associated with lower QoL in some studies (Abbot et al., 2008; Brown et al., 2000; Lazarus, 1991). Conversely, acceptance, optimism, and hopefulness were associated with higher QoL (Berglund et al., 2003). Other studies have identified social support as the most effective mechanism in helping people cope and adapt to ill health (Helgeson 2003; Schwartz and Frohner, 2005).

Phillips (2006) identified that self-reported QoL could appear to increase after disablement or diagnosis of terminal illness. Adaptive preference theories seek to clarify this seemingly unexpected finding by suggesting that some individuals adopt evasive adaptive action. This refers to people trivialising or ignoring their disability and/or debility or focusing on the important things in life which remain or which they retain control over (Gasper, 2007b; Lazarus, 1991; Schwartz et al., 2006). See

Sen (1979, 1987, 1993, and 2002) for a comprehensive critique of adaptive preference theories.

## 2.4.2: Biographical disruption

Theories of biographical disruption review adaptive theory by arguing that adaption theories often view disability, debility, and adaptive processes as a relatively stable and linear occurrence (Bury, 1982). The concept of biographical disruption was introduced by Bury (1982) as a means to understand experiences of chronic ill health and how people respond and adapt to chronic ill health (Hubbard et al., 2010). Since Bury's (1982, 2001) research indicates that biographical disruption may make the transition from ill health to managing one's health condition particularly difficult for those who experience sudden onset chronic ill health, impairment\* or disability\* (Hobbs and Sixsmith, 2010; Hubbard et al., 2010), being able to adapt to a changing daily life routine in this context is more challenging for those with concomitant psychological issues and may have a particularly detrimental impact on wellbeing (Hobbs and Sixsmith, 2010).

Since Bury's (1982) theory of chronic ill health as biographical disruption, criticism of biographical disruption has increased, especially from disability studies (Williams, 2000). Within disability studies there has been an increasing call for the perception of chronic health and biographical disruption to be understood within the social model of disability\* (Goodley and Lawthom, 2006; Williams, 2000). The relevance of body image, for example, has also been identified as an important factor within lay perceptions of chronic health (Charmaz, 1995; Goodley and Lawthom, 2006). Disability, impairment and ill health often involve fluctuating symptoms, deterioration and improvement over periods of time (Hobbs and Sixsmith, 2010; Barnes et al., 2012; Taylor, 1977; Ward et al., 2013). These fluctuations may undermine the usefulness of adaptation and biographical disruption theories to understanding the wellbeing of people in ill health (Hubbard et al., 2010; Reeve et al., 2010).

The large number of theories, which propound to help explain the relationship between health and wellbeing highlights the challenges wellbeing researchers face in explicating understandings of the highly complex relationship between wellbeing and changes in health. However, further wellbeing research on the role of coping mechanisms within adaption theories may offer a greater appreciation of how changes in health affect wellbeing.

### 2.4.3: Self-reported health

The ONS (2012c, 2012f, and 2013c) highlighted a clear and positive relationship between self-reported health and personal wellbeing. The ONS (2011a) used regression analysis to explore the size and strength of the relationship between self-reported health and personal wellbeing. The findings indicated that the size of the relationship between self-reported health and personal wellbeing is larger than any other variable included in the model. One important caveat for this data is that it is not known how the subjective and objective interaction between health and wellbeing operates, a matter subject to much discussion (Adams, 2011; Allen, 2008; Fortin, 2006, 2010). However, self-reported health aligns more to personal wellbeing than objective measures of health (Adams, 2011, ONS, 2013c). Dolan et al. (2008) explored subjective aspects of health and wellbeing and argued that people who consider themselves subjectively to be in good physical and mental health may be objectively happier. However, the direction of causality remains unclear; for example, happier people may be less likely to report feeling ill than unhappy people (Dolan et al., 2008). Further research is required to explore the extent to which the relationship between personal wellbeing and health is the result of personality characteristics rather than objective health conditions (Dolan et al., 2008; ONS, 2011a).

### 2.4.4: Disability and impairment

The ONS reviewed research and reported clear evidence of a relationship between self-reported disability and personal wellbeing (ONS, 2011a). People who reported a disability gave lower ratings on average for life satisfaction and 'happiness yesterday' and higher ratings for 'anxiety yesterday' than people who did not report

a disability. However, the size of the relationship between self-reported disability and personal wellbeing was small (ONS, 2011a).

Edwards and Imrie (2008) "Explored the moral discourse of wellbeing and its implications in the context of disability in the UK" (p.338). They claim that structural issues which act as barriers to people with disabilities or impairments are more detrimental for personal wellbeing than the disability or impairment.

Edwards and Imrie (2008) argued that disability *per se* is not detrimental to wellbeing, but rather it is the current wellbeing agenda (WBA) in the UK which is detrimental for the wellbeing of those with disabilities. This is grounded in their assertion that mainstream representations of wellbeing undermine societal understanding of disablement in society. This emphasises "Biologism, personality and character traits, and a policy prognosis that revolves around self-help and therapy, or individuated actions and (self) responsibilities" (Edwards and Imrie, 2008:338).

The 'Social Model of Disability' suggests that body image in those with disabilities is an important mediating factor on wellbeing (Goodley and Lawthom, 2006). Evidence suggests those with a positive body image tend to have higher levels of wellbeing (Charmaz, 1995; Goodley and Lawthom, 2006).

Whilst there is evidence that disability has a negative influence on wellbeing, the causal direction remains unclear. Increasingly, critics argue that it is the societal representations of disability and the structural barriers and constraints which people with disabilities and impairments face which truly undermine wellbeing (Edwards and Imrie, 2008; Goodley and Lawthom, 2006; Hobbs and Sixsmith, 2010).

In summary, health is widely perceived as one of the key determinants of wellbeing. Multiple health conditions, which develop in advanced older age, have a particularly negative impact on personal wellbeing. Self-reported health has a particularly strong relationship with wellbeing. However, there remains a shortage of longitudinal studies, which would help to clarify the direction of causality between them (Dolan et al., 2008). Self-reported disability has a significant but

much smaller relationship with personal wellbeing when other variables are held equal, including self-reported health. However, little is known about how long-term chronic physical or mental health conditions affect wellbeing.

Research indicates physical health and psychological health are strongly associated with personal wellbeing. Psychological wellbeing also correlates strongly with personal wellbeing and is discussed in the following section.

# 2.5: Psychological\* wellbeing

Bradburn's (1969) seminal work on the structure of psychological wellbeing laid the foundations for growing research interest in psychological wellbeing. Bradburn (1969) was able to demonstrate that positive and negative affect were independent variables. Bradburn (1969) used this to define happiness as the balance between positive and negative affect. This altered what researchers perceived as important for psychological wellbeing and spearheaded new approaches to the study of psychological wellbeing.

Early empirical research into psychological wellbeing was criticised for narrow conceptions of positive functioning (Ryff, 1989a; Ryff and Keyes 1995). Research focused on the importance of happiness, and failed to recognise the importance of aspects of psychological wellbeing such as sense of purpose, self-acceptance and self-realisation (Ryff and Keyes, 1995).

Much of the early theoretical research around psychological wellbeing emphasised the importance of loving, trusting inter-personal relationships (Ryff, 1989a; Ryff and Keyes, 1995; Ryff and Singer, 2000a; Sellers et al., 2006; Sen, 1993). Theoretical hypothesis such as 'Adult Developmental Stage Theory' also emphasised the importance of intimacy (Hatch et al., 2007). Positive relations with others are now widely recognised as an important aspect of psychological wellbeing (Diener and Biswas-Diener, 2008; Skinner et al., 2008; Winefield et al., 2012). Winefield et al. (2012) for example, used 'Positive Relations with Others', 'Environmental Mastery' and 'Life Satisfaction' as measures of overall wellbeing. The importance of positive

relations with others has been consistently emphasised in conceptualisations of psychological wellbeing (Ryff, 1989, 1998; Ryff et al., 2006; Winefield et al., 2012). Ryff (1989a) was an important figure in advancing theoretical debates about aspects of psychological wellbeing and functioning. Ryff's research laid much of the groundwork for modern conceptualisations of psychological wellbeing (Alkire, 2013; Bilancini, 2012; Bloodworth, 2005; Burman, 2008; Winefield et al., 2012).

One of the most important domains of wellbeing is psychological wellbeing (Alkire, 2013; Ryan and Deci, 2000; Ryff, 1985, 1989a, 1995, 2006). The extent to which aspects of psychological wellbeing are binary opposites or separate constructs is of key interest. For example, Winefield et al. (2012) conducted research to distinguish between psychological wellbeing and psychological distress. Winefield et al. (2012) indicated that psychological wellbeing was not the binary opposite of psychological distress. Instead, there exists a continuum between psychological wellbeing and psychological distress.

Psychological wellbeing has been widely perceived as having a number of interrelated domains. These domains include autonomy, environmental mastery, purpose in life, personal growth, positive functioning, self-realisation, informed choice and control and interpersonal relationships (Devine et al., 2006; Doyal and Gough, 1991; Ryff and Keyes, 1995, 2006; White and Pettit, 2004; Veenhoven, 2000).

Ryan and Deci (2000), however, reduced the number of psychological needs associated with psychological wellbeing. Ryan and Deci (2000) identified three innate psychological needs – autonomy, competence, and relatedness – which their research indicated were necessary to achieve psychological wellbeing. It is important to note, however, that they are concerned with those things which foster wellbeing, rather than those which are indicative of wellbeing (Chavez et al., 2005).

Chavez et al. (2005) reviewed wellbeing and reported that literature concerned with psychological wellbeing continued to focus on domains such as self-

acceptance, sense of purpose or fulfilment in life, continuing personal development, feelings of interpersonal connectedness, happiness and SWB. However, the work of Ryff (1989a, 1989b, 1995, 1999) and Ryff and Keyes (1995) has encouraged new research into psychological wellbeing which contests the earlier focus on the binary nature of the aspects of psychological wellbeing such as absence of positive affect and the presence of negative affect (Chavez et al., 2005). Ryff et al. (2006) for example, called for "Clarification of whether psychological well-being and ill-being comprise opposite ends of a bipolar continuum, or are best construed as separate, independent dimensions of mental health" (p.85).

Theoretical research into psychological wellbeing has advanced understandings of mental health and applied notions of psychological wellbeing. In the UK, mental health has been linked to the psychological wellbeing debate through the work of Sir Richard Layard (Bird, 2007; Newton, 2007).

Layard (2005, 2006, 2009) concluded, "In Britain mental illness has now taken over from unemployment as our greatest social problem" (P.1030). Figures suggest that 300 out of 1,000 people in the UK experience mental health problems every year (Bird, 2007). This has led to increasing research into mental and psychological wellbeing within policy contexts.

Alternative representations of psychological wellbeing have increasingly emerged in recent years, perhaps because of the positive psychology movement. The concept of wellbeing is integral to positive psychology, which is concerned with the strengths and virtues that enable individuals and communities to thrive (Kahneman et al., 1999). Non-biological approaches to the study of psychological wellbeing have come forward to explore alternative perceptions of psychological wellbeing. These include Alkire (2013), who defined psychological wellbeing as "Reflective life satisfaction, positive affect, spirituality and mind training" (p.6).

Alkire (2013) is part of a new generation of human development researchers who explore wellbeing in relation to non-industrialised/developing countries. Human development researchers from the 'Wellbeing in Developing Countries' group offer new paradigmatic approaches to understanding wellbeing. These tend to build on

theoretical work such as Doyal and Gough's (1991) 'Theory of Human Needs', McClelland's 'Human Motivation Theory' (1961) and Ryan and Deci's (2000) 'Self-determination Theory'. Current human development researchers are constructing theoretical expositions, which explore wellbeing and basic needs in relation to wider aspects of wellbeing such as environmental sustainability and ecological preservation.

In summary, psychological wellbeing is highly correlated with personal wellbeing and is considered one of the most important domains of wellbeing (Allen, 2008; Dolan et al., 2008; Oguz et al., 2013). Positive relations with others are now widely recognised as an important aspect of psychological wellbeing. Autonomy may be a key component of psychological wellbeing; however, opportunity and the capability to choose are also important. Autonomy, Environmental Mastery, Personal Growth, Positive Relations with Others, Purpose in Life and Self- Acceptance are all statistically significant in predicting psychological wellbeing. Positive psychological functioning, which incorporates self-acceptance, self-actualisation, and optimal functioning, is also perceived to be an important characteristic of psychological wellbeing.

The premise of psychological wellbeing often rests on models of rational choice, which posit that people use logic and rationality in their pursuit of positive functioning (Ahmed, 2010; King, 2007; Mc Gregor, 2006; Ryan and Deci, 2000). However, Kahneman (1994, 1999) argued that people do not always make rational choices, and that these choices do not necessarily enhance psychological wellbeing.

The impact of psychological distress on psychological wellbeing has become increasingly important to those interested in psychological wellbeing (Oswald and Powdthavee, 2007a; Winefield et al., 2012). This may be because the incidence of reported psychological distress has consistently risen in the UK over the past twenty years (Oswald and Powdthavee, 2007a). However, inconsistent findings on the relationships between psychological wellbeing and psychological distress means that further research is required to unpack this (Winefield et al., 2012).

### 2.6: Physical wellbeing

Higher levels of physical wellbeing have been identified as having a significant impact on health and holistic wellbeing (Carlisle and Hanlon 2007; Chavez et al., 2005; Griffin, 2010; Penedo and Dahn, 2005; Reiner et al., 2013). This is also associated with better emotional and mental health (Allen, 2008; McCormick et al., 2009; Reiner et al., 2013).

Penedo and Dahn (2005) identified that physical wellbeing has a positive association between better functional capacity and better mood states. Physical wellbeing has also been identified with decreased risk of disease, illness, and injury; better immune functioning; speedier recovery; and increased longevity (Diener and Biswas-Diener, 2008; Ostir et al., 2000, 2001; Pressman and Cohen, 2005). Penedo and Dahn (2005) reviewed cross-sectional and longitudinal studies, which evaluated the relationship between exercise, physical activity, physical wellbeing and mental health and wellbeing. Penedo and Dahn's (2005) results support a growing literature base which indicates that exercise, physical activity and physical wellbeing have beneficial effects across several domains of wellbeing, most significantly in psychological, emotional and social wellbeing (Allen, 2008; McCormick et al., 2009; Reiner et al., 2013).

Much of the UK research around physical wellbeing indicates that exercise and physical activity are associated with mental health, psychological wellbeing and health outcomes (Biddle et al., 2003; Callaghan et al., 2011; Reiner et al., 2013). However, to date much of the UK evidence originates from quantitative clinical trials (Reiner et al., 2013; Trivedi et al., 2011) which tend to contain limited sample sizes and have short follow-up periods (Callaghan et al., 2011).

Physical health is considered hugely important for physical wellbeing (Allen, 2008; Barnes et al., 2013; McCormick et al., 2009; Ward et al., 2012). Research indicates physical wellbeing underpins holistic wellbeing (Allen, 2008; McCormick et al., 2009). In addition to which it has been identified as an important mechanism for combating illnesses and improving functionality (Penedo and Dahn, 2005; Reiner et al., 2013; Scheier et al., 2006).

Factors which undermine physical wellbeing may often result from long-term chronic ill health, impairment or disability (Allen, 2008; McCormick et al., 2009). These often result in depression and other mental health issues, which further undermine physical wellbeing (Allen, 2008; McCormick et al., 2009).

In summary, much of the UK research around physical wellbeing indicates that exercise and physical activity are also associated with psychological wellbeing, emotional wellbeing and holistic wellbeing. Further research is required to explore the interaction between the environmental determinants of physical wellbeing and the opportunities to facilitate greater physical wellbeing. Similarly, the associations and relationships between physical wellbeing, emotional wellbeing and personal wellbeing need clarification around the direction of causality.

# 2.7: Spiritual, religious and emotional wellbeing

When considering the impact religion\* and spirituality may have on wellbeing, it is important to take into consideration the relationship between spiritual dimensions and wellbeing and to acknowledge the differences between spirituality and religiosity (Chavez et al., 2005; McCullough et al., 2000). Spirituality is associated with religion, and the terms overlap but are not synonymous. An individual may report being spiritual but claim no affiliation with religion (Sartori, 2010).

# 2.7.1: Spiritual wellbeing

Visser et al. (2010) defined spirituality as an experience of a connection with the essence of life. Piedmont (1999) emphasised the importance of spiritual wellbeing in fostering a sense of being non-judgmental. Additionally, it is also associated with altruistic behaviour, having meaning in life and contributing to a sense of connection with others and the wider world (Bartlett et al., 2003; Vilhena et al., 2014; Woolryche and Sixsmith, 2008).

Koenig (2008) selectively reviewed the role of religion and spirituality on health and wellbeing and concluded that religion and spirituality act as powerful coping mechanisms, which help to facilitate wellbeing. These mechanisms help people

make sense of challenges, inculcate a sense of control and support the socio-cultural rules and observations that facilitate mutual support and cooperation (Clark and Lelkes, 2005; Graham et al., 2010; Shobhna, 2008). More recently, Vilhena et al. (2014) explored spirituality and wellbeing in 774 cancer patients. They concluded that spirituality might facilitate and improve emotional wellbeing and resilience by encouraging a sense of positivity (Bartlett et al., 2003; Shobhna, 2008; Vilhena et al., 2014).

There are those, however, who dispute evidence of a positive association between wellbeing and religion/spirituality. Hebert et al.'s (2006) systematic review of the relationships between religion/spirituality and wellbeing suggested that the effects of religion and spirituality on wellbeing remain unclear (Hebert et al., 2006).

## 2.7.2: Religious wellbeing

Early research on religion and wellbeing suggested church-related activity was associated with wellbeing (Clark and Lelkes, 2005; Cohen, 2002; Joshi et al., 2008). Similarly, Inglehart (1990) reported that religious affiliation and worship attendance was positively associated with happiness and life satisfaction. Religiousness has been identified by some as one of the best predictors of wellbeing amongst older people as it can facilitate resilience through finding meaning in life circumstances and events (Okun and Stock, 1987). This claim has been substantiated by research which suggests religiosity is important for wellbeing, particularly for older people because it fosters emotional wellbeing and helps with difficult life transitions (Donovan and Halpern, 2002; Joshi et al., 2008; McCullough et al., 2000). However, there is some evidence to indicate that social standing influences some of the relationship between religion and wellbeing (Dolan et al., 2006, Dolan and White, 2007).

Dolan et al. (2006) reviewed factors affecting wellbeing and identified church attendance and a belief in God as being positively associated with SWB. There has been much speculation concerning why religion may be particularly important for

older peoples' wellbeing (Donovan and Halpern, 2002; Graham et al., 2010; Inglehart, 1990; McCullough et al., 2000). Donovan and Halpern (2002) theorised this is because religion confers a sense of purpose and continued participation within a social and supportive social network.

More recently, Oguz et al. (2013) analysed results from the UK Annual Population Survey (2011–2012). Those participants who reported a religious affiliation compared to those who did not were more likely to feel 'satisfied' with their life and feel they lived a 'worthwhile' existence (Oguz et al., 2013).

However, other studies have found that religious faith can impact negatively on wellbeing (James and Wells, 2003; Koenig, 2008). For example, religiosity may exert a detrimental impact on personal relationships and emotional wellbeing if those within one's close social circle do not share the same religious convictions (James and Wells, 2003). Koenig (2008) also cautioned against a tendency to focus predominantly on the positive associations between religion, health and wellbeing. "While religious beliefs and practices can represent powerful sources of comfort, hope and meaning, they are often intricately entangled with neurotic and psychotic disorders, sometimes making it difficult to determine whether they are a resource or a liability" (P.289).

### 2.7.3: Emotional wellbeing

Emotional wellbeing is shaped by factors including gender, ethnicity, socio-economic position, environment, inequalities, health and disability (Allen, 2008; Cameron, 1975; Clifton, 2009; Demakakos et al., 2010; Diener and Larsen, 1993; McCormick et al., 2009). Emotional wellbeing has a close association with physical health and wellbeing (Allen, 2008; Demakakos et al., 2010; Ostir et al., 2000, 2001). The relationship between health and emotional wellbeing has become of increasing interest to wellbeing researchers in recent years (Abdallah, 2012; Allen, 2008; Pahl, 2007). This may be the result of ageing populations and the detrimental impact that

ill health and conditions such as dementia have on emotional wellbeing (Abdallah, 2012; Allen, 2008).

Allen (2008) argued that the UK has a large and growing number of older people with poor emotional wellbeing because of a greater prevalence of mental health issues. This is compounded by an increase in reported depression in older people and 'older' old people, i.e. those over 85 (Allen, 2008). Depression, the most common mental health issue for those aged 65+ (Allen, 2008), has been frequently identified as having a negative impact on emotional wellbeing (Abdallah, 2012; Allen, 2008; Barnes et al., 2013; Diener and Larsen, 1993; McCormick et al., 2009). Poor access to 'talking therapies' for older adults (Layard, 2005) has exacerbated the incidence of depression (Allen, 2008). This has concomitantly led to increased levels of poor emotional wellbeing (Allen, 2008; McCormick et al., 2009).

There is also strong evidence of an association between physical health and poor emotional wellbeing (Abdallah, 2012; Allen, 2008; DoH, 2008; ONS, 2011, 2012). This association, exacerbated by health inequalities, means those from deprived backgrounds are more likely to experience poor emotional wellbeing than those from affluent backgrounds (Allen, 2008; ONS, 2011, 2012). "Poverty has a clear relationship with poor emotional wellbeing across the life cycle and worsening income inequalities compound that" (Allen, 2008:22).

Poverty and deprivation have been widely recognised as significant risk factors for poor emotional wellbeing in later life (Allen, 2008; Barnes et al., 2013; DoH, 2008; Pahl, 2007). Research suggests that this is likely to be related to factors such as fear of crime and poor quality housing (Allen, 2008; Barnes et al., 2013; Dolan et al., 2006; ONS, 2011, 2012). Factors which appear to be particularly relevant for poor emotional wellbeing include ill health, depression, disability, loneliness, poverty and isolation (Abdallah, 2012; Allen, 2008; Barnes et al., 2013; Diener and Larsen, 1993; DoH, 2008; Ward et al., 2012).

Conversely, there is evidence to suggest that positive emotional wellbeing is principally associated with social interaction and community wellbeing (Abdallah,

2012; Allen, 2008; Barnes et al., 2013; Pahl, 2007; Turk, 2009). Participation and interaction within close social contexts such as family life, one's neighbourhood and wider community are seen as particularly important for fostering and protecting emotional wellbeing (Abdallah, 2012; Allen, 2008; Barnes et al., 2013). This is especially the case for older adults whose social circle tends to diminish as they age (Allen, 2008; McCormick et al., 2009).

Opinions differ about what is most important for emotional wellbeing. Turk (2009) suggests respect and social status within one's community underpins emotional wellbeing. However, there is general consensus that emotional wellbeing, particularly for older adults, depends to a large degree on social interaction (Allen, 2008; Barnes et al., 2013; Ward et al., 2012).

## 2.8: Economic wellbeing

Traditionally, the study of economic wellbeing assumed wellbeing was a function of income (Easterlin, 1974, 2009; Frey and Stutzer, 2002; Paim, 1995; Qizibash, 1998). Research suggests that a number of variables are important for economic wellbeing (King, 2007; Paim, 1995; Qizibash, 1998). However, the effects of employment, unemployment and income levels have a significant influence on economic wellbeing (Abdallah, 2012; Biswas-Diener, 2008; Diener et al., 1999; Diener and Biswas-Diener, 2002; Dolan et al., 2006; Easterlin, 2009; Jeffrey et al., 2014; ONS, 2012, 2013). This recognition has led to them being widely used as the most popular proxy measures of economic wellbeing, a format followed here.

# 2.8.1: The relationship between economic wellbeing and income

The relationship between income and wellbeing has been extensively explored (Seaford, 2011; Stoll, 2012). It is widely agreed that the interaction between them is complex and multi-layered (Dewe and Cooper, 2012; Diener, 1999; Frey and Stutzer, 2002; Stevenson and Wolfers, 2008). One of the most contested points in

this field of study centres on whether the relationship between economic wellbeing and personal income is linear or curvilinear (Dewe and Cooper, 2012; Frey and Stutzer, 2002; Jeffrey et al., 2014).

Evidence indicates that wealth has a greater influence on wellbeing in poor countries than in rich, and has a greater impact on people who are poor than those who are rich (Easterlin, 1974, 1995, 2001, 2009; Frey and Stutzer, 2002; Wilkinson and Picket, 2009). This phenomenon has been explained by the 'Theory of Basic Needs' (Maslow, 1957; Doyal and Gough, 1991). It has been widely documented that since World War II, despite a considerable increase in standards of living and personal income there has not been a commensurate rise in wellbeing (Blanchflower and Oswald, 1999; Easterlin, 1974, 1995, 2001, 2009; Marks and Shah, 2004; Layard 2006). These findings helped to establish claims of a curvilinear relationship between wellbeing and income (Dewe and Cooper, 2012; Stevenson and Wolfers, 2008). Evidence suggests this is linked to the 'Theory of Diminishing Returns' (Biswas-Diener, 2008; Easterlin, 1974, 2009; Stevenson and Wolfers, 2008; Stoll et al., 2012).

The 'Theory of Diminishing Returns' suggests once wealth has reached subsistence level, its effect on wellbeing is greatly reduced (Diener and Diener, 1995; Inglehart and Klingemann, 2000; Stevenson and Wolfers, 2008). This paradox is widely referred to as the 'Easterlin Paradox' (Macunovich and Easterlin, 2008). The Easterlin Paradox has come under increasing criticism in recent years for being based on short-term data (Jeffrey et al., 2014; Stevenson and Wolfers, 2008). Some research suggests that the Easterlin paradox does not exist and wellbeing has a linear relationship to personal income (Booth, 2012; Hagerty and Veenhoven, 2003).

Booth (2012), for example, argued, "Contrary to popular perception, new statistical work suggests that happiness is related to income and that this relationship holds between countries, within countries and over time. The relationship is robust and also holds at higher levels of income as well as at lower levels of income" (P.19). There is now growing support for claims that long term data indicates the existence

of a positive and linear relationship between economic wellbeing and personal income (Stevenson and Wolfers 2008; Diener et al., 1993; Deaton, 2008).

Criticisms of the Easterlin Paradox led Easterlin et al. (2010) to revisit the phenomenon. Easterlin et al.'s (2010) findings indicated that there was a positive correlation between income and happiness/wellbeing in the short term. This was mitigated, however, as happiness/wellbeing tends to fall during periods of economic decline and rise during periods of economic expansion. Furthermore, Easterlin et al. (2010) argued that data from developing countries demonstrates that the long-term relationship between income and wellbeing still conforms to the Easterlin Paradox.

Despite growing support for the existence of a positive and linear relationship between economic wellbeing and personal income there remains considerable support for the Theory of Diminishing Returns and a curvilinear relationship between economic wellbeing and personal income (Biswas-Diener 2008; Clark et al., 2008; Dolan et al., 2006; Easterlin et al., 2010; Kahneman and Deaton, 2010).

Kahneman and Deaton (2010) found that income does correlate with economic wellbeing/happiness, but that the rate of positive association diminishes past a certain income threshold. Research suggests higher earners generally report better life satisfaction, but diminishing returns means that people's daily emotional wellbeing only rises with earnings until a threshold annual income of \$75,000 is reached (Kahneman and Deaton, 2010). This has been linked to claims that higher incomes enable better health, increased life expectancy and less stressful life events (Ballas and Dorling, 2007; Jeffrey et al., 2014; Kahneman and Deaton, 2010).

However, some argue this could be counteracted by a diminished opportunity to undertake leisure activities or to establish strong social networks, both of which enhance wellbeing (Deiner and Biswas-Deiner, 2002; Jeffrey et al., 2014; Lewis and Purcell, 2007). Kasser and Ryan (1993) reported that the desire for material gain may have a negative impact on wellbeing, where the desire for a higher income increases workplace stress. Workplace stress has been associated with having a negative impact on close personal relationships (Dewe and Cooper, 2012; Lewis and

Purcell, 2007). Marriage particularly suffers where "Possessing an upper income is associated with a doubled to quadrupled likelihood of divorce" (Clydesdale, 1997:605).

Advances in research have generated a broad consensus that income is only modestly correlated with wellbeing (Clark and Senik, 2011; Easterlin, 2009; Easterlin et al., 2010). Associations between income and wellbeing appear strongest for those in Western nations with relatively low-income rates, although research has also have found effects for those at higher income levels (Argyle, 1999; Blanchflower and Oswald, 1999; Easterlin, 2009).

Many studies indicated that national wealth has a greater influence on wellbeing than personal income (Diener and Diener, 1995; Diener and Suh, 1999; Easterlin, 2009; Inglehart and Klingemann, 2000; Kahneman and Deaton, 2010; Veenhoven, 1991). However, as with personal income, evidence suggests there is a point of diminishing returns (Biswas-Diener, 2008; Kahneman et al., 2006: Stoll et al., 2012).

### 2.8.2: Unemployment

Jahoda's (1982) seminal research on the psychology of unemployment and employment produced one of the earliest and most influential studies on the effects of unemployment and employment on economic wellbeing. Jahoda (1982) identified that the unemployed were deprived five categories vital to wellbeing "Time structure, social contact, collective effort or purpose, social identity or status, and regular activity" (p.87). Jahoda (1982) maintained that unemployed people were unhappy because they were deprived of these qualities and not simply because they were poor. This interpretation led to research such as Clark and Oswald (1994) and Clark et al. (2004), which furthered understanding of the relationship between wellbeing and employment (Dewe and Cooper, 2012; Jeffrey et al., 2014).

Research has subsequently identified that unemployment negatively affects wellbeing in the short and the long term (Abdallah, 2012; Argyle, 1999; Deaton,

2011; Dolan et al., 2006; Lewis and Purcell, 2007). Individuals who are unemployed experience significantly lower wellbeing (Abdallah, 2012; Dolan et al., 2006; Jahoda, 1982; Warr et al., 2010). The unemployed, however, cannot be considered a homogenous group (Abdallah, 2012; Clark et al., 2004). Those who have been unemployed for more than six months have a significantly lower wellbeing than those who have been unemployed for less than six months (Abdallah, 2012). There is also evidence that unemployment is not unremittingly bad for wellbeing (Abdallah, 2012). In communities where a significant number of the population is unemployed, unemployment does not appear to have detrimental effect on wellbeing (Abdallah, 2012; Argyle, 2002; Dewe and Cooper, 2012). One explanation for this finding is that the social comparison and social standing components of the impact of unemployment on wellbeing is moderated when living in a social group in which unemployment is the norm (Abdallah, 2012; Ballas et al., 2007; Lewis and Purcell, 2007).

### 2.8.3: Employment

Paid employment has been widely reported as key in the wellbeing of individuals (Argyle, 2002; Ballas et al., 2007; Clark and Senik, 2011; Dewe and Cooper, 2012; Haworth, 2007; Lewis and Purcell, 2007; Warr, 2007; Warr and Clapperton, 2010). Paid employment is important because it has economic benefits and facilitates access to resources. It can also be an indicator of feeling valued by others and by ourselves (Ballas et al., 2007; Haworth, 2007; Layard, 2005; Lewis and Purcell, 2007; Warr and Clapperton, 2010). In addition to which, paid employment has also been identified with fostering satisfaction, meaning and purpose in life (Blanchflower and Oswald, 1999; Clark and Senik, 2011; Deaton, 2011; Lewis and Purcell, 2007; Warr, 2003).

Job satisfaction also appears to play an important part in the positive impact employment has on wellbeing (Dewe and Cooper, 2012; Jeffrey et al., 2014; Warr and Clapperton, 2010). Jeffrey et al. (2014) reported that most aspects of an individuals' psychological wellbeing such as sense of purpose, positive emotions,

morale, motivation and life satisfaction were enhanced for those in paid and unpaid employment with job satisfaction.

Research has increasingly focused not only on the impact of employment on wellbeing but also on the importance of meaningful work (Dolan et al., 2006; Lewis and Purcell, 2007; Prilleltensky and Prilleltensky, 2007; Warr, 2007; Warr et al., 2010). Employment, both paid and unpaid, particularly that which is stimulating and meaningful has been identified as engendering factors, related to improved wellbeing; for example, feelings of self-control, flow and the opportunity for creative experiences (Haworth, 2007; Lewis and Purcell, 2007; Warr, 2007; Warr et al., 2010).

In addition, employment helps to facilitate social standing, a sense of pride and belonging, which helps people "Construct their social identity" (Myers and Diener, 1995:14). Employed individuals also demonstrate better wellbeing (Abdallah, 2012; Dolan et al., 2006; Jahoda, 1982). However, evidence, suggests that employment does not have a consistent independent relationship with wellbeing (Abdallah, 2012).

For many economists it is the earning potential of paid employment which facilitates a higher sense of economic wellbeing. However, Carr (2004) identified that "In economically advanced countries people who value earning money more than other goals are less satisfied with their standard of living and their lives" (P.33). Studies have increasingly focused on the individual differences such as goals, values and personality traits that could moderate the relationship between personal income and wellbeing (Abdallah, 2012; Dewe and Cooper, 2012; Warr and Clapperton, 2010).

Malka and Chatman (2003), for instance, proposed that individual differences in intrinsic and extrinsic work orientation might moderate these relationships. Their research showed a connection between wellbeing and income, which changed depending on the individuals' extrinsic and intrinsic orientations towards work (Malka and Chatman, 2003). Individuals with a more extrinsic work ethic exhibited a

stronger association between income and SWB. Those who valued an intrinsic approach to work exhibited a stronger association between income and emotional wellbeing (Malka and Chatman, 2003).

Other studies have since confirmed that the impact of work on economic wellbeing varies depending on factors such as personality traits and values such as the extrinsic and intrinsic orientations of the individual (Dewe and Cooper, 2012; Warr, 2007; Warr and Clapperton, 2010). For those with an intrinsic orientation paid and unpaid employment can facilitate an improved sense of economic and holistic wellbeing (Haworth, 2007; Lewis and Purcell, 2007; Prilleltensky and Prilleltensky, 2007).

Ballas and Dorling (2007b) explored the impact of major life events on happiness and wellbeing and concluded that it was employment-related gains rather than employment status which was a significant predictor of wellbeing. This interpretation is supported by those who argue that immersion in work, 'flow' and the pleasure from that pursuit has the most significant impact on wellbeing (Haworth, 2007; Lewis and Purcell, 2007; Prilleltensky and Prilleltensky, 2007). These help people to establish and accomplish goals, achieving one's aspirations and personal fulfilment, all of which are experiences which embody the notion of eudaimonic wellbeing (Delle Fave and Massimini, 2003a; Lewis and Purcell, 2007).

It is important to note, however, that not all employment *per se* has an equal effect on economic wellbeing (Biswas-Diener, 2008; Deaton, 2008; Dolan et al., 2006; Easterlin et al., 2010). For example, individuals who have temporary employment contracts experience lower wellbeing than those on other contracts, even when controlling for other factors (Abdallah, 2012; Donovan and Halpern, 2002). Self-employed people meanwhile are more satisfied and experience a greater sense of wellbeing than employed people doing equivalent work (Dolan et al., 2006; Donovan and Halpern, 2002). This was also true when taking into consideration confounding factors such as the self-employed working longer hours, having fewer holidays and earning less (Donovan and Halpern, 2002).

In summary, research indicates that income is important for economic wellbeing, but beyond a certain level of earnings commensurate with the earning potential of those of a similar socio-economic position the impact of income on wellbeing declines.

Similarly, employment is also important for economic wellbeing but it is the presence of meaningful, interesting, stimulating and satisfying employment which has a positive impact on economic wellbeing. Conversely, routine, monotonous, repetitive employment has a negative impact on economic wellbeing. Factors such as job insecurity, work/life balance and zero hour contracts have a significant and negative impact on economic wellbeing.

Unemployment has perhaps the most significant and negative impact on economic wellbeing. However, even in this area there is evidence which indicates that social comparison has an important mitigating role to play. That is, those unemployed living in a community of high unemployment suffer less from the stigmatising factors of unemployment, which lessens the significant and negative impact on economic wellbeing.

It is difficult to ascertain the relationship between income, employment, unemployment and wellbeing. The number of possible diverging and confounding factors which may interact with and affect the data make the task extremely difficult. This review indicates a need for greater research into the individual factors that could affect findings.

# 2.9: Social wellbeing\*

Some have argued that wellbeing has been narrowly perceived as that which pertains to the personal (Haworth and Hart, 2007; Prilleltensky and Prilleltensky, 2007; Seligman 2011). At a theoretical level there has been a tendency for some wellbeing theories to underplay the social domain (Bruni and Porta, 2007b;

Haworth and Hart, 2007; Phillips, 2006). In a comparison of wellbeing theories Phillips (2006) identified that those with an economic focus assume social wellbeing pertains solely to the wellbeing of individuals. Bruni and Porta (2007b), who claimed that economic theories of wellbeing tend to ignore inter-personal relationships and social wellbeing, support this observation. Gasper (2010) acknowledged this oversight and argued that wellbeing theories should do more to evaluate the social needs and requirements of individuals.

Conversely, at an empirical level research has endeavoured to explore wellbeing in connection to relationships (Bjørnskov, 2007; Bowling et al., 2003; Myers and Deiner, 1995). Indeed Bowling et al. (2003) claimed, "The largest body of empirical research on the various facets of well-being has focused on the structure of social networks and the functioning and supportiveness of human relationships" (p.272). The evidence has resulted in a consensus that social relationships and social wellbeing are integral to personal wellbeing (Allen, 2008; Barnes et al., 2013; Dolan et al., 2006; Donovan and Halpern, 2002; Gasper, 2010; McCormick et al., 2009; Ward et al., 2012). There is strong evidence that important social networks\*, which include friendships, close relatives, neighbours, colleagues and romantic partners all play an important role in facilitating social wellbeing (Allen, 2008; Barnes et al., 2013; McCormick et al., 2009; Ward et al., 2012).

Evidence suggests that social wellbeing is underpinned by social acceptance, and the fundamental need to belong (Allen, 2008; Barnes et al., 2013; Haworth and Hart, 2007; Li et al., 2005). Taken together these studies suggest that social wellbeing contributes to personal wellbeing by inculcating a sense of meaningfulness, belonging, happiness and social acceptance. There is some evidence to suggest aspects of social wellbeing such as support networks are particularly important for psychological wellbeing because they act as important buffers against stress, anxiety, isolation and depression (Barnes et al., 2013; McCormick et al., 2009; Ward et al., 2012).

There is also strong evidence that associations between social participation and support is of key importance for social wellbeing (Barnes et al., 2013; Lawton, 1980;

Li et al., 2005; Sixsmith and Boneham, 2007; Ward et al., 2012). This is particularly the case when participation and support stems from relationships which are felt to be reciprocal and offer a balance between dependence and independence (Allen, 2008; Barnes et al., 2013; Ward et al., 2012). This corpus of research has enabled researchers to argue that social wellbeing in the shape of social connectedness is one of the key determinants of personal wellbeing (Allen, 2008; Barnes et al., 2013; Taylor, 2011; Ward et al., 2012).

Conversely, research indicates that poor social wellbeing characterised by social isolation leads to feelings of depression and anxiety (Allen, 2008; Bjørnskov, 2007; Li et al., 2005). "Lack of attachments is linked to a variety of ill effects on health, adjustment and well-being" (Baumeister and Leary, 1995:497). However, the benefits to social wellbeing derived from social networks depend on the nature and quality of those networks (Barnes et al., 2013; Bjørnskov, 2007; Li et al., 2005). Social networks, which are superficial or one-dimensional, may have a negative impact on social wellbeing (Allen, 2008; Barnes et al., 2013; Bjørnskov, 2007; Li et al., 2005).

In addition to the development of social relationships, social wellbeing is linked to social capital. This association focuses on the key components of participation, trust and reciprocity and the social networks of bonding, bridging and linking ties (Li et al., 2005; Haworth and Hart, 2007; Sixsmith and Boneham, 2007). However, much of the research concerned with social capital is more closely aligned with community wellbeing and so will be addressed in the following section.

In summary, the degree to which social wellbeing affects personal wellbeing and community wellbeing remains a matter for debate. However, evidence suggests social wellbeing through mechanisms such as connectedness and relationships has a significant impact on personal wellbeing. Further research is necessary to try to unpack this.

## 2.10: Environmental wellbeing\*

A relationship between an individual and the environment is seen as critical to perceptions of identity, belonging and familiarity (Ballas et al., 2007a; Ballas and Dorling, 2013; Newton, 2007; Pickering, 2007). Sayer (2000) argued that these are integral to inclusion, participation and engagement, themselves seen as key facets of wellbeing. The Foresight Report (Environment Agency, 2004) and National Institute for Clinical Health Excellence guidance (NICE), (2008) both reinforced the importance of the environment as a key determinant of wellbeing.

### 2.10.1: Natural environment\*

Much of the early work, which identified the natural environment as essential for personal wellbeing, was predicated on an intrinsic assumption that humans had an inherent relationship with nature (Bird, 2007; Naess, 1973; Newton, 2007; Pickering, 2007; Wells et al., 2010; Wilson, 1984).

Wilson's 'Biophilia Hypothesis' (1984) asserted that at a fundamental level human identity and self-actualisation relies on our relationship with the natural environment. This has been influential in encouraging others to develop theoretical accounts, which help provide frameworks for understanding the complex relationship between humans and the natural environment (Galloway, 2006; Newton, 2007). In addition to which, Wilson's Biophilia Hypothesis (1984) offers theoretical support for those who argue that humans have an emotional attachment to the natural environment (Newton, 2007). For many this attachment meets subconscious and deep-rooted human needs and promotes wellbeing (Abdallah, 2012; Abdallah et al., 2012; Bird, 2007; Newton, 2007; Pickering, 2007; Stoll, 2012).

However, much of the early evidence linking the natural environment and wellbeing was generated through small-scale qualitative studies. This led to disagreement over whether these studies contributed robust and reliable evidence, which met the perceived requirement for a robust clinical and quantitative evidence base (Newton, 2007). The concerted move towards larger quantitative

studies accompanied research which increasingly sought to offer theoretical suppositions to support empirical evidence (Newton, 2007).

Newton (2007) undertook a literature review to substantiate claims that the natural environment plays an important role in wellbeing. Newton's review was a significant step forward for those interested in conceptualising and exploring the connections between the natural environment and wellbeing. Its main value lay in its presentation of a wide body of evidence of the role the environment may play in wellbeing within the context of sustainable development.

Newton (2007) reported a positive and significant association between the natural environment and wellbeing. This, however, was primarily based on qualitative findings from outside the UK. Further research is required to determine the extent to which global findings are replicated in the UK (Abdallah et al., 2012; Berry, 2014; Newton, 2007).

Since Newton's review (2007), much of the UK research undertaken to explore interactions between the natural environment and wellbeing has continued to utilise quantitative approaches (Abdallah et al., 2012; Berry, 2014; Sustainable Development Commission, 2007a). White et al.'s (2013) longitudinal study based on 18 years of panel data (1991–2009) identified that the natural environment has the potential to have a considerable and negative impact on psychological health and wellbeing. There is also evidence that some people do not respond positively to natural landscapes in urban areas (Abdallah et al., 2012; Newton, 2007; Stoll et al., 2012).

The safety implications of green spaces are particularly pertinent for women and older people (Butterworth, 2000; Commission for Architecture and the Built Environment, 2009; Stewart and Bushell, 2011). Fears for personal safety, issues over access and perceptions of the natural environment as unkempt or valueless offers support for claims that the natural environment may have a negative impact on wellbeing (Giles-Corti and Donovan, 2002; Giles-Corti et al., 2005; Newton, 2007). UK quantitative research findings largely support global findings that greener environments and interaction with nature have a predominantly positive impact,

particularly for psychological, personal and community wellbeing. However, further qualitative research is required to help identify why and how different aspects of the natural environment enhance feelings of wellbeing (Berry, 2014; Stoll et al., 2012; Thompson and Marks, 2008).

In summary, there is a strong evidence base to support claims that the natural environment has a positive impact on wellbeing. In addition to which, living in a greener area has a significant impact on mood and cognitive functioning. However, much of the research used to support this claim has been undertaken in the US, Japan and Scandinavia and is restricted by an over reliance on observational research. An important caveat to note is that whilst findings from previous research have indicated a correlation between wellbeing and green space, research has not been able to prove the direction of causality (Dolan et al., 2006; Newton, 2007; White et al., 2013).

In recent years, the UK has experienced a growth of quantitative studies seeking to explicate understandings of the interactions between the natural environment and wellbeing. However, this review has identified that further research is required to explore the potential benefit of more urban green spaces at a population level in promoting wellbeing. Furthermore, there is a need in the UK for qualitative research to unpack the relationship between aspects of the natural environment and the positive and negative impact they may have on wellbeing.

#### 2.10.2: The built environment

The World Health Organization (WHO) (1986) identified the need for research, which recognised the impact of the built environment on health and wellbeing. Since then there have been concerted efforts to investigate the relationship between the built environment and wellbeing (Crawshaw, 2008; Sciven and Garman, 2005; Seedhouse, 1998, 2001; South and Woodall, 2010; WHO, 1998).

The Local Government Act (2000) swelled this interest, encouraging disciplines such as public health/health promotion and social geography to focus on exploring the

impact of the built environment on wellbeing (Ballas et al., 2007a; Cameron, 2006; Carlisle and Hanlon, 2008; Crawshawe, 2008). This interest was primarily aligned to the interactions between building and urban planning policy, design and wellbeing (Ballas, 2010, Ballas and Dorling, 2013; Butterworth, 2000; Carlisle and Hanlon, 2005, 2008). This interest sparked research, which subsequently identified that active involvement in planning for the built environment was an under-utilised arena for improving health and wellbeing (Black, 2015; Commission for Architecture and the Built Environment, 2007, 2009; Carlisle and Hanlon, 2005, 2007).

Research which explores the relationship between wellbeing and the built environment tends to coalesce around studies that explore wellbeing in relation to housing and its immediate locale, and those which explore wellbeing in relation to the wider aspects of the built environment (Black, 2015; Stewart and Bushell, 2011). This approach is used in the following sections.

### 2.10.3: Housing

There is a well-documented and established evidence base which supports claims of a relationship between wellbeing and housing (Allen, 2008; Black, 2015; Commission for Architecture and the Built Environment, 2007, 2009; Stewart and Bushell, 2011). Amongst the earliest research to demonstrate this relationship were two studies employing longitudinal designs, which showed sustained improvements in wellbeing among people relocating to improved housing (Carp, 1968; Lawton and Cohen, 1974). Three other studies also suggested that differences in housing affect wellbeing (Martin, 1973; Schooler, 1970; Smith and Lipman, 1972). The differences reported in these studies appeared to suggest that wellbeing improved in spite of a reduction in social interaction. This implied that the physical aspects and conditions of a dwelling directly affect wellbeing (Schooler, 1970; Smith and Lipman, 1972). More recently, there have been sustained efforts to isolate the interaction between housing conditions and wellbeing/QoL (Black et al., 2015; OECD, 2011; ONS, 2012b; Potter et al., 2012).

Since the worldwide recession there is evidence to suggest that the relationship between wellbeing and housing conditions is often predicated on pecuniary resources (Gibb, 2014). However, it is important to acknowledge, "Measuring housing conditions and their effects on people's well-being is a complex task" (OECD, 2011:82). Despite these challenges, research has identified that well maintained housing has a significant impact on wellbeing (Black et al., 2015; Gibb et al., 2013; OECD, 2011; ONS, 2012b). Whilst the importance of housing has been acknowledged in relation to wellbeing, Black et al. (2015) noted that in order to facilitate a sense of personal wellbeing "Beyond the house itself, our homes need to be in well-designed neighbourhoods with strong communities" (p.5).

The importance of the dwelling space on wellbeing is therefore not limited to personal housing conditions. Weich et al. (2002), provided evidence that those living in deprived urban areas characterised by derelict buildings, graffiti and neglected public spaces were more likely to suffer episodes of depression than those who lived in areas with well-maintained buildings and public spaces. Such research offers support for the 'Environment Stress Hypothesis', which implies that the quality of a neighbourhood has an impact on mental wellbeing (Cairney et al., 2010; Cairney et al., 2013).

# 2.10.4: Public spaces

Empirical research suggests that there are a number of public spaces which facilitate wellbeing (DCLG; 2008; Liu et al., 2009; Spokane et al., 2007). These include open public spaces such as parks, gardens and other recreational areas and other settings such as post offices, libraries and civic centres (Butterworth, 2000; DCLG, 2008). Empirical research indicates that informal public spaces such as high streets, pubs, cafes, post offices, and other 'Third Places' (Oldenburg, 1989, 2000) are the heart of a community's social strength and cohesion (Appleyard and Lintell, 1982; Burton et al., 2011; Liu et al., 2009; Oldenburg, 1989, 2000; Spokane et al., 2007).

Oldenburg (1989, 2000) identified the home as the 'first place', work as the 'second place' and informal public spaces as the 'third place'. These spaces are of key importance for wellbeing as they facilitate interaction, access to support and those ties of social wellbeing which connect communities (Oldenburg 1989, 2000).

Research, however, suggests that the way people use public space and place varies according to factors such as age, gender, ethnicity and a variety of other factors (Dooris, 2004, 2006). These may create issues over territoriality and ownership, which can help to reinforce problems of social isolation and exclusion for certain sectors of society (Burton et al., 2011; Butterworth, 2000; Department for Communities and Local Government, 2008). Evidence indicates that aspects of the built environment have both a positive and negative impact on wellbeing, depending in part on how an individual perceives the space (Burton et al., 2011; Butterworth, 2000; Department for Communities and Local Government, 2008).

Recent research suggests that wellbeing in later life is closely related to the built environment, which in itself is an important mediator of ageing experiences and opportunities (Burton et al., 2011). Burton et al. (2011) suggested that the physical characteristics of neighbourhoods appear to have a significant impact on the mobility, independence and wellbeing of older people and women.

Clarke and Nieuvenhuijsen (2009) argued that "Consideration of the built environment is particularly pertinent for older people: as they age, they are likely to spend more time in their home and community environments, and declining health and functional status can make them more susceptible to barriers in them" (p. 14).

Promoting environments which enable older people to flourish has been identified as an important aspect of local government wellbeing actions (The Foresight Mental Capital and Wellbeing, 2008). Changes which will enhance the built environment include improving the design of homes and towns, improving access to public spaces and making open public spaces safer and more appealing (Foresight Mental Capital and Wellbeing, 2008). These spaces need to be maintained, attractive, well lit, and networked to other resources to reduce fears for personal safety (Butterworth, 2000; Dooris, 2004, 2006).

Housing has both a direct and indirect impact on wellbeing: the physical aspects and conditions of a dwelling directly affect wellbeing, but housing conditions also have an indirect impact on wellbeing (Black et al., 2015; Commission for Architecture and the Built Environment, 2007, 2009; Stewart and Bushell, 2011).

Dwelling space is considered a fundamental human need as it provides protection from the elements. In addition to which housing offers space for familial interaction and is associated with privacy, personal identity, a sense of belonging and connection to where we live (Black et al., 2015). Poor housing conditions have a negative association with psychological wellbeing and undermine holistic wellbeing (Allen, 2008; Black et al., 2015; Commission for Architecture and the Built Environment, 2007, 2009).

In summary, public spaces within the wider context of the built environment are of key importance for wellbeing as they facilitate interaction and those bonds of social wellbeing which connect communities, enhancing community wellbeing. However, the way people use public spaces is dependent upon factors such as age, gender and ethnicity. Fears for personal safety, for example, may restrict access to public spaces and undermine wellbeing. Current perspectives on the impact of the built environment on wellbeing tend to concur that it is a combination of satisfaction with one's dwelling and one's opportunity to access public spaces, which is key for wellbeing.

One of the most significant problems facing those who seek to explicate the relationship between the built environment and wellbeing is that the built environment comprises of a huge number of individual elements such as street layout, road design, quantity and quality of green spaces. Future research may need to focus explicitly on which aspects of the built environment makes the most impact on wellbeing and why these particular elements or characteristics of the built environment are so important. Much of the research to date originates from single discipline studies. There exists therefore a need for multi-disciplinary research, which explores the impact of the built environment on wellbeing.

### 2.11: Temporality

The movement away from positioning wellbeing within quantifiable boundaries (Tomlinson and Kelly, 2013) encouraged the expansion of wellbeing research in

disciplines not traditionally associated with wellbeing research (Bilancini, 2012; Stoll et al., 2012). Social geographers Ballas and Dorling (2007), for example, began to measure the impact of major life events upon happiness. Research conducted by social geographer El-Shaarawi (2015) around the interplay between temporality, displacement and distress in refugees found that geographical displacement led to an altered experience of time. For the refugees in this study, the altered experience of time had a detrimental impact on envisaging the future and the affective forecasting of future feelings of wellbeing. This and other emerging research are using empirical research to explore the importance of temporal perspectives, that is, the way we think about the past, present, and future and the effect this has on our behaviour. This has been explored at a theoretical level over a considerable period of time, for example, Lewin (1942), Heidegger (1962) Zimbardo and Boyd (2008) and Durayappah (2010).

Zimbardo and Boyd (2008), for example, expanded the 'Zimbardo Time Perspective Inventory'. This uses self-reported data to identify an individual's orientations to the past, the present and the future. Zimbardo and Boyd's (2008) research indicated that temporal perspectives are an important individual characteristic that affects psychological and societal functioning.

However, until recently little qualitative empirical research had explored the role of temporality on wellbeing (Durayappah, 2010; Freund, 2010). Recent research has begun to address this deficit, this includes Schwanen et al. (2012), Galvin and Todres (2013), Guell et al. (2014), Åström et al. (2014), Zajenkowska et al. (2014), Griva et al. (2015) and El-Shaarawi (2015). In a forthcoming book, Wunderlich (2016) explores the importance of the 'everyday sense of time' by investigating the intersections between temporality, patterns of wellbeing and engagement in urban spaces. Literature such as this is part of a growing trend, which seeks to understand the influence temporality and temporal perspectives may exert on wellbeing. However, further empirical research is needed to substantiate theoretical notions of temporality and the intersections with wellbeing.

# 2.12: Component review summary

This review outlined the rationale for the organisation and sequence of topics covered in this domain approach to wellbeing. It documented the impact of domains of wellbeing widely recognised as key for personal wellbeing (Barnes et al., 2013; Marks and Shah, 2005; Ward et al., 2012). The review detailed the broad context of the research area under investigation, highlighting the relevant issues or debates which characterise these fields of research. At the end of each section, the current state of the research was summarised and gaps in knowledge and issues with the existing research were identified.

This literature review highlighted the paucity of multi-disciplinary approaches to wellbeing. Historically, wellbeing research has been undertaken within the confines of single disciplines. This research has therefore drawn upon a multi-disciplinary evidence base to contribute to wider understandings of wellbeing.

The research presented in the first part of this review highlighted the following gaps in knowledge:

- The extent to which coping mechanisms and adaptation are used to offset the impact of ageing and health issues on wellbeing.
- Unpacking the relationship between temporality and wellbeing.
- Unpacking the relationship between the natural environment and wellbeing
- Unpacking the relationship between the built environment and wellbeing.
- The extent to which wellbeing is associated with personal responsibility.

This study will build on existing research by explicating understandings of these gaps in knowledge. This will be achieved through a qualitative study conducted with older adults with LTCs. This will make an important contribution to knowledge as the absence of an established evidence base for these aspects of wellbeing constrains efforts to enhance citizen wellbeing through evidence-based policies.

The first part of this literature review has been structured to offer an appreciation of the complexity inherent within the subject area and provides the evidential base

for the empirical component of this study. The second part of this literature review, which follows, documents the chronological development of wellbeing research since the Local Government Act (2000). This provides the evidential base for the theoretical component of this study. It also seeks to present an appreciation of the changing narratives of wellbeing. These are espoused in critical questions such as 'What purpose does the wellbeing agenda serve?' (Gasper, 2004; Scott, 2012a, 2013; Scott and Bell, 2013) and 'Whose purpose does the wellbeing agenda serve?' (Atkinson and Joyce, 2011; Scott, 2012, 2013; Stenner and Taylor, 2008). These questions reflect increasing concerns regarding 1) agentic versus structural representations of wellbeing (Bache and Reardon, 2013; Bache et al., 2015; Seaford, 2011); 2) how wellbeing is represented to different sections of society (Atkinson and Joyce, 2011; Scott, 2012a, 2012c; Sointu, 2005; Taylor, 2011); and 3) how these representations impact on citizens (Atkinson and Joyce, 2011; Edwards and Imrie, 2008; Sointu, 2005; Scott, 2012c, 2013).

### 2.13: Chronological approach to wellbeing

Since the introduction of the Local Government Act (2000), several new terms such as 'wellbeing agenda', 'wellbeing politics' and the 'science of wellbeing' have emerged within the field of wellbeing (Carlisle and Hanlon, 2007; Edwards and Imrie, 2008; Shah, 2005). These terms reflect the upward trajectory of wellbeing within academia, the media and political rhetoric (Barnes et al., 2013; Scott, 2012c, 2014; Scott and Bell, 2013; Taylor, 2011). Whether this trajectory can be witnessed within lay narratives remains under-explored (Barnes et al., 2013; Hobbs and Sixsmith, 2010; Scott, 2012; Seedhouse, 1995; Ward et al., 2012).

Seedhouse (1995, 2002) raised concerns about wellbeing and a wellbeing agenda (WBA), which will be explored in detail in chapter three. Since Seedhouse (1995), a small but growing number of researchers suggest that wellbeing holds little currency for lay people (Eraurt and Whiting, 2008; Hobbs and Sixsmith, 2010; Mathews and Izquierdo, 2009). Wellbeing is a concept widely used and understood in academic/professional discourse, but it is not well understood and is rarely used

by lay people (Eraurt and Whiting, 2008; Mathews and Izquierdo, 2009; Scott, 2012c; Seedhouse, 1995).

## 2.13.1: The impact of the Local Government Act 2000

The Local Government Act (2000) gave local authorities the power to promote wellbeing as part of their duty to citizens (Department of the Environment, Transport and the Regions, 2001; Department for Communities and Local Government, 2008; Local Government Association, 2003). As such, it was the introduction of these new powers which stimulated efforts to coordinate the WBA (Gasper, 2004; Seaford, 2011; Scott, 2012c).

The UK national government sought to promote the economic, social and environmental wellbeing of communities through strategic plans for local wellbeing (Atkinson and Joyce, 2011; Atkinson et al., 2012; Scott, 2014; Scott and Bell, 2013). Scott (2011, 2014) proposed that the Local Government Act (2000) encouraged a marked increase in the WBA, in part because the Act provided little direction about what measures local councils might adopt to enhance citizen wellbeing. The arrival of local government wellbeing powers encouraged a surge in wellbeing research (Atkinson and Joyce, 2011; Atkinson et al., 2012; Scott, 2012a, 2014).

There was an expansion and extension of wellbeing research after the introduction of the Local Government Act (2000) (Abdallah et al., 2008; MacKerron, 2011; Nettle, 2005; O'Donnell et al., 2014; Ryan and Deci, 2001). In many respects, however, research proceeded along divergent lines with different disciplines devoting attention to largely separate lines of inquiry (Chavez et al., 2005). These lines of inquiry often made little or no attempt to provide an explicit definition of the concept (Bilancini, 2012; Galloway, 2006; Gasper, 2004b, 2010; Ryan and Deci, 2001; Seedhouse, 1995).

In addition, there was growing concern that research drew upon similar but not analogous concepts such as happiness, life satisfaction and QoL, which used different methodological approaches to the study of wellbeing (Edwards and Imrie, 2008; Gasper, 2010; King, 2007). All of these issues were perceived as undermining

efforts to homogenise findings, which could be applied across a range of policy settings.

In response to this, the Economic and Social Research Council series of academic seminars (2001, 2002) brought together experts from diverse backgrounds to explore wellbeing (Scott, 2012c). These seminars were important steps in encouraging new ways of thinking about wellbeing (Scott, 2012c). They also reflected a growing trend in moving beyond traditional methods and approaches when exploring wellbeing (Carlisle and Hanlon, 2007; Edwards and Imrie, 2008; Gasper, 2010; Stenner and Taylor, 2010).

# 2.13.2: The wellbeing agenda

The UK government was also involved in advancing and promoting wellbeing research (Donovan and Halpern, 2002; O'Donnell et al., 2014) as part of the wider WBA. In 2002, the UK Prime Minister's Strategy Unit explored the potential for promoting 'happiness policies' at a 'Life Satisfaction Seminar' in Whitehall. This prompted debates about the WBA and its role in society and governance (Furedi, 2008; King, 2007; Layard, 2005; Marks and Shah, 2004).

Chavez et al. (2005) provide a useful overview of the way in which the WBA developed in the light of two important QoL developments. The first was the conceptual introduction of SWB into empirical research. The second was the importance of recognising the role of utilising subjective assessments of wellbeing in a range of policy settings. This was particularly important for this thesis as it gave some recent historical contextualisation of the rationale for incorporating subjective accounts of wellbeing into the "Politics of wellbeing" (Edwards and Imrie, 2008:341). Before this development, it could be argued that the WBA in the UK was primarily concerned with drawing upon objective measures of wellbeing (Qizilbash, 1998, 2009). This paradigm shift led to recognition of the need for research to explore subjective accounts of wellbeing using qualitative methods (Clark, 2002b, 2005; Clark and Gough, 2005; O'Donnell et al., 2014).

# 2.13.3: The role of personal responsibility

Other developments in the field of wellbeing research included efforts to investigate wellbeing and the role of personal responsibility. Halpern et al. (2004), for example, explored the "Appropriate division of responsibility between the individual, community and state" (p.7) with regard to personal responsibility, behaviour change and the maximisation of wellbeing. They argued that personal responsibility and personal agency were fundamental for the attainment of eudaimonic wellbeing. Furthermore, Halpern et al. (2004) claimed a WBA which promoted personal responsibility in order to elicit behaviour and lifestyle changes would prove beneficial for the individual and the state.

These developments were perceived by some as part of a transition from former social welfare approaches to more neoliberal welfare agendas (Ferguson, 2007; Foucault, 1982, 1985, 1991; Froggert, 2002; Kingfisher, 2013; Rose, 1996, 1999). The appropriation of wellbeing to inculcate personal responsibility and behaviour change was increasingly evident in the policy documents of national government (Atkinson, 2011; Atkinson and Joyce, 2011; Edwards and Imrie, 2008; Fullager, 2002, 2009; Scott, 2011, 2014; Scott and Bell, 2012). This is discussed in detail in the following chapter.

Sointu (2005) offers a sociological interpretation of the appropriation of wellbeing within a wider WBA. Sointu (2005) deconstructed UK media text and representations of wellbeing over the 1985–2003 period to assess the appropriation and framing of wellbeing by the media within specific discourse. Sointu (2005) argued that wellbeing discourse has increasingly relied upon exploiting an individual's sense of self in order to facilitate a WBA. This places the responsibility for the creation and maintenance of wellbeing in the hands of the individual. In addition to which it "Emphasises proactive agency and self-responsibility as meaningful and normal" (p.255). A more in-depth sociological critique of the appropriation and mobilisation of wellbeing follows in chapter three.

### 2.13.4: Structure versus agency debate

The expansion of wellbeing research in disciplines not traditionally associated with wellbeing led to increasing concerns about the role of structure\* versus agency\* in efforts to enhance public and personal wellbeing (Ballas and Dorling, 2013; Scott and Bell, 2012; Wollny et al., 2010).

The Foresight Mental Capital and Wellbeing project (2008) identified a "Trend in recent years towards a model of public services based on greater levels of personal choice, active citizenship, personal responsibility, and 'co-production'" (p.18). This model of public services recognised the difficulty in determining the balance of responsibility for wellbeing between structural and agentic factors (Jordan, 2008; Scott, 2012a, 2012c; Tomlinson and Kelly, 2013). The role of the State\*/statutory agents\* in the promotion of wellbeing has become an issue of growing concern (Atkinson, 2011; Atkinson and Joyce, 2011; Scott, 2012a, 2012c; Scott and Bell, 2012). This will be considered in detail in chapter three.

The work of Edwards and Imrie (2008), for example, raised critical objections to the role of the WBA across UK society. Edwards and Imrie (2008) considered the wider implications of a WBA which sought to promote a 'politics of happiness' which had been embedded in government policy and discourse (Edwards and Imrie, 2008). Edwards and Imrie (2008) drew attention to the role of the New Economics Foundation (NEF), a prominent and prolific organisation concerned with wellbeing research. Edwards and Imrie (2008) argued that NEF's representation of wellbeing emphasised that wellbeing is attained by "happy, healthy, capable and engaged citizens" (p.338). Edwards and Imrie (2008) argued that this representation of wellbeing could only be aspired to by citizens without disabilities. Edwards and Imrie (2008) concluded that current government sponsored representations of wellbeing emphasised "Personality and character traits, and...individuated actions and (self) responsibilities" (Edwards and Imrie, 2008:340).

The use of expert voices\* such as NEF and Lord Layard to promote wellbeing have been in evidence in the UK from the middle of the decade (Shamir, 2008a).

However, the appointment of David Halpern, in 2010, as director of the government's new Behavioural Insight Team, informally known as The Nudge Unit' (Shamir, 2008a), indicated that the government was seeking to influence people's behaviour by "Using wellbeing as a lever to nudge people towards specific behaviour change under the subterfuge of promoting wellbeing" (Abdallah et al., 2012:12).

In 2011, the UK government commissioned the ONS to consult and develop measurements of wellbeing as part of the ongoing move away from using structural economic measures to assess social progress (Self et al., 2012). Thus began the systematic measurement of national wellbeing through agentic measures of wellbeing (Atkinson, 2013). This approach was pursued through publications such as the ONS National Debate "Measuring National Well-being" (2011a) and the "Initial investigation into Subjective Well-being Data from the ONS Opinions Survey" (ONS 2011b).

Agentic representations of wellbeing became a key element in mainstream efforts to promote personal responsibility within public sector models (Atkinson, 2011; Atkinson and Joyce, 2011; Lammy and Tyler, 2014; Scott, 2012a). Agentic representations of wellbeing and the WBA were disseminated through mechanisms such as the creation of the Wellbeing Czar\* Lord Layard, the work of the Behavioural Insight Team (formerly the Behaviour Change Unit) and the inclusion of wellbeing questions in ONS data sets (Abdallah et al., 2012). These advents meant that the WBA and its association with a PRA continued to flourish in the UK (Abdallah et al., 2012; Lammy and Tyler, 2014).

There were, however, dissenting voices about the role of the individual versus the state; for example, The Foresight Mental Capital and Wellbeing project (2008) noted, "A major issue will be to...determine the balance of responsibility for action – between the State, employers, families and individuals" (p.47). Similarly, the Organisation for Economic Cooperation and Development (OECD) (2011) explored the structure versus agency debate, concluding that structural determinants of wellbeing "Were also essential for positive wellbeing" (p.3).

The Care Act (2014) also considered the role of structural versus agentic factors on wellbeing. The Care Act (2014) introduced a general duty on local authorities to promote an individual's wellbeing (Care Act, 2014). The act identified the duty of local authorities to promote individual wellbeing and emphasised the role of governance in ensuring the wider determinants of wellbeing were addressed (DoH, 2014; Perkins and Hunter, 2014). The Care Act (2014) drew attention to the importance of structural factors and the role of governance by declaring that physical, mental and emotional wellbeing were the responsibility of statutory agencies. Oishi et al. (2015) contributed to the structure versus agency debate, calling for historical research to address important questions regarding the balance between agentic and structural representations of wellbeing and how these have changed over time.

### 2.13.5: Situating wellbeing within historical and linguistic contexts

Inglehart and Klingemann (1998) argued that historical factors play a powerful role in wellbeing, but receive insufficient attention in the majority of wellbeing research because they are complex and difficult to disentangle.

Kingfisher (2013) argued that wellbeing has deep historical roots which should be understood within historical contexts and noted the importance of considering how historical contextualisations may help to answer the 'how' and 'why' questions underlying the appropriation and mobilisation of wellbeing.

There has also been a growing recognition that linguistic analysis can help reveal how historical contexts impact on representations of complex concepts (DeWall et al., 2011). Similarly, historical and linguistic changes in concepts can be particularly important for research which aims to examine subjective states such as wellbeing (Mogilner et al., 2011; Oishi, 2012; Oishi et al., 2015). Research undertaken by Oishi et al. (2015) and McMahon (2006) in reviews of happiness/wellbeing offer new appreciations of the historical development of concepts.

In recent years, efforts have been made to research the historical and linguistic development of happiness such as Mogilner et al. (2011), McMahon (2006, 2006b, 2010), Oishi (2012) Oishi et al. (2015) and Peterson (2005). These works explored the shifting meaning of happiness and its allied concept of wellbeing. Oishi et al. (2015) posited, "That the linguistic analysis of the term happiness is critical to advance psychological theory and the scientific understanding of well-being" (p.2). In addition to which, it "Demonstrate the utility of a historical perspective on psychological science" (p.3). Mogilner et al. (2011), McMahon (2006, 2006b) and Oishi et al. (2015), whilst using wellbeing and happiness coterminously, recognised and highlighted the need for specific historical and linguistic contextualisations of wellbeing.

Kingfisher (2013), meanwhile, proposed that future research needs:

To situate wellbeing/happiness in historical context as there is nothing in our current conceptualisations or practices that is transparently self evident...that can be easily universalized across time and space. It is only by reflecting on historical and cultural contexts of our categories of analysis, investigation and action that we can gain insight into how our work relates to forms of governance and the interests that they serve.

(p.79)

### 2.13.6: Chronological review summary

The chronological review summarised the current state of UK research since the Local Government Act (2000), and provided an appreciation of the changing narratives and chronological development of wellbeing research.

This review has highlighted the impact of the Local Government Act (2000) on wellbeing research; this included the expansion and extension of wellbeing research into disciplines not previously associated with wellbeing. The review then considered some previously adumbrated aspects of wellbeing research. These included the WBA, and the expansion and extension of wellbeing research. This was followed by representations of wellbeing and the role of personal responsibility, which documented the appropriation and mobilisation of wellbeing. This

documented strong theoretical evidence that wellbeing has been mobilised within a wider PRA.

The structure versus agency debate highlighted critical objections to the role of the WBA through expert voices and mainstream efforts to promote wellbeing in association with personal responsibility. Finally, situating wellbeing within historical and linguistic contexts documented the need for historical and linguistic contextualisations of wellbeing.

# 2.13.7: Integration of theoretical and empirical reviews

The empirical review identified the need for research to explore lay understandings of the association between wellbeing and personal responsibility. This is supported by the chronological review, which identified the need for research to establish whether lay people associate wellbeing with personal responsibility. There is also a need for research which focuses efforts to support the theoretical evidence of the mobilisation of wellbeing with corroborative empirical evidence. In order to facilitate the integration of theoretical and empirical gaps in knowledge, this study intends to consider the theoretical and empirical evidence for the association between wellbeing and personal responsibility. This is of key importance, given that there is a perception that "Government input to improve wellbeing is less influential than the personal choices we all make" (Cabinet Analysis and Insight Team Blog, 2015) and that wellbeing has become well-established as 'an explicit policy goal' (Bache et al., 2015).

Taken together, this literature review identified that further research is required to explore agentic versus structural representations of wellbeing and the balance between government and personal responsibility for wellbeing (Ferguson, 2007; Naumova, 2014; Seaford, 2011; Scott, 2012a, 2012b, 2014). Chapter three seeks to address this by exploring how and why wellbeing has been utilised in association with personal responsibility.

As a whole, this literature review domain considered the evidence for understanding wellbeing. It also noted that wellbeing is influenced by factors such as historical and linguistic context, which have received little research attention to date. Chapter five addresses these gaps in knowledge by investigating how historical and linguistic developments have influenced modern conceptualisations and representations of wellbeing.

The following chapter presents the theoretical framework underpinning this study and offers an appreciation of the raft of sanctioned extensions to power, which facilitates the mobilisation of wellbeing.

### **Chapter 3: Theoretical Critiques of Wellbeing**

The following theoretical chapter documents the theoretical evidence that wellbeing has been mobilised within a PRA. It references wellbeing theories and theoretical positions in relation to the theoretical framework. This is contextualised in relation to sociological, health and political critiques of wellbeing.

This chapter explores and critiques how mechanisms of governance and discourse can be orchestrated to influence the way citizens think, act and conduct themselves. This is used to help frame an appreciation of the appropriation and mobilisation of wellbeing within a wider PRA.

This chapter begins by outlining the theoretical framework; then moves on to present a sociological critique, which considers the first theoretical premise that wellbeing has been mobilised to persuade people to accept responsibility for aspects of wellbeing which may be beyond their control.

Following this, health-orientated critiques are drawn together to consider the second theoretical premise that a WBA has been utilised to facilitate the reduction of government expenditure on health and welfare to an ageing population.

This is followed by a utilisation of the concepts of governmentality and governance to discuss the final theoretical premise that wellbeing has been mobilised as a socio-political tool to validate particular behaviours and lifestyles choices.

The final section outlines alternative theories regarding the mobilisation of wellbeing and concludes with a summary.

#### 3.1: Theoretical framework

The literature review established a strong evidence base to support claims that wellbeing has been mobilised within a PRA. The following chapter will present a theoretical framework which seeks to explicate understandings of how and why wellbeing has been mobilised within a PRA. Theoretical frameworks are important because they introduce and describe the theory that explains why the research problem under study exists (Swanson, 2013). The research problem anchors the study, forming the basis from which the theoretical framework is constructed (Jacard and Jacoby, 2010; Swanson, 2013; Swanson and Holton 3<sup>rd</sup>, 1997).

Theoretical frameworks also demonstrate understandings of theories and concepts relevant to the topic of research, "In addition to which [they explain] the meaning, nature and challenges associated with the phenomena" (Torraco, 1997:115).

Theoretical frameworks help define the specific viewpoint the researcher will take in analysing and interpreting the data (Jacard and Jacoby, 2010; Swanson, 2013). They also facilitate the construction of new knowledge by validating or challenging theoretical assumptions (Swanson, 2013). This enables the researcher to discern which assumptions or factors are important and which are not (Jacard and Jacoby, 2010). The following theoretical framework was constructed to explore those theoretical assumptions widely identified as underpinning the research problem, i.e. 'Why and how has wellbeing been mobilised within a PRA framework?'

The theoretical framework underpinning this thesis proposes that wellbeing is utilised:

- To obligate/persuade people to accept responsibility for aspects of their wellbeing, which may be beyond their control (Ahmed, 2010; Ferguson, 2007; Jordan, 2008; Kingfisher, 2013; Scott, 2012a, 2014; Seedhouse, 1995).
- To facilitate the reduction of government expenditure on health and welfare to an ageing population (Ferguson, 2007; Jordan, 2008; Miller, 2008; Kingfisher, 2013; Seedhouse, 1995).

 As a socio-political tool concerned with approving and validating particular behaviours and lifestyles choices (Ahmed, 2010; Bache and Reardon, 2011; Fullager, 2002, 2009; Sointu, 2005).

This particular framework is primarily concerned with exploring the mobilisation of wellbeing within the broader context of PRA. It has a multi-disciplinary focus, i.e. it draws from multiple disciplines. A predisposition towards certain theoretical positions is often linked to the dominant epistemological positioning adopted by those in particular disciplines (Carlisle and Hanlon, 2007; Chavez et al., 2005; Gasper, 2010; Poon and Cohen-Mansfield, 2011; Ryan and Deci, 2001). This tends towards a 'natural' inclination to perceive wellbeing within certain discipline boundaries, indeed the majority of wellbeing theories predominantly utilise a single discipline focus (Chavez et al., 2005; Gasper, 2010; Gough, 2005; Hubin, 2005). This perpetuates research which is constrained by boundaries which lay people may neither understand nor relate to (Clark and Gough, 2005; Gasper, 2010). These approaches provide a framework, internally consistent within a particular discipline but one-dimensional for those more concerned with multi-disciplinary perspectives (Galloway, 2006; Gasper, 2010).

Wellbeing research increasingly needs to accommodate the role wellbeing plays as a cross discipline concept by undertaking research which utilises multi-disciplinary approaches and theoretical understandings (Clark and Gough, 2005; Gasper, 2004b, 2010). Involving different disciplines helps to generate more holistic understandings of the topic under discussion (Chavez et al., 2005). In addition to which it is more appropriate for topics of research such as wellbeing, which fall in the intersections between disciplines (Gasper, 2010; Scott, 2012c).

This framework utilises multi-disciplinary perspectives, which recognise intersections between disciplines (Chavez et al., 2005).

### 3.2: Sociological critiques

The first premise of the theoretical framework proposed that wellbeing is utilised to obligate/persuade people to accept responsibility for aspects of wellbeing which may be beyond their control.

#### 3.2.1: Sense making

Contemporary modes of subjectivity such as those discussed by Charles Taylor (1989), Michel Foucault (1991), Antony Giddens (1991), and Nikolas Rose (1999) have tended to underpin sociological critiques of wellbeing (Kroll, 2011; Peck, 2013). These form a key sociological understanding of wellbeing as a neo-liberal tool in which notions of subjectivity shift from a sense of self pertaining to the duties and obligations laid down by external or higher authorities to a sense of self pertaining to the unique subjective experiences of the individual (Ahmed, 2010; Ferguson, 2007; Peck, 2013).

Sociological critiques of wellbeing have a propensity to argue that wellbeing has been mobilised to encourage citizens to accept responsibility for structural aspects of wellbeing, which may be beyond their control. This mobilisation has facilitated a shift from government-led welfare-based actions to improve wellbeing to a PRA in which citizens are expected to enhance wellbeing through their own actions (Ahmed, 2010; Kingwell, 1998; Peck, 2013; Scott, 2012c).

Perri's (2007) 'Sociological Theory of Wellbeing' (2007) conceptualises wellbeing in relation to the ways in which people make sense of their lives and social environment. Perri's (2007) theory seeks to negate the role of personal responsibility for wellbeing, instead asserting that the role of political responsibility has much more to contribute to the attainment of wellbeing. This contrasts with most mainstream wellbeing theories, which intimate that wellbeing is the personal responsibility of the individual attained through personal happiness or fulfilment (King, 2007; Gough, 2005; Nordbakke and Schwanen, 2014). Perri's (2007) neo-Durkheimian theory of wellbeing proposed that social institutions\* and forms of social organisation such as healthcare and education bodies underpin how people

make sense of wellbeing. Furthermore, Perri argued that the "Capabilities for wellbeing inhere in social relations and social organisation, not in the individual and still less in individually owned resources" (p.128). Perri (2007) argued that structural aspects of wellbeing are most important for wellbeing because of the way "They are organized, refracted, filtered, selected, articulated and even mobilized through sense-making" (p.139).

### 3.2.2: Challenging the wellbeing agenda

Sociologists challenge mainstream interpretations of the mobilisation of wellbeing, offering alternative critiques of the promotion of wellbeing (Baltatescu, 2007, unpublished PhD thesis; Kroll, 2011, unpublished PhD thesis; Sayer, 2011).

Fullager (2002, 2009), Furedi (2004, 2006), Sointu (2005) and Veenhoven (2004) encouraged sociological debate about the role of wellbeing. Fullager (2002, 2009) and Sointu (2005) in particular highlighted and questioned the mobilisation of wellbeing as part of wider efforts to persuade people to accept responsibility for aspects of their wellbeing beyond their control. They also contributed to the wider debate about who should be considered responsible for wellbeing. Some sociological approaches to wellbeing argue that the western neo-liberal style of governance seeks to shift responsibility for individual wellbeing from the state to the citizen (Fisher, 2008; Foucault, 1982; Fullager, 2002, 2009; Furedi, 2004, 2006; Sointu, 2005; Stacey, 1999).

This commodification of wellbeing (Bartram, 2012; Carlisle and Hanlon, 2007; Williams, 2000) facilitated representations of wellbeing primarily as the responsibility of the individual (Atkinson, 2011; Atkinson and Joyce, 2011; Fullager, 2005, 2009; Miller and Rose, 1990, 1995; Rose, 1989, 1996; Scott, 2012c; Sointu, 2005; Stacey, 1999).

Sointu (2005), for example, propounded a sociological critique of the mobilisation of wellbeing across society. Sointu's analysis documented the context within which wellbeing is discussed in two UK newspapers over the 1985–2003 period.

This critique utilised understandings of the links between subjectivity, personal responsibility and political governance and their influence on the prominence of wellbeing across multiple discourses.

Sointu proposed that in the early years (1985–86), wellbeing tended to be discussed and used in relation to the functioning of state institutions, especially with regard to the national economy. However, by the end of the 1980s wellbeing had become increasingly commercialised with mainstream media discourse representing wellbeing as attained through the actions of individuals rather than through the state. By the end of 2003, however, wellbeing discourse contained an implied injunction for the renunciation of certain types of unhealthy behaviour for more wholesome behaviour. This formed part of wider efforts to encourage citizens to accept responsibility for aspects of wellbeing which may be beyond their control. This injunction was framed simultaneously within consumerist language and located within the PRA (Sointu, 2005).

### 3.2.3: The role of government

Some sociological critiques of wellbeing centre around concerns that wellbeing research implies that governments have a duty (moral and/or ethical) to actively promote policies aimed at influencing citizen wellbeing (Bartram, 2011; Bjornskov, 2012; Duncan, 2010; Furedi, 2003, 2004; Veenhoven, 2008). This is founded on a long-standing concern of whether the pursuit of happiness or wellbeing is an appropriate ethical-political goal of governments and/or state agencies (Bjornskov, 2012; D. Bok, 2010; Booth, 2012; Briggs, 1970; Duncan, 2010; Furedi, 2004; Hobbes, 1651; Locke, 1689; Smith, 1776). These authors primarily associate wellbeing with the individual and concepts such as self-efficacy and agency. This provides the basis for their understanding that wellbeing is the responsibility of the individual not government.

Furedi (2004) was amongst the earliest sociologists who questioned the appropriation of wellbeing within political rhetoric (Baltatescu, 2007, unpublished PhD thesis). In an early sociological critique of the mobilisation of wellbeing, Furedi (2006) argued that the WBA was an "Over-intrusion of politics into the subjective

realm and an attempt to dress up social problems as emotional problems" (Furedi, 2006: online).

Since then, a growing number of sociologists have questioned the role of the state and state responsibility in wellbeing policies (Bartram, 2011; Duncan, 2010). A growing number of sociologists are uncertain whether the new science of happiness/wellbeing has a sufficient evidence base to be utilised by governments for the pursuit of state objectives (Bjornskov, 2012; Duncan, 2010; Furedi, 2004, 2006). Bartram (2011) argued that sociologists are also "Sceptical about the idea that we can offer meaningful suggestions for government actions more generally, as long as we lack a more systematic understanding of well-being as a foundation for "policy implications" (p.17).

For some sociologists, this raises concerns that the WBA frequently attempts to inform or advise what action should be undertaken within policy agendas to increase civic wellbeing. This approach has been criticised on two counts. The first for presuming that there is a direct and linear path from wellbeing findings to policy implementation (D. Bok, 2010; Cooper and Huppert, 2014). The second for the assumption that those versed in wellbeing research understand the wider issues connected with how public policy works (Bartram, 2011; Duncan, 2010).

### 3.2.4: Sociological contributions to the wellbeing debate

In the past five years, there has been an increase in wellbeing research undertaken by those using sociological approaches. This has been driven in part by concerns that qualitative research is marginalised across mainstream wellbeing research (Bartram, 2011; D. Bok, 2010; Newton, 2007; King, 2007; Scott, 2012). Qualitative research has a strong tradition within sociology and offers sociologists an opportunity to engage with and contribute to this particular domain of wellbeing research (Bartram, 2011; Kroll, 2011, unpublished PhD thesis). Sociologists have also expressed concerns that failing to participate in direct engagement with wellbeing research may undermine the ability of sociology to influence policy shaping, public discourse and political rhetoric (Baltatescu, 2007b; Bartram, 2011, 2012; Duncan, 2010). These are areas of significance and importance for

sociologists as they are concerned with societal processes and the wider societal determinants of wellbeing (Bartram, 2011; Duncan, 2010).

Concern about the lack of sociological input into wellbeing research has led to sociological studies such as those undertaken by Ahmed (2010), Frost and McClean (2014) and Kroll (2011). These have explored wellbeing in relation to sociological notions of power, discourse and structural factors and their impact on individual and societal wellbeing. Sociologists are increasingly considering the contribution sociology can make to wellbeing research. Bartram (2011) proposed that sociologists might be "Particularly well placed to investigate changes in how people talk about and act on happiness, analyzing those changes with conventional sociological tools to perceive inequalities, relations of power, etc. that might result from discourses and practices of happiness" (p.19). Sociological commentators and theorists are increasingly critical of a WBA which appears to prioritise agentic aspects of wellbeing over structural aspects (Baltatescu, 2007, unpublished PhD thesis; Kroll, 2011, unpublished PhD thesis; Perri, 2007).

In summary, it is apparent that sociological critiques of wellbeing have explored the underlying use of social norms within modern wellbeing discourse. These have been considered in relation to issues of governance, power and responsibility. They have also raised concerns about the role of government and the promotion of a WBA. Mainstream and traditional sociological critiques of wellbeing tend to suggest that wellbeing has been utilised to encourage citizens to accept responsibility for aspects of their wellbeing which may be beyond their control. This is achieved by negating the role of structural determinants of wellbeing in direct contrast to previous periods in recent history "When the UK government via the welfare state was the primary agent in the creation and maintenance of health and wellbeing" (Sointu 2005:255-6). This is indicative of a style of governance which seeks to shift responsibility for individual wellbeing from the state to the citizen. The following section will explore whether a WBA, which negates the role of structural determinants of wellbeing, is also associated with reducing health and welfare expenditure.

### 3.3: Health critiques

The second premise of the theoretical framework proposed that wellbeing is utilised to facilitate the reduction of government expenditure on health and welfare to an ageing population.

With the rise of neoliberalism, wellbeing in association with health became increasingly perceived as a consumable commodity (Gould and Gould, 2001; Sointu, 2005; Stacey, 1997). This led to government rhetoric and discourse which represented health and wellbeing as the responsibility of the citizen (Carlisle and Hanlon, 2007; Seedhouse, 2005; Williams, 2000). Health approaches to wellbeing have until recently assumed that physical health and functioning is a fundamental condition for wellbeing (Buchanan, 2000; Scott and Bell, 2013; Shah, 2004). A small number of health critics began to argue that government representations of wellbeing use these assumptions to promote agendas which prioritise individual endeavour over state responsibility (Buchanan, 2000; Seedhouse, 1995; Stacey, 1999; Williams, 2000).

### 3.3.1: Concerns about the role of wellbeing

Seedhouse (1994) was amongst the earliest health critics to raise concerns over the role of wellbeing (Carlisle and Hanlon, 2007). Seedhouse claimed theories of wellbeing should not be mobilised within health contexts because there is an implicit assumption that the promotion of wellbeing equates to the promotion of health. Seedhouse (1995) criticised the emerging WBA for promoting wellbeing as a fashionable replacement term for health. Seedhouse (1995) suggested part of the rationale for the mobilisation of wellbeing is its conceptually ambiguity, which has helped the language of wellbeing replace the language of health. This has helped revolutionise the way government thinks about and plans for health and wellbeing. This in turn stimulated an emerging WBA, in which some conceptualisations are preferred over others (Seedhouse, 1995). For example, white middle-class academic conceptualisations may be authenticated whilst others conceptualisations underpinned the emerging WBA.

This WBA could then be utilised to facilitate the reduction of government spending on health bills to an ageing population by approving and validating behaviours and lifestyles choices which endorse healthy living (Griffin, 2010; Nordbakke and Schwanen, 2014; Seedhouse, 1995). Thus, the WBA is less concerned with improving health practices and more concerned with reducing health expenditure (Fullager, 2009; Nordbakke and Schwanen, 2014; Seedhouse, 1995). The WBA also encouraged citizens to accept responsibility for aspects of wellbeing beyond their control (Ahmed, 2010; Ferguson, 2007; Nordbakke and Schwanen, 2014; Peck, 2013; Seedhouse, 1995).

As outlined in chapter two, alternative interpretations of the role of wellbeing emerged after the Local Government Act (2000). At a theoretical level, health approaches to wellbeing increasingly perceived wellbeing as a pragmatic tool which could be utilised to influence risk-taking behaviours and poor lifestyles (Cooper, 2014; Griffin, 2010; Ogden, 2012; Ryff and Singer, 2000). These approaches tended to emphasise multidisciplinary/interdisciplinary understandings of wellbeing and conferred primacy to structural representations of wellbeing (Carlisle and Hanlon 2007; Cooper, 2014; Ogden, 2012).

#### 3.3.2: Consumer culture

Critics from health backgrounds increasingly began to question whether mainstream wellbeing research focused too much on "Maximising individual happiness, rather than developing the kind of societies in which all humans can flourish" (Carlisle and Hanlon, 2007:266). Some public health critics suggested consumer culture has been mobilised to promote a representation of wellbeing which fits the Western neoliberal model, i.e. the achievement of wellbeing through consumerism (Carlisle and Hanlon, 2007, 2008; Crawshaw, 2008; Williams, 2000).

The notion of individual responsibility created through a consumer culture permeated healthcare approaches to wellbeing (Rose, 1999; Sointu, 2005; Stacey, 1997, 2000). Stacey (1997) coined the term 'self-health' to make explicit the links between self-responsibility, consumerism and health. Sointu (2005) argued that

health approaches have been extended to "Incorporate the notion of wellbeing to such a degree that we now perceive our ability to be healthy as part of our identity to be bought and bargained, for example, through the use of league tables for hospitals" (p.268).

The focus on consumer culture, which prioritised the agentic determinants of wellbeing, has been vigorously challenged in recent years. The expansion of dominant mainstream discursive approaches within health such as 'Empowerment Approaches' have contributed to the promotion of representations of wellbeing which focus on agentic determinants of wellbeing (Crawshaw, 2008; Scott and Bell, 2013; South and Woodall, 2010). Empowerment approaches have tended to overlook structural determinants of wellbeing and shifted the emphasis of responsibility from the healthcare system to the patient (Edwards and Imrie, 2008; Scott and Bell, 2013).

Critics of empowerment approaches suggest that agentic representation of wellbeing may be more concerned with the reduction of health and welfare expenditure than with health issues (Berry, 2014; Buchanan, 2000; Crawshaw, 2008; Edwards and Imrie, 2008; Froggett, 2002). This appraisal has received support from those interested in the role and function of neo-liberalism. A growing cadre of authors experienced in politics and governance have questioned the role of wellbeing and personal responsibility within modern society (Miller and Rose, 2008; Peck, 2013).

### 3.3.3: Wellbeing and its relationship to welfare

As documented in chapter two, there is growing concern that personal responsibility is connected with neoliberal efforts to make citizens self-examining, self-governing and autonomous (Ferguson, 2007; Miller and Rose, 2008; Peck, 2013). These facilitate the piecemeal dismantling of welfare, thereby reducing overburdened welfare systems through discursive practices and policies which promote self-sufficiency and empowerment (Ferguson, 2007; Kingfisher, 2013).

Recent years have witnessed a growing debate about wellbeing and its relationship to welfare (Jordan, 2008; Searle, 2008; Taylor, 2011). Some suggested the language of wellbeing was replacing the language of welfare (Dean, 2010; Stenner and Taylor, 2008; Taylor, 2011). This has caused some degree of concern that wellbeing has been mobilised to restrict welfare services and the social provisions of the welfare state (Dean, 2010; Taylor, 2011). In fact, it has been suggested "That a preoccupation with individual wellbeing alone has the potential to detract from the continued importance of collective welfare and the social provision of the material conditions in which much individual wellbeing is lived and felt" (Taylor, 2011:779). Furthermore, "Wellbeing is increasingly supplanting welfare as a central political goal for social and public policy" (Taylor, 2011:777). This replacement of state maintained welfare with individual maintained wellbeing facilitated the removal of state financing from aspects of the welfare system which are increasingly perceived as the responsibility of the individual (Fleuret and Atkinson, 2007; Stenner et al., 2009; Taylor, 2011)

In recent years, Seedhouse's (1994, 1995) concerns of a WBA and its attendant motivations to reduce government health and welfare expenditure have been reignited. Wright (2014), for example, argued that "The Wellbeing agenda contributes to cost-benefit approaches to health provision that do indeed evaluate various psychological complaints not in terms of subjective suffering but of the loss to the economy through sick leave" (p.797).

Those critical of the WBA have once again begun to question the commodification of wellbeing which promotes agentic representations of wellbeing within health contexts (Barnes et al., 2013; Edwards and Imrie, 2008; South and Woodall, 2010; Ward et al., 2012). Critics suggested that the prevailing "Focus on well-being which reinforces personal responsibility... may present a distraction from efforts to address enduring structural inequalities in health" (Carlisle and Hanlon, 2008:263).

# 3.3.4: Blurring the boundaries between wellbeing and QoL research

At a theoretical level wellbeing in health disciplines remains relatively under-explored, despite health being widely recognised as one of the key drivers of wellbeing (Allen, 2008; Abdallah and Shah, 2012; Berry, 2014; Galloway, 2006; Hausman, 2015). Health approaches to wellbeing have often relied upon findings from QoL research. Examples include, providing evidence of an association between physical health, optimal functioning and SWB (Nussbaum and Sen, 1993; Scrivan and Garmen, 2005; Vilhena et al., 2014). Whilst health is recognised as fundamentally important for QoL, Dodge et al. (2012) and Vilhena et al. (2014) cautioned researchers against applying health-related QoL research to wellbeing. Vilhena et al. (2014) and Dodge et al. (2012) proposed that whilst QoL is related to wellbeing, it should not be seen as synonymous with wellbeing.

Lhussier (2009), however, suggested that regardless of whether wellbeing and QoL are different concepts, researchers in these fields should blur the boundaries between these concepts in order to revitalise and progress research. Lhussier (2009) argued that keeping within disciplinary boundaries has stymied research and undermined efforts to operationalise QoL/wellbeing. Blurring these boundaries will release QoL research from current constraints and facilitate the development of alternative approaches to research theory and practice (Lhussier, 2006, 2009).

Lhussier (2006, 2009) offers an alternative approach to understanding QoL/wellbeing within health contexts. Lhussier highlighted a growing recognition in QoL research that empowerment discourse has promoted an environment in which the patient is held responsible for their LTCs (Lhussier, 2006; South and Woodall, 2010). "This discourse simultaneously shifts the emphasis of responsibility from the healthcare system to the patient which is financially advantageous for the health services" (Lhussier, 2006:162).

Lhussier (2006) investigated discourse and representations of QoL/wellbeing within the NSF for LTCs (DoH, 2005), which is "Expressly concerned with improving patient QoL and providing services to support independent living" (DoH, 2005:5).

Within the NSF, enhancing QoL and independence are promoted as fundamental aspirations for patient with LTCs (Lhussier, 2006). Lhussier (2006) argued that this approach centres on patients being forced to accept responsibility for improving their QoL through their care and management of their LTCs. This "Biomedicalisation utilises discourse of patient empowerment and independence to facilitate patient management of long-term conditions and reduced healthcare costs" (p.219).

Lhussier (2006, 2009) offered evidence that patients with LTCs can circumvent prevailing representations of QoL promulgated by healthcare professionals. Patients are capable of constructing conceptualisations of QoL, which are appropriate for them and their circumstances. These alternative conceptualisations of QoL/wellbeing can be utilised within alternative health approaches to the operationalisation of QoL/wellbeing (Lhussier, 2006, 2009).

In summary, this section has provided evidence that health is generally recognised as one of the key drivers of wellbeing. Indeed, some have argued that health has the potential to act as a bridge between competing wellbeing theories. Despite this, health approaches and theories in particular remain a relatively underexplored area of research. Health approaches to wellbeing have often relied upon findings from QoL research. However, more recently researchers have sought alternative approaches to understanding QoL/wellbeing within health contexts. Discursive approaches, for example, have suggested that empowerment discourse adopted within health approaches to wellbeing shifts the emphasis of responsibility from the healthcare system to the patient. Within the past decade, mainstream health approaches to wellbeing have conceptualised wellbeing as a commodity. These approaches, however, have been vigorously challenged in recent years.

The "Creative struggles between competing interests and discourses" (Scott and Bell, 2013:537) have had an important part to play in health approaches to wellbeing. At a theoretical level, health critics of the WBA originally tended to confer primacy to structural representations of wellbeing. However, more recent mainstream health approaches to wellbeing have mobilised representations of

wellbeing which fit Western neoliberal models of consumerism. Dominant discourse and mainstream discursive approaches within health such as empowerment approaches may have contributed to the promotion of conceptualisations of wellbeing, which focus on the agentic determinants of wellbeing (South and Woodall, 2010). These have tended to overlook structural determinants of wellbeing. Critics of these approaches suggest that this commodification of wellbeing is likely to isolate certain sections of society, particularly those with health issues (Berry, 2014; Buchanan, 2000; Crawshaw, 2008; Edwards and Imrie, 2008).

## 3.4: Where wellbeing fits within citizenry and governance

### 3.4.1: Governmentality

The final premise of the theoretical framework proposed that wellbeing is utilised as a socio-political tool concerned with approving and validating particular behaviours and lifestyles choices. Using notions of governance will help frame understandings of why a WBA has been mobilised within UK government legislation, discourse and rhetoric.

'Governmentality' is a concept originally conceived by Michel Foucault (1901–1984) (Jones 1998). Governmentality offers a framework for analysis of political discourse and can be understood as "a 'guideline' for the analysis...of historical reconstructions embracing a period starting from Ancient Greece right through to modern neo-liberalism" (Foucault, 1983; 1984). Governmentality forms an important strand of this thesis because "It provides a rationale for the ways in which governments assemble, retain and employ power through mechanisms which utilises modern technology, apparatus and stratagems to control and govern the thought processes and behaviours which regulates modern societies" (Jones 1998:965).

Governmentality is a complex concept, which alludes to different notions of governance. Foucault (1991) variously identified that it referred to the processes by which government seeks to create a citizenry that adheres to its conceptualisation of a governable society (Foucault, 1991). However, it could also be considered as the 'art of government' (Foucault, 1991), which utilises techniques and stratagems to facilitate a governable society (Foucault, 2002). Governmentality is a tenet primarily associated with westernised liberal democracies, i.e. neo-liberalism (Rose, 1999). Foucault uses governmentality to explore how governments sanction extensions to their power through social control via the mediums of state institutions such as police authorities, schools, hospitals, criminal justice system, etc, 'Czars'\* and public rhetoric and discourses.

### 3.4.2: Mechanisms of governance

More established neo-liberal governments use the political-media complex, official discourse, historical and cultural ties, propaganda and indoctrination as tools to create a population which ostensibly controls its own behaviours (Rose, 1989). The political-media complex refers to "The mutually beneficial relationships which have been established between a state's political agents, [and] its special interest groups for example expert professionals" (Semetko and Valenburg, 2000:96). The political-media complex endorses the socio-political control used by governments to reinforce governance by engendering a society in which citizens are expected to act as instruments of their own control (Miller and Rose, 1990; Sointu, 2005; Stacey, 1997, 2000).

The following example demonstrates how mechanisms of governance can be manipulated to influence citizen behaviour.

Lupton (1995, 1999b) and Petersen and Lupton (1996) explored the utilisation of health discourses in relation to the concept of 'risk' to demonstrate how governance and discourse can be orchestrated to influence citizen perceptions and behaviour (Baltrescu, 2007b). Lupton (1999) identified changes in the meaning of risk and documented how social perceptions of risk have shifted over time. Lupton (1999b) suggested that in the Middle Ages, the concept of risk did not include the

idea of human fault and/or responsibility; risk was beyond an individual's control. However, this began to change during the Industrial Revolution and from the 19<sup>th</sup> century onwards, risk was no longer associated exclusively with natural phenomenon but with human actions and behaviour. Perceptions of risk continued to shift until by the late 20<sup>th</sup> century risk was less associated with natural phenomenon or divine retribution and more closely associated with rational thought, human efforts, control and manipulation of one's environment (Lupton, 1999b). Lupton (1999b) proposed that the concept and language of risk in the late 20<sup>th</sup> century, which flourished in 'expert discourses'\*, may be grounded in perceptions that selected risks "have an important ontological status in our understandings of selfhood and the social and material worlds" (p.14).

Sociologists have applied Lupton's (1999b) proposal to understand and help explain why the concept and language of wellbeing has flourished in expert discourses (Fullager, 2002, 2009; Sointu, 2005). Expert discourse has become a substantial force in the promotion of health and wellbeing (Bergdolt, 2008; Hughes, 1988; Scott, 2012c). Similarly, "Self responsibility has become the cultural norm used to target a notion of an idealised individual based on health behaviours under the auspices of the sphere of wellbeing" (Sointu, 2005:262).

Fullager (2002, 2009) utilised the work of Lupton in her study of two Australian campaigns, which analysed "The way discourses of leisure and healthy lifestyles have been produced through the governmental objectives of health policy and promotion aimed at the body" (Fullager, 2002:69). Fullager (2002) focused on the process of normalising risk reduction practices (Lupton, 1995, 1999a) to examine how leisure discourse, which utilised risk, has been mobilised as a governmental strategy.

Fullager (2002) identified some significant shifts in the way Australian health policy had problematised the role of leisure, recreation and physical activity in relation to the identification of lifestyle disease risks to individual and social wellbeing.

Fullager's (2002) findings indicated that health promotion had mobilised agentic

forces through particular discourses of leisure, technologies of the self, management strategies and normative lifestyle practices.

Fullager (2002, 2009) argued that governmental power was exercised through a form of rationality in which particular truths and logics about healthy living were identified implicitly via the promotion of self-examination, self-care and self-improvement (Lupton, 1995; Dean, 1999a; Rose, 1999). These discourses were then used to inculcate a sense of personal responsibility for individual behaviour and lifestyle choices (Fullager, 2002, 2009). This sense of personal responsibility has increasingly been appropriated within a wider PRA.

### 3.4.3: Governance, responsibilisation and the WBA

The trend towards a PRA and a responsibilisation process\* within neo-liberalism is perceived by some as a regressive aspect of the shift towards governance of the self in modern societies (Peck, 2013; Rose, 1999; Taylor, 1989). The PRA acts as a sociopolitical tool, which promotes particular behaviours and lifestyles choices through notions of empowerment, autonomy and control (Ferguson, 2007; Kingfisher, 2013; Micklethwait and Woolridge, 2003). Governments, state institutions and the political-media complex promote a responsibilisation process, which persuades societies and individuals to take personal responsibility for their actions and behaviours (Ahmed, 2010; Micklethwait and Woolridge, 2003; Miller and Rose, 2008; Rose, 1999).

The responsibilisation process is covertly powerful, as it does not demand compliance with official regulations or absolute authority (Micklethwait and Woolridge, 2003). Instead, it instils values in-line with governing socio-political ethos and is rooted in the prevailing cultural and historical positioning of individualistic motivation for action (Micklethwait and Woolridge, 2003; Shamir, 2008b). Research suggests the responsibilisation process has been utilised in the mobilisation of a WBA, particularly in its association with health (Seedhouse, 1994, 1995; Sointu, 2005; Stacey, 1997).

Alternative interpretations of the responsibilisation process suggest it is a progressive aspect of the principles of neo-liberalism which ensures certain standards of behaviour are maintained by the least responsible sections of the business and government communities (Micklethwait and Woolridge, 2003; Shamir, 2008a). This aspect of neo-liberalism lays the foundations for how neo-liberalism can benefit wider populations (Auld et al., 2008; Cashore et al., 2004; Shamir, 2008b), keeping it alive as a major political force by invigorating and infusing it with new realities (Shamir, 2008b).

Traditionally the relationship between responsibilisation and neo-liberalism has been seen as the politico-economic organisation of state and citizens, based upon historical and cultural positioning (Moloney, 2006; Rose, 1999; Shamir, 2008a). The PRA underpins neoliberalism, which favours indirect governance rather than direct interventionism (Ahmed, 2010; Ferguson, 2007; Rose, 1999). When considered in relation to the WBA this can be exemplified by "The commandment to 'be happy' which amounts to a form of insidious social control, in which we are encouraged to look inwards (and blame ourselves) for the causes of our trouble" (Moloney, 2006:27).

The mobilisation of the PRA and the WBA by neo-liberal governments contributes to discourse and rhetoric which seeks to manage and control citizens and the increasing demands on national resources (Rose, 1999; Taylor, 2011). This has been achieved in part by employing expert voices\* from economic backgrounds (Ahmed, 2010; Edwards and Imrie, 2008; King, 2007). These expert voices add their weight and influence behind government wellbeing rhetoric, discourse and policy (Ahmed, 2010; Edwards and Imrie, 2008).

The UK government mobilisation of wellbeing through a PRA has utilised the expert voice of one of the main proponents of wellbeing (Lord Layard) who was appointed as the 'Wellbeing Czar' in 2008 (Ahmed, 2010). Other neo-liberal nations such as Australia, Germany and the US have also extended the ambit of the PMC and the promotion of wellbeing by bringing on board celebrated and ostensibly impartial experts (Ahmed, 2010; Miller, 2008). For example, pre-eminent stalwarts of the

WBA such as Stiglitz, Sen and Fitoussi (2011) were employed to orchestrate French public opinion around wellbeing (Taylor, 2011).

Neo-liberal governments increasingly apply the happiness principle to influence thoughts and actions whilst simultaneously shifting issues formerly conceived of as public into the private domain (Furedi, 2004). This enables governments to shift the blame for societal problems onto individuals rather than confront those wider determinants, which are beyond the control of individuals (Seedhouse, 1995; Taylor, 2011).

Current representations of wellbeing have been used extensively through government discourse and rhetoric. These have been criticised for being more concerned with approving and validating particular behaviours and lifestyles choices in line with consumerist ethics than with establishing positive constructions of responsible citizens and holistic communities (Edwards and Imrie, 2008; Ferguson, 2007; Kingfisher, 2013).

"Many have questioned the role of the WBA within the political system as an attempt to legitimise socio-economic control of the population particularly those members of society who fail to conform to the idealised image of the responsible citizen" (Edwards and Imrie, 2008:346). In so doing, individual responsibility and a private rather than public commitment to personal wellbeing takes precedence over the functions of state responsibility (Barnes et al., 2013; Fleuret and Atkinson, 2007; Furedi, 2004; Taylor, 2011).

### 3.4.4: Alternative accounts of the role of wellbeing

A number of alternative perceptions of the role of wellbeing as a socio-political tool have been proposed. Jordan (2008), for example, argued that at a theoretical level wellbeing allows us to consider 'social value' as underpinning societal progression. This is contrasted with economic theories of societal progression, which promote 'utility' as a concept associated with welfare (Jordan, 2008; King, 2007; Qizilbash, 1998; Taylor, 2011). Dean (2010) argued that "The advantage of wellbeing as a term is that it can turn our attention to the positive aspects of social policy, as opposed

to the negative aspects relating to social problems" (p.100). This last point has been widely posited by wellbeing proponents such as the new economic foundation (NEF) and those associated with the positive psychology movement (Seligman, 2011a,b).

Wood and Newton (2005), meanwhile, proposed that a WBA which replaced notions of welfare with wellbeing facilitated social development and social policy. The influential Wellbeing and Development (WeD) group draws on universal theories of human need (Doyal and Gough,1984, 1991) to argue that wellbeing as a socio-political tool is more relevant than welfare because it encompasses wider notions of what is important for human development (Taylor, 2011).

### 3.4.5: Alternative theories for the mobilisation of wellbeing

A number of alternative theories have been proposed regarding the mobilisation of wellbeing in the UK. Perhaps one of the most widely promulgated suggested that wellbeing has been mobilised as an alternative indicator of societal advancement other than traditional measures such as Gross Domestic Product (Aked et al., 2010; Marks, 2006a; Marks and Shah, 2004; Shah, 2005a). Easterlin (2008) posited that until recently, "Economists have tended to conceptualise wellbeing as pertaining to objective proxy measures of life satisfaction e.g. Gross Domestic Product per capita, life expectancy, educational attainment" (p.6). These conceptualisations have come under increasing criticism for their tendency to infer that all things being equal, people will choose those things which enhance their wellbeing, and thus their choices act as proxies for what can be perceived as constituting their wellbeing (see Ryan and Deci, 2001; Gasper, 2004b).

Increasingly, those within economic disciplines have spearheaded a growing interest in wellbeing as an indicator of societal advancement (Easterlin, 2003; Frey, 2008; Layard 2005; Marks et al., 2006a). This has been perceived as a response to research, which suggests that despite increasing levels of Gross Domestic Product per capita, life expectancy and educational attainment, citizens' life satisfaction had reached a plateau (Easterlin, 1976).

Some suggest that wellbeing has been mobilised as part of wider efforts to emphasise the non-material determinants of wellbeing in order to enhance the environmental sustainability agenda (Aked et al., 2010; Marks et al., 2006b; Morphet, 2008; Stern, 2006). The environmental sustainability agenda first came to global prominence when the United Nations Conference on Environment and Development (1992) introduced the 'Local Agenda 21' (Bosselmann et al., 2008). This promoted the role of government in association with local communities in enhancing environmental sustainability (Bosselmann et al., 2008).

The environmental sustainability agenda was subsequently redefined in the UK as the broader sustainability agenda (Bosselmann et al., 2008; Morphet, 2008). Since then the Local Government Act (2000), which granted wellbeing powers to local governments in the UK, has been used to promote a sustainability agenda (Bosselmann et al., 2008; Morphet, 2008). Indeed, Morphet (2008) suggested that in the UK local governments have sought to mobilise wellbeing powers in order to position themselves at the forefront of the sustainability agenda. Bosselmann et al. (2008), however, argued that despite localised efforts to mobilise wellbeing, local government wellbeing powers have proved largely ineffectual. Despite Bosselmann et al.'s (2008) claims, there is some support for the claim that wellbeing has been mobilised within the wider environmental sustainability agenda (Aked et al., 2010; Marks et al., 2006b; Morphet, 2008; Stern, 2006).

The Sustainability Development Strategy 'Securing the Future' (2005), for example, linked wellbeing explicitly with environmental sustainability policies (Aked et al., 2010). Aked et al. (2010) suggested that placing wellbeing at the heart of the sustainable development agenda recognises that the WBA has an important role to play in environmental sustainability. Furthermore Aked et al. (2010) proposed that the wellbeing powers bequeathed to local government enable them to mobilise wellbeing as part of wider efforts to encourage citizens to initiate behaviour change and lifestyle which will help alleviate some local environmental issues.

Stuff (2016) has also contributed to a growing debate about the WBA by suggesting that wellbeing can and has been used to facilitate work productivity. This

complements the work of Sointu (2005), who proposed that language used in the media represented wellbeing within work productivity discourse as part of wider efforts to make citizens responsible for aspects of wellbeing formerly the responsibility of organisations.

### 3.4.6: **Summary**

This chapter has investigated the "Developing wellbeing agenda, adding a critical and dissonant voice to the current conceptualisation of where wellbeing fits within citizenry and governance" (Stenner and Taylor, 2008:435). It has documented diverse theoretical approaches and alternative perceptions of the mobilisation of wellbeing. The accumulated evidence documented here suggests there is increasing support for the theoretical premises underpinning this study.

There is increasing theoretical evidence that wellbeing is one of a raft of measures utilised to reduce government expenditure, promote behaviour modification and inculcate personal responsibility. Some argue that wellbeing has been re-cast as a civic duty, moulded to satisfy the requirements of governmentality. The evidence presented here suggests the current WBA is helping to shape how the public perceive and conceptualise wellbeing. This is promulgated through its use of:

- Legislation, for example, the Local Government Act (2000).
- 'Expert voices', for example, the Wellbeing Czar Lord Layard.
- Agentic representations and publications of wellbeing, for example, 'Five Ways to Wellbeing' (Aked et al., 2008).
- Research, for example, the Whitehall Wellbeing Working Group (W37G)
   (2005).
- Consultation, for example, 'How should we measure wellbeing' (ONS, 2010).

This chapter has provided evidence to support the theoretical premise of this thesis that wellbeing has been appropriated and mobilised within governance frameworks. This has chiefly addressed the question of why wellbeing has been appropriated and mobilised.

The following chapter describes the methodology and methods used in this study.

### **Chapter 4: Methodology and Methods Chapter**

This chapter discusses the epistemological perspectives, paradigms, data collection, analysis and the research methods which underpin this study. The chapter also incorporates an overview of the participant and co-researchers' selection process. It presents the participants' profile, discusses the co-researchers' involvement in the research process and concludes by summarising the reflective process and ethical considerations.

### 4.1: Background to study

As detailed in chapter one, this PhD thesis originated from a participatory research project. During the course of this project, service users elected to refer to themselves as co-researchers/co-advisors. They worked alongside the researcher throughout the research process. Two of the co-researchers/co-advisors (Alison and Jim) agreed to take part in this thesis as co-researchers. Henceforth, when the term co-researchers is used, it will refer to Alison and Jim.

### 4.2: Methodological approach

This section will set forth the theoretical underpinnings which inform the methodology utilised in this PhD thesis. It will include discussions around epistemological perspectives, paradigms, data collection, analysis and research methods.

The distinction between epistemology, methodology and method can be fluid and hard to categorise (Cutcliffe and Harder, 2012; Glasper and Rees, 2012). Harding (1986) defined epistemology as the philosophical perspective which facilitates the decision-making process. This concerns what type of knowledge is possible and what can be known. Method and methodology are terms which are often used interchangeably; however, this can be confusing as they refer to different aspects of the research process (Cutcliffe and Harder, 2012; Glasper and Rees, 2012).

Cutcliffe and Harder (2012) propose that methodology concerns the process whereby the researcher describes, explains and justifies the methods used but not the actual methods themselves. Cutcliffe and Harder (2012) define methods as the specific research procedures, tools and techniques of research (Cutcliffe and Harder, 2012:4). It is generally accepted that the methodology justifies and is followed by the method (Carter and Little, 2007; Cutcliffe and Harder, 2012; Glasper and Rees, 2012; Harding, 1986).

### 4.2.1: Epistemology

Epistemology refers to the theory of knowledge. It "Provides much of the justification for particular methodologies i.e., the aim, function, and assumptions of method. Our epistemological positioning within the context of our research informs our theoretical framework, which in turn informs how we 'read' and relate to the text we explore" (Schwandt, 2001:71). The primary epistemological perspective is framed within a social constructionist approach and takes into account the doctrine of historicism.

The theoretical framework has guided the respective positions adopted within this thesis. At a theoretical level, this study is underpinned by social constructionism and historicism. Social constructionism plays an important role within this thesis and is congruent with the theoretical and empirical components of this study.

The theoretical framework is concerned with exploring lay narratives in order to explicate understandings of how and why wellbeing has been mobilised within a PRA. It was important therefore to adopt an approach which facilitates the presentation of lay narratives and their representations of knowledge and understanding (Andrews, 2012; Morley et al., 2014). Social constructionism draws attention to the fact that human experience, including perception, is mediated historically, culturally and linguistically (Willig, 2001). It therefore offers the researcher a paradigm consistent with the theoretical framework.

Historicism gives an indication of my postionality with regard to the historical/linguistic chapter and the importance of the utilisation of historical

analysis and contextualisation within the context of this thesis. The theoretical framework enables the researcher to discern which assumptions or factors are important and which are not (Jacard and Jacoby, 2010). Historicism facilitates the researcher's perspective in understanding the theoretical premises within a historical context. These epistemological positions are linked to principles that value subjective experiences and allow for a diverse appreciation of ways of knowing.

### 4.2.2: Paradigm

The paradigm underpinning this study utilises social constructionism. A paradigm is a set of assumptions, concepts, values, and practices that constitutes a way of viewing reality for the community that shares them, especially in an intellectual discipline (Easterby-Smith et al., 1991; Göktürk, 2007). In recent years, social constructionism has become a widely utilised approach in social science research (Andrews, 2012; Burr, 1995, 1998; Morley et al., 2014). Social constructionism proposes that human experience and inter-action is cultivated through historical, cultural and linguistic context. This is particularly important in this study, which is contextualised by an investigation of how historical/linguistic factors explicate understandings of the mobilisation of wellbeing within a PRA. Drawing on social constructionism as a paradigm facilitated the researcher's ability to appreciate how socio-historical contexts affect and change word meanings and their conceptualisations and representations in society.

Social constructionism proposes that what is perceived and experienced is never a direct reflection of environmental conditions, but must be understood within the specific context of these conditions (Andrews, 2012; Burr, 1998; Schwandt, 2003). This implies that no form of knowledge is superior to other forms of knowledge (Burr, 1998; Morley et al., 2014; Parker, 1998).

Traditional academic research tends to locate academic empirical knowledge as superior knowledge, i.e. that which we should strive to create; thereby, producing a hegemonic construction of 'valid knowledge' (Andrews, 2012; Bryman, 2008; Morley et al., 2014). Social constructionism, however, facilitates the presentation of

lay narratives and their representations of knowledge and understanding. It espouses equal validity for academic, professional and lay narratives and representations of knowledge and understanding. This positioning is consistent with the epistemological perspective employed in this thesis.

# 4.2.3: Theoretical perspective

The theoretical perspective underpinning this study is historicism. Historicism originated during intellectualism in 19th century German scientific enquiry (Felluga, 2003; Morera, 2013; Schwandt, 2001; Young, 2004). Historicism proposes that all knowledge and understanding are historically conditioned and contextual (Felluga, 2003; Morera, 2013; Schwandt, 2001). Historicism seeks to ground and locate understanding and knowledge within a historical context (Hamilton, 2003; Mikics, 2007; Morera, 2013). Historicism challenged the prevailing view of history which interpreted history as a progressive linear process, operating according to universal laws (Mikics, 2007; Morera, 2013). This predominant view had been widely held by historical thinkers since the Enlightenment (Dixon, 2005; Felluga, 2003; Mikics, 2007). Historicism stresses the unique diversity of historical contexts and the importance of developing specific approaches, which are appropriate to each unique historical context (Felluga, 2003; Morera, 2013). Historicism also questions the notion of rationality and truth, embedded in traditional historical thinking, arguing instead for the pursuit of a historical contextualisation of knowledge and reason (Dixon, 2005; Morera, 2013).

This thesis utilises Wilhelm Dilthey's (1833–1911) understanding of historicism. Dilthey argued that individual's lived experiences project into the past and future and are intimately bound to socio-historical context (Dixon, 2005; Felluga, 2003; Hamilton, 2003; Young, 2004). This contrasts with Karl Popper's (1957) understanding of historicism, that there are historical laws in existence. Popper (1957) proposed that historical laws form cyclical patterns that can be observed and used as templates to inform and govern action and policy within social contexts (Hamilton, 2003). Popper's (1957) understanding of historicism, however, has been challenged for not being consistent with the fundamental philosophy of historicism,

which refutes the notion that true and objective knowledge exists and can be discovered (Dixon, 2005; Hamilton, 2003; Morera, 2013).

## 4.3: Historical analysis

Historical analysis considers and balances differing perspectives in order to appreciate competing or corroborating narratives (Dixon, 2005; Gardner, 2010; Wyche et al., 2006; Young, 2004).

The historical analysis undertaken in chapter five contextualises the mobilisation of wellbeing through the prism of the historical and linguistic development of wellbeing. The historical analysis is textual data, which contributes to the literature review and the theoretical premise that wellbeing has been mobilised within a wider PRA. This offers an appreciation of the role of historical and linguistic factors in the current mobilisation of wellbeing. The historical documentary evidence selected was determined by the available data, which in this field of research remains scarce (Bergdolt, 2008; S. Bok, 2010; Kingfisher, 2013; McMahon, 2006, 2010). Sources were drawn upon in order to explicate how and why conceptualisations of wellbeing may have changed over time. The main contributors to the historical documentary evidence were the works of Bergdolt (2008), McMahon (2006) and Oishi et al. (2015). Additional sources where available were also included. These combined sources were used to authenticate or challenge the validity and reliability of the researcher's interpretations and analysis.

Historical analysis is an approach which draws on narratives based on available evidence (Gardner, 2006, 2010; Marshall and Rossman, 2009). It reflects upon 'how' and 'why' events occurred and the contexts in which they are presented (Marshall and Rossman, 2009). Historical analysis seeks to provide exploratory answers such as how and why changes occurred in society. Additionally, it considers how the evidence and methods of verification utilised may have influenced findings (McMahon, 2006; Yılmaz, 2007). Importantly, historical analysis does not *per se* aim to generalise applicable arguments to other similar cases (Gardner, 2006). Historical analysis involves reviewing and interpreting a diverse range of sources by drawing

upon a wide range of analytical skills (Mahoney and Rueschemeyer, 2003; McMahon, 2006; Yılmaz, 2007).

"Historical analysis is commonly used in social research as an introductory strategy for establishing a context or background against which a substantive contemporary study may be set" (Wyche et al., 2006:47). Wyche et al. (2006) argued that historical analysis "Forces people to become more aware of their preconceptions about a topic and...Because of its ability to defamiliarize the present, historical analysis can be a powerful recourse for inspiring innovative approaches and conceptualisations" (p.48). Historical analysis is often combined with other methods as it has the power to challenge dominant assumptions attached to social research questions (Gardner, 2010; Mahoney and Rueschemeyer, 2003; Marshall and Rossman, 2009; Morera, 2013; Wyche et al., 2006).

History should not be perceived as a series of facts, but as a series of competing narratives based on an individual's interpretation of the evidence (Gardner, 2006). "Historical accounts always carry the status of interpretations of real past events rather than straightforward factual representation of them" (Gardner, 2010:5). Thus, the agency of the authors and historians must be perceived in relation to the historical accounts they utilise and the manner in which they interpret them (Dixon, 2005; Gardner, 2010; McMahon, 2006; Morera, 2013). It is important to recognise and comprehend the positionality of a source in order to assess the source's validity and reliability (Gardner, 2006, 2010; Morera, 2013; Young, 2004).

Historical analysis endeavours to present well-balanced and transparent interpretation through a number of important procedures (Gardner, 2010; Mahoney and Rueschemeyer, 2003; Waring and Bentley, 2012; Yılmaz, 2007). The historical documentary evidence was analysed through the procedures set out below:

 Comparing and contrasting the evidence provided by sources, paying due regard to diverging ideals, values and the institutions from which the data originated.

- Considering the multiple narratives and perspectives of the sources, paying due regard to motives, belief systems, political interests and expectations.
- Assessing the author's credibility as a means of scrutinising the source's validity and reliability.
- Considering proposed cause and effect relationships.
- Paying due regard to issues such as the socio-political status and/or profile of the source.
- Paying due regard to the influence of conventional and unconventional ideas and beliefs and their dissemination.
- Comparing evidence across time, cultural groups and geographical regions in order to highlight issues which transcend regional, cultural and temporal boundaries.
- Distinguishing between unsubstantiated opinion or anecdotal evidence and theories based on robust evidence.
- Providing interpretation of competing historical narratives.
- Challenging implicit representations of historical inevitability and linearity,
   i.e. the depiction of a direct line between past and present as if there were some predestined outcome.
- Challenging theories [your own and others] that may require amendments
  as new evidence emerges, new interpretations are offered and contextual
  perceptions change.
- Analysing and evaluating major debates concerned with alternative interpretations of the evidence.
- Taking into consideration the influence the evidence represents for the present by reflecting on the constraints and opportunities apparent in the past.
- Guarding against judging the past within current norms and value systems.

### 4.3.1: Historical comprehension

The historical analysis undertaken in this thesis has been grounded in the utilisation of historical comprehension. This is a key process which acknowledges the difference between historical facts and historical interpretations whilst recognising that they are related (Mahoney and Rueschemeyer, 2003; Waring and Bentley, 2012; Waring and Robinson, 2010). Sources purposively select evidence and facts, which reflect what may be construed as the most important, salient and significant aspects. Differing perspectives therefore require interpretation and analysis in order to assess the source's validity and reliability (Drost, 2011; Mahoney and Rueschemeyer, 2003). Whilst primary and secondary sources are widely utilised, literary sources such as novels, poetry, plays and music are often helpful in illustrating how the theories and analysis presented in historical narratives operated in society (Waring and Bentley, 2012; Waring and Robinson, 2010). This helps to generate a more rounded and holistic understanding of the subject matter (Drost, 2011; Waring and Robinson, 2010; Waring and Bentley, 2012).

#### 4.3.2: Historical Postmodernism

The work of Yýlmaz (2007) details the impact post-modernism has had on the way historical analysis and interpretation is undertaken and represented. Historical Postmodernism argues that the representation of history is influenced by historians' philosophical orientations. These orientations include commitments to a positivist or idealist paradigm, epistemological position, political alignment, belief systems, personal backgrounds and academic trainings (Yılmaz, 2007).

"Being aware of how historians of different historical orientations construct differing interpretations of the past" (Yılmaz, 2007:178) is one of the preconditions for transparent historical analysis. This "Rejection of historical realism is a major theme in the philosophy of postmodernist historicism" (Yılmaz, 2007:181), and one which is congruent with the epistemology underpinning this thesis.

Yılmaz (2007) suggests that whilst historical postmodernism represents a marginalised perspective, its positionality adds an important dimension to historical

analysis. Whilst this historical analysis does not claim to be written from a post-modernist perspective, it has utilised its cogency. This is reflected in the choices of sources, the diverse range of historical perspectives and the analytic and interpretative conclusions drawn. This has meant drawing upon traditional historical accounts which may present a euro-centric colonial reading of events (Brown, 2009; Endacott, 2010; Young, 2004) and the perspectives of post-modernists who seek to offer more inclusive analysis and interpretation (Yılmaz, 2007). Both perspectives offer insight into competing narratives such as the 'Grand Narratives' (Yılmaz, 2007) and under-represented 'Micro Narratives' (Young, 2004) in order to highlight the difficulties in conducting historical analysis.

## 4.4: Methods for empirical study

### 4.4.1: Qualitative approach

This study has been undertaken using a qualitative approach because little qualitative research has been undertaken in the UK to explore lay conceptualisations of wellbeing (Barnes et al., 2013; Eraurt and Whiting, 2008; Ward et al., 2012). This is despite the argument that "The concept of wellbeing lends itself to qualitative assessment as people conceptualise wellbeing individually and socially through the cognitive and affective appraisal of their lives" (Diener et al., 1999:237). Additionally, this study seeks to explicate understandings of complex concepts and theoretical positions. Qualitative research is ideal for this type of study because it helps to unpack complex issues (Barnes et al., 2013; Doyal and Gough, 1991; Clark and Gough, 2005). These factors help justify the adoption of a qualitative research approach to this study.

Qualitative research is a broad methodological approach which encompasses a diverse number of different research methods (Creswell, 1998, 2008; Denzin and Lincoln, 2005; Miles and Huberman, 1994). Some propose that qualitative research is primarily imbued with an exploratory agenda (Willig, 2001). Others suggest that qualitative research is frequently used to reveal patterns and relationship between

data (Creswell, 2008; Denzin and Lincoln, 2005; Savin-Baden and Major, 2013). The aim of qualitative research, however, varies widely and is frequently predicated on the disciplinary background of the researcher (Gasper, 2010).

Qualitative research and qualitative research methods have been criticised by some (Creswell, 2008; Savin-Baden and Major, 2013). For example, there are concerns that qualitative research and qualitative methods are too concerned with issues not directly related to the research problem (Savin-Baden and Major, 2013). Other concerns raised include claims that findings based on qualitative research are not generalisable as they are concerned with specific populations or contexts (Savin-Baden and Major, 2013). Other concerns raised about qualitative methods argue that it is not robust because procedures such as probability sampling are not employed (Creswell, 2008; Savin-Baden and Major, 2013). It is important therefore when using qualitative methods that one draws upon reflexive practice. The role of reflexive practice in mediating methodological issues is discussed in section 4.9.

Qualitative researchers, however, have countered criticisms, arguing that qualitative research seeks to understand a research topic from the perspectives of the local population and is not explicitly concerned with generalisable findings (Creswell, 2008; Savin-Baden and Major, 2013). Qualitative research is particularly effective when investigating issues within the context of a particular population or specific period of time (Braun and Clarke, 2006). Here, the primary objective is to "Describe and understand how people feel, think, and behave within a particular context relative to a specific research question" (Braun and Clarke, 2006:77).

Qualitative research methods such as discourse or thematic analysis offer numerable approaches to the research problem and facilitate diverse, complex and nuanced accounts of a research problem (Creswell, 2008; Holloway and Todres, 2003; Savin-Baden and Major, 2013).

Qualitative analytic methods have been roughly subdivided into two traditions. The first tradition consists of those methods associated with a particular theoretical and/or epistemological position. These include conversation analysis in which there is limited variability in the application of methods and discourse analysis which has

"Different manifestations of the method, from within the broad theoretical framework" (Braun and Clarke, 2006:77).

The second tradition consists of those methods that are essentially independent of theory and epistemology and can be applied across a range of theoretical and epistemological approaches. One such method is the one adopted in this study, i.e. thematic analysis, which "Provides a flexible and useful research tool, which can potentially provide a rich and detailed, yet complex, account of data" (Braun and Clarke, 2006:77).

The particular qualitative method adopted here was identified by the coresearchers as congruent with their research expertise (this is discussed in detail in section 4.7). Section 4.8 discusses the use of thematic analysis as the analytic method in this study.

Excerpt from Researcher's Reflexive Journal (December 2007)

My consideration of using discourse analysis, an approach well suited to this PhD, is dashed after conferring with Alison and Jim. They find it intimidating and complex!!

Critical theory, which involves both understanding and theoretical explanation focus on historical specificity, improving the understanding of the topic through the integration of all the major social sciences. Critical theory is congruent with my epistemological perspective, however, appears oriented toward critiquing and changing the phenomenon under study. This contrasts with traditional approaches, which are oriented to explaining the phenomenon under study. Given this study is orientated towards the more traditional approach, this isn't the method either.

## 4.4.2: Sampling framework

Sampling has huge implications for the internal and external validity of research and how representative it might be in wider contexts (Keiding and Clayton, 2014; Mason, 2010; Palys and Atkinson, 2008; Suen et al., 2014). It can be challenging to choose a sampling framework consistent with one's wider epistemology. Given the constraints under which PhD study is often undertaken, such as availability of funds and time limitations, sampling frameworks often present the researcher with difficult decisions (Charmaz, 2006; Keiding and Clayton, 2014; Mason, 2010; Ritchie et al., 2003).

Sampling is usually categorised as being based on probability or non-probability sampling. The non-probability sampling framework utilised in this study was part purposive\*, part snowball\* and part self-selecting\*. Purposive sampling is widely associated with qualitative research where the intention of the research is to explore particular groups in society who fit a certain criterion rather than to conduct research across a population (Palys, 2008; Palys and Atkison, 2008). "Researchers who use this technique carefully select subjects based on study purpose with the expectation that each participant will provide unique and rich information of value to the study" (Suen et al., 2014:105).

Purposive sampling ensures that the profile of those involved is congruent with the characteristics under investigation. In this study, the parameters for those involved were not considered overly restrictive.

With regard to sampling, this study sampled a cross section of older adults from a broad base. The participants in this study included those living in institutional settings and those living in the community. This sampling framework aimed to reduce the likelihood of the samples producing biased results not indicative of the wider population (Fadem, 2009; Jacobs et al., 2009; Keiding and Clayton, 2014; Mokhtarian and Cao, 2008). Non-probability sampling, however, has particular issues which must be addressed by the researcher: e.g. selection bias. The following sections address these issues.

#### 4.4.3: Selection bias

Self-selection bias is a considered to be a problem in research which utilises non-probability sampling and may undermine the robustness of the research (Gail and Benichou, 2000; Keiding and Clayton, 2014; Mokhtarian and Cao, 2008). Self-selection may make the determination of causality difficult to establish.

Additionally, it may undermine how representative the sample is and whether the findings in the study population are generalisable and valid in other populations (Fadem, 2009; Keiding, 2015; Keiding and Clayton, 2014).

Self-selection bias arises in any situation in which individuals select themselves for inclusion in a study (Gail and Benichou, 2000; Jacobs et al., 2009; Mokhtarian and Cao, 2008). This can lead to a biased sample that is a situation where the characteristics or demographics of the people who select themselves lead to an over-representation of a particular group in the study (Gail and Benichou, 2000). Conversely, non-response bias may also occur, i.e. a situation where a particular demographic fails to select themselves for inclusion (Gail and Benichou, 2000).

Given this study was undertaken by a lone researcher on a limited budget, the ability to address self-selection bias was limited. The researcher promoted and recruited to the study across the administrative districts of Manchester to try to ensure the demographic profile of participants was in broad agreement with the demographic profile of Manchester, as reported in the ONS data figures (ONS, 2009i).

The ability to influence non-response bias was also hampered by the researcher's limited resources. However, the researcher made every effort to promote and recruit across the demographic spectrum including hard-to-reach groups. However, given the absence of certain groups from this study such as homeless, migrants and refugees, there may be qualified concerns about non-response bias in this study.

Selection bias on the part of the researcher is also a recognised problem (Fadem, 2009; Gail and Benichou, 2000). Despite efforts to eliminate bias from research there may be a purposeful or subconscious intent on the part of the researcher to

select people congruent with the research aims and objectives. In much the same way as with self-selection bias, this has implications for the generalisability and validity of study findings (Fadem, 2009; Gail and Benichou, 2000).

It is difficult to say with any certainty if the researcher is guilty of having a subconscious intent to select people congruent with the research aims and objectives. However, it is certain that there was no purposeful intent on the part of the researcher to select people congruent with the research aims and objectives.

## 4.4.4: Sample size

There are a number of factors which may influence the potential size of a sample in qualitative research. These include budgetary and resource considerations; time constraints; heterogeneity of the population; the selection criteria; the inclusion of special interest groups; need to obtain multiple samples within one study; and types of data collection methods used (Charmaz, 2006; Guest et al., 2006; Mason, 2010; Ritchie et al., 2003).

Mason's (2010) review of qualitative research suggested that fifteen was the smallest acceptable sample size. Charmez (2006), however, suggested a sample size up to twenty-five people could be considered fit for purpose in qualitative research. The growing utilisation of qualitative research has led to calls for a greater focus on the need for qualitative researchers to generate robust research (Fadem, 2009; Guest et al., 2006; Jacobs, 2009; Keiding and Clayton, 2014; Mason, 2010).

Mason (2010) suggested that few researchers identify why they have chosen their particular sample size. This lack of clarity and little practical guidance for estimating sample sizes has been critiqued for undermining efforts to establish qualitative research as robust (Guest et al., 2006; Mason, 2010; Morse, 2000).

Mason (2010) argued that most researchers avoid or shy away from discussing what constitutes a sufficient sample size. "The point of saturation is, as noted here, a rather difficult point to identify and of course a rather elastic notion. New data (especially if theoretically sampled) will always add something new, but there are

diminishing returns, and the cut off between adding to emerging findings and not adding, might be considered inevitably arbitrary" (p.12).

Efforts to establish practical guidance for estimating sample sizes in qualitative research led Mason (2010) to propose that the concept of saturation should be the guiding principle for sample size in qualitative research. Mason (2010) investigated how many people were involved in PhD studies utilising qualitative interviews. The findings indicated that the extent to which these numbers varied depending on the methodological approach. Mason (2010) identified that a wide range of sample sizes was observed in PhD studies, but "The most common sample sizes were 20 and 30" (p.16).

After the saturation point for recruitment and retention was reached, the final sample size for this study was 20 participants and 2 co-researchers. The following sections document the study population.

### 4.5: Recruitment of participants

The term 'participants' refer to all those taking part in research; this is in line with the British Psychological Society Code of Human Research Ethics. This code underpins the ethical approval granted by Manchester Metropolitan University (MMU) for this study. The use of the term participants does not intend to infer that this research is participatory research. In this study, "Participation is understood more as the involvement of any groups of people who are not professional researchers" (Bergold and Thomas, 2012:4).

A recruitment drive was initiated with posters and leaflets detailing the study. These were distributed to community centres, day centres, residential centres, residential care homes, warden-assisted accommodation and events connected with the original project. The events connected with the original project included one workshop session and two focus group events. Potential participants who learnt about the study after attending events connected with the original project contacted named Manchester City Council staff responsible for promoting these events. These staff members had agreed, prior to the recruitment drive, to act as

intermediaries on behalf of the researcher. Staff members conveyed the contact details of potential participants who had enquired about the study to the researcher. Potential participants who learnt about the study from material distributed to community centres, day centres, residential centres, residential care homes, warden-assisted accommodation contacted the managers of these institutions. These managers had agreed, prior to the recruitment drive, to act as intermediaries on behalf of the researcher and conveyed the contact details of potential participants who had enquired about the study to the researcher.

Criteria for taking part in the study were limited to those aged 50 or over, who identified themselves as having an LTC and who lived in Manchester. The researcher contacted twenty seven potential participants matching the inclusion criteria and provided them with all the relevant information on the study. This included an invitation letter, participant information sheet, interview guide and consent form. Depending on individual preferences, potential participants could obtain more information about the study from the researcher either in person or over the telephone.

Twenty of the twenty seven potential participants self-selected to participate in the study. None of these participants had taken part in the original project. The participants were able to choose where they would like to be interviewed. Some chose to be interviewed in the day centres they attended and some chose to be interviewed in their accommodation. Pseudonyms are used throughout the report to refer to participants.

### 4.5.1: Participants' profile

Excerpt from Researcher's Reflexive Journal (11 May 2009)

In the spirit of reflexivity, I reflected at length upon the terminology used in this thesis. The term 'researcher' promotes a representation of myself to myself and to others, while within the term is located a position of authority and power, particularly in relation to 'participant' which perpetuates an in-balance of power. This places both myself and the 'participant' in pre-determined roles in which I hold the balance of power. I have reflected at length on how I would choose to portray those who took part in this thesis. However, after much deliberation, I have concluded that 'researcher' and 'participant' conveys our relationship as closely as the alternatives and is in keeping with the ethos of the study.

The participants were selected from across the three administrative districts of Manchester (Central, North and South) in order to capture the diversity of people's experiences. Nine participants lived in North Manchester, seven lived in South Manchester and four lived in Central Manchester. The participants included men and women from different ethnic and cultural backgrounds within an age range of 50 to 76 years. In this study, 'younger end of the age spectrum' refers to participants aged 60 or under (n=9) and 'older end of the age spectrum' refers to participants aged 70 and over (n=4).

The nature of participants' health conditions varied considerably. Some participants' conditions were lifelong such as cerebral palsy, which occurs at birth and may result in long-term disability (Hobbs and Sixsmith, 2010; Iddons, 2009; The Neurological Alliance, 2000). Other conditions included depression, stroke, cancer and traumatic acquired brain injury (TABI). These may occur at any point in the lifecourse (Hobbs and Sixsmith, 2010; Iddons, 2009; The Neurological Alliance, 2000). Other conditions tend to manifest at particular stages in the life-course. One participant had pyriecis, which tends to manifest in childhood, while three had multiple sclerosis, which tends to manifest in middle adulthood. Other conditions

included angina, arthritis, cataracts and dementia which tend to manifest in later life (Hobbs and Sixsmith, 2010; Iddons, 2009; The Neurological Alliance, 2000).

Five participants lived in residential care accommodation, five lived in wardenassisted living accommodation, six lived in council-rented accommodation and four lived in owner-occupied accommodation.

Warden-assisted living accommodation consisted of on-site care staff (24 hour / 7 days a week) and non-resident management staff. Warden staff contributed to the residents' wellbeing by facilitating independent living through activities such as shopping trips and shared social activities (Manchester City Council, 2005a).

Residential care normally provides accommodation, which includes meals and personal care but does not include nursing care. Personal care includes help with medication, assistance in getting up or going to bed, and help with eating, washing and dressing (NHS, 2014). These usually consist mainly of single room dwellings but may also have kitchens (NHS, 2014).

Council residential accommodation varied according to the tenants' requirements.

Participants in this study held varying tenancy agreements. Information regarding tenancy agreements was not routinely collected in this study, and therefore is not included in Table 1. However, issues around tenancy agreements emerged as important in the participants' narratives, so the following sections outlining tenancy agreements and rights are included for information purposes.

There are two types of tenancy agreements in council residential accommodation: an introductory tenancy and a secure tenancy. Introductory tenant residents have fewer legal rights than a secure tenancy, which is only offered to tenants who demonstrate that they "Can act responsibly" (Manchester City Council, 2015:online).

Tenancy rights for those who live in supported accommodation of either wardenassisted living accommodation or residential care depended on the type of accommodation and level of support received (Hobbs and Sixsmith, 2010). Tenancy agreements ranged from assured short-hold tenancy, to assured tenancy and secure tenancy. Assured short-hold tenancy usually lasts between 6 to 12 months and requires notice of eviction. Assured tenancies may be a fixed-term tenancy or a periodic rolling tenancy (Manchester City Council, 2015). Secure tenancies have the longest tenure but also include introductory tenancy agreements, usually between 12 to 24 months, which frequently limit residents' notice of eviction (Hobbs and Sixsmith, 2010).

# 4.5.2: Table 1- Participants' profile

Name	Age	Status of LTCs	Ethnicity	Employment	Living
				status	accommodation
Sara	50	Congenital -		Unemployed	Warden-assisted
		Cerebral palsy	White		accommodation
		and stomach			
		ulcer			
Shawn	51	Late Onset -		Unemployed	Residential care
		MS and clinical	Black		home
		depression			nome
Ben	52	Late Onset -		Unemployed	
		Stroke and	White		Warden-assisted
		clinical	vviiite		accommodation
		depression			
Caroline	52	Congenital -		Unemployed	Warden-assisted
		Cerebral palsy	White		accommodation
		and gall stones			
Audrey	52	Congenital -	White	Unemployed	Residential care
		Cerebral palsy			home
Tim	57	Late Onset -	Mixed Race	Unemployed	Owner occupier
		TABI and heart			
		attack	Nace		

Davindra	58	Late Onset -		Unemployed	Mandan
		TABI, clinical	Asian		Warden-assisted
		depression			accommodation
lvy	60	Late Onset -		Retired	Council-rented
		Stroke and	White		
		angina			accommodation
Jake	62	Late Onset -		Retired	
		MS,	White		Council-rented
		depression			accommodation
		and diabetes			
Silvia	62	Late Onset -		Retired	Residential care
		MS and clinical	White		home
		depression			
Simon	63	Late Onset -		Retired	
		TABI, clinical	White		Warden-assisted
		depression			accommodation
		and angina			
May	64	Late Onset -		Retired	
		Stroke,	White		Council-rented
		arthritis and	vviiite		accommodation
		dementia			
Jack	67	Congenital -		Retired	
		Pyriecis and	White		Residential care
		dislocated	vvnite		home
		shoulder			
Moira	68	Late Onset -		Retired	
		Stroke and	White		Council-rented
		clinical			accommodation
		depression			
Raj	68	Late Onset -	Asian	Retired	Owner coorsies
		Stroke,			Owner occupier

		diabetes and arthritis			
Andrea	69	Late Onset - Cancer, kidney failure and diabetes	Jewish	Retired	Owner occupier
Delphine	70	Late Onset - Stroke, angina and arthritis	Black	Retired	Council-rented accommodation
Jenny	72	Late Onset - Stroke and diabetes	White	Retired	Owner occupier
Brenda	74	Late Onset - Stroke, angina, arthritis, and cataracts	White	Retired	Owner occupier
Denise	75	Late Onset - Stroke, arthritis, angina	White	Retired	Council-rented accommodation

# 4.6: Co-researcher involvement

As previously detailed, the two people (Jim and Alison) who eventually became the co-researchers in this thesis had been involved in the original project as community researchers/advisors. They were engaged as part of a recruitment drive for the original project, which took place in 2007. The term 'co-researcher' is an empowering term selected by participants during their first training event (see Appendix H).

# 4.6.1: Lay versus professional dichotomy

User controlled research may have some justification for debating whether coresearchers should still be considered 'lay' (Evans et al., 2004). Some suggest that once research participants have received training there is room to argue the extent to which they may be considered lay (Bergold and Thomas, 2012; Durham Community Research Team, 2011; Staley, 2009). It is true to say "Co-researchers step back cognitively from familiar routines, forms of interaction, and power relationships in order to fundamentally question and rethink established interpretations of situations and strategies" (Bergold and Thomas, 2012:1). However these acquired skills do not mean they are no longer constitute being lay (Flick, 2009). The distinction between terms lay and professional revolves around knowledge not training (Abrahamson and Rubin, 2012; Bolam et al., 2003; Earl-Slater, 2004). Another clear distinction between lay and professional is that professional researchers receive a salary whilst lay researchers receive expenses or fees as recompense for travel costs, etc.

The debate to whether co-researchers can still be considered lay can be illuminated within the context of this research by the understandings of the co-researchers themselves. The co-researchers recognised that they had a level of awareness, skill-set and knowledge of the research process, which the other participants did not possess. However, they continued to identify themselves as lay participants. Indeed, the study data supports this sense of identity as the co-researchers' data were coterminous with the participants' data although in some respect, and in some cases more detailed. This detail, however, may be attributable to the greater number of interviews undertaken with the co-researchers. For an in-depth discussion around lay involvement in research, see McLaughlin (2009a,b).

The researcher recognises that there are different degrees of participation (Bergold and Thomas, 2012) and that co-researchers have contributed to the research process unlike the participants. However, their engagement in this study was circumscribed by their skill-set, knowledge and expertise, which marks them as lay rather than professional.

The researcher also recognises that there is a relationship between her and the coresearchers, which distinguishes them from the participants. However, this does not alter their status as lay. Lay in this study perceives their contribution to the research as being their lived experiences, and their everyday knowledge of the topic under discussion (Bergold and Thomas, 2012). For an in-depth discussion around lay involvement in research, see Appendix D.

## 4.6.2: Co-researcher training

Guidelines from INVOLVE (2007) indicated that training was a key undertaking for service users brought on board in research projects to ensure their full and informed participation (Branfield and Beresford, 2006). In consultation with the co-researchers a training agenda was devised and delivered in three half days over the course of one month.

The co-researchers undertook three training sessions. An external training provider delivered the first two sessions. Training aimed to familiarise the co-researchers with the research methods, research techniques, power sharing, an appreciation of ethics, committee membership, data analysis and the writing up process. Training also covered learning about the basic premises about qualitative research, the development of critical thinking skills and equipping the co-researchers with the necessary skills to engage in reflective practices. For full disclosure on the training process, see Appendix H.

A third training session focused on equipping the co-researchers with data analysis skills and was delivered by the researcher and her line manager. This involved a data workshop session which explored discourse analysis, content analysis, thematic analysis and grounded theory.

#### 4.6.3: Remuneration

The researcher consulted with the co-researchers about study payment and an agreement was reached that £10 per interview/consultation session was sufficient to cover the out-of-pocket expenses and as compensation for the time spent. Alison accepted this payment but Jim declined. In addition to the £10 per interview, Alison

and Jim were given refreshments and paid a small sum to cover the costs of travel and parking when attending progress meetings.

#### 4.6.4: Collaboration

Before the co-researchers became involved in this study, the researcher ensured that they knew the aims and objectives of the research and the personal time commitment that would be required. This included a commitment from the researcher to provide the co-researchers with ongoing support and a commitment from the co-researchers to undergo training.

# Support included:

- · Preparation, briefing and debriefing
- Progress meetings
- Information sharing
- Training and mentoring
- Regular contact and one-to-one support outside meetings
- Explaining things and checking understanding
- Transport, remuneration and refreshments

In addition to the training and support, the researcher established regular progress meetings. The progress meetings took place in a variety of settings; these included the co-researchers' homes, community spaces and the workplace of the researcher. This process enabled the co-researchers and the researcher to periodically reflect on the progress of the research and to establish action points collaboratively. This process involved a number of meetings in order to clarify the focus and direction of the study and ensured the acknowledgement of each person's input. In addition to the progress meetings, regular communication was maintained between the co-researchers and the researcher. Communication was achieved through a variety of mediums including personal one-to-one contact, telephone conversations and emails.

Clear roles were established for the co-researchers, taking into account their different skill-sets, interests, knowledge and expertise. Discussions were held to demonstrate how each role contributed to the research process.

Whilst this research does not claim to be participatory research, the involvement of the co-researchers throughout the research process means that it utilises aspects of the participatory research process (Bergold and Thomas, 2012; Durham Community Research Team, 2011). When utilising aspects of participatory research it is important to manage expectations (Barnes and Cotterell, 2011; Bergold and Thomas, 2010; Duffy and McKeever, 2011; Simpson et al., 2014). This was done by establishing the co-researchers' expectations at the beginning and revisiting these at intervals during the research process. This enabled reflection upon original expectations to consider if they had changed and if these changes were acceptable to the co-researchers.

The co-researchers contributed to the design and methodology of the study. This involved contributing their points of view to the means of data collection, data analysis and the writing up of findings. The co-researchers' involvement in this study influenced the focus of the research questions as well as its design and content.

The following sections detail the extent to which the co-researchers were involved in the different stages of the research process.

#### 4.6.5: Choice of methods

The co-researchers were involved in discussions around the choice of data collection and data analysis methods. Having received training in these aspects of the research process, they felt they had the necessary skills to make a meaningful and informed contribution to this aspect of the research process. The researcher and co-researchers collectively decided how the data would be collected and analysed. However, the co-researchers considered that involvement in the epistemological perspectives and paradigmatic approach was beyond their skill-set, and this aspect of the research process was the sole responsibility of the researcher.

## 4.6.6: Analysing and interpreting data

The researcher and Alison worked together on the data analysis both individually and collaboratively throughout the data collection period and for an additional three-year period (2010–2013). The researcher sent Alison drafts of the data analysis two weeks prior to face-to-face consultation sessions, which took place in her living accommodation. During this consultation session the researcher and Alison assessed and discussed each other's solo analysis and interpretation of the data. During this time, the researcher and Alison engaged in the process of data immersion and crystallisation, i.e. a process that distils understanding from text. This facilitated the emergence of concepts organically in addition to those identified through a deductive coding process. This process also involved working together collaboratively in order to synthesise our individual analysis. Subsequent to the preliminary analysis, themes were further informed by the ongoing literature review and discussions were held between the researcher and Alison with regard to the identification of themes.

Alison also reviewed final drafts of the full data analysis chapter and made comments concerning her perception of the validity of findings. Additionally, she was consulted with respect to presentation and the clarification of the findings chapter.

Jim throughout the course of the research intended to be involved with the data analysis; however, a decline in his health led to his decision to step back from active engagement with the data analysis process.

The analysis of the data has therefore been a collaboration between the researcher and Alison in order to frame understandings in lay terms and perspectives.

### 4.6.7: Writing up findings

The co-researchers did not undertake any of the writing up, but were involved in reading through drafts of specific chapters of the thesis in accordance with their particular and individual interests. Alison read through and commented on chapters

one, six and eight. Jim read through and commented on summarised drafts of chapters two and five.

#### 4.6.8: Dissemination

The co-researchers attended and contributed to two MMU conferences in which the study researcher presented early findings of the study. The co-researchers involvement included reading through the presentations, helping the researcher pick which aspects of the study to present and in what format (PowerPoint). One of the researcher's [Alison] was also involved in answering questions from conference attendees about her level of engagement and involvement with the study.

The co-researchers also attended and contributed to an MMU knowledge transfer event. This highlighted the opportunities and challenges of utilising aspects of participatory approaches in research. Their involvement included participating in a question and answer session to explore some of the opportunities and barriers to engagement they had experienced whilst working in academic research.

#### 4.6.9: Ownership

The two co-researchers felt that involvement in this thesis gave them the opportunity to "Contribute to the accumulation of knowledge" (Jim) and to "Give something back" (Alison). For Jim, contributing to an academic endeavour which would advance knowledge formed an important motivation for participating in the study. This motivation and dedication to intellectual development was important for him. He considered his involvement in the production of a PhD thesis to be part of his legacy. Jim therefore declined the option to have his name replaced with a pseudonym. Alison, however, was more interested in the applied outcomes of the research than with academic recognition and she elected to have her real name replaced with a pseudonym. Their reflective diaries in Appendices J and K contain a fuller account of their perceptions of being involved in this study.

However, mutual ownership of a PhD thesis cannot truly exist because in this context the research process is intended to lead to an academic qualification for the researcher alone (Durham Community Research Team, 2011). However, the co-

researchers were involved with the design and implementation of this thesis.

Without their time, effort and commitment this thesis would be less an account of lay personal wellbeing and perhaps more of an account of the researcher's perception of how lay people conceptualise personal wellbeing.

In some methods of analysis it is only the researcher's 'voice' that is heard and the researcher's account of the data (Barnes and Cotterell, 2011; Duffy and McKeever, 2011; Simpson et al., 2014). The decision to use thematic analysis in the production of this thesis facilitated and enabled the researched to have a voice, to account for the data and to some degree feel a sense of ownership.

## 4.6.10: Areas of research beyond the co-researchers' jurisdiction

The co-researchers were not involved with the ethics process as the ethics process was completed prior to their recruitment. The co-researchers were not involved with the funding process because funding was obtained prior to the recruitment of the co-researchers.

Neither Jim nor Alison were involved in reading or commenting on chapters three, four, and seven. The co-researchers were not involved in any aspects of data collection due to periods of ill health during this process.

### 4.7: Method of data collection

In seeking to present new understandings of how conceptualisations of wellbeing are generated, refined and redefined, this thesis required a flexible, pragmatic and focused method of data collection. Semi-structured interviews lend themselves to active engagement with participants and enable the researcher to explore complex issues with sensitivity (Bryman, 2001; Denzin and Lincoln, 2005; Silverman 2005).

The co-researchers proposed that face-to-face semi-structured interviews would also facilitate participant understanding of the complex concepts in this study. In addition to which they believed it would facilitate their engagement in the research process. In line with my commitment to their involvement in the process, it was

agreed that data would be collected through a series of semi-structured interviews from a purposively sampled group of individuals. This is an appropriate choice for data collection as semi-structured interviews are capable of generating a high level of participant engagement. It also encourages the development of reflexivity, which helps to focus the researcher's attention on how to use successive interviews to explore and probe issues and themes identified in previous interviews (Bryman, 2001). This lends itself to an iterative process, which enables the researcher to explore complex issues, building knowledge and understanding successively over the course of the data collection stage (Bryman, 2001; Silverman 2005).

However, the co-researchers and the researcher shared concerns that undertaking a series of single episodic interviews with participants would only provide a generalised appreciation of personal wellbeing. These were unlikely to generate the level of engagement, reflexivity, knowledge and understanding which the exploration of a complex and contested concept required. In addition to which the researcher began to reconsider whether a series of stand-alone semi-structured interviews would facilitate the researcher's task of synthesising theoretical understandings with emerging themes. An additional methodological concern surfaced with regard to the limited opportunity to undertake a series of interviews with the same participant.

To address these methodological issues a decision was made between the coresearchers and researcher that semi-structured interviews would be undertaken with twenty participants and that these stand-alone interviews would be supplemented with a series of semi-structured face-to-face interviews with the coresearchers. However, the co-researchers worried that the interview process might be constrained by the restrictive nature of their health. Further discussion suggested supplementing interviews with additional telephone and email communications and reflective diaries in order to address this concern.

#### 4.7.1: Interview process

The interviews took place from January 2008 to December 2010. They consisted of a series of stand-alone face-to-face semi-structured interviews with twenty

individuals with LTCs undertaken during 2008. These took place in day centres and residential homes.

Concerns raised by the co-researchers about the length of interviews led to a decision to restrict interviews to between 30 and 45 minutes, with regular breaks as and when required. The participants received a copy of the interview schedule at the onset of recruitment and again one week prior to the interview in order to facilitate considered responses and reduce potential anxiety regarding the process.

From January 2008 to December 2010, each co-researcher was interviewed separately six times in order to facilitate a more in-depth exploration of personal wellbeing. The co-researchers received a copy of the interview schedule one week prior to the interview. Interviews with the co-researchers were not time limited as the co-researchers agreed that they would end the interview when they felt it was appropriate. The face-to-face interviews with two co-researchers tended to last between approximately 45 and 60 minutes. The interviews took place in a range of settings including the researcher's workplace, the co-researchers' homes and public places such as cafes and libraries. Research diaries were used to contextualise understandings, whilst telephone communications were utilised to clarify understandings and explore commonality and contradictory findings.

The interviews, while being open-ended and flexible, utilised a semi-structured schedule which included prompts and probes to explore wellbeing in relation to topic areas widely recognised in the literature as important for wellbeing. This facilitated the participants' ability to explore understandings of personal wellbeing within the context of their own lives and offered the opportunity to discuss and prioritise these understandings.

The following section provides a rationale for the utilisation of thematic analysis in this thesis.

# 4.7.2: Analytic method

Thematic analysis is a versatile, pragmatic and intuitive method of data analysis (Braun and Clarke, 2006). As previous documented, thematic analysis is one of the

methods of data analysis included on the co-researcher training programme. The community researchers identified thematic analysis as the method of data analysis which they felt most comfortable and confident in understanding and using (see Appendix H).

Many have argued that thematic analysis is not a method in its own right (Boyatzis, 1998; Ryan and Bernard, 2000), but rather a generic process of generating themes in data across many different methods of analysis. Some have argued that "Thematic analysis is a poorly demarcated and rarely acknowledged, yet widely used qualitative analytic method" (Boyatzis, 1998; cited in Braun and Clarke, 2006:77). Braun and Clarke's (2006) seminal article argued that thematic analysis is a method in its own right, claiming, "Through its theoretical freedom, thematic analysis provides a flexible and useful research tool, which can potentially provide a rich and detailed, yet complex, account of data" (p.78). Braun and Clarke (2006) argued that despite a reluctance to ascribe thematic analysis as a method of analysis, it holds an important place in qualitative research and is often used widely as a 'generic method' of qualitative research.

Braun and Clarke (2006) identified some of the arguments cited against the use of thematic analysis as a method of analysis. They acknowledged that thematic analysis can be undermined by the absence of transparent and concise guidelines.

To assuage these arguments Braun and Clarke (2006) proposed clear and concise guidelines for undertaking thematic analysis. Their guidelines are intended "to celebrate the flexibility of the method and provide a vocabulary and 'recipe' for people to undertake thematic analysis in a way that is theoretically and methodologically sound" (p.78).

Thematic analysis in the context of my epistemological position is an important and pragmatic method of data analysis. It is a process which allows the researcher freedom to acknowledge that themes emanate as much from themselves and their relationship with their co-researchers as from the research process. In the years since Braun and Clarke's (2006) validation of thematic analysis it has become a

more widely accepted and popular method of data analysis, perhaps in recognition of its pragmatic approach and versatility (Yin, 2011).

## 4.7.3: Analytic process

Thematic analysis was utilised as a vehicle to emphasise lay perceptions of personal wellbeing by concentrating on the narratives of participants. Thematic analysis which "Focuses on identifiable themes and patterns of living and/or behaviour" (Aronson, 1994:8) was undertaken using procedures for categorising and coding key meanings (themes) identified in the data.

Braun and Clarke (2006) identified the analytical steps used in this thematic analysis:

- Familiarisation, i.e. reading and rereading transcripts and notes, identifying initial ideas.
- Code generation, i.e. systematic identification of interesting ideas.
- Theme identification, i.e. collating codes into potential themes supported by relevant data.
- Review, i.e. holistically checking themes against data.
- Labelling themes, i.e. refining themes and defining holistic analytical accounts in relation to research questions.
- Report writing.

An inductive\* or emergent generation of coding was primarily utilised to generate themes. The structuring of the interview questions, however, relied upon deductive coding\*. These were used to explore those elements of personal wellbeing widely identified within the literature as important contextual factors for understanding wellbeing, for example, 'health' and 'social interaction'. These codes mapped key themes that recurred during the interviews. Following the emergent coding procedure (Stemler, 2001) the data were examined utilising thematic analysis, which facilitates the regularisation of narratives from transcripts.

Observing the suggestions proposed by Granenheim and Lundman (2004), the data were transcribed, read for general comprehension and then re-read thoroughly.

Core meanings were identified and condensed into categories. These gradually coalesced into the emerging themes.

The researcher conducted the analysis during which the process of immersion in the data and crystallisation (a process that distils understanding from text) facilitated the emergence of concepts organically rather than through a preconceived agenda. Provisional themes identified during open coding were labelled based on recurring words, which occurred in the data sets. Subsequent to the preliminary analysis, themes were further informed by the ongoing literature review. In addition, discussions were held between the researcher and Alison (coresearcher) with regard to the identification of themes within her data set. As previously documented, the co-researchers had received training in thematic analysis and had undertaken its usage in data analysis workshops. Alison's reflective diary (see Appendix J) acknowledged that her training in thematic analysis and its subsequent application facilitated her engagement in the data analysis phase and the wider research process.

## Excerpt from Alison's Reflective Diary

Being part of this research has been great, as its [sic] an opportunity to get more involved with the university, to meet different people, to widen my experiences and to try and look at things from the professionals [sic] side of things. I did struggle at times in the data analysis workshop but by the end, I began to understand exploring themes, if I have to choose I would definitely pick the thematic analysis approach. I can see myself being able to get more involved with you in the future now that I have had some grounding and experience in actually doing it.

In line with ongoing data collection, minor additions and changes were made to the emergent framework of themes. This ensured that the analysis was grounded in practice and that aspects of the data were not over or under represented in the findings.

Themes sharing common characteristics were grouped together and further analysed to unpick sub themes. This was achieved by critically comparing,

contrasting and reviewing the data until no new themes could be discerned in the data. During this analytical process, identified as analytic generalisation (Yin, 2011), the interview data and concepts were continually condensed. This enabled often "Large, unwieldy categories to be refined into more specific, integrative themes whilst retaining important aspects of the properties and dimensions of the umbrella categories" (Corbin and Holt, 2005:51). Further exploration, analysis, cross-referencing and reflective practice unpicked the similarities and differences between themes, which were eventually refined into seven themes. These seven themes form the basis of the findings, but sub-themes were retained to highlight the complexities which existed in participants' narratives. The participants and co-researchers' data is presented collectively as the same themes emerged from their data sets. However, the co-researchers' familiarity with the parameters of the research may have led to more considered and reflective responses than the participants.

### 4.8: Reflexivity

"We do not learn from experience...we learn from reflecting on experience" (Dewey, 1933:17). Reflection may be perceived as the precursor to reflexivity (Moon, 1999). Conceptually, reflexivity moves beyond 'reflection' and seeks to initiate deeper understanding about oneself, one's socio-cultural conditioning, how this affects all aspects of one's life and how to use this knowledge to improve oneself (Salzman, 2002).

Reflexivity was first conceptualised by ethnographers working in anthropology in the early 20<sup>th</sup> century (Moon, 1999). Reflexivity is an important methodological issue in qualitative social science research (Savin-Baden and Major, 2013). It is a process in which researchers undertake to explore and experience those aspects of themselves and their rationalisations which are likely to have influenced every aspect of the research process (Nightingale and Cromby, 1999; Salzman, 2002; Savin-Baden and Major, 2013). Reflexivity has grown in stature as part of a broad academic movement, which began to critique and question the existence of any objective reality to discover (Salzman, 2002).

Reflexivity is now acknowledged as a methodological principle in qualitative research, in much the same way as the observer effect is acknowledged in quantitative research (Salzman, 2002, Savin-Baden and Major, 2013). It has therefore assumed a significance which requires the researcher to provide a full account of their socialised existence in relation to their research project (Hiles, 2008). This knowledge allows others to gain insights; for example, on the belief systems which directed their analytical approach. It is this conferment of the explicit assumptions made and the methods and procedures used which enable others to apply a more rigorous appraisal of the underlying nature of research and to compare and contrast it with other like-minded research (Attride-Stirling, 2001).

The notion of transparency therefore has become a key and over-arching concern in establishing the quality of qualitative research (see Hiles, 2008). However, fundamentally transparency has become the gold standard for the construction and dissemination of qualitative research findings (Hiles, 2008). Nightingale and Cromby (1999) reflected that reflexivity requires the researcher to acknowledge that their socialisation contributes to the meaning-making process. Thus, it is impossible to be detached or separate from the research process and what ensues from it.

Reflexivity is a key process in which researchers and those engaged in the research process must seek to explore the myriad ways in which their belief systems influence, and inform the research outcomes (Harper, 2002). This creates new challenges for qualitative researchers who must resist adopting an approach which offers insight into the subjectivity of the researcher rather than the researched (Finlay, 2002; Harper, 2002; Holloway and Todres, 2003; Nightingale and Cromby, 1999).

Personal reflexivity, however, is an important process in which researchers reflect and consider how their beliefs, values, experiences, interests and identities have shaped the research and how we might have been influenced and changed by the research we are doing (Willig, 2001).

Excerpt from Researcher's Reflexive Journal (20 February 2010)

It is over a year since I began to discuss with Alison and Jim how language, socialisation, class, gender, educational attainment, and position within society affects the choices we make in the research. E.g., what wellbeing means, how we internalise and externalise our inner and outer representation of what personal wellbeing means to us. Initially we all struggled with this but Alison's recent email confirms that she is beginning to get to grips with this process. My work on reflexivity for the PG CAP has also helped me enormously. I have read much more widely and appreciate how significantly your views and practices affect how you relate to people and ideas and how these in turn are affected by your upbringing, local environment, family structures, etc. These all impact on what we choose to discuss in our interviews and how we 'talk' about them. I was reflecting that Alison probably projects a certain image of herself to me because we like each other and thoughts or opinions often need to be aligned with image. Likewise I believe the reason Alison and Jim and I began this collaboration has much to do with our shared sense of identities. Alison and Jim certainly know I have strong views on the environment and I wonder how much this may have subconsciously influenced their narratives.

## 4.8.1: Epistemological reflexivity

Epistemological reflexivity is an important consideration, concerned with assumptions and how these have influenced the research questions, the methodology and the conclusions reached. Willig (2001) reflects that epistemological reflexivity obliges researchers to engage with questions such as: How the research question was defined, the effect this has on what can be 'found', and how the method of analysis has 'constructed' the data and the research findings. These lead to further reflections concerning how the research aims could have been explored through different channels and the degree to which this may have led to different findings. Epistemological reflexivity has informed this study by encouraging the researcher to "Reflect upon those assumptions (about the world, about knowledge) that we have made in the course of the research, and it helps us to think about the implications of such assumptions for the research and its findings" (Willig, 2001:10).

This approach to the structuring of knowledge creation has implications for the research and its findings. For example, the co-researchers and researcher all

identified themselves as working class with strong political affiliations. These shared identities are likely to have affected the topics chosen to explore and our 'reading' of the data. However, as with all research which seeks to investigate complex social phenomena, every effort was made to limit this impact through the utilisation of reflective practice and transparency.

#### 4.8.2: Reflective process

An important aspect of being involved with research is the capacity to reflect or think in-depth about one's own personal perspective of events, experiences and how one's background, gender, ethnicity, and education affects one's perception of things. However "Reflection is not solely a cognitive process; emotions are central to all learning" (Boud and Walker, 1998:194).

During the research training the co-researchers often struggled to feel confident in their ability to undertake reflective practices. Discussions between the co-researchers and researcher were undertaken via email and telephone to assist the reflective process. Reflective papers such as Moon (1999) and Brookfield (1994) were given an abridged read through with the co-researchers in order to help guide and inform their learning. Whilst these were considered to be useful reference points, the researcher's summary of Dahlberg and Halling (2001) appeared most salient.

Dahlberg and Halling (2001) identified different kinds of openness to be achieved by researchers such as open-mindedness, open-heartedness, and openness to dialogues. Dahlberg and Halling's (2001) perception of open-mindedness refers to the capacity to be receptive to research participants and their narratives. Open-heartedness is a type of reflexivity, in which researchers look at themselves in order to reflect upon their actions and behaviours. This includes being able to recognise attitudes, beliefs, inferences and biases that may affect the interpretation of the participants' comments and actions.

Excerpt from Alison's Reflective Diary (Email - 25 March 2008)

I enjoyed the training session with X, I enjoyed X's manner which didn't make you feel you had to choose your words carefully or be afraid of making a mistake. I was also reminded about making assumptions and prejudices, which I found very useful.

The research process sought to promote reflective spaces through co-researcher diaries and informal reflective conversations in which the researcher and co-researchers utilised personal reflection and shared dialogue. This facilitated a deeper exploration of lived experiences in relation to the research questions and conceptualisations of wellbeing. The reflective diaries kept by the co-researchers were not included as data; the data came solely from the interviews. However, they probably informed the co-researchers' responses to questions and may have informed Alison's analysis and interpretation. This assessment of the reflective diaries, however, is perforce subjective and cannot be realistically evaluated. However, the reflective diaries and the researcher's reflexive journal identified that the cycles of reflection engendered feelings of supported openness, camaraderie and mutual support. The co-researchers identified that the reflective process contributed to a sense that the study was a shared endeavour. In time, the co-researchers became increasingly aware of the importance their positionality, amongst other things, had on the research process.

#### 4.9: Ethical considerations

Application for ethical approval for the research was submitted to the National Research Ethics Service (NRES) in April 2007. Approval was granted on 2<sup>nd</sup> August 2007, with the MMU faculty ethics committee subsequently accepting this approval. Ethical practice within the research was guided by the British Psychological Society guidelines (2005), containing the following principles:

- Attention to facilitating the inclusion of people with language and communication impairments
- Informed consent

- Anonymity
- Confidentiality
- Privacy
- Ensure comfort and facilities necessary to be available for people with physical/mental/spiritual vulnerabilities
- Protection from harm for co-researchers, participants and the project researcher
- Free withdrawal from the research process

Research has an ethical duty to ensure that projects pay due consideration to encouraging and involving participants whose views may not be routinely consulted. The participants who found it difficult to understand verbal or written explanations, or had communication/language difficulties were not excluded from this study. Frequent breaks were taken as necessary for those with concentration or physical problems.

The researcher and co-researchers were trained to the NHS Research Ethics

Committee standard. A list of support organisation contact details was supplied for support during the research process. Both co-researchers and participants were given contact numbers for trained counsellors at MMU, should any problems arise.

Ethical practice was constantly reviewed as the research progressed and regular discussions were held to ensure support was available, should difficult or contentious issues arise. Ethical procedures were adhered to throughout the research process.

The following chapter frames the role of linguistic/historical influences in the PRA and on modern conceptualisations of wellbeing.

### **Chapter 5: Linguistic and Historical Contextualisation of Wellbeing**

#### 5.1: Rationale

As previously documented, little is known about the role of the historical and linguistic factors in the appropriation and mobilisation of wellbeing (Kingfisher, 2013; McMahon, 2006; Oishi et al., 2015). Additionally, little is understood about the role of historical and linguistic factors in conceptualisations of wellbeing (Hughes, 1988; McMahon, 2006; Oishi, 2012; Oishi et al., 2015).

This chapter aims to address this by exploring and synthesising the historical/linguistic conceptual development of wellbeing. The chapter starts with an investigation of the linguistic progression of wellbeing. The following linguistic section includes a section on the ontology and semantic extension of wellbeing and a discussion of the role of logomachy in the appropriation of wellbeing. This is followed by an investigation of the etymological roots of health, welfare, happiness and wellbeing. This documents the associations between wellbeing and its closest linguistic/conceptual terms in order to advance understandings of how modern conceptualisations of wellbeing have been mobilised.

### 5.2 Linguistic contextualisation of wellbeing

## 5.2.1 Ontology of wellbeing

A consideration of why wellbeing has attained its current eminence adds an important dimension to our understanding of how and why the ontology\* of wellbeing may be important in conceptualisations of wellbeing.

The first recorded use of wellbeing in written English was in the poem 'A Wife' in 1613 by Thomas Overbury (1581–1613) (Oxford English Dictionary (OED), 1971). The poem was first published in 1614, and by 1664 it had gone through eighteen reprints and was one of the most popular books of the 17<sup>th</sup> century in the UK (OED, 1971). It is likely this popularity helped disseminate the word wellbeing, if not the

concept to a wider population. However, the Greek origins of wellbeing assign it linguistic status and probably contributed to the fact that the first recorded use of wellbeing occurred in high literature (Hughes, 1988). This postulation is supported by the fact there are no documented examples of the word wellbeing in the language of the common people such as songs, rhymes or stories from this period (Hughes, 1988). Hughes (1988) explained this by arguing that wellbeing is primarily an "Academic literary concept which seeks to conceptualize it within polite boundaries, failing to acknowledge 'baser' understandings. It focuses on self-actualization for example and ignores sexual gratification or hero culture. It is perhaps this gentrification of the concept which alienates it from the average person" (p.28).

Others suggest that it is not so much wellbeing's Greek linguistic status which has facilitated the gentrification and dissemination of wellbeing, but rather its association with ancient philosophy and philosophers (Kingwell, 1998; McMahon, 2006; Walker and Kavedzija, 2015). Philosophy and philosophers have played an important part in the propagation of wellbeing within academic and literary language (Bergdolt, 2008; McMahon, 2006; Walker and Kavedzija, 2015). Philosophers have theorised about how and why wellbeing is conceptually important. Gewirth (1985) argued that the word wellbeing encompasses three fundamental human aspirations (health, wealth and happiness) which map onto efforts to make sense of our existence. These relate directly to the three states or dimensions of being: the physical world, the physical self and the non-physical self (Gewirth, 1985). In this three-dimensional ontology onto which wellbeing is mapped, sense is made of the physical world by aspiring to wealth, sense is made of the physical self by aspiring to health, and sense is made of the non-physical self by aspiring to happiness. It is important to note that the three concepts used to encompass fundamental human aspirations of health, wealth and happiness are the same concepts identified as being the etymological roots of wellbeing.

### 5.2.2 Logomachy

Logomachy is a term which stems from Greek and means "To fight about words" (Hughes, 1988:10). Logomachy is a natural semantic reaction to arguments "Concerning how we seek to define, understand and relate to contested concepts" (Gallie, 1964, cited in Hughes, 1988:157). This thesis presented a theoretical argument that governance mechanisms and the wider PMC had undertaken the mobilisation and moralisation of wellbeing as a means to promote personal responsibility and behaviour change. The role of logomachy can be understood to have played a part in this mobilisation which "Occurs as a result of socio-political competition for power through linguistic capitalism i.e. the appropriation and use of words as brand names or 'boosters' ... [which emerged as] the right of politicians to appropriate /manipulate language" (Hughes, 1988:10). Linguistic capitalism is disseminated through educational and state machinery as statements of authority. In this way "Words are mobilised, through powerful vested interests such as the media, academia and governments" (Hughes, 1988:93).

Whilst "Printing increases the mobility of words and accelerates their currency the widespread proliferation of access to the Internet effectively acts like a printing press increasing the rate of...semantic extensions (new meanings added to established words)" (Hughes, 1988:101). Thus the moralisation and mobilisation of wellbeing has been accentuated through the proliferation of the World Wide Web (Hughes, 1988).

# 5.2.3: Etymology – Wellbeing, health, welfare and happiness

The meaning of wellbeing is not fixed...its meaning will always be shifting. What it means at any one time depends on the weight given at that time to different philosophical traditions, world views and systems of knowledge.

(Ereaut and Whiting, 2007:7)

The etymological development of words offers understandings of how words were originally conceived (Epps, 2014). It is also suggestive of aspects which are perhaps no longer relevant to modern society or which are so deeply embedded within the concept that we only respond subconsciously to the inherent meanings (Epps,

2014; Healey-Ogden and Austin, 2010; Hughes, 1988). In order to begin to unravel the complexity which surrounds the concept it is necessary to return to the roots of the word wellbeing.

The Oxford English Dictionary (OED, 1922) definition of wellbeing identified the etymological roots of wellbeing as "happy, healthy, or prosperous condition, welfare". The following section will therefore offer a brief account of the etymological roots of 'prosperity', 'health', 'happiness' and 'welfare'. The understandings generated here will explicate the shared conceptual foundations, which have led to the increasing use of wellbeing in association with health within a PRA (Edwards and Imrie, 2008; Fleuret and Atkinson, 2007; Rose, 1999; Scott, 2012c; Seedhouse, 1999; Sointu, 2005).

To be 'prosperous' or 'prosperity' occurs in the definition of wellbeing and 'wellbeing' occurs in the definition of prosperity. This circularity of definition helps to create a strong association between the concepts. Etymologically, 'prosperity' is believed to derive from the root to 'prosper', which emerged from uncertain origins (OED, 1989). The earliest dictionary definitions of prosperity included references to being auspicious, favourable, flourishing, enjoying success and economic wellbeing and having vigorous and healthy growth (Guralnik, 1984; OED, 1989; Websters, 1971). There is no evidence to date of a linguistic association between prosperity and wellbeing; however, there exists a historical association (Mogilner et al., 2011; Oishi, 2012; Oishi et al., 2015). The historical association and circular definition connects the two concepts and remains embedded within normative understandings of both concepts (Hughes, 1988; McMahon, 2006; Ulman, 1962).

Historically, 'happiness' and wellbeing have been perceived by many as being coterminous (Kingfisher, 2013; McMahon, 2006; Oishi et al., 2015). However, semantic additions and extensions have led to these words having different nuances of meaning and understanding attached to them (Hughes, 1988; McMahon, 2006; Mogilner et al., 2011; Oishi, 2012). McMahon (2006) reported that happiness has its etymological root in happenstance. "The word for happiness is a cognate with the word for luck. *Hap* is the Old Norse and Old English root of

happiness, and it just means luck or chance" (p.17). Happenstance has a direct root in the concept of chance and has a long association with external happenings which are shaped by chance.

McMahon (2006) argued that this association with chance meant that historically to be happy was an uncertain and unlikely occurrence for all but the most select or lucky members of society. Furthermore, the association with happenstance suggested that the majority of human lives were "Likely to be beset with misery and ill fortune as neither happiness nor wellbeing were under the control or will of humans" (p.70). Historical evidence suggests that wellbeing and happiness share both historical and linguistic roots (Kingfisher, 2013; McMahon, 2006; Mogilner et al., 2011; Oishi, 2012).

Evidence clearly indicates that happiness and prosperity share conceptual roots with wellbeing (Guralnik, 1984; McMahon, 2006; Mogilner et al., 2011; Oishi, 2012). However, to date little research has been conducted to explore the conceptual roots between wellbeing, welfare and health. The circularity of definition which exists between wellbeing, health and welfare may reflect the lack of conceptual clarity which exists for these words (Alexander, 2008; Hughes, 1988; Williams, 1985).

The word 'welfare' has been widely acknowledged as that which formed part of the linguistic root of the term wellbeing (Friedichson, 1966; Guralnik, 1984; OED, 2002), particularly in reference to the philosophical, root of wellbeing, which tends to encompass the negative as well as the positive aspects of a person's life. Welfare is believed to derive from Old English 'wel faran' meaning faring or doing well (Crystal, 2010; OED, 1922), but which also alludes to how a person is faring as a whole, whether well or badly. Historically, the terms health, well-being, and welfare were used interchangeably (OED, Compact Edition, 1971).

The word 'health' has been widely acknowledged as that which formed part of the conceptual root of wellbeing (Guralnik, 1984; OED, 2002). Its etymological roots originate from 'heal' (Guralnik, 1984; McMahon, 2006), i.e. practices which restore the body. The first recorded use of health is documented circa 1000 (OED, 1989). Its

definition included 'soundness of body and the conditions in which bodily functions are efficiently discharged. During the Middle Ages, the meaning attached to health shifted and began to incorporate notions of spiritual, moral and mental soundness/wellbeing (OED, 1989). The association with wellbeing/welfare continued to expand on the arcane representation of health as that pertaining solely to the physical body (OED, 1989). However, the notion of health was still underpinned by assumptions that health referred to physical vigour, vitality and strength, ideas that are derived from the earliest understandings of health (Bergdolt, 2008; Porter, 1994; Rosen, 1958). By the 17<sup>th</sup> century, more holistic understandings perceived health as that pertaining to physical strength, efficient bodily functions and moral/mental soundness/wellbeing (Bergdolt, 2008; McMahon, 2006).

This review of the etymological roots of wellbeing helps us to understand how associations between wellbeing, prosperity, health and welfare underpin current conceptualisations of wellbeing. However, research suggests wellbeing is linguistically and historically closest to welfare and health. For example, they all share a concept of wholeness; however, there exists a greater depth of inherent meaning within the concept of wellbeing (Bergdolt, 2008; Donnal, 2006).

The following section explores semantic extensions and the interconnections between wellbeing, welfare and health in order to provide an emic perspective of the origin of the concept of wellbeing and its appropriation and assimilation into the English language.

### 5.2.4 The semantic extension of wellbeing

During the 17<sup>th</sup> century, semantic extensions became increasingly common (Hughes, 1988; Mugglestone, 2006; Pinker, 2007; Shay, 2008; Ulman, 1962). Semantic extensions refer to the process whereby what was ostensibly a new word appropriates a wealth of meanings from an older word, often one which has a number of different language stems (Epps, 2014; Hughes, 1988; Shay, 2008; Ulman, 1962). It is apparent that the conjoined word 'well-being' was an early semantic extension (Crystal, 2010; Guralnik, 1984; Hughes, 1988; Jawad, 2012).

The word "'well' is related to the German word 'wohl' which has its origins in the Indo-European branch of language development and referred to 'will'" (Guralnik, 1984:1613). During the 16<sup>th</sup> century, the religious schism which occurred in the Catholic Church had a profound impact on language development (McMahon, 2006; Pinker, 2007; Shay, 2008). Words of Germanic origin increasingly took root in those nations which sympathised with or converted to the new Protestant religions (McMahon, 2006; Shay, 2008). These new religions increasingly emphasised the importance of the 'will' of the laity as opposed to the will of the church (Harrison, 1990; Hill, 1985, 1990, 1997; McMahon, 2006). The theological and intellectual dissemination of these new religions did much to enhance the importance and influence of the Germanic language (Bergdolt, 2008; McMahon, 2006). The growing influence of the Germanic language may help to explain why an increasing number of conjoined words starting with 'well' took root during the 16<sup>th</sup> and 17<sup>th</sup> centuries (Shay, 2008). Alternatively, it may have been the result of the existing historical/linguistic relationship between 'well' and the German word 'wohl' (Guralnik, 1984). Whilst we may never 'know' which explanation is 'true', there is evidence to suggest wellbeing was increasingly associated with the 'will' of the laity as opposed to the 'will' of God as ministered through the Church (Bergdolt, 2008; McMahon, 2006).

The definition of the word 'being' is more complex, but at a base level refers to "Existence" (Guralnik, 1984:128). However, at a philosophical level its definition includes "Fulfilment of possibilities" (Guralnik, 1984:128). Donnal (2006) proposed that 'being' can communicate material existence, a sense of self and a sense of identity. Shakespeare's soliloquy from Hamlet, which starts 'To be, or not to be that is the question!' is a useful example of how literature in this period increasingly began to refer to mental processes and states of mind, questioning the nature and meaning of human existence (Wootton, 2015). Hamlet's soliloquy represents an internalised debate increasingly mirrored within society, which began to create words/expressions to articulate new ways of thinking about selfhood and the human condition (Crystal, 2010; Wootton, 2015).

Compound words were fashioned extensively in this period as literary agents sought to give greater voice to sentiments, beliefs, and contemplations that the Renaissance inspired (Crystal, 2010, 2014; Mugglestone, 2006; Shay, 2008). Combining 'well' with 'being' in its most rudimentary form may have been a means to confer a notion of one being 'well' in the physical sense. Conversely, it may be an example of semantic change which "Involves the development of polysemy, motivated by pragmatic inferencing that allows speakers and listeners to draw associations between particular ideas" (Epps, 2014:587). However, it may also be suggestive of an emerging reflective process, at the heart of 16<sup>th</sup>/17<sup>th</sup> century literary works, which addressed fundamental moral and ethical questions (Crystal, 2010; Shay, 2008; Wootton, 2003).

Jawad (2012) offers an alternative interpretation of the conjunction of 'well' and 'being' based on the work of Martin Heidegger (1889–1976) which explored the difficulties inherent in defining 'being'. Jawad (2012) perceived the illusive nature of 'being' as "neatly reduced by the addition of a prefix 'well'" (Jawad, 2012:17). This had the effect of altering the course of social policy into "The pursuit of some ideal, of a minimal state of wellness and happiness, generally identified in objective and money-metric terms" (Jawad, 2012:17).

Conceptually the self, defined as "That conscious thinking thing..." (Locke, 1690:100) flourished during the 17<sup>th</sup> century. Harrison (1990) suggests that 'self' became intimately associated with 'being' in part as the result of the schism which occurred in the Catholic Church during the 16<sup>th</sup> century. This upheaval in religious thought contributed to a newly found appreciation of Man's free will as a conscious being distinct from that which was the will of God (Hill, 1985, 1990, 1997; McMahon, 2006; Oishi et al., 2015). It is likely that this played a part in the flourishing of languages in tandem with the dissemination of complex concepts (Crystal, 2010; Mugglestone, 2006; Pinker, 2007). These were vital tools for expressing hitherto unconceivable or under explored cogitations (Crystal, 2010; Hughes, 1988; McMahon, 2006; Pinker, 2007; Wootton, 2003). These factors may have been instrumental in facilitating the utilisation of wellbeing in English literary texts (Hughes, 1988; McMahon, 2006). Thus, the foundations were laid for the

assimilation of wellbeing into the utilitarian movement of the 18<sup>th</sup> and 19 centuries (Hughes, 1988). This will be explored in detail in the following historical sections.

In summary, wellbeing was explored across linguistic domains, i.e. it included the influences of ontology, logomachy, etymology and semantics in order to explicate understandings of its linguistic development. The linguistic contextualisation also offered divergent and corroborative explanations in order to establish a clearer appreciation of the conceptual development of wellbeing.

Evidence suggests the semantic extension of 'well' to incorporate original notions of 'will' facilitated the conceptualisation of wellbeing as relating to the will of the laity as opposed to the will of the Church. This pragmatic inferencing may have facilitated an association between the 'will' and/or responsibility of the laity for wellbeing, rather than wellbeing as a divine gift administered through the Church.

Additionally, the linguistic contextualisation documented the association between wellbeing and health, wealth and happiness, concepts used to underpin and map human efforts to make sense of our existence. It also drew attention to wellbeing's linguistic status, historical associations and the gentrification process, which facilitated its assimilation into literary works. This is supportive of the premise identified in chapter one that wellbeing's linguistic/historical legacy has facilitated its mobilisation within professional and academic discourse whilst simultaneously undermining its transferability to lay usage.

It is to be hoped that this linguistic contextualisation offers an appreciation of the linguistic factors underpinning why wellbeing has achieved its current eminence within the language of academics and professionals, and moreover to demonstrate how wellbeing's linguistic distinction facilitates its mobilisation within the language and discursive practices of powerful voices.

The following sections will document the historical development of wellbeing and the role of historical factors in the appropriation and mobilisation of wellbeing.

### 5.3: Historical contextualisation of wellbeing

This historical contextualisation is applied to understanding the role of historical factors in the mobilisation of wellbeing. It is specifically concerned with investigating historical associations between wellbeing and personal responsibility in order to understand how and why historical factors facilitate current efforts to mobilise wellbeing within a PRA.

It also seeks to understand how and why historical changes in conceptualisations of wellbeing continue to influence current conceptualisations of wellbeing. These are important considerations, given that current conceptualisations of wellbeing are utilised to facilitate wellbeing's assimilation into a wider PRA.

### 5.3.1: Rationale for historical analysis

Chapter four documented the methods and procedures underpinning this historical analysis. The following section outlines the importance of historical analysis within the context of this study.

Historical analysis is increasingly recognised as an important tool in social research as a strategy for establishing a context or background for substantive studies (Epps, 2014; Gardner, 2010; Marshall and Rossman, 2009; Wyche et al., 2006). Historical analysis is often combined with other methods in order to provide "Insights into social phenomena" (Wyche et al., 2006:47). Inglehart and Klingemann (1998) used historical analysis and quantitative data sets to interpret "A substantial and enduring decline" (p.169) of the set point of wellbeing in Belgium. Inglehart and Klingemann (1998) proposed that a decline in wellbeing in Belgium was the result of "Historic factors that have afflicted Belgian society" (p.170). Based upon their findings they posited that varying levels of wellbeing recorded across nations were closely linked with the historical development of nations' political institutions.

How we understand and interpret history and its relationship to modern society directly influences the way we relate to communicating and interpreting words (Bowern and Evan, 2015; Hughes, 1988; Seymour, 2006; Ulman, 1962; Wootton, 2003). It is important therefore to recognise the role of history, power, politics,

rhetoric and linguistic development as integral components of a continuous discourse which links the past, present, and future and has a direct bearing on current ways of constructing meanings (DeWall et al., 2011; Epps, 2014; Kingfisher, 2013; Oishi et al., 2015).

Historical factors may play an important role in the mobilisation of wellbeing, but receive insufficient attention in the majority of wellbeing research because they are complex and difficult to unpack (Inglehart and Klingemann, 1998; Kingfisher, 2013; Kingwell, 1998; McMahon, 2006). The following section titled the 'Trans-historical account of wellbeing' is a historical analysis of selective periods of human history. This will document how historical events, human development and shifting normative values have influenced conceptualisations of wellbeing. The remainder of the chapter contextualises how shifting conceptualisations of wellbeing have been adopted and adapted within government practice in order to promote agendas consistent with shifting notions of governance and responsibility.

### 5.3.2 Trans-historical account of wellbeing

This section aims to provide a broad overview of the historical development of conceptualisations of wellbeing. The work of Bergdolt (2008), McMahon (2006) and Oishi et al. (2015) make important contributions to the shape of this chapter.

McMahon (2006) reviews wellbeing and happiness (as synonymous terms) in a historical-cultural analysis of conceptualisations, attitudes and approaches to wellbeing/happiness over time. However, there is a suggestion that McMahon (2006) offers a distorted interpretation of the history of Western philosophy in relation to wellbeing (Sirgy and Estes, in press) and rather one-sided summaries in the intellectual debate about happiness (Wright, 2014).

Oishi et al. (2015) undertook a historical/linguistic exploration of happiness/wellbeing to investigate how their meanings have shifted and changed over time. However, Oishi et al. (2015) have been challenged for not recognising in sufficient detail that the compass and language of happiness is constrained within a range of contexts (Walker and Kavedzija, 2015).

Bergdolt (2008), McMahon (2006) and Oishi et al. (2015) have all been criticised for conflating wellbeing with the closely aligned concepts of health and happiness (Bowern and Evan, 2015). This can obfuscate matters, particularly for those who perceive wellbeing as a distinct concept. It also demonstrates that complex concepts such as wellbeing can be hard to explore without reference to closely aligned concepts such as health, welfare and happiness (Bowern and Evan, 2015; Crawshaw, 2008; Forth, 2009; Seymour, 2006).

Bergdolt, for example, equates wellbeing with health, generally perceiving wellbeing as relating to inner balance and the equilibrium of bodily functions (Forth, 2009:413). Bergdolt argued wellbeing has always been associated with health, but over time it has become increasingly associated with physical health and beauty. In time, responsibility for health/wellbeing was "Placed...squarely on the shoulders of individuals...by leading moderate and balanced lifestyles" (Forth, 2009:413).

It is important to note that Bergdolt's (2008) account has been considered more a trans-historical account of the theoretical and practical conceptualisations of health than wellbeing (Crawshaw, 2008). Other criticisms of Bergdolt's (2008) historical synthesis include Boughan (2008), who claims Bergdolt overlooks important secondary sources written in non-Germanic languages; while Forth (2009) suggests Bergdolt focuses on the principle of dietetics and is primarily concerned with Western civilisations. Gardner (2010) criticises the utilisation of anecdotal evidence such as legends and old tales adopted by Bergdolt (2008). These are generally perceived as unreliable sources in traditional historical analysis. However, historical postmodernism accepts the use of anecdotal and literary evidence as helpful in illustrating how the theories and analysis presented in historical narratives operated in society (Gardner, 2010; Wyche et al., 2006; Yýlmaz, 2007). Additionally, these resources augment academic interpretations with more holistic and lay conceptualisations of complex concepts (Gardner, 2010; Wyche et al., 2006; Yýlmaz, 2007).

Current conceptualisations of wellbeing are rooted in the earliest meditations concerning the origins of health via balance and life force (Bergdolt, 2008). The earliest civilisations strove to explore those features of physical existence, which went beyond daily activity and fundamental health concerns (Epps, 2014). These were influenced by wider conceptual understandings of the interaction between humans and life-force balance (Bergdolt, 2008; Epps, 2014; McMahon, 2006). These ideas were influenced by deeply forged connections to the natural world, spirituality and the mystical world, and perhaps gave rise to a conscious appreciation of the more elusive condition of wellbeing (Bergdolt, 2008; McMahon, 2006). However, the expansion of the ancient advanced civilisations such as Egypt, Babylon, Mesopotamia and Persia led to health becoming more closely associated with a conscious concern for physical and spiritual wellbeing and happiness (Bergdolt, 2008; McMahon; 2006; Oishi et al., 2015).

In the ancient advanced civilisations, health and wellbeing were widely ascribed as divine gifts which could be negated through daemons or magicians (Bergdolt, 2008; McMahon, 2006). As McMahon (2006) noted, "Eudaimonia thus contains within it a notion of fortune—for to have a good daimon on your side, a guiding spirit, is to be lucky—and a notion of divinity, for a daimon is an emissary of the gods who watches over each of us" (p.3-4). McMahon (2006) and Walker and Kavedzija (2015) corroborate Bergdolt's assertion that in ancient Greece happiness/wellbeing was deemed beyond human agency, i.e. that which is controlled mainly by luck and the Gods.

Oishi et al. (2015) also substantiate Bergdolt's (2008) assessment that luck was considered to be of primary importance in the pursuit and attainment of wellbeing/happiness. This system of belief, which perceived health, welfare, wellbeing and physical form as products of religious faith persisted as the foundation of health dogma into early pre-classical Greece (Bergdolt, 2008; McMahon, 2006; Oishi et al., 2015). However, Greek historians and philosophers increasingly sought knowledge and understanding from the ancient civilisations and began to apply a scientific approach towards health and wellbeing (Bergdolt, 2008). They explored and codified ancient practices and began to catalogue illness and

anatomy in order to conceptualise new ways of thinking about health and wellbeing (Bergdolt, 2008; McMahon; 2006). By the 5<sup>th</sup> century BC, philosophers such as Democritus, Alcmaeon and Herodotus increasingly contested this pre-historic faith in the mystical nature of health and wellbeing. They introduced and disseminated notions that the concept of individual agency was key in matters of health and wellbeing (Bergdolt, 2008; McMahon; 2006).

These new theories facilitated the development of credible theories of health such as dietetics (Bergdolt, 2008). Proponents were able to establish that individual's poor or inappropriate behaviour resulted in sickness, which could be remedied by lifestyle and behavioural change. A similar perspective was applied to the growing number of wellbeing theories, and thus the connection between personal responsibility and wellbeing/happiness was cemented (Bergdolt, 2008; McMahon, 2006). This theoretical perspective grew in importance and eventually this belief system permeated from physicians and philosophers to lay people (Bergdolt, 2008).

The notion of personal responsibility was not a new idea, however, having been utilised by ancient societies alongside the more traditional understanding of health and wellbeing as a gift to be given or withheld by a higher force (Bergdolt, 2008; McMahon, 2006; Walker and Kavedzija, 2015). This duality is ancient and rooted in both practical and theoretical understandings of how the environment in its entirety influences human balance and harmony. The paradigm of personal responsibility, however, became enshrined in dietetics, a doctrine that emerged to challenge other theories of health and wellbeing. In ancient Greece, dietetics increasingly underpinned many of the cultural understandings of those philosophers preoccupied with the human condition (Bergdolt, 2008; McMahon, 2006). Dietetics as "An exhortation to take personal responsibility for one's health" (Bergdolt, 2008:5) has continued to shape conceptualisations of the intersections between holistic health, wellbeing and human agency (Seedhouse, 2001).

Thus far, I have primarily addressed how ancient beliefs have helped to shape modern conceptualisations of health. This can be seen as underpinning wider notions of wellbeing, as ancients perceived health within a framework which

incorporated a sense of the spiritual, mental and ethereal conditions under which the human species existed (Bergdolt, 2008; McMahon, 2006). The concept and terminology of wellbeing, however, was first documented and widely established during the Greek civilisation (McMahon, 2006; Ryan and Deci, 2001; Tiberius, 2004).

Aristotle pursued understandings of wellbeing through a eudaimonic perspective that is wellbeing as a synonym for equilibrium or harmonious balance in one's life (Ryan and Deci, 2001). Other philosophers such as Aristippus pursued understandings of wellbeing through a hedonic perspective, perceiving it as a striving for personal pleasure and happiness (Ryan and Deci, 2001). It is in this period that history documents wellbeing as having a health-related inflection, a "Philosophical and cultural-historical dimension" (Bergdolt, 2008:33) and a conceptual association with the environment, society and religion (Bergdolt, 2008; McMahon, 2006).

"The assumption that physical and spiritual wellbeing was also dependent on the individual disposition became increasingly prevalent in the course of the 4<sup>th</sup> century BC" (Bergdolt, 2008:37). However, alternative theories and practices challenged this assumption by suggesting that happiness/wellbeing was "beyond human agency... and highly contingent upon external conditions (Walker and Kavedzija, 2015:8). Aristotle and Plato were influential in establishing limits to the viability of applying the tenet of personal responsibility within health and wellbeing contexts (Bergdolt, 2008; McMahon, 2006). Aristotle and Plato argued that the holistic nature of existence meant that it was also the responsibility of administrative states to manage those elements of human existence which went beyond that which individuals could reasonably be expected to be responsible for, such as the built environment and its impact on wellbeing (Bergdolt, 2008; McMahon, 2006).

### 5.3.3: Changing conceptualisation of wellbeing and welfare

Attitudes to wellbeing/happiness shifted gradually over time, but accelerated during particular periods (McMahon, 2006, 2006b, 2010; Oishi, 2012; Oishi et al., 2015). Prior to the medieval period, wellbeing aside from that pertaining to physical health was primarily perceived as beyond the responsibility of man (McMahon,

2006; Oishi et al., 2015; Walker and Kavedzija, 2015). McMahon (2006) argued that the historical/linguistic shift in the meaning attached to wellbeing followed a trajectory. This started with wellbeing/happiness as being the province of the Gods, whereby their divine intervention was necessary for humans to experience it. However, as religion replaced mysticism, wellbeing/happiness became attainable through human endeavour (Bergdolt, 2008; McMahon, 2006; Walker and Kavedzija, 2015).

During the Middle Ages, the association between wellbeing and responsibility expanded (Kingswell, 1998; McMahon, 2006; Oishi et al., 2015). In particular, there was a growing sense that monarchs had a duty and responsibility to consider their subjects' welfare (Bergdolt, 2008; McMahon, 2006). The Reformation in the 16<sup>th</sup> century also heralded a period in which attitudes and approaches to wellbeing/happiness and welfare shifted radically (McMahon, 2006). McMahon (2006) argued that during this period the rise of religious expression, Protestantism and free will meant wellbeing/happiness began to be thought of as something which humans could and should strive for independent from the will of God (Bergdolt, 2008; McMahon, 2006; Oishi et al., 2015), whilst the provision of welfare to the poorest in society emerged as a societal duty (Mokyr, 2010).

The OED (1971) interpreted the term 'welfare' as having two meanings: how well people live, or what is done by others to help those in need. The Poor Law (1598/1601) was the first legislative statute introduced in England to provide welfare for the very poorest in society (Mokyr, 2010). The concept of welfare was, however, well established amongst the population of England, dating back to Anglo-Saxon rule (Mokyr, 2010). However, as attitudes changed and meanings shifted to take account of this, words which described complex ideas also shifted (Bowern and Evan, 2015; Crystal, 2010; Oishi et al., 2015; Ullman, 1962).

Between 1650 and 1800 the growth of Puritanism, nationalism and desacralisation led to an increase in 'conscientious industry' (Wootton, 2015) and self-improvement (Briggs, 1970, 2000). Politically reformist notions gathered credence and the mercantile interests of society began to perpetuate altered understandings

of prosperity, wellbeing and welfare (Briggs, 1970, 1983, 2000; Schama, 2002; Wootton, 2015). Romanticism, humanism and idealism, for example, were key concepts which began to enlighten the existence of the gentry and mercantile classes (Black and Levy, 1969; Briggs, 1970, 1983; McMahon, 2006; Wright, 2014). Traditionally, philanthropic observances were perceived as enhancing 'welfare'. The concept of welfare was used to envisage assisting the less fortunate to attain a level of existence which went beyond subsistence and a life of abject poverty (Briggs, 1970, 1983; McMahon, 2006; Trevelyan, 1973). Critics, however, argue that this perception of welfare is an overly romanticised view of the situation (Jawad, 2012; Royle, 2012).

Alternative theories attempt to account for the emergence and promotion of welfare in 19<sup>th</sup> century England (Walsh et al., 2000). These include Jawad (2012), who claimed it is more plausible that religious duty and observance were the driving factors which impelled the more affluent members of society to provide basic assistance to their fellow humans. Other theoretical perspectives include Stone (1984, cited in Gerard, 1987), who claimed philanthropic acts of welfare provision were pastimes or hobbies of the well to do. Martin (1965, cited in Gerard, 1987) argued that the ruling classes provided welfare assistance to reinforce their socio-political hegemony, whilst Bogardus (1922) claimed that unhealthy workers were not productive and the deliverance of welfare acted as a means to preserve the ruling elite's pecuniary interests. Ballas and Dorling (2013), however, noted that "In Britain it was not until the decades after the 17<sup>th</sup> century European Enlightenment that the focus for understanding happiness shifted to how people actually felt as they went about their earthly lives" (P.467).

### 5.3.4: The period of Enlightenment

Wellbeing was increasingly perceived as a fundamental criterion for human advancement during the Enlightenment. The adoption and wider dissemination of the term wellbeing emerged during the latter part of the 18<sup>th</sup> century as philosophers began to question the nature of wellbeing, human existence and the

role of governments and individuals in promoting it (Bergdolt, 2008; Kingwell, 1998; McMahon, 2006; Oishi et al., 2015; Wright, 2014).

Utilitarianism\*, "The theory of the greatest happiness of the greatest number" (Briggs, 1970:323), was perhaps the single biggest contributor to the mobilisation and dissemination of wellbeing conceptually in the modern era (McMahon, 2006; Scarre, 1995; Wright, 2014). However, Briggs (1970) argued that "Utilitarianism... has been perceived historically as important in advancing the cause of self improvement and industrialisation but for failing to support the holistic development of human beings in complex societies" (p.323).

Jeremy Bentham (1748–1832) is widely regarded at the founding father of the British Utilitarian movement (Crisp, 1998; Mitra, 2002; Mokyr, 2010; Morgan, 2010). Bentham's 'utility' (usefulness) equated to human 'happiness' or hedonic wellbeing (Crisp, 1998; Ryan and Deci, 2001). Under Bentham, utilitarians regarded society as a collective of individuals who pursued their self-interest and were governed by a government whose purpose was to balance individual and public happiness (Bentham, 1789). Bentham argued that citizen wellbeing should also be the legitimate responsibility of government and proposed that precept hedonic wellbeing should be the fundamental foundation upon which to organise social and political life (Bentham, 1789; Crisp, 1998).

John Stuart Mill's 'On Liberty' (1859) expanded Bentham's position and was an important treatise on those preconditions which were considered necessary for the fulfilment of human wellbeing (Mitra, 2002). David Hume's 'Treatise on Human Nature' (1738) associated utility (wellbeing) with welfare and happiness (Wootton, 2015). John Stuart Mill in his role as an MP and the wider utilitarian movement were important influences on British legislation and policy (Churchill, 1956; Mokyr, 2010; Morgan, 2010).

Utilitarianism challenged the prevailing social and political ethos of laissez-faire, which held as its central tenet that government should interfere as little as possible in the affairs of its citizens (Mokyr, 2010; Schama, 2001). The utilitarian movement strengthened the demands of those who perceived that welfare, or as the

utilitarians increasingly called it 'wellbeing', should perhaps approximate more closely to the notion of wellbeing for the poor as well as the rich (Kingwell, 1998; Scarre, 1995). This emerging opinion was held by people who began to perceive that societal ills such as poverty and ill health were not necessarily a result of lifestyle choices and irresponsible behaviour (Porter, 1994). A more empathetic view of the impoverished and a growing appreciation of the socio-political climate of reform engendered a realisation that social justice was primarily the responsibility of those with political power (Mitra, 2002; Mokyr, 2010; Morgan, 2010).

Thus, the utilitarian movement was instrumental in shifting perceptions of wellbeing from that which was a matter for personal responsibility to that which was also a state responsibility (Kingfisher, 2013; Kingwell, 1998; McMahon, 2006; Oishi et al., 2015). The emergence of the utilitarian movement under Bentham encouraged a shift in how wellbeing was conceptualised. A minority of the educated elite now perceived wellbeing as the legitimate responsibility of government (Crisp, 1998). This normative shift was accelerated by the French and American revolutions (Bergdolt, 2008; McMahon, 2006).

The French and American revolutions disseminated political, philosophical and practical ideals of state responsibility to provide systems of welfare which sought to promote wellbeing/happiness (McMahon, 2006; Oishi et al., 2015; Poynter, 1969; Schama, 2002). The view that humans should be free to strive for happiness was taking shape amongst enlightened industrialising nations (Bergdolt, 2008; McMahon, 2006; Oishi et al., 2015; Wright, 2014). This was to become a major force in western culture with the inclusion of 'The pursuit of happiness as an inalienable right' into the American Declaration of Independence (1776/8) (Easterlin, 1974; Ho, 2005; Oishi et al., 2015; Wright, 2014).

John Stuart Mill (1812–1890), Mary Wolstencroft (1759–1797), Adam Smith (1723–1790) and Rev Thomas Malthus (1766–1834) were amongst an expanding number whose theoretical treatises illuminated new avenues to explore connections between the nation state, individuals and societal responsibilities (Bergdolt, 2008;

McMahon, 2006; Porter, 1994; Wootton, 2015). These led to ideas and assertions that the nation state had a responsibility for the health and welfare of the individuals who made up its constituent parts (McMahon, 2006; Morgan, 2010; Porter, 1994).

During the Enlightenment, perceptions of wellbeing shifted from that "Best left to the individual to that which is the appropriate purview of government policy" (McMahon, 2006:71). Indeed, Oishi et al. (2015) noted "That the Enlightenment movement of the 18<sup>th</sup> century shifted the main question from the religious "How can I be saved?" to the secular "How can I be happy?" (Oishi et al., 2015:8).

# 5.3.5: Wellbeing, self-improvement and shifting normative values

The early 19<sup>th</sup> century heralded a period in which self-help pamphlets proliferated (Briggs, 1970). Injunctions such as "Heaven helps those who help themselves" (Smiles, 1832; cited in Briggs, 1970:324) reflected the mood of the rapidly industrialising UK. Those in positions of authority promoted self-improvement as a sensible route to improved health, social behaviour and wellbeing, and as a means to inculcate moral fortitude (Briggs, 1983; McMahon, 2006; Walsh et al., 2000). Social enterprises such as the temperance movement, charity schools and reading groups were established with the intention of improving people's minds and spirits as well as their bodies (Jawad, 2012). The rise of individualism (Briggs, 1970) and the work of writers such as Adam Smith and Rev Thomas Malthus supported those who argued that welfare was not the responsibility of the state (Briggs, 1970; McMahon, 2006). There was, however, a growing sense that life should encompass more than mere existence and that life even for the working classes could and should hold some meaning and fulfilment (Briggs, 2000; Bergdolt, 2008; Zanden and Luiten, 2014).

Conceptualisations of wellbeing are interwoven with histories of what constitutes liveable lives. These are culturally bound up within time-related notions of how lives are expected to be lived (Bergdolt, 2008; McMahon, 2006; Walsh et al., 2000). Periods within history have generated quite different notions of previously well-accepted theories of existence (Briggs, 2000; Bergdolt, 2008; McMahon, 2006). For

example, the changing attitudes in the early 1800s towards the practice of slavery were at odds with centuries of accepted practice (Schama, 2001).

Changing normative values like these were dispersed across social boundaries. They encouraged some members of the aristocracy to argue that workers lived and worked in similar conditions to that of some slaves (Birn et al., 2009; Briggs, 2000; Schama, 2002). However, some argue that these progressive views were at this time still uncommon (Walsh et al., 2000). Many people with wealth and in positions of power retained an antipathy towards the working classes (Gunn, 2006; Schama, 2002), whilst evidence documents that in parliament, both political parties (Whigs and Tories) were politically positioned against improving living conditions, extending welfare and/or political representation to the working classes (Briggs, 1970, 1983, 2000; Schama, 2002; Walsh et al., 2000).

However, the extension of the right to vote in the 'Great Reform Bill' (1832) resulted in the new industrial cities winning the right of representation in parliament and the political process for the first time. This gave urban middle-class men increasing participation in the political process and in shaping the cultural ideology of the nation (Mokyr, 2010; Schama, 2002; Walsh et al., 2000).

In addition, the working classes began to form more cohesive groups and organisations to fight against the most repressive conditions (Birn et al., 2009; Briggs, 2000; Clark, 1986; Schama, 2002). Research undertaken by philanthropists, campaigners and reformers such as Edwin Chadwick and Charles Dickens during the 19<sup>th</sup> century began to change the way people thought about welfare, wellbeing and the nature of people's existence (McMahon, 2006; Walsh et al., 2000). The flourishing industrial revolution and Britain's reliance upon the working classes to maintain it and hence the wealth, power and prestige of the nation (Porter, 1994) became an important political realisation for all classes (Briggs, 2000; Mokyr, 2010; Morgan, 2010; Schama, 2002).

Socio-economic legislation and civic works were undertaken to enhance certain aspects of life, which directly influenced the ability of the nation to maintain its position as a global power (Mokyr, 2010; Porter, 1999; Rosen, 1974). National

policy directives concerned with improving the lives of all citizens became increasingly associated with the targeted provision of welfare (Birn et al., 2009; Rosen, 1958; Walsh et al., 2000). The following four sections will detail the development of welfare provision in the UK and highlight its contribution to shifting conceptualisations of wellbeing across UK society (Stenner and Taylor, 2008; Taylor, 2011).

#### 5.3.6: Welfare

"Man's desire for happiness, health and wellbeing was bound up more closely with the hedonistic striving of the community for 'welfare'" (Bergdolt, 2008:220). Philanthropic ideals of welfare increasingly envisaged human existence as more than survival, procreation and duty (Briggs, 1983, 2000; Royle, 2012; Schama, 2002). Alternative interpretations of the role of philanthropy in shifting conceptualisations of welfare propose that philosophical ideas were far removed from the legislative and practical implementation of welfare (Royle, 2012; Schama, 2002). England's first legislative attempts to ensure some provision of welfare was made available for the poorest in society were the poor laws set out in acts of Parliament in 1572, 1597/8 and 1601 (Webb and Webb, 1927). However, Royle (1987) argued that the poor laws were less concerned with philanthropy and acted as an "Instrument of control in an age of population growth and increasing vagrancy" (p.172). In 1834, the poor laws were amended to ensure that this piecemeal system of welfare was offered only to those whose existence was threatened. This ensured that welfare provision did not support those who could support themselves (Morgan, 2010; Royle, 2012; Schama, 2002).

However, it is generally accepted that philosophical, literary and political debate began to exert an influence on public attitudes towards the provision of welfare and notions of wellbeing (Bogardus, 1922; Kingwell, 1998; McMahon, 2006; Mokyr, 2010; Poynter, 1969). However, there are contentions to whether improvements in "the provision of welfare were primarily achieved through the work of philanthropists and philosophic radicals" (Rosen, 1958:175) or as a result of religious and faith-based bodies (Jawad, 2012).

The social, political and economic legislation of the 19<sup>th</sup> century did, however, begin to regulate the worst aspects of life for the under privileged. Many argue that this began to encourage a new public perception of the working classes as citizens of the Commonwealth (McMahon, 2006; Mokyr, 2010; Price, 1999; Royal, 1987). There was a growing commitment to raising expectations, access to opportunity and providing welfare to less fortunate members of society (Mokyr, 2010; Morgan, 2010; Royle, 2012). These incorporated ideals such as encouraging citizens to strive for a life well lived so that "Issues of social welfare became tightly connected with issues of the personal well-being of citizens" (Stenner and Taylor, 2008:416).

# 5.3.7: Beginning of the shift from welfare to wellbeing

The 19<sup>th</sup> century heralded a shift in the historical development of wellbeing, as it became less associated with the work of classical Greek scholars (Ryan and Deci, 2001). Documentary evidence from the time suggests what could be considered as a meaningful existence, particularly for the working classes, became more empathetic (Briggs, 1983, 2000; McMahon, 2006; Mokyr, 2010; Royle, 2012). It has been suggested that this facilitated a societal shift from providing welfare which supported subsistence living to wellbeing which supported higher human needs (Kingwell, 1998; McMahon, 2006). This is challenged by Stenner and Taylor (2008), who argued that the shift from welfare to wellbeing began with the emergence of psychology as a social science in the late 19<sup>th</sup> century.

Psychology was first acknowledged as a legitimate science in Germany three years before Bismarck proposed one of the most important and comprehensive rafts of welfare reforms introduced in 19<sup>th</sup> century Europe (Stenner and Taylor, 2008). Stenner and Taylor (2008) argued that the adoption of psychology and welfare reforms led to a wider concern to conceptualise welfare not only as a social endeavour, but also as fundamentally linked with the wellbeing of individuals. The emergence of state welfare systems led to the adoption and growth of other new scientific disciplines such as sociology, social policy and social administration (Stenner and Taylor, 2008). Within these disciplines the prevailing understanding of social welfare\* "Proximally denotes those 'provisions' a society puts in place to

guarantee a minimum level of 'life quality' or 'well-being' on the part of its citizens, then this points to a clear relationship between social welfare and the well-being of citizens" (Stenner and Taylor, 2008: 417/8).

### 5.3.8: The influence of the welfare state on conceptualisations of wellbeing

It is widely believed that the term 'welfare state' first emerged in the UK during World War II (Finlayson, 1994). The welfare state "Authorises: what services should be provided, how they should be provided, by whom and on what authorisation and the nature of provision" (Staeheli, 2001:179). There are diverse understandings of what constitutes a 'welfare state'. However, at a fundamental level, there is consensus that welfare state institutions and policies are intended to ensure a population's basic needs are met (Walsh et al., 2000).

Originally, the provision of welfare described systems of socio-economic provision implemented during the 19<sup>th</sup> century (Mokyr, 2010; Walsh et al., 2000). Welfare provided shelter and subsistence-level food, promoted the notion of duty to the monarch and country, and in places offered schooling in religious education (Mowat, 1952; Gunn, 2006).

During the 20<sup>th</sup> century, understandings of what constituted welfare advanced and the emergence of new political parties, such as the Labour Party and Liberal Democrat Party heralded new ideological principles and pressed for greater state intervention in the lives of citizens (Staeheli, 2001, 2012). This led to increasing socio-economic and political pressures being exerted upon the government to extend welfare beyond its former conceptualisation (Mowat, 1952). Mowat (1952) speculated that this fundamental shift in ideological thought brought into existence the now widely accepted rationale of providing welfare as a social responsibility into the national consciousness (Staeheli, 2012; Walsh et al., 2000). Mowat (1952) argued that the emergence of the 'welfare state' in the UK was in direct response to the need to improve conditions for working people. Welfare provision formed part of "The duty of the state to maintain the wellbeing of all its members by

guaranteeing them a minimum of income and services and insuring them against the hazards of sickness, unemployment and old age" (Mowat, 1952:22).

"'Wellbeing' and 'welfare' are often bracketed together, in particular wellbeing and state-welfare" (Veenhoven, 2000:91). Evidence from other western societies indicates that this may be part of a concerted effort on the part of governments to moderate and control access to the welfare system. In New Zealand, the term "Welfare was withdrawn from health literature because of its underlying association with 'Government Giving'" (Murdoch, 1999:6). This may help to explain why wellbeing is now increasingly used in preference to 'welfare' (Birn et al., 2009; Ferguson, 2007; Kingfisher, 2013; Murdoch, 1999).

### 5.3.9: From welfare to wellbeing

Froggett (2002) identified the 1945–1979 period as the era of the 'Classic Welfare State'. During this period the provision of societal and individual wellbeing were envisaged as part of the state's responsibility to co-ordinate social welfare (Froggett, 2002; Stenner and Taylor, 2008). During the Classic Welfare State period the coordination of social welfare encouraged governments to measure the impact of welfare policies on their citizens (Froggert, 2002; Walsh et al., 2000; Weir, 1997a). Proxies for a life well lived such as QoL and life satisfaction were increasingly used to measure the efficacy of welfare interventions (Froggert, 2002; Walsh et al., 2000). This meant that conceptualisations of welfare became proximally associated with QoL and life satisfaction (Walsh et al., 2000; Weir, 1997a).

Shifting political dynamics continued to stimulate new conceptualisations of welfare, e.g. 'Social Democratic Welfarism' (Froggett, 2002; Staeheli, 2001, 2011; Walsh et al., 2000). Social Democratic Welfarism predominated under successive Conservative governments from 1979 to 1996 and the 1996–2009 Labour government (Froggett, 2002). Social Democratic Welfarism emerged in response to efforts to shift societal conceptualisations of welfare from the 'old welfare' model towards a new market led and private model of providing welfare (Froggett, 2002; O'Brien and Penna, 1998; Walsh et al., 2000). The adoption of Social Democratic

Welfarism illuminates changing attitudes to welfare, and what it signified within state machinery. It also marks a period in UK history when changing conceptualisations of welfare explicitly mobilised its historical/linguistic association with wellbeing (Stenner and Taylor, 2008).

The following section starts with an international example of governance efforts to mobilise wellbeing within the context of welfare provision by replacing the language of welfare with the language of wellbeing.

## 5.3.10: A 'welfare to wellbeing' strategy

A 'welfare to wellbeing' strategy was promoted in New Zealand by the Department of Social Welfare to reduce the incidence of long-term welfare dependency (Murdoch, 1999:5). Concerted efforts were made to mobilise vested interests to disseminate the strategy across the nation (Murdoch, 1999). The business sector and media were broadly supportive of the strategy, whilst other government departments adopted it as a "Vehicle for mobilising and focusing efforts" (Murdoch, 1999:7). Other sectors, however, challenged the appropriateness of the strategy, suggesting structural rather than agentic factors were more important in the campaign to reduce welfare dependency and improve wellbeing (Howell, 1997; Robinson, 1997, cited in Murdoch, 1999). The branding and active mobilisation of the concept of wellbeing utilised by a New Zealand state department highlights the utilisation of the WBA in ageing neo-liberal westernised societies (Murdoch, 1999). The WBA seeks to replace the language of welfare with the language of wellbeing, with the stated intention of reducing citizen reliance on what many perceive as overburdened welfare systems (Ahmed, 2010; Ferguson, 2007; Miller and Rose, 2008; Peck, 2013).

This overt approach has not been undertaken in the UK. However, the DoH's 'Wellbeing: Why it matters to health policy' (2014) publication identified that wellbeing is associated with positive health behaviours. Furthermore, it claims that focusing policies on wellbeing "Could reduce the healthcare burden associated with an ageing population" (DoH, 2014:7). Replacing the language of welfare with the

language of wellbeing encouraged a shift in how wellbeing is conceptualised across government departments (Murdoch, 1999; Peck, 2013). During the Classic Welfare State conceptualisations of wellbeing focused on the structural determinants of wellbeing such as economic progress, health promotion and educational attainment (Stenner and Taylor, 2008). Replacing the language of welfare with the language of wellbeing meant wellbeing became proximally associated with strategies which focused on inculcating a sense of personal responsibility and the need to undertake behavioural change (Ahmed, 2010; Peck, 2013; Rose, 1995; 1999; Seedhouse, 1995; Sointu, 2005).

## 5.3.11: The emergence of wellbeing into the collective consciousness

The World Health Organisation (WHO) is credited with introducing the modern day concept of wellbeing into the collective consciousness (Cameron et al., 2008; Crawshawe, 2008) by focusing on the positive nature of health as the responsibility of all (Seedhouse, 2001; Stewart-Brown, 1998).

The WHO defined health as: "A state of complete physical, mental and social well-being and not merely the absence of disease or infirmity" (WHO, 1946:online). This situates wellbeing within the context and auspices of health, and explicitly takes into account the psychological and social determinants of health as well as the biological determinants of health (Knight and McNaught, 2011). This definition of health, which shifted the parameters of health discourse, helped to propel wellbeing to prominence in policy and planning (Cameron et al., 2006; Knight and McNaught, 2011; Stewart-Brown, 1998).

The WHO is the only multi-national agency which has the power and responsibility for improving the health and wellbeing of the world's populations (Stewart-Brown, 1998; Tones et al., 1990). As such, it is amongst the loudest and most influential of 'expert voices' which have been mobilised to reassert the symbiotic relationship between health and wellbeing (Cameron et al., 2006; Crawshawe, 2008). The current manifestation of the symbiotic relationship between health and wellbeing has existed since wellbeing was first conceptualised in early civilisations (Bergdolt, 2008; McMahon, 2006; Oishi et al., 2015). However, the WHO's endorsement of

this relationship enhanced the conceptual profile of wellbeing (Kingwell, 1998; Knight and McNaught, 2011).

The WHO's power as a universal 'expert voice' has ensured that wellbeing within a social model of health enjoys prominence as an adjunct to health (Cameron et al., 2006; Knight and McNaught, 2011). Since its inclusion in the WHO's universal declaration, wellbeing is increasingly employed by policy makers as a generic term that embraces physical, mental and emotional health (Knight and McNaught, 2011). Wellbeing has also become increasingly prevalent in government policy documents (Knight and McNaught, 2011). Examples include 'Every Child Matters' (DoH, 2003), 'Choosing Health' (DoH, 2004) and 'Health, Work and Well-being – caring for our future' (DWP and DoH, 2005). Similarly, wellbeing indicators are now included in The Public Health Outcomes Framework, the NHS Outcomes Framework and the Adult Social Care Outcomes Framework (DoH, 2014).

The growing importance of wellbeing as a function within society has been widely documented (Atkinson and Joyce, 2011; Atkinson et al., 2012; D. Bok, 2010; S. Bok, 2010; Ferguson, 2007; Seligman, 2009; Scott, 2012a). During the 21<sup>st</sup> century, the concept of wellbeing transitioned from its position as an adjunct to health and is increasingly utilised as a proxy or holistic indicator of individual and societal welfare (Ferguson, 2007; ONS, 2013; Scott, 2012a; Scott and Bell, 2013). It has also increasingly come to the fore in social policy documentation (Ferguson, 2007; Cameron et al., 2008; Scott, 2012b). The following two sections outline wellbeing in relation to the PRA and the challenges this places on governments and citizens.

### 5.3.12: The wider application of wellbeing

The latter part of the 20<sup>th</sup> century and the beginning of the 21<sup>st</sup> century has witnessed wellbeing join "The pantheon of contested concepts such as democracy, freedom, justice" (Hughes, 1988:33). The wider application and use of the term wellbeing has now become central to worldwide health agendas (United Nations International Children's Emergency Fund (UNICEF), 2009; Organisation for Economic Cooperation and Development (OECD), 2011).

Increasingly, wellbeing is perceived as an active process which concerns how well people function as well as how well they feel (Atkinson et al., 2012; D. Bok, 2010; S. Bok, 2010; Haworth, 2008; Scott, 2012a). Thus, the concept of wellbeing has increasingly been utilised within the PRA. This focuses on promoting the role of citizens as active agents of their own wellbeing, underscored by a call to "Perceive wellbeing as an active process of 'doing' rather than the passive role of 'being' " (Stenner and Taylor, 2008:417).

Dinham (2006) asserted that wellbeing is a power term used in government rhetoric alongside 'community', 'participation', and 'empowerment' to endorse behaviour which motivates individuals to be compliant and dutiful citizens. Whilst it continues to serve that purpose, it will continue to be utilised within governance frameworks which promote more responsible citizenry (Dinham, 2006; Ferguson, 2007; Rose, 1999; Stenner and Taylor, 2008; Scott, 2012a; Taylor, 2011).

Ageing populations and increasing health expenditure consumes an increasing proportion of a country's national revenue (UNICEF, 2009; OECD, 2011). These factors have contributed to increasing injunctions for citizens to help themselves and take responsibility for their wellbeing (Ahmed, 2010; S. Bok, 2010; Ferguson, 2007; Scott, 2012a; Taylor, 2011). Westernised governments has been increasingly promulgated a PRA for wellbeing in the 21<sup>st</sup> century (Ferguson, 2007; Peck, 2013; Rose, 1995, 1999; Scott, 2012a). This agenda encourages citizens to improve their wellbeing through individual endeavours rather than through the actions of the state (Ferguson, 2007; Fullager, 2002; Peck, 2013; Rose, 1995, 1999; Sointu, 2005).

## 5.3.13: Challenges of the 21st century

The 21<sup>st</sup> century faces different challenges to those of the 19<sup>th</sup> and 20<sup>th</sup> centuries. This includes diverse aspects such as the globalised economy, global fears concerning sustainability, displacement through ecological and environmental disasters and the dispersal of economic migrants and refugees (Ballas and Dorling, 2013; D. Bok, 2010).

Traditional disciplines long associated with wellbeing such as economics, psychology and philosophy have prioritised wellbeing theories, which fail to take into account ecological conceptualisations of wellbeing (Abdallah and Johnson, 2008; Newton, 2007). This has led to growing criticisms of traditional approaches to wellbeing, conceptualisations of wellbeing and the role of the WBA (Ahmed, 2010; Kingfisher, 2013; Peck, 2013). However, as documented in chapter two, wellbeing research is increasingly undertaken by diverse disciplines such as ecology, social geography, sociology, health and political science (Ballas and Dorling, 2013; Galloway, 2006; Scott, 2012c). The global challenges referred to above are likely to influence global and local conceptualisations of complex concepts such as wellbeing. Those working within ecological frameworks, for example, have sought to develop conceptualisations of wellbeing which recognise sustainability theories (Pickering, 2007). The Millennium Ecosystem Assessment (2003), for example, was developed to provide a framework for wellbeing theorists which contributes to understanding the links between wellbeing and the natural environment (Newton, 2007). The link between wellbeing and sustainability may already be shifting conceptualisations of wellbeing. Sustainability theories argue that current conceptualisations of wellbeing and wider wellbeing discourse ignore the importance of connectivity with the natural world (Abdallah and Johnson, 2008; Berry, 2014; Pickering, 2007; Newton, 2007).

Wilson's 'Biophilia Hypothesis' (1984) (see chapter 2) suggests that human identity and personal fulfilment is dependent on our relationship to nature. This theory argues that wellbeing is not confined to the human need to "Exploit the material components of environment for sustenance (objective wellbeing), but also related to its role in human's emotional, cognitive, aesthetic, and spiritual development (subjective wellbeing)" (Newton, 2007:11). In time, conceptualisations of wellbeing which acknowledge our relationship with nature may replace existing conceptualisations.

Critics of the WBA suggest that in the current conceptualisations of wellbeing there is a tension between individual rights and the wider structural determinants of

wellbeing such as social justice and environmental considerations (Carlisle and Hanlon, 2007; Edwards and Imrie, 2008; Haworth and Hart, 2007; Newton, 2007).

The UK National Ecosystem Assessment (NEA), 2011), however, has been influential in trying to meet the environmental challenge of the 21<sup>st</sup> century by establishing action for environmental wellbeing within the wider WBA. The NEA argued that WBAs must acknowledge the importance of the environment for wellbeing. "The natural world, its biodiversity and its ecosystems are critically important to our well-being and economic prosperity, but are consistently undervalued in conventional economic analyses and decision making" (UK NEA, 2011:online).

This challenge was taken up by the Department of Environment, Food and Rural Affairs' (DEFRA) white paper 'The Natural Choice: Securing the value of nature' (2011). In this publication, environmental considerations and a connection to the natural environment were reported as key for personal wellbeing. This white paper (DEFRA, 2011) incorporates a new conceptualisation of wellbeing: one which recognises the intrinsic value of the natural environment for health and wellbeing (DEFRA, 2011). Indeed, the report identified that "People cannot flourish without the benefits and services our natural environment provides. Nature is a complex, interconnected system. A healthy, properly functioning natural environment is the foundation of sustained economic growth, prospering communities and personal wellbeing" (DEFRA, 2011:3).

The NEA research was important in promoting approaches to wellbeing, which recognised that one of the challenges of the 21<sup>st</sup> century is to incorporate the intrinsic connection between people and the natural environment into conceptualisations of wellbeing. Indeed, its research included "A complete assessment of the benefits that nature provides, how they have changed over the past, the prospects for the future and their value to our society... and was the reason for many of the actions proposed in the White Paper" (DEFRA, 2011:7).

At a global level, the South American concept of 'buen vivir' (living well or collective wellbeing) exemplifies changing conceptualisations of wellbeing. Buen vivir has emerged to challenge the centrality of the individual in current predominant

conceptualisations of wellbeing (Kothari et al., 2014; Walsh, 2010). Buen vivir has increasingly promoted ideas that wellbeing is not concerned with the individual, but rather the individual within the context of their community and their environmental and ecological habitat (Kothari et al., 2014; Walsh, 2010). These ideas, built into policy in Ecuador and Bolivia, offer a view of collective wellbeing, a useful counterbalance to the individual focus of wellbeing in the UK (Kothari et al., 2014).

Discursive shifts, which occur during the lifespan of a complex concept, are often influenced by and reflected in perceptions held by people. A local example of this is the notion of what constitutes older people's welfare/wellbeing, which has changed radically in the last century (Jones and Clapson, 2009; Royle, 2012; Walsh et al., 2000). As the population has aged, so ideas about age have changed (Jones and Clapson, 2009; Royle, 2012; Battersby, 1998). During the Victorian period, the societal perception of older people's welfare/wellbeing was proximally associated with notions of morality, sexual abstinence and civic and filial guidance (Briggs, 1970, 1983). Modern societal perceptions of older people's positive wellbeing is now more closely aligned with concepts such as active ageing, volunteering, intergenerational contact and empowerment (Gibson, 1991; Steptoe and de Oliverira, 2012). One challenge of the 21st century is how to manage the growing incidence of mental health problems among older adults, which results in negative wellbeing (Donovan and Halpern 2002; Layard, 2005).

However, "Historical analyses suggest that the health concerns and issues prevalent in a society at any one point in time are defined by powerful groups" (Research Unit in Health and Behavioural Change, 1989:18). Lord Layard, as the UK's Wellbeing Czar, has considerable power invested in his work and it is worth re-iterating that the majority of powerful wellbeing voices in the UK are those from economic backgrounds (King, 2007; Edwards and Imrie, 2008; Scott, 2012c). It may be therefore that the discursive shifts in wellbeing and determinants of wellbeing identified by 'expert voices' are not reflected in the language of citizens (Scott, 2012a; Seedhouse, 1995).

### 5.4: Summary

This chapter provided a linguistic/historical account of wellbeing in order assess the role of historical/linguistic factors in the mobilisation of wellbeing and whether historical/linguistic factors influence current conceptualisations of wellbeing.

The linguistic contextualisation provided evidence that wellbeing's linguistic status has facilitated its appropriation and mobilisation. In addition to which, wellbeing has a strong linguistic/historical legacy which until recently was relatively underused. This makes wellbeing particularly malleable for the mobilisation of civic behaviour through a diverse range of powerful discourses and rhetoric.

The historical analysis has given us a greater appreciation of the long-standing association between wellbeing and personal responsibility. Additionally, it facilitates understanding that whilst personal responsibility may be considered integral to wellbeing, certain aspects of human existence central to wellbeing require administrative intervention.

The association between wellbeing and personal responsibility is not linguistically explicit. However, wellbeing has a historical resonance which has been utilised to mobilise it within a PRA to facilitate a reduction in state responsibility for wellbeing. The close historical and linguistic association between health and welfare have also contributed to this. Wellbeing retains its connections with health and welfare due in part to the historical and linguistic understandings we are subconsciously aware of. Wellbeing acts as a conceptual bridge between health and welfare, enabling and facilitating its incorporation into modern governance, particularly in relation to health and welfare agendas.

This chapter also assessed whether historical/linguistic factors influence current conceptualisations of wellbeing. The linguistic synthesis explored the ontology and semantic construction of wellbeing and the etymological roots of health, welfare, happiness and wellbeing. This synthesised understandings about associations between wellbeing and its closest linguistic/conceptual terms. The synthesis

suggests the linguistic roots of wellbeing and its closest linguistic/conceptual terms continue to influence modern conceptualisations of wellbeing and its usage.

The historical analysis documented the transition from early conceptualisations of wellbeing through to current conceptualisations. This contextualisation documents the conceptual fluidity and versatility of wellbeing, which facilitates its appropriation and mobilisation across divergent agendas.

However, it is important to note that "We must not lose sight of the fact that the inferences we draw from historical/linguistics are necessarily hypotheses" (Epps, 2014:592). Furthermore, it has not been my intention to present the linguistic/historical development of wellbeing as an apparently linear progression. Rather, it is to be hoped that the consistent application of procedures employed to authenticate the linguistic contextualisation and historical analysis offer a well-balanced interpretation and synthesis of the historical/linguistic roots of wellbeing and its conceptual development.

The following chapter documents the findings from the empirical study, which investigated conceptualisations of wellbeing amongst older adults with LTCs.

### Chapter 6: Findings: Experiential Conceptualisations of Personal Wellbeing

The order in which the themes are addressed in the following sections reflects the deductive coding process which was used in the semi-structured interview schedule. The findings from the literature review indicated that certain contextual experiential aspects of wellbeing act as important determinants for personal wellbeing. Ageing, health, markers of personal identity and social interaction were all widely recognised in the literature as key for personal wellbeing and were included in the semi-structured interview schedule. For this reason, these four themes are discussed first. The three remaining themes of the natural environment, the built environment and temporal states were also identified in the literature review as important for personal wellbeing, but emerged in more detail from the inductive coding process of the semi-structured interviews.

The seven key themes explored in the following chapter relate to the ageing process, health, personal identity, social interaction, the natural environment, the built environment and temporality. These themes were recounted through participants' narratives of their lives conceptualised in relation to personal wellbeing. Their narratives will help to expand our understandings of how personal circumstances and structural factors interact with and shape what is important to participants' conceptualisations of wellbeing.

### 6.1: The ageing process

Half of the participants (n=12), thought what was important for their wellbeing changed as they grew older. This led many to report that the ageing process played an important but not necessarily negative role in their personal wellbeing.

A - As you get older I think what's important to your wellbeing changes, wellbeing is different when you were younger than as a young married couple. When you get over 50 I would say your health is a bigger worry

and finances too because kids always come to the bank of mum and dad. For me now I'm older health and feeling financially ok are probably the most fundamental for my wellbeing. (Jim)

There were some aspects of the ageing process, which participants identified as having a positive impact on wellbeing.

A - By 50 I'd learnt what my strengths and weaknesses were and that's given me more confidence and [pause] it allows me to do stuff I'm good at so I feel like I'm being useful and helpful for my family. (Ivy)

Alison's account of her role as a mother and grandmother was a rich and consistent vein running through her past, current and projected future sense of wellbeing. Her recollections of her past performance as a mother often impinged upon her assessment of her current mothering. This was usually referred to negatively and with a sense of regret and sorrow. However, some aspects of the ageing process had a beneficial impact on her wellbeing. Alison's experience of the menopause meant she no longer felt pressurised to portray herself as sexually attractive to the opposite sex. Concomitantly, she also perceived a decline in societal pressure to portray herself as sexually attractive. Freed from spending time making herself sexually attractive left more time for familial interaction. These physiological, psychological and psychosocial aspects of the ageing process enhanced Alison's maternal confidence. Alison's shift in focus away from the pursuit of sexual validation to her role as mother and grandmother enhanced her wellbeing.

A - I don't feel driven by my urges as I did when I was younger, that urge to be sexually driven, stay slim, look good and be attractive to men....That social mixing has also gradually waned and now my home means more to me...Being able to help my daughter write a thank you letter showed me how important it is for me that I can help my kids...It makes me very proud that they come to me, that's my reward and so it makes me think that whilst I may not have been a great mum early on I've come up with the goods now. I'm giving back and I've grounded their personal development and without that they would not have the lives they have now. (Alison)

# **6.1.1: Premature ageing**

For some the perceived decline in personal wellbeing was attributed to age and/or because of LTCs. This was considered indicative of the ageing process and regarded as a normal or inevitable part of the ageing process. Those participants with conditions which they identified as being part of the natural ageing process, for example arthritis, were not inclined to perceive themselves as having an LTC (n=6). Others, particularly those diagnosed with sudden onset or degenerative conditions such as traumatic or acquired brain injuries (TABI), Multiple Sclerosis (MS) and Parkinson's Disease (PD) felt that their condition led to premature ageing (n=8).

Premature ageing led to a sense of marginalisation and loss of identity through mechanisms such as employment and social activities. These had a directly attributable and detrimental impact on their personal wellbeing, more specifically their emotional, psychological and social wellbeing. Davindra had suffered a TABI and lived in warden-assisted accommodation. This life-changing event had resulted in the loss of his employment and home, both of which affected his sense of self, independence and autonomy. Davindra recounted that his sense of premature ageing and moving to warden-assisted accommodation had prohibited him from managing his own finances, thereby undermining his self-respect.

A - It's my money but I'm not allowed to spend it as I wish. I'm not a child and it's my money so I should be able to manage my own affairs for my own self-respect...I'm not the same man as I was before and I feel like I've aged ten years at least... I lost my job, my home; my partner as a result of this [taps his head]. I have to live here, no one comes to visit me and I feel lonely and depressed. (Davindra)

Supporting mechanisms such as access to rehabilitation, aids, adaptations, and employment support advice were sometimes cited as having helped participants adjust to the premature ageing process brought on by their LTCs.

Raj had experienced two strokes, which severely impaired his speech. Raj recounted that access to rehabilitation had facilitated improved speech and enhanced his psychological and emotional wellbeing.

A - Doctors did a marvellous job for my mental health. Speech therapy was very good too. I'm a speaker so I need to talk and I couldn't, but through speech therapy I've got my confidence back and now I volunteer at the Sikh centre in Moss Side. (Raj)

# **6.1.2: Intimacy**

Participants consistently identified that companionship, intimacy and sexual intimacy were important but understated components of wellbeing. These were recognised as one aspect of the ageing process which had an important influence on perceptions of self and perceptions of self through societal norms. Gendered differences were apparent in these findings. For some participants the loss of a partner or intimate relationship was associated with the ageing process and reduced opportunities to meet new people. Many participants were able to talk freely about a changing association between their personal, emotional and psychological wellbeing and companionship (n=14), but struggled to articulate their desire for intimacy and sexual intimacy (n=9). This was particularly the case for the male participants. The majority of participants (n=6) who implied or expressed a desire for sexual intimacy were men (n=4). Three female participants reported the ageing process had reduced their desire for intimacy and sexual intimacy.

The sexual self-image of the female participants appeared to be primarily portrayed through the eyes of non-filial others such as neighbours and friends or through societal attitudes. Some recognised this and reflected how this aspect of the ageing process constrained their options for companionship and encroached on their sense of personal and social wellbeing.

A - Being single definitely has had an impact on my wellbeing...being on your own is just more scary [pause] and you're looked at like a second class citizen. You realise that you're going to have make future friends with people who are single... because people see you as a threat. (Alison)

Male participants who were in relationships felt that desire for intimacy and sexual intimacy was an important facet of their personal, emotional and psychological wellbeing, but they struggled to articulate its impact.

A - Since my marriage broke down...I've been living with a woman that I can be close with again. (Tim)

Q – By 'close' Tim, do you mean sexually intimate?

A - Yeah, that side of things hadn't been there for a long time, my wife no longer saw me in that way and it was hard for me, I felt unwanted and didn't feel comfortable in my own home anymore. The woman I'm with now [long pause] it's hard to explain [long pause] but I feel happier in myself again. (Tim)

Sexual intimacy was a delicate matter to discuss in interviews and forms part of my reflections (see Appendix I).

# Excerpt from Researcher's Reflexive Journal

I have reflected at length on the difficulties I faced in broaching aspects of wellbeing concerned with sexual self-image, sexual identity and sexual intimacy, particularly with Jim. My relationship with Jim is more formal than my relationship with Alison and starting a conversation about sexual dysfunction and sexual intimacy has perhaps been made particularly difficult because of my original research project. During this project, I spoke with younger men about how their condition affected their opportunity to engage in sexual intimacy. Whilst I found it easier to broach this with the younger male participants than the older male participants, I still felt that this was an area which required a cautious and sensitive approach. I am aware that men with neurological conditions such as PD and MS are likely to suffer from reduced sexual intimacy and that this has implications for their wellbeing and their identity as males, but I struggle to feel comfortable and confident in raising these issues. I have pondered how I might start these conversations with participants, particularly with my co-researcher Jim.

Would he prefer me to ask him face to face, via email, not at all? Would he be happy to talk about these things openly? Would a conversation initiated by me about sexual self-image, sexual intimacy and sexual identity alter his perception of himself or alter his perception of how I 'see' him?

The male participants who were not in relationships lived predominantly in assisted accommodation. Three of these participants believed their LTCs had resulted in premature ageing, which had had a negative impact on their confidence to form relationships, particularly sexual relationships. They appeared reluctant to consider the possibility of forming an intimate relationship and experiencing sexual intimacy again.

A - I do miss female company but I've got no chance of meeting anyone. (Simon)

### Q - Does that affect your wellbeing?

A - I suppose so yeah, [pause] I try not to think about it but I do catch myself day dreaming sometimes about when I was normal and could have relationships with women [sigh]. That's not gonna happen though so I try not to let myself dwell on it. (Simon)

Simon appeared to employ a coping mechanism, which consisted of refusing to dwell on his past. Associating and confining intimacy with his past helped Simon adapt to his current situation.

Life changing events and life transitions emerged as sub themes within the ageing process. These had a significant impact on older people's sense of personal and emotional wellbeing. Adaptability was perceived as a key faculty helping to anchor wellbeing.

A - My wellbeing is negatively affected by feeling anxious about paying the mortgage and financial worries. My divorce hit me hard too and it took me a long time to adapt, but once I accepted it I felt able to start thinking about enjoying life again, that I could be happy and satisfied with my life. (Tim)

### 6.1.3: Transitioning

LTCs, which threatened to foreshorten life, often led participants to reflect upon premature ageing and the prospect of death. Whilst only a minority of participants (n=5) had conditions which were likely to shorten their lifespan, this was perceived

as a real danger to both current and future levels of wellbeing. These participants appeared to lack the resources to cope with this (Lupton, 1996). Jim had been diagnosed with PD for over twenty years and frequently questioned the probability of his continued existence. This led to his apparent pre-occupation with death from complications. To some extent, this had changed his perspective on the ageing process and what constituted a life well lived.

The following extract reveals that for Jim, making a successful transition through the ageing process was by no means assured. His temporal perspective (i.e. a tendency to reflectively project into the future) undermined his sense of present wellbeing. The findings suggest that temporal perspectives, i.e. "Our attitudes about the past, present and future" (Durayappah, 2010:6), may play an important role in wellbeing behaviour.

A - I think about living as long as possible...at times this year I thought my time had come...I can get morbid about it...and sometimes I dwell on it and get really down thinking I won't see my grandkids grow up. My condition is advancing only slowly but I know that it's the complications from other things that might see me off and I worry about that. (Jim)

The notion of filial support dominated older people's expectations of what could be expected from the ageing process and transitioning into older age. This appeared to be proximally related to ideas of filial responsibility. For many participants (n=9), the notion of relying on their children dominated expectations of making the transition into the next life course stage and this frequently undermined their current and projected sense of future wellbeing. The participants who were still relatively independent felt challenged by envisaging themselves as reliant on their children. This conflicted with their perception of themselves as head of the family, but many appreciated that the ageing process led to changing family dynamics.

A - As I get older I come to rely on my kids more and more but that's a natural part of things isn't it, I did it with my mum and dad and I'm watching it start to happen now with my daughter and her kids. (Andrea)

### 6.1.4: Summary

This study identified new areas of interests in relation to the ageing process, personal wellbeing and associations with notions of temporality. Temporal attitudes, perspectives and salience\* appeared to act as important filters through which participants experienced the ageing process. Temporality, and in particular temporal perspectives, appeared to act as a conduit through which adaptation to the ageing process and its impact on wellbeing was regulated.

The findings from this research indicated that internalised characteristics such as self-esteem and self-efficacy were important for dealing with the ageing process, LTCs and wellbeing. However, participants also identified structural supporting mechanisms as being equally important. Structural supporting mechanisms, provided through professional services which were easy to access, often facilitated those adaptive behaviours, which acted as ballast for personal wellbeing. As previously documented, life changing events such as the diagnosis of LTCs or marital breakdown substantially affected participants' ability to accommodate the ageing process. These aspects of the ageing process had significant impacts on psychological and emotional wellbeing. The findings from this study indicate that successful ageing and maintenance of personal wellbeing is often underpinned by attitudes towards the adoption of adaptive behaviours.

The findings also indicate that what is important to older people's personal wellbeing changed throughout the ageing process. These changes did not necessarily lead to negative wellbeing. Some participants perceived changes as part of the natural ageing process in what is often referred to as 'The Paradox of Ageing'. Other participants reported that the ageing process increased self-awareness, facilitating an improved sense of psychological and filial wellbeing.

Changing perspectives on companionship and sexual intimacy associated with the ageing process formed a key aspect of some participants' social and emotional wellbeing, particularly for single men and females under 60. As previously documented, gendered differences were apparent. Age also appeared to influence the association between the ageing process, wellbeing and intimate relationships.

Sexual intimacy was more likely to be intimated as important for male participants and younger females. Conversely, females at the older end of the age spectrum appeared more likely to express a reduced desire for additional companionship and sexual intimacy.

# 6.2: Health

It is important to note that participants were recruited to this study because they had LTCs. Health, therefore was used as a structured prompt to explore the potential intersections between health and personal wellbeing.

Feelings of wellbeing, which changed over time and in relation to health, were often dependent upon a number of factors. These included time of onset, causation, disease trajectory, chronic pain\* and manifestation of physical symptoms. Seven participants reported that these factors could often leave a lasting imprint on wellbeing, particularly social, physical and psychological wellbeing. Five participants reported that the chronic pain brought on by LTCs led to feelings of long-term negative physical, social and psychological wellbeing. Fluctuating symptoms throughout the lifetime of a condition (e.g. during times of deterioration) followed by periods of improved wellbeing as conditions stabilised led to feelings of insecurity, which undermined short and long-term wellbeing.

## 6.2.1: Adaptation

As time spent living with a condition progressed, the majority (n=15) appeared to adapt to the changes which their health needs made to relationships, aspects of daily living, financial circumstances, functioning and sense of personal identity. However, for many, feelings of a reduced sense of personal wellbeing were particularly prominent at onset and/or diagnosis and during periods of condition deterioration. These were often followed by periods of an improved sense of personal wellbeing as people's lives stabilised. This is consistent with 'Set Point Theory' (Lykken and Tellegen, 1996) and 'Adaptation Level Theory' (Brickman and Campbell, 1971,

Brickman et al., 1978). These propose that people tend to adapt and accept their condition over time and revert to previous 'set levels' of wellbeing.

Jim had been diagnosed with PD for twenty years and had learnt to adapt over time to the limitations this placed on his physical and psychological wellbeing.

Q - Is health important to your sense of personal wellbeing Jim?

A - Well, it depends on the individual, the degree to which it affects their wellbeing, but once you have a health problem you start to think about health more and worry about other areas of your health such as your mental health and health issues which may arise as a result of your underlying health problem. Having a positive outlook on your future helps. (Jim)

Q - Did being diagnosed with PD impact on your personal wellbeing?

A - You accept what you've got and you get on with it or you don't accept it. I quickly came to the conclusion I'd got it and so I would work around it and I've done that fairly successfully [laughs] those people who remain in denial struggle mentally. (Jim)

However, despite adapting Jim was fearful that PD would result in his premature death. LTCs, which could foreshorten life, played an important part in the everyday prospects of wellbeing. Individuals felt constrained to envisage and accommodate the real and immediate prospect of death and struggled to construct the psychosocial resources which were needed to cope with this (Lupton, 1995).

LTCs which foreshorten life lead individuals to envisage and accommodate the real and immediate prospect of death (Gough, 2005; Hobbs and Sixsmith, 2010; Kuh et al., 2014). The findings suggest that individuals with life-limiting conditions often struggle to construct the psychosocial coping mechanisms to deal with this. Studies such as Hobbs and Sixsmith (2010), Marsh et al. (2011) and Taylor and Stanton (2007) have also observed this. Interpretation of the data indicates the failure to construct psychosocial coping mechanisms may play an important role in undermining short and long-term emotional and psychological wellbeing.

Some participants adopted adaptive behaviours associated with making a successful transition to older age. These included allowing family members to take a greater role in aspects of daily living such as allowing children to cook and clean. Female participants appeared to trade filial assistance by providing emotional support for the wellbeing of their children and grandchildren. Thus, personal supporting mechanisms appeared to enable the adoption of adaptive behaviours.

Structural supporting mechanisms such as access to rehabilitation, aids and adaptations appeared important for facilitating adaptive behaviours.

Interpretation of findings suggests access to supportive services underpinned adaptive behaviours such as having a positive mental outlook. Positive attitudes towards the adoption of adaptive behaviours appeared to minimise the effect that life-limiting conditions had on physical and psychological wellbeing.

A sizeable minority (n=7), however, described experiencing a loss of identity which negated the adoption of adaptive behaviours. This was particularly the case for those who had been diagnosed with an adult onset condition such as MS or lived with impairments resulting from a TABI. The occurrence of an adult onset condition/impairment appeared to be a key underlying factor for those who were maladjusted to ill health. Of the seven participants who did not recount adaptive behaviours, three had been diagnosed with MS and three lived with impairments resulting from a TABI. Two of the three with MS had been diagnosed with clinical depression, whilst all three living with TABI had been diagnosed with clinical depression. Hobbs and Sixsmith (2010) identified an apparent relationship between clinical depression and adaptive behaviours which has an important impact on social and psychological wellbeing. This suggests depression and adult onset conditions exert a substantial and detrimental impact on one's ability to draw upon adaptive behaviours.

## 6.2.2: Coping mechanisms

The findings indicated that coping mechanisms were used extensively to offset the impact of health on personal wellbeing. Research suggests coping mechanisms are an important way for older adults to offset the impact of ill health on personal wellbeing (Abbot et al., 2008; Cameron and Stuart, 2011; Keogh and Herdenfeldt, 2002; Hobbs and Sixsmith, 2010; Lazarus, 1991). The most frequently reported coping mechanism in this study was refusing to 'dwell on one's past' (n=7). Other common coping mechanisms included using religious faith and spirituality such as attending religious services (n=5), seeking support from fellow believers and spiritual leaders (n=3), using meditation and relaxation techniques (n=3) and undertaking physical activity (n=3).

The existence of chronic pain affected the participants' ability to manage day-to-day tasks, often exacerbating feelings of loneliness and isolation by severely reducing opportunities to engage in social activities. This impacted on physical, psychological and emotional wellbeing and overall quality of life. Three participants described using positive coping mechanisms for pain management. These included relaxation, stretching and deep breathing techniques, physical exercise and meditation. These underpinned adaptive behaviours such as reducing alcohol consumption, undertaking non-weight bearing exercise and adopting healthier diets.

A - I've got to say I have noticed that I can do more since I've cut down on my drinking, started taking some exercise and am eating more healthily. That's had a knock on effect of feeling bit more positive with life. I don't dwell on the past so much now I'm getting out and about a bit more and I don't feel as lonely as I did. (Raj)

However, the participants identified the need to be able to access additional professional resources to alleviate chronic pain such as a pain management clinics, physiotherapy and complementary therapies.

A - I'm in constant pain with my arthritis and I take that many painkillers it can't be good for me. Since coming to the centre I've learnt to use some relaxation techniques which have been quite helpful and have helped me cope a bit better with all the day-to-day stuff that needs doing. (May)

The adoption of positive coping mechanisms played an important part in the extent to which participants felt able to maintain their personal wellbeing in spite of declining health. Coping mechanisms were perceived as important in helping individuals accept reduced functioning and modifications required in their lifestyles. The findings suggest that a strong sense of self-image and personal identity were important supporting coping mechanisms for dealing with LTCs in relation to personal wellbeing.

Some participants (n=6) reported that changes associated with LTCs diminished their self-image. This was most apparent in those who had LTCs which were not apparent to others but led to impaired cognitive functioning and those whose conditions had result in impaired communication functioning. These participants frequently withdrew from social contact and social activities. In these cases negative coping mechanisms such as avoidance, prescribed drug taking and excessive alcohol consumption were employed to countermand the cumulative effects of their health.

Research suggests negative coping mechanisms may be particularly detrimental to psychological and social wellbeing (Gale et al., 2014; Hobbs and Sixsmith, 2010; Kuh et al., 2014). The most frequently cited in this study were excessive alcohol intake (n=5), ignoring negative emotions (n=3), prolonged use of licensed drugs (n=3), problem avoidance (n=3) and drug taking (n=2).

### 6.2.3: Social sequelae of health

Being unable to clearly articulate and thus communicate with others had a considerable negative impact on participants' social and psychological wellbeing. Those with TABI and those with affected speech patterns most frequently conveyed this. For these participants communication issues undermined wellbeing and underpinned their disinclination to socialise in wider society.

Q - Do you find that people struggle to understand what you say?

A - Yes and it's a common problem for people with head injuries. That's one of the hardest things to adapt to, being able to make people understand you, it's so frustrating and depressing. (Tim)

The social sequelae of health conditions appeared particularly detrimental to social and psychological wellbeing when they undermined characteristics which formed part of participants' sense of identity.

Q - Has your health condition had any indirect impact on your personal wellbeing?

A - [Sigh] Yes I became less confident, I lost my independence, my self-respect, all of a sudden there was nothing to get up for and I couldn't do much for myself. I've worked hard all my life and I felt like I'd lost part of my identity. (Ivy)

A - I had stroke twice, I've also been diagnosed now with diabetes and arthritis and I got help and support but went into deep depression, doctors did marvellous job though for my mental health. Speech therapy was very good too. I'd lost my confidence for speech because people couldn't understand me. I was shamed and scared to speak to people and I'm a people person so it was like I'd lost touch with myself. (Raj)

There were a wide range of factors which affected the extent to which becoming ill became a central component of self-identity and impacted on wellbeing. These included access to services, the extent to which participants experienced physical pain and disruption to their daily lives, and the degree to which their ill health was outwardly visible to others. The extent to which participants experienced pain, sleep loss and interference with activities of daily living because of their LTCs varied considerably. These factors were recognised as important in terms of the degree to which having LTCs affected wellbeing.

### 6.2.4: Structural issues

Structural issues in relation to government health policies and health and social care practices were identified as impacting negatively on personal wellbeing (n=6).

A - I definitely think this whole government talk around empowerment, choice, responsibility and wellbeing is part of a wider scheme to try and persuade, fool even, people into thinking they should be responsible for improving their health. By making us feel responsible it lets them off the hook so, they don't have to stump up for the services which will help people feel as healthy as possible. I'm a community activist and read the Guardian [newspaper] so I can see through it but if you just watched the BBC news or listened to Radio 2 they might have you believing that the state is no longer responsible for people's health. To be honest it makes my blood boil. (Alison)

There was some variation in people's levels of satisfaction and concerns regarding structural issues, such as the availability of health and social care services. The participants at times drew attention to structural factors beyond their control, which had important influences on personal wellbeing. These included referral and access to services. The majority of participants (n=13) reported dissatisfaction with access to services and service provision.

Q - How could services better reflect your wellbeing needs Brenda?

A - Since my stroke I received very little physio to help me improve and that really gets me down. I also have arthritis, angina and I'm going in for a cataracts operation on Monday. I think that services should be there to support people's needs, if I'd had more access to physio I would be more mobile than I am now and that has a big impact on my wellbeing. (Brenda)

Dissatisfaction with service was proximally related to the divergence between participants' perceived requirements and what they actually received. The participants recounted the inflexibility of services and commonly spoke about the need to 'fight' to obtain access to resources. Some participants did not pursue claims for resources, such as aids and adaptations, as they had previously found this process unproductive, stressful and detrimental to personal wellbeing.

Half of all participants (n=12) felt that they struggled to obtain the services they required to maintain their wellbeing. This was widely considered indicative of the inflexibility of service providers. Three participants reported that much needed aids and adaptations were not provided, or that the inflexibility of providers meant that equipment was not received until they were at crisis point or when they no longer needed the equipment or had purchased it themselves. Jack, who lived in residential

care accommodation, highlighted what he perceived to be a culture of inflexibility within service provision.

A – A perfect example of how inflexible providers are is the way they manage care arrangements. Instead of coming when you need them you have to block book visits. So you end up in bed by 7.30/8 o'clock and this is common practice. I've changed my care provider a couple of times and I think I've finally got one that seem prepared to be flexible and will take into account my feelings. I'm lucky though, I used to care for my parents so I know how the system works and I'm able to push for what I want. There are too many in here who don't have the experience and wherewithal and just have to put up with things the way they are. (Jack)

Some participants had conditions, which fluctuated in severity, and this prevented them from establishing a daily routine or structure in their life. In the case of fluctuating conditions, prearranged care packages which were delivered at the same time each day often failed to provide targeted assistance when needed. Many felt that service providers should focus on providing care which was flexible and tailored to meet the requirements of individuals whose conditions might fluctuate in severity on a daily basis.

Some opined that health and social care services failed to address or promote personal wellbeing. Amongst some participants (n= 5) there was a perception that health and social care services focused on health to the exclusion of wellbeing.

A - This place used to be run by social services and was fully staffed and then they cut the staff down and brought care agencies in...The agencies don't address wellbeing, they focus on your health and managing your condition and that's about it. (Jack)

Q - Is wellbeing a meaningful concept to you Davindra?

A - Well I wasn't really aware of it till I started talking to you; although I have heard the staff use it sometimes when they talk about putting on communal things like barbeques. But I'm more interested in my health and I would say that the staff here are only really interested in making sure they look after your health; they pay lip service to the other things but that's all" (Davindra)

#### 6.2.5: Acclimatisation

Despite acclimatising to changing health, nearly one third of participants (n=7) reported that experiencing LTCs led to health occupying a more central role in their wellbeing. For many participants attention to health became more important for their wellbeing than it had previously.

Q - Has your health become more important to your sense of wellbeing?

A - I've got a 1 in 10 chance of dying from an aneurism in the next year, but after all that's happened to me in health terms this year my wellbeing is still pretty good. I think everyone is probably born with the resilience to deal with it, it just needs switching on in some people. (Jim)

A - I never used to really think about my health until I got diagnosed, then all of a sudden it began to consume me and now my QoL is dictated by how well or poorly I feel. (Andrea)

Some who had not acclimatised to their LTCs reported feelings of depression. These participants described feeling that they had received a life sentence, which undermined their daily wellbeing and their projected future sense of wellbeing. This is consistent with the 'Theories of Biographical Disruption' documented in chapter 2. Seven participants reported feelings of depression, six of whom had a clinical diagnosis of depression.

Depression appeared particularly prevalent for participants diagnosed with fluctuating adult onset condition such as MS or living with impairments resulting from TABI. Two of the three diagnosed with MS had also been diagnosed with clinical depression. Two of the three participants living with impairments resulting from a TABI had also been diagnosed with clinical depression. The remaining two living with a clinical diagnosis of depression were stroke survivors. Episodic depression was most strongly associated with a sudden decline in condition.

Others reported the diagnosis of their LTCs resulted in the onset of depression, which subsequently persisted.

Q - Does having an LTC impact on your wellbeing?

A - I can and do get really depressed, just thinking this is it! This is my life! I have nothing to live for. (Silvia)

A - Yes. I get depressed sometimes, especially when I have a bad few days and I'm in pain and I haven't slept. I'm not sure I would cope mentally without my wife; I know I wouldn't cope physically. She makes me laugh even when I feel like crying. I would be lost without her!" (Jake)

A substantial proportion of participants (n=14) reported that services failed to adequately address progressive mental health deterioration. The participants who lived in residential or warden-assisted accommodation appeared particularly inclined to draw attention to the failure of services to adequately address mental health as part of a wider commitment to wellbeing.

The actual configuration of participants' health needs and their health condition played an important part in its perceived relationship with their wellbeing. As noted in the previous section, some perceived their ill health as a sign of the natural ageing process and did not perceive themselves as 'ill'. Others, particularly those with a congenital condition, appeared more likely to perceive their condition as distinct from their health. Concurrently, they also appeared less likely to perceive it as detrimental to their personal wellbeing. This may be indicative of the fact that they had not experienced biographical disruption.

Q - Is health an important part of your wellbeing?

A - My health is good, apart from the odd cold I'm as healthy as the next person, just because I have cerebral palsy people seem to think I must be ill all the time but it's not like that. (Caroline)

A - I don't allow my health condition to be seen by myself, at least as part of my wellbeing. (Sara)

A - Aspects of me are in good health, if you're managing your health condition it will be less of a thing to your wellbeing, it's all about your outlook and managing your condition. (Audrey)

Excerpt from Researcher's Reflexive Journal

Today was really illuminating because whilst I was interviewing X she spoke about her health and condition as being quite distinct from each other. I immediately realised I have been guilty of assuming that because she has an LTC this perforce indicates that she has poor health. However, this assumption is clearly misplaced as X told me that it is a common misconception that her cerebral palsy means she has poor health; in fact, she rated herself as "healthy as the next person". This has really made me reflect upon how many other assumptions I am making are misjudged and the influence this may have on my findings.

Participants' narratives also identified psychosocial factors such as the degree to which their illness was outwardly visible to others as detrimental to their emotional, social and psychological wellbeing. This appeared related to how others perceived them. This was sometimes internalised, affecting self-perception, ability to function and undermining efforts to feel valued as a member of society. Self-perception appeared particularly important to those who had few external symptoms of illness such as a mental health condition or TABI.

A - Don't really go out now because I feel awkward when I'm out because people stare at me and think I'm drunk because I'm off balance when I walk and I slur my words. They just assume I'm drunk, they don't know the way I am is the result of an accident. It's easier to stay here where people understand the nature of my problems. (Simon)

As documented in the literature review, the social sequelae of people with LTCs has been well researched. The participants who experienced the social sequelae of their condition reported it as detrimental for their wellbeing. Jim's accounts of living with PD were often located around its social sequelae and how this affected his wellbeing. Jim, like other participants, believed that gait and balance problems combined with dysarthria made it appear to those unaware of his condition and/or unfamiliar with PD that he was drunk. This perception undermined participants' self-confidence and constrained social interaction. Research suggests that it is common practice for people who have few or no physical manifestations of illness

to feel that society does not extend them the sympathy and compassion afforded to those with observable physical symptoms (Byrne, 2000; Thornicroft et al., 2007).

## **6.2.6: Summary**

The majority of participants identified that health, and in particular fluctuating health had a considerable impact on their personal wellbeing, especially psychological, emotional and social wellbeing. These were recurring underlying themes, particularly in relation to the extent to which health affected wellbeing.

The findings also indicated that sense of identity and social and structural factors were important supporting mechanisms for dealing with LTCs in relation to wellbeing. The participants documented the need to 'fight' to obtain access to resources. The participants reported feeling disadvantaged in accessing appropriate services as the result of ageism, uncertainty regarding the availability of resources, how to access those resources and how to challenge the refusal of services. These were often mentioned proximally with failure of service providers to provide integrated health and social care services.

The findings revealed that paucity of social compassion for participants with LTCs, particularly those with few or no discernible physical symptoms, led to negative social and emotional wellbeing. For a large minority of participants this led to reduced social confidence and increased feelings of social isolation.

Adaptation, coming to terms with LTCs and the adoption of positive coping mechanisms played an important part in the extent to which participants felt able to maintain their personal wellbeing in spite of declining health. The understanding gained from this study indicated that avoidance, distraction, and disengagement techniques appeared to be associated with lower wellbeing. This largely mirrors the findings of QoL studies such as Abbot et al. (2008), Brown et al. (2000) and Lazarus (1991). Conversely, acceptance was widely identified as an important coping mechanism in efforts to offset the impact of ageing and health issues on personal wellbeing. QoL studies have also identified social support as the most effective

mechanism in helping people cope and adapt to ill health (Helgeson, 2003; Schwartz and Frohner, 2005).

Being able to adapt and/or construct psycho-social coping mechanisms may be more challenging for those with sudden onset LTCs, and/or concomitant psychological issues (Hobbs and Sixsmith, 2010; Kuh et al., 2014). This can be understood within the Theory of Biographical Disruption, which suggests that making the transition from ill health to managing one's health condition is particularly difficult for those who experience sudden onset LTCs, impairment or disability (Hobbs and Sixsmith, 2010; Hubbard et al., 2010).

The notion of health and wellbeing in connection to participants' sense of identity emerged as a recurrent theme underlying participants' narratives. Marginalisation and a loss of identity through socio-economic mechanisms such as employment were reported as undermining personal wellbeing. A more in-depth exploration of personal identity and its perceived relationship with personal wellbeing is addressed in the following section.

### 6.3: Markers of personal identities

Personal identities are understood within this study to refer to a person's selfdefinition based on personal ideals and attributes (Ashforth and Mael, 1989). Personal identities are defined by the categorisations to which individuals perceive they belong (Elsbach and Kramer, 1996).

## 6.3.1: Life aspirations

Societal, filial and personal expectations of what was considered suitable for participants to aspire to appeared to exert a considerable effect on wellbeing. Alison viewed her life-course trajectory as channelled by societal expectations to become a wife and mother in her teenage years. Alison frequently reflected upon her past and her conformity to social norms when discussing what was important

for her personal wellbeing. During these reflections she made unfavourable comparisons between her own life-course trajectory and that of her contemporaries. She reflected that these comparisons continually undermined her current and projected future sense of wellbeing.

# Excerpt from Alison's Reflective Diary

I met a friend from childhood recently they are thin on the ground as I left Withington at 16 and never went back. I am really happy to meet up with her, especially as she is such an intelligent woman who came from a family with as many problems as mine but different, she lives in Hebden Vale now. I met up with her a few months ago. It was great! I saw the life I may have had if I had been able to use the opportunity of grammar school, I think about that often, and it gets me down about my life now and what the future holds for me. I flunked that opportunity but I can see why now because I was not well I was bipolar but no-one picked it up!!!!

This was not uncommon amongst participants. Nearly one third of the female participants (n=4), all of whom had children, identified that failure to fulfil early personal life aspirations undermined wellbeing. This led to long-term regrets, which threatened to undermine projected future wellbeing and future aspirations of personal development. Four female participants reflected that their wellbeing and their children's wellbeing might have been better had they not conformed to social norms and societal expectations.

A - Sometimes I feel like my whole life has been about looking after my kids and then helping to look after their kids. Don't get me wrong, I love them all to bits but sometimes I think about how different my life and my kids might have been if I'd had waited a bit. I don't think I achieved anything but having kids! See it's different now, but when I was growing up women got married, had kids and stayed at home; that was it. (Ivy)

Social norms and values also affected the range of options participants perceived were available to them, constraining choice and guiding behaviour in a number of

areas including employment and uptake of state benefits. These had profound impacts on economic and psychological wellbeing.

A - I wanted to be a doctor when I was young, I was bright enough too but it wasn't really a viable option for working class girls back then. So I got married instead and have lived hand to mouth ever since. (Moira)

Q- By 'living hand to mouth', do you mean you have struggled financially?

A - That's exactly what I mean. It's been a daily struggle to survive at times. Had I refused to kowtow to people I could have lived a comfortable life doing a job I felt called to do. It's still a source of regret; it's made me quite bitter with life. (Moira)

#### 6.3.2: Power and control

Power and control were among the most commonly recorded narratives in relation to participants' personal identity and those conditions which undermined or reinforced a sense of positive wellbeing. The lack of power and control appeared particularly prevalent for those participants who lived in residential care and warden-assisted accommodation. Audrey lived in residential care accommodation. She was one of the youngest participants and lived with cerebral palsy. Audrey epitomised participants with congenital conditions who perceive autonomy and independence as important in their conceptualisations of wellbeing.

Q - How could services better reflect your wellbeing needs Audrey?

A - Well, things like they should have a residents' representative here at the meetings they hold. Because I've experience of the staff being heavy handed and out of order; I would be a candidate because there have been residents' meetings and they've [staff] wanted to attend. They [staff] won't let you have any power or control over the system, which is wrong, especially as they go on about empowering people and managing your condition. (Audrey)

Those participants who perceived that they had a certain degree of control and power within the power sharing structures in society felt able to negotiate and influence situations and events around them. This reinforced a sense of positive wellbeing. Jim related a story which gave insight into his perception that he was

able to influence the workings of his local council and thereby exert control and instigate action in his community. For Jim, this made an important contribution to his sense of personal wellbeing.

A - We had a problem with a burst main and no one would take responsibility for it and it was getting everyone on the street down, but as X [Jim's wife] is a local councillor we were able to ring up and get it sorted within days, being able to contribute like that gives me a sense of wellbeing. (Jim)

It is important to note an apparent collective failure on the part of participants to perceive personal responsibility for aspects of their built environment. These aspects were considered the responsibility of local government, and as such beyond the responsibility and control of residents. Interpretation of the findings suggested most participants exhibited little sense of agency concerning their environment. Chapter seven will explore in greater depth the human agency/structure dynamic, which remains widely contested within wellbeing research.

For some participants diagnosis of an LTC had led to loss of employment and a reliance on benefits. This appeared to inculcate a perception that their socio-economic position and sense of identity (i.e. being working-class and claiming welfare benefits) marked them as politically un-important or insignificant. This deficiency of control and power appeared to amplify their sense of disengagement from power sharing structures in society. Many participants (n=11) reported feelings of resignation towards their current situation and a perception that there was little or nothing they could do to improve their position in the political process. Research such as Diener and Suh (2000), Ryan and Deci (2001), Ryff (1989a,b) and Ryff et al. (2006) consider agency a key facet of psychological wellbeing. The majority of participants, however, displayed little sense of purposive agency. In much the same way as autonomy and independence, agency appeared to be superfluous in their conceptualisations of wellbeing. This was particularly so for those diagnosed with clinical depression (n=3), those with TABI or MS (n=6) and those at the older end of the age spectrum, i.e. those over 70 (n=4).

Audrey lived in residential care accommodation and recounted long-standing problems with her introductory tenancy agreement.

A - My tenancy has been threatened a few times because of my relationship with the staff. I feel powerless and threatened and it's not just me. X, Y and Z [fellow residents] have also been threatened with eviction because they refused to be bullied by staff. (Audrey)

Audrey and three other participants reported issues around tenancy insecurity and voiced concerns their tenancy agreements would be revoked. This was a source of negative wellbeing, which impacted daily on their psychological and emotional wellbeing. There was a strong perception that residents' sense of power and control was undermined by the fear of losing one's tenancy. In addition to which, it was perceived as a tool to curb or manipulate particular behaviours.

A - I worry constantly they will take my tenancy agreement away because it's not permanent. I didn't know that when I moved here, otherwise I wouldn't have moved in. The worry of it stresses me out every day and really gets me down. (Jack)

A - They are trying to control me by hanging this over my head [tenancy agreement]. Sometimes X [a member of staff] will tell me to behave or quieten down or else I will be evicted. Things like that can make or break your day when you're stuck in here all day. (Silvia)

Those participants who owned their own property (n= 6) appeared more likely to feel that they were directly involved in power sharing structures. For men this tended to relate, but was not exclusive to the exercise of power within wider society (n=3). For female homeowners exercising power and control was recounted in relation to their living space (n=3). The following extracts convey the contrasting experiences revealed in relation to exercising power and control. Homeowners reported that being free to control, adapt or change their living space was particularly important for their physical and psychological wellbeing.

A - We own our house and so I feel like we have some control over our home environment and neighbourhood. Having that control and respect makes life easier because you know you can go ahead and change things without having to get permission or wait till the council can afford to do the work like some people I know. It's important to have respect, it's part of wellbeing. (Jim)

Some participants felt that they had little recourse to power sharing within society. This appeared particularly endemic in participants who were not homeowners. Those participants who lived in council-rented, residential or warden-assisted accommodation felt constrained by the strictures which living in council accommodation placed them under.

A - Since my stroke it's been a real struggle to get around the house. I've requested a new bathroom so I can get in it with my walker but I've been waiting forever, same with the ramp and the handrails. See your age goes against you. (Delphine)

Alison frequently used emotive vocabulary such as 'appalled', 'devastated' and 'resentful' in reference to her lack of power and control. This was an ongoing source of negative wellbeing. Alison's feelings of negative wellbeing and the concomitant negative emotions this engendered gave rise to feelings of anger, frustration and antipathy towards members of her family and community. These feelings significantly affected Alison's sense of psychological, emotional and social/community wellbeing.

A - I get so angry sometimes and annoyed with my family and neighbours because they can't understand why that makes me so down, feeling powerless and depressed, but feeling powerless when all the talk is about empowering people and communities to take control of their lives makes me so angry. (Alison)

A - I often feel powerless...I want to help older and more vulnerable members of the community to be involved and be listened to and to have their opinion heard. They should be treated with the respect they deserve, the respect we all deserve. (Alison)

Some participants articulated a strong sense of community (n=6). For them access to power and influence was not solely concerned with personal power or identity, but was concerned with enabling a network of power. They considered this network to be a key driver of community wellbeing, helping communities to nurture and support each other and live according to their own principles.

A - Personally I am powerless....Without the input of professionals, I would have struggled to get what I'm entitled to but being Jewish and having that strong community support is priceless. The Jewish community is quite influential here. (Andrea)

### 6.3.3: Prioritisation and allocation of services

Disenchantment with the way local and national services were prioritised and allocated was widely articulated in participants' narratives. Geographical disparities in the way resources and services were funded within the local community were commonly perceived as reinforcing cultural, class and socio-economic divisions. Thus, participants' personal identities were illuminated by those personal ideals attributed as consistent with their class/socio-economic position. Personal identity categorisations indicate how one is distinct from or similar to other in-group members. Social identity categorisations indicate group distinction and how this compares to other groups (Brickson, 2000). Whilst participants associated themselves as recipients of older adults' services, they also identified the differences which existed within the provision of services for older adults. The following quote illustrates in-group identity and out-group perceptions.

A - It's better for people in Trafford because it's a middle class area, they're posh people and so they've got more stuff and got more facilities. (Ben)

Participants were particularly likely to highlight that the lack of services and opportunities afforded to people from poor working-class areas undermined wellbeing. The participants from affluent predominantly middle-class areas also noted that institutions perceived as middle class such as the 'Women's Institute' and 'Joan and Derby' clubs provided them with social interaction and community support, both of which were important for social and community wellbeing but which were not available in poorer working-class areas. The perceived differences in services and resources were also highlighted with regard to different areas of the city. The participants from South Manchester, even those from poorer working class areas, felt their wellbeing was enhanced by their access to services and resources not available to those in North Manchester. This was endorsed by participants in North Manchester who felt their wellbeing was undermined by having fewer resources and fewer services than people living in Central and South Manchester.

Some participants were acutely aware of what they saw as the unfair allocation of services and resources. The participants generally perceived that those of a higher socio-economic standing benefited more from service provision aimed at improving wellbeing than those in poorer areas and communities. Proximally related to allocation was the participants' perception of cuts to welfare provision, which was firmly located within the parameters of government responsibility and perceived to be beyond the influence and control of most participants.

A - The Council used to do more but I suppose times are getting hard and there is less money to go round. I think it's wrong though that the poor and the old always seem to draw the short straw. My husband and I worked all are lives and paid our dues but when it comes to our time when we need a bit of help and support it's not there. It's like we were sold this dream of being looked after 'from the cradle to the grave'! Ha, [grimaces] you've gotta be kidding, I mean so many people I know can't afford to put the heating on in winter and the winter cold payment is laughable. Decent food, especially meat is so expensive now and our pensions [puffs cheeks out] well they just about keep your head above water, I'd be snookered if it wasn't for my daughter's help. (Brenda)

### 6.3.4: Cultural and group identification

Narratives, which addressed local and national politics in relation to wellbeing, were often referred to in terms of structural and social factors beyond an individual's control. These were often specifically concerned with decisions to provide or maintain service provision for some groups in society whilst failing to provide for others. This resulted in some participants (n=5) feeling that they lacked agency, power and control within the political process. The participants in general exhibited little signs or narratives which identified a sense of agency. The participants who identified themselves as working class appeared to feel that their position in society had been subsumed by the arrival of minority groups. These groups were perceived as receiving preferential treatment and access to scant resources by those wielding political power.

A - The thing is if I was lesbian say or Muslim or a refugee the Council would make sure I got the support I need; there are services available to them and

that's what's unfair. Those in power decide where to spend the money and there is nothing I or anyone else can do about it. (Tim)

Cultural differences and group identities were identified by some participants as important factors which underpinned differentiations of what might be considered important for wellbeing.

A - I think there are so many different aspects to wellbeing that come from cultural differences or religious differences that it makes little sense in a multi-cultural society like ours to try and define wellbeing. (Raj)

A - It's impossible to define wellbeing and so I wish you good luck. Everyone has a different slant. It's probably 90% core wellbeing and the rest is individual, unique, but there is probably so much variation within that 10% that you couldn't pin it down if you wanted to. (Jim)

Jim had been involved in researching wellbeing as a representative on the 'Valuing Older People Committee' and had written a wellbeing article for their newssheet (see Appendix P). The excerpt below reveals how Jim perceived cultural and group identification as having a strong interaction with wellbeing.

### Excerpt from Jim's Reflexive Diary

I was thinking about culture and wellbeing and I have decided that I don't think wellbeing affects culture but I think culture could affect wellbeing. So in terms of culture I feel a culture connection to people who live in same region of world, have a similar skin colour, similar earning power, same religion, same traditions, same moral values and treat others as equals. I don't think it makes sense though to try and define wellbeing which is relevant to everyone although of course there are things we all want for our wellbeing regardless of our culture, such as love, health, friendship, family. In some cultures though wellbeing is about the collective good whilst in western cultures it's more about the individual and I think things like that make it tricky if you're trying to define wellbeing.

Jim regarded wellbeing as a complex and multilayered concept, which was influenced to a significant degree by historical and cultural legacies. For Jim this meant that in multi-cultural countries such as the UK there were likely to be significant socio-cultural group differences in what was considered important for wellbeing.

Andrea had been diagnosed with cancer, kidney failure and diabetes, and was in poor health. She was proud of her Jewish heritage and felt that her cultural attachments were an important source of emotional wellbeing. Intergenerational contact facilitated through the Jewish community enhanced her sense of social and community wellbeing. These in turn helped alleviate the loneliness she felt living alone. For Andrea, her ability to cope with her failing health was strengthened through her sense of community and social wellbeing.

Q - So, is a sense of community important to your personal wellbeing Andrea?

A - Yes, because we look out for each other. Now and again, I meet up with friends and family at the synagogue and it's nice to watch the next generation coming through. I have some friends who aren't Jewish and they don't have the same bonds we do. We have a different culture and we live by our own strong sense of principles. (Andrea)

# 6.3.5: Respect/disrespect

Respect and/or disrespect were important sub-themes, which emerged in participants' narratives as having an important bearing on levels of personal wellbeing.

Participants' perception of ageism, experienced in particular as the inequality of access to resources, was reported by the four participants aged over 70. Being considered old appeared to have been assimilated into these participants' personal identities. This was often narrated proximal to the disrespect which participants perceived modern society assigned to older people.

A - As an old person I feel invisible, like I don't count anymore. I used to be very outgoing and outspoken but now it's like my opinions count for nothing. (Jenny)

Alison reported that council housing employees routinely displayed a lack of respect towards her and her fellow residents. This damaged her sense of self-worth and sapped her psychological and emotional wellbeing. The following extract from Alison's reflective diary helps to contextualise how feeling disrespected undermined certain aspects of Alison's personal identity.

# Excerpt from Alison's Reflective Diary (2009)

My immediate neighbour moved out, after she had gone her smoke alarm was beeping. After the first weekend of it bleeping every 30 seconds I called housing repairs. No one came. I then reported it to the manager...Now 1 month has gone by, every night when I went to bed, usually feeling relaxed and happy I'd hear it bleep and my heart and stomach would drop into a miserable feeling, of feeling worthless, that I didn't matter in the scheme of things. Although the housing manager is ok, I sense no respect for tenants.

Many participants felt that those with power or in positions of authority must be challenged in order to obtain respect or control from them.

Q - Why don't you think the staff are respectful towards you Audrey?

A - Oh they can be when they want, when it suits them, but it's more for show. I don't think they don't have any respect for us. I've had dealings with social services for years, they're cute to your face but behind your back! They have all the control and I wouldn't tell them anything really. (Audrey)

Some participants (n=6) felt that being treated respectfully by professionals, young people, family members, friends and wider society underpinned their sense of personal identity. Concurrently, these participants reflected that it was also important to show respect and behave respectfully towards others. Mutual respect was seen as helping to reinforce social cohesion and community wellbeing.

## 6.3.6: Autonomy and independence

Autonomy and independence were also amongst the most commonly recorded narratives and were a rich and consistent theme throughout participants' narratives. For a significant number of participants, however, autonomy and independence were regarded as redundant concepts as their LTCs meant they were reliant on others to assist them with the most basic activities of daily living. The findings suggest that current policy and practice orientations towards independent and autonomous decision making may not reflect the desires of those living with LTCs. The participants, especially those living in residential or warden-assisted accommodation (n=5), indicated that the level of control they wished to exert over the decision-making process was not consistent with those promoted by policy and practice. These participants felt dependent on others and did not feel capable or sufficiently able to make decisions, especially decisions related to financial budgeting. In these cases, many felt that the professionals with whom they had a relationship were best placed to make decisions on their behalf.

Three male participants aged 60 or under, diagnosed with adult onset conditions and living in residential or warden-assisted accommodation felt that autonomy though important was difficult to operationalise. These participants felt that a sense of autonomy and agency was difficult to articulate. This was symptomatic of the restrictive nature of their condition and their dependence upon others.

A - After getting MS I lost my confidence and I would like to be more independent but... (Shawn)

Q - Have you asked your care manager if you can live more independently?

A - I wanna see how I get on here first because I do need a lot of support and I don't want to make a hash of things and have to come back here with egg on my face. (Shawn)

A significant minority of older participants (n= 6) believed that their LTCs meant they had neither the opportunity nor capability to choose, and thus autonomy was a superfluous concept for them.

A - A lot of things like autonomy and independence, which they [centre staff] talked about in relation to wellbeing, well a lot of people don't have

them so it makes no sense to consider them in relation to our wellbeing. (Delphine)

Conversely, autonomy and independence were not considered redundant concepts for those at the younger end of the age spectrum. The participants at the younger end of the age spectrum with congenital conditions which restricted their independence (n=3) and who were to varying degrees reliant on others appeared less likely to perceive autonomy and independence as redundant concepts. As documented in chapter two, adaptation and biographical disruption theories have been used to explain this phenomenon.

A - I enjoy living independently, I've got the support of the carers so I can do more for myself here but more importantly I've made friends with others who live here and it gives me a sense of control knowing that we have some choice in what to do with our time and that I've got people to socialise with. (Sara)

A - I always wanted to be able to go out and meet new people and mix more but I never had the opportunity until I moved here. I think helping people to live in their own flat but also to have others in the same boat living alongside you has been the best thing to happen recently. You talk with some of the older ones and they've spent all their lives in institutions and so they don't really know how to do things for themselves and that saddens them, it's sad to see really. (Caroline)

The findings suggest that those participants with a congenital condition had often made necessary life adjustments and were physically, mentally and emotionally adapted to managing their condition. As a result, they appeared more likely to value their own sense of self and autonomy in the decision-making process. This did not extend to overall control of all decisions, as the role of professionals in critical health-related decisions was valued. However, the majority of participants, especially those with mobility issues, wanted more control over decisions concerning activities of daily living such as shopping and eating.

#### 6.3.7: Financial considerations

The majority of participants' (n=17) identified that their sense of personal identity was intimately connected to their financial situation and their economic wellbeing. The participants often drew attention to structural factors beyond their control that affected the form and quality of family relationships. Being able to financially provide for loved ones and earning one's own money underpinned participants' self-respect, self-worth and self-esteem. Men in particular reported that a key feature of their economic wellbeing was having the freedom to manage their own financial affairs. Financial management, financial stability or financial anxieties played an important role in how they perceived themselves. The opportunity to provide financial assistance to one's descendants was consistently identified by those men with children when conceptualising and articulating what was important for their personal wellbeing.

A - Having financial security and savings means we can help the children out with things before we die. The pleasure and satisfaction I get from being able to help them is a big thing for my wellbeing. It's nice to be able to take the worry off them before you're dead. (Jim)

Male and female participants indicated that financial concerns underpinned their economic and personal wellbeing. This was often proximally related to the day-to-day pressures to provide for those material effects which form the foundations of living beyond that which constitutes living on or below the poverty line.

Q - Can you think of any examples of how services can better reflect the wellbeing needs of older adults with LTCs?

A - They could take us out more, on trips and things. I've not been on holiday since my husband died because I can't afford it so I would love to go on holiday. It wouldn't have to be abroad or anything, even just to Blackpool or Southport for a few days, it would give us something to look forward to and a chance to do things we wouldn't normally get to do. (Brenda)

For those who struggled to afford necessities, such as warm clothing, adequate heating and decent food, the resultant continuous worry, anxiety and strain undermined economic wellbeing.

A - The financial security of having a phone so I can check on X's [grandchild] welfare is paramount to my wellbeing. A lot of my wellbeing is practical, about being able to afford stuff, for example, decent warm clothes, etc, the things that have always been a problem. Car boots and charity shops help my wellbeing. I can get gifts/toys for the children, makeover the house, etc. All the things that others take for granted and buy new. I always felt bad about not being able to buy much for the kids so I feel better now it is easy to pick bits up. Definite wellbeing when I find nice winter coats for myself, daughters or grandkids for £2! (Alison)

The majority of participants believed different people had different financial requirements and what was sufficient for one might not be sufficient for another. In addition to which, the majority of participants disclosed that whilst money was important for economic wellbeing it should not necessarily be perceived as underpinning wellbeing. Argyle (2002) and Ballas and Dorling (2007) reported similar findings.

A - Youngsters nowadays think it's their right to have a television in their room, and the newest mobile phone and computers. When we were kids, we had none of those things but we were happy playing hide and seek in the park, or climbing trees or whatever. I blame the media; they've put these ideas in their heads, that and the pop stars and footballers who have ridiculous amounts of money. All kids seem to want now is fame and fortune and they can't seem to take pleasure in the small things in life like we do. (Ivy)

Q - Do you think the media and advertising influences what is considered important for wellbeing?

A - Oh, yes. Advertising definitely does impact on and influence your wellbeing. (Ivy)

A number of participants suffered financial hardship and difficulties after being diagnosed with LTCs. This was usually the result of losing their employment and becoming financially reliant on state benefits. This had a long-term damaging impact on their personal identity and an associated negative impact on their economic and personal wellbeing. The participants identified that these effects on

personal wellbeing were diverse and varied in intensity and duration. These included financial hardship, a reduced sense of pride, self-esteem, depression and social isolation.

A - I'm not the same man as I was before and I feel like I've aged ten years at least. Things I used to be able to do without thinking about I now struggle with. I've lost my job, X [partner], my boys and all my mates. I have to live here, no one comes to visit me and I feel lonely and depressed. (Davindra)

Jim lost his employment after being diagnosed with PD but received a comfortable pension, which gave him a better standard of living than previously.

A - The fact that I'm secure in financial terms makes it easier for me to deal with things that life throws at me. I'm now better off not working so it's become easier to live. I may get depressed sometimes and at times, this year, with having 2 serious operations my wellbeing has taken a bit of a dip but the fact that I'm financially secure has acted as a ballast for my wellbeing. It's definitely one of the pillars of my wellbeing. (Jim)

However, the majority of participants who had lost their employment spoke of becoming financially compromised and struggled to make ends meet. This had a considerable negative impact on their personal identity and wellbeing. The loss of income resulted in a reduced sense of purpose, self-worth and self-fulfilment. In addition to which, financial constraints reduced the opportunity to engage in activities outside the home, undermined opportunities for social contact, interaction within the wider community and was detrimental to social wellbeing.

A - I lost my job and it just really hit me, after twenty odd years of working suddenly I was stuck in the house all the time and to make it worse there were no women in the house and I missed that female companionship. I was lucky that the nurse sussed what was wrong and encouraged and supported me to get out of the house and use the community centre to meet people. (Ivy)

Adaptability to changing circumstances frequently arose in the participants' narratives and was often perceived as an important mechanism for underpinning one's identity and encouraging positive thinking. Some participants, however,

appeared to have struggled to adapt to changing circumstances throughout their lives and referred to events in their past which continued to exert a significant influence on current levels of personal wellbeing.

A - The more I think about it the more I think that there were 2 periods in my life which impact most on my wellbeing. The AA [Alcoholic Anonymous] and going to secondary school. My life changed at 11 because I went from having an authoritarian school environment to a school which lacked any of that planned, laid out structure. Nobody at school noticed that I wasn't engaged or that I wasn't able to deal with the school situation because I was bi-polar; if they had I might have been able to make the most of the opportunity I had going to grammar [school]. (Alison)

## 6.3.8: Summary

Markers of an individual's personal identity are intertwined with an individual's sense of personal wellbeing, but structural issues were judged more pertinent to an individual's sense of personal wellbeing. Power and control, and more specifically their deficiency, were amongst the most widely cited aspects of participants' personal identity identified as undermining personal wellbeing. These appeared to amplify participants' disengagement from power sharing structures and intensify a sense of alienation from the political process.

The findings suggest that lack of power, control and disengagement from power sharing structures are important for personal wellbeing. The findings indicated that these were of central importance for participants' psychological and emotional wellbeing. This was particularly the case for participants who lived in residential or warden-assisted accommodation and felt particularly disempowered.

It is important to note that feeling respected was consistently identified by participants as an important aspect of personal wellbeing. Respect, however, was rarely explicitly associated with notions of self-respect. The participants appeared to demonstrate little sense of agency in this. Respect did not appear to be contingent on self-respect, but rather was perceived as facilitating community and

cultural cohesion and intergenerational appreciation. Being treated respectfully was perceived as symbolic of valuing participants' membership of society.

Autonomy and independence were also identified as important markers of an individual's personal identity and wellbeing. For those with adult onset conditions, autonomy and independence were often felt to have been compromised through the loss of employment. This frequently resulted in financial hardship, which further undermined personal wellbeing through mechanisms of personal identity. For those with debilitating LTCs, which meant they were reliant on others, autonomy and independence, agency, choice and control were considered redundant concepts. Autonomy and independence, whilst often being perceived as redundant concepts were, nevertheless, considered important in participants' conceptualisations of wellbeing. Autonomy and independence were often perceived as being linked to the accessibility of forging meaningful social relationships; sense of belonging; place in society; freedom and capacity to be involved in decision-making; and being able to participate in society. The relationship between wellbeing, social intercourse and social isolation is explored in the next section.

### 6.4: Social interaction

For many participants, narratives relating to social interaction, supportive networks and filial affiliation were routinely articulated when formulating conceptualisations of what was important for personal wellbeing. This was often proximally related to the sense of security they promoted.

A - For me, wellbeing is knowing that my family is here for me, that they are safe and happy and well, like my granddaughter who comes and cooks for me every night. (Ivy)

Q - Why is that important for your personal wellbeing Ivy?

A - Well Jo, I would be lonely if it wasn't for my family...I live on my own and don't get out much so I feel safer knowing that she will be here in the evenings. (Ivy)

Having a network of supportive people was vital for social wellbeing as those who felt isolated tended to report poor levels of personal wellbeing. The ability to access community centres and day-care centres reduced feelings of isolation and depression and were seen as essential for social wellbeing.

Conversely, without social contact participants experienced a sense of loneliness\*, social isolation\* and increased levels of depression. The majority of participants identified that kinship and ties of affiliation were fundamentally important to their social and personal wellbeing. For many this involved playing a strong matriarchal or patriarchal role within their family, as typified by Alison.

A - Having a home that's hospitable is important that I can be hospitable in and have the wherewithal to put my family up comfortably in, to have enough bedding and beds and the space and peace for them to feel happy here. I personally would not be happy if I didn't have enough space. I know that my son can come and stay here with his wife and feel happy and not worry about his car being stolen or having to go out and face rowdy kids making noise and trouble like in my old place. That's mega important, knowing I have somewhere my kids and grandkids can come and stay because it means you can give back to your family and it's not always them doing the giving. This is good for family dynamics and relationships and the wellbeing of myself and my family. (Alison)

Q - How important is your family for your wellbeing Jenny?

A - Oh very important, some of them in here never get to see their grandkids, not like in our day when you saw your gran and granddad all the time. Young people and old people don't mix much now which is a shame really. I'm not well off but I love being able to get my grandkids little gifts and see their faces light up. (Jenny)

Male and female participants both commonly reported a deep desire to engage in greater intergenerational inter-action with young people. This was commonly expressed through narratives, which revealed that their social wellbeing was undermined by what they perceived as a generation gap within social networking systems. Thirteen participants reported negligible contact with young people. The paucity of intergenerational inter-action was deemed detrimental to social and personal wellbeing. Furthermore, this generation gap was prominent amongst families and wider society. The participants talked frequently about modern family

life, which meant that older and younger relatives rarely socialised together or lived together in extended families. Half of all participants (n=11) felt they were less actively involved in helping to shape their familial and community life, and existed on the periphery of family and community life. This had a detrimental impact on personal, social and community wellbeing.

# 6.4.1: Filial legacy

Male and female participants both commonly reported a deep aspiration to bequeath a legacy for their descendants. For many of the participants, contributing to a filial legacy was key for personal wellbeing and appeared to impart a strong sense of social wellbeing. Although people had different ideas about what that legacy might be, their commonality was that they experienced a deep sense of wellbeing from their motivation and efforts to provide a legacy for their families. Female participants were more likely than male participants to talk about personal wellbeing in relation to filial connections. For example, personal wellbeing was frequently reported as negatively affected by concern for children and grandchildren. It was positively affected by enhanced social wellbeing through female companionship and greater interaction with family members. For female participants, this tended to be expressed in terms relating to their role in nurturing their children and grandchildren in order to facilitate their emotional growth and personal development.

Q - Can you explain Jenny why 'feeling like you're still part of things' is important for your personal wellbeing?

A - [Sigh] Erm [pause] no one wants to think they are a spare part Jo. It's good being able to read to my grandkids and sometimes I can even help them figure out things that are worrying them. Knowing that I'm being of some use and that I am contributing to their values and how they behave adds greatly to my wellbeing. (Jenny)

For the male interviewees, filial legacy tended to be expressed in terms relating to their role as financial providers.

A - My family is the most important for me. I suppose I grew up thinking a man's role is to look after his family and I've always tried to be a good husband and father. Losing my job undermined that because I was no longer the bread winner, suppose it hurts my pride that I can't help the kids as much financially as I used to do. (Tim)

Jim's patriarchal role within the family was a rich and consistent vein running through his narrative and entwined in his current and projected future sense of wellbeing (this will be reported in greater detail in the 'Temporality' section).

When reflecting upon his family, he reported that being perceived as a good family man by his peers gave him a keen sense of personal wellbeing.

Analysis suggests the notion of filial legacy dominated participants' expectations of the transition to their next life-course stage. This appeared to be proximally related to preparing for a time when they may become physically reliant on family members. Thus, family support mechanisms become fundamental for ongoing wellbeing and the wellbeing of close family members. Undertaking a matriarchal or patriarchal role acted as a significant contributor to feelings of positive personal wellbeing for many. However, at times it was also perceived as having circumscribed personal development. There was a complex interaction between filial proximity and wellbeing, with family support contributing to both negative and positive wellbeing. It was clear that for many participants the family was their primary source of support and care. It was also apparent that this support was not always freely given or could be taken for granted. This could lead to a sense of apprehension and negative wellbeing.

A - Living with family may be good in that you're looked after, but may be bad in that you lose your financial independence. Your family may come and bring you food but if it's not to your taste and you fall out because of it then what looks like something which should be improving your wellbeing is actually having negative effect on it. (Jake)

## 6.4.2: Participation in society

Participation in society appeared to be proximally related to social isolation, loneliness and depression. The participants identified the importance of addressing social isolation and exclusion. They felt that services needed to be more oriented to the social wellbeing of participants. Isolation and loneliness were seen as two key contributing factors to negative social wellbeing amongst people with LTCs. Participant narratives failed to indicate a sense of personal responsibility for tackling social isolation, depression and loneliness. Indeed, there was broad vocalised consensus that it was a duty and responsibility of statutory services to address these through the facilitation of access and opportunity for social interaction and integration. This was most frequently verbalised by those who lived in residential care settings.

A - I get lonely at night when everyone's in their own rooms. Being stuck inside all the time looking at the same four walls is depressing, especially when you're dependent on others because of your health problems. Social services and the agencies in here are responsible for us, it's down to them to help people have a life, we used to have barbeques and things and it felt like a proper community...Coming from Scotland where community is strong, you have a sense of belonging and being with people helps pass the time and gives you something to look forward to, a reason to get up. (Silvia)

Participants frequently reported that services failed to address social isolation by facilitating participation and engagement in informal and formal social activities. The participants reported increased levels of depression and loneliness from reduced contact with others. They argued that services could and should be held accountable and responsible for those determinants of personal wellbeing such as participation, which impacted on isolation, depression and loneliness.

Social wellbeing appeared to exert a large effect on personal wellbeing, mirroring the findings of Barnes et al. (2013) and Ward et al. (2012). Interpretation of the data suggests that there were gender differences regarding which aspects of social wellbeing interacted most significantly with personal wellbeing.

#### 6.4.3: Relationships

Females participants (n=8) reported that friendship, female companionship, a sense of contributing to filial legacy and greater interaction with family members were the primary sources of their social wellbeing. For female participants, social wellbeing related primarily to their subjective experience of connections with close and significant others, and the strength of those relationships. Recent research has identified that the number of real-life friends compared to online friends is positively correlated with subjective well-being (SWB) after controlling for income, demographic variables and personality (Helliwell, 2013). In addition to which, Helliwell (2013) reported that doubling the number of real-life friends has a comparable effect on wellbeing as a 50% rise in income.

Research has also demonstrated that nourishing relationships are a key component of social wellbeing, supporting self-worth, resilience, psychological wellbeing and mitigating the effects of stress (Barnes et al., 2013; Taylor, 2011; Ward et al., 2012). However, relationships within wider social circles were also important. Engaging in conversation, whether with close friends, acquaintances or strangers was particularly important for the social and personal wellbeing of females. Conversational exchanges in addition to strengthening bonds were valued for facilitating information and knowledge exchange. Information and knowledge exchange contributed to the participants' ability to access resources, which supported personal wellbeing and facilitated a sense of interdependency. Barnes et al. (2013) and Ward et al. (2012) explored wellbeing in older people and reported that older people perceived interdependency to be more congruent with their wellbeing than independence.

#### 6.4.4: Social activities

Social activities were seen as vitally important to the social and psychological wellbeing of participants. However, a reduction in both residential and community centre staffing levels had reduced informal opportunities for socialising. This, in turn, undermined participants' access to socially supportive networks. The

participants commented on the need for social trips and activities, which facilitated social interaction and satisfied desires to be part of a normal social world. This helped participants develop a sense of self in relation to others and to feel part of a wider community. This was particularly important at night and during the winter months when people felt particularly lonely. The majority of participants argued that this would make an overwhelming difference to personal wellbeing.

A - I get so bored and lonely, especially at night. Services should help with things like that, I mean it wouldn't cost them much and they know how depressed people can get being stuck in the house all day and when you have health problems it can become a vicious circle. They have a responsibility to help people get out and about and feel like they still have a place in their community. (Jenny).

Shared social activities appeared particularly important for male participants. Seven male participants recounted that shared social activities such as playing snooker/pool, fishing and watching sports matched greatly enhanced personal wellbeing. This was an important aspect of their social wellbeing. Conversely, three males with cognitive impairments all reported feeling little or no inclination to engage in shared social activities. Concomitantly, they exhibited little interest in social interaction. One possible explanation for this apparent disassociation from social interaction is that these participants had become disengaged from society. Disengagement theory has been used to explore the relationship between social activity and wellbeing, as research indicates these two variables are positively related (Di Tella and MacCulloch, 2005; Havighurst et al., 1968; Larson, 1978;).

An alternative explanation for this finding is offered in studies such as Baker et al. (2005), Biddle and Ekkekakis (2005), and DeNeve and Cooper (1998), which suggest that the lower levels of wellbeing reported in people with reduced social interaction and social wellbeing may be associated with personality characteristics.

Shared social activities appeared to have a smaller influence on personal wellbeing for females. Only three female participants identified this aspect of social wellbeing as important for their wider sense of personal wellbeing. Alison identified the

importance of shared social activities in respect to facilitating community empowerment and cohesion. The other two participants reported shared social activities in respect of undertaking religious duties. For these women, feeling socially engaged and having a sense of social productivity appeared to exert a significant and positive impact on personal wellbeing. Keyes (1998, 2000) and Myers (1999) reported similar findings.

Some commonality existed between male and female participants with respect to social interaction. Male and female participants identified that a lack of social interaction manifested in feelings of isolation and loneliness, which contributed to negative social wellbeing and undermined personal wellbeing.

#### 6.4.5: Isolation and loneliness

Isolation appeared particularly problematic for participants with TABI who opined that their former established relationships had been irretrievably broken since their injury. In terms of impact on personal wellbeing, this appeared to lead to withdrawn or reclusive individuals who struggled to articulate their wellbeing needs.

A - I haven't seen my family since my accident. I don't think they know how to handle my disability and depression. I don't really go out either because people stare at me and think I'm drunk because I'm off balance when I walk and I slur my words. It's easier to stay here where people understand the nature of your problem and you don't have to explain yourself. (Simon)

Nine participants (the majority of whom were men, n=7) expressed acceptance about their lack of opportunity for social interaction. These nine participants all lived with LTCs resultant from an adult onset condition such as a TABI or a degenerative condition (n=9). Concomitantly, they all described having lived different lives prior to their accident or development of their condition. This demonstration of biographical disruption apparently led to increased negative psychological and emotional

wellbeing, as the majority of participants reported spending considerable periods reflecting on and regretting the loss of their former selves.

It is important to note, however, that five of the nine participants had been diagnosed with clinical depression. Clinical depression is a common development in people diagnosed with degenerative conditions and brain injuries and often leads to reduced social interaction (Hall and Havens, 2003; Sixsmith, 2012). The nine participants diagnosed with degenerative conditions/brain injuries were less likely to attend daycentres or participate in activities outside their home environment. Simultaneously, they also reported a predisposition to remain within the confines of their home environment. However, they perceived their failure to access activities outside their home environment as exercising choice rather than as a service failure.

The remaining participants (n=13), however, reported that when provided with social events they valued the opportunity to socialise, meet new people and become involved in shared activities as these helped to alleviate boredom and loneliness. This gave them the opportunity to feel that they belonged to a community and increased their sense of self-worth. Additionally, participants also reported that these encounters often facilitated information and knowledge exchange.

A - I filled in an advert in paper and I've got this booklet called *vitalise* and that's great for telling you what you can get and where to get things like funding and help for holidays. Oh, that reminds me. I said I'd bring that in for one of the women here. That's one good thing about coming here, you get to find out about stuff you would never know about otherwise. (Denise)

Bonding and bridging social capital, which Putnam (2000) defined as the "Connections among individuals – social networks and the norms of reciprocity and trustworthiness that arise from them" (p.9), played an important role in alleviating isolation and loneliness. Many talked about the importance of feeling well enough to access informal systems of support in a natural environment.

A - I've always spent a lot of time in my garden and I've got it looking really nice. The birds come in and the cats too of course. I like to feel connected with nature. Sitting outside in my garden on a warm sunny day with my neighbours and a couple of old pals having a drink or two [laughs] and

talking about the old days, you forget your troubles for a bit and then you go to get up and remember that you're old [laughs]. It doesn't get much better than that though Jo. (Moira)

#### 6.4.6: **Summary**

Participation in society via engagement in informal and formal social activities appeared to be proximally related to a reduced sense of isolation, loneliness and depression. Having a network of supportive people was vital for social wellbeing as those who felt lonely and isolated tended to report poor levels of personal wellbeing. Interpretation of the data suggests without social contact participants experienced a sense of loneliness, social isolation and increased levels of depression. Social wellbeing appeared to impact significant on psychological wellbeing by reducing feelings of depression, and thereby improving overall levels of personal wellbeing.

Biographical disruption and the interaction between cognitive impairments, depression and social isolation appeared to exert a significant impact on personal wellbeing.

Loneliness and social isolation were widely identified by participants as undermining personal, psychological and social wellbeing. The participants also identified that feeling socially isolated and lonely amplified feelings of depression, which further undermined personal and psychological wellbeing.

Social interaction and participation in society, particularly with close friends or those who shared common interests, was proximally related to an improved sense of psychological wellbeing. These reduced feelings of loneliness, boredom and isolation, which in turn were perceived as helping to alleviate feelings of depression. The participants reported little sense of personal responsibility for the alleviation of loneliness, depression and social isolation. Indeed, the participants almost uniformly identified this as the responsibility of statutory agencies. The failure of services to tackle social isolation was highlighted as indicative of a

tendency to focus almost exclusively on health and part of a wider failure to address wellbeing. This has been widely reported in QoL research.

The participants' narratives, however, accepted personal responsibility for undertaking actions and behaviours in order to facilitate filial proximity, a filial legacy and a matriarchal/patriarchal role. These were perceived as central to notions of social interaction and jurisdictions over which participants had some influence.

Assuming a matriarchal/patriarchal role within the family, whilst personally satisfying, could often marginalise efforts for personal development. Actions taken to ensure a filial legacy were, however, widely associated with those adaptive behaviours associated with making a successful transition to older age.

The availability of communal green spaces was widely perceived amongst participants as important for community cohesion and social interaction amongst neighbours and concomitantly personal, social and community wellbeing. The following section will explore in greater depth how the participants' narratives framed personal wellbeing with regard to the natural and built environments.

#### 6.5: The natural environment

The findings suggested that the natural environment exerted significant positive effects on personal wellbeing. The natural environment was perceived by nearly half of participants (n=9) as facilitating informal intergenerational and cross-cultural social interaction with non-filial others. This can be understood within the context of 'Self Determination Theory', which argues that "Wellbeing is both constructed and largely achieved through relationships" (Ryan and Deci, 2000:10). Thus, relationships and connectedness with others are central to basic need fulfilment, which enables people to thrive. Opportunities to connect with others were highly valued and communal green spaces were identified as important for wellbeing.

Our sense of place and the sense of space around places define our places and the sense of who we are (Tuan, 1974).

A - This place I live now has a sense of civic pride, the gardeners are smart and polite and it's like you're being cared for as well as the environment. Psychologically, I feel safer in this place because there is a resident who is big guy and he has made it his job to make his presence known on the green spaces out front. (Alison)

The natural environment was an aspect of personal wellbeing, which was often felt to have an inherent effect on participants' wellbeing, particularly their psychological and emotional wellbeing. Six participants' reported that feelings of depression were often lifted through immersion into the natural environment.

A - When I'm feeling a bit depressed I take myself off to Edale and I find myself feeling that life isn't all bad. The natural beauty of the place is uplifting and just being there for a few hours does more good for my spirits than anything else. (Jake)

For these participants, the natural environment exerted a calming influence. Some participants' (n=6) narratives reflected that the natural environment held intrinsic value for their personal wellbeing; this often appeared to be felt at a subconscious level.

A - I don't know why but just sitting in my garden and watching the birds flitting about and the squirrels scampering around gives me an inner sense of calm. (Jenny)

# 6.5.1: Communal green spaces

Nearly one third of the participants (n=7) identified communal green spaces as conduits through which social interaction took place irrespective of age, socioeconomic, religious and cultural barriers. Interpretation of the data suggests this social interaction helped transcend group's tensions, making a strong contribution to community spirit and cohesion and positively influencing both personal, social and community wellbeing.

Five participants were apprehensive about the preservation of communal green spaces, fearing that these areas may be lost to the community. Such a loss concerned participants because these spaces were valued for facilitating intergenerational contact and interaction between different social groups. Duany et al.

(2001) noted that in communities deficient in green public spaces "People of diverse ages, races, and beliefs are unlikely to meet and talk" (p.60).

A - Taking Tess (dog) out for walks is really the only time I feel truly happy. She is my best friend and having a dog means that you have to go out. It gets me out in all weathers and it's good to see how the park changes through the seasons. I love to see the flowers come up in spring and on top of that you tend to end up chatting with other dog walkers from all walks of life. (May)

For a number of participants, preserving local green spaces in the community was considered one way of enhancing personal and social wellbeing. Green spaces such as parks and commons were one of the limited spaces in the community where there was some degree of interaction between socio-cultural groups and intergenerational contact. These spaces were considered important for maintaining a sense of community spirit and community cohesion whilst simultaneously enhancing personal wellbeing.

A - Where I live about ten years ago they cleaned up this old derelict site, they planted trees, put in a little lake with picnic tables and play areas and you should see it now [sighs], it's beautiful. You get people of all ages there, dog walkers, courting couples. It's become the heart of things really. You can bump into people from all walks of life and have a chat. Little things like that are important to my wellbeing. (Raj)

#### 6.5.2: Gardens

Some participants spoke about the effects that gardening or being outside in the garden had for their personal wellbeing, such as better physical and mental health and an inner peace and contentment, which enhanced their sense of psychological wellbeing.

A - The natural environment is really important for my personal wellbeing. I'm lucky I've got a big garden which was the main reason we bought the house. We even put the conservatory on to 'bring the outside in' as they say. It's a haven for me really. I sit and watch the birds and the bees because my girlfriend has planted lots of flowers so it's a hive of activity out there. When I'm feeling low and the weather's ok I will go out and do a bit of gardening. Not only does it lift my spirits but being active means that I usually sleep better too. (Tim)

A - I love to sit here and look out onto the garden, to look at the birds and the trees and that; I can sit here for hours. (Caroline)

Q - It is a lovely room Caroline. You look right out into the garden, don't you? Why does that give you so much pleasure?"

A - It's so peaceful and pretty, with the flowers and trees, I can day dream to my heart's content. (Caroline)

Jim frequently talked about the importance of the natural environment for his emotional and psychological wellbeing. Jim, however, used his garden as an aid memoir, reliving his childhood interaction with nature to facilitate an inner sense of contentment. Jim's amalgamation of his childhood memories with present experiences in his garden sustained him during periods when he doubted his continued earthly existence, while additionally strengthening and sustaining his emotional and spiritual wellbeing. This can be understood within the context of research undertaken by Henwood and Pidgeon (2001), which suggested that a positive interaction with the environment during childhood leads to a sense of personal wellbeing, which endures into adulthood.

A - When I was younger I had fields and woods to play in and that was probably the best period of my life, when I think about those times now it makes me feel good, contented. (Jim)

Jim's garden facilitated reminisces of a period when his emotional wellbeing was to a large degree dependent on the experiences which he had in the woods of his childhood. Jim reported that his childhood experiences in the natural environment facilitated both his childhood and adulthood emotional wellbeing.

A - When I was a kid I spent all my time playing with my friends in the woods, they were simple times. All that was important to me then was having three or four meals a day, [drinks water] and the freedom to roam. I think being cocooned like that was what made my childhood so special [clears throat]. In the woods we could be anything, we played without a care

in the world. My thoughts often drift back to memories of playing in the woods, that's why nature and being close to nature is so important to me. I get a high from sitting and reflecting on when I was a kid playing in the woods. (Jim)

## 6.5.3: Unregulated green spaces

The findings indicated one aspect of the natural environment in particular had a significant and negative impact on specific domains of personal wellbeing.

Unregulated poor quality green spaces such as wasteland or overgrown derelict plots of land was most frequently identified as undermining emotional and psychological wellbeing.

Six participants (3 females and 3 males with mobility issues) reported that unregulated poor quality green spaces raised concerns for personal safety, primarily because they were perceived as crime hotspots. This was particularly troublesome in the long dark winter months when many felt a disinclination to leave the relative safety of their home environment.

A - My house is opposite a common and there's always kids hanging around and it's not very well lit so in winter when it gets dark early its really scary going to the shops. A couple of people have been mugged round there, the council should do something with it and there should be more police. They'd think twice if the police were regularly around. (Andrea)

A - It's £125 for a stand up scooter, because I can stand just can't walk, [stammers] guess I'd get used to it and then I could get out more. Mind you, we have to pass the park to get to the shops and that means running the gauntlet of the local thugs. The council let it get so overgrown that two or three of them could be waiting in the bushes to jump on you. (Simon)

This had a considerable negative impact on physical, emotional and psychological wellbeing because it restricted opportunities to meet and interact with others. This self-enforced seclusion had a particularly marked impact on psychological wellbeing as the participants felt increasingly lonely and depressed.

#### 6.5.4: Summary

The findings suggest that the negative associations between the natural environment and wellbeing are caused primarily because of fears for personal safety. The participants, particularly females and those with mobility issues, widely reported that unregulated green spaces had the potential to exert a negative impact on personal wellbeing. These concerns inhibited social interaction and concomitantly hindered connectedness with others.

The findings identified that a sense of accord with the natural environment was widely attributed as important for a holistic sense of personal wellbeing. The findings documented here largely replicate global findings identified in the existing literature. However, a widespread concern for the ongoing preservation of communal green spaces emerged as an important finding. Communal green spaces acted as conduits through which social interaction across age, social, cultural and class categorisations could flourish. These spaces were perceived as providing environments which facilitated and encouraged inter-generational and cross-cultural contact. These helped to transcend group's tensions, thereby making a strong contribution to community spirit and cohesion, and positively influencing social and community wellbeing.

The findings indicated that access to natural environments made an important contribution to feelings of emotional, social, psychological and community wellbeing. The majority of participants valued the natural environment, its calming influence and the benefits this contributed to energy levels and physical wellbeing. For those with mobility issues, regulated and private green spaces such as private and communal gardens offered opportunities to engage with nature.

### 6.6: Built environment\*

The majority of participants (n=17) noted the importance of living in a built environment that was well cared for, clean and provided a safe environment. The physical characteristics of the built environmental most frequently identified as

having a positive impact on wellbeing were well-kept environments; safe and sociable community hubs; quality green-space; and well-lit, well-maintained public footpaths. These physical characteristics appeared to be particularly important because they had a significant impact on the ability of participants to access, navigate and interact within their immediate surroundings. This in turn facilitated their sense of interdependence and ability to remain living within their local community. These findings mirror those reported in Burton et al. (2011) and Clarke and Nieuvenhuijsen (2009).

## 6.6.1: Aesthetic appeal

The aesthetic appeal of participants' built environmental was also reported as having a significant impact on wellbeing. The participants lauded council efforts to improve the environment, such as providing more street cleaners and recycling facilities. However, a significant number of participants (n=9) identified a need for local councils to increase efforts to re-energise the built environment and improve its aesthetic appeal. Interpretation of the data suggests that caring for the local built environment made participants feel that the local council were simultaneously caring for them too, and showing respect for the local people and their communal environs.

Q - How could services better reflect your wellbeing needs?

A - At times it's made me feel really low, like when every street was covered in broken glass and when the sun shined you could see it even more clearly sparkling. Now that we're recycling and have more street cleaners in residential areas I feel much better, less depressed by my surroundings. Services should look after the environment more. The built environment has a massive impact on my wellbeing, this used to be a subconscious thing but now I'm very much conscious of it. (Alison)

Participants commented on the importance of the local built environment and the vibrancy of community for wellbeing. The local neighbourhood and environment were perceived as contributing to that vibrancy and being important for encouraging a sense of common belonging. The participants also noted the

importance of living in clean, safe environments and inclusive, cohesive communities in which they felt socially accepted despite visible physical manifestations of their illness.

A - I don't feel part of the community at all because I don't get out into it. If I do manage to persuade staff to take me out I get so many stares that I feel like a leper. It really undermines how I feel about myself. (Silvia)

The physical characteristics of the built environmental most frequently identified as having a negative impact on wellbeing were housing and poorly maintained neighbourhoods. Research suggests that wellbeing in later life is closely related to the physical environment, which is an important mediator of ageing experiences and opportunities (Clarke and Nieuvenhuijsen, 2009; Burton et al., 2011).

Seven participants identified the degeneration of their built environment as a key factor in undermining community cohesion and civic life, which impacted negatively on social and community wellbeing.

#### 6.6.2: Community spaces

The findings indicated that community spaces were important for wellbeing and their decline was concomitantly important for wellbeing. The dissolution of community hubs was particularly detrimental for those participants who lived alone or who had impairments or disabilities, which reduced their opportunities to socialise. The eldest participants and those with mobility issues were particularly likely to highlight the loss and/or decline of community hubs and the profoundly negative effect this exerted on social and psychological wellbeing. One could argue that this finding reflects the importance of local meeting places, which provide opportunities to maintain contact with friends, develop social networks and engage in informal social activity. Oldenburg (1989, 2000) identified that in modern urban communities community hubs offer a neutral public space for people to connect and establish social relatedness, and bridging and bonding capital (Putnam, 2000). It has now been well documented that the availability of personal and public community spaces is integral to the creation of both bridging and bonding social

capital, which underpins community wellbeing (Berry, 2014; Christakopoulou et al., 2001; Putnam, 2000).

The closure of community hubs such as libraries, community centres, pubs and local shops had a detrimental impact on the participants' wellbeing. These local meeting places had provided an opportunity for members of the community to maintain contact with friends, develop social networks and engage in informal social activity.

A - I used to go to the local library or cafe and I would meet up with old friends and I also got to know other people who'd moved into the neighbourhood, and now they are gone most of my social life has gone too. There is no real sense of community anymore because there are no places left to have a brew and chat. Outside of the Jewish community there is no real sense of community round here anymore. (Andrea)

The findings suggest that the loss of community hubs constrained participation, engagement, friendships and involvement in social activities undermining social, psychological and community wellbeing.

#### **6.6.3: Community resources**

The provision and maintenance of community resources, which were well-run, secure and hospitable environments, were identified as having a positive impact on wellbeing, especially social and emotional wellbeing. These were also seen to cut across cultural and generational wellbeing requirements. Living in an environment which retained intergenerational and socio-cultural community hubs, as well as more traditional hubs such as public houses was of great importance for wellbeing. For many participants, a diverse range of community hubs were required to ensure the built environment facilitated social and community wellbeing.

## 6.6.4: Structural concerns

Structural concerns such as the poor design of accommodation, disrepair of public thoroughfares, inadequate public lighting, unkempt communal spaces and the dissolution of community hubs were seen as having significant negative impacts on

psychological, emotional, social and community wellbeing. The majority considered these aspects of the built environment to be beyond their power and control.

Consequently, these were identified as matters of local interest which local authorities should attend to in order to enhance citizen and community wellbeing.

The majority of participants reported the presence of police foot patrols as reassuring, particularly in dilapidated areas or areas which had not received regeneration funding. A strong police presence played a crucial role in helping allay participants' sense of physical vulnerability, and fears for personal safety. This was particularly the case in areas which had large unregulated green spaces, overgrown tree-lined paths, poorly illuminated entryways or derelict waste ground. Those with visual impairments or mobility issues felt particularly at risk in these environments. They often preferred to stay at home, especially after daylight hours, rather than risk venturing into these areas.

It has long been recognised that poorly maintained physical environments undermine wellbeing and contribute to depression, isolation and loneliness (Christakopoulou et al., 2001; Putnam, 1995, 2000; Rutter, 1995). Weich et al. (2002) reported that those living in deprived urban areas characterised by derelict buildings, graffiti and neglected public spaces were more likely to suffer episodes of depression than those who lived in areas with well-maintained buildings and public spaces. Older participants who lived in more deprived areas were particularly likely to describe feelings of depression and loneliness, which they associated with living in run-down environs. This lends further support for the environment stress hypothesis, which argues that the quality of a neighbourhood has a particularly significant impact on personal development, social relatedness and psychological wellbeing (Cairney et al., 2010; Cairney et al., 2013).

When reflecting upon the interactions between their personal wellbeing and the built environment, the participants prioritised the importance of living within environments which were safe and secure, in much the same way as participants noted safety concerns when accessing green spaces in the natural environment.

A - The way the place I live in now has been designed is really good because it's well lit and all the paths are overlooked by windows. This means that even in winter, the route to the shops is overlooked and so you don't think people would chance attacking you. House design is so important for my wellbeing. Proximity to your neighbours can really get to you, especially when you move from having kids and living in a house to being in a 1 bed flat where you're so much closer to your neighbours and you're surrounded on all sides. (Alison)

Regeneration schemes had been undertaken in areas of North and East

Manchester, and participants from these areas reported that local government

efforts which focused on re-energising the built environmental had an important
and positive impact on personal, social and community wellbeing.

- A- Getting involved with the In-Bloom competition was great. The area looked brighter and seeing flowers in baskets and planters lifted most peoples' spirits. (Alison)
- A- Where I live was really run down but the regeneration team have done wonders. They have tried to get people of all ages involved; now the place looks so much better, and there is more of a sense of community now. The council should be spending more money on things like that. (May)

## 6.6.5: Living space

The participants' living space was identified as an important dimension of wellbeing. Older adults and those with LTCs spend more time in the home environment than any other (Allen, 2008; Black et al., 2015; Stewart and Bushell, 2011). Housing and the importance of well-designed and hospitable homes were frequently cited as being of great significance for wellbeing. Housing issues were frequently associated with reduced sense of independence and appeared to have a complex relationship with the participants' sense of wellbeing.

- Q- "What things could services do then which would improve your personal wellbeing?"
  - A Well I can't get in my kitchen because the doors aren't wide enough. I need an adapted flat for myself but they won't let me have one. I would be

more independent if I had an adapted flat. I would be able to do some cooking and all that. (Denise)

Housing issues such as inadequate and poorly-designed living space were important sub-themes consistently identified by participants as impacting negatively on wellbeing. Issues such as housing design, condition, tenure and so on led to depression and anxiety, both of which had a major impact on psychological and emotional wellbeing, and to lesser degree physical wellbeing. Poorly designed living spaces frequently undermined independence and exacerbated feelings of powerlessness. These had a negative effect on the participants' day-to-day and long-term future prospects of wellbeing.

In some cases, participants noted that the council house design prevented them from accessing resources which could improve their independence. Many of the participants who lived in council accommodation felt that the living space had been poorly designed.

A - Adaptations have supplied my chair, bed and shower chair as well. They tried to install a monkey pole [a grab rail] but the walls are breeze block and you can't drill through breeze-block. That's just one of the many problems you face living in poorly designed council property. (Jake)

Those participants who did not own their homes (n=16) spoke about feeling 'at the mercy' of professionals who could rescind tenancy agreements for what were often considered spurious reasons. Those who lived in residential care accommodation (n=4) appeared to feel particularly at risk of this and reported this as a constant source of negative personal wellbeing

A - My tenancy has been threatened a few times because of my relationship with the staff and I've been bullied and it was when I was highly strung and felt threatened and powerless and when you have the staff shouting at you it doesn't help...The issue with tenancies isn't just a problem for me, X [fellow resident] has been threatened with losing his tenancy because he stands up for himself too and Y [fellow resident] has a long running feud with them about his tenancy. (Audrey)

The majority of participants who lived in the new warden-assisted accommodation reported that their design, construction and layout had a profoundly positive

impact on wellbeing because it facilitated social interaction and freedom of movement. The newer and smaller warden-assisted living accommodation was split into individual or two person flats, rather than single occupancy rooms. In stark contrast to those living in residential homes, participants in the warden-assisted accommodation were less likely to feel their flat wasn't their own or fear reprisals from staff such as the revocation of their tenancy agreement. The warden-assisted living accommodation, introduced in 2005, was considered a great improvement on residential housing accommodation. This was primarily because residents appreciated the personal and communal space it afforded them which allowed people to come together and socialise. This was important as it encouraged residents to leave their respective flats and socialise with other residents. This facilitated social interaction, itself seen as a fundamental platform for improving social and emotional wellbeing. The communal area was particularly popular in the warmer months, when the area was utilised for barbeques and parties.

Additionally, warden-assisted living accommodation heightened the participants' sense of independence and interdependence as they were often given the opportunity to share a flat with a friend. The majority of residents who moved to this accommodation from residential accommodation felt that this gave them the support they needed to establish a more interdependent lifestyle. This was not always the case, however.

#### Q - What undermines your personal wellbeing Ben?

A - The lack of living space and its design. I can't reach anything in my kitchen, it's too small for me to turn my wheelchair round in and I can't reach the cupboards, the fridge, anything really. That means I'm reliant on the staff to cook and clean and that reduces my independence, the very thing that coming here was supposed to achieve. (Ben)

Feelings of powerlessness exacerbated the effect which poor living spaces had on the participants' day-to-day and long-term future prospects of wellbeing. The participants desired housing which would enable and facilitate amongst other things, some measure of independence or interdependence.

#### 6.6.6: **Summary**

The findings indicated that the built environment made a strong contribution to feelings of social, emotional and community wellbeing and to a lesser degree physical and psychological wellbeing. The findings suggest housing concerns such as design, layout, condition and tenancy agreements are pre-eminent. Other aspects which appear to play an important role in wellbeing include dissolution of community hubs and unregulated, poorly lit unkempt communal spaces, which increased fears for personal safety.

The interior and exterior design and layout of dwelling spaces had a particularly important bearing on wellbeing. For those participants who lived in council-rented and residential accommodation, poor housing design undermined independence and led to negative wellbeing. This was associated with a sense of powerlessness and daily anxiety, which frequently undermined the participants' present and future sense of wellbeing.

Conversely, homeowners (n=6) reported a sense of empowerment as homeowners. They felt in control of their personal dwelling space and felt able to make any necessary structural or cosmetic adaptations to suit their individual requirements. This was particularly the case in relation to the reconfiguration of space to accommodate disabilities associated with their LTCs.

Structural constraints were particularly important for participants who felt they had little power to exert influence on the built environment beyond that pertaining to those who owned their dwelling space. The participants reported that regeneration schemes had led to cleaner and more aesthetically pleasing aspects of the built environment. These enhanced emotional, social and community wellbeing primarily because these improvements were widely perceived as facilitating community spirit and cohesion in the same way as the natural environment.

Many believed that personal wellbeing could be improved through the funding of regeneration schemes, which focused on re-energising the built environment.

These were perceived to be more likely to be achieved through structural changes initiated at local and national government level. The endeavours of individuals or

community networks were perceived as ineffectual and too small to compete with other vested interests.

Structural concerns such as the poor design of accommodation, disrepair of public thoroughfares, inadequate public lighting, unkempt communal spaces and the dissolution of community hubs were seen as having significant negative impacts on personal, social and community wellbeing. The majority of participants considered these aspects of the built environment to be beyond their power and control. Consequently, these were identified as matters of local interest, which local authorities should attend to in their duty to enhance citizen and community wellbeing.

#### 6.7: Temporality

"Temporality has long been considered a fundamental context and mechanism for the construction of meanings, and for individuals' reflexivity and self-evaluations" (Guell et al., 2014:15).

Temporality\*, particularly in reference to the three temporal states of attitudes\*, preferences\* and perspectives\* emerged throughout the participants' narratives and appeared to play an important role in perceptions of long and short-term wellbeing.

The co-researchers' accounts of their wellbeing in particular exhibited a fluidity, which spanned the past, present and future in such a way that they appeared intricately interlinked.

Q - Do you feel that your early years were important for your current sense of personal wellbeing Alison?

A - I think it has played a large part in my wellbeing. My feelings of insecurity, issues with self-worth, self-esteem, handling disappointment, issues of addiction, lack of education and learning difficulties have and

continue to have an important bearing on my day-to-day sense of wellbeing and what I expect for my future wellbeing. (Alison)

For participants with illnesses which were likely to shorten their life span, narratives often arose in relation to future deterioration of the condition. In this context, participants discussed storing up current feelings of positive wellbeing, which could be used to negate future negative wellbeing. In addition to which, the participants frequently reminisced about past happy memories to enhance and reinforce present wellbeing.

A - On my good days I try and do as much as possible so that in times down the line when I'm stuck inside because I'm too poorly I can think over my good days and what I've achieved and that helps me feel a bit happier about my lot. (Simon)

#### 6.7.1: Temporal preferences

Some participants' accounts of current wellbeing appeared to be grounded in experiences of wellbeing and fundamentally related to their temporal attitudes, perspectives and preferences. Jim appeared to exhibit a past temporal preference. Jim's recollections of past wellbeing and its impact on present wellbeing were almost exclusively discussed in relation to childhood memories and the connections he had forged to the natural environment during this period of his life.

A - When I was young we lived at the back of some woods and me and my friends spent all our time there. When I watch the conkers come out now it instantly transports me back to a time when I was gloriously happy without even knowing it. Putting the conservatory on was the best thing we did because it brings me closer to nature. When I sit there it takes me back to those days spent playing in the woods and I've spent many a happy time there listening to the birds and dreaming of my childhood. (Jim)

This quote illustrates the process whereby people experience a sense of wellbeing by drawing upon past feelings of wellbeing. A significant minority (n=5) appeared to have a present temporal preference. They reported amassing current feelings of positive wellbeing in anticipation that they can be used to generate future feelings of wellbeing.

A - Sometimes when I'm out with my dog and I'm enjoying our walk I make a mental note to remember it and how good I feel at that moment because I know that I might need to relive that feeling later on when I'm a bit depressed and lonely. (May)

Bryant (2003) and Galvin and Todres (2013) noted a past temporal preference such as an individual's tendency to reminisce about the past helps them to project a more positive sense of wellbeing for the present and future.

#### 6.7.2: Temporal attitudes

A - It helps to have a clear picture of my past. Many conflicts have arisen between me and [my] only remaining sibling re childhood memories...This matters big time as I have no one to fix those memories with. That conflict of different memories continues to make me question who I am. Putting the pieces of my past together and reflecting on how that's affected my wellbeing undermines my wellbeing now and how I view my future wellbeing both consciously and subconsciously. (Alison)

Durayappah's (2010) model would suggest that Alison's personal wellbeing is affected by her retrospective bias. Using Durayappah (2010), Guell et al. (2014), Galvin and Todres's (2013) synthesis of temporality, Alison's narrative appears to reveal a negative temporal attitude to the past. Alison displayed within her wellbeing narrative a persistent retrospective analysis of her life. These included references to opportunities not taken and a projected visualisation of the life she might have led had she made use of the opportunities which she felt she spurned during her teenage years. Alison constantly revisited these feelings and consciously ascribed them as intimately connected with her current negative sense of wellbeing and perception of future wellbeing. Alison and Jim (co-researchers) both appeared to display a past temporal attitude, but whilst this afforded Jim feelings of positive wellbeing, it provided Alison with feelings of negative wellbeing.

## 6.7.3: Temporal perspectives

Over half of all participants (n=13) appeared to use temporal perspectives as a means of contrasting past, present and future levels of wellbeing. The majority

(n=7) appeared to exhibit a past-positive temporal perspective. Two appeared to exhibit present-hedonistic perspectives (i.e. taking pleasure in the present), while the remaining four displayed past-negative perspectives.

Q - Can you think of any examples of how local government could improve the wellbeing of older adults with LTCs Denise?

A - I would love to go on an organised holiday like they [community centre] used to do for us. Having something to look forward to helps you get through and staves off depression. I can while away many an hour thinking about past holidays: walking on the beach, smelling the sea air, eating fish and chips. My husband and I loved doing that. I think about those good times a lot, especially when I'm stuck in the house of a night it gives me a real lift and helps me face tomorrow. (Denise)

The participants frequently retold stories from their past or vocalised aspirations for their future, which appeared aligned to a temporal dimension of their sense of wellbeing. Having something to look forward to (future wellbeing) was repeatedly located within the participants' narratives as making a meaningful impact on present wellbeing. This finding is validated by Rozin (2008a, 2008b) and Galvin and Todres (2013), who highlighted that happiness can be achieved temporally through experience, memory, and anticipation.

Q - Why does 'that' [her garden] give you so much pleasure Caroline?

A - It's so peaceful and pretty, with the flowers and trees, I can daydream to my heart's content. (Caroline)

Q - What do you daydream about?

A - [sigh and long pause] Lots of things [pause], think about when I was a kid playing with my brother and sister, when we used to go on holiday and my dad would take us fishing, stuff like that. (Caroline)

Conversely, the anticipation of feeling unhappy, particularly in relation to health, was also noted as having a detrimental impact on wellbeing.

A - It's important to me to socialise outside the house. I used to go to the pub but I drink less now because I would go out and have a few drinks but I was having to get up too many times in the night and I felt tired and rotten

in the morning. So it was reducing my wellbeing because I didn't look forward to feeling poorly in the morning. (Raj)

## 6.7.4: Summary

Temporal attitudes, preferences and perspectives appeared to play an important role in conceptualisations of long and short-term wellbeing. Wellbeing narratives frequently included reminisces of happy memories and experiences which enhanced and reinforced present wellbeing. For those with illnesses which threatened to reduce their natural life-course, temporal narratives often emerged in relation to some future period of time when their health condition had deteriorated. There was a perception this future sense of negative wellbeing could, to some extent, be offset by amassing current feelings of positive wellbeing, which could be drawn on in the future.

The findings suggested that for many, wellbeing was directly influenced by temporal attitudes, preferences and perspectives. These often emerged in relation to connections forged during childhood.

The majority of participants widely reported that anticipation of future wellbeing (i.e. having something pleasurable to look forward to) had an important bearing on present wellbeing. Conversely, the expectation of prospectively feeling discontented or unhappy with life, particularly in relation to health, was also noted as having an important impact on negative wellbeing.

#### 6.8: Summary of study findings

The findings revealed that seven key thematic domains were perceived as the most important and non-negotiable aspects of wellbeing for older adults with LTCs. Within these domains there were numerous sub-themes which were closely interrelated. The participants widely acknowledged wellbeing as a multi-dimensional concept encompassing a broad range of socio-economic, physical, socio-political,

environmental and psychological components. Personal wellbeing was considered highly subjective and situational, and linked to notions of cultural belonging and social identity. The findings indicated that feelings of wellbeing were inconsistent and liable to change over relatively short periods. In addition to which, some participants reflected that what was important for wellbeing changed during different periods of one's life, particularly during life transitions such as becoming a parent or being widowed. It is also important to note that some aspects of an individual's wellbeing, particularly those of a sensitive nature, may not be expressly vocalised or may be couched in obtuse language. Examples documented in this study included the role of sexual intimacy and political circumstances on wellbeing.

Personal wellbeing was also influenced by factors such as severity of condition, age at time of onset of condition, familial and financial circumstances, access to social intercourse, temporality, environmental circumstances, illness trajectory, perception of the condition as a 'natural' part of ageing, and length of time living with a condition.

Living with LTCs appeared to amplify feelings of disenchantment with service provision. Concurrently, it enhanced feelings of loss of power and control, social isolation, reliance on welfare, systemic poverty, mental health issues, depreciation of a role as a valued member of society and a reliance on others. All of these had a sustained and long-term impact on personal wellbeing.

There was real accord amongst participants that conceptualisations of wellbeing are highly subjective. The majority of participants believed efforts to define wellbeing were misplaced, preferring personalised conceptualisations of wellbeing.

#### **Chapter 7: Discussion**

#### 7.1 Introduction

There were a number of key findings from the research. The following section briefly synthesises the theoretical and empirical analysis before proceeding to those sections which put the findings into context.

The historical/linguistic analysis identified a latent, long established association between wellbeing and personal responsibility. This has facilitated the mobilisation of wellbeing within a PRA. However, the empirical evidence reported that the majority of study participants did not associate wellbeing with personal responsibility. The theoretical investigation facilitated an appreciation of how and why the UK government has mobilised agentic representations of wellbeing consistent with a wider PRA. Notwithstanding the role of wellbeing in government policy, the empirical component of the study suggested that wellbeing is not considered a meaningful concept, and it held little currency for the study participants. It was most frequently associated with the language of professionals, media and state.

This thesis established that there is theoretical evidence that wellbeing has been mobilised within a PRA (Ahmed, 2010; Miller, 2008; Peck, 2013). However, as Kingfisher (2013) noted, little research has been undertaken to explore the role of historical and linguistic factors in this mobilisation, particularly how and why wellbeing has been mobilised.

# 7.2: How wellbeing has been mobilised

Interpretation of the findings suggests a number of interconnected reasons, which explain how historical/linguistic factors have facilitated the mobilisation of wellbeing.

First, the findings suggest wellbeing has been mobilised through agentic representations of wellbeing, which rely on latent and historical associations between wellbeing and personal responsibility. For example, the ancient Greek theory of dietetics (i.e. personal responsibility for health) and its association with the ancient Greek concept of wellbeing have shaped modern conceptualisations of wellbeing as underpinned by personal responsibility and human agency. These largely disregarded associations between wellbeing and its inherent, latent or intrinsic historical meanings facilitate the current discursive practice of promoting wellbeing within a PRA. Bergdolt (2008), McMahon (2006) and Oishi et al. (2015) have reported elements of these findings. However, their studies conflate wellbeing with happiness and health, clouding specific understandings of wellbeing as a distinct concept. This study, whilst recognising the associations between wellbeing and coterminous terms, is specifically concerned with wellbeing.

Second, findings indicate that wellbeing has been mobilised within the discursive language of the intellectual and the state (Atkinson and Joyce, 2011; Bache et al., 2015; Scott, 2012a). Interpretation of findings suggests this mobilisation has been achieved in part through its linguistic Greek heritage, which marks it as a high status concept popular within professional and academic discourse (Hughes, 1988).

Third, mobilisation has been achieved through a range of options. Government policy documents and 'expert voices' such as NEF and the Wellbeing Czar, Lord Layard, have utilised agentic representations of wellbeing which promotes a PRA (Ahmed, 2010; Bache et al., 2015; D. Bok, 2010; Edwards and Imrie, 2008). In addition to which, wellbeing has been mobilised through government rhetoric, legislation, consultation exercises, the politico-media complex and neoliberal discourse.

Elements of these findings have been reported in previous theory-based research, for example, Bache and Reardon (2013), Abdallah et al. (2012) and Lammy and Tyler (2014). However, this study considered examples of the mobilisation of wellbeing through theoretical and applied contexts. In addition to which, previous research has tended to work within single discipline boundaries. This study expands

on these through the utilisation of a multi-disciplinary perspective, which synthesised evidence from across a broad range of disciplines.

Having addressed 'how' wellbeing has been mobilised, the following section considers 'why' historical/linguistic factors have facilitated the mobilisation of wellbeing.

# 7.3: Why wellbeing has been mobilised

Health, welfare and wellbeing were shown to share linguistic roots and a shared historical association with notions of personal responsibility. This shared historical/linguistic heritage has facilitated the appropriation of wellbeing as a conceptual bridge. The role of wellbeing as a conceptual bridge between health and welfare has enabled its appropriation within a wider PRA (Stenner and Taylor, 2008; Taylor, 2011). This bridging concept has been mobilised within a PRA for governance purposes. The UK government has used the language of wellbeing to promote agentic representations of wellbeing. These representations promote the adoption of lifestyle choices and behaviour change, which aim to reduce health and welfare expenditure to a population that is ageing and has a growing number of citizens with preventable LTCs (Fullager, 2002, 2009; Seedhouse, 1995; Taylor, 2011).

Alternative theories about the mobilisation of wellbeing include those which suggest that wellbeing has not been mobilised to promote a PRA, but instead represents over intrusion by the state into the private matters of the individual (Bjornskov, 2012; Furedi, 2004; Vos, 2012). The NEF acknowledges the mobilisation of wellbeing, but disagree that this has been to promote a PRA. Instead, they propose wellbeing has been mobilised to provide governments with a more holistic measure of societal progression than economic measures (Aked et al., 2008; Marks and Shah, 2005; Seaford, 2011). These are credible claims. However, this study has documented evidence that wellbeing has been mobilised to help reduce health and welfare expenditure in a nation with an ageing population and increasing numbers of people with preventable LTCs.

This study argued that wellbeing has been mobilised within a PRA to move away from the promotion of wellbeing as a state responsibility based on economic justifications, to the promotion of wellbeing as a personal responsibility based on socially progressive rationalisations (Abdallah et al., 2012; Self et al., 2012; Shamir, 2008a). This resonates with Barnes et al. (2013) and Taylor (2011) who argued that wellbeing has become "Discursively linked to policy objectives related to individual responsibility and self-governance" (Taylor, 2011:793).

# 7.4: Is there empirical evidence participants associated wellbeing with personal responsibility?

Evidence collated in the literature documented the debate about the division of responsibility for wellbeing between the individual and state (Halpern and Bates, 2004; Kingfisher, 2013; Perri, 2007; Seedhouse, 1995). This raised the question of who can and who should be considered responsible for wellbeing (Booth, 2012; Kingfisher, 2013; Seaford, 2011). Whilst theoretical research has investigated the division of responsibility for wellbeing, little empirical research has investigated lay opinion about this division of responsibility. The current study helped to explore this by investigating whether participants associated wellbeing with personal responsibility.

As documented in the previous section, findings indicated the majority of study participants did not associate wellbeing with personal responsibility, except in relation to certain aspects or domains of wellbeing. This appeared to be most closely aligned to aspects which pertained to physical wellbeing and to a lesser degree filial and community wellbeing. The participants rarely offered narratives in which wellbeing was associated with personal responsibility. This was primarily done in relation to government health and welfare interventions. The participants' narratives occurred alongside a subtext of assuming personal responsibility for aspects of physical health such as adopting healthy diets and undertaking more physical exercise. Less frequently the participants accepted personal responsibility for personal relationship, especially within familial contexts and some aspects of community wellbeing through volunteering. There was some evidence to suggest

the participants perceived assuming responsibility for these aspects of wellbeing assisted local government efforts to reduce public spending by encouraging individuals to undertake duties which had previous been performed by statutory or charitable organisations.

One possible explanation for this is that the discursive language of wellbeing has focused on lay responsibility for physical health, family wellbeing and community cohesion (Atkinson and Joyce, 2011; Kingfisher, 2013; Scott, 2012c). The UK government widely publicised the notion of 'The Big Society' and the Localism Act (2011), in which citizens were exhorted to enrich community wellbeing through behaviours such as volunteering (Bache and Reardon, 2013; Scott, 2014). This exhortation was well supported through the utilisation of 'expert voices' such as Lord Layard and NEF, policy documentation and media channels (Scott, 2012a; Seaford, 2011). As previously documented, the NEF publication 'Five Ways to Wellbeing' (Aked et al., 2008) was widely disseminated to health and social care professionals. This publication promoted personal responsibility for physical and community wellbeing in the name of self-sufficiency and empowerment (Kingfisher, 2013; Peck, 2013; Scott, 2014).

The convergence between governance rhetoric and lay narratives with regard to personal responsibility for physical wellbeing and aspects of community wellbeing may represent an intersection between lay narratives and politico-media discourse. Alternatively, the discursive language of wellbeing may have been assimilated into lay perspectives, which is suggestive of the successful appropriation and mobilisation of wellbeing for certain aspects of wellbeing within the context of the PRA. Alternatively, notions of individual responsibility for physical health, family and community wellbeing may be indicative of the latent historical/linguistic associations identified in chapter five.

However, it is difficult to intimate that this provides sufficient evidence to support the theoretical premise that wellbeing has been mobilised to facilitate the reduction of government spending on welfare and health bills to an ageing population. These findings are interesting, however, in the light of the proposed

health and social care transformations under the North-West devolution deal, which depends upon citizens taking personal responsibility for their health and wellbeing (Greater Manchester Combined Authority, 2015).

Participants frequently located responsibility for much of personal wellbeing with the state/statutory agencies. The participants, for example, reported little sense of personal responsibility for the alleviation of psychosocial aspects of wellbeing such as loneliness, depression and social isolation. Indeed, the participants widely identified this to be the responsibility of statutory agencies. Many perceived that psychological wellbeing, economic wellbeing, environmental wellbeing, and community wellbeing were primarily the responsibility of the state or its statutory agents.

Other issues which undermined wellbeing such as the reduction and restriction of welfare provision and the loss of community assets and community networks of power were widely considered the responsibility of the state. For some the loss of these assets was perceived as an act of conscious disempowerment, perpetrated by those in positions of authority. Others construed these as unfortunate events, which might be mitigated in the future by improving politico-economic conditions. These research findings expand on the findings and understandings of others, such as Sointu (2005), Ferguson (2007) and Scott (2014a).

Some participants, particularly the two co-researchers, reported that they thought wellbeing had been mobilised by the media, professionals and marketing campaigns as a means to 'sell' a lifestyle or to influence ways in which people lived their lives (Edwards and Imrie, 2008; Foucault, 1988, 1991; Rose, 1999; Sointu, 2005). This suggests that there is some evidence that wellbeing was perceived as concerned with approving and validating particular behaviours and lifestyle choices (Ahmed, 2010; Fullager 2009; Rose, 1999). However, again it is difficult to claim this provides sufficient evidence to support the theoretical premise that wellbeing has been mobilised as a socio-political tool concerned with approving and validating particular behaviours and lifestyles choices.

This study found little evidence to support the theoretical premise that wellbeing has been mobilised to obligate/persuade citizens to accept responsibility for aspects of wellbeing which may be beyond their control. The majority of the participants in this study had not adopted personal responsibility for aspects of their wellbeing which could be considered beyond their control, such as urban planning, housing design, or the allocation of resources. Concepts such as agency, autonomy and control, which many argue are fundamental to wellbeing, had apparently not been assimilated into the participants' narratives. Those familiar with wellbeing through contact with health and social care professionals reported that these concepts had been promoted as central to wellbeing. However, whilst acknowledging these concepts as important *per se*, a significant number of participants reported that they were inappropriate in their conceptualisations of wellbeing.

The findings indicated that the majority of participants displayed little sense of agency. One can only speculate upon the reasons for this. However, it is widely acknowledged that feelings of insecurity, hopelessness and powerlessness all impact on people's agency (Barker, 2005; Camfield, 2006; Littlejohn and Foss, 2009). These in turn are widely recognised as undermining wellbeing (Giroux, 2013; Hughes et al., 2014; Lawler, 2014). Many participants reported feeling disrespected by and powerless in the face of officialdom. This led to a perceived failure to exert power and control over their lives, which may have contributed to a reduction in their confidence and ultimately their agency. This finding is mirrored in a study by Barnes et al. (2013) and Ward et al. (2012), which reported that notions of agency or autonomy were not perceived as particularly meaningful or appropriate for older people as they failed to recognise 'interdependence' and the importance of friendship and mutual dependence.

There could be a number of potential explanations for why participants did not associate wellbeing with personal responsibility.

One explanation is that feelings of powerlessness and lack of agency influenced perceptions of responsibility for wellbeing. The theoretical literature suggests it is

the subjective feelings of powerlessness and lack of control which fundamentally undermine personal wellbeing (Baltatescu, 2007a, unpublished PhD thesis; Ehrenreich, 2010; Perri, 2007).

An alternative explanation is that the findings are cohort specific and reflect historic/linguistic associations between wellbeing and welfare. The rise of welfarism in the 20<sup>th</sup> century marked a period when the provision of welfare via state legislation and policies was considered to underpin citizen wellbeing (Mowat, 1952; O'Brien and Penna, 1998). This period of welfare expansion encouraged a shift in conceptualisations of wellbeing in which structural representations of wellbeing predominated. These prioritised state responsibility for wellbeing. The majority of the participants' formative and early adulthood years occurred during the 1945—1979 period, identified by Froggett (2002) as the era of the 'classic welfare state'. During this period, the provision of societal and individual wellbeing were envisaged as part of the state's responsibility to co-ordinate social welfare (Froggert, 2002; O'Brien and Penna, 1998). This may have influenced this cohort of participants' perception that the state is fundamentally responsible for most aspects of citizen wellbeing within the context of the welfare state.

Whilst the participants did not perceive all the discursive elements of an association between wellbeing and the PRA, interpretation suggests that there was an appreciation that wellbeing had been utilised within broader health and welfare agendas.

## 7.5: Is wellbeing a meaningful concept for lay people?

Seedhouse (1995, 2002) raised concerns that wellbeing is an empty concept for lay people and only has real meaning for those in a professional capacity. This prompted the question: Is wellbeing a meaningful concept for lay people? (Edwards and Imrie, 2008; Eraurt and Whiting, 2008; Mathews and Izquierdo, 2009). The findings from this study indicate that wellbeing is perceived predominantly as a professional/academic concept which has little transferability or meaning for those from non-professional/non-academic backgrounds. This may be explained by its

historical/linguistic legacy as a linguistic marker of social and educational status.

Until recently, wellbeing was used almost exclusively in the language and discourse of academics and professionals. The findings suggest this has not changed.

Wellbeing was not a meaningful concept for the majority of participants in the study and was rarely used with peers. This resonates with claims that the term 'wellbeing' is not well understood and is rarely used by lay people (Eraurt and Whiting, 2008; Mathews and Izquierdo, 2009; Seedhouse, 1995). Interpretation of the data suggest wellbeing has limited lay currency, as the findings revealed low recognition rates amongst the majority of participants. This is despite high media profiling and extensive government and professional rhetoric. Whilst this may be a reflection of the age of participants and exposure to media messages, it does not account for the lack of usage reported within everyday interaction or the reported lack of identification with the term. Scott (2012c) has also documented similar findings.

Wellbeing is a term which the majority of participants felt little compunction to reflect upon unless directed by those pursuing an agenda (including myself). When the participants did reflect on its meaning it was primarily perceived proximally in relation to acceptable forms of behaviour, predominantly in relation to health and happiness. As previously documented, this may be because primarily their health and social-care professionals had introduced wellbeing to them. Alternatively, it may be indicative of the latent historical associations between these concepts.

Despite barriers to enhanced understanding, there was at a fundamental level a general perception that wellbeing roughly equated to QoL, and to a lesser degree welfare. These two terms were widely preferred as relevant and understandable to people from non-professional and non-academic backgrounds. These findings have been reported previously by the Department for Children Schools and Families (2007) and Eraurt and Whiting (2008). Both concluded that wellbeing was not a familiar term to lay people and that lay people struggled to conceptualise it. Conversely, Chapman and West-Burnham (2009) argued that "The concept of wellbeing is well understood in everyday usage" (p.37).

The participants perceived wellbeing as a highly subjective, complex, and multi-layered concept; this may help to explain why it was not considered meaningful in everyday contexts. In addition to which, it was conceptualised as subject to additional cultural, language and historical differences and interpretations, which rendered it a highly subjective personalised concept.

Alternatively, wellbeing may not be considered a meaningful concept because of the participants' lack of exposure to the concept. Vernon (2008) argued that wellbeing has currency as "A useful word because it is relatively unfamiliar" (p.44). However, since the study's inception, the ONS has run a series of public consultation exercises to underpin the collection of national wellbeing (ONS, 2011a, 2011b, 2012c). This may have led to an increased awareness of the concept and a greater appreciation of the meaning of wellbeing across society.

The following sections review two previously adumbrated aspects of wellbeing this study has helped to illuminate.

# 7.6: The role of the natural environment on wellbeing

Much has already been written about the links between the natural environment and wellbeing (Bird, 2007; Newton, 2007; Townsend and Weerasuriya, 2010). However, much of the research originates from the US, is quantitative and tends to focus on the spatial planning and health effects of green spaces (Ballas, 2007a, 2010; Ballas and Dorling, 2013; Lee and Maheswaran, 2011). Global research indicates that the natural environment has the potential to have a considerable impact on wellbeing (Abdallah et al., 2012; Commission for Architecture and the Built Environment, 2009; Stewart and Bushell, 2011). However, the extent to which these findings are replicated in the UK required investigation (Newton, 2007; White et al., 2013). Further research was required to unpack the relationship between the natural environment and wellbeing, and particularly which aspects of the natural environment impact on wellbeing (Berry, 2014; Sustainable Development Commission, 2007a). The following section begins by considering why unregulated

poor quality green spaces and concern for the preservation of communal green spaces contributed most to negative wellbeing.

The findings indicated that unregulated poor quality green spaces, such as wasteland or overgrown derelict plots of land, were most frequently identified as undermining psychological, emotional, social and community wellbeing. There are a number of potential explanations for this finding.

First, because unregulated poor quality green spaces act as a barrier to social interaction and human contact (Commission for Architecture and the Built Environment, 2009; Newton, 2007; Stewart and Bushell, 2011), this undermines a fundamental psychological and emotional human need, that is, 'connectedness with others', which has been widely identified as a basic psychological human need. 'Self Determination Theory' (Ryan and Deci, 2000) and the 'Theory of Human Need' (Doyal and Gough, 1991) recognise the significance of relatedness and relationships to wellbeing and acknowledge the importance of nature within this (King, 2007; Newton, 2007).

Second, unregulated aspects of the natural environment stirred fears for personal safety which led to self-imposed confinement in isolated first spaces, i.e. the home (Oldenburg, 1989, 2000). This had a significant negative impact on psychological, emotional and social wellbeing because it restricted opportunities to meet and interact with others. Concomitantly, it also contributed to feelings of loneliness, isolation and depression. 'Human Flourishing Theory' (Nussbaum, 2000) argues that the attainment of wellbeing is contingent upon social interaction. Paucity of social contact undermines wellbeing at a fundamental level.

Widespread concern for the preservation of communal green spaces also made an important contribution to negative wellbeing. The natural environment, particularly communal green spaces, was perceived by nearly half of the participants as facilitating informal intergenerational and cross-cultural social interaction with non-filial others. This can be understood within the context of 'Self Determination Theory', which argues that "Wellbeing is both constructed and largely achieved

through relationships" (Ryan and Deci, 2000:10). Thus, relationships and connectedness with others are central to basic need fulfilment, enabling people to thrive.

Taken together, these findings suggest that concerns over unregulated poor quality green spaces and the preservation of communal green spaces need to be incorporated into existing debates regarding how best to enhance individual and community wellbeing.

However, the study findings suggest interaction with the natural environment has a predominantly positive impact on wellbeing, particularly psychological, emotional and spiritual wellbeing.

Over half of the participants highlighted the value of their gardens in reducing tension, facilitating a sense of calm and as a source of ongoing pleasure. The participants also highlighted the role the natural environment played in facilitating reflective contemplation. This heightened and enhanced emotional, spiritual and psychological wellbeing. This was primarily attained in quiet naturalised settings where the participants felt free to disconnect from the chaos of modern living and psychologically and emotionally unwind. A possible explanation for this relates to Kaplan's (1995) 'Attention Restoration Theory'. Kaplan (1995) argued that natural environments provide opportunities to distance oneself from the stress of routine living. Thus it would appear that being in the presence of nature "Leads to a contemplative state of mind, where people can attain a state of calm or balance" (Townsend and Weerasuriya, 2010:12), thus facilitating emotional, spiritual and psychological wellbeing.

The notion of 'therapeutic landscapes' documented by social geographers such as Ballas and Dorling (2007a, 2007b, 2013) may also help to understand why some participants reported that feelings of stress and depression were assuaged through immersion in the natural environment. Stress Reduction Theory, which proposes that natural environments promote recovery from stress (Konijnendijk, 2008a; Velarde et al., 2007), offers an alternative explanation for this finding.

In addition to relieving stress, the findings indicated that immersion in the natural environment facilitated contemplation and reminiscences of happy memories, which enhanced emotional and psychological wellbeing. This finding can be understood within the context of studies such as Sempik et al. (2002), Henwood (2003) and Wells et al. (2010). These studies reported that interacting with nature leads to improved mood and psychological wellbeing.

Additionally, some participants indicated that the natural environment also held intrinsic value. Time spent in the natural environment frequently propagated a sense of harmony with the wider world, which generated a holistic sense of wellbeing. This finding is comparable to the findings of the existing theoretical literature, which argues that wellbeing is enhanced when humans recognise and appreciate their inherent relationship with nature (Haworth and Hart, 2007; Naess, 1973; Newton, 2007; Pickering, 2007; Pretty et al., 2005b; Wells et al., 2010; Wilson, 1984).

One could argue that this finding reflects the theoretical position established in Wilson's 'Biophilia' Hypothesis' (1984), which argues that humans have an innate emotional attachment to the natural environment. Wilson (1984) proposed that at a fundamental level, human identity and self-actualisation rely on our relationship with the natural environment. Taken together, these findings suggest that valuing the natural environment needs to be incorporated into existing debates regarding how best to enhance psychological, emotional, social and community wellbeing.

#### 7.7: The role of temporality on wellbeing

Until recently, little qualitative empirical research had explored the role of temporality on wellbeing (Durayappah, 2010; Freund, 2010; Guell et al., 2014). However, as outlined in the literature review, this is beginning to change. Recent examples include El-Shaarawi (2015), Schwanen et al. (2012), Galvin and Todres (2013), Guell et al. (2014), Åström et al. (2014), Zajenkowska et al. (2014), Griva et

al. (2015) and Wunderlich (2016). However, these studies largely focus on exploring associations between temporal perspectives and particular behaviours and/or an individual's orientations to a particular temporal dimension (Aström et al., 2014; Guell et al., 2014). The findings from this study therefore offer new insights into the role of temporality within the context of lay conceptualisations of personal wellbeing. These include the role of affective forecasting and mechanisms for establishing temporal perspectives.

Throughout this thesis, the notion of temporality and temporal perspectives emerged as a theme underlying the participants' narratives of personal wellbeing. One could argue that this suggests "An awareness that their own histories are germane to their present personal wellbeing" (Rubinstein, 1989:109). Interpretation of the findings suggest that for the individuals in this study, temporal perspectives are not confined to an understanding of time within a linear organisation of continuity in a chronological sense. The temporal perspectives appeared to influence how the participants perceived and related to past, current and future levels of personal wellbeing. Future temporal perspectives appeared particularly influential in how the participants perceive present levels of personal wellbeing.

Affective forecasting (i.e. predicting future feelings of events such as holidays or the birth of a grandchild) was frequently used to augment present feelings of wellbeing. This can be understood within the context of Gilbert and Wilson's (2007) theoretical premise that we pre-experience the future by envisaging it in our contemplations and use this cognitive process to predict future feelings of the event (affective forecasting). The findings suggest that affective forecasting was an important conduit for enhancing the participants' present levels of personal wellbeing.

The study findings indicated that early childhood experiences are important mechanisms for establishing temporal perspectives. This supports Åström et al.'s (2014) theoretical claim that "Our temporal frames of reference are shaped by our personal histories and experiences" (p.4). Early childhood experiences appeared to

regulate how participants coordinate and integrate the three temporal states. In addition to which, early childhood experiences appeared to help the shape participants' sense of time and its importance within the context of their daily lives. This finding adds empirical support to the research of Henwood (2001). Henwood (2001) reported that early childhood associations, particularly in relation to experiences in natural environments, have a particular value in triggering happy memories and promoting positive feelings of wellbeing in times of stress or anxiety.

Additionally, a considerable number of participants reported storing up feelings of positive wellbeing, which could be revisited in the future, during periods of negative wellbeing. The participants also frequently recounted spending time reflecting on past happy memories to enhance current levels of personal wellbeing. This fluidity between temporality and wellbeing suggests that wellbeing should be envisaged as temporally contingent. Envisaging wellbeing as temporally contingent supports the recent work of Galvin and Todres (2013), who explored the associations between ageing, temporality and wellbeing (Galvin and Todres, 2013). Galvin and Todres (2013) call for a humanising approach to healthcare services which recognises and accommodates amongst other things the role of temporality in the wellbeing of older people.

Taken together, these findings add evidential support to studies such as El-Shaarawi (2015), Guell et al. (2014) and Griva et al. (2015). They also contribute to the emergent research such as Wunderlich (2016), Schwanen et al. (2012) and Schwanen and Wang (2014), which explores the intersections between temporality and wellbeing. This study suggests the role of temporality, and in particular that temporal perspectives need to be incorporated into existing discussions regarding how best to enhance wellbeing in older adults with LTCs.

### 7.8: Limitations

Although this study revealed some interesting findings, it has several limitations. It is important to acknowledge that whilst qualitative research affords detailed and indepth understandings of lived experiences, there are limitations inherent in

qualitative research. The narratives in this thesis may have under or over represented the importance of particular themes. One must be cognisant of people's propensity to only acknowledge those aspects of wellbeing they are willing to declare in public, and to perhaps ignore those which they want to keep private (Dolan and White, 2007). It is also worth remembering that the data in this study relate to those who agreed to participate in the research; this should be borne in mind.

This research may have been limited by participant bias and memory distortion. Previous research suggests retrospective self-reports are vulnerable to bias and distortion of memory (Blore, 2008; Oishi, 2002). This is related to the overestimation effect (Wirtz et al., 2003) and may have distorted the findings, particularly in relation to the intersections between temporality and wellbeing. Durayappah (2010) noted that if an individual is in a positive mood they are more likely to think about positive aspects of their life and offer a positive evaluation of their life. This may have influenced the study findings.

There is also a danger that research can become self-fulfilling in nature, that is, researchers may be prone to find results that confirm their pre-existing expectations. For example, the nature of semi-structured interviews means the researcher has some input into the material highlighted for discussion. It is apposite to declare the researcher, Alison and Jim shared similar political affiliations and working-class sense of identity. These shared ideological principles may have led to bias in this study. Efforts were made to mitigate this by ongoing reflexive practice, discussions between the researcher, her supervisory team and the co-researchers. However, the extent to which this may have influenced findings can only be surmised, although it is worthy of reflection and deliberation. It must also be noted that the relationship between the researcher and co-researchers may have had an undue influence on the study finding. The extent to which this may have influenced findings can only be surmised, but it is worthy of reflection and deliberation.

Finally, the sample size for this study was small and selected from within one geographic area, and therefore may not reflect the typical experiences of older adults with LTCs. However, qualitative research tends to have a local focus and

emphasises the specifics of experiences rather than generalised understandings (Bartova, 2014, unpublished PhD thesis; Cresswell et al., 2007; Davies and Dodd, 2002).

### 7.9: Strengths

Despite the limitations, this study offers a number of strengths. First, it makes an original contribution by investigating wellbeing through the prism of historical/linguistic contextualisation. This provides insights into the relationship between conceptualisations of wellbeing and historical/linguistic factors identified as under-researched (Inglehart and Klingeman, 1998; Kingfisher, 2013; Sirgy and Estes, In Press).

Researcher reflexivity and co-researcher reflections contribute an additional layer of contextualisation in this thesis. In addition to which, the co-researchers were immersed in much of the research process. Our personal reflections and shared dialogues contributed to aspects of the research process associated with participatory research. The co-researchers' reflective diaries and the researcher's reflexive journal provided a source of rich, in-depth and contextual data.

There is increasing recognition of the value of lived experience and the important role this aspect contributes to research (Bartova, 2014; Basset et al., 2010; Seaford, 2011). This study explored the lived experiences of lay people within a historical and linguistic context and helps to give voice to lay narratives which are underrepresented in wellbeing research (Clark and Gough, 2005; Dodge, 2012; Scott, 2012c).

It has been widely recognised that the field of wellbeing research is so vast it is beyond the capabilities of a lone researcher to conduct an extensive literature review (Gasper, 2010; Langlois and Anderson, 2002; Sarvimaki, 2006; Scott, 2012c). Nevertheless, a comprehensive literature review was undertaken and identified gaps in knowledge, which this study has contributed to. This comprehensive review incorporated a components' approach to explore in substantial detail domains of

wellbeing. It also utilised a chronological approach, documenting the trajectory of wellbeing research since the Local Government Act (2000). This two-pronged approach provides a broad and holistic overview of the development of wellbeing research.

# **7.10: Summary**

This study has contributed to a clearer theoretical and empirical understanding of an association between personal responsibility and wellbeing. Feelings of powerlessness and lack of agency may influence perceptions of responsibility for wellbeing. Those familiar with the term 'wellbeing' through contact with healthcare/social care professionals were most likely to feel personal responsibility for physical wellbeing. This suggests that the language used by professionals may be important. Conceptually, wellbeing was most frequently associated with the language of professionals, media and state. This may indicate that the historical/linguistic legacy of wellbeing as a linguistic indicator of education and social positioning (Hughes, 1988) continues to influence current conceptualisations of wellbeing.

The study findings suggest that subjective experiences of wellbeing in older adults cannot be understood in isolation from the structural realities, which constrict their ability to assume personal responsibility for aspects of wellbeing beyond their control. Conceptually, these findings reinforce the need to look beyond agentic representations of wellbeing. Lay conceptualisations of wellbeing appeared to incorporate recent historical associations between wellbeing and welfare, which focused on structural representations of wellbeing. This has important implications for policy and the way in which wellbeing is researched.

There is a relationship between the way wellbeing is represented in legislature and discursive policies and the way in which it is researched (Schwanen and Ziegler, 2011; Nordbakke and Schwanen, 2014). There is an onus therefore on researchers and policy makers to offer different ways of understanding and promoting wellbeing and to acknowledge the limitations of agentic representations of

wellbeing. Attempts to represent and define wellbeing, which embody normative values associated with autonomy, independence and agency, may be positively detrimental to older people's wellbeing (Taylor, 2011).

Wellbeing was shown to have been mobilised within a PRA as a response to an ageing population, increasing numbers of people with preventable LTCs and the demands this places on health and social care resources. As a result, wellbeing has become discursively linked to policy objectives related to individual responsibility and self-governance. This alternative theoretical interpretation of the role of wellbeing and its civic functions should help to expand our appreciation of the perceived nature, meaning and purpose of wellbeing in civic society.

### **Chapter 8: Conclusion**

This chapter begins by considering the extent to which the aims and objectives of the study have been met and synthesising the most notable findings. This is followed by a summary of the study's contribution to knowledge. The concluding paragraphs will highlight recommendations for future research, implications for practice before ending with a reflective comment.

# 8.1: A consideration of whether the study aims have been addressed

- 1 The theoretical aim of this thesis was to explicate understandings of wellbeing and its association with personal responsibility. The findings established strong evidence to support theoretical claims that wellbeing has been mobilised within a PRA for governance purposes. The historical/linguistic contextualisation indicated that latent historical/linguistic associations between wellbeing and personal responsibility have facilitated this agenda. Furthermore, the historical/linguistic legacy of wellbeing continues to influence current conceptualisations of wellbeing.
- 2 The empirical aim of this thesis was to investigate and synthesise the participants' conceptualisations of wellbeing. The study identified a disjunction between the participant and professional conceptualisations of wellbeing, particularly in relation to the perceived responsibility for wellbeing. The participants conceptualised wellbeing primarily within the parameters of state responsibility. This finding contributes to the debate about the division of responsibility for wellbeing between the individual and state.
- 3 The overarching aim of this thesis was the synthesis of the theoretical and the empirical evidence. Whilst there is compelling theoretical evidence of an association between wellbeing and personal responsibility, the empirical evidence demonstrated that participants rarely acknowledged any personal responsibility for wellbeing. Where this was apparent, personal responsibility was identified in relation to those aspects of personal wellbeing most widely promoted within the

neo-liberal discourse. This concurrence may be indicative of the partial success of neoliberal efforts to mobilise wellbeing to promote a PRA.

### 8.2: Notable findings

The study revealed that:

- Wellbeing was not commonly associated with personal responsibility.
   Domains such as environmental wellbeing were commonly conceptualised as the responsibility of the state and beyond the control of average citizens.
- Feelings of powerlessness and lack of agency may influence perceptions of responsibility for wellbeing.
- Wellbeing was not considered a meaningful concept by participants.
- Those familiar with the term wellbeing, through contact with healthcare/social care professionals, were most likely to feel personal responsibility for physical wellbeing. This suggests language used by professionals may be important in apportioning responsibility for wellbeing.

Conceptualisations of personal wellbeing were coalesced into seven thematic domains: health, the ageing process, personal identity, social interaction, the built and natural environments and temporality. The study revealed some notable findings regarding how older adults with LTCs conceptualise personal wellbeing:

- Conceptualisations of wellbeing were considered highly subjective and situational. Efforts to operationalise it were regarded as problematic and potentially divisive.
- Wellbeing was rarely conceptualised in agentic terms, on the contrary it was widely conceptualised as the responsibility of the state.
- Autonomy and independence of personal value were generally considered redundant in personalised conceptualisations of wellbeing.
- Temporality played an important role in conceptualisations of long and short-term wellbeing. Childhood experiences were important mechanisms for establishing temporal perspectives.

The natural environment has a predominantly positive impact on wellbeing,
 particularly psychological, emotional, and social and community wellbeing.

The study revealed some notable findings about whether wellbeing is considered a meaningful concept for older adults with LTCs, in particular:

- Unfamiliarity with the concept and its functional modus operandi led most to avoid using it favour of more widely recognised terms such as QoL and welfare.
- Wellbeing was not associated with the language used amongst peers.
- Wellbeing was considered conceptually complex and associated with the language of professionals, the media and the state. This suggests that wellbeing is a linguistic indicator of education and social positioning.

### 8.3: Policy recommendations

Within the complex data produced through the study, a number of policy recommendations emerged as warranting further investigation. These touch on different facets of wellbeing. Health and wellbeing boards may wish to address these as part fulfilment of their wellbeing duties, established in the Local Government Act (2000) and the amended Act (2007).

Since the worldwide recession, there is evidence to suggest that the relationship between wellbeing and housing conditions is often predicated on pecuniary resources (Gibb, 2014). For an increasing number of people, however, affordable well-maintained housing is beyond their budget. In addition to which, falling levels of housing stock has meant that housing costs constitute an increasingly large proportion of household income (Black et al., 2015; Gibb et al., 2013). Whilst "Measuring housing conditions and their effects on people's well-being is a complex task" (OECD 2011:82), future policies may wish to focus on how falling levels of housing stock and associated issues have impacted on wellbeing.

Temporality played an important role in conceptualisations of long and short-term wellbeing. This has important implications for those who commission or regulate local wellbeing services. Health and wellbeing boards may wish to explore how aspects of temporality such as temporal perspectives, attitudes and preferences enhance or undermine long and short-term wellbeing. This may help to explicate theoretical notions of temporality and its intersections with wellbeing, as recommended by Guell et al. (2014) and Griva et al. (2015).

Naumova (2014) sounded a cautionary note concerning the disproportionate emphasis on personal responsibility employed within health and wellbeing services. We need a clearer understanding of the role of structural and agentic factors on wellbeing (Bache and Reardon, 2013; Seaford, 2011; Scott, 2012a; Scott and Bell 2013). Those responsible for wellbeing services need to acknowledge there is a valid and mounting debate about the mobilisation of agentic representations of wellbeing within a wider PRA. Future policies should endeavour to align wellbeing measures with the Social Care Act (2014), which emphasised the role of governance in ensuring that wider structural determinants of wellbeing are addressed.

There was little evidence to suggest that participants associated wellbeing with personal responsibility. Indeed, wellbeing appeared to be more closely aligned with welfare provision and was primarily considered the responsibility of the state. This suggests that future wellbeing policies should acknowledge the role of social and institutional responsibility alongside that of individual responsibility, as recommended by Seaford (2011) and Scott (2012a, 2012c).

## 8.4: Implications for practice

Poor psychological wellbeing, frequently undermined by feelings of anxiety and depression, was particularly detrimental for personal wellbeing. Limited access to dwindling psychological services were identified as particularly detrimental. The participants reported that improving psychological wellbeing was unlikely to be achieved unless service provision provided high intensity and long-term interventions. These findings are likely to bring important implications for practice,

as evidence suggest that at any one time one in six people in the UK suffer from mental health issues (ONS, 2014). Increasing provision of psychological services for wellbeing has ostensibly been addressed by the UK government's widespread adoption of Layard's (2005) proposal to increase access to cognitive behavioural therapy. However, this has been challenged as an approach based on cost—benefit analysis rather than a well-evidenced approach to the provision of psychological services (Taylor, 2011, 2015; Wright, 2014). Similarly, new measures such as 'mindfulness' promoted within psychological services have been criticised for failing to address more serious mental health issues or psycho-social aspects of wellbeing (Scott, 2012c; Wright, 2014).

This suggests local government services should prioritise attention to psychological wellbeing. This could focus on increasing access to a broad range of wellbeing services such as long and short-term psychological services and psychosocial therapies (Taylor, 2011, 2015).

This study also has practical implications for the professional treatment of those who struggle with psychosocial aspects of mental health such as isolation, loneliness and depression. These exert a considerable impact on personal wellbeing. The findings suggest that the provision of services should prioritise attention to wellbeing alongside practical and health-related support. In particular, consideration should be given to enhancing wellbeing in terms of social network development to reduce the psychological and physical health complications which may arise from inactive and lonely lifestyles. Local government services can better reflect the wellbeing needs of older adults with LTCs by creating opportunities for formal and informal social interaction. This could be particularly effective if undertaken in the natural environment, which induces a calming effect and facilitates opportunities for intergenerational and cross-cultural contact. Situating social interaction within the natural environment has an additional advantage of valuing natural habitats, considered beneficial for social and community wellbeing.

Fear for personal safety restricted access to public green spaces and undermined the largely positive impact the natural environment had on personal wellbeing.

Public spaces need to be well maintained, informally policed, promoted as a community resource and networked to other resources to restrict territorial claims and reduce fears for personal safety. The findings suggest strategic policies, which include a range of measures to enhance perceptions of safety, may encourage greater levels of physical engagement and activity in outdoor spaces, concomitantly enhancing physical, psychological, social and community wellbeing.

This study provides clear and grounded evidence that loneliness and feelings of social isolation contribute significantly to the negative wellbeing of older people. Joint commissioning strategies based on joint strategic needs assessments may increasingly need to support multi-disciplinary services which address issues of loneliness, before they result in isolation and depression.

In the main, wellbeing was not considered a meaningful concept by those who took part in this study. This was due in part to low recognition rates, poor applicability in wider society and its perception as a professional term. If wellbeing has little lay currency, despite its inclusion in extended areas of policy and professional practice, then it may have little intrinsic worth, particularly as an outcome measure. This suggests that those responsible for commissioning wellbeing services may wish to reconsider wellbeing as an outcome measure for service user satisfaction.

### 8.5: Research recommendations

This thesis was a small scale, qualitative study that focused on older adults with LTCs, and therefore may not be widely generalisable. However, a number of areas worthy of further research were identified.

Biographical disruption and the interaction between cognitive impairments, depression and social isolation exerted a significant impact on personal wellbeing. These findings raise an important question concerning how these interactions manifest, and further highlight the importance of longitudinal studies in order to explicate understandings.

Negative associations between the natural environment and wellbeing were caused primarily because of fears for personal safety. Further research is required to establish whether there are causal relationships between certain aspects of the natural environment and specific wellbeing domains.

Temporality played an important role in conceptualisations of wellbeing. Research, which is attuned to temporality and its impact on an individual's subjective state, may help us to understand an individual's wellbeing in greater depth. Further research is required, for example, to discern how temporal perspectives interact with an individual's sense of wellbeing.

Agentic and structural representations of wellbeing continue to exert a significant influence on conceptualisations of wellbeing. The combined influence of structural and individual factors on personal wellbeing needs to be better understood, and investigated further.

### 8.6: Contributions to knowledge

- 1. This is the first UK study to contextualise current understandings of wellbeing through a historical/linguistic prism. This study documented a long established historical/linguistic association between wellbeing and personal responsibility, which continues to influence modern conceptualisations of wellbeing.
- 2. This study explored people's lived experiences of wellbeing, contextualised by the historical/linguistic prism. This offers an appreciation of two factors rarely addressed in current wellbeing research.
- 3. This study has contributed to a clearer theoretical and empirical understanding of an association between personal responsibility and wellbeing.
- 4. This research has elucidated some adumbrated aspects of personal wellbeing, which were under-researched at the onset of this thesis. In particular:
  - New insights into the role of temporality and temporal perspectives within the context of lay conceptualisations of personal wellbeing.

- ii. New insights into aspects of the natural environment and their impact on wellbeing.
- iii. An appreciation of the extent to which coping mechanisms and adaptation are used to offset the impact of ageing and health issues on wellbeing.
- iv. An appreciation of lay perceptions of the links between the built environment and wellbeing.
- v. An appreciation of lay perceptions of the links between social and personal wellbeing.
- 5. This thesis has contributed to the theoretical debate regarding conceptual differences between wellbeing and QoL. Whilst QoL and wellbeing may allude to similar aspects of the human condition, they differ in respect to linguistic reference, linguistic register, semantic construction and historical legacy.
- 6. Operationalised definitions should balance agentic representations of wellbeing, recognising the importance of structural factors and incorporating notions of state responsibility alongside notions of personal responsibility.
- 7. Definitions of wellbeing should not be thought of as being immutable or universally applicable but as historically, linguistically, culturally and temporally contingent.
- 8. Methodologically, the combination of a theoretical, linguistic and a qualitative study enabled the conceptual complexity of wellbeing to be explored.

### 8.7: Concluding comments

This study documented strong theoretical evidence of the mobilisation of wellbeing within a PRA. Historical/linguistic analyses indicated that latent associations between wellbeing and personal responsibility have facilitated this mobilisation. The empirical analysis suggested that recent historical/linguistic associations between wellbeing and welfare have influenced lay conceptualisations of wellbeing. These may have undermined government efforts to establish a sense of personal responsibility for wellbeing, at least for those involved in this study. The majority of

participants in this study identified the state/statutory agencies as primarily responsible for key domains of wellbeing such as psychological, economic, social and environmental wellbeing. Only a minority of those involved in this study felt personal responsibility for wellbeing. This was primarily for aspects of wellbeing pertaining to physical health, personal relationships and the community. Much of the discursive representations and conceptualisations of wellbeing advocated by professional/academic 'expert voices' were not reflected in the language of participants. This study therefore suggests there may be a disjunction between lay and professional conceptualisations of who may be considered responsible for wellbeing.

Wellbeing is a continually evolving socially constructed concept. However, it retains former inherent meanings which are embedded within the term. It is hard for us today to think about wellbeing without our modern frame of reference. In this modern era, we compartmentalise it into terms such as 'physical wellbeing' or 'mental wellbeing'. As a society which does not think primarily in holistic terms, we fail to see wellbeing in holistic terms.

We can, however, bequeath more holistic and less human-centred conceptualisations of wellbeing to future generations. These conceptualisations may draw upon existing historical and linguistic notions. Simultaneously, they may employ visionary, cultural and historical foresight to promote representations of wellbeing, which are future facing and incorporate notions of sustainability, ecohumanism and cultural heterogeneity.

Representations of wellbeing may be better served by alluding to a greater sense of the collective than that which it currently enjoys under the neo-liberal platform of individualism and the primacy of humans. The conceptual clarifications and definitions which we operationalise today, may lay the foundations upon which future literary luminaries create works we are not able to conceive. Perhaps it is also time to revisit notions which were formerly inherent or implicitly understood as integral to the concept, such as mutual interdependence and reliance on the natural world for survival.

#### References

Abbott, J., Hart, A., Morton, A., Gee, L., Conway, S. (2008) "Health-related quality of life in adults with cystic fibrosis: The role of coping". *Journal of Psychosomatic Research*, 64:149–157.

Abbot, P., and Wallace, C. (2012) "Social quality: a way to measure the quality of society". *Social Indicators Research*, 108(1) 153-167.

Abdallah, S., Thompson, S. and Marks, N. (2008) "Estimating worldwide life satisfaction". *Ecological Economics*, 65, 35–47.

Abdallah, S. and Johnson, V. (2008) "Sustainability and well-being: Connects and disconnects". In Proceedings of the National Transport Conference, London.

Abdallah, S. (2012). "Well-being patterns uncovered" London: New Economics Foundation.

Abdallah, S., Michaelson, J., Seaford, C., and Stoll, L. (2011) "Measuring our progress. The power of wellbeing". London: New Economics Foundation. Available at: http://www.neweconomics.org/publications/P10 [Accessed:26/7/2012].

Abdallah, S., Michaelson, J., and Stoll, L. (2012) "Happy Planet Index: 2012 Report". London: New Economics Foundation.

Abdallah, S., Michaelson, J., and Stoll, L. (2013) "Quality of life in Europe: Subjective wellbeing". Brussels: Eurofound.

Abebe, A., Flicker, S., George, P., Shimeles, H., Yassin, A. (2011) "Community based participatory research. Available at:

http://avnu.ca/avnu/uploads/2014/04/GOAL\_CommunityBasedParticipatoryResear ch.pdf [Accessed:12/9/15].

Abouleish, E., J. (1979) "Islamic contributions to Medicine". *Islamic Medical Association* 3 (4), 28-45,

Abraham, A., Sommerhalder, K. and Abel, T. (2010) "Landscape and well-being: a scoping study on the health-promoting impact of outdoor environments". *International Journal of Public Health*, 55, 59–69.

Abramovitz, M., and Zelnick, J. (2010) "Double jeopardy: The impact of neoliberalism on care workers in the United States and South Africa". *International Journal of Health Services*, 40 (1), 97-117.

Abramson, C., M. (2003) "Qualitative Research in the Positivist-Behavioral Tradition Resources for Addressing Type I and Type II Errors in Code Associations Using

ATLAS.ti". Available at:

http://www.academia.edu/1149421/Qualitative\_Research\_in\_the\_Positivist-Behavioral\_Tradition\_Resources\_for\_Addressing\_Type\_I\_and\_Type\_II\_Errors\_in\_Code\_Associations\_Using\_ATLAS.ti [Accessed:10/9/11].

Abrahamson, J., A., Rubin, V., L. (2012) "Discourse structure differences in lay and professional health communication". *Journal of Documentation*, 68 (6), 826 – 851.

Acemoglu, D., Johnson, S., Robinson, J.A. and Yared, P. (2008). "Income and Democracy". *American Economic Review*, 98(3), 808–842.

Acheson, D. (1998) "Independent Inquiry into Inequalities and Health". London: The Stationery Office.

Adams K., B., Leibbrandt, S., and Moon, H. (2011) "A Critical Review of the Literature on Social and Leisure Activity and Wellbeing in Later Life". *Ageing and Society* 31, 683–712.

Adams, L., Amos, M., and Munro, J. (eds) (2002) "Promoting Health: Politics and Practice". London: Sage.

Age Concern and National Consumer Council (2006)"As fit as butchers dogs: A report on healthy lifestyle choice and older people". London: Age Concern Reports Available at: http://www.ageconcern.org.uk/AgeConcern/ [Accessed:2/7/12].

Agnew, J. (1998) "Geopolitics: Re-envisaging world politics". New York: Routledge.

Ahmed, S. (2010) "The promise of Happiness". London: Duke University Press.

Ahmed, S., Berzon, R., A., Revicki, D., A., Lenderking, W., R., Moinpour, C., M., Basch, E., et al. (2012) "The Use of Patient-reported Outcomes (PRO) Within Comparative Effectiveness Research: Implications for Clinical Practice and Health Care Policy". *Medical Care*, 48 (1).

Ajzen, I., and Fishbein, M. (1980) "Understanding Attitudes and predicting Social Behavior". New Jersey: Prentice-Hall Inc.

Ajzen, I. (2005) "Attitudes, Personality and Behavior". Maidenhead: Open University Press.

Aked, J., Michaelson, J., and Steuer, N. (2010) "The role of local government in promoting wellbeing". London: New Economics Foundation. Available at: http://www.nmhdu.org.uk/silo/files/the-role-of-local-government-in-promoting-wellbeing.pdf [Accessed:8/7/2008]

Aked, J., Marks, N., Cordon, C., and Thompson, S. (2008). "Five Ways to Wellbeing: The evidence". London: New Economics Foundation.

Aknin, L., B., Barrington-Leigh, C., P., Dunn, E., W., Helliwell, J., F., Biswas-Diener, R., Kemeza, I., Nyende, P., Ashton-James, C., E., and Norton M., I. (2010). "Prosocial spending and well-being: cross-cultural evidence for a psychological universal". NBER Working Paper 16415. Cambridge: National Bureau of Economic Research.

Albrecht, G., Fitzpatrick, R., and Scrimshaw, S., C. (eds) (2009) "The Handbook of Social Studies in Health and Medicine". London: Sage.

Alcock, I., Wheeler, B., Depledge, M. (2013). "Green Spaces May Boost Well-being for City Slickers". Association for Psychological Science, Press Release April 22, 2013. Available at:

https://www.psychologicalscience.org/index.php/news/releases/green-spaces-may-boost-well-being-for-city-slickers.html [Accessed:14/09/14].

Alcock, I., White, M.P., Wheeler, B.W., Fleming, L., Depledge, M.H. (2014) "Longitudinal Effects on Mental Health of Moving to Greener and Less Green Urban Areas". *Environmental Science and Technology*. 48 (2), 1247-1255.

Alexander, J., M. (2008) "Capabilities and Social Justice". London: Ashgate Publishing Ltd.

Alexander, J., M. Hughes, J., and Hughes, K. (2014) "Towards a Developmental understanding of happiness". *Sociological Research Online*, 19 (2), 2.

Alexopoulos, G., S. (2005) "Depression in the elderly". Lancet, 365(9475), 1961–70.

Alkire, S. (2007) "Why the Capability Approach?". Journal of Human Development, 6 (1), 20 -25.

Alkire, S. (2005) "Valuing Freedoms: Sen's Capability Approach and Poverty Reduction". Oxford: Oxford University Press.

Alkire, S. (2013) "Wellbeing, happiness and Public Policy". Oxford Poverty and Human Development Initiative (OPHI) Research in Progress series 2013.

Allen, J. (2008) "Older adults and Wellbeing". London: Institute for Public Policy Research.

Allin, P. (2007) "Measuring societal wellbeing". *Economic and Labour Market Review*, 1(10), 46–52.

Allport, G., W. (1954) "The historical background of modern social psychology". In G. Lindzey (ed) 'Handbook of Social Psychology', Reading, MA: Addison-Wesley.

Altman, I. (1975) "The environment and social behaviour". Monterey, California: Brooks/Cole.

Ananta K.,G. (2000) "Rethinking Human Well-being: A Dialogue with Amartya Sen." *Journal of International Development* 12 (7), 1003-1018.

Anderson, M. (1996) "Population change in north-western Europe, 1750-1850." In M. Anderson, (ed) 'British Population History: from the Black Death to the present day'. Cambridge: Cambridge University Press.

Anderson, P., Jané-Llopis, E., Cooper, C. (2011). "The imperative of well-being". *Stress and Health*, 1-3.

Andrew E. Clark, Paul Frijters, and Michael A. Shields (2008). "Relative Income, Happiness, and Utility: An Explanation for the Easterlin Paradox and Other Puzzles," *Journal of Economic Literature*, 46(1), 95-144.

Andrews, F., M., and Withey, S., B. (1976) "Social Indicators of Well-being: Americans' Perceptions of Life Quality". New York: Plenum Press.

Andrews, T. (2012) "What is Social Constructionism"? Grounded theory and International Journal. 1 (1), 12-20.

Andrews, F., M., and Withey, S., B. (1976) "Social Indicators of Well-being: Americans' Perceptions of Life Quality". New York: Plenum. Available at: https://lemosandcrane.co.uk/resources/European%20Forum%20on%20Population %20Ageing%20Research%20-%20Models%20of%20Quality%20of%20Life.pdf [Accessed:12/9/15].

Andrews, M; Sclater, S; Squire, C., and Treacher, A. (2000) "Lines of Narrative' Psychosocial Perspectives". London: Routledge.

Antonovsky, A. (1996) "The salutogenic model as theory to guide health promotion". *Health Promotion International*, 11, 11-18.

Appleyard, D. and Lintell, M. (1982) "The environmental quality of city streets: The residents' viewpoint". In S., Kaplan and R. Kaplan (eds) 'Humanscape: Environments for people' Ann Arbor, USA: Ulrich's Books.

Argyle, M. (2002) (2<sup>nd</sup> edn) "The Psychology of Happiness". London: Routledge.

Argyle, M. (1999) "Causes and correlates of happiness". In: D., Kahneman, E., Diener, N. Schwarz (eds) 'Well-being: the foundations of hedonic psychology'. New York: Russell Sage Foundation.

Armstrong, D., and Caldwell, D. (2004) "Origins of the Concept of Quality of Life in Health Care: A Rhetorical Solution to a Political Problem". *Social Theory and Health*, 2 (4), 361-371.

Arnstein S. R. (1969) "A ladder of citizen participation". Journal of the American Planning Institute, 35, 216-224.

Aronson, J., A. (1994) "Pragmatic view of thematic analysis". *The Qualitative Report,* 2 (1), Spring, Available at: http://www.nova.edu/ssss/QR/BackIssues/QR2-1/aronson.html [Accessed:26/1/2011].

Arts, W., and Gelissen, J. (2002) "Three worlds of welfare capitalism or more? A state-of-the-art report". *Journal of European Social Policy*, 12, 137-149.

Ashforth, B., and Mael, F. (1989) "Social identity and the organization". *Academy of Management Review*, 14 (1), 20-39.

Ashton J., and Seymour, H. (1989) "The New Public Health". Buckingham: Open University Press.

Åström, E., Wiberg, B., Sircova, A., Wiberg, M., and Carelli, M. G. (2014) "Insights into features of anxiety through multiple aspects of psychological time". *Journal of Integrative Psychology and Therapeutics*, 2 (1), 3. Available at: http://www.hoajonline.com/psychology/2054-4723/2/3. [Accessed:4/1/16].

Atkinson, S. (2011) "Care is needed in moves to measure wellbeing". *BMJ*, 343:1051.

Atkinson, S., Joyce, K., E. (2011) "The place and practices of wellbeing in local governance". *Environmental Planning*, 29, 133–148.

Atkinson, S., Fuller, S., Painter, J. (2012) "Wellbeing and place". In S., Atkinson, S,. Fuller and J. Painter (eds) (2012) 'Wellbeing and place'. Ashgate, Farnham

Atkinson, S. (2013). Beyond Components of Wellbeing: The Effects of Relational and Situated Assemblage. Linguistic Landscapes Research Report no. DCSF-RW073. Available at:

 $http://download.springer.com/static/pdf/779/art%253A10.1007\%252Fs11245-013-9164-0.pdf?auth66=1426868321\_b48e9561caf47fb8e8c472f623fee1b0andext=.pdf [Accessed:12/03/15].$ 

Attride-Stirling, J. (2001) "Thematic networks: an analytic tool for qualitative research". Qualitative Research, 1 (3), 385-405.

Auld, G., Gulbrandsen, L, and McDermott, C. (2008) "Certification Schemes and the Impacts on Forests and Forestry." *Annual Review of Environment and Resources*, 33, 187–211.

Avner, O. (ed) "In Pursuit of the Quality of Life". Oxford: Oxford University Press.

Ayer, A. (2011) "Why Things Matter to People: Social Science, Values and Ethical Life". Cambridge: Cambridge University Press.

Babcock-Gove, P. (ed) (1971) (3<sup>rd</sup>edn) "Websters New International Dictionary" vol 3. Chicago: Encyclopaedia Britannica.

Bache, I., and Reardon, L. (2013) "An idea Whose Time has come? Explaining the Rise of Well-Being in British Politics". *Political Studies*, 61(4), 898-914

Bache, I., Reardon, L., and Anand, P. (2015) "Wellbeing as a wicked problem: navigating the arguments for the role of government". *Journal of Happiness Studies*, 17 (3), 893-912.

Baggott, R. (2000) "Public Health, Policy and Politics". Basingstoke: MacMillan Press.

Baker, L. A., Cahalin, L. P., Gerst, K., and Burr, J. A. (2005) "Productive activities and subjective well-being among older adults: The influence of number of activities and time commitment". *Social Indicators Research*, 73, 431–458.

Ball, M., M., Perkins, M., M., Whittingham, F., J., Hollingworth, C.(2004) "Independence in assisted living". *Journal of Aging studies* 18 (4), 476-483.

Ballas, D., Dorling, D., and Shaw, M. (2007) "Social inequality, health, and well-being". In J., Haworth and G. Hart, (eds) 'Well-Being: Individual, Community, and Social Perspectives'. Basingstoke: Palgrave.

Ballas, D. and Dorling, D. (2007b) "Measuring the impact of major life events upon happiness". *International Journal of Epidemiology*, 36, 1244-1252.

Ballas, D. (2010) "Geographical simulation models of happiness and well-being". In J., Stillwell, P., Norman, C., Thomas, and P. Surridge, (eds) 'Spatial and Social Disparities Understanding Population Trends and Processes'. Cambridge: Cambridge University Press.

Ballas, D., and Tranmer, M. (2012) "Happy people or happy places? A multi-level modelling approach to the analysis of happiness and well-being". *International Regional Science Review*, 35(1), 70–102.

Ballas, D., and Dorling, D. (2013), "The Geography of Happiness". In D., Boniwell, S., Conley, and A. Ayers (eds) 'The Oxford Handbook of Happiness'. Oxford University Press.

Baltatescu, S. (2007) "Happiness in the Social Context of Post-Communist Transition in Romania". Unpublished PhD. University of Bucharest.

Baltatescu, S. (2007b) "Towards a sociological theory of subjective well-being" Excerpt from the unpublished thesis "Happiness in the social context of post-communist transition in Romania". Translated by S. Baltatescu, (2007). University of

Bucharest. Available at: file:///C:/Users/Wayne/Downloads/SSRN-id2196346.pdf [Accessed: 12/03/15]

Bandura, A. (2001) "Social Cognitive Theory: An Agentic Perspective". *Annual Review of Psychology*, 52:1-26.

Banks, S., Armstrong, A., Carter, M. (2013) "Everyday ethics in community-based participatory research". *Contemporary Social Science*, 8 (3), 263-277.

Bardsley, M., Steventon, A., Smith, J., Dixon, J. (2013) "Evaluating Integrated and Community-Based Care: How do we know what works"? London: Nuffield Trust.

Barresi, C., M., Ferraro, K., F., and Hobey, L., L. (1984) "Environmental Satisfaction, Sociability, and Well-being among the Urban Elderly". *International Journal of Aging and Human Development*, 18, 227-293.

Barker, C. (2005) "Cultural Studies: Theory and Practice". London: Sage.

Barnes, C., and Mercer, G. (2006) "Independent Futures: Creating User-led Disability Services in a Disabling Society". Bristol: Policy Press.

Barnes, M., Cotterell, P. (2011) "Critical Perspectives on User Involvement". Bristol: Policy Press.

Barnes, M., Taylor, D., and Ward, E. (2013) "Being well enough in old age". *Critical Social Policy*, 33 (3).

Barrington-Leigh, C., P., and Helliwell, J., F. (2008) "Empathy and Emulation: Subjective Well-Being and the Geography of Comparison Groups". *NBER Working Paper* No. 14593.

Barry, A., Osborne, T., and Rose, N. (eds) (1996) "Foucault and political reason". London: UCL Press Ltd.

Bartels, M. (2015) "Genetics of wellbeing and its components satisfaction with life, happiness, and quality of life: a review and meta-analysis of heritability studies". *Behaviour Genetics*. 45(2), 137-56.

Bartels, M., and Boomsma, D., I. (2009) "Born to be happy? The etiology of subjective well-being". *Behaviour Genetics*, 39(6), 605-615

Bartlett, S., J., Piedmont, R., Bilderback, A., Matsumoto, A., K., Bathon, J. (2003) "Spirituality and well-being in persons with Rheumatoid Arthritis" *Arthritis Rheum* 49(6):778-83.

Bartolini, S., and Bilancini, E. (2010) "If not GDP, what else? Using relational goods to predict the trends of subjective well-being". *International Review of Economics*, *57*, 199–213.

Bartram, D. (2012) "Elements of a Sociological Contribution to Happiness Studies". *Sociology Compass*, 6, 644–656.

Bartova, Z. (2014) "Learning from the Experts: Qualitative Study of the Lived Experience of Mental Illness". Unpublished PhD, Edith Cowan University.

Basset, T., Faulkner, A., Repper, J., and Stamou, E. (2010) "Lived Experience Leading The Way for Peer Support in Mental Health". London: Together.

Battersby, D. (1998). "Successful ageing and challenges". In C., Bevan, and B. Jeeawody, (eds) 'Successful ageing: Perspectives on health and social Construction'. Sydney: Mosby.

Baum, A., Fleming, R., and Singer, J., (1985) "Understanding Environmental Stress: Strategies for conceptual and methodological integration". In A. Baum and J. Singer, (eds) 'Advances in environmental psychology', New Jersey: Lawrence Erlbaum Associates.

Baumeister, R., F. (1987) "How the self became a problem: A psychological review of historical research". *Journal of Personality and Social Psychology*, 52, 163-176.

Baumeiste, R., F., and Leary, M., R. (1995) "The Need to Belong: Desire for Interpersonal Attachments as a Fundamental Human Motivation". *Psychological Bulletin*, 117 (3), 497-529.

Bazeley, P. (2007) "Qualitative Data Analysis with Nvivo". London: Sage Publications.

BBC (2013) 'Whitehall "Nudge" to be part privatised'. Available at: http://www.bbc.co.uk/news/uk-22362727 [Accessed: 20/9/13].

Beacon North East Community-University Research Partnerships (2011) "Beacon North East Co-inquiry Action Research (CAR) Project". Durham: Beacon North East and the National Coordinating Centre for Public Engagement.

Beattie, A., and Gott, M. (1993) "Health and Wellbeing – A reader". Buckingham: Open University Press.

Beaumont, J. (2011) "Measuring National Well-being" Discussion paper on domains and measures. London.

Becker, S., Sempik, J. and Bryman, A. (2010) "Advocates, agnostics and adversaries: Researchers' perceptions of service user involvement in social policy research". *Social Policy and Society*, 9(3), 355-366.

Beckie, T., M., and Hayduk, L., A. (1997). "Measuring Quality of Life". Social Indicators Research, 42 (1), 21-39

Beesley, K., B., and Russwurm, L., H. (1989). "Social Indicators and Quality of Life Research: Toward Synthesis". *Environments*, 20, 22-39.

Bell, D. (2005) Annex 1 "Review of research into subjective wellbeing and its relation to sport and culture". In S. Galloway, (2006) 'Education Research Programme Research Findings series number 12'. Edinburgh: Scottish Education Department.

Bell J., E., and Staeheli, L., A. (2001) "Discourses of diffusion and democritization". California: Sage.

Bell, J; Cheney, G., Hoots, C., Kohrman, E., Schubert, J., Stidham, L., and Traynor, S. (2004). "Comparative similarities and differences between action research, participative research, and participatory action research". Available at: http://www.arlecchino.org/ildottore/mwsd/group2final-comparison.pdf [Accessed: 20/1/2012].

Ben-Arieh, A. (2000) "Beyond welfare - measuring and monitoring the state of children: New trends and domains". *Social Indicators Research*, 523, 235-257.

Bengston, V., Elder, G., Putney, N. (2005) "The life course perspective on ageing: linked lives, timing and history". In J., Katz, S., Peace, and S, Spurr, (eds) (2012) 'Adult lives: A life course perspective'. Bristol: Policy Press.

Bennet, P., and Murphy, S. (1997) "Psychology and health promotion". Buckingham: Open University Press.

Benneworth, P., Conway, C., Charles, D., Humphrey, L., and Younger, P. (2009) "Characterizing modes of university engagement with wider society: A literature review and survey of best practice". Newcastle: Office of the Pro Vice Chancellor, Newcastle University.

Benzeval, M., Judge, K., and Whitehead, M. (1995) "Tackling Inequalities in Health an Agenda for Action". London: Kings Fund.

Berg, M., and Veenhoven, R. (2010) "Income inequality and happiness in 119 nations". In B. Greve, (ed) 'Social policy and happiness in Europe'. Cheltenham: Edgar Elgar.

Bergdolt, K. (2008) "Wellbeing: A Cultural History of Healthy Living". Cambridge: Polity Press.

Berger-Schmitt, R. (2002) "Considering social cohesion in quality of life assessments: Concept and measurement". *Social Indicators Research*,59(10–3), 403–428.

Bergin, A., E., Masters, K., S., and Richards, P., S. (1987) "Religiousness and psychological well-being re-considered: A study of an intrinsically religious sample". *Journal of Psychiatry*, 34, 197-204.

Berglund, P., Andrade, L., Caraveo-Anduaga, J., J. (2003) "The epidemiology of major depressive episodes: results from the International Consortium of Psychiatric Epidemiology Surveys". *International Journal of Methods Psychiatric Research*, 12(1), 3-21.

Bergold, J., and Thomas, S. (2012) "Participatory Research Methods: A Methodological Approach in Motion". *Forum qualitative social research*. 13 (1), Art. 30.

Bergold, J., (2007) "Participatory strategies in community psychology research—a short survey". In A. Bokszczanin, (ed) 'Poland welcomes community psychology: Proceedings from the 6th European Conference on Community Psychology'. Opole: Opole University Press.

Bergold, J., and Thomas, S. (2010) "Participative Research". In G., Mey and K., Mruck, (eds) 'Handbook of Qualitative Research in Psychology'. Wiesbaden: VS Verlag.

Berkman, L., Kawachi, I. (2000) "Social Epidemiology". Oxford: Oxford University Press.

Bernard, M., and Scharf, T. (eds) (2007) "Critical Perspectives on Ageing Societies". Bristol: Policy Press.

Bernhard C., and Noll, H., H. (2003) "Subjective Well-Being in the European Union During the 90s". *Social Indicators Research*, 64, (3) 180-197.

Berry, C. (2014) "Wellbeing in four policy areas". London: New Economics Foundation.

Betancourt, J and Quinlan, J. (2007) Editorial – "Personal Responsibility Versus Responsible Options: Health Care, Community Health Promotion, and the Battle Against Chronic Disease". *Preventing Chronic Disease: public health research practice and policy*, 4 (3).

Bhondekar, R., D. (2015) "Love - The Key to Optimism: Path towards happiness". National Bureau of Economic Research working paper 18667. New York: Notion Press.

Bickenbach, J., Felder, F., and Schmitz, B. (eds) (2013) "Disability and the Good Human Life". Cambridge: Cambridge University Press.

Biddle, S. J. H., Wang, C. K. J., Chatzisarantis, N. L. D. and Spray, C. M. (2003a) Motivation for physical activity in young people: Entity and incremental beliefs about athletic ability. Journal of Sports Sciences, 21, 973–89.

Biddle, S., J., H., and Ekkekakis, P. (2005) "Physical active lifestyles and well-being". In F., A., Huppert, N., Baylis, and B. Keverne, (eds) 'The science of well-being'. Oxford: Oxford University Press.

Biesecker B., B., and Erby, L. (2008) "Adaptation to living with a genetic condition or risk: a mini-review". *Clinical Genetics*, 74(5):401-7.

Bilancini, E. (2012) "Utility Theory, Subjective Well-Being and Social Comparison". Unpublished PhD, University of Bucharest. Available at: http://www.econ-pol.unisi.it/dottorato/pagine/doc/tesi/Bilancini\_PhD\_Thesis.pdf [Accessed:8/3/2014]

Billson, J., M., and Fluehr-Lobban, C. (eds) (2005) "Female Well-Being: Towards a Global Theory of Social Change". London: Zed Books.

Billson, J., M. (2005) "Defining Female Well-Being". In J. M. Billson, and C. Fluehr-Lobban, (eds) 'Female Well-Being: Towards a Global Theory of Social Change'. London: Zed Books.

Binkley, S. (2012) "Happiness as Enterprise: A meditation on governmentality and neoliberal life". Sound recording of a lecture at Birkbeck College, London 26 June 2012. Available at http://www.mixcloud.com/player/ [Accessed: 27/2/2013).

Birn, A., Pillay, Y., Holtz, T. (3<sup>rd</sup> edn) (2009) "Textbook of International Health: Global Health in a Dynamic World". Oxford: Oxford University Press.

Bird, W. (2007) "Natural Green Space: Viewpoint". *British Journal of General Practice*, 57, (534) 65-79.

BIS (Department for Business, Innovation and Skills) (2008) "Foresight Mental Capital and Wellbeing Project. Final Project report". London: The Government Office for Science. Available at http://www.bis.gov.uk/foresight/our-work/projects/published-projects/mental-capital-and-wellbeing/reports-and-publications. [Accessed: 27/2/2013).

Biswas-Diener, R.,M. (2008) "Material wealth and subjective well-being". In M., Eid, and R., J. Larsen, (eds) 'The science of subjective well-being'. New York: The Guilford Press.

Bjørnskov, C. (2007) "The multiple facets of social capital". *European Journal of Political Economy*, 22, 22–40.

Bjørnskov, C. (2012) "Wellbeing and the size of government". In P. Booth, (ed) "...and the Pursuit of Happiness Wellbeing and the Role of Government". London: The Institute of Economic Affairs.

Bjørnskov, C., Dreher, A., and Fischer, J., A., V. (2008) "Cross-country determinants of life satisfaction: Exploring different determinants across groups in society". *Social Choice and Welfare*, 30, 119–173.

Black E.,C., and Levy, L., W. (eds) (1969) "European society in the 18<sup>th</sup> century". New York: Harper and Row.

Black, D., Morris, J., Smith, C. and Townsend, P. (1980) "Inequalities in Health: Report of a Research Working Group". London: Department of Health and Social Security.

Black, R., Armstrong, M., Boyle, S., Donald, G., Hanlon, P. (2015) "Housing and wellbeing. The commission on housing and wellbeing: A blue print for Scotland's Future". Available at:

http://housingandwellbeing.org/assets/documents/Commission-Final-Report.pdf [Accessed: 6/8/14]

Blackburn, H., Hanley, B., and Staley, K. (2010) "Turning the pyramid upside down". Eastleigh: INVOLVE.

Blackmore, C. (2009) "Responsible Wellbeing and its Implications for Development Policy". WeD Working Paper 09/47.

Blake, N., F. (1996) "A history of the English language". London: Palgrave.

Blanchard, J., M. (2013) (ed) "Aging in Community". Chapel Hill, NC: Second Journey

Blanchflower, D., G., and Oswald A., J. (2004a) "Money, sex and happiness: an empirical study". *Scandinavian Journal of Economics*, 106(3), 393–415.

Blanchflower, D., G., and Oswald A., J. (2004b) "Well-being over time in Britain and the USA". *Journal of Public Economics*, 88, 1359–1386.

Blanchflower, D., G., and Oswald A., J. (2005) "Happiness and the Human Development Index: The Paradox of Australia". *The Australian Economic Review*, 38(3), 307–318.

Blanchflower, D., G., and Oswald A., J. (2008a) "Is well-being U-shaped over the life cycle?". Social Science and Medicine, 66, 1733–1749.

Blanchflower, D., G., and Oswald, A., J. (2010) "Well-Being, Insecurity and the Decline of American Job Satisfaction". Available at: www.dartmouth.edu/~blnchflr/papers/JobSat.pdf Accessed:11/05/14]

Blane, D., Netuveli, G., Bartley, M. (2007a) "Does Quality of Life at Older Ages Vary with Socio-Economic Position?" *Journal of Epidemiology and public health* 54, 23–30.

Blane, D., Netuveli, G., Stonem, J. (2007b) "The development of life course epidemiology". *Journal of Epidemiology and public health*, 55, 31–38.

Blankenship, K., M., Friedman, S., R., Dworkin, S., and Mantell, J., E. (2006) "Structural Interventions: Concepts, Challenges and Opportunities for Research". *Journal of Urban Health*. 83(1), 59–72.

Blaxter, M. (1990) "Health and Lifestyles". London: Routledge.

Bloodworth, A. (2005) "Theories of Wellbeing". Nursing Philosophy, 6 (3) 213-215.

Bloom, J., R., Petersen, D., P. and Kang, S., H. (2007) "Multi-dimensional quality of life among long-term (5+ years) adult cancer survivors". *Psycho-Oncology*, 16 (8).

Blore, J., D. (2008) "Subjective Wellbeing: An Assessment of Competing Theories". Faculty of Health, Engineering and Science. Unpublished PhD, Deakin University.

Bogardus, E. (1922) "A History of Social Thought". California: California Press.

Bok, D. (2010) "The politics of Happiness: What government can learn from new research on wellbeing". Princeton: Princeton University Press.

Bok, S. (2010) "Exploring Happiness". New Haven: Yale University Press.

Bolam, B., Gleeson, K., and Murphy, S. (2003) "Lay Person" or "Health Expert"? Exploring Theoretical and Practical Aspects of Reflexivity in Qualitative Health Research". Qualitative Social Research 4, (2) Art. 26.

Boniwell I., Osin, E., Linley, P., A., and Ivanchenko, G., V. (2010) "A question of balance: Time perspective and well-being in British and Russian samples". *Journal of Positive Psychology*. 5:24-40.

Bond, J. and Corner, L. (2006) "The future of wellbeing: Quality of Life of older people in the twenty first century". In M. Bond, (2003) 'The pursuit of happiness'. London: Polity Press.

Borg, M., Karlsson, B., Hesook, S., and McCormack, B. (2012) "Opening up for many voices in knowledge construction". *Qualitative Social Research*, 13 (1), Art. 1.

Bourdieu, P. (1991) "Language and Symbolic power". London: Polity Press.

Bourdieu, P. (1992) "Invitation to a Reflexive Sociology". Chicago: University of Chicago Press.

Bosselmann, K., Engel, R., and Taylor, P. (2008) "Governance for Sustainability: Issues, Challenges an and Successes". Garlan, Switzerland: International union for conservation of nature (IUCN).

Boud, D., and Walker, D. (1998) "Promoting Reflection in Professional Courses: the challenge of Context". Studies in Higher Education, 23, (2) 191-206.

Boughan, K., M. (2008) "A review of Well-being: A Cultural History of Healthy Living". The journal of the historical association, 94 (316), 551-554.

Bowern, C., Evan, B. (eds) (2015) "The Routledge Handbook of Historical Linguistics". Routledge: Abingdon.

Bowling, A. (2000) "Adding quality to quantity. Older people views on their quality of life and its enhancement". Available at : http://www.shef.ac.uk/uni/projects/gop/index.htm [Accessed:27/2/2013).

Bowling, A (2001) "Measuring Disease". Buckingham: Open University Press.

Bowling, A. (2005a) "Measuring health. A review of quality of life measurement scales". (3rd edn) Berkshire: Open University Press.

Bowling, A. (2005b) "Implications for ageing well in the twenty first century". In J., Katz, , S., Peace, S. Spurr, (eds) (2012) 'A life course perspective'. Buckingham: Open University Press.

Bowling, A. (2005c) "Ageing well: Quality of Life in old age". Buckingham: Open University Press.

Bowling, A. (2008) "Enhancing later life: how older people perceive active ageing?". *Aging and Mental Health*, 12(3), 293-301.

Bowling, A., Gabriel, Z., Dykes, J., Dowding, L., M. Evans, O. (2003) "Lets ask them: A National Survey of definition of Quality of Life and its Enhancement among people aged 65 and over?". *International Journal of Aging and Human Development*, 56(4) 269-306.

Bowling, A., and Gabriel, Z., (2003) "Lay theories of Quality of Life in older age". *Ageing and Society*, 27, 827-848.

Bowling, A., Banister, D., Stenner, P., Titheridge, H., Sproston, K., McFarquhar, T. (2009) "QoL in older age: Psychometric testing of the multi-dimensional older people's QoL (OPQOL) questionnaire and the causal model under-pinning it". London: New Dynamics of Ageing.

Bowling, A., and Iliffe, S. (2011) "Psychological Approach to successful Ageing predicts future of Quality of Life in older adults". *Health and Quality of Life* 9 (13) 27-38.

Bradburn, N., M. (1969) "The structure of psychological well-being". Chicago: Aldine.

Bramley, G., Dempsey, N., Power, S., Brown, C., Watkins, D. (2009)"Social sustainability and urban form: evidence from five British cities". *Environment and Planning*, 41(9), 2125-2142.

Brandstetter, S., McCool, M., Wise, M., and Loss, J. (2014) "Australian health promotion practitioners' perceptions on evaluation of empowerment and participation". *Health Promotion International*, 29(1), 70-80.

Branfield, F., Beresford, P., Danagher, N. and Webb, R. (2005) "Independence, Wellbeing and Choice: A Response to the Green Paper on Adult Social Care. Report of a Consultation with Service Users". London: National Centre for Independent Living and Shaping Our Lives.

Branfield, F. and Beresford, P. (2006) "Making user involvement work: Supporting service user networking and knowledge". York: Joseph Rowntree Foundation. Available at: http://www.jrf.org.uk/sites/files/jrf/1410-user-networking-knowledge.pdf [Accessed: 5/5/2009].

Braun, V., and Clarke, V. (2006) "Using Thematic Analysis in Psychology". *Qualitative Research in Psychology*, 3, 77-101.

Brereton, F., Clinch, J.P. and Ferreira, S. (2008) "Happiness, geography and the environment". *Ecological Economics*, 65(2), 386–396.

Brett, J., Staniszewska, S., and Mockford, C.(2010) "The PIRICOM study: A systematic review of the conceptualisation, measurement, impact and outcomes of patient and public involvement in health and social care research". United Kingdom Clinical Research Collaboration, London. Available at www.ukcrc.org. [Accessed:29/8/ 2015]

Brickman, P., D., and Campbell, D.,T. (1971) "Hedonic relativism and planning the good society". In M.,H. Appley, (ed) 'Adaptation Level Theory'. New York: Academic Press.

Brickman, P., Coates, D. and Janoff-Bulman, R. (1978). "Lottery winners and accident victims: Is happiness relative"? *Journal of Personality and Social Psychology*, 36(8), 917-927.

Brickson, S. (2000) "The impact of identity orientation on individual and organizational outcomes in demographically diverse settings". *Academy of Management Review*, 25, 82-101.

Briggs, A. (ed) 1970) "The Nineteenth Century". London: Thame and Hudson,

Briggs, A. (1983) "A social history of England". London: Weidenfield and Nicolson.

Briggs, A. (2000) "The age of Improvement' 1783-1867". (2<sup>nd</sup> edn) Essex: Longman publishing.

British Psychological Society (2000) "Code of Conduct, Ethical principles and guidelines". Available at: www.scutrea.ac.uk/library/bpscode.pdf [Accessed: 25/3/2007]

Brockmeier, J. and Carbaugh, D. (2001) "Narrative and Identity: Studies in Autobiography, Self and Culture". Amsterdam/Philadelphia: John Benjamins.

Brookfield, S. (1994) "Tales from the dark side: a phenomenography of adult critical reflection". *International Journal of Lifelong Education*, 13 (3), 203–216.

Brookfield, S. (1995) "Becoming a Critically Reflective Teacher". San Fransisco: Jossey-Bass.

Brown, S., D. (2009) "History circles: The doing of teaching history". *History Teacher*, 42(2), 191-203.

Brown, J., E., Butow, P., N., Culjak, G., Coates, A., S., Dunn, S., M. (200) "Psychosocial predictors of outcome: time to relapse and survival in patients with early stage melanoma". *British Journal of Cancer*, 83, 1448–1453.

Brown, J., Bowling, A., Flynn, T. (2004) "Models of Quality of Life: A Taxonomy, Overview and Systematic Review of the Literature". *Ageing and Society*, 29, 657-678.

Brown, K., W., and Kasser, T. (2005), "Are Psychological and Ecological Wellbeing Compatible? The Role of Values, Mindfulness and Lifestyle". *Social Indicators Research* 74 (2), 349-368.

Brown, R., I., Schalock, R., L., Brown, I, (2009) "Quality of Life: Its application to persons with intellectual disabilities and their families – Introduction and overview". *Journal of Policy and Practice in Intellectual Disabilities*, 6 (1), 2-6.

Brubaker, R., Cooper, F. (2000). "Beyond 'Identity". Theory and Society 29,1–47.

Bruni, L., Porta, P.L., (2007a). Introduction. In L. Bruni and P., L. Porta, (eds) 'Handbook on the Economics of Happiness'. Cheltenham: Edward Elgar.

Bruni, L., Porta, P.L. (eds) (2007b) "Handbook on the Economics of Happiness". Cheltenham: Edward Elgar.

Bryant, F., B., and Veroff, J. (1982) "The structure of psychological wellbeing: A socio-historical analysis". *Journal of Personality and Social Psychology*, 43, 653-673

Bryant, F., B. (2003) "Savoring Beliefs Inventory (SBI): A scale for measuring beliefs about savouring". *Journal of Mental Health*, 12, 175-196.

Bryman, A. (2001) "Social Research Methods". Oxford: Oxford University Press.

Bryman, A. (2008) "Social Research Methods". (2nd edn) Oxford: Oxford University Press.

Buchan, N., R., Brewer, M., B., Grimalda, G., Wilson, R., K., Fatas, E., Foddy, M. (2011). "Global social identity and global cooperation". *Psychological Science*, 22, 821-828.

Buchanan, D. (2000) "An Ethic for Health Promotion: Re-thinking the Sources of Human Well-Being". New York: Oxford University Press.

Buntinx, W., H., E., and Schalock, R., L. (2010) "Models of Disability, Quality of Life, and Individualized Supports". *Journal of Policy and Practice in Intellectual Disabilities*. 7 (4), 283–294.

Bunton, R. and Macdonald, G. (1992) "Health Promotion". London: Routledge.

Burchardt, T. (2006) "Happiness and social policy: barking up the right tree in the wrong neck of the woods". In L. Bauld, K. Clarke and T. Maltby, (eds) 'Social policy review 18'. Bristol: The Policy Press.

Burchi, F., and Gnesi, C. (2013) "A Review of the Literature on Wellbeing in Italy: A Human Development Perspective". Working Paper n° 175,

Burman, E. (2008) "Deconstructing developmental psychology". (2<sup>nd</sup>edn) London: Routledge.

Burns, N., and Grove, S., K. (2007) "Understanding Nursing Research: Building An Evidence-Based Practice". (4th edn) St. Louis: Saunders Elsevier.

Burr, V. (1995) "An introduction to social constructionism". Florence: Taylor and Frances/Routledge.

Burr, V. (1998) "Realism, Relativism Social constructionism and Discourse". In I. Parker, (ed) 'Social constructionism, Discourse and Realism'. London: Sage.

Burt, J., Roland, M., Paddison, C., Reeves, D., Campbell, J., Abel, G., Bower, P. (2012). "Prevalence and benefits of care plans and care planning for people with long-term conditions in England". *Journal of Health Services Research and Policy*, 17, suppl 1, 64–71.

Burton, E., Mitchell, L., and Stride, C.(2011) "Good places for ageing in place: development of objective built environment measures for investigating links with older people's wellbeing". *BMC Public Health*. Available at: http://bmcpublichealth.biomedcentral.com/articles/10.1186/1471-2458-11-839 [Accessed: 20/11/2012].

Bury, M. (1982). "Chronic illness as biographical disruption". Sociology of Health and Illness, 4, 167-182.

Bury, M. (1997) "Health and Illness in a Changing Society". London: Routledge.

Bury, M. (2001) "Illness narratives: fact or fiction?". Sociology of Health and Illness, 23, 263-285.

Butterworth, J., Steere, D.,E., and Whitney-Thomas, J. (1997) "Using person-centered planning to address personal quality of life". In R.L. Schalock, (ed) 'Quality of life volume II: Application to persons with disabilities'. Washington DC: American Association on Mental Retardation.

Butterworth, J., (2000) "The relationship between the built environment and wellbeing: a literature review". Victorian Health Promotion Foundation, available at:

file:///C:/Users/Wayne/Downloads/built\_environment.pdf [Accessed: 10/10/2014].

Buunk, B., P., and Gibbons, F., X. (eds) (1997) "Health, coping, and well-being: Perspectives from social comparison theory". New Jersey: Lawrence Erlbaum Associates Publishers.

Buunk, B., P., and Mussweiler, T. (2001) "New directions in social comparison research". *European Journal of Social Psychology* 31, 467-475.

Byrne, M. (2001) "Hermeneutics as a methodology for textual analysis" *Aorn Journal*, 73 (5), 968-970.

Cabinet analysis and Insight Team (BLOG) Available at: https://coanalysis.blog.gov.uk/2015/03/25/well-being-is-everyones-responsibility-local-partnerships-can-help/ [Accessed: 2/11/ 2015].

Cai, J., Clark, J., Croft, S., Deakin, S., Fowajah, A. (2008) "Indications of public health in the English Regions: Older people". York: Association of Public Health Observatories.

Cairney, J., Corna, L., Streiner, D., L. (2010) "Mental health care in later life: results from a national survey of Canadians. *Canadian Journal of Psychiatry* 55(3), 157-164.

Cairney, J., Rigoli, D., Piek, J. (2013) "Developmental coordination disorder and internalizing problems in children: The environmental stress hypothesis elaborated". *Developmental Review*, 33 (3) 224–238.

Caldon, L., Marshall-Cook, H., Speed, G., Reed, M. and Collins, K. (2010) "Consumers as researchers – innovative experiences in UK National Health Service Research". *International Journal of Consumer Studies*, 34 (5), 547-550.

Callaghan, P., Khalil, E., Morres, I., Carter, T. (2011) "Pragmatic randomised controlled trial of preferred intensity exercise in women living with depression". *BMC Public Health*, 11(1) 465-479.

Cambridge Dictionary online (2016). Available at: http://dictionary.cambridge.org/dictionary/english/professional [Accessed: 14/1/16].

Cameron, P. (1975) "Mood as an indicant of happiness: Age, sex, social class and situational differences". *Journal of Gerontology*, 30, 216-224.

Cameron, E., Mathers, J., and Parry, J. (2006) "Health and well-being': Questioning the use of health concepts in public health policy and practice". *Critical Public Health*, 16 (4) 347-354.

Cameron, P., A., and Stewart, C. (2011) "What's in a name? How the lack of an agreed definition of self management in chronic pain prevents the development of guidelines". *British Pain Society*, Winter Edition 2011, 36-38.

Cameron, M., Crane, N., Ings, R., Taylor, K., (2013) "Promoting well-being through creativity: how arts and public health can learn from each other". *Perspectives in Public Health*, 133(1), 52-59.

Camfield, L. (2006) "The Why and How of Understanding 'Subjective' Wellbeing: Exploratory Work by the WeD Group in Four Developing Countries". WeD Working Paper 26, Bath: University of Bath.

Camfield, L., and Skevington, S., M. (2008) "On Subjective Well-being and Quality of Life". *Health Psychology*, 13 (6), 764-775.

Campbell, A., Converse, P., and Rogers, W.,L. (1976) "The Quality of American Life: Perceptions, Evaluations and Satisfaction". New York: Russell-Sage Foundation.

Care Act (2014) London: DoH. Available at: http://www.legislation.gov.uk/ukpga/2014/23/section/1/enacted. [Accessed: 12/08/14].

Carlisle, C., and Hanlon, P. (2005) "Complex territory of well-being: contestable evidence contentious theories and speculative conclusions". *Journal of public mental health*, 6 (2),8-13.

Carlisle, C., and Hanlon, P. (2007) "Wellbeing and consumer culture: A different kind of public health problem?" *Health Promotion International*, 22 (3), 127-135.

Carlisle, C., and Hanlon, P. (2008) "Well-being as a focus for public health? A critique and defence". *Critical Public Health*, 18 (3), 263-270.

Carlisle, S., and Hanlon, P. (2009) "Is modern culture bad for your health?". *Global Health Promotion*, 16 (4), 27-34.

Carlisle, S., Henderson, G., and Hanlon, P. (2009) "Wellbeing: A collateral casualty of modernity". *Social Science and Medicine*, 69 (10), 1556-1560.

Carp, F. (1968) "Effects of improved housing on the lives of older people". In B., N. Neugarten, (ed) 'Middle age and aging'. Chicago: Chicago University Press.

Carr, S. (2004) "Has Service User Participation Made a Difference to Social Care Services?". Position Paper No. 3. London: Social Care Institute for Excellence.

Carr, S., M., Lhussier, M., Forster, N., Geddes, L., Deane, K., Pennington, M. Et al.(2011) "An evidence synthesis of qualitative and quantitative research on component intervention techniques, effectiveness, cost-effectiveness, equity and acceptability of different versions of health-related lifestyle advisor role in improving health". *Health Technology Assess*, 15(9) iii-iv, 1-284.

Carstensen, L., L. (1998) "A life span approach to social motivation". In J., Heckhausen and C. Dweck, (eds) 'Motivation and self-regulation across the life span'. Cambridge: Cambridge University Press.

Carstensen, L., L. (2009) "A long bright future". New York: Random House.

Carter, S. M., and Little, M. (2007) "Justifying knowledge, justifying method, taking action: Epistemologies, methodologies and methods in qualitative research". *Qualitative Health Research*, 17, 1316-1328.

Cashore B., Auld G., Newsom D. (2004) "Governing through markets: forest certification and the emergence of non-state authority". New Haven: Yale University Press.

Center for disease control and prevention (2013) Available at: http://www.cdc.gov/hrqol/wellbeing.htm [Accessed: 12/08/14].

Centre for Well-Being (2011) "Measuring our Progress". London: nef.

Cervinka, R., Röderer, K., and Hefler, E. (2012) "Are nature lovers happy? On various indicators of well-being and connectedness with nature". *Journal of Health Psychology* 17 (3), 379-388.

Chadwick, E. (1842) "Report on the Sanitary Conditions of the Labouring Population of Great Britain". London: Home Office.

Chambers, R. (1997) Editorial: "Responsible Wellbeing – A Personal Agenda for Development". World Development, 25(11), 1743-1754.

Chambers, R. (2005) "Ideas for Development". London: Earthscan.

Chambers, R. (2008) "PRA, PLA and pluralism: Practice and theory". In P. Reason and H. Bradbury (eds) 'The Sage handbook of action research: Participative inquiry and practice'. (2nd edn) London: Sage.

Chapman, L., and West-Burnham, J. (2008) "Education for social justice: Achieving wellbeing for all". London: Continuum.

Charmaz, K. (1993) "Good days and bad days: The self in chronic illness and time". New Jersey: Rutgers University Press.

Charmaz, K. (1995) "The body, identity, and self: Adapting to impairment". *The Sociological Quarterly*, 36, 657-680.

Charmaz, K. (2000) "Experiencing chronic illness". In G., Albrecht, R., Fitzpatrick, and S., C. Scrimshaw, 'The Handbook of Social Studies in Health and Medicine'. London: Sage.

Chavez, A., Milburn, K., Parry, O., and Platt, S. (2005) "Understanding and researching wellbeing: Its usage in different disciplines and potential for health research and health promotion". *Health Education Journal* 64 (1),70-87.

Checkland, K (2014) "Leadership in the NHS: does the Emperor have any clothes"? *Journal of Health Services Research and Policy*, Online first, 19 (4), 253-256.

Cheek, J. (2000) "Postmodern and postcultural approaches to nursing research". London: Sage.

Chirkov, V., and Ryan, R., M. (2001) "Control versus autonomy support in Russia and the U.S.: effects on well-being and academic motivation". *Journal of Cross-Cultural Psychology*, 12, 27-36.

Christakopoulou, S., Dawson, J., and Aikaterini, G. (2001) "The Community Well-Being Questionnaire: Theoretical Context and Initial Assessment of Its Reliability and Validity". *Social Indicators Research*, 56 (3), 314-328.

Christoph, B., and Noll, H. (2003) "Subjective Well-Being in the European Union During the 90s". Social Indicators Research, 64 (3), 197-209.

Christopher, J., C. (1999) "Situating psychological well-being: Exploring the cultural roots of its theory and research". *Journal of Counselling Development* 77,141–152.

Churchill, W., S. (1956) "The Great Democracies: A History of the English Speaking Peoples". Vol IV. London: Cassell and Co.

Cianci, R., Gambrel, P., A. (2003) "Maslow's hierarch of needs: Does it apply in a collectivist culture". Journal of Applied Management and Entrepreneurship, 8 (2), 143-161.

Cieslik, M. (2013) "Not Smiling but Frowning": Sociology and the 'Problem' of Happiness". *Sociology online first*, 49 (3), 422-437.

Cieslik, M., and Bartram, D. (2014) "Introduction: Happiness Studies". *Sociological Research Online*, 19 (2), 17-20.

Clark, J., C., D. (1986) "Revolution and Rebellion: State and Society in 17<sup>th</sup> and 18<sup>th</sup> century England". Cambridge: Cambridge University Press.

Clark, D., A. (2002) "Visions of Development: A Study of Human Value". Cheltenham: Edward Elgar.

Clark, A. (2003a) "Unemployment as a social norm: Psychological evidence from panel data". *Journal of Labor Economics*, 21(2), 323–351.

Clark, E., H. (2004) "Quality of life: a basis for clinical decision-making in community psychiatric care. *Journal of Psychiatric Mental Health Nursing*, 11(6), 725-730.

Clark, D., A. (2005) "Sen's capability approach and the many spaces of human well-being." The Journal of Development Studies, 41(8), 1339-1368.

Clark, D., A. (2010) "The Capability Approach: Its Development, Critiques and Recent Advance". London: Sage

Clark, A., and Oswald, A. (1994) "Unhappiness and Unemployment". *The Economic Journal*, 104 (424), 648–659.

Clark, A. and Lelkes, O. (2005) "Deliver us from evil: Religion as insurance". Paris: PSE working paper 43.

Clark, A., E., Georgellis, Y., Lucas, R., E., and Diener, E. (2004) "Unemployment alters the set point for life satisfaction". *Psychological Science*, 15, 8-13.

Clark, D., A., and Gough, I. (2005) "Capabilities, Needs and Wellbeing: Relating the Universal and the Local." In L. Manderson, (ed) 'Rethinking Wellbeing'. Perth: API Network, Australia Research Institute.

Clark, A., Frijters, P., and Shields, M. A. (2008) "A survey of the income happiness gradient". *Journal of Economic Literature*, 46 (1), 95-144.

Clarke, M. (2007) "Overview of Methods" In C. Webb and B. Roe (eds) 'Reviewing Research Evidence for Nursing Practice'. London: Blackwell Publishing.

Clarke, A. and Warren, L. (2007) "Hopes, fears and expectations about the future: what do older people's stories tell us about active ageing?". *Ageing and Society*, 27 (4), 465-488.

Clark, A. E., Frijters, P., and Shields, M., A. (2008) "Relative income, happiness, and utility: An explanation for the Easterlin paradox and other puzzles". *Journal of Economic Literature*, 46, 95–144.

Clarke, P., and Nieuvenhuijsen, R. (2009) "Environments for healthy ageing: a critical review". *Maturitas*, 64, 14-19.

Clark, A., E., and Senik, C. (2010) "Who compares to whom? The anatomy of income comparisons in Europe". *Economic Journal*, 120, 573–594.

Clark, A., and Senik, C. (2011) "Is Happiness Different from Flourishing? Cross-country Evidence from the ESS". Paris: Paris School of Economics Working Paper, no. 4.

Clydesdale, T. (1997) "Family behaviors among early U.S. baby boomers: Exploring the effects of religion and income change, 1965-1982". *Journal of Social Forces*, 76: 605-635.

Coffey, A. and Atkinson, P.(1996) "Making sense of qualitative data". Thousand Oaks: Sage.

Cohen, A. (2002) "The importance of spirituality in well-being for Jews and Christians". *Journal of Happiness Studies*, 3, 287–310.

Cohen, L., Manion, L., Morrison, K. (2007) "Research Methods in Education" (6th edn) London: Routledge.

Cohrs, J., C., Christie, D., J., White, M., P., Das, C. (2013) "Contributions of positive psychology to peace: Towards global well-being and resilience". *American Psychologist*, *68*, 590-600.

Cole, P. (2006) "Are Children Ready for School: Health and Physical Well-Being". Indiana: Indiana Institute on Disability and Community, Early Childhood Center.

Cole-Hamilton, I., and Lang, T. (1986) "Tightening Belts – A report on the impact of poverty on food". London: The London Food Commission Promotions Ltd.

Collard, D. (2006) "Research on Well-Being: Some Advice from Jeremy Bentham". *Philosophy of the Social Sciences*, 36 (3) 330-354.

Collins, C., McCrory, M., Mackenzie, M., and McCartney, G. (2015) "Social theory and health inequalities: Critical realism and a transformative activist stance? *Social Theory and Health*, 13, 377-396.

Colman, A. (2008) "A dictionary of psychology". Oxford: Oxford University Press.

Commission for Architecture and the Built Environment (CABE) (2007) "What are we scared of? The value of risk in designing public space". London: CABE.

CABE (2009) "Making the Invisible Visible – the Real Value of Park Assets". London: CABE.

CABE (2009) Homes for our old age: living by design". London: CABE.

CABE (2010) "Community green: using local spaces to tackle inequality and improve health". London: CABE.

Commission for Social Care Inspection (2007) "Benefit Barriers to Involvement: Finding solutions", London, Commission for Social Care Inspection. Available at: http://www.scie.org.uk/publications/misc/bbi/index.asp [Accessed: 28/8/ 2015].

Conrad P., Barker, K., K. (2010) "The Social Construction of Illness: Key Insights and Policy Implications". *Journal of Health and Social Behavior*, 51 (1), supplement.

Cook, T. (2012) "Where participatory approaches meet pragmatism in funded (health) research: The challenge of finding meaningful spaces". *Qualitative Social Research*, 13 (1), Art. 18. Available at: http://nbn-resolving.de/urn:nbn:de:0114-fqs1201187. [Accessed:14/08/15].

Cooper, C., L. (ed)(2014) "Wellbeing: A Complete Reference Guide". Oxford: Wiley-Blackwell.

Cooper, C., L. (ed) (2014) "Wellbeing: A complete reference guide: Wellbeing and the environment" Vol 1, Oxford: Wiley Blackwell

Cooper, C., L. (ed) (2014) "Wellbeing: A complete references guide: Wellbeing in later life". Voluntary 2 Oxford: Wiley Blackwell

Cooper, C., L. And Huppert, F., A. (eds) (2014) Wellbeing: A Complete Reference Guide, "Interventions and Policies to Enhance Wellbeing". Vol 6, Oxford: Wiley Blackwell

Corbin, J., and Holt, N. L. (2005) "Grounded theory". In B., Somekh, and C. Lewin, (eds) 'Research methods in the social sciences'. California: Thousand oaks.

Costa, P.,T., and McCrae, R.,R. (1980) "Influences of extraversion and neuroticism on subjective well-being". *Journal of Personality and Social Psychology*, 38, 668-78.

Costanza, R., (2006) "Quality of life: An approach integrating opportunities, human needs, and subjective well-being". *Ecological economics* 61 (2), 67–76.

Coulter, D.,L. (1997) "Health-related application of quality of life". In R.,L. Schalock (ed) 'Quality of life volume II: Application to persons with disabilities' Washington DC: American Association on Mental Retardation.

Coulter, A., Roberts, S., Dixon, A. (2013) "Delivering better services for people with long-term conditions: Building the house of care". London: Kings Fund.

Coulton, C. (1992) "Potential and problems in developing community-level indicators". In R., Hausser, B., Brown, W., Prosser (eds) 'Indicators of Children's Well-Being'. New York: Russell Sage Foundation.

Courtney, W. (2000) "Constructions of masculinity and their influence on men's wellbeing: a theory of gender and health". *Social Science and Medicine* 50, 1385-1401.

Cracknel, R. (2010) "The ageing population: Key issues for the new parliament". London: House of Commons Library Research.

Craig, R. and Mindell, J. (2007) "The health of older people". Vol 1-6. Leeds: The Information Centre.

Crawshaw, P. (2008) "Whither wellbeing for public health?". *Critical Public Health*, 18 (3), 259-261.

Creswell, J. (1998) "Qualitative inquiry and research design". Thousand Oaks: Sage.

Creswell, J. (2008) "Research design: Qualitative, Quantitative and Mixed Methods Approaches". Thousand Oaks: Sage.

Creswell, J., W., Hanson, W., E., Clark, V., L., P., and Morales, A. (2007) "Qualitative research designs: Selection and implementation". *Counseling Psychologist*, 35, 236-264.

Crisp, R. (1998) (ed) "Utilitarianism". Oxford: Oxford University Press.

Cronin de Chavez, A., Backett-Milburn. K., Parry, O., (2005) "Understanding and researching wellbeing: its usage in different disciplines and potential for health research and health promotion". *Health Education Journal*, 64, 70-87.

Cronin, P., Ryan, F., and Coughlan, M. (2010) "Concept analysis in healthcare research". *International Journal of Therapy and Rehabilitation* 17(2),62-68.

Crotty, M., J. (1998) "The Foundations of Social Research: Meaning and Perspective in the Research Process". London: Sage publications.

Cruikshank, B. (1996) "Revolutions within: self-government and self-esteem". In A., Barry, T., Osborne, and N. Rose (eds) 'Foucault and Political Reason: Liberalism, Neo-Liberalism, and Rationalities of Government'. Chicago: University of Chicago Press.

Crystal, D. (2002)"The English language". (2<sup>nd</sup> edn) London: Penguin.

Crystal D. (2010) "Evolving English: One language many voices". Oxford: Oxford University Press.

Crystal D. (2014) "Words in time and Place: Exploring Language through the Historical Thesaurus of the Oxford dictionary". Oxford: Oxford University Press.

Csikszentmihalyi, M. (1990) "Flow: The Psychology of Optimal Experience". In H. Collins, (ed) 'Annals of Physics'. New York: Harper and Row. Available at: http://www.amazon.com/Flow-Psychology-Experience-Mihaly-Csikszentmihalyi/dp/0060920432 [Accessed:18/7/2014]

Csikszentmihalyi, M., Csikszentmihalyi, I. (eds) (1988) "Optimal Experience: Psychological Studies of Flow in Consciousness". New York: Cambridge University Press.

Csikszentmihalyi, M., and Csikszentmihalyi, I. (2006) "A Life Worth Living: Contributions to Positive Psychology". Oxford: Oxford University Press.

Cummins, R., A. (1996) "Domains of life satisfaction: An attempt to order chaos". *Social Indicators Research*, 38, 303-328.

Cummins, R., A (1997) "Assessing Quality of Life". In R. Brown, (ed) 'Quality of Life for People with Disabilities: Models, Research and Practice'. (2nd edn) Cheltenham: Stanley Thornes (Publishers) Ltd.

Cummins, R., A. (2000a) "Personal income and subjective well-being: A review". *Journal of Happiness Studies*, 1, 133–158.

Cummins, R., A. (2000b) "Objective and Subjective quality of life: An interactive model". *Social Indicators Research*, 52 (1), 55-68.

Cummins, R., A. (2005) "Moving from the quality of life concept to a theory". *Journal of Intellectual Disability Research*, 49, 699–706.

Cummins, R., A., and Nistico, H. (2002) "Maintaining life satisfaction: The role of positive cognitive bias. *Journal of Happiness Studies*, 3 37-69.

Cutcliffe, J., R., and Harder, H., G. (2012) "Methodological Precision in Qualitative Research: Slavish Adherence or "Following the Yellow Brick Road?". *The Qualitative Report*, 17, Article 82, 1-19. Available at: http://www.nova.edu/ssss/QR/QR17/cutcliffe.pdf . [Accessed:2/5/15].

Cutler, S. (1973) "Volunteer association participation and life satisfaction: A cautionary research note". *Journal of Gerontology*, 25, 96-100.

Dahlberg, K. and Halling, S. (2001)."Human science research as the embodiment of openness: swimming upstream in a technological culture". *Journal of Phenomenological Psychology*, 32, 12-21.

Danna, K., and Griffin, R. (1999) "Health and Wellbeing in the workplace: A review and synthesis of the literature". *Journal of management* 25 (3), 357-384.

Dasgupta, P. (1989) "Well-being and the extent of its realisation in poor countries". Discussion paper, Development Economics Research Programme, no. 19. London: London School of Economics.

Dasgupta, P. (1993) "An Inquiry into Well-Being and Destitution". Oxford: Clarendon Press.

Dasgupta, P. (2001) "Human Well-Being and the Natural Environment". Oxford: Oxford University Press.

Davidson, N., Townsend, P., and Whitehead, M. (eds) (1998) "Inequalities in Health". London: Pelican.

Davies, D., and Dodd, J. (2002) "Qualitative research and the question of rigor". *Qualitative Health Research*, 12, 279-289.

Davies, M., and MacDowall, W. (2006) "Health Promotion Theory". Buckingham: Open University Press.

Day, M., D. (1998) "Ecopsychology and the restoration of home". *The Humanistic Psychologist*, 26(13), 51-67.

Dean H. (1997) "Multiple instruments for measuring quality of life: Instruments for Clinical Health Care Research". London: Jones and Barlett Publications Inc.

Dean, M. (1999a) "Risk, calculable and incalculable". In D. Lupton (ed) 'Risk and sociocultural theory: New directions and perspectives'. Cambridge: Cambridge University Press.

Dean, M. (1999b) "Governmentality: Power and Rule in Modern Society". London: Sage.

Dean, H. (2010) "Understanding Human Need". Bristol: Policy Press.

Deaton, A (2008) "Worldwide, residents of richer nations more satisfied". Available at:

http://www.gallup.com/poll/104608/Worldwide-Residents-Richer-Nations-More-Satisfied.aspx [Accessed:12/5/13].

Deaton, A. (2011) "The Financial Crisis and the Well-being of Americans". Oxford Economic Papers, 64, 1–26.

Deci, E. L., Ryan, R. M., Gagné, M, Leone, D., R., Usunov, J., and Kornazheva, B., P. (2001) "Need satisfaction, motivation, and well-being in the work organizations of a former Eastern bloc country: A cross-cultural study of self-determination". *Personality and Social Psychology Bulletin*, 27, 930–942.

Deci, E., L., and Ryan, R., M. (2008) "Hedonia, Eudaimonia and well-being: An introduction". *Journal of Happiness Studies*, 9:1–11.

Delle Fave, A., and Massimini, F. (2003a) "Optimal Experience in Work and Leisure among teachers and Physicians: Individual and Bio-cultural Implications". *Leisure Studies*, 22, 323-342.

Demakakos, P., and Steptoe, A., (2010) "The English Longitudinal Study of Ageing (ELSA) shines a light on the changing lives of People as they get older". London: ELSA.

Demakakos, P., McMunn, A., Steptoe, A., (2010) "Wellbeing in older age: A multi-dimensional perspective". London: ELSA.

Dempsey, N. (2008) "Quality of the built environment in urban neighbourhoods". *Planning Practice and Research* 23(2), 249-264.

Deneulin, S., and McGregor, J. (2009) "The Capability Approach and the politics of a social conception of wellbeing". WeD.09/93. Available at: http://www.gprg.org/pubs/workingpapers/pdfs/gprg-wps-032.pdf [Accessed: 2/1/2014]

Denzin, N., K., Lincoln, Y. (2005) (3<sup>rd</sup> edn) "The Sage Handbook of Qualitative Research". Thousand Oaks, CA: Sage.

De Neve, K., M. (1999) "Happy as an extraverted clam? The role of personality for subjective well-being". *Current Directions in Psychological Science*,8(5),141–144.

De Neve, K. M., and Cooper, H. (1998) "The happy personality: A meta-analysis of 137 personality traits and subjective well-being". *Psychological Bulletin*, 124, 197–229.

Dentith, A., M., Measor, L., and O'Malley, M., P. (2012) "The research imagination amid dilemmas of engaging young people in critical participatory work". *Qualitative Social Research*, 13 (1), Art. 17. Available at: http://nbn-resolving.de/urn:nbn:de:0114-fqs1201176 . [Accessed: 2/5/15].

Denzin, N.,K., and Lincoln, Y.,S. (eds) (1994) "Handbook of Qualitative Research". California: Sage.

Department for Communities and Local Government (DCLG) (2008) "Practical use of the Well-Being Power. London: DCLG;

Department of Communities and Local Government (2008) "Lifetime Homes, Lifetime Neighbourhoods: A National Strategy for Housing in an Ageing Society". London: DCLG.

Department of Communities and Local Government (2008) "Building a Society for all Ages". London: DCLG.

Department for Education and Skills (DfES) (2003) "Every Child Matters: Change for Children. Summary". London: HMSO. Available at: www.everychildmatters.gov.uk/aims. [Accessed: 14/2/2008].

Department for Education and Skills (2005) "Youth Matters". London: HMSO.

Department of Environment Food and Rural Affairs (DEFRA) (2005) "Securing the future: Delivering the UK sustainable Development Strategy". London: Department of Health, HMSO. Available at: http://www.dh.gov.uk/publications. [Accessed: 10/6/2008]

Department for Environment, Food and Rural Affairs (DEFRA) (2011) "The Natural Choice: securing the value of nature". London: The Stationery Office. Available at: www.tsoshop.co.uk [Accessed:4/1/16].

Department of the Environment, Transport and the Regions (DETR) (2001) "Power to promote or improve economic, social or environmental well-being". Available at: http://www.communities.gov.uk/index.asp?id=1133748. [Accessed:10/6/ 2008]

Department of the Environment, Transport and the Regions and Commission for Architecture and the Built Environment "By design. Urban design in the planning system: towards better practice". London: The Stationery Office.

Department of Health (DoH) (1999) "Saving Lives – Our Healthier Nation: A Contract for Health". London: Department of Health, HMSO.

DoH. (2001) "The expert patient: a new approach to chronic disease management for the 21st century". London: Department of Health, HMSO

DoH. (2003) "Health Equity Audit: A guide for the NHS". London: Department of Health, HMSO. Available at:

http://www.dh.gov.uk/en/Publicationsandstatistics/Publications/PublicationsPolicy AndGuidance/DH\_4084138 [Accessed: 3/3/2009]

DoH. (2003a) "Every Child Matters". London: Department of Health, HMSO.

DoH, (2004) "Choosing Health". London: Department of Health, HMSO.

DoH. (2004a) "The NHS Improvement Plan: Putting People at the Heart of Public Services". London: Department of Health, HMSO.

DoH. (2004b) "Choosing Health: Making Healthy Choices Easier". London: Department of Health, HMSO. Available at:

http://www.dh.gov.uk/en/Publicationsandstatistics/Publications/PublicationsPolicy AndGuidance/DH 4094550 [Accessed:1/7/2008]

DoH. (2004-9) "The quality and Outcomes Framework". London: Department of Health, HMSO. Available at:

http://www.dh.gov.uk/en/Healthcare/Primarycare/Primarycarecontracting/QOF/index.htm [Accessed: 21/6/2009]

DoH. (2005a). "Improving Care Improving Lives - Supporting People with Complex health needs- An NHS and Social Care Model to Support Local Innovation and Integration". London: Department of Health, HMSO.

DoH. (2005b) "The National Service Framework for Long term Conditions". London: Department of Health, HMSO. Available at:

http://www.dh.gov.uk/en/Publicationsandstatistics/Publications/PublicationsPolicy AndGuidance/DH 410536 . [Accessed:30/3/2007]

DoH. (2005c) "Independence, Well-being and Choice: Our vision for the future of Social Care for Adults in England". Social Care Green Paper. London: Department of Health, HMSO. Available at:

http://www.dh.gov.uk/en/Publicationsandstatistics/Publications/PublicationsPolicy AndGuidance/DH 4106477 . [Accessed:30/3/2007]

DoH. (2005d) "Supporting people with long term conditions: An NHS and social care model to support local innovation and integration". London: Department of Health, HMSO.

DoH. (2005e) "The National Service Framework for (Neurological) Long-term Conditions". London: Department of Health, HMSO.

DoH. (2005f) "Independence, Well-being and Choice: Our vision for the future of Social Care for Adults in England". Social Care Green Paper. London: Department of Health, HMSO. Available at:

http://www.dh.gov.uk/en/Publicationsandstatistics/Publications/PublicationsPolicy AndGuidance/DH\_4106477. [Accessed:30/3/2008]

DoH (2005) "Our health, our care, our say: a new direction for community services". London: Department of Health, HMSO. Available at:

http://www.dh.gov.uk/en/Publicationsandstatistics/Publications/PublicationsPolicy AndGuidance/DH\_4127453 [Accessed:25/3/2007]

DoH. (2006a) "Our health, our care, our say: a new direction for community services". London: Department of Health, HMSO. Available at: http://www.dh.gov.uk/en/Publicationsandstatistics/Publications/PublicationsPolicy AndGuidance/DH 4127453 [Accessed:30/3/2007]

DoH. (2006b) "Supporting people with complex health needs to self care: A guide to developing local strategies and good practice". London: Department of Health, HMSO.

DoH. (2007a) "Commissioning framework for health and well-being". London: Department of Health, HMSO. Available at:

http://www.dh.gov.uk/en/Publicationsandstatistics/Publications/PublicationsPolicy AndGuidance/DH\_072604 [Accessed:30/3/2007]

DoH. (2007b) "Putting People First, A shared vision and commitment to the transformation of adult social care". London, Department of Health, HMSO. Available at:

http://www.dh.gov.uk/en/Publicationsandstatistics/Publications/PublicationsPolicy AndGuidance/DH\_081118 [Accessed:24/1/2008]

DoH. (2007c) "The Expert Patient Programme". London: Department of Health, HMSO.

http://www.dh.gov.uk/en/Aboutus/MinistersandDepartmentLeaders/ChiefMedical Officer/ProgressOnPolicy/ProgressBrowsableDocument/DH\_4102757 [Accessed:4/4/2008]

DoH. (2007d) "National Service Framework for Complex health needs- Good Practice Guide". London: Department of Health, HMSO.

DoH. (2008a) "Long term conditions compendium of information". London, Department of Health, HMSO. Available at:

www.dh.gov.uk/en/Publicationsandstatistics/Publications/PublicationsPolicyAndGuidance/DH\_082069 [Accessed:3/3/2009].

DoH. (2008b) "High Quality Care For All, NHS Next Stage Review Final Report". London: Department of Health, HMSO.

DoH. (2008d) "The future regulation of health and adult social care in England. A consultation on the framework for regulation of health and adult social care providers". London: Department of Health, HMSO. Available at:

http://www.dh.gov.uk/en/Consultations/Liveconsultations/DH\_0836 25 [Accessed:19/5/2008].

DoH. (2009) "The Quality and Outcomes Framework".

http://www.dh.gov.uk/en/Healthcare/Primarycare/Primarycarecontracting/QOF/index.htm [Accessed: 21/6 /2009]

DoH. (2010) "Improving the health and well-being of people with long-term conditions". London: Department of Health, HMSO

DoH. (2011) "NHS Outcomes Framework". London: Department of Health, HMSO.

DoH. (2012a) "Public Health Outcomes Framework". London: Department of Health, HMSO.

DoH. (2012b) "Long Term Conditions Compendium of Information, (3rd edn) London: Department of Health, HMSO. Available at: www.dh.gov.uk/prod\_consum\_dh/groups/dh\_digitalassets/@dh/@en/documents/digitalasset/dh\_134486.pdf [Accessed:14/2/2013].

DoH. (2012c). "Adult Social Care Outcomes Framework". London: Department of Health, HMSO.

DoH. (2014) "Wellbeing Why it matters to health policy". London: Department of Health, HMSO. Available at:

http://www.vision2020uk.org.uk/core\_files/DOH\_Wellbeing\_why\_it\_matters\_to\_Health\_Policy\_\_January\_2014\_.pdf [Accessed: 18/5/2014]

Department for Work and Pensions and Department of Health (DWP and DoH) (2005) "Health, Work and Well-being – caring for our future". London: Department of Health, HMSO.

Department for Work and Pensions (2009) "Opportunity Age Indicators: 2008 update". London: Department of Work and Pensions, Older People and Ageing Society Division.

Devereux, S. (2001) "Sen's Entitlement Approach: Critiques and Counter-critiques". *Oxford Development Studies*, 29, (3) 1-20.

Devine, J., Camfield, L., Gough, I. (2006) "Autonomy or Dependence or Both? Perspectives from Bangladesh". ESRC Group on Well-Being in Developing Countries (WeD) Working Paper 13. Available at:

http://www.welldev.org.uk/research/workingpaperpdf/wed13.pdf [Accessed: 1/7/2008]

DeWall, C. N., Pond, R. S., Campbell, W. K., and Twenge, J. M. (2011) "Turning in to psychological change: Linguistic markers of psychological traits and emotions over time in popular U.S. song lyrics". *Psychology of Aesthetics, Creativity, and the Arts*, 5, 200-207

Dewe, P., Cooper, C. (2012) "Wellbeing and work: Towards a balanced agenda". Basingstoke: Palgrave MacMillan.

Dewey, J. (1933) "How We Think". Boston: Heath & Co.

Diener, E. (1984) "Subjective well-being". Psychological Bulletin, 95, 542-575.

Diener, E. (2006) "Guidelines for national indicators of subjective well-being and ill-being". *Journal of Happiness Studies*, 7 (4), 397–404.

Diener E. (2009) "Assessing well-being: the collected works of Ed Diener". New York: Springer.

Diener, E. (2009b) "Subjective well-being". In M., Eid and R., J. Larsen, (eds) 'The science of subjective well-being'. New York: The Guilford Press.

Diener, E., and Larsen, R., J. (1993) "The experience of emotional wellbeing". In M. Lewis and J. M. Haviland (eds) 'Handbook of emotions'. New York: Guilford Press.

Diener, E., Sandvik, E., Seidlitz, L., and Diener, M. (1993b) "The relationship between income and subjective well-being: Relative or absolute?" *Social Indicators Research*, 28,195–223.

Diener, E., and Suh, E. (1997) "Measuring Quality of Life: Economic, Social and Subjective Indicators". *Social Indicators Research*, 40, 189-216.

Diener, E., and Lucas, R. (1999) "Personality and subjective wellbeing". In D., Kahneman, E., Diener, and N. Schwartz, (eds), 'Wellbeing: the foundations of hedonic psychology'. New York: Russell Sage Foundation.

Diener, E., Suh, E., M., Lucas, R., E., and Smith, H., E. (1999) "Subjective wellbeing: three decades of progress". *Psychological Bulletin*, 125, 276 – 302.

Diener, E., and Lucas, R., E. (2000) "Explaining differences in societal levels of happiness: Relative standards, need fulfilment, culture and evaluation theory". *Journal of Happiness Studies*,1,41–78.

Diener, E., and Suh, E. (eds) (2000) "Culture and subjective well-being". Cambridge, MA: MIT Press.

Diener, E., and Biswas-Diener, R. (2002) "Will Money Increase Subjective Well-Being?". *Social Indicators Research*, 57 (2), 119-169.

Diener, E., and Seligman M., E., P. (2002) "Very happy people". *Psychological science*, 13 (1), 81-84.

Diener, E., Oishi, S., and Lucas, R. (2003) "Personality, Culture, and Subjective Well-Being: Emotional and Cognitive Evaluations of Life". *Annual Review of Psychology* 54, 403-425.

Diener, E., and Seligman, M., E., P. (2004) "Beyond money: Toward an economy of well-being". *Psychological Science in the Public Interest*, 5, 1-31.

Diener, E., Lucas, R., E., and Scollon, C. (2006) "Beyond the hedonic treadmill: revising the adaptation theory of well-being". *Psychological Science*, 61, 305-14.

Diener, E., and Biswas-Diener, R. (2006) "Will Money Increase Subjective Well-Being?: A Literature Review and Guide to Needed Research" *Journal of Happiness Studies*, 7 (4),397–404.

Diener, E. and Biswas-Diener, R. (2008) "Happiness: Unlocking the Mysteries of Psychological Wealth". Oxford: Wiley-Blackwell.

Diener, E., Lucas, R.,E., and Schimmack, U. (2008) "National Accounts of Well-Being". Oxford: Oxford University Press.

Diener, E., Lucas, R., Schimmack, U., and Helliwell, J. (2009) "Well-being for public policy". New York: Oxford University Press.

Diener, E., Helliwell, J., and Kahneman, D. (eds) (2010) "International Differences in Well-Being" Oxford: Oxford University Press.

Dinham, A. (2006) "Raising expectations or dashing hopes?: Well-being participation in disadvantaged areas". *Community Development Journal* 42 (2): 181-193.

Director, B. (2004) "Using an action research approach to involve customers in the assessment of professional competence". *European Journal of Social Work*, 8 (2), 165 – 179.

Di Clemente, R., Crosby, R., and Kegler, C. (2002) "Emerging Theories in Cockerham". Cambridge: Polity Press.

Di Tella, R., MacCulloch, R., and Oswald, A. (2003) "The macroeconomics of happiness". *The Review of Economics and Statistics*, 85(4), 809-827.

Di Tella, R., and MacCulloch, R. (2005) "Partisan social happiness". *Review of Economic Studies*, 72(2), 293–367.

Dixon, C. (2005a) "New Historicism". Oxford: Routledge.

Dixon, C. (2005b) "Important people in New Historicism". Oxford: Routledge.

Dodge, R., Daly, A., Huyton, J., Sanders, L. (2012) "The challenge of defining wellbeing". *International Journal of Wellbeing*, 2(3), 222-235.

Dolan, P., Peasgood, T., and White, M. (2006a) "Review of research on the influences on personal wellbeing and application to policy making". London: Defra.

Dolan, P., Peasgood, T., Dixon, A., Knight, M., Philips, D., Tsuchiya, A. and White, M.(2006b) "Research on the relationship between wellbeing and sustainable development". London: Defra.

Dolan, P., and White, M. (2007) "How can measures of subjective well-being be used to inform public policy?". *Perspectives on Psychological Science*, 2(1), 71 -79.

Dolan, P., Peasgood, T., and White, M. (2008) "Do we really know what makes us happy? A Review of the economic literature on the factors associated with subjective well-being". *Journal of economic psychology* 29, 94-122.

Dolan, P., Layard, R., and Metcalfe, R. (2011) "Measuring Subjective Well-being for Public Policy". London: Office for National Statistics (ONS).

Dolan, P., and Metcalfe, R, (2011) "Comparing measures of subjective well-being and views about the role they should play in policy". London: Office for National Statistics (ONS).

Donnal C., W. (2006) "Defining Wellbeing". *Philosophy of the Social Sciences*, 36 (3), 330-354.

Donovan, N., and Halpern, D. (2002) "Life Satisfaction: The State of Knowledge and Implications for Government." London: Strategy Unit, Cabinet Office.

Dooris, M. (2004) "Joining up settings for health: a valuable investment for strategic partnerships?". *Critical Public Health*, 14 (1), 49-61

Dooris, M. (2006) "Healthy settings: challenges to generating evidence of effectiveness". *Health Promotion International*, 21 (1), 55-65

Dorfman, L., Wallack, L., Woodruff, K. (2005) "More Than a Message: Framing Public Health Advocacy to Change Corporate Practices". *Health Education and Behavior*, 32 (3), 320-336.

Doyal, L., and Gough, I. (1991) "A Theory of Human Need". London: MacMillan.

Doyal, L. (1983) "The Political Economy of Health". London: Pluto Press.

Drake, L., Duncan, E., Sutherland, F., Abernethy, C., and Henry, C. (2008) "Time perspective and correlates of wellbeing". *Time and Society*, 17 (1), 47–61

Drost, E., A. (2011) "Validity and Reliability in Social Science Research". *Education Research and Perspectives*, 38 (1), 105-123.

Duffy, J., and Mc Keever, B. (2011) "Looking out from the middle: influencing policy change through user involvement". In M., Barnes and P. Cotterell 'Critical Perspectives on User Involvement'. Bristol: Policy Press.

Duany, A., Plater-Zyberk, E., and Speck, J (2001) "Suburban Nation: The Rise of Sprawl and the Decline of the American Dream". New York: North Point Press.

Duncan, P. (2007) "Critical Perspectives on Health". Basingstoke: Palgrave Macmillan.

Dunnell, K. (2008) "Ageing and mortality in the UK: National statistician's annual article on the population". Population trends 134, 6–23.

Durayappah, P. (2010) "The 3P Model: A General Theory of Subjective Well-Being". *The Journal of Positive Psychology*, 5(1), 24-40.

Durham Community Research Team (2011) "Community-based participatory research: Ethical challenges". Available at:

www.dur.ac.uk/resources/beacon/CCDiscussionPapertemplateCBPRBanksetal7Nov 2011.pdf [Accessed: 1/7/2015].

Dutta, M. (2008) "Communicating Health: A culture-centred approach". Cambridge: Polity Press.

Earl-Slater, A. (2004) "Lay Involvement in Health and Other Research". London: Radcliffe Publishing.

Earle-Slater, A., Lloyd, C. Sidell, M., and Spur, S. (eds) (2007) "Theory and Research in Promoting Public Health". London: Sage.

Easterby-Smith, M., Thorpe, R., Lowe, A. (1991) "Management Research: An Introduction". Sage: London.

Easterlin, R.A. (1974) "Does economic growth improve the human lot? Some Empirical Evidence". In P., A., David and W., R. Melvin, (eds) "Nations and Households in Economic Growth: Essays in Honor of Moses Abramovitz". New York: Academic Press.

Easterlin, R., A.(1995) "Will Raising the Incomes of All Increase the Happiness of All?". *Journal of Economic Behavior and Organization*, 27(1), 35-47.

Easterlin, R., A. (1996) "Growth triumphant: the twenty-first century in historical perspective". Michigan: University of Michigan Press.

Easterlin, R., A. (2000) "The Worldwide Standard of Living since 1800". *Journal of Economic Perspectives*, 14 (1), 7-26.

Easterlin, R., A. (2001) "Income and Happiness: Towards a Unified Theory". *The Economic Journal*, 111(473), 465-484.

Easterlin, R., A. (2005a) "Feeding the Illusion of Growth and Happiness: A Reply to Hagerty and Veenhoven". *Social Indicators Research*, 74(3), 429 – 443.

Easterlin, R., A. (2005b) "Building a better theory of well-being". In L. Bruni and P. Porta, (eds) 'Economics and Happiness: Framing the Analysis'. Oxford: Oxford University Press.

Easterlin, R., A. (2006) "Life cycle happiness and its sources. Intersections of psychology, economics and demography". *Journal of Economic Psychology*, 27, 463–482.

Easterlin. R., A. (2009) "Lost in Transition: Life Satisfaction on the Road to Capitalism". *Journal of Economic Behavior and Organization*, 71 (2), 130–145.

Easterlin, R., A., McVey, L. A. Switek, M.; Sawangfa, O.; Zweig, J. S. (2010) "The happiness-income paradox revisited". Proceedings of the National Academy of Sciences, 107 (52).

Ebstein, R., P. (1996) "Dopamine D4 receptor (D4DR) exon 111polymorphism associated with the human personality trait of Novelty seeking". *Nature Genetics*, 12, 78-80.

Eckermann, L. (2000) "Gendered indicators of Health and wellbeing: Is QoL gender neutral?" *Social Indicators Research*, 52, 29-54

Eigner, S. (2001) "The Relationship Between 'Protecting the Environment' as a Dominant Life Goal and Subjective Wellbeing". In P., Schmuck, and K.,M. Sheldon, (eds) 'Life Goals and Wellbeing: Towards a Positive Psychology of Human Striving', Seattle: Hogrefe and Huber.

Eckersley, R. (2004) "Well and Good: how we feel and why it matters". Melbourne: Text publishing.

Edelman, N., Cassell, J., Ward, L. (2007) "Engaging lay people in research on sensitive topics: A Case Study". London: INVOLVE.

Edwards, C., and Imrie, R. (2008) "'Disability and Implications of the Wellbeing Agenda: Some Reflections from the United Kingdom". *Journal of Social Policy*, 37, (3), 337-355.

Edwards-Jones, G., Davies, B., Salman, S., Hussain, J. (2000) "Ecological Economics: An Introduction to Manfred Max-Neef's typology of needs and satisfiers". London: Blackwell Science Ltd.

Ehrenreich, B.(2010) "Bright sided: How the relentless promotion of positive thinking has undermined America". New York: Henry Holt.

Ehrenreich, B. (2010) "The delusion of positive thinking" *Therapy today*, 21 (4), 10-17.

Ekins, P., and Max-Neef, M. (eds) (1992) "Real-Life Economics: Understanding Wealth Creation". London: Routledge.

Elsbach, K., D., and Kramer, R., M. (1996) "Members' responses to organizational identity threats: Encountering and countering the Business Week rankings". *Administrative Science Quarterly*, 41, 442-476.

El-Shaarawi, N. (2015) "Living an Uncertain Future: Temporality, Uncertainty, and Well-Being among Iraqi Refugees in Egypt". Social Analysis 59, (1), 38-56 (19).

Emmons, R., A. (1986) "Personal strivings: An approach to personality and subjective well being". *Journal of Personality and Social Psychology*, 51, 1058-1068.

Emmons, R. A. (1996) "Striving and feeling: Personal goals and subjective well-being". In P., M., Gollwitzer and J., A. Bargh, (eds) 'The psychology of action: Linking cognition and motivation to behavior'. Guilford Publications: New York.

Endacott, J., L. (2010) "Reconsidering affective engagement in historical empathy". *Theory and Research in Social Education*, 38(1), 6-49.

English Longitudinal Study of Ageing: Wave 0 (1998,1999 and 2001) and Waves 1-4 (2002-2009). Colchester, Essex: UK Data Archive.

Entwistle, V., Renfrew, A., Yearley, M.,J., Forrester, S., Lamont, T. (1998) "Lay perspectives: advantages for health research". *BMJ*, 316: 463–466.

Epps, P. (2014) "Historical linguistics and socio-cultural reconstruction". Available at:

https://www.researchgate.net/profile/Patience\_Epps/publication/292655743\_Hist orical\_linguistics\_and\_socio-

cultural\_reconstruction/links/571df68d08aeaced7889c277.pdf [Accessed 3/6/15].

Eraut, G., and Whiting, R. (2008) "What do we mean by wellbeing? And why might it matter?" Research report DCSF-RW073, Linguistic Landscapes, Department for Children, Schools and Families, (DCSF), HMSO, London

Esping-Andersen, G. (1990) "The Three Worlds of Welfare Capitalism". Princeton: Princeton University Press.

Esping-Andersen, G. (1999) "Social Foundations of Post-Industrial Economies". New York: Oxford University Press.

Evans, C., and Jones, R. (2004) "Engagement and empowerment, research and relevance: Comments on user-controlled research". *Research Policy and Planning*, 22 (2), 5-13.

Evans, D. and Jackson, T. (2007) "Towards a Sociology of Sustainable Lifestyles", RESOLVE Working Paper 03-07 Research Group on Lifestyles, Values and the Environment. Surrey: University of Surrey.

Ewels, L., and Simnett, I. (2001) "Promoting Health: a practical guide" (4<sup>th</sup> edn) London: Bailliere Tindall.

Fadem, B.(2009) "Behavioral Science". New York: Lippincott Williams and Wilkins.

Fahey, T., Nolan, B., Whelan, C. (2003) "Monitoring quality of life in Europe: European Foundation for the Improvement of Living and Working Conditions". Brussels: Office for official publication of the European Communities.

Fallowfield, L. (2002) "Quality of life: a new perspective for cancer patients". *National Review of Cancer*, 2, 873–879.

Farquhar, M. (1995) "Elderly People's Definitions of Quality of Life". *Social Science and Medicine*, 41: 1439-1446.

Farquhar, M. (1995b) "Definitions of Quality of Life: a Taxonomy". *Journal of Advanced Nursing*, 22 (3), 498 -520.

Farquhar, M. (1996) "Older People's Definitions of Quality of Life and the Relevance of Various Scales" Available at:

https://www.rcn.org.uk/\_\_data/assets/pdf\_file/0011/414659/Farquhar\_M\_comple te.pdf [Accessed: 2/11/ 2012].

Faulkner, A. (2004) "The ethics of survivor research. Guidelines for the ethical conduct of research carried out by mental health service users and survivors". Bristol: The Policy Press. Available at:

http://www.jrf.org.uk/publications/browse/category/u#user-involvement [Accessed: 27/12/ 2011].

Faulkner, A. (2010) "Changing Our Worlds: Examples of user controlled research in action". Eastleigh: INVOLVE. Available at:

http://www.invo.org.uk/pdfs/6823\_INVOLVE\_UCRCS\_brochure\_WEB.pdf [Accessed:28th August 2015]

Faulkner, A. (2015) "Payment for involvement". Eastleigh: INVOLVE.

Faulkner, A., and Thomas, P. (2002) "User-led research and evidence-based medicine". *The British Journal of Psychiatry*, 180, 1 – 3.

Felce, D. (1997) "Defining and applying the concept of quality of life". *Journal of Intellectual Disability*, 41(2), 126-135.

Felce, D., and Perry, J. (1995) "Quality of life: its definition and measurement". *Research in Developmental Disabilities*, 16 (1), 51-74.

Felce, D., and Perry, J. (1996) "Assessment of quality of life". In R., L., Schalock, and G., N. Siperstein, (eds) 'Quality of life volume I: Conceptualization and measurement'. Washington DC: American Association on Mental Retardation.

Felluga, D. (2003) "General Introduction to New Historicism: Introductory Guide to Critical Theory". (3rd edn) Upper Saddle River, NJ: Prentice Hall,

Ferguson, I. (2007) "Neoliberalism, happiness and wellbeing". *International Socialism*, 117, 1-15.

Fernandez, R., and Cornes, J. (2009) "A critical review of history and current status of positive psychology". *Annuary of Clinical and Health Psychology*, 5, 7-13.

Ferrell, B., Grant, M., Chan, J., Ahn, C., Ferrell, B. A. (1995) "The impact of cancer pain education on family caregivers of elderly patients". *Oncology Nursing Forum*, 22 (8), 1211-1218.

Ferrer-i-Carbonell, A. (2005) "Income and well-being: An empirical analysis of the comparison income effect". *Journal of Public Economics*, 89, 997–1019.

Findlay, A., M., Gayle, V., van Ham, M., and Nowok, B. (2011) "Does Migration Make You Happy? A Longitudinal Study of Internal Migration and Subjective Well-Being". IZA Discussion Paper 10.

Figueras, J., and McKee, M. (eds) (2012) "Health Systems, Health, Wealth and Societal Well-being: Assessing the case for investing in health systems". Maidenhead: Open University Press. Available at: http://www.euro.who.int/\_\_data/assets/pdf\_file/0007/164383/e96159.pdf [Accessed:22/3/2014]

Finlay, L. (2002) "Negotiating the swamp: the opportunity and challenge of reflexivity in research practice". *Qualitative Research* 2: 209, Available at: http://grj.sagepub.com/content/2/2/209 [Accessed: 30/3/2009]

Finlayson, G. (1994) "Citizen, State, and social welfare in Britain, 1830-1990". Oxford: Clarendon Press.

Fisher, P. (2008) "Wellbeing and empowerment: the importance of recognition". *Sociology of Health and Illness*, 30(4), 583–598,

Flavell, L., and R (1999) "The chronology of words and phrases". London: Kyle Cathie Ltd.

Fleck, J. R. (1981) "Dimensions of personal religion: A response to J. R. Fleck and J. D. Carter" (eds), 'Psychology and Christianity'. New York: Harper Row.

Fleuret, S., and Atkinson, S. (2007) "Wellbeing, health and geography: A critical review and research agenda". *New Zealand Geographer*, 63, (2) 106–118.

Flick, U. (2009) "An Introduction to Qualitative Research". (4th edn) London: Sage.

Flint, C., and Taylor, P., J. (2007) "Political Geography: World economy, Nationstate, and Locality". New Jersey: Pearson/Prentice Hall.

Foresight Mental Capital and Wellbeing Project (2008) "Foresight Mental Capital and Wellbeing Project: Final Project Report". London: The Government Office for Science.

Foresight Mental Capital and Wellbeing Project (2008) "Mental Capital and Wellbeing: Making the most of ourselves in the 21st century". London: The Government Office for Science.

Forgeard, M., J., C., Jayawickreme, E., Kern, M. and Seligman, M., E., P., (2011) "Doing the right thing: Measuring wellbeing for public policy". *International Journal of Wellbeing*, 1(1), 79–106.

Forth, C. E. (2009) "Well-being: A Cultural History of Healthy Living". *Social History of Medicine*, 22(2), 413-414. Available at: http://shm.oxfordjournals.org/content/22/2/413.extract [Accessed:7/2/2014]

Fortin, M., Bravo, g., Hudon, C., Lapointe, L., Almirall, J., Dubois, M., and Vanasee, A. (2006) "Relationship between multimorbidity and health related qol of patients in primary care." *International Journal of QOL Aspects of Treatment, Care and Rehabilitation*, 15 (1), 83-91.

Fortin, M., Bravo, G., Haggerty, G., Hudon, C. (2010) "Prevalence estimates on multi-morbidities: Comparative study of two sources". *British Medical Council Health Services Research*, 10 (1).

Foucault, M. (1997) "Ethics: Subjectivity and Truth". (ed) P. Rabinow. New York: New Press.

Foucault, M. (1991) "Governmentality". Translated by R. Braidotti (1991), revised by C. Gordon. In G., Burchell, C., Gordon and P. Miller (eds) 'The Foucault Effect: Studies in Governmentality'. Chicago: University of Chicago Press.

Foucault, M. (1982) "Technologies of the Self". In Huck, L., Gutman, M., and Hutton, P.,H. (eds) Technologies of the Self: A Seminar with Michel Foucault, Amherst: The University of Massachusetts Press.

Foucault, M. (1997) "Ethics: Subjectivity and Truth". (Ed) Rabinow, P. New York: New Press.

Foucault (2002) "Governmentality and Critique". Available at:http://www.andosciasociology.net/resources/Foucault\$2C+Governmentality\$2C +and+Critique+IV-2.pdf [Accessed:5/5/2010]

Foucault, M. (2008) "The Birth of Biopolitics Lectures At The College de France" (1978-1979). Available at: http://rauli.cbs.dk/index.php/foucault-studies/article/view/3127/3299.pdf [Accessed:5/5/2010]

Foucault, M. (2010) "The Government of Self and Others". Lectures at the Collège de France 1982-1983". Available at: http://rauli.cbs.dk/index.php/foucault-studies/article/view/3127/3298.pdf [Accessed:5/5/2010]

Foucault, M. (2011) "The Courage of Truth". Lectures at the Collège de France 1983-1984 (2011). Available at: http://rauli.cbs.dk/index.php/foucault-studies/article/view/3128/1298.pdf [Accessed:5/5/2010]

Frey, B., and Stutzer, A. (2002) "Happiness and Economics: How the Economy and Institutions Affect Human Well-Being". Princeton: Princeton University Press.

Freire, P. (1996) "Pedagogy of the Oppressed". (3rd edn) London: Penguin.

Freund, P. (2010) "Capitalism, Time-Space, Environment, and Human Well-Being: Envisioning Ecosocialist Temporality and Spatiality". Capitalism Nature Socialism 21 (2) Available at:

http://www.tandfonline.com/doi/abs/10.1080/10455752.2010.489684. [Accessed:17/11/ 2015].

Frey, B., and Stutzer, A. (1999) "Measuring preferences by subjective well-being". *Journal of Institutional Theoretical Economic*, 155, 755–78.

Frey, B., and Stutzer, A. (2002) "Happiness and Economics: How the Economy and Institutions Affect Well-being." New Jersey: Princeton University Press.

Frie, R. (2012) "Psychological Agency: Theory, Practice, and Culture". Bristol: Policy Press.

Friedichson, G., W., S. (1966) "The Oxford dictionary of English Etymology". Oxford University Press: Oxford.

Froggert, L. (2002) "Love, Hate and Social welfare: Psychosocial Approaches to policy and practice". Bristol: Policy Press.

Fullager, S. (2002) "Governing the Healthy Body: Discourses of Leisure and Lifestyle within Australian Health Policy". *Health*, 6 (1), 69-84.

Fullagar, S. (2005) "The paradox of promoting help-seeking: a critical analysis of risk, rurality and youth suicide". *International Journal of Critical Psychology*, 14, 31–51.

Fullager, S. (2009) "Governing healthy family lifestyles through discourses of risk and responsibility". *Biopolitics and the 'obesity epidemic'*, 108-126.

Furedi, F. (2004) "Therapy Culture: Cultivating Vulnerability in an uncertain age". London: Routledge.

Furedi, F. (2005) "The Politics of Fear. Beyond Left and Right". New York: Continuum International Publishing Group.

Furedi, F. (2006) "Why the politics of happiness makes me mad". Available at: http://www.frankfuredi.com/articles/happiness-spiked-20060523.shtml [Accessed:3/3/2011].

Furedi, F. (2008) "Fear and Security: A Vulnerability-led Policy Response". *Social Policy and Administration*, 42 (6), 645-661.

Furedi, F. (2009) "Medicalisation in a Therapy Culture". In D. Wainwright, (ed) 'A Sociology of Health'. London: Sage.

Gabriel, Z and Bowling, A. (2004) "Quality of life from the perspectives of older people". *Ageing and Society*, 24 (5), 675 - 691.

Gail, M., and Benicho, J. (2000) (eds) "Encyclopedia of Epidemiologic Methods". London: John Wiley and Sons.

Gaillard, G. (2004) "The Routledge Dictionary of Anthropologists". Translated by P., J. Bowman, 1997 (ed). London and New York: Routledge.

Gale, C., R., Booth, T., Mõttus, R., Kuh, D., Deary, I. (2014) "Neuroticism and Extraversion in youth predict mental wellbeing and life satisfaction 40 years later". *Journal of Research in Personality*, 47 (6), 687-697.

Galloway, S. (2006) "QoL and well-being: Measuring the benefits of culture and sport". Education Research Programme Research Findings series number 12. Edinburgh: Scottish Education Department.

Galvin, K., Todres, L. (2013) "Caring and Well-being: A Lifeworld Approach". London: Routledge.

Gardner, P. (2010) "Hermeneutics, History and Memory". Oxford: Routledge.

Gartner, J. (1996) "Religious commitment, mental health, and pro-social behavior: A review of the empirical literature". In E., P. Shafranske, (ed) 'Religion and the clinical practice of psychology'. Washington: American Psychological Association.

Gaskell, M., G., and Dumay N. (2003) "Lexical competition and the acquisition of novel words". *Cognition*, 89, 105-132.

Gasper, D. (2002) "Is Sen's Capability Approach an Adequate Basis for Considering Human Development?" *Review of Political Economy*, 14 (4): 435-461.

Gasper, D. (2004b) "Human Well-being: Concepts and Conceptualizions". Working paper 388, Institute of Social Studies, The Hague, and 'WIDER' Working paper for project on Measuring Well-being, Helsinki: 'WIDER'.

Gasper, D. (2007a) "Human Rights, Human Needs, Human Development, Human Security –Relationships between four international 'human' discourses". Paper presented at: *Forum for Development Studies*. NUPI, Oslo, January 4<sup>th</sup> 2007.

Gasper, D. (2007b) "From 'Hume's Law' To Policy Analysis For Human Development – Sen after Dewey". *Review of Political Economy*, 20 (2), 233-256.

Gasper, D. (2007c) "Conceptualising human needs and wellbeing". In I., Gough and J. McGregor, (eds) 'Wellbeing in Developing Countries: From Theory to Research'. Cambridge: Cambridge University Press.

Gasper, D. (2010) "Understanding the diversity of conceptions of well-being and quality of life". *The Journal of Socio-Economics*, 39, 351–360.

Gedžūne E., A. (2011) "Educational Action research to generate ecological wisdom of insight for inclusion and sustainability". *Discourse and Communication for Sustainable Education*, 2 (1), 5–31.

Gerard, R. (2002) "The EU: The European Miracle in Reverse". Historical Notes No. 41. London: An occasional publication of the Libertarian Alliance. Available at: www.libertarian.co.uk/lapubs/histn/histn041.pdf [Accessed 15/6/12].

Gergen, K. (1985) "The social constructionist movement in modern psychology". *American Psychologist*, 40 (3), 266-275.

Gergen, K. '(2001) "Social constructionism in context". London: Sage.

Gewirth, A (1985) "Why There Are Human Rights". Social Theory and Practice 11 (2), 235-248.

Gibb, K. (2014) "The multiple policy failures of the UK bedroom tax: International Journal of Housing Policy". *International Policy of Housing Journal*, 15 (2) 148-166.

Gibb, K., Maclennan, D. and Stephens, M. (2013) "Innovative financing of affordable housing: international and UK perspectives". York: Joseph Rowntree Foundation.

Gibson, C., H. (1991) "A concept analysis of empowerment". *Journal of Advanced Nursing*, 5, 51-57.

Giddens, A. (1976) "New Rules of Sociological Method: a Positive Critique of interpretative Sociologies". London: Hutchinson.

Giddens, A. (1984) "The Constitution of Society. Outline of the Theory of Structuration". Cambridge: Polity Press.

Giddens, A. (1991) "Modernity and Self-Identity: Self and Society in the Late Modern Age". Cambridge: Polity Press.

Gilbert, D., T., and Wilson, T., D. (2007) "Prospection: Experiencing the future". *Science*, 317, 1351–1354.

Giles-Corti, B., and Donovan, R., J. (2002) "The relative influence of individual, social and physical environment determinants of physical activity". *Social Science and Medicine* 54 (12), 1793–1812.

Gilroy, R. (2007) "Taking a capabilities approach to evaluating supportive environs for older people". *Applied research in QoL*, 1 (3), 343-356.

Gilroy, R. (2008) "Places that support human flourishing; lessons from later life". *Planning theory and planning* 9(2), 145-163.

Giroux, H. (2013) Youth in Revolt: Reclaiming a Democratic Future, Boulder, CO: Paradigm Publishers.

Glaser, B., G. (2001) "The Grounded Theory perspective. Conceptualization contrasted with description". Mill Valley, CA: Sociology Press.

Glaser, B., and Strauss, A. (1967) "The Discovery of Grounded Theory: Strategies for Qualitative Research". Chicago: Aldine Publishing Company.

Glasgow Centre for Population Health (2013) "The built environment and health: an evidence review". Briefing paper 11. Available at: http://www.gcph.co.uk/assets/0000/4174/BP\_11\_\_Built\_environment\_and\_health\_-\_updated.pdf

Glasper, A., and Rees, C. (2012) "How To Write Your Nursing Dissertation". UK: Wiley-Blackwell.

GMCA (2015) "Taking Charge of Our Health and Social Care". Available at: http://www.gmhealthandsocialcaredevo.org.uk/assets/GM-Strategic-Plan-Final.pdf [Accessed: 25/1/ 2016].

Göktürk, E. (2007) "What is "paradigm"? Available at: http://folk.uio.no/erek/essays/paradigm.pdf [Accessed: 25/1/ 2016].

Gonzalez-Guerra, M., Bergdolt, K., and Lev, E. (2010) "Further Comments on Teaching the History of Medicine 2". Available at: http://www.bium.univ-paris5.fr/ishm/buda2006/medical%20humanities.pdf [Accessed:7/2/2014]

Goodley, D., Lawthom, R. (eds) (2006) "Disability and Psychology: Critical Introductions and Reflections". Basingstoke: Palgrave MacMillon.

Gordon, C. (1991) "Governmental rationality: an introduction". In G., Burchel and C., Gordon and, P. Miller (eds) 'The Foucault Effect: Studies in Governmentality'. Chicago: University of Chicago Press.

Gough, I. (2004) "Human well-being and social structures: relating the universal and the local". *Global Social Policy*, 4 (3), 289-311

Gough, I. (2005) "Theories of Wellbeing". Available at: http://www.bath.ac.uk/soc-pol/welldev/spa-presentations/theories-of-wellbeing-text.pdf [Accessed: 25/1/2016].

Graddol, D., Leith, D., and Swann, J. (eds) (1996) "English – history, diversity and change". London: Routledge.

Graham, H. (2000) "Understanding Health Inequalities". Buckingham: Open University Press.

Graham, P.(2002) "Does inequality matter to individual welfare?". *Journal of economic Inequality*, 4(1), 107-122.

Graham, J., and Haidt, J. (2010) "Beyond beliefs: Religions bind individuals into moral communities". *Personality and Social Psychology Review*, 14, 140-150.

Graham, E., Atherton, J., R., and Steadman, T. (2010) "The practices of wellbeing: Political Economy, Religion and Wellbeing". London: Taylor Francis.

Graneheim, U.,H., and Lundman B. (2004) "Qualitative content analysis in nursing research: concepts, procedures and measures to achieve trustworthiness". *Nurse Education Today*, 24, 105–112.

Greater Manchester Combined Authority (2015) "Taking Charge of Our Health and Social Care". Available at:

file:///C:/Users/Wayne/Downloads/GM\_Strategic\_Plan\_Final.pdf [Accessed:2/1/2016].

Green, L., W., George, A., Mark, D. Frankish, C. (2003) "Guidelines for participatory research in health promotion". In M., Minkler and N. Wallerstein (eds) 'Community-based participatory research for health'. San Francisco: Jossey-Bass.

Griffin, J. (1986) "Well-being: Its meaning, measurement, and moral importance". Oxford: Clarendon Press.

Griffin, J. (2010) "The Lonely Society"? London: Mental Health Foundation

Griggs, S. and Howarth, D. (2011) "Discourse and practice: using the power of well being". *Evidence and Policy*, 7(2), 213-226

Griva, F., Tseferidi, S. I., and Anagnostopoulos, F. (2015) "Time to get healthy: Associations of time perspective with perceived health status and health behaviours". *Psychology, health and medicine*, *20*(1), 25-33.

Gruber, J., Mauss, I. B., and Tamir. M. (2011) "A dark side of happiness? How, when and why happiness is not always good". *Perspectives on Psychological Science*, 6, 222-233.

Gu"ell, P., Orchard, M., Yopo, M., and Jime´nez-Molina, J. (2014) "Time Perspectives and Subjective Wellbeing in Chile". *Social Indicator Research* 1-15.

Guest, G., Bunce, A., Johnson, L. (2006). "How many interviews are enough? An experiment with data saturation and variability". *Field Methods*, *18* (1), 59-82.

Guite H., F., Clarke C., Ackrill, G. (2006) "The impact of the physical and urban environment on mental wellbeing". *Public Health*, 120, 1117-1126.

Gunn, S (2006) "History and Cultural Theory". London: Longman Pearson,

Guralnik, D., B. (ed) (1984) "Webster's New World Dictionary of the American Language". (2<sup>nd</sup>edn) New York: Simon and Schuster.

Haas, K., Simons, S. and Stevenson, N. (1998) "Older persons and horticulture therapy practice". In S., Simpson and M. Strauss, (eds) 'Horticulture as therapy: Principles and Practice'. New York: The food product press.

Haas, Barbara K. (1999a) "A Multidisciplinary Concept Analysis of Quality of Life". Western Journal of Nursing Research, 21(6).

Haas, Barbara K. (1999b) "Clarification and Integration of Similar Quality of Life Concepts". *Journal of Nursing Scholarship*, 31 (3).

Habermas, J. (1962) "Knowledge and Human Interest". Available at: http://www.marxists.org/reference/archive/habermas/1968/theory-knowledge.htm [Accessed:5/4/2008]

Hagerty, M. (2000) "Social comparisons of income in ones community: Evidence from national surveys of income and happiness". *Journal of Personality and Social Psychology*,78(4),764–771.

Hagerty, M. R. and Veenhoven, R. (2003) "Wealth and Happiness Revisited – Growing National Income Does Go with Greater Happiness". *Social Indicators Research*, 64, 1-27.

Haidt, J., and Graham, J. (2007) "When morality opposes justice: Conservatives have moral intuitions that liberals may not recognize". *Social Justice Research*, 20, 98-116.

Hall, E. (2010) "Spaces of wellbeing for people with learning disabilities". *Scottish Geography Journal*, 126, 275–284.

Hall, M and Havens, B. (2003) "Social Isolation and Social Loneliness". Division of Aging and Seniors: Health Canada. Available at: http://www.hc-sc.gc.ca/seniorsaines/naca/writings\_gerontology/writ18/writ18\_3\_e.htm [Accessed:22/7/2009].

Halpern, D. (1995) "Mental Health and the Built Environment: More Than Bricks and Mortar"? London: Taylor and Francis.

Halpern, D., Bates, C., Mulgan, G., Aldridge,S. (2004) "Personal Responsibility and Changing Behaviour: the state of knowledge and its implications for public policy". Prime Minister's Strategy Unit. Available at:

http://webarchive.nationalarchives.gov.uk/+/http:/www.cabinetoffice.gov.uk/media/cabinetoffice/strategy/assets/pr2.pdf [Accessed:30/8/2014].

Hamamura, T. (2012) "Are cultures becoming individualistic? A cross-temporal comparison of individualism-collectivism in the U.S. and Japan". *Personality and Social Psychology Review*, 16, 3-24.

Hamer, D., H. (1996) "The Heritability of Happiness". Nature Genetics, 14, 125-26.

Hamilton, P. (2003) "Historicism". (2<sup>nd</sup> edn) London: Routledge.

Hamilton, C., and Scullion, A. (2006) "Why do definitions matter? "Wellbeing, culture and sport". In S. Galloway, (2006) 'Education Research Programme Research Findings' series number 12. Edinburgh: Scottish Education Department.

Harding, S. (1986) "The Science Question in Feminism". Ithaca: Cornell University Press.

Harkness, V., Cameron, D., Latter, J., Ravat, M., and Bridges, L (2012) "Preparing for an Ageing Society: Evaluating the Ageing Well programme Parts 1 and 2". London: Department for Work and Pensions Research Report No 807.

Harper, D., J, (2002) "Talking about pictures: a case for photo elicitation". *Visual Studies*, 17 (1),13-26.

Harrison, P. (1990) "Religion' and the Religions in the English Enlightenment". Cambridge: Cambridge University Press.

Harrison, T., and Stuifbergen, A. (2005) "Ageing with a childhood onset disability". In J., Katz, S., Peace and S. Spurr, (eds) 'A life course perspective'. Bristol: Policy Press University.

Hart, C. (1999) "Doing a Literature Review: Releasing the Social Science Research Imagination". London: Sage.

Hartwell, R., M. (1961) "The Rising Standard of living in England 1800-1850". *Economic History Review*, 13, 377-416.

Hartwell, R., M. (1963) "The Standard of living in England". *Economic History Review*, 16, 135-146.

Hartwell, R., M. (1975) "Rising Standards of living". In A., J. Taylor, (ed) 'The Standard of Living in Britain in the Industrial Revolution'. London: Methuen and Co.

Hatch, G., (2007) "A life course approach to well-being". In J., Haworth and G. Hart, (eds) 'Well-being: Individual, Community and Social Perspectives'. London: Palgrave Macmillan.

Hatfield-Dodds S. (1988) "Beyond consumerism: Ideas of progress and well-being in the search for sustainable development". Environmental Series 6: Cambridge. University of Cambridge, Department of Land Economy.

Hausman, D., M. (2015) "Valuing Health: Wellbeing, Freedom and Suffering". Oxford University Press.

Havighurst, R., Neugarten, B., and Tobin, S. (1968) "Disengagement and patterns of aging". In B., N. Neugarten (eds) 'Middle age and aging'. Chicago: University of Chicago Press.

Haworth, J. (2007) "Work, Leisure and Wellbeing in Changing Social Conditions". In J., Haworth and G. Hart, (eds) 'Well-being: Individual, Community and Social Perspectives'. London: Palgrave Macmillan.

Haworth, J., and Hart, G. (eds) (2007) "Well-being: Individual, Community and Social Perspectives". London: Palgrave Macmillan.

Haybron, D. (2008) "The Pursuit of Unhappiness". Oxford: Clarendon Press.

Hayes, H., Buckland, S., Tarpey, M.(2012) "Briefing Notes for Researchers: Briefing Note 3; Why involve members of the public in research"? Eastleigh.

Hayes H. Buckland S. Tarpey M.(2012) "Briefing Notes for Researchers: Briefing Note 5; How to involve members of the public in research"? Eastleigh.

Heady B. (2007) "The Set-point Theory of Well-being Needs Replacing: On the Brink of a Scientific Revolution"? German institute of economic research Discussion paper 753.

Healey-Ogden, M., J., and Austin, W., J. (2012) "Uncovering the Lived Experience of Well-Being" *Qualitative Health Research*, 21(1) 85–96.

Hebert, R., S., Weinstein, E., Martire, L., M., Schulz, R. (2006) "Religion, spirituality and the well-being of informal caregivers: a review, critique, and research prospectus". *Aging and Mental Health*, 10(5), 497-520.

Heidegger, M. (1962) "Being and Time". Translated by J., Macquarrie and E. Robinson, 1962. London: SCM Press.

Held, D. and McGrew, A. (2007) "Globalization/Anti-Globalization: Beyond the Great Divide". (2nd edn) Cambridge: Polity Press.

Heinz-Herbert, N. (2011) "Social Indicators and Quality of Life Research: Background, Achievements and Current Trends". Available at: www.gesis.org/fileadmin/upload/institut/wiss\_arbeitsbereiche/soz\_indikatoren/Publikationen/isscnoll.pdf [Accessed: 2/4/ 2016].

Helgeson, V., S. (2003) "Social support and quality of life". *Quality of Life Research* 12, Suppl 1, 25-31

Helliwell, J., F. (2012) "Comparing the Happiness Effects of Real and On-line Friends". NBER Working paper No 18690. NBER Program.

Helliwell, J., F. (2012) "Understanding and Improving the Social Context of Well-Being". NBER Working paper No 18486. NBER Program.

Helliwell, J., F., and Putnam, R. (2004) "The social context of well-being". *Philosophical Transactions of the Royal Society London*, 359, 1435–1446.

Helliwell, J., F. and Putnam, R. (2006) "The social context of wellbeing". In F., Huppert, B., Keverne, and Baylis, N. (eds) "The Science of Wellbeing". Oxford: Oxford University Press.

Helliwell J.,F., and Huang, H. (2008) "How's your government? International evidence linking good government and well-being". *British Journal of Political Science*;38:595–619

Helliwell, J. F., and Wang, S. (2011) "Trust and Wellbeing". *International Journal of Wellbeing*, 1(1): 42–78.

Hemingway, A. (2011) "Lifeworld-led care: Is it relevant for well-being and the 5th wave of public health action"? *International Journal of Qualitative Studies in Health and Well-being*, 6 (4) 21-30.

Hemingway, A., and Steven, P. (2011) "Innovating to achieve sustainable wellbeing inside the built environment". *Perspectives in Public Health*, 131:117-118.

Henderson, S., and Petersen, A. (eds) (2002) "Consuming health: the commodification of health care". London: Routledge.

Hendry, F., and McVittie, C. (2004) "Is quality of life a healthy concept? Measuring and understanding life experiences of older people". *Qualitative Health Research*, 14(7), 961–975.

Henwood, K., and Pidgeon, N. (2001) "Talk about woods and trees: threat of urbanization, stability and biodiversity". *Journal of Environment Psychology*, 21, 125-147.

Heron, J. (1996) "Co-operative inquiry: Research into the human condition". London: Sage.

Heron, J., and Reason, P., (2006) "The practice of cooperative inquiry: research 'with' rather than 'on' people". In P., Reason, H. Bradbury, (eds) 'Handbook of action research: participative inquiry and practice'. Sage, London

Heron, J., Reason, P., (2008) Extending epistemology within a cooperative inquiry. In P., Reason, H. Bradbury, (eds) 'Handbook of action research: participative inquiry and practice', Sage, London.

Hewitt-Taylor, J. (2008b) "Young people with complex health needs: The organisation of support". *Primary Health Care*, 18(8), 36-39.

Hewitt-Taylor, J. (2010) "Supporting children with complex health needs". *Nursing Standard*, 19 (24), 50 -56.

Hibbard J., H., and Greene, J. (2013) "What the evidence shows about patient activation: better health outcomes and care experiences; fewer data on costs". *Health Affairs*, 32 (2), 207–214.

Hicks, S. (2011) The measurement of subjective well-being Paper for Measuring National Well-being Technical Advisory Group, Office for National Statistics. Available at file:///C:/Users/Wayne/Downloads/measurementwellbein\_tcm77-225913.pdf [Accessed:7/2/2014].

Hicks, S. (2011) "Spotlight on: Subjective Well-being, Office for National Statistics". Available at: http://www.statistics.gov.uk/articles/social\_trends/spotlight-on-subjective-wellbeing.pdf [Accessed:7/2/2014].

Higgs, M. (2008) "Emotional intelligence, well-being and personality: an empirical study of their interrelationship". Discussion Papers in Management Working Paper Series - Management, M-08-0, Southampton: University of Southampton.

Hilbert, D., R. (2014) "Scientific and Lay Knowledge: The Need for Communication and Integration". Available at:

http://www.eckerd.edu/academics/ford/files/12/DeborahHilbert.pdf [Accessed:30/8/2015]

Hiles, D., R. (2008) "Transparency". In L., M. Givens, (ed) 'The Sage Encyclopaedia of Qualitative Research'. Thousand Oaks: Sage.

Hill, C. (1985) "The collected essays of Christopher Hill: Religion and politics in 17th century England". Vol 2. Massachusetts: University of Massachusetts Press.

Hill, C. (1990) "A nation of change and novelty: Radical politics, religion, and literature in seventeenth century England". London: Routledge.

Hill, C. (1997) "Intellectual Origins of the English Revolution – Revisited". Oxford: Oxford University Press.

Hird S. (2003) "What is well-being? A brief review of current literature and concepts". Edinburgh: NHS Scotland.

Ho, L., S. (2005) "Happiness and public policy" Centre for Public Policy Studies: CPPS Working Paper Series. Paper 72.

Hobbs, G., J, Sixsmith, J. (2010) "Changing Service for People with Long-Term Neurological Conditions: Promoting Well-being within Service Provision". Manchester: Research Institute for Health and Social Change (RIHSC). Available at: http://www.rihsc.mmu.ac.uk/Publications\_order\_form.pdf [Accessed:3/5/2010].

Hobbes, T. (1968) "Thomas Hobbes", *Leviathan*, (ed) J.,C.,A. Gaskin,Oxford: Oxford University Press.

Hobsbawm, E., J. (1963) "The Standard of living during the Industrial Revolution: A discussion". *Economic History Review*, 16, 119-134.

Hobsbawm, E., J. (1971) "From Social History to the History of Society". *Daedalus*, 100 (1), *Historical Studies Today*, 20-45.

Hobsbawm, E., J. (1990) "Nations and Nationalism from 1780". Cambridge :Cambridge University Press.

Holloway, I., and Todres, L (2003) "The Status of Method: Flexibility, Consistency and Coherence". *Qualitative Research*, 3 (3), 345-357

Holstein, M. B. and Minkler, M. (2007) "Critical gerontology: reflections for the 21<sup>st</sup> century". In M. Bernard and T. Scharf, (eds) 'Critical perspectives on ageing societies'. Cambridge: Polity press.

Hooker, B. (2015) "The elements of Well-being." *Journal of Practical Ethics*, 3 (1), 15-35.

Horowitz, C., R., Robinson, M., and Seifer, S. (2009) "Community-Based Participatory Research From the Margin to the Mainstream. Are Researchers Prepared?". *Circulation*, 119, 2633-2642.

Hörnquist J.,O. (1982) "The concept of quality of life". Social care and Soc Medicine, 10, 57-61.

Hothi, M., Brophy, M., and Bacon, N. (2009) "Going Green and beating the Blues: The local approach to improving wellbeing and Environmental sustainability". London: Young foundation discussion paper. Available at: www.cipd.co.uk/wellbeing. [Accessed:12/4/16].

Hotze. T (2011) "Identifying the Challenges in Community-Based Participatory Research Collaboration". *Journal of Ethics*, 13, (2) 105-108.

Hovard L., Crowe S., and Hussain, N. (2004) "Effectiveness and Quality in customer Involvement Projects'" (EQUIP) Final Report. Public Health Resource Unit. Available at:www.invo.org.uk/All\_Publications.asp .Accessed:2nd March 2008[Accessed:19/3/2012]

Howard-Jones, P. (2011) "The Impact of Digital Technologies on Human Wellbeing: Evidence from the Sciences of Mind and Brain". Available at: http://www.nominettrust.org.uk/sites/default/files/NT%20SoA%20-%20The%20impact%20of%20digital%20technologies%20on%20human%20wellbein g.pdf [Accessed: 25/1/ 2016].

Hu, J., and Gruber, K., J. (2008) "Positive and negative affect and health functioning indicators among older adults with chronic illnesses". *Mental Health Nursing*, 29(8), 895-11.

Hubbard, G., Kidd, L., and Kearney, N. (2010) "Disrupted lives and threats to identity: The experiences of people with colorectal cancer within the first year following diagnosis". Available at:

https://dspace.stir.ac.uk/bitstream/1893/2105/1/Health%20in%20press%20-%20GH%20080109.pdf [Accessed: 15/11/ 2015].

Hubin, D., C. (2003) "Desires, whims, and values". *The Journal of Ethics, 7*, 315–335.

Hubley, J., and Copeman, J. (2008) "Practical Health Promotion". Cambridge: Polity Press.

Hughes, G. (1988) "Words in time". Oxford: Blackwell.

Hughes, C., and Hwang, B. (1996) "Attempts to conceptualize and measure quality of life". In R.L. Schalock, and G.N. Siperstein (eds) 'Quality of life volume I: Conceptualization and measurement'. Washington: American Association on Mental Retardation.

Hughes, G., Cooper, C., Gormally, S., and Rippingale, J. (2014) "The state of youth work in austerity England – reclaiming the ability to 'care'". *Youth and Policy*, 20-29.

Huppert, F. (2005) "Positive mental health in individuals and populations". In F., Huppert, N., Baylis and B. Keverne, (eds) 'The Science of Well-Being'. Oxford: Oxford University Press.

Huppert, F., Keverne, B., and Baylis, N. (eds) (2005) "The Science of Well-Being". Oxford: Oxford University Press.

Huppert, F. (2006) "Positive mental health in individuals and populations". In F., Huppert, N., Baylis and B. Keverne, (eds) "The Science of Wellbeing". Oxford: Oxford University Press.

Huppert, F. and Linley, P. (eds)(2011) "Happiness and well-being" (Vols. 1-4). Oxford, UK: Routledge.

Huppert, F. (2009) "Psychological well-being: Evidence regarding its causes and consequences". *Applied Psychology: Health and Well-being*, 1(2), 137-164.

Huppert, F. (2013) "The state of well-being science: concepts, measures, interventions and policies". In F., Huppert, and C. Cooper, (eds) (2013) 'Interventions and policies to enhance well-being'. Oxford: Wiley-Blackwell.

Hyde M., Wiggins R., Higgs P., and Blane D. (2003) "A measure of quality of life in early old age: the theory, development and properties of a needs satisfaction model (CASP-19)". Aging and Mental Health, 7 (3), 186–194.

Hyland, P., Gomez, O., and Greensides, F. (eds) (2003) "The enlightenment: A sourcebook and reader'. Oxon: Routledge.

Ibarra, H. (1999) "Provisional selves: Experimenting with image and identity in professional adaptation". *Administrative Science Quarterly*, 44, 764-791.

Iddons, P. (2009) "Survey Report". Unpublished, Manchester: Manchester PCT.

Imrie, R. (2012) "Universalism, universal design and equitable access to the built environment". *Disability Rehability*, 34(10), 873-82.

Inglehart, R., and Klingemann, H., D. (1998) "Genes, culture, democracy and happiness". In E., Diener and E. Suh, (eds) 'Culture and Subjective Well-Being'. Cambridge: Cambridge University Press.

Inglehart, R. (2002) "Gender, aging, and subjective well-being". *International Journal of Component Sociology*, 43 (3-5), 391–408.

INVOLVE (2007) "Good practice in active public involvement in research". Available at: http://www.invo.org.uk/pdfs/GoodPracticeD3.pdf [Accessed:7/6/2008]

INVOLVE (2012) "Briefing notes for researchers". Eastleigh: INVOLVE

Ipsos MORI (2009) "Long term health conditions: research study conducted for the Department of Health". London: Department of Health.

Israel, B., A., Eng, E., Schulz, A., J., and Parke, E., A. (eds) (2009) "Methods for Community-Based Participatory Research for Health". Oxford: Wiley-Blackwell.

Jacobs, B., Hartog, J., Vijverberg, W. (2009) "Self-selection bias in estimated wage premiums for earnings risk", *Empirical Economics*, 37 (2), 271–286.

Jacord, J., and Jacorby, J. (2010) "Theory construction and model building skills: A practical guide for social scientists". New York: Guidford.

Jahoda, M. (1982) "Employment and Unemployment: a Social Psychological Analysis". Cambridge: Cambridge University Press.

James, A., and Wells, A. (2003) "Religion and Mental health: towards a cognitive-behavioural framework". *British Journal of Health Psychology*, 8 (3), 359-376.

James, A., and Hockey, J. (2007) "Embodying Health Identities". Basingstoke: Palgrave Macmillan.

Jawad, R. (2012) "Religion and Faith-based Welfare: From Wellbeing to Ways of Being". Bristol: Policy Press.

Jeffreys, E. and Sigley, G. (2009) "Governmentality, governance and China'. In E. Jeffreys, (ed) 'China's Governmentalities: governing change, changing government'. New York: Routledge.

Jeffrey, K., Abdallah, S., Michaelson, J., Mahoney, S. (2014) "Wellbeing at Work". London: New economics foundation.

Jones, L., Sidell, M., and Douglas, J. (eds) (2002) "The Challenge of Promoting Health". Basingstoke: Exploration and Action Open University Press and MacMillan.

Jones, R., L. (2008) "Imagining old age". In J., Katz, , S., Peace and S. Spurr, (eds) 'A life course perspective'. Bristol: Policy Press University.

Jones, M. (2007) "An Introduction to Political Geography Space, Place, and Politics". New York: Routledge.

Jones, H., and Clapson, M. (eds) (2009). "The Routledge Companion to Britain in the Twentieth Century". New York: Routledge.

Jones, R.,L., and Bornat, J. (eds) (2012) "Imagining futures". London: The centre for policy on ageing.

Jordan, B. (2008) "Welfare and Well-being: Social Value in Public Policy". Bristol: Policy Press.

Joseph Rowntree Foundation (2004) "From welfare to well-being -planning for an ageing society: Summary conclusions of the Joseph Rowntree Foundation Task Group on Housing, Money and Care for Older People". York: Joseph Rowntree Foundation. Available at: http://www.jrf.org.uk/sites/files/jrf/034.pdf [Accessed:8/11/2012]

Joseph, S., and Wood, A. (2010) "Assessment of positive functioning in clinical psychology: Theoretical and practice issues." *Clinical psychology review*, 30 830-838.

Joshi, S., Kumari, S., and Varanasi, M. (2008) "Religious Belief and Its Relation to Psychological Well-being". *Journal of the Indian Academy of Applied Psychology*, 34 (2), 345-354.

Jost, J., T., and Banaji, M. R., (1994) "The role of stereotyping in system-justification and the production of false consciousness". *British Journal of Social Psychology*, 33, 1-27.

Joyce, P. (2003) "The Rule of Freedom: Liberalism and the Modern City". London: Verso.

Kagan, C. (2008) "Pillars of support for wellbeing in the community: the role of the public sector". Available at: http://www.e-space.mmu.ac.uk/e-space/bitstream/2173/17972/2/Pillars-of-support-for%2520wellbeing.pdf [Accessed:17/6/2010].

Kagan, C., Burton, M., and Siddiquee, A. (2008) "Action Research". In C. Willig and W. Stainton-Rogers (eds) "Handbook of Qualitative Methods in Psychology". London: Sage.

Kagan, C., and Kilroy, A. (2007) "Psychology in the Community". In J., Haworth and G. Hart, (eds) 'Well-being: Individual, Community and Social Perspectives'. Basingstoke: Palgrave Macmillan.

Kahneman, D. (1999) "Objective happiness". In D., Kahneman, E., Diener, and N. Schwartz, (eds) 'Well-being: the foundations of hedonic psychology'. New York: Russell Sage Foundation.

Kahneman, D., Diener, E., and Schwarz, N. (eds) (1999) "Wellbeing: the foundations of hedonic psychology". New York: Russell Sage Foundation

Kahneman, D., Krueger, A., Schkade, D., Schwarz, N., and Stone, A. (2004) "A survey method for characterising daily life experience: The day reconstruction method (DRM)" *American Economic Review*, 94(2), 429-434.

Kahneman, D., and Deaton, A. (2010) "High Income Improves Life Evaluation But Not Emotional Well-being", Proceedings of the National Academy of Sciences, 107,(38) 16489–93.

Kaplan, S. (1987) "Mental fatigue and the designed environment". In J., Harvey and D. Henning, (eds) 'Public Environments'. Edmund, CA: Environmental Design Research Association.

Kaplan, S. (2000) "Human Nature and Environmentally Responsible Behavior". *Journal of Social Issues*, 56 (3) 491–508.

Kaplan, R., M., and Anderson, J., P. (1988) "The quality of well-being scale: Rationale for a single quality of life index". In S., R., Walker, R. Rosser, (eds) 'Quality of Life: Assessment and Application'. London: MTP Press.

Kaplan, R., and Kaplan, S. (1989) "The Experience of Nature: A Psychological Perspective". Cambridge: Cambridge University Press.

Kaplan, S. (1995) "The Restorative Benefits of Nature: Towards an integrative framework". *Journal of Environmental Psychology*, 15,169–182.

Kapteyn A., Smith, J., and Van Soest, A. (2010) "Comparing Life Satisfaction". In A., Clark and C. Senik (eds) 'Happiness and Economic Growth: Lessons from Developing Countries'. Oxford: Oxford University Press.

Katz, J., and Peberdy A. (eds) (1997) "Promoting health, Knowledge and Practice". Basingstoke: Open University Press and MacMillan.

Katz, J., Peace, S., Spurr, S. (eds) (2012) "Adult lives: A life course perspective". Bristol: Policy Press University.

Kawulich, B., B. (2005) "Participant Observation as a Data Collection Method". *Qualitative Social Research*, 6 (2), Art. 43

Keiding, N., and Clayton, J., R. (2014) "Perils and potentials of self-selected entry to epidemiological studies and surveys". *Statistical. Society Association*, 179, Part 2, 1–28.

Keiding, N., and Clayton, D (2014) "Standardization and Control for Confounding in Observational Studies: A Historical Perspective". Statistical Science, 29 (4) 529-558.

Keiding, N., and Louis, T., A. (2016) "Perils and potentials of self-selected entry to epidemiological studies and surveys". *Journal of Royal Statistical Society*, A 179, Part 2, 1–28.

Kemmis, S. (2001) "Exploring the relevance of critical theory for action research: Emancipatory action research in the footsteps of Jürgen Habermas". In P. Reason and H. Bradbury (eds) 'Handbook of action research: Participative inquiry and practice'. London: Sage.

Kemmis, S., and McTaggart, R. (2005) "Participatory action research: Communicative action and the public sphere". In N., K., Denzin and Y., S. Lincoln (eds) 'Handbook of qualitative research' (3rd edn) Thousand Oaks: Sage.

Kendall, L; Harker L (eds) (2002) "From Welfare to Wellbeing: The Future of Social Care". London: Institute for Public Policy Research.

Kenrick, D., T., Griskevicius, Neuberg, S., L., and Schaller, M. (2010) "Renovating the pyramid of needs: Contemporary extensions built upon ancient foundations." *Perspectives on Psychological Science*, 5, 292-303.

Kenrick, D., T. (2010) "Rebuilding Maslow's pyramid on an evolutionary foundation". *Perspectives on Psychological Science*, 5, 253-263.

Kerr, D. (1999) "Beheading the king and enthroning the market: A critique of Foucauldian governmentality" in *Science and Society*, 63, (2), 173-203.

Kesebir, P., and Diener, E. (2008). "In pursuit of happiness: Empirical answers to philosophical questions". *Perspectives on Psychological Science*, 3, 117–125.

Kette, J. (1991) "Prison: A Social Psychological Analysis". In D. Kahneman, E. Diener, E. and N. Schwarz, (eds) (1999) 'Wellbeing: the foundations of hedonic psychology'. New York: Russell Sage Foundation.

Keyes, C., L., M. (1998) "Social Wellbeing". Social Psychology Quarterly 61(2), 121-140.

Keyes, C., L., M. (2000) "Subjective change and its consequences for emotional well-being". *Motivation and Emotion*, 24, 67-84.

Kickbusch, I. (2003) "The Contribution of the World Health Organization to a New Public Health and Health Promotion". *Health Promotion International*. *American Journal of Public Health*, 93(3), 383-388.

Kiefer, R. (2008) "An Integrative Review of the Concept of Well-Being". *Holistic Nurse Practitioner*, 22(5), 244–252.

Kimweli, D., and Stilwell, E. (2002) "Community Subjective Well-Being, Personality Traits and Quality of Life Therapy". *Social Indicators Research*, 60, (1-3).

Kim-Prieto, C., and Diener, E. (2009) "Religion as a source of variation in the experience of positive and negative emotions". *Journal of Positive Psychology*, 4, 447-460.

Kindon, S., Pain, R., Kesby, M. (eds) (2007) "Participatory action research approaches and methods: connecting people, participation and place". Abingdon, Oxon: Routledge.

King, N. (2004) "Using templates in the thematic analysis of text". In: C. Cassell, G. Symon, (eds) 'Essential guide to qualitative methods in organizational research'. London: Sage.

King, P. (2007) "The concept of wellbeing and its application in a study of ageing in Aotearoa New Zealand". EWAS Working Paper 8. Available at: http://www.ewas.net.nz [Accessed:3/11/2008].

King, S., and Waldron, J. (1988) "Citizenship, social citizenship and the defence of welfare provision". *British Journal of political science*, 18 415 -443.

King, L., A. Hicks, J., Krull, J., Gonso, D., K. (2006) "Positive affect and the experience of meaning in life". *Journal of Personality and Social Psychology*, 90 179- 196.

Kingfisher, C. (2013) "Happiness: notes on history, culture and governance" *Health, Culture and Society,* 5, 67-80.

Kingwell, M. (1998) "Better Living: In pursuit of happiness: From Plato to Prozak". Toronto: Penguin.

Knight, A., and McNaught, A. (eds) (2011) "Understanding wellbeing: An Introduction for Students and Practitioners of Health and Social Care". Basingstoke: Lantern Publishing Ltd.

Koenig, H., G. (1997) "Is religion good for your health? The effects of religion on physical and mental health". Binghamton, New York: Haworth Press.

Koenig, H., G. (2008) "Research on Religion, Spirituality and Mental Health: A Review". Canadian Journal of Psychiatry, 54 (5), 283-91.

Konijnendijk, C. (2008a) "The Healthy Forest' in The Forest and the City: Cultural landscapes of urban woodland". Netherlands: Springer.

Kothari, A., Demaria, F., and Acosta, A (2014) "Buen Vivir, Degrowth and Ecological Swaraj: Alternatives to sustainable development and the Green Economy". *Development*, 57, 362-375.

Krause, I., B. (1998) "Therapy across culture". London: Sage Publications Ltd

Krishnakumar, J. (2007) "Going beyond Functionings to Capabilities: An Econometric Model to Explain and Estimate Capabilities". *Journal of Human Development*, 8(1), 39-63.

Kroll, J., and Bachrach, B. (1984). "Sin and mental illness in the Middle Ages". *Psychological Medicine*, 14 (3), 507–514.

Kroll, C. (2011b) "Towards a Sociology of Happiness: Examining social capital and subjective well-being across subgroups of society" Unpublished PhD. London School of Economics and Political Science.

Kroll, C. (2014) "Towards a Sociology of Happiness: The Case of an Age Perspective on the Social Context of Well-Being". *Sociological Research Online*, 19 (2), 1.

Kuh, D., Karunananthan, S., Bergman, H., and Cooper, R. (2014) "A life-course approach to healthy ageing: maintaining physical capability". *The Proceedings of the Nutrition Society*, 73 (2), 273-248.

Kuo, F. E. and Sullivan, W. C. (2001) "Environment and crime in the inner city: does vegetation reduce crime?" *Environment and Behaviour* 33, 343–367.

Kyung-Man, K. (2004) "Critical Theory Criticized: Giddens's Double Hermeneutic and the Problem of Language Game Change". *Cultural Studies, Critical Methodologies*, 4 (1), 28-44.

Lammy, D., and Tyler, C. (2014) "Wellbeing in four policy areas Report by the All-Party Parliamentary Group on Wellbeing Economics". Available at: http://b.3cdn.net/nefoundation/ccdf9782b6d8700f7c\_lcm6i2ed7.pdf [Accessed: 25/1/2016].

Lane, R., F. (2000) "The Loss of Happiness in Market Democracies". New Haven and London: Yale University Press.

Langlois, A., and Anderson, D., E. (2002) "Resolving the Quality of Life/Well-being Puzzle: Toward a New Model". *Canadian Journal of Regional Science*, XXV(3): 501-512.

Larsen, R., J., Eid, M., and Diener, E. (2008) "The Science of Subjective Well-Being". In: RJ Larsen and M Eid, (eds) 'The Science of Subjective Well-Being'. New York: Guildford Press.

Lane, R., E. (1996) "Quality of Life and Quality of Persons: A New Role for Government". In O. Avner, (ed) 'In Pursuit of the Quality of Life'. Oxford: Oxford University Press.

Lawler, S. (2014) "Identity: Sociological Perspectives". (2nd edn) Cambridge: Polity Press.

Lawton, M., P. (1983) "Environment and other determinants of well-being in older persons". *Gerontologist*, 23, 349-357.

Lawton, M., P., and Cohen, J. (1974) "The generality of housing impact on the well-being of older people". *Journal of Gerontology*, 28, 194–204.

Lax, W., and Galvin, L. (2002) "Methodological problems: Reflections on a community action research project: Inter-professional issues and methodological problems". *Journal of Clinical Nursing, UK Community action research*, 377 (11) 376–386.

Layard, R. (2005) "Happiness: Lessons from a New Science". New York: Penguin.

Layard, R., Nickell S., and Jackman, R. (2005) "Unemployment: Macroeconomic Performance and the Labour Market", (2nd edn) Oxford: Oxford University Press.

Layard, R. (2006) "The depression report: a new deal for depression and anxiety disorders". Mental Health Policy Group, Centre for Economic Performance, London: London School of Economics.

Layard, R., (2009): "Well-Being Measurement and Public Policy." In A., B. Krueger (ed) 'Measuring the Subjective Well-Being of Nations: National Accounts of Time Use and Well-Being'. Chicago: University of Chicago Press.

Layard, R., Mayraz, G., and Nickell, S. J. (2010) "Does Relative Income Matter? Are the Critics Right?" In E., Diener, J., F., Helliwell and D. Kahneman, (eds) 'International Differences in Well-Being'. Oxford: Oxford University Press.

Layard, R., Clark, A. E., Cornaglia, F., Powdthavee, N. and Vernoit, J. (2013) "What Predicts a Successful Life? A Life-course Model of Well-being", CEP Discussion Paper No. 1245. London: London School of Economics (LSE) Centre for Economic Performance (CEP).

Layard, R., and Clark, D. M. (2014) "Thrive: The Revolutionary Potential of Evidence-Based Psychological Therapies" London: Penguin.

Layard, R. (2014) "Can We Afford More Therapy?" LSE CEP mimeo, Available at cep.lse.ac.uk/layard [Accessed: 12/5/16].

Lazarus, R., S. (1991) "Emotion and adaptation". London: Oxford University Press.

Lazarus, R., S., Folkman, S. (1984) "Stress, appraisal, and coping". New York: Springer.

Leary, M., R., Tangney, J., P. (2003) "Handbook of self and identity". New York: Guilford Press.

Lee, M., and Rubin, V. (2007) "The Impact of the Built Environment on Community Health: The State of Current Practice and Next Steps for a Growing Movement". Los Angeles: Policy Link and the California Endowment.

Lee, M. (2006) "Promoting mental health and wellbeing in later life: A first report from the UK Inquiry into Mental Health and Wellbeing in Later Life". London: Mental Health Foundation and Age Concern.

Lee, A., C., K., and Maheswaran, R. (2011) "The health benefits of urban green spaces: a review of the evidence." *Journal of Public Health*, 33(2), 212-222.

Leibfried, S., and Obinger, H. (2000) "Welfare State Futures: An Introduction". *European Review*, 8, 277-289.

Leitch, T., M. (2003) "Twelve Fallacies in Contemporary Adaptation Theory". *Criticism*, 45, (2) 149-171.

Leith, D. (1997). "A social history of English". (2<sup>nd</sup> edn) London: Routledge.

Lemke, T. (2001) "The birth of bio-politics: Michael Foucault's lectures at the College de France on neo-liberal governmentality". *Economy and Society*, 30 (2), 190-207.

Lemke, T. (2004) "Foucault, Governmentality, and Critique" in Rethinking Marxism, 14, (3) 49 – 64.

Leventhal, H., and Colman, S. (1997) "Quality of Life: A process view". *Psychology and Health*, 12, 753-767.

Lewin, K. (1942) "Time Perspective and Morale". In G. Lewin, (ed) 'Resolving Social conflicts'. New York: Harper.

Lewin, K. (1951) "Field theory in social sciences". New York: Harper.

Lewis, S., and Purcell, C. (2007) "Wellbeing, paid work and personal life". In J., Haworth and G. Hart, (eds) 'Well-being: Individual, Community and Social Perspectives'. Basingstoke: Palgrave Macmillan.

Lhussier, M. (2006) "A nomadic exploration of quality of life in long-term conditions". Unpublished PhD. Northumbria University.

Lhussier, M. (2009) "Quality of Life: Nomadic Insights – A nomadic exploration of quality of life in long-term conditions. VDM: Verlag.

Li, Y., Pickles, A., and Savage, M. (2005) "Social capital and social trust in Britain". *European Sociological Review*,21(2), 109–123.

Liu, E., March, T., and Jones, S. (2009) "Wellbeing in later life is closely related to the physical environment". *Environment and Behaviour*, 45 (3), 210–233.

Lilienfeld, S., O. (2012) "Public scepticism of psychology: Why many people perceive the study of human behavior as unscientific". *American Psychologist*, 67(2), 111–129.

Lim L.,Y. (2001) "Quality of life case studies for university teaching in sustainable development". *International Journal of Sustainability in Higher Education*, 2 (2), 127 – 138.

Lin, N., Ye, X., and Ensel, W. (1999) "Social support and depressed mood: A structural analysis". *Journal of Health and Social Behavior*, 344–359.

Lincoln, Y., S., and Guba, E., G. (1985) "Naturalistic inquiry". London: Sage.

Lindenmeyer, A., Hearnshaw, H., Sturt, J., Ormerod, R., and Aitchison, G. (2007) "Assessment of the benefits of user involvement in health research from the Warwick Diabetes Care Research User Group: A qualitative case study". *Health Expectations*, 10, 268-277.

Lindstrom, B. (1994) "QoL as domain of wellbeing: The Essence of Existence: on the Quality of Life of Children in the Nordic Countries. Theory and Practice in Public Health". Goteborg, Sweden: Nordic School of Public Health.

Lindström, B., and Henriksson, B. (1996) "The essence of existence On the quality of life of children in the Nordic countries". *International Journal of Social Welfare*, 5(2), 117-118.

Little, B., R. (1989) "Personal projects analysis: Trivial pursuits, magnificent obsessions, and the search for coherence". In D. Buss and N. Cantor, (eds) 'Personality psychology: Recent trends and emerging directions'. New York: Springer-Verlag.

Littlejohn, S., W., and Foss, K., A. (2009) "Agency". In S. Littlejohn, and K. Foss (eds), 'Encyclopedia of communication theory' Thousand Oaks: Sage.

Liu, B. (1975) "QoL: Concept, measure and results". *American Journal of Economic Sociology*, 34, 4-13.

Liu, C., W., Everingham, J., A., Warburton, J., Cuthill, M., Bartlett, H. (2009) "What makes a community age-friendly: a review of international literature". Australasian Journal on ageing, 28(3), 116-121.

Livneh, H. (2001) "Psychosocial adaptation to illness and disability: A conceptual framework". *Rehabilitation Counseling Bulletin*, 44(3), 151-161.

Lloyd, L. (2010) "The individual in social care: The ethics of care and the 'personalisation agenda' in services for older people in England". *Ethics and social welfare*, 4 (2), 188-200.

Local Government Association (2003) "Powering up: Making the most of the Well-Being Power LGA". Available at:

http://www.lga.gov.uk/Documents/Publication/poweringup.pdf [Accessed:14/08/10].

Locke, J (1689) 'The Two Treatises of Government'. In R. Ashcraft, (1986) 'Revolutionary Politics and Locke's "Two Treatises of Government'. Princeton: Princeton University Press.

Lucas, R., E., Clark, A., E., Georgellis, Y., and Diener, E. (2004) "Unemployment alters the set point for life satisfaction". *Psychological Science*, 15(1), 8-18.

Lucas, R., E, Clark, A., E., Georgellis, Y., Diener, E. (2003) "Re-examining adaptation and the set-point model of happiness: Reactions to changes in marital status". Journal of Personality and Social Psychology, 84, 527–539.

Luke, T., W. (1997) "Ecocritique: Contesting the Politics of Nature, Economy and Culture". Minneapolis: University of Minnesota Press.

Luo, Y., Hawkley, L., C., and Cacioppo, J., T. (2012) "Loneliness, Health, and Mortality in Old Age: A National Longitudinal Study". *Social Science and Medicine*, 74 (6), 907-914.

Lupton, D. (1995) "The imperative of health: Public health and the regulated body". London: Sage.

Lupton, D. (ed) (1999a) "Risk and sociocultural theory: New directions and perspectives". Cambridge: Cambridge University Press.

Lupton, D, (1999b) "Risk". London and New York: Routledge.

Lund, R., Nilsson, C. J., and Avlund, K. (2010) "Can the Higher Risk of Disability Onset Among Older People Who Live Alone be Alleviated by Strong Social Relations? A Longitudinal Study of Non-disabled Men and Women", *Age and Ageing*, 39, 3.

Lykken., D., and Tellegen, A. (1996) "Happiness is a stochastic phenomenon". *Psychological Science*, 7, 186–89.

Lyubomirsky, S., King, L., Diener, E. (2005) "The benefits of frequent positive affect: does happiness lead to success"? *Psychological Bulletin*, 131(6), 803–855.

McAllister, F. (2005) "Wellbeing concepts and challenges: Discussion Paper". SDRN Discussion paper.

McAndrew, R. (1993) "Environmental psychology". California: Brookes/Col.

McCartan, C., Schubotz, D., Murphy, J. (2012) "The self-conscious researcher—Post-modern perspectives of participatory research with young people". *Qualitative Social Research*, 13(1), Art. 9.

McCormick J., Clifton, A., Sachrajda, M., and McDowell E. (2009) "Getting On: Wellbeing in later life". Institute for Public Policy Research (ippr).

McCullough, M., Hoyt, W., Larson, D., Koenig, H., Thoresen, C. (2000), "Religious involvement and mortality: a meta-analytic review". *Health Psychology*, 19,211-222.

MacDonald, T.,H. (1998) "Rethinking Health Promotion – A Global Approach". London: Routledge.

McEvoy, R., Keenaghan, C., Murray, A. (2008) "Service users Involvement in the Irish Health Service: A Review of the Evidence". Available at: http://www.invo.org.uk/resource-centre/library-resource/?id=319andsection=invonet [Accessed:14/08/14]

McGillivray, M. (2006) "Human Wellbeing: Concept and Measurement" New York: Palgrave Macmillan.

McGregor, J.,A. (2004) "Researching well-being: communicating between the needs of policy makers and the needs of the people". *Global Social Policy*, 4 (3), 337-358.

McGregor, G. (2006) "Researching Wellbeing: From Concepts to Methodology and challenges". Wellbeing in Developing Countries (WeD) Working Paper 20.

Mclaughlin, H. (2006) "Involving Young Service Users as Co-Researchers: Possibilities, benefits and costs" *British Journal of Social Work*, 36 (8), 1395-1410.

Mclaughlin, H. (2007). "Service User Research". In R., Thorpe and R. Holt (eds) 'A Dictionary of Qualitative Management Research'. London: Sage.

Mclaughlin, H. (2009a) "What's in a name: client, patient, customer, consumer, expert by experience, service user: what's next?" *British Journal of Social Work*, 19 (6),1101-1117.

Mclaughlin, H. (2009b) "Service user research in Health and social care". London: Sage.

Mclaughlin, H. (2010) "Keeping service user involvement in research honest" *British Journal of Social Work*, 40 (5), 1591-1608.

Mclaughlin, H. (2011) "Ethical issues in involving young service users in research". In T., Ghaye, A., Melander-Wikman, C., Kostenius, S., Marchi, U., Bergmark and P. Chambers (eds) 'Participatory and Appreciative Action and Reflection'. Gloucester: New Vista Publications.

Mclaughlin, H. (2013) "Alternative futures for service user research". In P. Staddon 'Mental Health Service Users in Research: A Critical Perspective'. Bristol: Policy Press.

McMahon, D. (2004) "From the Happiness of Virtue to the Virtue of Happiness." *Daedalus* 133:2 (Spring 2004): 5–17.

McMahon, D. (2006) "Happiness: A history". New York: Atlantic Monthly Press.

McMahon, D. (2006a) "Market and the pursuit of happiness". Society, 43(2), 53-61.

McMahon, D. (2006 b) "The History of Happiness and the Contemporary Happiness Studies." In A.,K., Dutt and B. Radcliff (eds) (2009) 'Happiness, Economics and Politics: Towards a Multi-Disciplinary Approach'. New York: Edward Elgar.

McMahon, D. (2010) "What does the ideal of happiness mean". *Social Research*, 77 (2) 469-490.

MacKerron, G. (2011) "Happiness economics from 35,000 feet". *Journal of Economic Surveys*, 26(4), 705 – 735.

Macy, J., and Brown, M.,Y. (1998) "Coming Back to Life". Philadelphia: New Society Publishers.

Macunovich, D., J., and. Easterlin, R., A. (2008) "Easterlin hypothesis," *The New Palgrave Dictionary of Economics*, (2nd edn) London: Routledge.

Mahoney, J., M., and Rueschemeyer, D. (2003) "Comparative Historical Analysis in the Social Sciences". Cambridge: Cambridge University Press.

Malka, A., and Chatman, J., A. (2003) "Work orientation and contingency of job satisfaction and SWB on annual income: a longitudinal assessment". *Personality and Social Psychology Bulletin*, 29, 737-746.

Manchester Joint Health Unit (2005) "Manchester's health: A picture of progress, Compendium of statistics 2005" Available at: http://www.manchester.gov.uk/downloads/Picture of Progress 2007 1 .pdf.

[Accessed:28th January 2008]

Manchester City Council (2005a) "The Connected City: Inner South Manchester Strategic Plan 2005-2015". Manchester: Manchester City Council.

Manchester City Council. (2015) *Private tenacies and tenancy agreements*. [Online] [12/01/16]

http://www.manchester.gov.uk/info/100007/homes\_and\_property/1834/private\_t enancies\_and\_tenancy\_agreements

Manzo, L.C., and Brightbill, N., (2007) "Towards a participatory ethics". In S. Kindon, R. Pain, M. Kesby, (eds) 'Participatory action research approaches and methods: connecting people, participation and place'. Abingdon, Oxon: Routledge.

Marcus, G., E. (2003) "What Comes (just) After "Post"? In N., K. Denzin and Y., S. Lincoln (eds) (2<sup>nd</sup> edn) 'The Landscape of Qualitative Research: Theories and Issues'. Thousand Oaks: Sage.

Marks, N., and Shah, H. (2004) "A Wellbeing Manifesto for a Flourishing Society: The Power of Wellbeing". London: New Economics Foundation (nef). Available at: http://www.neweconomics.org/gen/ [Accessed:7/6/2008].

Marks, D., Murray, M., Evans, B., Willig, C., Woodall, C., and Sykes, C. (2005) "Health Psychology: Theory, Research and Practice". London: Sage.

Marks, N., Abdallah, S., Simms, A and Thompson, S. (2006a) "The Happy Planet Index". London: New Economics Foundation.

Marks, N., Thompson, S., Eckersley, R., Jackson, T., and Kasser, T. (2006b) "Sustainable Development and Wellbeing: relationships, challenges and policy implications". London: New Economics Foundation.

Marks, D., F., Murray, M., Evans, B., Estacio, E., V. (2011) (3<sup>rd</sup> edn) "Health Psychology. Theory-Research- Practice". Thousand Oaks: Sage.

Marsh, S., Cameron, M., Duggan, M., Eisenstadt, N., Iskander, R., Stone J. (2011) "Young people with life-limiting conditions: transition to adulthood". London: Public Service Works.

Marshall, C., and Rossman, G., B. (2009) "Designing qualitative research". (5th edn) Thousand Oaks: Sage Publications.

Marmot, M., and Wilkinson, R., G. (1999) "Social determinants of health". Oxford: Oxford University Press.

Marshall, J., and Reason, P. (2007) "Quality in research as "taking an attitude of inquiry"". *Management Research News*, 30, 368-380.

Maslow, A., H. (1943) "A Theory of Human Motivation". *Psychological Review*, 50 (4), 370-96.

Maslow, A., H. (1954) "Motivation and Personality". New York: Harper and Row.

Maslow, A., H. (1962) "Towards a Psychology of Being". Princeton: D. Van Nostrand Company.

Mason, M. (2010) "Sample Size and Saturation in PhD Studies Using Qualitative Interviews". Qualitative social Research, 11 (3), Art. 8.

Mathews, G., and Izquierdo, C. (eds) (2009) "Pursuits of Happiness: Well-Being in Anthropological Perspective". New York: Berghan Books Publishing.

Max-Neef, M. (1991) "Human-Scale Development - conception, application and further reflection". London: Apex Press.

Max-Neef, M. (1992). "Development and human needs". In P. Ekins, and M. Max-Neef (eds) 'Real-life economics: Understanding wealth creation'. London: Routledge.

Mayhew, S. (ed) (2004) "A Dictionary of Geography: Governmentality". Oxford: Oxford University Press.

Mazanec, S., R., Daly, B., J., Douglas, S., L., Lipson, A., R. (2010) "The relationship between optimism and quality of life in newly diagnosed cancer patients". *Cancer Nurse*, 33(3), 235-247.

Mehrabian, A. (1971) "Silent Messages". California: Wadsworth.

Mercier, C., Peladeau, N., and Tempier, R. (1998) "Age, gender and quality of life". *Community Mental Health Journal*, 34 (5),487–500.

Meyer, J. (2012) "Centenarians: 2010". Washington, DC: U.S. Dept. of Commerce, Economics and Statistics Administration, U.S. Census Bureau.

Michalos, A.,C. (ed) (2014) "Encyclopedia of Quality of Life and Well-Being Research". MMLXIV, 7347.

Micklethwait, J., and Woolridge, A. (2003) "The Company: A Short History of a Revolutionary Idea". New York: Modern Library

Mikics, D. (ed) (2007) "A New Handbook of Literary Terms, 2007, "New historicism". Yale: Yale University Press.

Miles, M., and Huberman, A (1994). Qualitative data analysis: An expanded source book (2<sup>nd</sup> edn) Thousand Oakes: Sage.

Mill, J., S. (1863) "Utilitarianism". R. Crisp, (ed) (1998) Oxford: Oxford University Press.

Miller, A. (2008) "A Critique of Positive Psychology or 'The New Science of Happiness'". *Journal of Philosophy of Education*, 42, 591–608.

Miller, P., and Rose, N. (1990) "Governing economic life". Economy and Society, 1990, 19 (1), 1-31.

Miller, P., and Rose, N (1995) "Political thought and the limits of orthodoxy: a response to Curtis". *British Journal of Sociology*, 1995, 46 (4), 590-597.

Miller, W., R., and Thoresen, C., E. (2003) "Spirituality, religion, and health: An emerging research field". *American Psychologist*, 58, 24–35.

Miller, E., Morrison, J. and Cook, A. (2006) "Brief encounter: Collaborative research between academic researchers and older researchers". *Generations Review*, 16(3/4), 39-41.

Miller, P., and Rose, N. (2008) "Governing the Present". Cambridge: Polity Press.

Miller, W., Pollack, C., Williams, D. (2011) "Healthy homes and communities: putting the pieces together. *American Journal of Preventive Medicine* 40(1), 48-57.

Minkler, M., and Wallerstein, N. (2003) "Community Based Participatory Research for Health". San Francisco: Jossey-Bass.

Mitchell, R., Dorling, D. and Shaw, M. (2000) "Inequalities in life and death: what if Britain were more equal?". Bristol: The Policy Press.

Mitra, A. (2002) (2<sup>nd</sup> edn) "History of Western Philosophy". Ahlulbayt School of Thoughts Researches and Publication Institute. Oxford: Routledge.

Mogilner, C., Kamvar, S., D., and Aaker, J. (2011) "The shifting meaning of happiness". *Social Psychological and Personality Science*, 2, 395-402.

Mokhtarian, P., L., Cao, X. (2007) "Examining the impacts of residential self-selection on travel behavior: A focus on methodologies". *Transportation Research* Part B, 42 (2008) 204–228.

Mokyr, J. (2010) "The Enlightened Economy: An Economic History of Britain 1700–1850". Bristol: The Policy Press.

Moloney, P. (2006) "Unhappiness is inevitable". The Guardian, 28 August, 2006 p. 27.

Moon, J. (1999) "Reflection in Learning and Professional Development". London: Kegan Paul.

Moons, P., Budts, W., and De Geest, S. (2006) "Critique on the conceptualisation of quality of life: a review and evaluation of different conceptual approaches'. *International journal of nursing studies*, 43(7), 891-901.

Morera, E. (2013) "Gramsci's Historicism: A Realist Interpretation". Oxford: Routledge.

Morgan, K., O. (2010) (ed) "The Oxford History of Britain". Oxford: Oxford University Press.

Morley, C., Ablett, P., Macfarlane, S. (2014) "Engaging with Social Work: A Critical Introduction". Cambridge: Cambridge University Press.

Morphet, J.(2008) "Modern local government". London: Sage.

Morrow, V., and Mayall, B. (2008) "What is wellbeing?". In A., Morgan, E., Ziglio, M., Davies and R. Barker, (eds) 'International health and development: investing in assets of individuals, communities and organisations'. York: Palgrave Macmillan.

Mowat, C., L. (1952) "The Approach to the Welfare State in Great Britain". The *American Historical Review*, 58 (1)b, 55-63.

Mroczek, D., K., and Kolarz, C., M (1998) "The Effect of Age on Positive and Negative Affect: A Developmental Perspective on Happiness". *Journal of Personality and Social Psychology*, 75 (5), 1333-1349.

Mruck, K., Roth, W., Breuer, F. (eds) (2002). "Subjectivity and reflexivity in qualitative research". *Qualitative Social Research*, *3* (3). Available at: http://www.qualitative-research.net/index.php/fqs/issue/view/21 [Date of access: 27/12/2011].

Myers, D., G., (1999) "Close Relationships and Quality of Life". In D. Kahneman et al.(eds), 'Well-being'. New York: Russell Sage Foundation.

Murdoch, S. (1999) "A Welfare to Well-being' strategy". Available at: http://www.msd.govt.nz/documents/about-msd-and-our-work/publications-resources/archive/2001-welfaretowellbeing.pdf [Accessed: 17/3/2010]

Mugglestone, L. (2006) "The Oxford History of English". Oxford: Oxford University Press.

Muraven, M., Gagne´M., Rosman, H. (2008) "Helpful self-control: Autonomy support, vitality, and depletion". *Journal of Experimental Social Psychology*, 44, 573–585.

Myers, D., G., and Diener, E. (1995) "Who Is Happy?". *Psychological Science*, 6 (1) 10-19.

Naess, A. (1973) "The shallow and the deep, long-range ecology movement. A summary". *Inquiry*, 16 (1) 95-100.

Naess, A. (1988) "Self Realization: An ecological approach to being in the world". In J., Seed, J., Macy, P., Fleming, and A Naess, 'Thinking like a mountain: Towards a council of all beings'. Philadelphia: New Society.

Naidoo, J., and Wills, J. (1994) "Health promotion – Foundations for Practice". London: Balliere Tindal.

Naidoo, J., and Wills, J. (2009). "Foundations for Health Promotion" (3rd edn) China: Bailliere Tindall Elsevier.

National Ecosystem Assessment (NEA) (2011). Oxford: Information Press Oxford. Available at :

http://www.wensumalliance.org.uk/publications/UKNEA\_SynthesisReport.pdf [Accessed:6/8/14]

National Health Service (NHS) England (2013) "The NHS belongs to the people". London: NHS England. Available at: https://www.england.nhs.uk/wp-content/uploads/2013/07/nhs\_belongs.pdf [Accessed:26/7/2015].

National Institute for Clinical Excellence (NICE) (2008) "Promoting or Creating Built Environments that Encourage and Support Physical Activity". NICE Public Health Guidance 8. Manchester: NICE.

NICE guidelines (2015) "Older people with social care needs and multiple long-term conditions". London: NICE. Available at:

http://www.nice.org.uk/guidance/ng22/resources/older-people-with-social-care-needs-and-multiple-longterm-conditions-1837328537797 [Accessed:16/12/2015].

National Youth Agency (2003) "Every Child Matters" – and every young person: the NYA's initial response to the green paper', *Spotlight*, 16, Leicester: National Youth Agency.

Naumova, E., N.(2014) "A cautionary note for population health: disproportionate emphasis on personal responsibility for health and wellbeing". *Journal of public health policy*, 35(3), 397-400.

Navarro, V. (1998) "Neoliberalism, globalization, unemployment, inequalities, and the welfare state". *International Journal of Health Services*. 28(4): 607–682.

Navarro, V. (1999) "The political economy of the welfare state in developed capitalist countries". *International Journal of Health Services*, 29(1), 1–50.

Navarro, V. (2002) "Power, politics and quality of life: A critique of social capital". *International Journal of Health Services*, 32(2), 423–432.

Naylor, C., Parsonage, M., McDaid, D., Knapp. M. et al.(2012) "Long term conditions and mental health – the cost of co-morbidities". London: The King's Fund and Centre for Mental Health.

Navarro, V. and Shi, L. (2001) "The political context of social inequalities and health". *Social Science and Medicine*, 52, 481–491.

New economics foundation(Nef)(2006) "Potential of wellbeing indicators". London: New Economics Foundation

Nef (2012) "Well-being Patterns Uncovered: An Analysis of UK Data". London: New Economics Foundation. Available at: http://www.neweconomics.org/ [Accessed:27/9/14]

nef (2011) "Measuring our progress". London: New Economics Foundation. Available at: http://www.neweconomics.org/publications/measuring-our-progress [Accessed:27/9/14]

Nettle, D. (2005) "Happiness: the science behind your smile." Oxford: Oxford University Press.

Nettleton, S. (1995) "The Sociology of Health and Illness". Cambridge: Polity Press.

Neuberger, J. (2008) "Unkind, risk averse and untrusting if this is today's society, can we change it"? York: Joseph Rowntree Foundation.

Neufeldt, V., and Guralnik, D., B (1997) "Websters new world dictionary". New York: MacMillan.

Neumayer, E. (2006) "Sustainability and wellbeing indicators". In M. McGillivray (Ed) 'Human Wellbeing: concept and measurement'. New York: Palgrave Macmillan.

Neurological Alliance (2001) "In search of a service: The experiences of people with neurological conditions". London, Neurological Alliance.

Neurological Alliance (2002) "Levelling up". London, Neurological Alliance. Available at:

http://www.sehd.scot.nhs.uk/nationalframework/Documents/neuro/key%20docs/levellingUP.pdf [Accessed: 5/7/2009]

Neurological Alliance (2003) "Neuro Numbers: A brief overview of the numbers of people in the UK with a neurological condition". London: Association of British Neurologists and Royal College of Nurses. Available at:

http://www.neural.org.uk/store/assets/files/20/original/NeuroNumbers.pdf [Accessed:5/7 2009].

Neurological Alliance. (2006) "Getting the Best from Neurological Services. Part 1, 2 and 3". London: Association of British Neurologists and Royal College of Nurses. Available at: http://www.neural.org.uk/publications-and-resources [Accessed:5/7/2009]

Newton, J. (2007) "Wellbeing and the Natural Environment: A brief overview of the evidence". Available at: www.sustainable-development.gov.uk/what/documents/WellbeingAndTheNaturalEnvironmentRepor

New Zealand Ministry for Culture and Heritage website. Available at: www.mch.govt.nz [Accessed:9/4/2014].

t.doc [Accessed:13/6/2009].

Nightingale, D., and Cromby, J. (1999) (eds) 'Social Constructionist Psychology'. Buckingham: Open University press.

Nilsson, J., Rana, A., and Kabir, Z., N. (2005) "Social capital and quality of life in old age: Results from a cross-sectional study in rural Bangladesh". *Journal of Aging and Health*, 18(3), 419–434.

Nolan, P., and Lenski, G. (2004) "Human societies: an introduction to macrosociology". Boulder, Colorado: Paradigm Publishers.

Nordbakke, S., and Schwanen, F. (2014) "Wellbeing and mobility: A theoretical framework and literature review focusing on older people". *Mobilities* 9 (1), 145-163.

Nordenfelt, L. (1991b) "Quality of Life and Health. Theory and Critique". Stockhelm, Sweden: Almqvist and Wiksell.

Nordenmark, M., and Strandh, M. (1999) "Towards a Sociological Understanding of Mental Well-Being among the Unemployed: The Role of Economic and Psychosocial Factors". *Sociology*, 33 (3) 577-597.

Norman R., M., G., Malla, A.,K., Mclean, T., Voruganti, L.,P.,N., Cortese, L., Mcintosh, E., et al.(2000) "The relationship of symptoms and level of functioning in schizophrenia to general well-being and the Quality of Life Scale". *Acta Psychiatr Scand*;102(4):303-309.

Nuffield Institute for Health (2007) "Living well in later life: From Prevention to Promotion". London: Nuffield Institute for Health.

Nussbaum, M. (1988) "Nature, functioning and capability: Aristotle on political distribution". *Oxford Studies in Ancient Philosophy, Supplementary Volume*, 145–184.

Nussbaum, M. (1992) "Human functioning and social justice. In defense of Aristotelian essentialism". *Journal of Political Theory*, 20(2), 202–246.

Nussbaum, M., and Sen, A. (1993) "The Quality of Life. In M., C., Nussbaum and A., K. Sen, (eds) 'Development'. Oxford: Oxford University Press.

Nussbaum, M. (1995) "Human capabilities, female human beings". In M., Nussbaum and J. Glover, (eds) "Women, Culture and Development". Oxford: Clarendon Press.

Nussbaum, M. (2000) "Women and Human Development: The Capabilities Approach". Cambridge: Cambridge University Press.

Nussbaum, M. (1995) "Aristotle on Human Nature and the Foundation of Ethics". In J., E., J. Altham and R. Harrison, (eds) 'World, Mind and Ethics: Essay on the Ethical Philosophy of Bernard Williams'. Cambridge: Cambridge University Press.

Nussbaum, M. (2003a) "Capabilities as fundamental entitlements: Sen and social justice". Feminist Economics, 9(2/3), 33–59.

Nussbaum, M. (2004) "On Hearing Women's Voices: A Reply to Susan Okin." *Philosophy and Public Affairs*, 32 (2), 193-205.

Nussbaum, M., C. (2005) "Wellbeing, contracts and capabilities." In I. Manderson (ed) 'Rethinking Wellbeing'. Perth: API Network, Australia Research Institute, Curtin University of Technology.

O'Brien, E. (2003) "Social and cultural values of woodlands and trees in northwest and southeast England". Unpublished report to the Forestry Commission.

O'Brien, M., and Penna, S. (1998) "Theorising welfare: Enlightenment and Modern Society". London: Sage.

O' Campo, P., Salmon, C., Burke, J. (2009) "Neighbourhoods and mental well-being: What are the pathways"? *Health and Place*, 15 (1), 56-68.

O'Donnell, G. (2014) "Wellbeing and Policy". London: Legatum Institute.

Office for National Statistics (ONS) (2008) "Internal migration estimates for local and unitary authorities in England and Wales, year to mid-2007". *Population Trends* 133, 81–89. Basingstoke: Palgrave.

ONS (2008) "More pensioners than under-16's for first time ever". Available at: www.statistics.gov.uk/cci/nugget.asp?ID=949 [Accessed: 17/3/2014]

ONS (2009a) "The impact of the recession on the labour market." Available at: www.statistics.gov.uk/downloads/theme\_labour/Impact-of-recession-on-LM.pdf [Accessed: 17/3/2014]

ONS (2009b) "Information on latest UK Population Estimates". Available at: www.statistics.gov.uk/CCI/nugget.asp?ID=6 [Accessed:17/3/2014]

ONS (2009c) "Ageing and mortality in the UK: National Statistician's annual article on the population". Population Trends 134. Basingstoke: Palgrave.

ONS (2009d) "Information on latest UK national population projections". Available at: www.statistics.gov.uk/statbase/Product.asp?vlnk=8519 [Accessed17/3/2014]

ONS (2009e) "Information on National Population Projections Expert Advisory Panel". Available at:

www.statistics.gov.uk/downloads/theme\_population/NPP2008/NatPopProj2008.pd f [Accessed17/3/2014]

ONS (2009i) "Population Trends No 138". Basingstoke: Palgrave.

ONS (2011a) "Measuring What Matters: National Statistician's: Reflections on the National Debate on Measuring National Well-being". Basingstoke: Palgrave.

ONS (2011b) "Initial investigation into Subjective Well-being data from the ONS Opinions Survey". Available at: http://www.ons.gov.uk/ons/guide-method/userguidance/well-being/publications/index.html [Accessed:14/08/14].

ONS (2012a) "Initial findings from the consultation on proposed domains and measures of national well-being". Available at: http://www.ons.gov.uk/ons/guide-method/user-guidance/well-being/publications/index.html [Accessed:14/08/14].

ONS (2012b), "Measuring National Well-being: The Natural Environment". Available at: http://www.ons.gov.uk/ons/rel/wellbeing/measuring-national-well-being/natural-environment/art-the-natural-environment.html [Accessed:14/08/14].

ONS (2012c) "Measuring National Well-being: Report on consultation responses on proposed domains and measures". Basingstoke: Palgrave.

ONS (2012d) "Analysis of experimental subjective well-being from the Annual Population Survey, April to September 2011". Basingstoke: Palgrave. Available at:

http://www.ons.gov.uk/ons/guide-method/user-guidance/well-being/publications/index.html [Accessed:14/08/14].

ONS(2012e) "Measuring National Well-being: Where We Live". Basingstoke: Palgrave.

ONS (2012f) "Measuring National Wellbeing: Life in the UK, 2012." Available at: http://www.ons.gov.uk/ons/dcp171778\_319478.pdf [Accessed: 17/3/2014]

ONS (2012g) "Subjective Well-being: A Qualitative Investigation of Subjective Wellbeing Questions". Basingstoke: Palgrave.

ONS (2013a) "Personal Well-being Across the UK 2012/13 Statistical Bulletin", Available at: www.ons.gov.uk/ons/rel/wellbeing/measuring-national-well-being/personal-well-being-across-the-uk--2012-13/sb---personal-well-being-acrossthe-uk--2012-13.html [Accessed:14/08/14].

ONS (2013b) "Personal Wellbeing in the UK, 2012/13". Available at: http://www.ons.gov.uk/ons/dcp171778\_319478.pdf [17/3/2014]

ONS (2013c) "Measuring National Wellbeing – What matters most to Personal Wellbeing?" Available at: http://www.ons.gov.uk/ons/dcp171766\_312125.pdf [Accessed: 17/3/2014]

ONS (2014) "Measuring National Well-being, Domains and Measures". Available at: http://www.ons.gov.uk/ons/rel/wellbeing/measuring-national-well-being/domains-and-measures---september-2014/index.html [Accessed:14/08/14].

Office of the Deputy Prime Minister (ODPM) (2006a) "The Social Exclusion of Older people: Evidence from the first wave of the English Longitudinal Survey of Ageing". London: ODPM.

Ogden, J. (2012) "Health Psychology: A textbook". (5<sup>th</sup> edn) Maidenhead: Open University Press.

Oguz, S., Merad, S., Snape, D. (2013) "Measuring National Well-being – What matters most to Personal Well-being?". Office for National Statistics. Available at: http://www.ons.gov.uk/ons/dcp171766 312125.pdf [Accessed:13/07/2014].

Oishi, S. (2001a) "Culture and memory for emotional experiences: on-line vs. retrospective judgments of subjective well-being". *Diss. Abstr. Int*. 61(10-B), 5625-5638.

Oishi, S. (2001b) "Cross-cultural differences in on-line and retrospective happiness: US vs. Japan". Symposium Presentation at: *13th Annual Meeting. Am. Psychol. Soc.* Toronto, 2<sup>nd</sup> July.

Oishi, S., and Diener, E. (2001a) "Goals, culture, and subjective well-being". *Personality and Social Psychology Bulletin*, 27, 1674-1682.

Oishi, S., Diener E. (2001c.) "Culture and Well-Being: The Cycle of Action, Evaluation, and Decision". Working Paper from the Department of Psychology. University of Minnesota.

Oishi, S. (2002) "The experience and remembering of wellbeing: a cross cultural analysis". *Personality and Social Psychology Bulletin*, 28 (10), 1398-1406.

Oishi, S. (2012) "The psychological wealth of nations: Do happy people make a happy society". Malden, MA: Wiley-Blackwell.

Oishi, S., Graham, J., Kesebir, S., Galinha, I. (2015) "The concept of Happiness across time and culture". *Personality and Social Psychology Bulletin*, 39 (5), 559-577.

Okun, M., A., and Stock, W., A. (1984) "Quality of research on correlates of subjective well-being in adulthood". *Experimental Aging Research*, 10(3), 161–163.

Okun, S and Stock, K (1987) "Wellbeing and Ageing". The journal of applied gerontology, 6, (1), 95-112.

Oldenburg, R. (1989) "The Great Good Place: Cafes, Coffee Shops, Community Centers, Beauty Parlors, General Stores, Bars, Hangouts, and How They Get You Through the Day". New York: Paragon House.

Oldenburg, R. (2000) "Celebrating the Third Place: Inspiring Stories about the "Great Good Places" at the Heart of Our Communities". New York: Marlowe and Company.

Oliver, Nicolas, Holloway, Frank and Carson, Jerome (1995) "Deconstructing Quality of Life". *Journal of Mental Health* (UK), 4, (1), 89-98.

O'Malley, P. (1996) "Risk and responsibility". In A., Osborne, T., Barry, T. and N. Rose, (eds) 'Foucault and Political Reason: Liberalism, Neoliberalism and Rationalities of Government'. London: UCL Press.

Ononiwu, C. (2013) "Coping with ageing life-adjustments: role of music, media, song and spirituality on wellbeing". Available at: https://publications.theseus.fi/bitstream/handle/10024/66287/THESIS%203.pdf?se quence=1 [Accessed:13/07/2014].

Organisation for Economic Cooperation and Development (OECD) (2001) "The well-being of nations: the role of human and social capital". Available at: http://www.oecd.org/site/worldforum/33703702.pdf [Accessed: 17/3/ 2014]

OECD (2013) "Guidelines on Measuring Subjective Well-being". Paris: OECD publishing.

OECD (2011) "How's Life: Measuring Well-Being". Paris: OECD Publishing.

OECD (2013) "OECD Guidelines on the Measurement of Subjective Well-being". Paris: OECD Publishing. Available at:

http://www.oecd.org/statistics/Guidelines%20on%20Measuring%20Subjective%20Well-being.pdf [Accessed:27/9/14].

Orley J, (1997) "Quality of life project". WHOQOL Group. Geneva: World Health Organisation.

Orley, J., Saxena, S., Herman, H. (1998) "Quality of life and mental illness. Reflections from the perspectives of the WHOQOL". *British Journal of Psychiatry*, 172, 291-293.

Orr, H., A. (2005) "The genetic theory of adaptation: a brief history". *Nature Review of Genetics* 6(2), 119-27.

Osoba, D., and King, M. (2005) "Meaningful differences". In: P. Fayers, R. Hays, (eds) 'Assessing quality of life in clinical trials'. Oxford Press: New York.

Ostir, G., V., Markides, K., S., Black, S., A. (2000) "Emotional well-being predicts subsequent functional independence and survival". *Journal of American Geriatric Society*, 48, 473–478.

Ostir, G., V., Markides, K., S., Peek, M., K, (2001) "The association between emotional well-being and incidence of stroke in older adults". *Psychosom Med* 63, 210–215.

Oswald, A., and Powdthavee, N. (2007a) "Obesity, Unhappiness and the Challenge of Affluence: Theory and Evidence". Economic Journal, 117 (521) 441-454.

Oxford English Dictionary (OED) (1922) (1st edn) Oxford: Clarendon Press.

OED (1989) (2<sup>nd</sup> edn) Oxford: Clarendon Press

OED (1989). Oxford: Clarendon Press.

OED (2002) (3<sup>rd</sup> edn) Oxford University Press.

Oxford English Dictionary (2015) [Online] [Accessed:12/3/2016] Available at: Http://dictionary.oed.com

Pacione, M. (1984) "The Definition and Measurement of Quality of Life". In M., Pacione and G. Gordon, (eds) 'Quality of Life and Human Welfare'. Norwick: Geobooks.

Pagano, I., S., and Gotay, C., C. (2006) "Modeling quality of life in cancer patients as a unidimensional construct". *Hawaii Medical Journal*, 65 (3) 76-85.

Pahl, R. (2007) "Friendship, Trust and Mutuality". In J., Haworth and G. Hart, (eds) 'Well-Being: Individual, Community, and Social Perspectives'. Basingstoke: Palgrave.

Paim, L. (1995) "Definitions and measurements of wellbeing: a review of literature" *Journal of Economic and Social Measurement*, 21, 297-309.

Pais-Ribeiro, J., L. (2004) "Quality of life is a primary end-point in clinical settings". Clinical Nutrition, 23(1), 121-30.

Pais-Ribeiro, J., L. (2004) "Quality of Life: Are We Talking about the Same Thing?". Available at: https://sites.google.com/site/jpaisribeiro/-proceedings [Accessed:14/08/14].

Palmer, S., S. (2000) "OED Online". Library journal 1: 161, Literature resource centre.

Palys, T. (2008) "Purposive sampling". In L., M. Given (ed) 'The Sage Encyclopedia of Qualitative Research Methods'. (Vol.2) Los Angeles: Sage.

Palys, T., and Atkison, R. (2008) "Research decisions: quantitative and qualitative perspectives". Toronto: Thomson Nelson.

Parfit, D. (1984) "Reasons and Persons". Oxford: Oxford University Press.

Parker, I. (ed) (1998) "Social constructionism, Discourse and Realism". London: Sage publications.

Parker, I. (2002) "Critical discursive psychology". Basingstoke: Palgrave.

Patrick D., L., Erickson, P. (1988) "Assessing health-related quality of life for clinical decision making". In S., R., Walker, R., M. Rosser (eds) 'Quality of life: assessment and application'. Lancaster, England: MTP Press.

Peck, J. (2013) "Explaining neoliberalism: Territory, Politics and Governance" London: Sage.

Penedo, F., and Dahn, J., R. (2005) "Exercise and Well-Being: A Review of Mental and Physical Health Benefits Associated With Physical Activity". *Current opinion in Psychiatry*, 18 (2), 189-193.

Perkins, N., and Hunter, D. (2014) "Health and Well-being Boards: a new dawn for public health partnerships"? *Journal of Integrated Care*, 22(5/6), 220-229.

Perri, 6. (2007) "Sense and solidarities: politics and human well-being". In J., Haworth, and G. Hart, (eds) 'Well-being: Individual, Community and Social Perspectives'. Basingstoke: Palgrave Macmillan.

Petersen, A., and Lupton, D. (1996) "The New Public Health: Discourses, Knowledges, Strategies". London: Sage Publications.

Peterson, E., S. (2005) "Light and liberty: Reflections on the pursuit of happiness". New York,: Modern Library.

Phillips, D. (2006) "Quality of Life: Concept, policy and practice". London: Routledge.

Phillips, D. (2006) "QoL". In J., Katz, S., Peace, and S. Spurr, (eds) (2012) 'Adult lives: A life course perspective'. Bristol: Policy Press.

Pickering, J. (2007) "Is wellbeing local of global? A perspective from ecopsychology". In J., Haworth and G. Hart, (eds) 'Well-being: Individual, Community and Social Perspectives'. Basingstoke: Palgrave Macmillan.

Piedmont, R., L. (1999). Strategies for using the five-factor model of personality in religious research. *Journal of Psychology and Theology*, 27, 338–250.

Pinker, S. (2007) "The stuff of thought: Language as a window into human nature". London: Viking penguin.

Polit, D., E., and Beck, C., T. (2008) "Nursing Research: Generating and Assessing Evidence for Nursing Practice". (8th edn) Philadelphia, USA: Wolters Kluwer, Lippincott Williams and Wilkins.

Poon, L., W., Cohen-Mansfield, J. (2011) "Understanding Well-Being in the Oldest Old". Cambridge: Cambridge University Press.

Poston, D., Turnbull, A., Park, J., Mannan, H., Marquis, J., and Wang, M. (2003) "Family Quality of Life: A Qualitative Inquiry". *Mental Retardation*, 41 (5), 313-328

Potter, J., Cantarero, R., and Wood, H. (2012) "The Multi-Dimensional Nature of Predicting Quality of Life". *Social and Behavioural Sciences*, 50, 781-790.

Pound, P., Gompertz, P., and Ebrahim, S. (1998). "Illness in the context of older age: The case of stroke". *Sociology of Health and Illness*. 20(4), 318-326.

Poynter, J., R. (1969) "Society and Pauperism: English Ideas on Poor Relief". London: Routledge and Kegan Paul.

Pressman S., D., and Cohen, S. (2005) "Does positive affect influence health"? *Psychological Bulletin*, 131, 925–971.

Pretty, J., Peacock, J., Sellens, M. and Griffin, M. (2005b) "The mental and physical health outcomes of green exercise". *International Journal of Environmental Health Research*, 15 (5), 319-337.

Prilleltensky, I., and Prilleltensky, O. (2007) "Webs of well-being: the interdependence of personal, relational, organisational and communal well-being".

In J., Haworth and G. Hart, (eds) 'Well-being: Individual, Community and Social Perspectives'. Basingstoke: Palgrave Macmillan.

Public Health England (2013) available at: http://www.wmpho.org.uk/topics/page.aspx?id=454 [ Accessed 12/7/15].

Putnam, R., D. (2000) "Bowling Alone: The Collapse and Revival of American Community". New York: Simon and Schuste.

Qizilbash, M. (1998) "The concept of well-being" *Economics and philosophy*, 14, 51-73.

Qizilbash, M. (2009) "Well-Being, Preference Formation and the Danger of Paternalism". Available at:

https://papers.econ.mpg.de/evo/discussionpapers/2009-18.pdf [Accessed:19/9/2010).

Raco, M. (2005) "Sustainable Development, rolled-out Neoliberalism and Sustainable Communities". *Antipode*, 37, 324-46.

Raco, M. (2009) "From expectations to aspirations: State modernisation, urban policy, and the existential politics of welfare in the UK". *Political Geography*, 28, 236-244.

Radnitsky, G (2002) "The EU: The European Miracle in Reverse". Historical Notes No. 41. An occasional publication of the Libertarian Alliance. Available at : www.libertarian.co.uk/lapubs/histn/histn041.pdf. [Accessed 2/8/15].

Rapley, M. (2003) "Quality of Life Research: A Critical Introduction". London: Sage.

Ranzijn, R., and Luszcz, M. (1999) "Acceptance: A key to well-being in older adults?" Australian Psychologist, 34 (2), 94-98.

Raz, J. (2004) "The role of Well-being". Philosophical Perspectives, 18, Ethics.

Rawls, J. (1993) "Political Liberalism". New York: Columbia University Press.

Reason, P. (1994) "Three approaches to participative inquiry". In N., K., Denzin and Y., S. Lincoln, (eds) 'Handbook of Qualitative Research', Thousand Oaks: Sage.

Reason, P. (1999) "Integrating Action and Reflection through Co-Operative Inquiry". *Management Learning*, 30 (2), 207-225.

Reason, P. (2002) "The practice of cooperative inquiry". Systemic Practice and Action Research, 15(3), 169-176.

Reason, P., and Bradbury, H. (2008a) "Introduction". In P., Reason and H. Bradbury, (eds) 'The Sage handbook of action research: Participative Inquiry and Practice' (2nd edn) London: Sage.

Reason, P., and Bradbury, H. (eds) (2008b). "The Sage handbook of action research: Participative Inquiry and Practice" (2nd ed.) London: Sage.

Rees, N. (1996) "Dictionary of word and phrase origins". London: Cassell.

Reeve, J., Lloyd-Williams, M., Payne, S., and Dowrick, C. (2010) "Revisiting biographical disruption: Exploring individual embodied illness experience in people with terminal cancer". *Public, Environmental and Occupational Health*, 14 (2), 178-195.

Reiner, M., Niermann, C., Jekauc, D., Woll, A. (2013) "Long-term health benefits of physical activity--a systematic review of longitudinal studies". *BMC Public Health*, 8 (13), 813-820.

Reis, H., T., Sheldon, K. M., Gable, S.L., Roscoe, J., and Ryan, R., M. (2000) "Daily well-being: The role of autonomy, competence, and relatedness". *Personality and Social Psychology Bulletin*, 26 (4), 419–435.

Reis, S., Steinle, S., Morris, G., Fleming, L., E., Cowie, H., Hurley, F. et al. (2013) "Integrating Health and Environmental Impact Analysis" Available at: http://www.sciencedirect.com/science/article/pii/S0033350613002424. [Accessed 24/4/15].

Rejeski, W., J., and Mihalko, S. (2001) "Physical Activity and Quality of Life in Older Adults". *Journals of Gerontology,* Series A: *Biological Sciences and Medical Sciences*, 56, (11).

Research Unit in Health and Behavioural Change, (1989) 'Changing the public health'. University of Edinburgh: John Wiley & Sons.

Ricoeur, P. (1988) "Time and narrative". (Vol. 3) Chicago: University of Chicago Press.

Ricoeur, P. (1992) "Oneself as another". Chicago: University of Chicago Press.

Ridder, D., Geenen, R., Kuijer, R., Van, Middendorp, H. (2008) "Psychological adjustment to chronic disease". *Lancet*, 372(9634), 246-255.

Riley, B., B., Perna, R., Tate, D., G., Forcheimer, M., Anderson, C., Luera, G. (1998) "Types of spiritual well-being among persons with chronic illness: their relation to various forms of quality of life". *Archives Physical Medical Rehabilitation*, 79 (3), 258–264.

Ritchie, J., Lewis, J., and Elam, G. (2003) "Designing and selecting samples". In J., Ritchie, and J. Lewis (eds) 'Qualitative Research Practice'. Oxford: Oxford University Press.

Ritchie, D., Parry, O., Gnich, W., Platt, S. (2004) "Issues of participation, ownership and empowerment in a community development programme: tackling smoking in a low-income area in Scotland". *Health Promotion International*, 19,51-59.

Riva, M., and Curtis, S. (2012) "The significance of material and social contexts for health and wellbeing in rural England". In S., Atkinson, S., Fuller, J, Painter, (eds) 'Wellbeing and place'. Ashgate, Farnham.

Robb, M. (ed) (2007) "Youth in Context: Frameworks, Settings and Encounters". London: Sage.

Robinson, E., H. (2013) "The distinction between State and Government". *Geography Compass*, 7/8, 556-566.

Rodger, J., J. (2008) "Criminalising Social Policy: Anti-Social Behaviour and Welfare in a De-Civilised Society". Cullompton: Willan.

Rodmell, S., and Watts, A. (eds) (1992) "The Politics of Health Education – Raising the Issues". London: Routledge.

Roe, B., Beynon, C., Pickering, L., Duffy, P. (2010) "Experience of drug use and Ageing: health, QoL relationships and service implications". *Journal of Advanced Nursing*, 66 (9) 1968-79.

Roffey. S. (ed) (2012) "Positive Relationships: Evidence Based Practice Across the World" New York: Springer.

Rose, N. (1989) "Governing the Soul". London: Routledge.

Rose, N. (1996) "Inventing Our Selves". Cambridge: Cambridge University Press.

Rose, N. (1999) "Powers of Freedom: reframing political thought". Cambridge: Cambridge University Press.

Rosen, G. (1958) "History of public health". New York: MD publications.

Ross, L., F., Loup, A., Nelson, R., M., et al. (2010) "The challenges of collaboration for academic and community partners in a research partnership: points to consider". *Journal of Empirical Research Human Research Ethics*, 5(1), 19-31.

Royle, E. (1987) "Modern Britain A Social History 1750- 1985". Agincourt, ON, Canada: Hodder & Stoughton.

Royle, E. (2012) (3<sup>rd</sup> edn) "Modern Britain: A Social History 1750–2011". Oxford: Oxford University Press.

Rozin, P. (2008) "New uncertain vs. familiar-positive experiences" [PowerPoint slides]. Available at: http://www.sas.upenn.edu/lps/graduate/mapp/login [Accessed:29/10/13]

Rozin, P., Haidt, J., and McCauley, C. (2008) "Disgust". In M., Lewis and J., M. Haviland-Jones, (eds) 'Handbook of emotions'. (3rd edn) New York: Guilford Press.

Rubinstein, R., L (1989)"Temporality and affect: The meaning of personal wellbeing". In (ed) L., E. Thomas, 'Research on Adulthood and Aging: The Human Science Approach'. New York: Suny Press.

Russo, J. (2012) "Survivor-controlled research: A new foundation for thinking about psychiatry and mental health". *Qualitative Social Research*, 13(1), Art. 8,

Rutter, M. (1995) "Clinical Implications of attachment concepts: retrospect and prospect". *Journal of child psychology psychiatry*, 36 (4) 549-571.

Ryan, P., and Coughlan, B. (2010) Psychology of ageing". East Sussex: Routledge.

Ryan, P., and Coughlan, B. (eds) (2010) "Ageing and Older Adult Mental Health: Issues and Implications for Practice". East Sussex: Routledge.

Ryan, B., Scapens, R., W., and Theobold, M. (1992) "Research method and Methodology in Finance and Accounting". London: Academic Press Itd.

Ryan, R., and Deci, E. (2000) "Self-determination theory and the facilitation of intrinsic motivation, social development, and well-being". *American Psychologist*, 55, 68-78.

Ryan, R., and Deci, E. (2001) "On Happiness and Human Potentials: A Review of Research on Hedonic and Eudaimonic Well-Being". *Annual review psychology*, 52, 141-166.

Ryff, C., D. (1989) "Happiness is everything, or is it? Explorations on the meaning of psychological well-being". *Journal of Personality and Social Psychology*, 57(6), 1069-1081.

Ryff, C., D. (1989b) "In the eye of the beholder: views of psychological well-being among middle-aged and older adults". *Psychological Aging*, 4, 195–210.

Ryff, C., D. (1991) "Possible selves in adulthood and old age: a tale of shifting horizons". *Psychological Aging*, 6, 286–95.

Ryff, C., D. (1995) "Psychological well-being in adult life". *Current Directory Psychological Science*, 4, 99–104.

Ryff, C., and Singer, B., H. (1998) "The contours of positive human health". *Journal of psychological Inquiry* 9 (1), 1-28.

Ryff, C., and Singer, B., H. (1999) "Know thyself and become what you are". *Journal of Happiness Studies*, 169-221.

Ryff, C., and Singer, B. (2000a) "Interpersonal Flourishing: A positive Health agenda for the new millennium". *Personality and Social Psychology Review*, 4, 30–44.

Ryff, C,. Love, G., D., Urry, L.,H., et al.(2006) "Psychological well-being and ill-being: do they have distinct or mirrored biological correlates"? *Psychotherapy Psychosomatic*, 75, 85–95.

Sabeti, S. (2014) "Creative Ageing? Selfhood, Temporality and the Older Adult Learner". *International Journal of Lifelong Education*, 12 (2), 15-21.

Sallis, J., F., and Nader, P.,R. (1988) "Family determinants of health behaviours". In D., S. Gochman, (ed) 'Health behaviour: Emerging research perspectives'. New York: Plenum Press.

Salzman, P., C. (2002) "On Reflexivity". American Anthropologist, 104: 805–811.

Sandvik, E., Diener, E., and Seidlitz, L. (1993) "Subjective well-being: The convergence and stability of self report and non self report measures". *Journal of Personality*, 61,317–342.

Sartori, P. (2010) Spirituality: should spiritual and religious beliefs be part of patient care? *Nursing Times*, 106, 28-37.

Sartorius N. (1996) "Quality of life in Psychiatry". European Psychiatric, 11, 228-229.

Sarvimaki, A. (2006) "Well-being as being well\*/a Heideggerian look at well-being". *International Journal of Qualitative Studies on Health and Well-being*, 1, 4/10.

Savin-Baden, M., and Major, C. (2013) "Qualitative research: The Essential Guide to Theory and Practice". London: Routledge.

Saxena S., and Orley J. (1997) "Quality of life assessment: The World Health Organization Perspective". *European Psychiatric* 12S3, 263s-266s.

Sayer, A. (2011) "Why Things Matter to People: Social Science, Values and Ethical Life". Cambridge: Cambridge University Press.

Scanlon, T. (1993) "Value, desire and quality of life". In M., Nussbaum, , A. Sen (eds) 'The Quality of Life'. Oxford: Clarendon.

Scarre, G. (1995) "Utilitarianism". London: Routledge.

Schalock, R., L. (1994) "Quality of life, quality enhancement, and quality assurance: Implications for program-planning and evaluation in the field of mental retardation and developmental disabilities". *Evaluation and Program Planning*, 17(2), 121-131.

Schalock, R., L. (1996) "Reconsidering the conceptualization and measurement of quality of life". In R., L.. Schalock, and G., N. Siperstein, (eds), 'Quality of life volume I: Conceptualization and measurement'. Washington: American Association on Mental Retardation.

Schalock, R., L. (1997) "Can the concept of quality of life make a difference"? In R., L. Schalock (ed) 'Quality of life volume II: Application to persons with disabilities'. Washington: American Association on Mental Retardation.

Schalock, R., L. (2000) (ed) "Quality of life volume II: Application to persons with disabilities". Washington: American Association on Mental Retardation.

Schalock, R.L., Bonham, G.,S., and Marchand, C., B. (2000) "Consumer based quality of life assessment: A path model of perceived satisfaction". *Evaluation and Program Planning*, 23(1), 77-87.

Schalock, R.L., Brown, I., Brown, R., Cummins, R.A., Felce, D., Matikka, L., Keith, K.D., and Parmeter, T. (2002) "Conceptualization, measurement, and application of quality of life for persons with intellectual disabilities: Report of an international panel of experts". *Mental Retardation*, 40(6), 457 – 470.

Schalock, R., L. (2004) "The Concept of Quality of Life: What We Know and Do Not Know". *Journal of Intellectual Disability Research*, 48 (3), 205-216.

Schalock, R. L., Keith, K. D., Verdugo, M. A., and Gomez, L. E. (2010) "Quality of life model development and use in the field of intellectual disability". In R. Kober (ed) 'Quality of life: Theory and implementation'. New York: Sage.

Schama, S. (2000) "A History of Britain" Vol. I. London: BBC Books.

Schama, S. (2001) "A History of Britain "Vol. II. London: BBC Books.

Schama, S. (2002) "A History of Britain" Vol. III London: BBC Books.

Schmidt, H. (2009) "Personal responsibility in the NHS Constitution and the social determinants of health approach: competitive or complementary". *Health Economics, Policy and Law,* 4(2), 129–138.

Schooler, K. (1970) "Effect of environment on morale". Gerontologist, 70, 194-197.

Schroeder, S., A. (2015) "Health, disability and wellbeing". In (Ed) G. Fletcher 'Routledge Handbook of Philosophy of wellbeing'. London: Routledge.

Scheufele, D., A. (1999) "Framing as a theory of media effect". *Journal of Communication*, Winter 1999, 103-122.

Schuessler, K.,F., Fisher, G.,A.(1985) "Quality of life research and sociology". *Annual Review of Sociology* 11, 129–149.

Schuz, B., Wurm, S., Warner, L., and Tesch-Romer, C. (2009) "Health and Subjective Well-being in later adulthood: Different health states – Different needs?" *Applied Psychology: Health and Well-being*, 1(1), 23-45.

Schwandt, T., A (2001) "Origins of historicism". In T., A. Schwandt, 'Dictionary of Qualitative Inquiry'. Thousand Oaks: Sage.

Schwandt, T., A. (2003) "Three epistemological stances for qualitative inquiry: Interpretativism, hermeneutics and social constructionism". In N., Denzin and Y. Lincoln (eds) 'The Landscape of Qualitative Research: Theories and issues'. Thousand Oaks: Sage.

Schwanen, T., and Paez, A. (2010) "The mobility of older people: An introduction". *Journal of Transport Geography*, 18 (5), 591-595.

Schwanen, T., and Ziegler, F. (2011) "Wellbeing, independence and mobility: an introduction. *Ageing and Society*, 31 (5), 719-733.

Schwanen, T., Hardill, I., Lucal, S. (2012) Spatialities of ageing: The co-construction and co-evolution of old age and space. *Geoforum* 43 (6), 21-30.

Schwanen, T., and Wang, D. (2014) "Wellbeing: Context and everyday activities in space and time". *Annals of association of American Geographers*, 104 (4), 833-851.

Schwartz, C., E., and Sprangers, M., A., G. (eds) (2000) "Adaptation to changing health: Response shift in quality of life research". Washington: American Psychological Association.

Schwartz, C., and Frohner, R. (2005) "Contribution of demographic, medical, and social support variables in predicting the mental health dimension of quality of life among people with multiple sclerosis". *Health and Social Work* 30(3), 203-12.

Schwarz, N., and Strack, F. (1999) "Reports of subjective well-being: Judgemental processes and their methodological implications". In D., Kahneman, E., Diener, and N. Schwarz, (eds) 'Well-being: The foundations of hedonic psychology'. New York: Russell Sage Foundation.

Schwartz, S., H. Verkasalo, M., Antonovsky, A., and Sagiv, L. (2006) "A Theory of Cultural Value Orientations: Explication and Applications". *Comparative Sociology*, 5, 2-3.

Scriven, A., and Garman, S. (2005) "Promoting Health: Global Perspectives". Basingstoke: Palgrave.

Scott, K. (2011) "Happiness on your doorstep: disputing the boundaries of wellbeing and localism in UK national policy agendas". https://www.psa.ac.uk/sites/default/files/312\_160\_0.pdf

Scott, K. (2012a) "A 21st Century Sustainable Community: Discourses of Local Wellbeing". In S., Atkinson, S., Fuller, and , J. Painter (eds) 'Wellbeing and Place'. Abingdon: Ashgate.

Scott, K. (2012b) "Happiness on your doorstep: The regressive dove-tailing of wellbeing and localism in UK national policy agendas". Available at: https://www.psa.ac.uk/sites/default/files/312 160 0.pdf [Accessed:5/9/2012].

Scott, K. (2012c) "Measuring wellbeing". London: Routledge.

Scott, K., and Bell, D. (2013) "Trying to measure local wellbeing: indicator development as a site of discursive struggles". *Environment and Planning C: Government and Policy*, 31(3), 522-539.

Scott K. (2014) "Happiness on your doorstep: Disputing the boundaries of wellbeing and localism". *The Geographical Journal*, 1-9.

Seaford, C. (2011) "The Practical Politics of Well-being". London: New Economics Foundation.

Searle, J. (1995) "The Construction of Social Reality". London: Penguin.

Searle, J. (2010) "Making the Social World: The Structure of Human Civilization". Oxford: Oxford University Press.

Searle, B. (2008) "Well-being: In Search of a Good Life"? Bristol: Policy Press.

Seedhouse, D. (1994) "The Trouble With Well-Being: A Response to Mild Mania and Well-Being". *Philosophy, Psychiatry, and Psychology*, 1 (3) 185-191.

Seedhouse, D. (1995) "Well-being: health promotion's red herring". *Health Promotion International*, 10 (1), 61-67.

Seedhouse, D. (1998) "Health Promotion, Philosophy, Prejudice and Practice". Chichester: John Wiley and Sons.

Seedhouse, D. (2001) "Health – The Foundation for Achievement". (2<sup>nd</sup> edn) Chichester: John Wiley and Sons.

Seedhouse, D. (2005) "Values-based decision-making for the caring professions". Chichester: John Wiley and Sons.

Self, A., Thomas, J., Randall, C. (2012) "Measuring national well-being: life in the UK, 2012". London: Office for National Statistics.

Seligman, M. (2011a) "Doing the right thing: Measuring wellbeing for public policy". *International journal of wellbeing*, 1 (1), 79-106.

Seligman, M. (2011b) "Flourish: A New Understanding of Happiness and Well-Being – and How to Achieve Them". London: Nicholas Breasly Publishing.

Sellers, S., Ward, E., and Pate, D. (2006) "Dimensions of Depression: A qualitative study of wellbeing among Black African Immigrant Women". *Qualitative social work* 5(1), 45-66.

Semetko, H., A., and Valkenburg, P., M. (2000) "Framing European Politics: A Content Analysis of Press and Television News," *Journal of Communication*, 50 (2), 93-109.

Sempik, J., Aldridge, J., and Becker, S. (2002) "Social and Therapeutic Horticulture: evidence and messages from research". Reading: Thrive in association with the Centre for Child and Family Research.

Sen, A. (1979) "Personal utilities and public judgements: or what's wrong with welfare economics?". *The Economic Journal*, 89, 537–558.

Sen, A. (1980) "Equality of what?". In S. McMurrin, (ed) 'The Tanner Lectures on Human Values'. Salt Lake City, Utah: Anchor.

Sen, A. (1984) "Rights and capabilities". In A. Sen 'Resources, Values and Development'. Cambridge: Harvard University Press.

Sen, A. (1985b) "Well-being, agency and freedom". *The Journal of Philosophy*, LXXXII (4),169–221.

Sen, A. (1987) "The standard of living". In G. Hawthorn, (Ed) 'The Standard of Living'. Cambridge: Harvard University Press.

Sen, A. (1993) "Capability and well-being". In M., Nussbaum and A. Sen (eds) 'The Quality of Life'. Oxford: Clarendon Press.

Sen, A. (1993) "Utility and Welfare: wellbeing agency and freedom". *Journal of Philosophy* 76 (9), 463-489.

Sen, A. (1999) "Development as Freedom". New York: Anchor.

Sen, A. (2002) "Health: perception versus observation". BMJ, 324 (7342), 860-871.

Seymour, D., B. (2006) "The history of English and Language change". New Jersey: Lawrence Erlbaum associates.

Shah, H (2005a) "The politics of well-being". Soundings, 30 (35), 33-44.

Shah, H. (2005b) "Well-being and the Environment". London: New Economics Foundation

Shah, H., and Peck, J. (2005) "Well-being and the Environment: achieving one planet living and quality of life". London: New Economics Foundation.

Shamir, R. (2008a) "The age of responsibilization: on market-embedded morality" *Economy and Society*, 37 (1), 1-19.

Shamir, R. (2008b) "Corporate Social Responsibility: Towards a New Market-Embedded Morality?" *Theoretical Inquiries in Law*, 9 (2), 32-52.

Shaw, M., Dorling, D., Gordon, D. and Davey-Smith, G. (1999) "The widening gap: health inequalities and policy in Britain". Bristol: The Policy Press.

Shay, S. (2008) "The History of English: A Linguistic Introduction". San Francisco: Wardja Press,

Sheldon K., M., and Kasser T. (1995) "Coherence and congruence: two aspects of personality integration". *J. Pers. Soc. Psychol*, 68, 531–43

Sherrrard, C., A. (1994) "Elderly wellbeing and the psychology of social comparison". *Ageing and society*, 14 (3), 341 – 356

Sidell, M., Jones, L., Katz, J., and Peberdy, A. (eds) (1997) "Debates and Dilemmas in Promoting Health. A Reader". Basingstoke: MacMillan and Open University Press.

Siegrist, J., Wahrendorf, M., Siegrist, J. (2010) "Are changes in productive activities of older people associated with changes in their well-being? Results of a longitudinal European study". *European Journal of Ageing*, 7(2), 59–68.

Silverman, D. (2005) "Doing Qualitative Research". (2<sup>nd</sup> Edn) London: Sage Publications.

Simpson, A., Jones, J., Barlow, S., Cox, L. and The Service User and carer Group Advising on Research (SUGAR) (2014) "Adding SUGAR: Service user and carer collaboration in mental health nursing research" *Journal of Psychosocial Nursing and Mental Health Services*, 52(1), 22-30.

Singh, K., and Jha, S., D. (2008) "Positive and negative affect, and grit as predictors of happiness and life satisfaction". *Journal of India Academic Applied Psychology*, 34, 40-45.

Sirgy, M., J. (2002) "The Psychology of Quality of Life". Dordecht, The Netherlands: Kluwer Academic Publishers.

Sirgy, M., J. and Estes, R. (In Press) "A history of Wellbeing: An ongoing project to explore the history of wellbeing". The Halloran Philanthropies Society. Available at: http://www.miqols.org/history%20of%20well-being/ [Accessed:19/10/15]

Sixsmith, J. (2012) "Well-Being Services for People with Long Term Neurological Conditions: Co-researchers Involvement in Research, Service Design and Development". *Global Journal of Community Psychology Practice*, 3(1), 50-58.

Sixsmith, J., and Boneham, M. (2007) "Health, well-being and social capital". In J., Haworth and G. Hart, (eds) 'Well-being: Individual, Community and Social Perspectives'. Basingstoke: Palgrave Macmillan.

Sixsmith, J; Callender, M., Hobbs, G., Corr, S., Huber, J (2013) "Implementing the National Service Framework for long-term (neurological) conditions: service user and service provider experiences". *Journal of Disability and Rehabilitation*. Available at: http://informahealthcare.com/doi/abs/10.3109/09638288.2013.804594 [Accessed:5/3/2014].

Skevington, S., M. (2002) "Advancing cross-cultural research on quality of life: Observations drawn from the WHOQOL development". *Quality of Life Research*, 11(2), 135–144.

Skevington,S. M., MacArthur, P., Somerset, M. (1997) "Developing items for the WHOQOL: A study of contemporary beliefs about quality of life related to health in Britain". *British Journal of Health Psychology*, 2, 55–72.

Skinner, E., Kindermann, T., Furrer, C. (2008) "A Motivational Perspective on Engagement and Disaffection". *Educational and Psychological Measurement*. 3, 108-113.

Smith, A. (1776) "An Inquiry into the Nature and Causes of the Wealth of Nations". Reprinted 1976, Oxford: Oxford University Press.

Smith, D., M. (1973) "The Geography of Social Well-being in the United States: An Introduction to Territorial Social Indicators". New York: McGraw-Hill.

Smith, A. (2000) "Researching QoL of older people: Concept, Measures and Findings". *Centre for Social Gerontology*, Working Paper No 7.

Smith, K., and Lipman, A. (1972) "Constraint and life satisfaction". *Journal of Gerontology*, 27, 77-82.

Smith, E., Ross, F., Donovan, S., Manthorpe, J., Brearley, S., Sitzia, J. and Beresford, P. (2008) "Service user involvement in nursing, midwifery and health visiting research: a review of evidence and practice". *International Journal of Nursing Studies*, 45 (2), 298-315.

Smith, M., Albanese, F., Truder, J. (2014) "A roof over my head: the final report of the Sustain project, a longitudinal study of housing outcomes and wellbeing in private rented accommodation". London: Shelter and Crisis.

Snoek, F.,J. (2000) "Quality of Life: A Closer Look at Measuring Patients' Well-Being". *Diabetes Spectrum*, 13 (4), 24-30.

Sointu, E. (2005) "The rise of an ideal: tracing changes discourses of wellbeing". *The sociological review*, 53, 255-274.

Sointu, E. (2006a) "Recognition and the Creation of Wellbeing" *Sociology*, 40(3), 493–510.

Sointu, E. (2006b) "The search for wellbeing in alternative and complementary health practices". Sociology of Health and Illness, 28 (3), 330–49.

South, J., and Woodall, J. (2010) "Empowerment and Health and Well-being: Evidence summary". Leeds: Leeds Centre for Health Promotion Research and Leeds Metropolitan University.

Spiro, A. III and Bosse, R. (2000) "Relations between Health-Related Quality of Life and Well-Being: The Gerontologist's New Clothes?" *The International Journal of Aging and Human Development*, 50 (4), 297 – 318.

Spokane, A., R., Lombard, J., L., Martinez, F., Mason, C.,A, Gorman-Smith D. (2007) "Identifying streetscape features significant to well-being". *Architectural Science Review*, 2007, 50(3), 234-245.

Stacey, J. (1997) "Teratologies: A cultural study of cancer". London: Routledge.

Stacey, J. (2000) 'The Global Within'. In S., Franklin, C., Lury and J. Stacey, (eds) 'Global Nature, Global Culture'. London: Sage.

Staddon, P.(2015) "Mental Health Service Users in Research: Critical Sociological Perspectives". Bristol: Policy Press.

Staeheli, L. (2001) "Of possibilities, probabilities and political geography". *Space and Polity*, 5, 177–89.

Staeheli, L.,A. (2011) "Political geography: Where's citizenship?". *Progress in Human Geography*. 2011, 35, 393-400.

Staley, K. (2009) "Exploring Impact: Public involvement in NHS, public health and social care research". Eastleigh: INVOLVE.

Staley, K. (2012) "An evaluation of service user involvement in studies adopted by the Mental Health Research Network (MHRN)". London: The MHRN.

Staley, K. (2013) "A series of case studies illustrating the impact of service user and carer involvement on research". London: The MHRN.

Staley, K., Tarpey, M., Hayes, H., and Buckland, S. (2012) "Public involvement in research: impact on ethical aspects of research". Eastleigh: INVOLVE. Available at: http://www.invo.org.uk/wp-

content/uploads/2012/06/INVOLVEevidenceresource.pdf [Accessed:28/8/2015]

Standing, G.(2010) "Work After Globalization: Building Occupational Citizenship". London: Edward Elgar Publishing.

Stark, J. and Faulkner, E. (1996) "Quality of life across the lifespan". In R., L., Schalock, and G., N. Siperstein (eds) 'Quality of life volume I: Conceptualization and measurement'. Washington: American Association on Mental Retardation.

Statham, J., and Chase, E. (2010) "Childhood Wellbeing: A brief overview". Childhood Wellbeing Research Centre Briefing Paper 1. Available at: https://www.gov.uk/government/uploads/system/uploads/attachment\_data/file/183197/Child-Wellbeing-Brief.pdf [Accessed:12/6/2014]

Stedman, T. (1996) "Approaches to measuring quality of life and their relevance to mental health". Australian and New Zealand Journal of Psychiatry, 30 (6), 731-740.

Steedman, P. (2000) "On the relations between seeing, interpreting and knowing". In F. Steier, (ed) 'Research and Reflexivity'. London: Sage.

Steel, R. (ed) (2004) "Involving the Public in NHS, Public Health and Social Care Research". Eastleigh: INVOLVE.

Steger, M., Frazier, P., Oishi, Kaler, M. (2006) "The meaning in life questionnaire". *Journal of Counselling Psychology*, 53, 80-93.

Stemler, (2001) "Emergent coding procedure suggestions". London: Sage.

Stenner, P., and Taylor, D. (2008) "Psychosocial welfare: reflections on an emerging field". *Critical Social Policy*, 28 (4), 415-417.

Stenner, P., Barnes, M., and Taylor, D. (2008) "Psychosocial welfare: Contributions to an emerging field". *Critical Social Policy*, 28 (4) 411-415. Available at http://csp.sagepub.com/content/28/4/415. [Accessed:23<sup>rd</sup> february 2011].

Stenner, P., McFarquhar, T., and Bowling, A. (2011) "Older people and 'active ageing': subjective aspects of ageing actively". *Journal of Health Psychology*, 16(3), 467–477.

Steptoe, D., and de Oliverira, C. (2012) "The psychological wellbeing and health functioning of older adults in England". In 'The dynamics of ageing: Evidence from the English Longitudinal Study of Ageing 2002-10'. London: DoH.

Stern, N. (2006) "The economics of climate change: The Stern Review". Cambridge: Cambridge University Press.

Steuer, N., Marks, N., Thompson, S. (2007) "Measuring well-being at the local level: a report for the Audit Commission". London: nef.

Stevenson, B., Wolfers, J. (2008) "Economic growth and subjective wellbeing: reassessing the Easterlin paradox". National Bureau for Economic Research Working Paper No. 14282. NBER, Massachussetts. Available at: http://www.nber.org/papers/w14282 [Accessed:19/11/13]

Stevenson, B., and Wolfers, J. (2009) "The paradox of declining female happiness". National Bureau of Economic Research. Working paper 14969. Available at: http://www.nber.org/papers/w14969 [Accessed:19/11/13]

Stewart, F. (2001) "Review of women and human development by Martha Nussbaum". *Journal of International Development*, 13, 1189-1202.

Stewart, J., and Bushell, F. (2011) "Housing, the Built environment and Wellbeing". In A., Knight, and A. McNaught, (eds) 'Understanding wellbeing'. Banbury: Lantern Publishing.

Stewart-Brown, S. (1998) "Emotional wellbeing and its relation to health". *BMJ*, 317, 1608-1609.

Stewart, J., and Bushell, F. (2011) "Housing, the Built Environment and Wellbeing". In (eds) A., Knight, and A. McNaught 'Understanding wellbeing'. Banbury: Lantern Publishing.

Stiglitz, J., Sen, A. and Fitoussi, J. (2011) "Report by the Commission on the Measurement of Economic Performance and Social Progress". Available at: http://www.stiglitz-sen-fitoussi.fr/documents/rapport\_anglais.pdf [Accessed:9/1/12].

Stoll, L., Michaelson, J., Seaford, C. (2012) "Well-being evidence for policy: A review". London: New Economics Foundation.

Stolle, D. (2001)"Clubs and congregations: The benefits of joining an association". In K. Cook, (ed) 'Trust'. New York: Russell Sage Foundation.

Stone, J., and Mennel, C. (eds) (1980) "Alexis de Tocqueville on Democracy, Revolution, and Society". Chicago: Chicago University Press.

Strachan, G., Wright, G., and Hancock, E. (2007) "An evaluation of a community health intervention programme aimed at improving health and wellbeing". *Health education Journal* 66(3), 277-285.

Stroebe, W., and Stroebe, M. (1996) "The social psychology of social support". In E., T., Higgins and A., W. Kruglanski, (eds) 'Social psychology: Handbook of basic principles'. New York: Guilford Press.

Stuff, R. (2016) "Growing the Health and Wellbeing Agenda: From first steps to full potential". Chartered Institute of Personnel and Development policy report. London: Chartered Institute of Personnel and Development.

Sturgeon, D. (2014) "The business of the NHS: The rise and rise of consumer culture and commodification in the provision of healthcare services". *Critical Social Policy* 34 (3), 405-416.

Suen L., J., Huang H., M., Lee H., H.(2014) "A comparison of convenience sampling and purposive sampling. *The Journal of Nursing*, 61(3),105-11.

Suh, E., Diener, E., and Fujita, F. (1996) "Events and subjective well-being: Only recent events matter". *Journal of Personality and Social Psychology*, 70(5), 1091-1102.

Sumner, L., W. (1996) "Welfare, Happiness and Ethics". Oxford: Clarendon Press.

Sundstrom, E., Bell, P.A., Busby, P.L., and Asmus, C. (1996) "Environmental psychology 1989-1994". In J., T., Spence, J., M., Darley and D., J. Foss, (eds) 'Annual Review of Psychology', Vol. 47, Palo Alto, CA: Annual Reviews Inc.

Sustainable Development Commission (2007a) "The natural environment, health and wellbeing". Healthy Futures no. 6, London: Sustainable Development Centre.

Swanson, R., A. (2013) "Theory building in Applied Disciplines". San Francisco: Berret Koehler.

Swanson, R., A. and Holton E., F. 3<sup>rd</sup> (eds) (1997) "Human Resource Development Handbook: Linking research and Practice". San Francisco: Berret Koehler.

Tabbush, P., and O'Brien, L. (2003) "Health and Well-being: Trees, Woodlands and Natural Spaces" Forestry Commission. Available at: www.forestry.gov.uk/pdf/health\_wellbeing.pdf/\$FILE/health\_wellbeing.pdf [Accessed:3/6/2010].

Taillefer, M., C., Dupuis, G., Roberge, M., and Lemay, S.(2003) "Health-Related Quality of Life Models: Systematic Review of the Literature". *Social Indicators Research*, 64(2), 293-323.

Tamminen, J., and Gaskell, M., G. (2008) "Learning New Words: Effects of Lexical Competition and Age of Acquisition". Available at http://csjarchive.cogsci.rpi.edu/proceedings/2006/docs/p822.pdf [Accessed:3/6/2010].

Tarpey, M. (2011) "Public involvement in research applications to the National Research Ethics Service". Eastleigh: INVOLVE.

Tarpey, M., and Bite, S. (2014) "Public involvement in research applications to the National Research Ethics Service: Comparative analysis of 2010 and 2012 data". Eastleigh: INVOLVE.

Taylor, C. (1989) "Sources of the Self". Cambridge, Massachusetts: Harvard University Press.

Taylor, C. (1991) "The Ethics of Authenticity". Cambridge, Massachusetts: Harvard University Press.

Taylor, C. (1994) "The Politics of Recognition". In A. Gutmann, (ed), 'Multiculturalism, Examining the Politics of Recognition'. Princeton: Princeton University Press.

Taylor, D. (2011) 'Wellbeing and Welfare: a psychosocial analysis of being well and doing well enough'. *Journal of Social Policy*, 40 (4), 777–794.

Taylor, T. (2015) The Markers of Wellbeing: A basis for a theory-neutral approach to well-being". *International Journal of wellbeing*, 2 (5). Available at: http://www.internationaljournalofwellbeing.org/index.php/ijow/article/view/373/478 [Accessed:25/9/2015].

Taylor, S., J., and Bogden, R. (1996) "Quality of life and the individual's perspective". In R.,L., Schalock, and G.,N. Siperstein, (eds) 'Quality of life volume I: Conceptualization and measurement'. Washington: American Association on Mental Retardation.

Thaler, R., and Sunstein C. (2008). "Nudge: Improving decisions about health wealth and happiness". New York: Penguin Books.

Taylor, S., E., and Stanton, A., L. (2007) "Coping resources, coping processes, and mental health". *Annual Review of Clinical Psychology*, 3, 377-401.

The economist (2012) "Age and happiness: The U-bend of life". Available at: http://www.economist.com/node/17722567?story\_id=17722567 [Accessed:2/6/2015]

The Institute of Wellbeing. "Wellbeing in Canada" Available at: http://www.ciw.ca/en/WellbeingInCanada/WhatIsWellbeing.aspx [Accessed:21/7/2014]

Thin, N. (2014) "Positive Sociology and Appreciative Empathy: History and Prospects". *Sociological Research Online*, 19 (2), 5.

Thornicroft, G., Ruggeri, M., and Goldberg, D. (eds) (2007) "Improving Mental Health Care: The Global Challenge". Oxford: Wiley-Blackwell.

Thompson, C., W. (2013) "Activity, Exercise and the Planning and Design of Outdoor Spaces". *Journal of Environmental Psychology*, 34, 79–96.

Thompson, S., and Marks, N. (2008) "Measuring well-being in policy: issues and applications" London: New Economics Foundation. Available at: http://s.bsd.net/nefoundation/default/page/file/575659b4f333001669\_ohm6iiogp. pdf [Accessed:14/8/2009]

Tiberius, V. (2004) "Cultural differences and philosophical accounts of well-being". *Journal of Happiness Studies*, 5, 293-314.

Tiberius, V. (2013) "The Best Within Us: Positive Psychology Perspectives on Eudaimonic Functioning". Washington: American Psychological Association.

Tiberius, V. (2015) "How theories of Well-being can help us help." *Journal of Practical Ethics*, 3 (1), 5-15.

Tiberius, V., and Plakias, A. (2010) "Well-Being". In J., M., Doris and the moral psychology research group (eds) 'The moral psychology handbook'. Oxford: Oxford University Press.

Tomlinson, M., W., and Kelly, G., P. (2013)" Is everybody happy? The politics and measurement of national wellbeing". *Policy and Politics*, 41 (2) 139-157.

Tones, K., Tilford, S., and Robinson, Y. (1990) "Health Education: Effectiveness, Efficiency, and Equity" (2<sup>nd</sup> edn) London: Chapman and Hall.

Tones, K., and Green, K. (2010) "Health promotion: planning and strategies". (2nd edn), London: Sage.

Tonkiss, F. (1998) "Analysing discourse". In C. Seale, (ed) "Researching Society and Culture". London: Sage.

Torfing, J. (1999) "New theories of discourse". Oxford: Blackwell Publishers Ltd.

Torraco, R., J. (1997) "Theory building research methods". In R.,A. Swanson and E., F. Holoton 3<sup>rd</sup> 'Human Resource Development Handbook: Linking research and Practice'. San Francisco: Berret Koehler.

Tov, W., Diener, E. (2008) "The well-being of nations: Linking together trust, cooperation, and democracy". In B., A., Sullivan, M., Snyder, J., L. Sullivan, (eds) 'Cooperation: The psychology of effective human interaction'. Malden, M.A.: Blackwell Publishing.

Townsend, M., and Weerasuriya, R. (2010) "Beyond blue to green: The benefits of contact with nature for mental health and well-being". Melbourne, Australia: Beyond Blue Limited.

Tracy, S., J., Tretheway, A. (2005) "Fracturing the Real-Self-Fake-Self Dichotomy: Moving Toward 'Crystallized Organizational Discourses and Identities"". *Communication Theory*, 15 (2), 168–195.

Trachtenberg, M. (2012) "Audience Costs: An Historical Analysis". *Security Studies*, 21 (1), 3-42.

Trevelyan, G., M. (1973) "English Social History: A Survey of Six Centuries from Chaucer to Queen Victoria". London: Book Club Associates.

Trevelyan, G., M. (1975) "Shortened History of England". London: Penguin Books.

Trivedi, P., Wykes, T. (2002) "From passive subjects to equal partners". *British Journal of Psychiatry*, 181, 468 – 472.

Trivedi, M., H., Greer, T., L., Church, T., S., Carmody, T. (2011) "Exercise as an augmentation treatment for non-remitted major depressive disorder: a randomized, parallel dose comparison". *Journal of Clinical Psychiatry*, 72(5, 677-84.

Tuan, Y. (1974) "Topophilia: a study of environmental perception, attitudes and values". Englewood Cliffs: Prentice Hall.

Turk, C., S. (2009) "Strength and Vulnerability Integration (SAVI): A Model of Emotional Well-Being Across Adulthood". *Psychol Bull*, 136(6), 1068–1091.

Turnbull III, H.,R., and Brunk, G., L. (1997) "Quality of life and public policy". In R., L., Schalock, (ed) 'Quality of life volume II: Application to persons with disabilities'. Washington: American Association on Mental Retardation.

Turner, J. C. (1991) "Social influence". Milton Keynes: Open University Press.

Turner, J. (1997) "The Institutional Order". New York: Longman.

Turner, M., and Beresford, P. (2005a) "Contributing on Equal Terms: Service user involvement and the benefits system". Adult Services Report 08. London: Social Care Institute for Excellence. Available at:

http://www.shapingourlives.org.uk/downloads/publications/benefitsreportoct05.pdf [Accessed:28/8/2015]

Turner, M., and Beresford, P. (2005b) "User Controlled Research: Its Meanings and Potential: Shaping Our Lives and Centre for Citizen Participation". Eastleigh: INVOLVE.

Ullman, S. (1962) "Semantics An introduction to the Science of Meaning". Oxford: Basil Blackwell and Mott Ltd.

Ulrich, R., and Parsons, R. (1992) "Influences of passive experiences with plants on individual wellbeing and health". In D. Relf, (ed) 'The role of horticulture in human wellbeing and social development', Portland: Timber Press.

Ulrich, R., Simons, R., Losito, B., Fiorito, E., Miles, M. and Zelson, M., (1991) "Stress recovery during exposure to natural and urban environments". *Journal of Environmental Psychology*, 11 (3), 201–203.

Ulrich, R., S. (1986) "Human responses to vegetation and landscapes". *Landscape and Urban Planning*, 13, 29–44.

United Nations (2007) "Article 2 of the United Nations Convention on the Rights of Persons with Disabilities". Available at: http://www.un.org/disabilities/[Accessed:27/6/2015].

United Nations International Children's Emergency Fund (UNICEF) (2009) "The State of the World's Children 2009". New York: UNICEF.

Vanhoutte, B. (2012) "Measuring subjective well-being in later life: a review". The Cathie Marsh Centre for Census and Survey Research. Manchester: University of Manchester.

Van den Berg, A., E., and Custers, M., H., G. (2011) "Gardening promotes neuroendocrine and affective restoration from stress". Journal of Health Psychology, 16, 3–11.

Van Manen, M. (2010) (2<sup>nd</sup> edn) "Researching Lived Experience: Human Science for an Action Sensitive Pedagogy". New York: State University of New York Press.

Van Zanden, J., and Luiten, J. (eds.) (2014) "How Was Life?:Global Well-Being since 1820". Paris: OECD Publishing.

Veenhoven, R. (1984) "Conditions of Happiness". Dordrecht: Reidel.

Veenhoven, R. (1991) "Is happiness relative?" Social Indicator Research, 24,1–34.

Veenhoven, R. (1993) "Happiness in Nations: Subjective Appreciation of Life in 56 Nations 1946–1992". Rotterdam: Erasmus University.

Veenhoven, R. (2000) "Happiness as an aim in public policy: The greatest happiness principle". In L., Alex and S. Joseph, (eds) 'Positive Psychology in Practice'. New Jersey: Wiley.

Veenhoven, R. (2002) "Why social policy needs subjective indicators". In V., I,. J., Hagerty and V. Moller, (eds) 'Assessing quality of life and living conditions to guide national policy: the state of the art'. Dordrecht: Kluwer Academic Publishers.

Veenhoven, R. (2003) "World Database of Happiness, Catalogue of Happiness Queries". Available at: www.eur.nl/fsw/research/happiness. [Accessed:7/6/2008]

Veenhoven, R. (2007) "Subjective measures of well-being". In M. McGillivray, (ed) 'Human Well Being: Concept and Measurement'. Basingstoke: Palgrave Macmillan.

Veenhoven, R. (2008) "Sociological theories of subjective well-being". In M., Eid and R., J. Larsen, (eds) 'The science of subjective well-being'. New York: Guilford Press.

Velarde, M., D., Fry, G., and Tveit, M. (2007) "Health effects of viewing landscapes: Landscape types in environmental psychology". *Urban Forestry and Urban Greening*, 6 (4), 199–212.

Vernon, M. (2008) "Wellbeing (Art of Living)". Stocksfield: Acumen Publishing.

Vilhena, E. (2014) "Psychosocial factors as predictors of quality of life in chronic portuguese patients". *Health and Quality of Life Outcomes*, 12 (3).

Vilhena E., Pais-Ribeiro, J., Silva, I., Pedro, L., Meneses, R. Et al. (2014) "Psychosocial factors as predictors of quality of life in chronic portuguese patients". *Health and Quality of Life Outcomes*, 12 (3).

Vincent, J., A. Phillipson, C., R. Downs, M. (2006) (eds) "The futures of old age" London: Sage.

Visser, A, Garssen, B, Vingerhoets, A (2010) "Spirituality and well-being in cancer patients: a review". *Psychooncology*, 19(6), 565-72.

Vittersø, J. (2004) "Subjective Well-Being Versus Self-Actualization: Using the Flow-Simplex to Promote a Conceptual Clarification of Subjective Quality of Life". *Social Indicators Research*, 65 (3), 300-323.

Vittersø, J., Røysamb, E., Diener, E. (2002) "The concept of life satisfaction across cultures: exploring its diverse meaning and relation to economic wealth". In E., Gullone and R. Cummins, (eds) 'Social Indicators Research Book Series: The Universality of Quality of Life Constructs'. Dordrecht: Kluwer Academic.

Vivenza, G., 2007. Happiness, wealth and utility in ancient thought. In L., Bruni and P., L. Porta, (Eds) (2007b) "Handbook on the Economics of Happiness". Cheltenham: Edward Elgar.

von Unger, H.(2012) "Participatory Health Research: Who Participates in What??". Qualitative Social Research, 13(1), Art. 7.

Wadsworth, Y. (1998) "What is participatory action research"? Action Research International, Paper 2. Available at: http://www.scu.edu.au/schools/gcm/ar/ari/p-ywadsworth98.html [Accessed:12/2/2011].

Waldron, S. (2010) "Measuring Subjective Wellbeing in the UK". ONS working paper. Available at http://www.statistics.gov.uk/cci/article.asp?ID=2578 [Accessed:17/3/2015].

Walker, H., and Kavedzija, I. (2015) "Values of Happiness". Special Issue Edition. *Journal of Ethnographic Theory*, 5 (3), 1-23.

Wallerstein, N., Duran, B. (2003) "The conceptual, historical and practical roots of community based participatory research and related participatory traditions". In M., Minkler and N. Wallerstein, (eds) 'Community-Based Participatory Research for Health'. San Francisco: Jossey-Bass.

Walsh, C. (2010) "Development as Buen Vivir: Institutional arrangements and (de)colonial entanglements". *Development* 53(1), 15–21.

Walsh, M., Stephens, P., and Moore, S. (2000) "Social Policy and Welfare". Cheltenham: Nelson Thornes.

Walker, A. (2003) "Quality of life in old age in Europe". Brussels: European Commission.

Ward, N., and Keith, K., D. (1996) "Self-advocacy: Foundation for quality of life". In R.,L., Schalock and G., N. Siperstein, (eds) 'Quality of life volume I: Conceptualization and measurement'. Washington: American Association on Mental Retardation.

Ward, L., Barnes, M., Gahagan, B. (2012) "Well-being in old age: findings from participatory research". Brighton: University of Brighton and Age Concern Brighton, Hove and Portslade. Available at: http://www.brighton.ac.uk/sass/older-people-wellbeing-and-participation/Executive-summary.pdf [Accessed:16/3/2014]

Waring, S., M., and Bentley, C., C. (2012) "Constructing historical profiles with digital natives". *Contemporary Issues in Technology and Teacher Education*, 12(2), 184-208.

Waring, S., M. (2012) "Conducting authentic historical investigations in the digital age". *Contemporary Issues in Technology and Teacher Education*, 12(1), 150-170.

Waring, S., M., and Robinson, K. (2010) "Developing critical and historical thinking skills in middle grades social studies". *Middle School Journal*, 42(1), 22-28.

Warner, M., E. (2010) "The Future of Local Government: 21st Century Challenges". *Public Administration Review*, 70(6).

Warr, P., B. (1987) "Work, unemployment, and mental health". Oxford: Oxford University Press.

Warr, P., B. (1990) "The measurement of well-being and other aspects of mental health". *Journal of Occupational Psychology*, 63, 193-210.

Warr P., B. (2003) "Well-being in the workplace". In: D., Kahneman, E., Diener, N. Schwarz, (eds) 'Well-Being: The foundations of hedonic psychology'. New York: Russell Sage Foundation Publications.

Warr, P., B. (2007) "Work, happiness, and unhappiness". New York: Routledge.

Warr, P., B., and Clapperton, G. (2010) "The joy of work? Jobs, happiness, and you". New York: Routledge.

Waterman, A., S. (1993) "Two conceptions of happiness: Contrasts of personal expressiveness (eudaimonia) and hedonic enjoyment". *Journal of Personality and Social Psychology*, 64, 678-691

Wasty, H., N.(1962) "Muslim Contribution to Medicine". Lahore: M. Sirajuddin and Sons.

Watts, S., and Guessous, J. (2006) "Agency". London: Sage.

Webb, S., and Webb, B. (1927) "English poor law history". (vol 7) London: Longmans, Green and co.

Websters (1971) (3<sup>rd</sup>edn) New International Dictionary (vol 3).

Weich, S., Blanchard, M., Prince, M., Burton, E., Erens, B., Sproston, K. (2002) "Mental health and the built environment: cross-sectional survey of individual and contextual risk factors for depression". *British Journal of Psychiatry*, 180, 428-433.

Weinreich, P., and Saunderson, W. (eds) (2003) "Analysing Identity: Cross-Cultural, Societal and Clinical Contexts." London: Routledge.

Weir, M. (1997a) "The Social Divide. Political Parties and the Future of Activist Government". Washington DC and New York: Brookings Institution/Russell Sage Foundation.

Weir, M. (1997b). "The Uncertain Future of Welfare Reform in the Cities." *The Brookings Review,* Winter 15 (1), 30-33.

Wells, N., M. (2000) "At Home with Nature: Effects of "greenness" on children's cognitive functioning". *Environment and Behaviour*, 32 (6), 775–795.

Wells, N., M., and Evans, G., W. (2003) "Nearby Nature: A buffer of life stress among rural children". *Environment and Behaviour*, 35 (3), 311–330.

Wells, N., M. (2011) "At home with nature: Effects of "greenness" on children's cognitive functioning". *Environ. Behav*, 32, 775–795.

Wenze, W., Gunthert, K., and German, R. (2012) "Biases in affective forecasting and recall in individuals with depression and anxiety symptoms". *Personality and Social Psychology Bulletin, 38*, 895–906

Westerhof, G., J., Dittman-Kholi, F., Thissen, T. (2001) "Beyond Life satisfaction: Lay conceptions of well-being amongst middle-aged and elderly adults". *Social Indicators Research*, 56 (2), 179-203.

Wierzbicka, A. (2004). "Happiness in cross-linguistic and cross-cultural perspective". *Daedalus*, 133, 34-43.

Wiseman, J., and Brasher, K. (2008) "Community wellbeing in an unwell world: Trends, Challenges and Possibilities". *Journal of Public Health Policy*, 29 (3), 353-366.

Wirtz, D., Kruger, J., Napa-Scollon, D. (2003) "What to do on spring break? The role of predicted, online, and remembered experience in future choice." *Psychological Science*, 14 (5), 520-524.

White, S., and Pettit, J. (2004) "Participatory Approaches and the Measurement of Well-Being". Bath: WeD Working Paper number 8.

White, A. (2007) "A Global Projection of subjective well-being: A challenge to Positive Psychology? *Psychtalk* 56, 17-20

White, M., Alcock, I., Wheeler, B., W., and Depledge, M., H. (2013) "Would You be Happier Living in a Greener Urban Area? A Fixed-effects Analysis of Panel Data", *Psychological Science*, 10 (3).

Whitbourne, S., K. (2016) "Differential aging". *International Encyclopaedia of Social and Behavioural Sciences*, 3654-3658.

Whitehall Wellbeing Working Group (W3G7) (2005) "Update summary on the use of wellbeing concepts in developing and appraising UK Government policy".

vi Prime Minister David Cameron's speech given on 25 November 2010. Available at: http://www.number10.gov.uk/news/pm-speech-on-well-being/[Accessed:15/12/2011].

Wiggins, R.,D., Higgs, P., F., D., Hyde, M., and Blane, D., B. (2004) "Quality of life in the third age: key predictor of the CASP-19 measure". *Ageing and Society*, 24, 693–708.

Wilkinson, R., G., and Picket, K., E. (2006). "Income Inequality and Population Health: A review and explanation of the evidence." *Social Science and Medicine*, 62 (7), 1768-84

Wilkinson, R., G., and Pickett, K. (2009) "The Spirit Level: Why More Equal Societies Almost Always Do Better". London: Allen Lane.

Williams, R. (1985) "Keywords: A Vocabulary of Culture and Society". Oxford: Oxford University Press.

Williams, S. (2000) "Chronic illness as biographical disruption or biographical disruption as chronic illness? Reflections on a core concept. Sociology of Health and Illness, 22, 40-67.

Willig, C. (2001) "Introducing qualitative research in psychology, Adventures in theory and method". Buckingham: Open University Press.

Wilson, E., O. (1984) "Biophilia". Cambridge: Harvard University Press.

Winefield, H., Gill, T., Taylor, A., and Pilkington, R.(2012) "Psychological Well-Being and Psychological distress: is it necessary to measure both"? *Psychology of Well-Being: Theory, Research and Practice*, 2,3.

Wiseman, J., Brasher, K. (2008) "Community Wellbeing in an Unwell World: Trends, Challenges, and Possibilities". *Journal of Public Health Policy*, 29, 353-366.

Wistow, G., Knapp, M., Hardy, B., Allen, C. (1994) "Social care in a mixed economy". Maidenhead: Open University Press.

Wollny, I,. Apps, J., and Henricson, C. (2010) "Can government measure family wellbeing"? London: Family and Parenting Institute.

Wood, G., and Newton, J. (2005) "From welfare to well-being regimes: engaging new agenda". Paper presented at: New Frontiers of Social Policy, Arusha Conference. Tanzania, 12<sup>th</sup> December. Available at: http://siteresources.worldbank.org/INTRANETSOCIALDEVELOPMENT/Resources/fromwelfaretowellbeing.pdf [Accessed:10/9/2012]

Woodward, K. (2004) "Questioning Identity: Gender, Class, Ethnicity". London: Routledge.

Wooffitt, R. (2005) "Conversation Analysis and Discourse Analysis: A Comparative and Critical Introduction". London: Sage.

Woolryche, R., and Sixsmith, J. (2008) "Understanding health and well-being in the context of urban regeneration - Manchester Case Study". Manchester: Research Institute for Health and Social Change (RIHSC). Available at: http://www.rihsc.mmu.ac.uk/Publications\_order\_form.pdf [Accessed:3/2/2010].

Wootton, D. (2003) "Divine Right and Democracy: An Anthology of Political Writing in Stuart England". Indianapolis/Cambridge: Hackett Publishing.

Wootton, D. (2015) "The Invention of Science: A New History of the Scientific Revolution. London: Penguin Books.

World Health Organisation (WHO) (1947) "The constitution of the World Health Organisation". WHO Chronicle 1947, 1. Geneva: World Health Organisation.

World Health Organization (1980) "The International Classification of Impairments, Disabilities, and Handicaps". Geneva: World Health Organisation.

World Health Organization. (1997). WHOQOL Measuring Quality of Life. Geneva: World Health Organisation. Available at:

http://www.who.int/mental\_health/media/68.pdf [Accessed:9/8/14].

World Health Organization (1978) "Copenhagen: WHO Europe; 1978. Declaration of Alma Ata (International Conference on Primary Health Care)". Geneva: World Health Organisation.

World Health Organization (1986) "Ottawa Charter for Health Promotion". Geneva: World Health Organisation.

World Health Organization (2002) "Active ageing: a policy framework". Geneva: World Health Organisation.

World Health Organization Quality of Life (WHOQOL) Group (1995) "The World Health Organization Quality of Life Assessment (WHOQOL): Position Paper From the World Health Organization". Social Science and Medicine, 41, (10), 1403-1409...

World Health Organization Quality of Life Group (1999) "The World Health Organization (WHOQOL): Tests of the universality of QoL in 15 different cultural groups worldwide". *Health Psychology*, 18 (5), 495-505.

World Health Organization (2010) "Definition of an older or elderly person". Geneva: World Health Organisation.

Wortman, C.,B., and Silver, R.,C. (1987) "Coping with irrevocable loss". In G.,R., Vanderbos and B.,K. Bryant, (eds) 'Cataclysms, crises, catastrophes: psychology in action'. Washington: APA.

Wright, C. (2014) "Happiness Studies and Wellbeing: A Lacanian Critique of Contemporary Conceptualisations of the Cure". Culture Unbound, 6, 791–813.

Wunderlich, F.(2016) "Place Temporality: Time, Rhythm and Urban Design". London: Ashgate Pub Co.

Wyche, S., Sengers, P., and. Grinter, R., E. (2006) "Historical Analysis: Using the Past to Design the Future". Available at: http://www.cc.gatech.edu/~beki/c29.pdf [Accessed:3/2/2010].

Yang, J., Li, S., J., Zheng, Y.,L. (2009) "Predictors of depression in Chinese community-dwelling people with type 2 diabetes". *Journal of Clinical Nursing*, 18(9),1295-1304.

Yılmaz, K. (2007) "Postmodernist Approach to the Discipline of History". *Kocaeli Üniversitesi Sosyal Bilimler Enstitüsü Dergisi* 14 (2) 176-188. Available at: http://kosbed.kocaeli.edu.tr/sayi14/yilmaz.pdf [Accessed:2/1/2013].

Yin, R., K. (1994) "Case Study Research: Design and Methods" (2<sup>nd</sup> edn) California: Sage.

Young, R., J., C. (2004) (2<sup>nd</sup> edn) "White Mythologies: Writing History and the West". Oxford: Routledge.

Zajenkowska, A., Zajenkowski, M., and Stolarski, M. (2014) "Why certain time perspectives makes people aggressive? Mediating effect of situational triggers of aggressive behaviours". *Personality and Individual Differences*, 60, S11.

Zahavi, D. (2005) "Subjectivity and Selfhood: Investigating the first-person perspective". Cambridge: MIT Press.

Zaidi, M., A., (2008) "Conceptualising wellbeing of older people". Guest editorial in M., A. Zaidi, (ed) 'Wellbeing of older people in ageing societies'. London: Ashgate publishing.

Zarri, L. (2007) "Happiness, morality and game theory". In: L. Bruni and P., L. Porta (eds) 'Handbook on the Economics of Happiness'. Cheltenham: Edward Elgar Publishing.

Zhang, J., H., Howell, R., T., and Stolarski, M. (2013b) "Comparing three methods to measure a balanced time perspective: The relationship between a balanced time perspective and subjective well-being". *Journal of Happiness Studies*, 14, 169–184.

Ziegler, F., and Schwanen, T. (2011) "I like to go out to be energised by different people': an exploratory analysis of mobility and wellbeing in later life". *Ageing and Society*, 31 (5), 758-781.

Zikmund, V. (G u e s t E d i t o r i a l) (2003) "Health, well-being, and the quality of life: Some psychosomatic reflections". *Neuro-endocrinol Letters* 24(6), 401–403. Available at: http://www.nel.edu/pdf\_/24\_6/NEL240603E03\_Zikmund\_.pdf [Accessed:3/2/2010].

Zimbardo P., and Boyd J. (1999) "Putting time in perspective: A valid, reliable individual differences metric". *Journal of Personality and Social Psychology,* 77 (6) 1271-1288.

Zimbardo, P., and Boyd, J. (2008) "The Time Paradox: The New Psychology of Time That Will Change Your Life". London: Simon and Schuster.

Zimmerman M., A., and Rappaport, J. (1988) "Citizen participation, perceived control, and psychological empowerment". *American Journal of Community Psychology*, 16:725-743.

# **Appendices**

# **Appendix A: Glossary**

**Active ageing**: "The process of optimizing opportunities for health, participation and security in order to enhance quality of life as people age" (WHO, 2002:10).

**Agency**: The capacity of individuals to act independently and to make their own free choices (Barker et al., 2005: 448).

Agentic determinants of wellbeing: Infers wellbeing is determined by self-directed actions and behaviours, influenced through "intentionality and forethought, self-regulation and self-reflective-ness" (Bandura, 2001:1). It assumes that the relationship between individual and wellbeing is one in which individuals have considerable autonomy to make and act on their choices.

**Built environment**: Can be broadly defined as "The sum total of what we design and construct in the places where we live, work, go to school, and play—from streets and highways to houses, businesses, schools, and parks. Since people create and experience communities in ways shaped by their cultures, understanding the built environment is as much about social processes as it is about physical ones" (Lee and Rubin, 2007:10).

**Chronic pain**: Any pain that has been experienced for more than 3 months. (DoH, 2010).

**Cohort effects**: Refers to "Data trends that arise because individuals were born at a particular point in time under particular circumstances that differ from those of individuals born at different times" (Schilling, 2005: 4).

**Community wellbeing:** "The combination of social, economic, environmental, cultural, and political conditions identified by individuals and their communities as essential for them to flourish and fulfil their potential." (Wiseman and Brasher, 2008:353).

**Cultural wellbeing:** "The vitality that communities and individuals enjoy through: participation in recreation, creative and cultural activities; and the freedom to retain, interpret and express their arts, history, heritage and traditions". (New Zealand Ministry for Culture and Heritage website).

**Czar**: "An informal term which refers to high profile appointments who devote their skills to one particular area of expertise) and the wider media industry" (Semetko and Valenburg, 2000:96).

**Deductive coding**: Codes which operationalise theories or accounts from existing research identified from the literature overview (Abramson, 2003:5)

**Differential ageing**: "A concept which reflects upon the process whereby some people remain physically and mentally healthy in older age whilst in others health begins to deteriorate in early adulthood (Schuz et al., 2009:23).

**Disability**: "Any restriction or lack (resulting from an impairment) of ability to perform an activity in the manner or within the range considered normal for a human being" (World Health Organization, 1980).

**Emotional wellbeing**: "The emotional quality of an individual's everyday experience – the frequency and intensity of experiences of joy, stress, sadness, anger and affection that makes one's life pleasant or unpleasant" (Kahneman and Deaton, 2010: 1).

**Environment**: "Including: (a) ecosystems and the constituent parts, including people and their communities; and (b) natural and physical resources; and (c) the qualities and characteristics of locations, places and areas; and (d) the social, economic and cultural aspects of a thing mentioned in paragraph (a), (b) or (c).

(Commonwealth Environmental Protection and Biodiversity Conservation Bill, 1998:6).

**Expert voices**: "Celebrated and ostensibly impartial professionals acknowledged as having specialized or professional knowledge in their field" (Shamir, 2008a:5).

**Frames:** "A composition of elements—visuals, values, stereotypes, messengers—which together trigger an existing idea. They tell us what this communication is about. They signal what to pay attention to (and what not to), they allow us to fill in or infer missing information, and they set up a pattern of reasoning that influences decision outcomes" (Dorfman et al. 2005: 323).

**Framing**: Refers to how "Frames tap complex moral structures that trigger how people react to a whole constellation of social and public policy issues in our society "(Dorfman et al. 2005: 323).

**Government**: "Governments can be interpreted as the major agent of the state and exists to carry out the day to day business of the state." (Flint and Taylor, 2007:39).

**Impairment**: "Any loss or abnormality of psychological, physiological or anatomical structure or function" (World Health Organization, 1980).

**Inductive coding**: "Codes generated after data collection during early data analysis which are used to group together common elements and quotes into meaningful categories or themes" (Abramson, 2003:7).

**Involvement in research**: "When research is carried out 'with' or 'by' members of the public rather than 'to', 'about' or 'for' them. (INVOLVE, 2012:4).

**Lay**: "A person who does not have specialized or professional knowledge of a subject" (Cambridge Dictionary online, 2016).

**Loneliness**: "Regarded as the psychological embodiment of social isolation, reflecting the individual's experienced dissatisfaction with the frequency and closeness of their social contacts or the discrepancy between the relationships they have and the relationships they would like to have" (Steptoe et al., 2012: 5798).

**Material wellbeing**: Financial status, employment status, housing status, possessions.

**Ontology**: "Philosophical study of the nature of being, becoming, existence, or reality, as well as the basic categories of being and their relations". (Oxford English Dictionary online).

**Participant:** Refers to all those taking part in research. Further information is available at: The British Psychological Society Code of Human Research Ethics http://www.bps.org.uk/sites/default/files/documents/code\_of\_human\_research\_e thics.pdf

**Personal identity**: "Refers to a person's self-definition based on personal ideals and attributes, furthermore personal identity categorizations may say something about how one is distinct and how one ranks compared to other in-group members" (Brickson, 2000: 82).

**Personal responsibility agenda**: "Influencing public behaviour, particularly with a view to increasing personal responsibility in areas like health and welfare" (Halpern et al., 2004:3).

**Personal wellbeing**: "People's own assessment of their own wellbeing" (Oguz et al., 2013:6).

**Physical wellbeing**: "The ability to be fully engaged, on a regular basis, in all developmentally appropriate activities" (Cole, 2006:2).

**Psychological wellbeing:** This study adopts Ryff (1989) understanding of psychological wellbeing as consisting of several dimensions: self-acceptance, autonomy, positive relationships, environmental mastery, personal growth, and sense of purpose. This offers an understanding of wellbeing that incorporates environmental aspects alongside personalised aspects.

**Political-media complex**: "The mutually beneficial relationships which have been established between a state's political agents, its special interest groups for example expert professionals" (Semetko and Valenburg, 2000:96).

**Positive Ageing**: "The concept of 'positive ageing' embraces a number of factors, including health, financial security, independence, self-fulfilment, community attitudes, personal safety and security, and the physical environment. The underpinning premise is that the years of 'older age' should be viewed and experienced positively. The focus is therefore not only on the experience of older individuals, but also on younger generations' attitudes, expectations, and actions regarding ageing and older people." (The New Zealand Government's Positive Ageing Strategy, Ministry of social development, 2000:9)

**Professional**: "A person who has the type of job that needs a high level of education and training". Cambridge Dictionary online (2016)

**Psychosocial:** Involving both psychological and social aspects and the interaction between social and psychological factors. Oxford English Dictionary online (2015).

**Purposive sampling**: "A non random selection of interviewees on purpose" (Suen et al. 2014:105).

**Religion:** "Involves beliefs, practices, and rituals related to the 'sacred.' Central to its definition, however, is that religion is rooted in an established tradition that arises out of a group of people with common beliefs and practices concerning the sacred" (Koenig, (2008: 283).

**Self selecting samples**: "A type of convenience sample comprising research interviewees or subjects who have volunteered to participate" (Colman, 2008:774).

**Social capital:** "Features of social organisation such as networks, norms, and trust, that facilitate co-ordination and co-operation for mutual benefit" (Putnam, 1995:67)

**Social isolation:** "A deficiency of contact between an individual and other people. It differs from loneliness, which reflects a temporary lack of contact with other humans" Hall and Havens (2003:2)..

**Social loneliness:** "The expression of dissatisfaction with a small number of regular social contacts" Hall and Havens (2003:2)..

**Social model of disability:** "Conceptualises disability as separate from impairment and focuses on the elimination of prejudices, oppression and barriers which prevent people with impairments from participating in society" (Lhussier, 2006:40).

**Social welfare:** "The provision of goods and services (including 'psychological' services) to meet human need...It is aimed at securing the social conditions of participation for subjects/citizens, and of course the regulation and limitations of participation for the socially excluded (Taylor, 2008: 417).

**Social wellbeing**: "The positive wellbeing effects of social cohesion, integration and inclusion" (Newton, 2007:29).

**Social networks**: Those networks which provide support to an individual ranging from those formed with neighbours, friends or family members to those which are formed within collective social groupings such as church membership or leisure groups (McCormick et al. 2009: 22).

**Snowball sampling:** "A type of purpose sampling where existing interviewees recruit future subjects from among their acquaintances. Thus the sample group appears to grow like a rolling snowball" (Ritchie et al., 2003:94).

**State:** "Claims common political and legal authority and a monopoly of legitimate force and other sovereign powers throughout its jurisdiction" (Agnew, 1998:7). It also "should possess the following qualifications: a permanent population, a defined territory, a government and a capacity to enter into relations with other states" (Montevideo Convention on Human Rights and duties of State 1933).

**Statutory body:** "A statutory body is set up by a law passed by parliament which explicitly mentions the objectives for their creation, their composition, and their powers. These bodies are authorised to enforce legislation on behalf of the relevant country or state". (Flint and Taylor, 2007:205).

**Structure:** "Those factors of influence (such as social class, religion, gender, ethnicity, customs, etc.) that determine or limit an agent and his or her decisions" (Baker et al. 2005:448).

Structural\* determinants of wellbeing: "These locate, often implicitly... problems in contextual or environmental factors ...rather than in characteristics of individuals...and tend to view individual agency as constrained or shaped by structures. They presume a certain degree of social causation of ...social, economic, political, or physical environments that shape... or otherwise affect outcomes. (Blankenship et al., 2006:59).

**Structure versus agency debate**: This may be understood within the context of determining whether an individual acts as a free agent or in a manner dictated by social structure. (Scott, 2012c).

**Subjective wellbeing:** "An individual's evaluation of their level of life satisfaction mapped against positive and negative aspects of affect" Diener et al. 1997:190).

**Temporality:** "Refers to time as it is experienced by us as humans" (Hemingway, 2011:23).

**Temporal states:** "The three temporal states, the past, the present and the future" (Durayappah, 2010:31).

**Temporal attitudes**: "A person's a prevailing outlook or way of thinking about time and temporality" (Durayappah, 2010:31).

**Temporal preferences:** "The temporal state which people tend to spend much of their cognitive time in". (Durayappah, 2010:31)

**Temporal perspectives:** "Reflects the strategic decision-making process involved in weighing the benefits of one particular state over another" (Durayappah, 2010:32).

**Temporal salience**: refers to "How evaluations of global well-being and life satisfaction could be significantly affected by ...how one feels at the time of evaluation" (Durayappah, 2010:29).

**Universal design:** "The design of products, environments, programmes and services to be usable by all people, to the greatest extent possible, without the need for adaptation or specialized design". (United Nations, 2007: Article 2).

**Utilitarianism:** A philosophical doctrine expounded originally by Jeremy Bentham (1700-1790) and then modified and contextualised by John Stuart Mill (1812-1890). Utilitarianism maintains that the ultimate goal of individuals is to maximise wellbeing or happiness. (Mokyr, 2010; Morgan, 2010)

**Welfare state:** "In essence it can be argued that it describes the institutions, policies and mechanisms dedicated to delivering those services, which seek to address issues around poverty and ill health, through the provision of education, housing, public health and pensions" (O'Brien and Penna, 1998:12).

**Wellbeing:** For the purpose of this thesis I have adopted Ryan and Deci (2001) appreciation of wellbeing as a "multidimensional phenomenon that includes aspects of both the hedonic and eudaimonic conceptions of well-being" (Ryan and Deci 2001:148).

### Appendix B: Conceptual Distinctions between Wellbeing and Quality of Life

#### Introduction

Some writers draw distinctions between the concepts wellbeing and quality of life (QoL) and urge others to follow their lead for consistency (Gasper, 2010; Haas, 1999; Helliwell and Putnam, 2004; Langlois and Anderson, 2002; Sarvimaki, 2006). Others prefer to treat the terms as synonymous (Camfield and Skevington, 2008; Krishnakumar, 2007) or propose blurring the conceptual the boundaries between QoL and wellbeing (Lhussier, 2009).

Wollny et al. (2010) argued that treating wellbeing and QoL as synonymous holds considerable appeal for those authors/researchers more concerned with practical applications than semantic erudition. This thesis however is directly concerned with investigating semantic understandings of wellbeing. In addition to which the empirical investigation is concerned explicitly with conceptualisations of wellbeing in relation to its mobilisation within a personal responsibility agenda. Thus, it is consistent with the aims and objectives of this study that the semantic distinction, which exists between the two concepts, necessitates a sole focus on the concept wellbeing. This thesis acknowledges these two concepts are closely aligned and share common characteristics. However, I propose that the concepts differ in respect to their linguistic, historical and conceptual identities and are two distinct concepts (Bowling et al., 2003; Camfield and Skevington, 2008; Gasper, 2010; Langlois and Anderson, 2002; Sarvimaki, 2006; Spiro and Bossé, 2000).

However, it is important to acknowledge the intersections between wellbeing and QoL (Snoek, 2000). Wellbeing research has utilised the findings of QoL research and the terms are often used interchangeably (Hamilton and Scullion, 2006). It is also important to note that distinctions between QoL research and wellbeing research are often blurred (Galloway, 2006; Gasper, 2010). The following pages therefore highlight some of the similarities and differences, which exist between

conceptualisations and theoretical and empirical understandings of wellbeing and QoL.

### The distinction between wellbeing and QoL within the parameters of this research

The utilisation of wellbeing in research and public policy is often grounded in QoL research and whilst these two concepts are used interchangeably they are distinct concepts (Bowling et al., 2003; Camfield and Skevington, 2008; Gasper, 2010; Langlois and Anderson, 2002; Sarvimaki, 2006; Spiro and Bossé, 2000).

Rapley (2003) reported that within research, QoL and wellbeing are commonly associated with each other as they both have two distinct and identifiable aspects, i.e. objective and subjective dimensions. However, there are a number of arguments why wellbeing and QoL may be considered distinct concepts.

First, where QoL may be sub-divided into objective and subjective domains, wellbeing have many more subdivisions. This enables wellbeing to be utilised with more specificity than QoL offers. Bond and Corner (2004), examined QoL research in social gerontology and health. They concluded researchers should draw a clear distinction between the objective indicators and psychometric measures of QoL and the subjective experiences of wellbeing. This has become increasingly important as domains such as social and emotional wellbeing are now widely utilised when assessing health (Eckermann, 2000).

Second, QoL is often used to describe how people feel at a particular stage in life in relation to their current life situation (Bowling, 2001; Bowling et al., 2003). Wellbeing however, is applied in relation to subjective experience but also has a philosophical aspect (Raz, 2004; Tiberius, 2004, 2011; Sumner, 1999). The philosophical aspect of wellbeing will be discussed in detail in section entitled 'Ontological perspectives'.

### **Historical identities**

QOL has a relatively short history (King, 2007; Ribeiro, 2004) coming into use during the 1950's (Armstrong and Caldwell, 2004; Snoek, 2000). It grew in prominence

during the 1960's because of the social indicators movement and became widely used during the 1970's (Andrew and Withey, 1976; King, 2007; Michalos, 2014). QoL is now widely recognised as a popular concept used in everyday language (Bowling, 2007; Katz et al., 2012; Pais-Ribeiro, 2004; Phillips, 2006).

Wellbeing is a much older construct than other overlapping concepts such as life satisfaction and QoL (Spiro and Bosse, 2000). Spiro and Bosse (2000) argued that the historical and linguistic legacy of wellbeing should be utilised by newer concepts such as QoL. Furthermore, they proposed that these fields of research should continue to "Maintain continuity with such classics as well-being" (p.297) in order to facilitate its absorption and adoption within mainstream gerontology.

Whilst wellbeing is not widely recognised as a popular concept, it is used extensively within media and government discourse (Fullager, 2002, 2009; Eraurt and Whiting, 2008; Rose, 1995, 1999; Sointu, 2005). It is a well-established and high status term (Crystal, 2002; Hughes, 1988; Ullman, 1962). However, research suggests this wellbeing is not widely understood or used in everyday language (Eraurt and Whiting 2008; Mathews and Izquierdo 2009; Seedhouse, 1995). Furthermore, it has failed to establish itself as a term widely used or recognised by lay populations (Clark and Gough, 2005; Edwards and Imrie 2008; Eraurt and Whiting, 2008; Hughes, 1988; Seedhouse, 1995; Whitehall Wellbeing Working Group W3G7, 2005).

The first recorded use of 'well-being' in written English was in 1613 (Oxford English Dictionary, 1971). During the 18<sup>th</sup> century, its legacy as a moral concept facilitated its appropriation within the doctrine of Utilitarianism (Mitra, 2002; Scarre, 1995). Utilitarianism is a philosophical doctrine, which espoused the pursuit of utility/wellbeing as the ultimate aim of citizens and governments (Scarre, 1995). The mobilisation of wellbeing utilises the notion of moral imperative (Raz, 2004; Sumner, 1999; Tiberius, 2004, 2011). It is this and the historical/linguistic legacy of wellbeing, which helps to, distinguishes it from QoL in the context of this study.

# The adoption of subjective assessments within advancing care practices

Many contend that the growing use of wellbeing and QoL arose during the 1960s within the social indicators research movement (Alkire, 2013; Andrews and Withey, 1976; Easterlin, 2007; Hörnquist, 1982; Snoek, 2000). This movement is widely associated with the changing societal norms and values of the 1960s. These shifting norms and values resulted in a reduced focus on the creation and measurement of standardised, objective conditions of peoples' lives. Instead, it became more concerned with the assessment of life satisfaction and quality of living (Abdallah and Shah, 2012; Christakopoulou, 2001; Easterlin, 2007). This growing concern primarily focused on the apparent discrepancies between objective economic and social indicators of life satisfaction and subjective evaluations of life satisfaction (Snoek, 2000).

Alongside the growth of QoL research, subjective wellbeing (SWB) also became an increasingly important consideration within healthcare and healthcare research (White, 2007). Advances in medical science, improved diets and better living conditions improved life expectancy but did not necessarily improve life satisfaction (White, 2007). This led to interventions and treatment becoming progressively more concerned with advancing care practices which simultaneously enhanced QoL and SWB (Scriven and Garmen, 2005; Seedhouse, 2001; Stewart and Bushell, 2011; Tones et al., 1990).

This paradigmatic shift in healthcare practice influenced the way in which patients were involved in healthcare decision-making (White, 2007). Tools were developed which could be used to assess patients' subjective evaluations of their health and healthcare experiences (White, 2007). This in turn led to the increasing use of measures of SWB (Abdallah and Shah, 2012; Stewart and Bushell, 2011).

Research undertaken in the 1960s and 70s, typically perceived wellbeing as that which denoted subjective experiences of life whilst QoL denoted that which was concerned with the objective conditions of life (Sarvimaki, 2006; Smith, 1973). Over

time, this conceptual distinction has been eroded until the terms became interchangeable (Langlois and Anderson, 2002).

### Conceptuality, circular definitions and definitional issue

Conceptually, wellbeing and QoL suffer from similar definitional problems (Camfield and Skevington, 2008; Gasper, 2010; Hamilton and Scullion, 2006). Both concepts are conceptually vague and there exists a plethora of definitions for QoL and wellbeing some of which tend to be discipline specific (Gasper, 2010). However, even within individual disciplines and organisations there are a broad range of definitions which are inconsistently applied (Eraurt and Whiting, 2008; Whitehall Wellbeing Working Group W3G7, 2005).

Additionally, authors often fail to provide an explicit definition of wellbeing or QoL (Gasper, 2010; King, 2007). Much of the literature and research associated with QoL and or wellbeing conflate the terms, or use them interchangeably without acknowledging contextuality, their epistemological positionality and the possible implications this may have (Gasper, 2010).

The circularity of definitions, which exists between QoL and wellbeing, represents a challenge for researchers (Sarvimaki, 2006) as the terms are frequently used as to help define each other (WHO, 1946). Definitions of QoL imply that what is worthwhile rests on the foundations of health and function (Bowling, 1995a, 1996, 2008). Definitions of wellbeing also imply that health is fundamental for a worthwhile existence. Definitions of wellbeing however, are often more complex and frequently include philosophical underpinnings and implications (Raz, 2004; Ryan and Deci, 2001; Sumner, 1999; Tiberius, 2004; Tiberius and Plakius, 2010).

Both health and QoL have commonly been defined as or conflated with wellbeing (Atkinson and Joyce 2011; Riva and Curtis 2012; Seligman 2011). In its landmark definition of health, the World Health Organisation (WHO, 1946) equated health and wellbeing by defining health as physical, mental and social wellbeing. In Eriksson's (1984) definition of health, wellbeing was perceived as one aspect of

health whilst wellbeing, was characterised as a concept which referred to how one feels or experiences life and health. Bowling (2005) however defined health in terms of quality of life and laid particular emphasis on psychological wellbeing.

These definitional issues are often compounded because "Researchers do not feel the need to explicitly define the term they are attempting to measure" (Galloway, 2006:30). These definitional issues are challenging for researchers, and pose significant challenges, for example, when conducting and writing a literature review (Galloway, 2006). These are further complicated by a tendency for some writers to conflate wellbeing and QoL with SWB, happiness, life satisfaction, and welfare (Galloway, 2006). In a perfect example of circular definitions, Christoph and Noll (2003) defined SWB as life satisfaction and represented SWB as a constituent of welfare, a term that they equated with QoL.

The association of SWB with life satisfaction and or QoL in itself can lead to conceptual ambiguity (Galloway, 2006; Hendry and McVitie, 2004; Snoek, 2000). This has been acknowledged as an additional issue for those seeking to explicate the meaning of wellbeing (Ryan and Deci, 2001). Ryan and Deci (2001), noted that some interpret SWB as the personal evaluation or assessments of one's own sense of wellbeing, whilst others interpret it as the balance between life satisfaction, positive affect and negative affect. Helliwell and Putnam (2004), for example, refer to happiness and life satisfaction in their efforts to distinguish between wellbeing and SWB.

"Generally speaking, self-ratings of 'happiness' turn out to reflect relatively short-term, situation-dependent (affective) expressions of mood, whereas self-ratings of 'life satisfaction' appear to measure longer-term, more stable (cognitive) evaluations". (p.1435).

The conflation of happiness and life satisfaction with wellbeing and QoL has led to further ambiguity (Galloway, 2006; King, 2007; Zikmund 2003). There are numerous examples across wellbeing and QoL research in which life satisfaction is explicitly or implicitly inferred as a synonym for wellbeing or QoL (Bowling, 1995a, 1996;

Farquhar, 1995; Haas, 1999b; Oliver et al., 1995; Taileffer et al., 2003).

Christakopoulou et al. (2001) used life satisfaction implicitly in their definition of wellbeing within urban populations in the UK, Ireland and Greece. They measured wellbeing in terms of satisfaction with a range of domains such as living space, the social community, economic community, political community (Christakopoulou et al., 2001). Galloway's (2006) extensive review of wellbeing and QoL reported that in economic SWB literature there are well-established associations between life satisfaction and the wider objective characteristics of QoL. Tailleffer et al. (2003) study of 63 QoL models reported that QoL, wellbeing and SWB were all equated with life satisfaction.

Haas (1999b) however, argued against the growing inclination and practice across healthcare research for researchers to conflate one term with another.

"The terms QoL, satisfaction with life, functional status, and well-being can no longer be used interchangeably. They represent different levels and aspects of the broad concept of QoL. If one chooses to focus on the subjective aspects of QoL, then it must be clear that that is what is being discussed is either 'well-being' or 'subjectively perceived QoL.' For those who choose to focus on objective indicators of QoL, it must be clearly identified as either 'functional status' or 'objectively perceived QoL.' Those who study satisfaction with life must either clearly state that as the purpose of their investigation or make it clear that they are interested in studying an aspect of well-being or subjectively perceived QoL". (p.8).

The inconsistent application of a researcher's chosen terms in association with inconsistent usage and a failure to offer any definition of terminology threatens to undermine efforts to synchronise research findings from the different fields of study (Christoph and Noll, 2003; Galloway, 2006; Gasper, 2010). These issues add to the already challenging nature of studying wellbeing and QoL. Without definitional and conceptual clarification it is not possible to conclude with any conviction if and what the relationship between wellbeing and QoL might be.

The practice of embedding concepts associated with wellbeing and QoL within definition is also problematic. Concepts such as health and happiness are frequently embedded within definitions of wellbeing and QoL. This has served to further obfuscate the boundaries between complex concepts and undermined efforts to establish consensual definitions (Atkinson 2013; Atkinson and Joyce, 2011).

The multiplicity of terms associated with wellbeing and QoL has stimulated a debate concerned with the degree to which the terms may be considered synonymous or distinct concepts (Bond and Corner, 2006; Langlois and Anderson 2002, Lim, 2001; Sarvimaki, 2006). Some identify QoL and wellbeing as distinct concepts such as Smith (1973), Langlois and Anderson (2002), Tiberius and Plakias (2010) and Gasper (2010). Others, perceive the terms to be synonymous, such as Bradburn (1969), Wiggins et al. (2004) and Burchi and Gnesi (2013). A considerable number of authors exist along this continuum including Snoek (2000), Sarvimaki (2006) and Galloway (2006). The following section endeavours to offer a précis of their respective positions.

## Wellbeing and QoL commonality

As previously noted, there has been an increasing trend amongst QoL and wellbeing research to distinguish between objective indicators and subjective evaluations of life satisfaction, QoL and wellbeing (Snoek, 2000). Simultaneously, however, there has been a considerable convergence of the terms (Langlois and Anderson, 2002). This may be because of the expansion of interest in wellbeing and QoL across academic disciplines (Brown et al., 2004; Galloway, 2006).

Langlois and Anderson (2002) reviewed QoL and wellbeing literature in order to integrate the terms. The result of this review led to a new 'Integrated Model of Quality of Life and Well-being' (Langlois and Anderson, 2002). This new model sought to understand the holistic nature inherent in QoL and wellbeing through the identification and integration of those domains identified as common to both concepts. Despite the considerable intersections between wellbeing and QoL, few researchers have taken up the 'Integrated Model of Quality of Life and Well-being'.

The intersections between wellbeing and QoL occur across policy, conceptual, empirical and academic contexts. These intersections were drawn upon to facilitate their adoption across a diverse range of subject areas after the Local Government Act, 2000 (Scott, 2012c). The promotion of wellbeing as part of Local Government duties to citizens led to both wellbeing and QoL being used widely in the early discourses of the New Labour government (Scott, 2012b, c). In these discursive domains the concept of wellbeing and or QoL were entwined with notions of sustainable development (Scott, 2012b, c).

The adoption of wellbeing/QoL in relation to sustainable development was possible because of existing associations in these concepts with the measurement of human progress (Lim, 2001; Smith, 1973; Snoek, 2000). It may be that the wider environmental factors associated with wellbeing and QoL, such as economic conditions were also important elements which could be drawn upon. Working conditions, for example, have been widely recognised as impacting upon peoples' sense of wellbeing and QoL (Bilanchini, 2012; Booth, 2012; Haworth, 2007; Galloway, 2006; Potter et al., 2012).

At policy level, wellbeing and QoL are both recognised as being influenced by factors and processes at the individual and global level. These may pertain to objective characteristic and or a subjective assessment (Atkinson 2013; Camfield and Skevington, 2008; Galloway, 2006). This duality has made them attractive as measurement tools (Scott, 2012c). QoL became one of the fastest growing areas of research and policy (Sarvimaki, 2006; Snoek, 2000). QoL is an important outcome measure, particularly in health and social care service delivery (Bowling and Gabriel, 2007; Galloway, 2006). In addition to which, it also plays an important role acting as a bridge for cross sector public partnership working (Hamilton and Scullion, 2006). Sarvimaki (2006) argued that this was because the concept has intuitive appeal as a measure of the wellbeing of individuals, communities and nation states.

Hamilton and Scullion (2006) noted that conceptually and pragmatically QoL and wellbeing have become important tools in public policy and service provision.

Wellbeing and QoL have both been promoted in recent years as measures or indicators of societal progress, which look beyond traditional economic measures of societal progress (Abdallah and Shah, 2012). Both terms, for example, are heavily used within service provision to evaluate the efficacy of service delivery (Lim, 2001; Snoek, 2000). This is particularly the case with health and social care services.

The shared characteristics of wellbeing and QoL mean the terms are also frequently used interchangeably across a range of government legislation (Galloway, 2006). The use of wellbeing and QoL interchangeably can be found in documents such as 'Every Child Matters: Change for Children' (DfES, 2003); NSF for LTCs (2005) and the white Paper 'Caring for our future: reforming care and support' (DoH, 2012).

At the theoretical level, there is general acknowledgement that conceptual similarities exist between the concepts. Both concepts are generally acknowledged as having both objective and subjective aspects (Brown et al., 2004; Camfield and Skevington, 2008; Galloway, 2006; Gasper, 2010; Sarvimaki, 2006). Implicit in many definitions of wellbeing and QoL are the connections between life satisfaction and their subjective and objective characteristics (Galloway, 2006). The attainment of QoL and wellbeing are both associated with objective conditions external to existence, such as living conditions or access to education and internal characteristics such as self-efficacy and self esteem (Bowling, 2000; Bowling et al. 2003; Carpenter, 1997; Christakopoulou et al., 2001; Diener and Lucas, 2000; Potter et al., 2012; Tailleffer et al. 2003).

Similarly, the importance of personal values and subjective assessments has been increasingly promulgated within wellbeing and QoL research (Galloway, 2006). Wellbeing and QoL are both perceived as being influenced by both positive and negative experiences and affect, and by valuations and self-evaluations of life circumstances, which may change over time in response to life experiences and changing health.

Bowling et al. (2003) claimed that the "Most frequently reported empirical associations with both well-being and quality of life in older age are good health and

functional ability, a sense of personal adequacy or usefulness, social participation, the existence of friends and social support, and level of income" (p.273). At an empirical level, there is long and well-established evidence to support these claims (Bowling et al. 1996, 2003, 2005; Brown et al., 2004; Gabriel and Bowling, 2004).

Within empirical research, researchers have identified that wellbeing and QoL suffer from a number of practical issue. For example, what constitutes the determinants of wellbeing/QoL, versus what constitutes its essence per se (Ryan and Deci, 2001; Shumaker et al.1990). There is therefore a need to acknowledge that both concepts require clarification in order to facilitate their application in practical and policy usage (Langlois and Anderson (2002).

At an academic level disciplines such as psychology and economics perceive both concepts as dynamic constructs (Bowling, 2005a; Bowling and Gabriel, 2003; Bowling et al. 2009; Haworth and Hart, 2006). Additionally both are widely adopted as umbrella terms and utilised across diverse academic disciplines (Galloway, 2006). Wellbeing and QoL research incorporates topics from across the academic spectrum (Galloway, 2006; Hamilton and Scullion, 2006; Langlois and Anderson, 2002). The multidisciplinary nature of these concepts facilitates diverse approaches which incorporate theoretical ideas from disciplines across the academic spectrum (Hamilton and Scullion, 2006; Ryan and Deci, 2001).

Despite areas of commonality between wellbeing and QoL there are conflicting views of what is meant when referring to QoL and wellbeing and whether these terms refer to the same thing. However, both terms have become part of the lexicon of public policy narratives (Association of Public health Observatories, 2006; Eraurt and Whiting, 2008; Seedhouse, 1995, 2005). Research, therefore, has an important role to play in the exploration of the commonalities and differences between theoretical and empirical understandings of wellbeing and QoL. This is important given that process and outcomes measures of wellbeing, and or QoL are used extensively particularly in health and social care settings (Abdallah et al., 2012; Association of Public health Observatories, 2006).

Having considered the commonality between wellbeing and QoL the following section explores the differences between them.

### Differentiating between wellbeing and QoL

Smith (1973) was an early advocate for differentiating between QoL and wellbeing. He argued that the term wellbeing referred to the objective life conditions of a population, whilst QoL referred to people's subjective assessments of their lives. However, as QoL research became widely established conceptual differences between the concepts lapsed. The emergence of wellbeing as a statutory duty of Local Authorities (Local Government Act, 2000) however focused greater attention on what was meant by the term wellbeing and whether wellbeing was distinct from similar terms such as QoL, life satisfaction and welfare (Scott, 2012c).

Langlois and Anderson (2002) provide an example of theoretical research which differentiates between wellbeing and QoL. Their research was based on the supposition that QoL pertains to process, whilst wellbeing pertains to a dynamic process and a state of being. In practical terms, one could equate this in service provision as utilising QoL as a process measure whilst utilising wellbeing as an outcome measure.

It has also been argued that QoL is more widely associated with functioning (Bowling et al., 2003; Gabriel and Bowling, 2004). Lhussier (2006) for example argued that QoL within healthcare has traditionally been based on improving functionality and "Prolonged life expectancy" (p.23). Whilst some theoretical accounts of wellbeing also associate wellbeing with functioning such as the Capabilities Approach (Sen, 1984, 1985b, 1993) and Capabilities List (Nussbaum, 1988, 1992, 1995, 2003a) there is growing evidence that this association has more to do with societal perceptions than lay realities (Edwards and Imrie, 2008).

Another area in which wellbeing may be said to differ from QoL is the binary nature often ascribed to it (Ryan and Deci, 2001; Woolryche and Sixsmith, 2008; Hobbs and

Sixsmith, 2009). QoL is often perceived as reducible to objective and subjective domains (Lui, 1976; Pais-Ribeiro, 2004). These can be assessed through objective and subjective criteria. However conceptually QoL is not thought of in binary terms.

Wellbeing however is often perceived in binary terms such as eudaimonic wellbeing and hedonic wellbeing, objective wellbeing and SWB, and as a multi-dimensional or a uni-dimensional concept (Bilancini, 2012; King, 2007; Ryan and Deci, 2001). These inherent contradictions in the nature of wellbeing reflect its long and complex historical linguistic development. If we take the example of wellbeing as a multi-dimensional or as a uni-dimensional concept one can see that there is a conflict between researchers at a basic level. Some perceive wellbeing as a multi-dimensional concept (Christakopoulou, 2001; Haworth, 2007). Others perceive it as a uni-dimensional concept (Beattie and Gott, 1993; Danna and Griffin, 1999). Whilst there are those who perceive it as both multi-dimensional and uni-dimensional (Ryan and Deci, 2001). This latter position holds increasing currency with wellbeing researchers' (Cooper, 2014).

Bond and Corner's (2004) research into QoL suggested there was a need for researchers to draw distinctions between the objective indicators used to assess QoL, from the subjective assessments used to measure the subjective experiences of wellbeing. They suggest that QoL has a populist appeal and utilisation in everyday language. This offers researchers a sensitising concept and a heuristic device, which can be used to investigate how lay people perceive life satisfaction. This may offer research a useful means of bridging lay and professional interpretations of what constitutes a life well lived.

The following section offers an overview of arguments concerning the extent to which wellbeing and QoL may be considered conceptually analogous. This debate has grown in recent years as utilisation of the two concepts within public policy has grown (Camfield and Skevington, 2008; Gasper, 2010; Hamilton and Scullion, 2006; Langlois and Anderson, 2002).

# **Ontological perspectives**

Gasper (2010) contended that whilst wellbeing and QoL are broadly based on evaluative judgements there remains insufficient consideration in current literature that usage of the terms resides in their differing value to researchers. Furthermore, Gasper (2010) positions himself in relation to Sens' precept, that these concepts are subjective and based on the value system of those who employ them. Therefore, a researcher's decision to use wellbeing or QoL is often predicated upon their judgement, valuation and positionality (Gasper, 2010). Gasper (2010) argues that this is an inevitable aspect of research concerned with the exploration of subjective experiences.

However, Gasper (2010) states that "Compared to work on well-being, work on quality of life more often tends to take a standpoint within a public decision-making context and thus looks also at necessary conditions within the social environment" (p.357). Gasper (2010) also notes that wellbeing tends to be used in reference to individuals and concrete experiences, whilst QoL is more likely to be used in reference to contextualise "Communities, localities, and societies" (p.251). Gasper (2010) argues that "We need and use different QoL and WB concepts for different purposes" (p.355). This utilisation of verstehen, i.e. an appreciation of others' meaning and perceptions is key in the debate regarding the degree to which wellbeing and QoL may be considered synonymous (Gasper, 2010).

For Gasper (2010), whether QoL and wellbeing are the same or different concepts revolves around conceptions of being and or the nature of human life. "Different conceptions of well-being and quality of human life reflect different understandings of be-ing and of the structures of human life, in other words different ontology's and perspectives about the nature of persons and societies." (Gasper, 2010:357). In addition to which research within these two subject areas use different research instruments and have differing foundations such as positionality, purpose, theoretical views and ontological presuppositions (Gasper, 2010).

In some disciplines, the terms wellbeing, QoL and welfare are used interchangeably as if they are synonymous (Galloway, 2006). However, in disciplines such as health, economics or sociology, wellbeing and welfare are generally considered quite separate concept, although wellbeing and QoL are often conflated (Galloway, 2006). In disciplines such as psychology and philosophy however the three terms are widely perceived to be related but distinct concepts (Galloway, 2006).

At a theoretical level, philosophy has contributed much to the debate about whether wellbeing and QoL are analogous. In philosophy, wellbeing is widely attributed as having an ethical and or moral dimension, which is not widely considered present in the concept QoL (Sumner, 1999; Tiberius and Plakias, 2010; Bilancini, 2012). Within the study of 'Ethics' for example, some of the most challenging theoretical debates relate to the nature, function and role of wellbeing and its relationship to moral responsibility (Griffin, 1986, 2010; Raz, 2004; Tiberius, 2004).

Questions such as 'What role can and should wellbeing play in morality'? What role can and should wellbeing play in the ethics of governance? (Bentham, 1789; Bilancini, 2012; J., S. Mill, 1859) have plagued philosophers since antiquity (Bergdolt, 2008; Ryan and Deci, 2001) and have led to a range of theoretical expositions (Wootton, 2015). These long established theoretical debates whilst common in wellbeing research are rarely addressed under the auspices of QoL research. When they are addressed, it is usually in relation to individuals with cognitive impairments, their capability to assess their own QoL and whether it is ethical for external authorities to impose an assessment of QoL (Galloway, 2006)

In seeking to clarify the conceptual difference between QoL and wellbeing, I have drawn upon the work of authors such as Sumner (1999). Sumner (1999) proposed that the strong philosophical dimension embedded within wellbeing differentiates it from other concepts such as QoL and welfare. In addition to which I would add that wellbeing might fruitfully be considered a concept, which seeks to guide action rather than describe action (Tiberius, 2004, 2013; Tiberius and Plakias, 2010). It is these philosophical arguments, which add weight to the supposition that wellbeing

has an explicit moral/ethical dimension which at best, may only be considered implicit in QoL.

### Can and should we use wellbeing and QoL interchangeably?

There are those who argue that wellbeing and QoL can be used interchangeably (Burchi and Gnesi, 2013; Christoph and Noll, 2003; Lhussier, 2009). These arguments are often predicated on quite different assumptions. However, there are significant numbers of researchers, policy makers and academics who use the terms synonymously. Christoph and Noll (2003) one such example, argued that wellbeing, welfare and QoL can all be used interchangeably. Their argument is based upon circularity of definition and linguistic association. Christoph and Noll (2003) claimed that SWB is a component of 'welfare'. Welfare is in itself a term, which they imply, is interchangeable ...with QoL . Thus for Christoph and Noll (2003) the terms SWB, welfare and QoL can legitimately be used interchangeably.

Schalock (2000, 2004) contributed much to the debate about whether wellbeing and QoL can and should be used interchangeably. Schalock argued that wellbeing and QOL are such closely related concepts that they can be used synonymously (Buntinx and Schalock, 2010; Schalock, 2000, 2004; Schalock et al. 2010). Schalock's (2000) definition of individual QoL for example intimated that wellbeing is a determinant of QOL. Within this definition, however he also argued that QoL is a constituent aspect of personal wellbeing (Schalock, 1996, 2000; Schalock et al., 2010). The circularity of definition within the work of Schalock is also evident in the work of others (Sarvimaki, 2006).

Schalock's argument however is undermined by the fact that the precise relationship between the concepts has not been established and is far from clear. This suggests that the relationship between the two concepts remains too ambiguous for researchers to use the terms synonymously with any great confidence (Sarvimaki, 2006).

Tailleffer et al. (2003) in a systematic review of health related QoL models reported that in one third of QoL models, which utilised the concept of wellbeing, QoL was

defined explicitly or implicitly as wellbeing. This means that two thirds of the models classed the two concepts as distinct. This suggests there is only qualified support for equating the two concepts and using them interchangeably.

The increasing use of QoL and wellbeing indicators in policy and planning since the Local Government Act (2000) has meant that the extent to which the terms wellbeing and QoL can be considered interchangeable continues to occupy the attention of researchers (Galloway, 2006; Gasper, 2010; Scott, 2012c).

Lhussier (2009) for example proposed that regardless of whether wellbeing and QoL are different concepts researchers should blur the boundaries between the concepts. Lhussier (2009) argued that maintaining these conceptual boundaries has led to a stagnation in QoL research and undermines efforts to operationalise it. Furthermore Lhussier (2009) claimed that blurring the boundaries between QoL and wellbeing will liberate QoL research from current constraints and facilitate the development of alternative approaches to QoL research and practices. These will challenge established understandings and ontological characterisations of QoL, facilitate wider representations and conceptualisations of QoL and assist in efforts to operationalise it (Lhussier, 2009).

More recently, research undertaken by Burchi and Gnesi (2013) suggests that wellbeing and quality of life are now considered almost analogous concepts. This assumption is grounding in the conviction that both concepts are multidimensional and is used by researchers who are motivated to investigate the conditions, which underpin peoples' life satisfaction.

Having considered the theoretical arguments for using wellbeing and QoL as interchangeable concepts we can consider the empirical arguments. Empirical QoL research, which focuses on lay conceptualisations of the good life, have tended to report that lay people conflate wellbeing and QoL (Bowling, 2005; Gabriel and Bowling, 2004, 2007; Gilroy, 2007, 2008; Ryff, 1989b). Gabriel and Bowling (2004) investigated lay definitions of QoL and reported that 99% of the 80 older people they interviewed identified health and psychological wellbeing as important

components in their definition of QoL. This offers empirical evidence of the way in which lay people conflate their understandings of QoL, health, and wellbeing. This may be indicative of lay people adopting a holistic understanding of life satisfaction, which is absent in professional and academic conceptualisations.

### Issues with using wellbeing and QoL interchangeably

Whilst empirical QoL research indicates that wellbeing and QoL can be used interchangeably, there are fundamental concerns in adopting this approach.

Camfield and Skevington (2008) acknowledged that whilst both concepts can be perceived as objective and subjective, further definitional and conceptual work is required to appreciate the relationship between them. Camfield and Skevington (2008) highlighted the broad problems in using wellbeing and QoL interchangeably. They suggested that researchers conducting empirical studies to explore what make human lives worthwhile need to posit fundamental enquiries. They proposed a series of questions, which might help clarify the relationship, and the extent to which the terms can be used synonymously. These questions suggest lines of enquiry such as:

"Is QoL synonymous with SWB or with WB in general? Is the QoL concept sub-ordinate to SWB or could SWB be nested within QoL? If QoL is not the same as SWB how much do these concepts differ semantically and conceptually? If they overlap conceptually, then in what ways and to what extent does this occur?" (Camfield and Skevington, 2008:764).

Camfield and Skevington (2008) concluded, "That the new definition of SWB derived by an expert panel (WHOQOL Group, 1995) now displays high convergence with an inter-national definition of QoL" (p.770). This led them to suggest that SWB and subjective QoL are virtually synonymous with each other (Camfield and Skevington, 2008). Research such as this and growing inter-disciplinary approaches to the study of peoples' life conditions, offers authors and researchers some

justification for using wellbeing and QoL interchangeably (Camfield and Skevington, 2008).

There are however, concerns that using wellbeing and QoL interchangeably is further complicated by issues such as whether wellbeing and QoL are multidimensional or uni-dimensional concepts and which dimensions are inherent or embedded within the concepts (Bowling and Gabriel, 2007; Gasper, 2010; Galloway, 2006). There has been a growing tendency for wellbeing and QoL researchers to argue that these concepts are multi-dimensional. Given this position, it becomes important for clarification purposes that researchers identify which domains are directly transferable to the other concept (Galloway, 2006). This is particularly important at an operational level, where the synthesis of the concepts may not always be a perfect fit and may undermine already complex assessments of these concepts as process and outcome measures (Galloway, 2006).

These are issues which future research could explore in order to explicate understandings of the relationship between wellbeing and QoL.

#### Wellbeing and QoL as similar but not synonymous concepts

Sarvimaki (2006) drew upon Heidegger's philosophy of 'being' to posit that whilst wellbeing and QoL are similar they are not synonymous. Wellbeing for example is widely perceived as referring to all three temporal states (Atkinson 2011, 2013; Durayappah, 2010; Schwanen, and Ziegler, 2011; Schwanen and Wang, 2014). QoL however, is traditionally and fundamentally associated with an individual's current life circumstances (Atkinson 2013; Bond and Corner, 2004; Sarvimaki, 2006; Smith, 1973; Zikmund, 2003). Indeed QoL is often perceived in relation to the "Everyday unfolding of life" (Sarvimaki, 2006: 9). Bond and Corner (2004) who suggested that QoL is a concept, which lends it to the language of the everyday, support this positionality.

There exists a continuum of opinion with regard to the extent to which QoL and wellbeing may be considered similar but not synonymous (Katz et al., 2012). Some perceive psychological wellbeing, SWB and personal wellbeing as a component,

dimension or domain of QoL (Katz et al., 2012; Nordenfelt, 1991b; Phillips, 2006; Vincent et al., 2006). This has been supported by empirical evidence (Bowling, 2003, 2005; Bowling and Gabriel, 2003; Bowling et al., 2003; Gabriel and Bowling, 2004).

Others perceive QoL to be a component, dimension or domain of wellbeing (Christoph and Noll, 2003). Bond and Corner, (2006) for example, argued that QoL is a "Concept that is widely used as a measure of wellbeing" (Bond and Corner, 2006:154). The following section will consider in greater detail the contrasting arguments regarding whether wellbeing is a domain of QoL or QoL a domain of wellbeing.

## Subsuming one concept within another

There are many competing views about the relationship between QOL and wellbeing (Haas, 1999b). There are a number of authors, who regard wellbeing as one constituent of QoL. These authors tend to considere QoL to be a broader concept than wellbeing (Diener and Suh, 1997; Lane, 1996; Vittersø, 2004). Discipline backgrounds can often play an important part in perceptions of the relationship between wellbeing and QoL (Gasper, 2010). The theoretical utilisation of particular concepts within particular disciplines was discussed previously in the section covering ontological perspectives. However, at a practical level this has important implications. For example, those from economic backgrounds are more inclined to distinguish between wellbeing and QoL on the basis that wellbeing relates to the individual, whilst QOL is considered to be a comparative concept related to welfare between individuals (Galloway, 2006). The association between wellbeing, QoL and welfare further complicates the extent to which one can subsume one concept within another.

## Wellbeing as a QoL domain

Authors such as Vittersø (2004) argue that wellbeing should be perceived as a component of QoL as the term QOL combines both objective and subjective dimensions. For Vittersø (2004) SWB is subsumed within the subjective domain of QoL and relates to an individual's assessment of both the cognitive and affective aspects of their existence. Similarly Schalock (1996) proposed that wellbeing is embedded within QoL and should be construed as being related to the subjective aspect of QoL (Schalock, 1996).

Christoph and Noll (2003) also argue that wellbeing is a QoL domain. They subsume SWB within the broader umbrella of QoL based on the premise that welfare and QoL are coterminous and thus by process of association one can also assume that SWB is a component of QoL.

Others consider the constituent parts of wellbeing when arguing that wellbeing is a component of QoL. Bowling (2005) Gabriel and Bowling (2004) and Lawton (1991) argued that psychological wellbeing is considered the pre-eminent component of wellbeing across empirical and theoretical research. Using this premise Gabriel and Bowling (2004) argue that as psychological wellbeing is located within the broader conceptual framework of QoL, this perforce means wellbeing can be perceived as a QoL domain.

Other authors such as Lindstrom (1994), Ferrell, (1995) and Langlois and Anderson, (2002) however propose that QoL can be more productively perceived as a domain of wellbeing.

# QoL as a wellbeing domain

Lindstrom's (1994) QoL model provides an early example of those who perceive QoL as a domain of wellbeing. Lindstrom (1994) developed a QoL model in which QoL, subdivided into four life realms, was embedded within the wider ambit of wellbeing. These four realms were represented as the global, external,

interpersonal and personal realms. Lindstrom (1994) identified a further subdivision of the personal realm, which she perceived was an amalgamation of physical, cerebral and spiritual facets. It may be that the spiritual facet of the model (widely perceived as central to wellbeing) played a role in Lindstrom's (1994) model embedding QoL within wellbeing.

Ferrell (1995) meanwhile identified QoL as underpinning wellbeing. Ferrell's (1995) understanding was predicated on the premise that wellbeing spans the four domains of human existence, i.e. physical, mental, social and spiritual wellbeing (Ferrell, 1995). Langlois and Anderson (2002) however perceived the relationship between wellbeing and QoL somewhat differently. Rather than perceiving QoL as underpinning wellbeing they suggest that "QoL can be seen as setting the stage for potential wellbeing" (p.509).

Lindstrom (1994), Ferrell (1995) and Langlois and Anderson (2002) perceptions of QoL as a domain of wellbeing has been enhanced by the work of Sarvimaki (2006). Sarvimaki (2006) conducted an overview of a range of definitions and models of 'wellbeing', 'health' and 'QoL'. Sarvimaki's (2006) findings indicated that wellbeing was a characteristic of health and QoL. This led Sarvimaki (2006) to argue that wellbeing could be perceived as the "Ultimate criterion of QoL" (p.4/5). In addition to which he suggested that wellbeing has the potential to act as a unifying concept across discipline boundaries.

#### Wellbeing and QoL as two distinct concepts

Smith (1973) wrote widely around the utilisation of social indicators and their application within US state policy. Langlois and Anderson (2002) identified Smith (1973) as one of the first authors credited with establishing the argument for conceptual distinction between the terms wellbeing and QoL. Smith's (1973) seminal study of social indicators differentiated between wellbeing and QoL, primarily on the premise that QoL was concerned with the objective assessment of conditions of life, whilst wellbeing was concerned with the subjective assessment of one's existence. Subjective assessment is now customarily recognised as a basic

requirement for the evaluation of wellbeing (Barnes et al., 2013; Bjornskov, 2012; Marks and Shah, 2004; Ryan and Deci, 2001). This distinction between the objective and subjective seems to offer wellbeing and QoL researchers reasonable grounds to support the notion that wellbeing can and should be used in both concept and application as distinct from QoL (Langlois and Anderson, 2002).

Haas (1999a) illustrated some of the associations, convergences and divergences between QOL and related concepts. Haas (1999a) identified wellbeing, life satisfaction, functional status and health status as concepts related to QoL. Haas (1999a) identified that whilst these concepts had broad and close relationships they were not identical. Haas (1999a) proposed that QoL was inadequately defined in part because QoL was equated with closely aligned but not synonymous concepts. She called upon researchers especially those within the field of QoL research to begin to unpack the subtle differences which exist between the concepts (Haas, 1999a),

Langlois and Anderson (2002) have perhaps been amongst the most vociferous in arguing that wellbeing and QoL are not synonymous. Langlois and Anderson (2002) reviewed QoL and wellbeing literature from 1970s to 2000. Based on their findings they argued that wellbeing and QoL are two discrete concepts. However, they identified that wellbeing and QoL researchers need to develop "A meaningful understanding of the quality-of-life and/or well-being literature" (p.502).

Unlike Smith (1973) however, Langlois and Anderson (2002) proposed that the conceptual difference between the concepts resides in healthcare practice. Here wellbeing is used in reference to objective life conditions whilst QoL is more readily aligned with individuals' subjective assessments of their lives. This contrasts with Smith (1973) position and may be indicative of the shift in emphasis, which developed during QoL research in the 1980s, and 1990s when increasing attention was paid to QoL as a subjective assessment.

Langlois and Anderson (2002) claim that the objective versus subjective distinctions used to distinguish between wellbeing and QoL, originated from those such as Smith (1973). They argue that this distinction is no longer applicable, as both

wellbeing and QoL are now widely perceived as incorporating objective and subjective assessments.

Langlois and Anderson (2002) contend that, whilst there is evidence to support the notion that the two concepts are conceptually linked, they are none the less distinct concepts. Langlois and Anderson (2002) suggest that there has been a conceptual erosion of the distinctions between the concepts. They suggest this erosion is the result of the recognition that wellbeing and QoL both have objective and subjective aspects.

Langlois and Anderson (2002) findings suggest there is a distinction between wellbeing and QoL. Moreover, this distinction is conceptually advantageous. Langlois and Anderson (2002) proposed that the distinction facilitates "Greater precision in the investigation of these multidimensional concepts" (p.506). This should be recognised within and across disciplines, particularly given that both are multidisciplinary and or trans-disciplinary concepts.

Zikmund (2003) has also proposed that wellbeing and QoL are distinct concepts though he acknowledges that conceptually wellbeing is closely related to QoL. Both concepts are concerned with the satisfaction of material, biological, psychological, social, and cultural needs and demands of an individual, which are necessary for life satisfaction. However, Zikmund (2003) proposed that use of the term of wellbeing is more closely associated with the present and immediate state of life satisfaction experienced by the individual. QoL on the other hand, is more concerned with the complex conditions which enable satisfaction with life. This may relate to an earlier concern raised by Gasper (2010) which cautions researchers against conflating determinants with inherent characteristics.

Bowling et al. (2003) a renowned QoL researcher proffered support for Spiro and Bossé's (2000) contention that "While overlapping to some extent, quality of life is conceptually distinct from well-being and satisfaction (Bowling et al., 2003:271). Bowling is one of the most widely published UK QoL researchers and is one of a strong contingent of researchers who argue that wellbeing and QoL are related but distinct concepts (Bowling and Gabriel; Gabriel and Bowling; Brown et al., 2004;

Haas, 1999a, 1999b; Gasper, 2010; Langlois and Anderson, 2002; Sarvimaki, 2006; Smith, 1973; Zikmund 2003).

It would appear that over time the arguments concerning whether wellbeing and QoL are distinct concepts have developed from early distinctions. These include those made based on the binary juxtapositioning of subjective and objective nature of the concepts (Gasper, 2010). In more recent times the distinction between wellbeing and QoL have developed into more sophisticated arguments. These argue that the concepts serve different purposes, perform different roles, have different relevance, stem from different disciplines and are an indication of researchers discipline backgrounds (Gasper, 2010).

#### Conclusion

It is to be hoped that this review has highlighted the complexity of the debate about whether wellbeing and QoL are synonymous concepts. Given the evidence presented above, one could say that whilst the majority accept that wellbeing and QoL are closely related concepts they do not as a matter of course accept that the two are synonymous.

As documented earlier, there are authors who use a multiplicity of terms when alluding to 'a life well lived'. However, given the paucity of conceptual clarity and the large number of definitions which exist for the three main conflated concepts 'wellbeing' 'QoL' and 'welfare' it would appear ill conceived for researchers to use these terms unreflectively and without offering the reader a clear rationale for using these terms interchangeably (Gasper, 2010).

Empirical research undertaken in the UK around lay understandings of QoL, most notably by Bowling (Bowling, 1995, 2003, 2005; Bowling and Gabriel, 2004) found that psychological wellbeing was of key importance in lay understandings and definitions of QoL. Similarly, other aspects identified as important such as health, social relationships, the home and neighbourhoods have considerable intersection with evidence of what is important for wellbeing (Gilroy, 2007, 2007). However,

little research has been undertaken to explore lay understandings of wellbeing. This study is therefore important in helping to explore the extent to which lay participants conflate understandings of wellbeing with those of QoL. As such, this study may help to ascertain the degree to which people in older age groups consider wellbeing analogous with QoL.

It is apparent from the discussion presented here that wellbeing and QoL share characteristics and domains. The degree to which wellbeing and QoL can be considered synonymous however, remains disputed and using the terms interchangeably warrants future investigation (Smith, 2000; Gasper, 2010). However, there is increasing support for those who argue that the term considered most appropriate for a particular research purpose is specific to each researcher (Galloway, 2006; Gasper, 2010; Sarvimaki, 2006; Smith, 2000).

Perhaps the final thought should come from Gasper (2010) who suggests that whilst wellbeing and QoL terminologies broadly overlap, their conceptual and applied differences are a strength we should celebrate.

# **Appendix C: Literature Review Search Methods**

#### Introduction

This following sections includes a detailed presentation of the process and procedures used to identify and assess the relevance of literature included in the literature review. A literature review is a scientific approach which identifies and describes the existing knowledge related to the problem under examination (Glasper and Rees, 2012). Literature reviews recognise the gaps in existing knowledge and provide an explanation of how the current study might generate new knowledge (Burns and Grove 2007; Clarke 2007).

The quality of any literature review may be judged by its "Claim to be comprehensive, have up-to-date references, be transparent, well-balanced and without bias. It should demonstrate credible analysis and synthesis, be well-organised, well-presented and offer sufficient detail to ensure it could be replicated" (Polit and Beck 2006:12). The literature review process has sequential steps, which provide a framework for a study (Glasper and Rees 2012). This study followed the nine steps required for an effective literature review, as suggested by Polit and Beck (2008). The parameters of the literature review were governed by the research questions and objectives. See chapter 1 for a summary of the research questions and objectives.

The researcher undertook manual searches of literature contained in the libraries of Manchester Metropolitan University and the University of Manchester.

Electronic databases and general internet searches were also utilised to identify relevant literature (Glasper and Rees 2012).

#### Literature search

The search strategy was formulated in order to identifying the most useful search terms and associated concepts. Recognition of search words is a requirement in searching studies (Glasper and Rees 2012). The 'keywords' or 'search terms' refer to a specific word or phrase for searching relevant literature from databases (Glasper

and Rees 2012). For this study keywords, concepts and alternative terms synonymous with each concept were identified and searched for accordingly. Additional methods such as 'truncation' i.e. using a term with \* after the word and 'Boolean' i.e. using AND/OR were also utilised with the terms to generate more inclusive returns.

#### Search terms

Search terms included wellbeing, personal wellbeing, subjective wellbeing, eudaimonic wellbeing, hedonic wellbeing, objective wellbeing, psychological wellbeing, social wellbeing, community wellbeing, physical wellbeing, environmental wellbeing. Searches for the term wellbeing and its variant spellings proved problematic owing to the generic nature of the term. For this reason, searches using this term were restricted to study titles and additional exclusion criteria were used. All searches were limited to English language articles or those which had been translated into English. No restrictions were placed on the publication date of articles.

Wellbeing searches included terms used to describe wellbeing where these were identified as a component of wellbeing. These included quality of life (QoL), life satisfaction, happiness, subjective wellbeing (SWB) and welfare. Health searches included terms to describe illness such as long-term conditions, complex health needs, disability, ill health etc. Age searches included terms to describe mature adults such as older adults, older people, aged populations, the elderly etc. These were combined with other terms such as 'health and wellbeing in the third age' used in relation to life-course, wellbeing and health. Historical searches included terms such as the 'historical development of wellbeing', 'utilitarianism and wellbeing' and 'historical analysis of wellbeing'. Linguistic searches included terms such as the 'linguistic development of wellbeing', 'etymology of wellbeing', 'ontology of wellbeing', 'logomachy', 'wellbeing lexicography' and 'semantics of wellbeing'.

# Decision pathway for selection of material

The literature search involved four stages. The first stage identified potential references using a range of search strategies and inclusion/exclusion criteria. The second stage was based on retrieving references which had been identified in stage one and reading the abstract to ascertain relevance to study parameters. The third stage involved using references retained in stage two as the basis for the literature review. The forth stage was undertaken to ensure the review was as comprehensive and up to date as possible.

The process of identifying relevant research for this literature review centred on reviewing general wellbeing literature identified through various search engines, databases and grey literature. An additional measure included the identification of sources from the bibliographies of pre-selected material. These were retrieved and read for inclusion into literature review. Searches were conducted from inception of thesis (September 2007) to June 2015.

## **Eligibility criteria**

Literature was included if they were published in a peer-reviewed journal and published or translated in English.

#### **Exclusion criteria**

This literature review was subject to the restraints necessary when conducted by a lone researcher (Glasper and Rees 2012; Polit and Beck, 2008). In searching for research on wellbeing, the objective was to identify articles in which the title suggested that wellbeing was explored in relation to its theoretical, conceptual or empirical application.

In order to ensure that this focus was achieved, a range of exclusion criteria was employed, these included:

- Wellbeing reviewed with specific regard to its measurement.
- Wellbeing reviewed within specific contexts, which were beyond the parameters of this research such as schools.

- Wellbeing reviewed within the context of a specific medical programme
   /intervention or treatment regimen, which were beyond the parameters of
   this research.
- Wellbeing reviewed within the parameters of specific lifestyle concern,
   which were beyond the parameters of this research such as those relating to sexual practices.

These exclusion criteria were relaxed when the result indicated that the research related to older adults and or complex health needs.

#### Stage one

Searches were conducted via:

- Electronic database searches
- Web searches
- Web sites of key organisations and research organisations
- Journal searches
- Bibliographies

#### **Electronic databases**

The main electronic databases used including the Social Sciences Citation Index (SSCI), Cumulative Index to Nursing and Allied Health Literature (CINAHL), The Cochrane Library, Applied Social Sciences Index and Abstracts (ASSIA), EMBASE, MEDLINE, PsycINFO, Pub med, Science Citation Index Expanded, Arts and Humanities Citation Index.

Unpublished literature such as PHD theses and literature not published by traditional publishing companies is known as grey literature (Glasper and Rees 2012). Grey literature might include university publications, policy publications from external bodies or from government departments (Glasper and Rees 2012). The World Wide Web is now widely recognised as the principal repository of grey literature (Glasper and Rees 2012). However, university repositories such as the

University of Oxford and Cambridge are also widely used in searches for grey literature (Glasper and Rees 2012). The researcher also undertook additional searches of grey literature. These articles or papers often proved to be a useful resource for the acquisition of background knowledge. Whilst they helped to inform knowledge acquisition, they were not included in the literature review.

### Web based searches for wellbeing

These searches often identified large numbers of irrelevant material because of the broad nature of the term so the following additional exclusion criteria were applied:

- Results which related specifically to QoL, life satisfaction, happiness and welfare
- Results which related specifically to the Welfare state
- Results which related specifically to the word 'well' or 'being' in the title
- Book reviews

## Web sites of key research centres/organisations

The main research centres and organisations used included:

- Cambridge University Wellbeing Institute
- New Economics Foundation (nef)
- Wellbeing in Developing Countries Research (WeD)
- Centre for Research on Ageing, Health and Wellbeing (CRAHW)
- The Australian National University of Canberra's Advanced Wellbeing Research Centre (AWRC)
- The What Works Centre for Wellbeing
- The Organisation for Economic Co-operation and Development (OECD)
- The World Health Organization (WHO)

# Journal searches for wellbeing

Manual searches were made of key journals such as the 'International Journal of wellbeing', 'British Journal of Wellbeing', 'Journal of wellbeing', 'The Journal of Happiness and Well-being' and 'Qualitative studies on Health and Wellbeing'.

#### Bibliography searches

The bibliographies of papers relating to wellbeing, older adults and complex health needs were searched for relevant references.

#### Results

The search returned thousands of references. Titles and abstracts were screened for relevance, those within the study parameters were retained whilst the remainder were excluded. Where there was any doubt with regard to the paper's relevance the source material was retained. Once the exclusion criteria had been applied and the references screened hundreds of resources had been identified.

## Stage Two

Over 300 references remained after stage one so the criterion was further refined. The next stage of the selection process was based on the knowledge of the researcher.

The selection of references for inclusion in the literature review was further refined by applying the following exclusion criteria:

- Where the key focus of paper was not related to wellbeing.
- Where the key focus of paper was concerned with the wellbeing of young people.
- Where the key focus of paper was not explicitly directed at the theoretical,
   conceptual or empirical exploration of wellbeing.

#### Results

Some of the papers portrayed wellbeing either relatively narrowly, for example as psychological wellbeing or physical functioning, or too broadly, for example as life

satisfaction or happiness. Other papers arbitrarily and unreflectively accepted conceptualisations of wellbeing as being synonymous with QoL, life satisfaction, happiness or welfare. Many failed to clarify their epistemological position, provide an explicit definition of what they understood wellbeing to mean or failed to identify the parameters they used to characterise wellbeing. Cameron et al. (2006) noted, "Wellbeing is often used in an unreflective manner, involving assumptions either implicit and or explicit with little systematic attention... and typically leaves the term well-being as an open-ended, catch-all category" (p.349). The researcher found that a significant number of articles used wellbeing in the title of their paper but closer reading of the abstract indicated that this might refer solely to specific aspects of wellbeing such as physical or mental functioning. Some articles found in the search used wellbeing in the title but closer inspection of the paper revealed that the study was concerned with areas of concern such as health-related QoL. QoL, happiness and life satisfaction are frequently conflated with wellbeing. However, whilst these are closely aligned concepts the researcher believes these terms are not synonymous with wellbeing and so these articles were discounted unless the study expressly identified wellbeing as a component of QoL/happiness /life satisfaction or vice versa. See Appendix B for a discussion of conflating wellbeing with other analogous terms.

## **Stage Three**

Following the process employed in stage two, and the ongoing identification of further references from bibliographies, over one hundred references were included in the literature review. The final stage of the search process involved reviewing the relevance of the material to the study parameters. This was challenging as the synthesis of qualitative findings in reviews is a developing discipline (Carr et al., 2011).

The Cochrane Qualitative Research Methods Group acknowledged that there is a need for methodological work to explore how to synthesise study findings which use different qualitative methods and data types. Carr et al. (2011) however, have identified that data mined from literature searches should undertake a process of

data synthesis. This synthesis involves "The collation, combination and summary of the findings of individual studies included in the systematic review" (Carr et al., 2011:20).

Titles and abstracts therefore were screened again for relevance and those outside the scope of the review were excluded. Weighted preference was based on factors such as: those with a qualitative research focus, multi-disciplinary approach, cross-disciplinary evidence base, papers based on primary research and systematic reviews. In addition to which, the inclusion process took into consideration year of publication, population sampled and the utilisation of robust research methods. This enabled the chosen material to be organised in line with the study parameters.

Full texts of all remaining inclusions were obtained, screened for relevance, and then read for critical appraisal and data extraction. Those, which were considered to fall within the study parameters, were included in the following literature review.

# Stage 4

An additional forth stage was undertaken to ensure the review was as comprehensive as possible. This contained the original search parameters but went beyond the specification of the original literature overview.

#### Literature parameters

Given the plethora of research material, which addresses wellbeing and QoL research and the study's word count constraints it was not possible to include a review, which contained reference to both wellbeing and QoL. Wollny et al. (2010) noted it is challenging for authors to conduct literature reviews of wellbeing and QoL given that "The research literature describing theoretical concepts of 'human wellbeing' and the closely related term 'quality of life' is ... extensive" (p.21). Gasper (2010) argued that the diverse conceptualisations of wellbeing and QoL means preference for one term over another is determined by their unique pertinence and relevance to the author, study and research occasions. Given the diverse positionality which drives the utilisation of one term in preference to another, Gasper (2010) proposed that the need to appreciate more fully that the terms can

and do have different roles and conceptual underpinnings. Given the widely acknowledged distinctions between wellbeing and QoL, this researcher contends that it is unnecessary for a review of wellbeing to include a review of QoL research. For a full appraisal of the evidence to support this decision, see appendix B.

There is a vast and continually expanding wellbeing literature evidence base (Atkinson, 2011; Dodge et al., 2012; Fabiola et al., 2013; Galloway, 2006; Scott, 2014). This makes it extremely challenging for authors to undertake comprehensive literature reviews (Carlisle and Hanlon, 2007; Chavez et al. 2005). It is widely acknowledged that it is beyond the capabilities of a lone researcher to undertake a fully comprehensive review of wellbeing (Atkinson and Joyce, 2011; Atkinson et al., 2012; Scott, 2013; Stoll et al., 2012). Literature reviews in the two closely aligned fields of wellbeing and QoL are primarily selective reviews, which focus on studies of particular relevance to the author's work (Gasper, 2010; Ribeiro, 2004; Sirgy, 2012). This literature review has therefore adopted the selective approach utilised by Lhussier, (2006, 2009) drawing from a purposively selected literature review in keeping with the parameters of this thesis.

The first part of the literature review is based upon the components approach to wellbeing which underpins the empirical component of this study. Atkinson and Joyce (2011) and Atkinson et al. (2012) suggested that research and policy attempt to deal with the abstract nature of wellbeing by breaking it down into constitutive dimensions in what has been called a 'Components Approach to Wellbeing' (Atkinson and Joyce 2011; Atkinson et al., 2012). In this approach, debate centres on the identification and theorisation of the independent elements that comprise wellbeing (Atkinson, 2013). This approach has its critics. However, those who critique the structuring of literature reviews in this way acknowledge that given the complex nature of wellbeing and wealth of data which one is required to assess this remains the most popular approach (Atkinson, 2013; Atkinson and Joyce 2011; Atkinson et al., 2012; Scott, 2012, 2015).

The first part of this literature review is therefore structured around a components approach to wellbeing and summarises those domains of wellbeing widely perceived to be the principal components for personal wellbeing. However, as previously noted there has been growing criticism of this approach (Atkinson 2013; Atkinson and Joyce, 2011; Atkinson et al., 2012; Scott, 2013; Wollny et al. 2010).

Scott (2012, 2014) and Atkinson (2011) for example question the efficacy of utilising domain approaches to wellbeing and QoL research, which mean that research is conducted in silo, and fails to reflect the holistic nature of the concepts'. Given these valid criticisms of the components approach and the requirements of the theoretical investigation, the second part of the literature review adopts a 'Chronological Approach'. This offers a broader and more holistic overview of the development of wellbeing research (Atkinson, 2013). The second half of the literature review therefore utilises a chronological approach, which underpins the theoretical component of this study.

This literature review therefore includes component and chronological approaches, which considers theoretical and empirical research across a broad discipline base. This facilitates "Integrating the diverse domains and dimensions of wellbeing through a relational and situated account of wellbeing" (Atkinson 2013:138).

# Appendix D: Lay Versus Traditional Researcher: strengths and weaknesses

#### Introduction

There is growing debate concerning the strengths and weaknesses of using lay/service users versus traditional researcher-based research. This may be attributed in part to a growing interest in participatory research approaches, particularly within qualitative social research (Bergold, 2007; Bergold and Thomas, 2010; Bergold and Thomas, 2012). Indeed the involvement of lay people in the research process is increasingly required by many funding bodies (Becker et al., 2010; Bergold and Thomas, 2012; Staley, 2013).

A cadre of professional researchers have become increasingly concerned with the development, practicalities, opportunities and barriers of involving participants in research. Approaches which facilitate meaningful participant involvement in the research process have been explored by researchers such as Branfield and Beresford (2006), Bergold, (2007), McLaughlin (2009b), Bergold and Thomas (2010) and Cook (2012).

Conversely, there is also concern that the use of lay/service users in participatory research approaches has a detrimental affect on the research process (Bergold and Thomas, 2012; Brandstetter et al., 2014). The following discussion offers an appreciation of the benefits and challenges that are associated with these two approaches.

# Differentiated use of terminology within research

There has been much debate about labels or terminology assigned to lay people involved in the research process (Barnes et al., 2013; Bergold and Thomas, 2012; Branfield and Beresford, 2006; Dinham, 2006; INVOLVE, 2007, 2009, 2011; McLaughlin, 2009a). Terminology remains an important and contested aspect of wider debate within participatory research (Brandstetter et al., 2014; Earl-Slater, 2004; Entwistle et al., 1998; McLaughlin, 2009a).

Terms such as 'service users', 'participants', 'clients', 'customers', 'expert by experience', 'patients', 'lay people', 'consumer', 'user', 'co-researcher' and 'research partners' are all widely used and have supporters and critics (McEvoy et al., 2008). Some terms have faced particular criticism, for example, service users and consumers, which represents people as a homogeneous group. McLaughlin (2009a) for example, suggested the term service users leads to people being perceived through "The prism of their service users status" (p.15). Others have argued that terms such as 'consumer' attempts to fit research into the economic market model whilst the term 'patient' assigns it to the medical model (Benneworth et al., 2009; Branfield and Beresford, 2006; Hilbert, 2014; McEvoy et al., 2008; Ritchie et al., 2004).

Whilst it is difficult to provide a rationale for one's preference for one term over another, I have chosen to use the term 'lay' in this study. I would argue that 'lay' differentiates from 'professional' without fitting people into one type of model, or emphasising their position solely within a service context.

For an in-depth discussion around the differentiated use of terminology within research, see McLaughlin (2009a,b).

### Differentiated understandings of participation

It is generally agreed that the first comprehensive assessment of lay involvement in the research process was through Arnstein's (1971) 'Ladder of citizen participation' (Benneworth, 2009; Brandstetter et al., 2014; McLaughlin, 2009a,b). This approach to valuing participation has proved useful in assessing the degree to which involvement in research may be considered participatory (McLaughlin, 2009a,b).

However, it is widely agreed that there is no standard uniform way of conceptualising participation (Brandstetter et al., 2014; Earl-Slater, 2004). It is important therefore to consider the extent and quality of involvement when assessing the degree to which people may be considered active participants or coresearchers (McLaughlin, 2009a,b). It is also important to consider the closely aligned concepts of ownership and empowerment (Brandstetter et al., 2014; Ritchie et al., 2004; Smith, 2008).

The differential interpretation and employment of participation, involvement, ownership and empowerment impacts on whether research may be considered truly participatory (Brandstetter et al., 2014; Benneworth, 2009; Hilbert, 2014). Conceptualisations of participation, involvement and differentiated understandings of what these involve often vary substantially amongst researchers and participants (Blackburn et al., 2010; Brandstetter et al., 2014; Brett et al., 2010). Participation operates on different levels and is operationalised differently across disciplines and settings (Brandstetter et al., 2014). Some interpretations of participation are regarded as little more than tokenism (Brandstetter et al., 2014; Earl-Slater, 2004; Petersen and Lupton, 1996; Ritchie et al., 2004).

### **Principles of Participatory research**

Participatory research incorporates the principle of 'co-research' and has its origins in development contexts where researchers worked collaboratively with vulnerable and marginalised communities (Bergold and Thomas, 2012; McLaughlin, 2009b; Staddon, 2015; Staley, 2012, 2013). Wick and Reason (2009), for example, identified three distinct phases within the process of participatory research: the inclusion phase, the control phase, and the intimacy phase. Wick and Reason (2009) suggested that emotional issues, task issues, and organisational issues arise within each phase. These issues must be addressed in order to support the participatory research process (Wick and Reason, 2009).

The Beacon North East approach (2011) to community-university research partnerships proposed a number of core methodological principles underpin the participatory research process.

#### These include:

- 1) Cooperation: This centres on the principle of 'working with rather than on people'. This principle highlights the importance of meaningful collaboration i.e. dialogue between academics and participants to ensure that the research project is of interest and significance to those involved.
- 2) Participation: This stresses the value of participation and cooperation. This principle acknowledges the interconnectivity of life, highlighting interdependence and participants as active agents within the participatory research process.
- 3) Equality: This focuses on ensuring there is an understanding of and mutual respect for the knowledge and contributions of all those involved in the research.
- 4) Co-production: This stresses the value of academics and participants working collectively throughout the research process without favouring one type of knowledge over another.
- 5) Social Justice: This stresses the value of research, which facilitates social justice outcomes, that make a real world difference for those involved.

There are those, however, who claim that whilst these principles are admirable, they are actually more in keeping with the tenets of participatory action research (Bergold and Thomas, 2012; Cook, 2012). This qualification is based on the premise that participatory research is not necessarily concerned with ensuring research delivers social change or emancipatory practice (Barnes et al., 2013; Bell et al., 2004; Bergold and Thomas, 2012). Some propose that participatory research can be distinguished from participatory action research in that it transfers the emphasis from research aimed at action and change to research aimed at collaborative research activities (Barnes and Cotterell, 2011; Bergold and Thomas, 2012; Cook, 2012; Green et al., 2003).

However, as Cook (2012) noted, the distinctions within participatory or collaborative research remain somewhat opaque. Many agree that differences have not been sufficiently well articulated between, for example, participatory action research, participatory research, community research and collaborative research (Bell et al., 2004; Bergold and Thomas, 2012; Cook, 2012).

Despite concerns about what constitutes participatory or collaborative research, funding in the UK, especially in health research, now widely considers the involvement of lay people in the research process a key undertaking (Barnes and Cotterell, 2011). Cook (2012) noted that many funding bodies now explicitly require lay participation within the research process. This is increasingly referred to as public and patient involvement (PPI).

#### What constitutes participation: consultation, collaboration, or control?

Recently there has been increasing debate around what constitutes consultation, collaboration, or control within participatory research (McLaughlin, 2006, 2010; Staley, 2009). The level of participant involvement in research remains contentious (Barnes and Cotterell, 2011; Barnes et al., 2013; Bergold and Thomas, 2012; Cook, 2012; Staley, 2009).

Some of the most hotly contested issues centre on:

- The degree to which the value of the participant's involvement is recognised and acknowledged.
- The opportunity for participants to access the process of involvement.
- Whether participants have the opportunity to be involved in the write up and dissemination stages.

(Bergold and Thomas, 2012; Cook, 2012; Trivedi and Wykes, 2002)

The following section attempts to offer a flavour of the literature pertaining to degrees of participation through the work of Staley (2009), The Durham Community Research Team (2011), Bergold and Thomas (2012) and Cook (2012).

Staley (2009) proposed that it is the level of involvement which constitutes whether research can be termed participator. Staley (2009) considered involvement to mean an active partnership between lay participants and professional researchers in the research process. Staley (2009) suggested that active involvement refers to consultation, collaboration or user control and that the level of participation is dependent on the individual requirements of participants and professional researchers. In this context participatory research could include "Public involvement in advising on a research project, assisting in project design, or carrying out the research" (Staley, 2009:16).

The Durham Community Research Team (2011), however, framed understandings of control, collaboration, or consultation through distinctions between community controlled/managed research, co-production and researcher managed research.

Professional researchers who are managed by and work for the community are an example of community control within participatory research.

Professional researchers and community members who are equal partners are an example of collaboration within participatory research.

Professional researchers who consult with participants but control the research process are an example of consultation within participatory.

The degree of partnership within these approaches can be qualified by a number of factors. These include the degree to which lay people are involved in research process and whether there are trained community researchers who undertake research elements such as data generation, data analysis and report writing.

Bergold and Thomas (2012) conversely, suggested that research cannot be considered participatory unless people are involved in the decision-making process as research partners or co-researchers.

Cook (2012), on the other hand, argued that participatory research is demonstrated by the degree of control participants exert in the research process or the extent to which participants are accorded the same rights as the professional researchers in terms of decision-making.

Despite differences between conceptualisations of what constitutes participation in research it is widely recognised that the relationship between professional researchers and co-researchers is fraught with issues of power and control (Banks et al., 2013; Benneworth et al., 2009; Dodson et al., 2007; Staley, 2009, 2011). The following sections help to frame aspects of this problem.

#### Remuneration

The relationship between professional researchers and co-researchers is often undermined by discrepancy between scales of payment (Dodson et al., 2007). Whilst professional researchers are salaried, the collaboration and the cooperation which co-researchers contribute to the research process is insufficiently rewarded (Hobbs and Sixsmith, 2010; Woolryche and Sixsmith, 2008). Research participants who receive remuneration receive payment, which covers out of pocket expenses, or payment, which is given on an adhoc basis (Turner and Beresford, 2005a,b,). There are no scales of remuneration for co-researchers based upon expertise, time commitments, skills, experience or level of involvement and this can cause a sense of inferiority for co-researchers (Cartwright et al., 2013; Hobbs and Sixsmith, 2010; Steel, 2006). Some have attributed this to the fact that remuneration recognises the worth of co-researcher input and is a societal indicator of the value of the individual's

contribution to research (Bergold and Thomas, 2012; Cartwright et al., 2013; Earl-Slater, 2004; McLaughlin, 2009).

Lack of remuneration for co-researchers has become an increasingly important consideration within participatory research (McLaughlin, 2009b). For some it acts as a barrier to full participation (Faulkner, 2010). To date efforts to provided co-researchers with adequate remuneration have been undermined by claims that payment may interfere with lay participants' benefit entitlements (Cartwright et al., 2013; Commission for Social Care Inspection, 2007; Turner and Beresford, 2005a,b).

It is this complex situation, which has led organisations such as, INVOLVE, the National Institute for Health Research, the Mental Health Research Network and the Joseph Rowntree Foundation to try to establish guidelines on payments. However, paying benefit recipients remains difficult to negotiate within the research process (Cartwright et al., 2013; Faulkner, 2015). Remuneration and the implications for research continue to occupy an important place in discussions about the degree to which participatory research is conducted on an equal footing (McCartan et al., 2012).

## **Capacity building**

Participation in participatory research requires the facilitation of capacity building in lay participants in order to allay some of the issues around power and control (Bergold and Thomas, 2012; Cook, 2012; von Unger, 2012). Capacity building includes inculcating specific knowledge and skills which participants may need training to acquire (von Unger, 2012). These include, for example, linguistic competencies, communicative skills and the ability to work systematically in the research process (Bergold and Thomas, 2012). Professional researchers can help to balance the power differential by undertaking training, mentoring and workshops in these specific areas. In addition to which, at a practical level they can impart these skills to lay participants in their day-to-day interactions with them (INVOLVE, 2012; von Unger, 2012). One approach to this undertaking is to design training agendas and choose methodological approaches which build on and develop lay participants

existing knowledge base and skill set (Abebe et al., 2011; Hilbert 2014; INVOLVE, 2012).

Having discussed theoretical considerations of what constitutes meaningful participant involvement, the following sections consider the wider practicalities of utilising participatory research. This will be contrasted with a consideration of the practicalities of utilising traditional research.

## Participatory research: opportunities and challenges

Many have argued that the participatory research process enables researchers to distance themselves from the traditional research viewpoint (Bergold, 2007; Bergold and Thomas, 2010, 2012; Branfield and Beresford, 2006; Cook, 2012; Kawulich, 2005; McLaughlin, 2009b). This enables researchers to reflect on the traditional and familiar research customs, which they have become habituated to. Participatory research, therefore, encourages professional researchers to question and rethink established interpretations of situations and strategies (Brandstetter et al., 2014; Caldon, 2010).

Abebe et al. (2011) proposed that participatory research offers professional researchers an opportunity to appreciate the lived experience of communities. Lay participants bring unique skills and experience to the research process. This enriches the research process and contributes to knowledge development (McLaughlin, 2010). Participatory research also facilitates partnership working, the exchange of innovative research ideas and a deeper understanding of the cultural environment within which the research is situated (Abebe et al., 2011; Brett et al., 2010; Faulkner, 2010; McLaughlin, 2010).

The utilisation of participatory research can provide professional researchers with opportunities to work with new people and organisations in innovative and challenging ways (Cook, 2012; Flick, 2009). This provides professional researchers with an opportunity to broaden traditional academic networks and share knowledge and learning across a variety of sectors (Cook, 2012; Flick, 2009; Green et al., 2003). Participatory research tends to encourage collaborative approaches. This offers new partnership working and promotes research endeavours, which

build on existing relationships (Abebe et al., 2011; Durham Community Research Team, 2011; Kawulich, 2005).

Collaborative, participatory and multi-sectoral research studies are increasingly well regarded by funding organisations (Cartwright et al., 2013; Earl-Slater, 2004; Faulkner, 2015). Research approaches which utilise participatory and collaborative approaches may therefore stand a better chance of successful funding bids (Barnes and Cotterell, 2011; Bergold and Thomas, 2012; Durham Community Research Team, 2011; Beacon North East Community-University Research Partnerships 2011).

Participatory research can bring many opportunities for the lay participants; these may include learning new skills, expertise and competencies, which can enhance amongst other things self-worth, self-esteem and confidence (Abebe et al., 2011; Bergold, 2012; Bergold and Thomas, 2012).

Lay participants, meanwhile, have identified that involvement in participatory research is fulfilling as they are contributing to knowledge development and service improvements (Hobbs and Sixsmith, 2010; INVOLVE, 2012; Woolryche and Sixsmith, 2008). In addition to which, involvement within a project may encourage greater interaction between lay participants from diverse socio-cultural backgrounds, enhancing social bridging and bonding (Putnam, 2001; Sixsmith and Boneham, 2007). Likewise, meeting and becoming involved with new people and feeling part of something greater than oneself has been identified as beneficial for both lay participants, and the community within which the research is situated (Heron and Reason, 2006; Hotze, 2011; Manzo and Brightbill, 2007; Trivedi and Wykes, 2002).

# Challenges

As qualitative research continues to flourish, interpretivist and constructivist methods are increasingly regarded as acceptable within research communities (Barnes and Cotterell, 2011). For example, qualitative approaches are now widely accepted in certain disciplines, such as sociology, health, psychology and ethnology (Bergold and Thomas, 2012). This has resulted in the growing utilisation of participatory research approaches across a number of disciplines (Barnes and

Cotterell, 2011; McLaughlin, 2006). However, within the academic disciplines, particularly those which have a strong focus on quantitative and positivist traditions, researchers who engage in participatory research risk becoming perceived as unconventional and/or as outsiders in the academic community (Benneworth, 2009; Bergold and Thomas, 2012; Hotze, 2011).

This problem is faced by qualitative research in general, but this is often made more difficult by those using participatory approaches within qualitative research (Bergold and Thomas, 2012; Mruck et al. 2002). For many within academia the closeness between professional researcher and lay participants and an onus on reflexivity in place of objectivity means that participatory research remains a less acceptable form of research than more 'conventional' qualitative research (Bergold, 2012; Bergold and Thomas, 2012).

Bergold (2012) highlighted further issues for those who adopt participatory approaches. Issues such as developing an academic career can be constrained for those who utilise participatory approaches. Bergold (2012) argued that few scholarly journals accept participatory research papers, partly because the marginalised groups who are frequently the focus in participatory research tend not to be the focus of interest of conventional science (Bergold, 2012). "This has an effect on the frequency with which the publication in question is cited. And because the Science Citation Index serves as an important indicator of scientific qualification, authors who apply participatory methods are disadvantaged" (p.7).

In addition to which, publications remain a widely used barometer of academic success. Participatory research is more time-consuming than traditional research. This means the production of publishable findings takes longer than more traditional. As a result the participatory researcher's list of publications is shorter (Bergold, 2012; Bergold and Thomas, 2012).

The diversity of requirements and the tasks which professional researchers undertake within participatory approaches necessitates different competencies and skill sets to those traditionally required within academic research (Bergold and

Thomas, 2012; Borg et al., 2012; Flick, 2009). For example being required to act as mediator, mentor, trainer, facilitator requires a high degree of flexibility and reflexivity which must be embedded within ones' practice (Bergold and Thomas, 2012; Borg et al., 2012). This can often mean professional researchers dedicate and devote extensive time to acquiring these skills and embedding them within the research process.

Other challenging issues identified include:

- The additional time required to train, mentor and support lay researcher's involvement in the research process.
- Establishing processes to facilitate periodical reflection on the progress of the research.
- Establishing and maintaining communication with and between research participants to ensure momentum and motivation.
- Establishing and maintaining group relations and connections between what may be disparate groups in the community.
- Establishing and maintaining group dynamics and partnership working. The dynamic nature of working alongside diverse groups is often complex and hard to manage.
- Encouraging diversity amongst group dynamic whilst recognising multiple identities.
- Managing and meeting the varied expectations of participants can be time consuming and challenging.
- Establishing outcomes, which take into account the needs and requirements of participants as well as the research.
- Establishing and maintaining clear roles and parameters for research participants.
- Establishing regular progress meetings to ensure research participants are kept informed and involved through out the research process.

These are all aspects of the participatory research process which are rarely built in to the timescale of a study and therefore puts additional time pressure on the professional researcher (Bergold, 2012; Borg et al., 2012; McLaughlin, 2010). Additional resources required to facilitate the involvement of lay participants' increases the pressure on the professional researcher. This requires a strong commitment from the researcher to set aside significant time and energy to facilitate the extended consultation process (Bergold, 2012; Bergold and Thomas, 2012; Borg et al., 2012; Cook, 2012; Green et al., 2003; Heron, 1996; McLaughlin, 2010).

Participatory research can also present significant challenges for lay participants (Barnes and Cotterell, 2011; Benneworth, 2009; Hilbert, 2014; INVOLVE, 2012). Many lay people who become involved in research place themselves in an unfamiliar and intimidating environment (Barnes and Cotterell, 2011; Benneworth, 2009; Bergold and Thomas, 2012; Durham Community Research Team, 2011). In addition to which, lay participants frequently take part in activities, which were previously unknown to them. Furthermore, as previously documented, involvement in participatory research frequently requires participants to engage in training, mentoring and learning workshops (Durham Community Research Team, 2011; Earl-Slater, 2004; Hobbs and Sixsmith, 2010). These often new and intimidating environments have frequently been cited as reasons why lay people disengage with research after the initial recruitment phase (Barnes and Cotterell, 2011; Hobbs and Sixsmith, 2010; Woolryche and Sixsmith, 2009). The need to communicate clearly and concisely, and integrate oneself in unfamiliar settings, has also been identified as particularly challenging for lay participants (Barnes and Cotterell, 2011; Benneworth, 2009; Hobbs and Sixsmith, 2010; INVOLVE, 2012).

#### The advantages of using traditional research

Professional researchers have the necessary methodological expertise to undertake diverse research studies. In addition to which, they are knowledgeable of research procedures and familiar with the importance of drawing upon theoretical frameworks to underpin research (Abebe et al., 2011; Brandstetter et al. 2014; Hayes et al., 2012; Hilbert, 2014). Traditional research relies on trained and experienced researchers who bring established research skills and experience. Traditional research, therefore has the advantage of being able to commence the

research process immediately, without the necessity of organising and providing training, support and mentoring for lay researchers.

In traditional academic research, detachment and objectivity are highly valued. Indeed, many consider it imperative to maintain objectivity within the research process (Bergold, 2012; Brandstetter et al. 2014). Research perceived as being subjective and lacking in rigor is frequently challenged for being neither representative nor universally applicable (Bergold and Thomas, 2012; Brandstetter et al., 2014). These perceptions often undermine efforts to achieve academic recognition through publications (Bergold, 2012; Bergold and Thomas, 2012; Staley, 2009). Traditional research however, does not tend to suffer from these challenges. In general terms traditional research findings are more likely to be accepted as having the rigor and validity required for evidence based policy and for academic or peer reviewed publications.

## The disadvantages of using traditional research

There are a number of disadvantages associated with using traditional research. For example within traditional research, gaining access to study participants can be challenging, particularly so with those from deprived or hard to reach groups. There is evidence to suggest that lay involvement has been effective in improving recruitment to research, especially for hard to reach groups and or where the demands of a research project are high (INVOLVE, 2006, 2012). This may be because peer networks encourage and support recruitment, negating perceptions that one is recruited as part of 'an agenda' (Entwistle, 1998; Earl-Slater, 2003; Hilbert, 2014). Improvements to recruitment may not always be facilitated by lay involvement however. McLaughlin (2006), for example, reported that young coresearchers had the potential to reduce recruitment if they were unpopular with their peers.

There is perhaps greater consensus that traditional research may fail to gain the diverse range of perspectives, which lay participants add to the research process. These perspectives add an extra dimension and enhance understandings of the topic being researched (Bergold, 2012; Brandstetter et al., 2014; Staley, 2009).

There is also evidence to suggest lay involvement adds an extra dimension to data analysis, by contributing alternative perspectives on emerging themes and patterns in the data (Bergold and Thomas, 2012; Durham Community Research Team, 2011). Without these additional perspectives, researchers may maintain existing traditional and habitual research customs. These may undermine their commitment to questioning and rethinking established interpretations of situations and strategies (Bergold and Thomas, 2010, 2012; Staddon, 2015; Staley, 2012, 2013).

Concerns raised by INVOLVE (2012), an organisation which promotes public involvement in research, suggests that traditional research may fail to identify research which is relevant to the public interests and lay concerns. Similarly, traditional research may be guilty of ignoring a wider array of topics, thereby failing to address lay concerns and perpetuating the study of narrow, academic areas of interest (INVOLVE, 2012). In addition to which, traditional research has systematic procedures which may stymie creative and reflexive processes (Bergold and Thomas, 2012).

Finally, funding bodies increasingly require the involvement of lay people in the research process (Becker et al., 2010; Bergold and Thomas, 2012; Staley, 2013). There is mounting evidence that traditional research which eschews the involvement of lay people will struggle to fulfil the research criteria expected by funding bodies (Becker et al. 2010; Cook, 2012; INVOLVE, 2012; Staley, 2013).

### The advantages of using participatory research

Staley (2009) reported that participatory research approaches have a number of positive influences on the research agenda. These include the identification of research topics or areas of concern which professional researchers were unfamiliar (Bergold and Thomas, 2012; Durham Community Research Team, 2011; INVOLVE, 2012).

As outlined above there is growing evidence to suggest that funding bodies looked favourably upon participatory research, particularly within healthcare. Some researchers have reported that public involvement added credibility to their

proposal, improved its feasibility and overall design and was instrumental in securing funding (Bergold and Thomas, 2012; Staley, 2009).

Other important advantages of using participatory research include influencing the research process. There is some evidence that lay participation often modifies the original academic focus of research, adding a dimension which accords more with public interests and concerns and is thus more relevant in real world contexts (Bergold and Thomas, 2012; INVOLVE, 2012; Staley, 2009).

INVOLVE has also put forward evidence which suggests that lay participation enhances the research design (INVOLVE, 2010, 2012, 2014). Improvements range from ensuring the language and wording used is appropriate and comprehensible, that questions are suitably phrased, and that research tools are suitable (Bergold and Thomas, 2012; Hotze, 2011; INVOLVE, 2012; Smith et al., 2008).

Evidence suggests that participatory research has a strong and positive impact on research ethics. For example lay involvement in patient information sheets has been reported as helping to ensure the information is clear and comprehensible (Brett et al., 2010; Smith et al., 2008; Staley, 2009). Research also suggests lay involvement during the early stages of a study helps to identify potential ethical concerns, and offers appropriate solutions to these ethical issues (Banks et al., 2013; Durham Community Research Team, 2011; Tarpey, 2011).

Staley (2009) identified that participatory research may have a positive impact on data collection as involving peer interviewers may facilitate the collection of more candid information. This may be associated with widely documented phenomenon of interviewees telling researchers what they think they want to hear (Bergold and Thomas, 2012).

A particularly important advantage of using participatory research is the impact it may have on data analysis. There is evidence to suggest that researchers who discuss initial readings of data with lay participants enhance the validity of their conclusions (Bergold and Thomas, 2012; Hilbert, 2014).

Public involvement enhances the clarity and depth of qualitative data by:

- Correcting researchers' misinterpretations (Bergold, 2012; Brandstetter et al., 2014).
- Identifying themes that researchers would have otherwise missed (Brandstetter et al., 2014).
- Highlighting findings most relevant to patients or the public (Ross et al., 2010)
- Challenging perceptions of researchers and changing the way in which results have been described in reports (Faulkner, 2010).

To date, little research has explored the impact of lay involvement on writing up findings. This may reflect the fact that lay participants are rarely involved at this stage of the research process (Smith et al., 2008; Staley, 2009). There is however, some evidence to suggest that lay involvement in the dissemination of research increases the likelihood of findings being implemented (Bergold and Thomas, 2012; INVOLVE, 2012).

## The disadvantages of using participatory research approaches

The objectivity versus subjectivity argument inherent in academic research is widely used as a barrier to the utilisation of participatory research (Barnes and Cotterell, 2011; Staley, 2009; McLaughlin, 2010). The dissolution of the subject-object relationship between the researcher and the researched is considered to be a real and valid concern for the academic recognition of participatory research (Barnes and Cotterell, 2011; Bergold and Thomas, 2012; McLaughlin, 2010). For many, the close working relationship, which exists between the research partners, prevents professional researchers from achieving the scientific distance, which is highly valued in academia. This leads to claims that in participatory research it is not possible to separate the researchers' contribution to the collected data from that of the researched (Hotze, 2011; Staley, 2009). Thus objectivity is compromised which in turn affects both the internal and external validity of the findings.

When "classical" quality criteria are applied, participatory research is frequently perceived to be unacceptable because it lacks objectivity, reliability and validity

(Abebe et al., 2011; Bergold and Thomas, 2012; McLaughlin, 2010; Staley, 2009). Associated with the need for professional researchers to remain objective is the basis of the relationship between the professional researcher and lay participants. Inherent power differentials and the balance of power have frequently been reported as having a detrimental impact on the research process (Brandstetter et al., 2014; Hayes et al., 2012; Hilbert, 2014).

A considerable disadvantage of using participatory research is the requirement to factor in time to address the additional requirements of working alongside lay participants (Hobbs and Sixsmith, 2010; Woolryche and Sixsmith, 2008). This may include recognising that the research hypothesis and questions may emerge only gradually during the process of engagement with the research participants. Furthermore, the research process can often be delayed by the need to facilitate lay participation in the research process. This makes it challenging to ensure that the research process runs according to the study timeframe (Hobbs and Sixsmith, 2010; Woolryche and Sixsmith, 2008).

Conflict therefore, may arise between the professional researchers need to implement research within a specific time frame and the lay participants need for additional time to become familiar with the research process (Benneworth,2009; Bergold and Thomas, 2012; Hotze, 2011; Staley, 2009). Lay participants involved in participatory research need to develop specialised skill-sets required for participation in the research process and more general competencies (Abebe et al., 2011; Bergold, 2012; Bergold and Thomas, 2012). Ensuring these competencies are achieved can be time consuming and costly (Durham Community Research Team, 2011; Hobbs and Sixsmith, 2010; Woolryche and Sixsmith, 2008).

Additional time which has usually not been written into the study timeframe must be found to ensure that partnership working alongside lay participants addresses issues such as recruitment, support, training, mentoring and facilitating lay participants access to resources (Blackburn et al., 2010; Faulkner, 2010; INVOLVE, 2013).

Staley (2009) suggests that there is little evidence of negative impacts of public involvement on recruitment. However, Viswanathan et al. (2004) noted a "Bias in recruitment caused by community members engaging highly motivated people who are not representative of the broader population" (p.9).

#### Conclusion

Given the numerous conditions attached to the utilisation of participatory research approaches compared with traditional research approaches it is challenging to argue that one approach is preferable to the other. In the realities of real world research, it seems pragmatic for the decision to be made according to the nature and circumstances surrounding individual research studies.

## **Appendix E: Research Participant Information Sheet**

#### RESEARCH PARTICIPANT INFORMATION SHEET

I would like to invite people with complex health needs to take part in the research project: 'Exploring the personal wellbeing of older people with long-term conditions'

You have been asked to participate in this research as part of the effort to ensure that the study is representative of the population of Manchester.

This research is in part fulfilment for a PhD undertaken by Jo Hobbs and is funded by Manchester Metropolitan University (MMU). The research will be conducted by Jo Hobbs a research student at MMU and supervised by her director of studies Carolyn Kagan of MMU.

The research will undertake a participatory research approach to articulate the voices of participants and co-researchers in the research process in order to obtain their views of what is important for personal wellbeing.

The study will recruit participants who will undertake a short one to one semistructured interview which will last between 30 – 60 minutes maximum.

Interviews will be conducted at a time and venue convenient for service users and family members/carers.

In addition to which 2 co-researchers will be recruited to take a more active role in the research process. They will receive training to aid their ability to take part in the research process. It is envisaged that the initial training requirements may take approximately four half days to deliver. Further training and developmental work will be delivered throughout the course of the project. Any travel and food costs will be reimbursed in full.

The main purpose of the study is to explore:

- 1-How personal wellbeing is perceived by older people with LTCs?
- 2- Whether wellbeing is a meaningful concept for older people with LTCs?

- 3 How services can better reflect the wellbeing needs of older people with LTCs
- 4 If a definition of wellbeing would help its usefulness as a measure of service user satisfaction?

Data that identifies the participants will be kept confidential and safely secured under lock and key in MMU to the extent permitted by the applicable laws and/or regulations and will not be made publicly available.

If the results of the research are published, the subject's identity will remain anonymous having first been labelled with an identifier code known only to the researcher and research supervisors.

Any information revealed during the course of the research which may cause harm to the participant or a member of the public may need to be disclosed to the relevant authorities.

There are no foreseeable risks involved in taking part in the study but should you become distressed as a result of research process you will be offered advice about counselling etc.

The study may not prove of any immediate benefit to you but the information we receive from this study and the inclusion of participants will help understand:

- 1- Perceptions of personal wellbeing in older people with LTCs.
- 2-The impact of services on the personal wellbeing of older people with LTCs.
- 3-Ensure that the voices of people with LTCs are included in this study

If you have any concerns regarding the study contact Jo Hobbs on 0161 247 2979.

If you have any concerns or issues regarding the study that you feel unable to discuss with Jo Hobbs please contact Carolyn Kagan on 0161 247 2545

As a participant in this research project you will be given a copy of this research participant information sheet and a signed consent form to keep.

# Appendix F: Consent Form Version 6 - 23<sup>rd</sup> July 2007

As a participant in this research project you will be given a copy of the research participant information sheet and a signed consent form to keep.

Title of Project: 'Exploring personal wellbeing with older people with long-term conditions.

Researcher: Jo Hobbs

# If possible please read carefully and write, sign and date your name at the bottom of the page under part 4

Jo has explained the research project to me and what will be asked of me. I have received a copy of the study participant information sheets, which have been explained to me.

- 1. I confirm that I have read and understand the information sheet dated 23/07/07 (Version 6) for the above study. I have been able to read and think about the information sheet, ask questions and have had these answered to my liking
- 2. I understand that my input is my choice and that I am free to withdraw at any time without giving any reason. This will not change my medical care, legal rights or any other benefits I am allowed.
- 3. I understand that some of my interview data collected during the study may be looked at by the researcher and her supervisors, that all interviews will be recorded and that if any of my comments are used as direct quotes they will be used in such a way that my identity will be protected. I give permission for these people to have access to my data.

	·	
Name of Participant	Date	
Signature of Participant	Date	
Witnessed by	 Date	

4. I agree to take part in the above study.

As a participant in this research project you will be given a copy of the research participant information sheet and a signed consent form to keep.

## Appendix G: Interview schedule

Q - Are you familiar with the term wellbeing?

Q - Is wellbeing a term you would use in general conversation/in your social circle?

Explore relationship between what is considered important for personal wellbeing

Explore relationship between why these are important for personal wellbeing

Explore relationship between those things which are considered to reduce participants' personal wellbeing.

Q - Is health important to your sense of wellbeing?

Explore relationship between participants' health and personal wellbeing.

Q - Has your health become more important to your sense of wellbeing since you've been diagnosed with complex health needs?

Q - Has your health condition had a direct influenced on your personal wellbeing? for example lost your job because of it?

Q - Has your health condition had an indirect influenced on your personal wellbeing? for example caused tension between you and your family?

Q - The local government act 2000 made local government responsible for improving the wellbeing of its citizens:

Explore examples of how participants' perceives local government has sought to improve the wellbeing of older people with complex health needs.

Explore examples of how participants' perceives local government could improve the wellbeing of older people with complex health needs?

Q - Have any of the services you use had a positive impact on your personal wellbeing?

Q - Have any of the services you use had a negative impact on your personal wellbeing?

Explore relationship between service provision/ access and improvements in personal wellbeing.

Explore how participants' perceives changes to service provision/ access might impact on older peoples' personal wellbeing.

- Q What sort of things could local government /services do that would have a positive impact on your personal wellbeing?
- Q What sort of things could local government / services do that would have an impact on those things that have a negative impact on your personal wellbeing?
- Q What sort of things could local government / services do for your personal wellbeing as you get older/ as your LTCs worsens?

Explore relationship between ageing/deteriorating health and personal wellbeing.

Explore relationship between participants' personal wellbeing and their life-course.

- Q has what is important to your wellbeing changed through your life-course?
- Q Do you think that what is important for your wellbeing as an older person is different to what it was as a younger person/child?
- Q Has the ageing process changed your perception of wellbeing/what wellbeing means to you?
- Q Is your sense of personal identity/ religion/spirituality/culture important for your wellbeing?

Explore relationships between participants' sense of identity and wellbeing.

Explore relationships between socio-economic position/class and wellbeing

Explore relationships between culture and wellbeing.

What is it about these markers of personal identity which make them important for wellbeing?

Q - Is social interaction important for your wellbeing?

Explore different aspects of social interaction and wellbeing.

- Q Does a lack of social interaction impact on your wellbeing?
- Q How does social interaction impact on your wellbeing?
- Q Would a workable definition of wellbeing help or hinder its usability as a concept for improving services for older people with LTCs?
- Q Why would a workable definition of wellbeing help or hinder its usability as a concept for improving services for older people with LTCs?

## **Appendix H: Co-researcher Training**

Three training sessions were delivered in January 2008 to the community researchers who were participating in the initial research project 'Changing services for people with long term neurological conditions: Promoting well-being within service provision' (2007-2009). Two of this original group of six went on to become the co-researchers who participated in the PhD study. The first two training session were delivered by an external training provider recruited after consultation with the co-researchers to ensure that he was considered an appropriate choice. In consultation with the community researchers I chose a trainer who had himself been a service user of mental health services and thus had an appreciation of the problems that users of services from other areas may have experienced. Consultation with the co-researchers identified a perception that this training provider could talk with knowledge and awareness of some of the potential issues within participant research involvement such as power differential between paid researcher and unpaid researchers, and feeling able and confident to challenge health and social care professionals. Funding for the training was provided by Manchester City Council, Manchester Primary Care Trust and Manchester Metropolitan University.

The intention of the first two research training sessions was to:

- Provide community researchers with knowledge about the Long Term
   Neurological Condition study
- Teach basic premises about qualitative research
- Generate a cooperative learning forum that encouraged community
   researchers to enhance their confidence and self-esteem
- Develop critical thinking skills
- Equip community researchers with necessary skills to engage in reflective practices
- Highlight the role of community researchers in the context of research

 Highlight the potentially contrasting positioning of the community researchers in terms of understanding, familiarity with research, culture, sexuality, class, gender and disability.

The sessions with the external training provider worked well according to the community researchers who felt that the trainer understood their context, where they were coming and talked in a language that they could identify with.

A third training session was delivered to the community researchers with the aim of introducing participants to different data analysis approaches. This involved a data workshop session delivered by the project researcher and her line manager. The methods of analysis chosen for the training were chosen to reflect the approaches which might be utilised in the study and which were congruent with the epistemological position of the research. The four types of analysis chosen were discourse analysis, content analysis, thematic analysis and grounded theory. An additional aim of the training session was to strengthen the relationships between the community researchers and the researcher thus facilitating their ability to be work alongside the researcher at all stages of the research process.

## **Training evaluation**

INVOLVE, an organisation which supports the engagement of lay participants in research, highlighted the need to rigorously evaluate training given to community researchers (INVOLVE, 2007). INVOLVE (2007) encourages research organisations to support participant engagement through networking and knowledge generation, and identified the importance of providing participants a comprehensive exploration of their training requirements in order to identify future training needs. To facilitate this process the community researchers completed an evaluation sheet and reflected upon and discussed the sessions with the researcher. The training evaluation was undertaken through feedback forms which contained some of the co-researchers reflections on the training sessions. The community researchers reported that the training sessions were engaging and stimulating and a positive learning process. Importantly, the participants felt that the training was non-

judgemental and represented a sharing of experiences as opposed to a top-down learning process

## Training evaluation feedback

Excerpt from Alison's Training Evaluation feedback form:

"I enjoyed the training session with X, it was nice to meet Jim and hear his perspective as a disabled person. Also because I know very little about Parkinson's and hope to learn more about how he copes.

"I enjoyed X's manner which didn't make you feel you had to choose you're (sic) words carefully or be afraid of making a mistake. The training was good too for reminding me about making assumptions and prejudices".

Alison identified that the training sessions amongst other things enabled her to challenge their own understandings and learning, to appreciate the value of different viewpoints and perspectives, to challenge with confidence the views of professionals and to feel she had some of the background in knowledge she would need for her involvement in research.

Excerpt from Jim's Training Evaluation feedback form:

"Since taking part in the training sessions I've questioned my own little prejudices which previously I wasn't aware of. It's made me think more about how much empathy you probably need when you're researching too so its been good for a range of things that relate to us as community researchers and us as people".

Jim identified that the training facilitated his ability to open up to new learning experiences and processes of reflection.

Overall the participants reported that being engaged in the research enabled them to begin the process of self-development through stimulation, participation and engagement. Furthermore, the training appeared to enable the co-researchers to experience richer and deeper insights in their reflective diaries.

## Appendix I: Researcher's Reflexive Journal

#### Dec 2007 - To hyphenate or not

Hughes (1988:171) stated that compounding is a fundamental method of word formation in English, and is a general feature of the Germanic languages". He goes on to add that "Compounds ....tend to produce ambiguity and opaqueness, particularly when they are made up of two general terms...furthermore the possibility of ambiguity increases as the language gains in grammatical flexibility" (Hughes, 1988:219).

I circumvent the hyphen in my spelling of wellbeing and that alone signifies a different conceptualisation of the term wellbeing. To hyphenate wellbeing would I believe separate what I have endeavoured to pursue in this thesis which is the case for addressing wellbeing as an holistic concept. Well-being, visually and lexographically is a composite word with an adjective and noun conjoined or in linguistic terminology a hyphenated compound. Wellbeing is quite different, which whilst retaining its composite nature has become more than that and has taken the linguistic form of a closed compound. I believe this depicts and resonates more completely with a sense of inner holism than QoL does. The Shorter Oxford English Dictionary, eliminated over 16,000 hyphens from the sixth edition arguing that "People are not confident about using hyphens anymore. They're not really sure what they're for" (Stevenson, 2010). Whilst 'well-being' in the OED retained its hyphen. As the Hughes noted hyphens denote how language changes, and can indicate when a word has not yet been completely assimilated into familiar or standardised usage and can be seen as being at a halfway point before it has been 'normalised' (Ereaut and Whiting, 2007). Email was originally written as e-mail but is now conventionally written as email.

Academic literate thought is that wellbeing should be hyphenated but my reflective accounts document that after a significant period of reflection about how to spell wellbeing I eventually concurred with (Ereaut and Whiting, 2007) that the variety of ways to spell the term reflects the ambiguity with which it is viewed and the shifting

nature in which it is often employed. My constant usage of the term however has led me to adopt what Ereaut and Whiting, (2007) term the 'normalising' process that is making the hyphen between 'well' and 'being' redundant.

#### March 2008

Emotional involvement: Speaking with x today who has deteriorated massively since I last saw him was quite upsetting and reminded me of Boud and Walker's assertion that reflecting isn't just about the cognitive rational process but must also involve emotional processes. X sort of remembered me but couldn't recall what we had talked about or when I had last visited. I told him it had been his second week here and he said I've been here about six months (so he remembered that much). He asked me what I wanted to talk about and I explained that this time I wanted to discuss his wellbeing to which he said "I haven't got a life". X despite the fact that his speech had deteriorated substantially was still trying to be upbeat! I felt sad seeing him like that which he must have sensed because he said that he was lucky as some people can't even smile and gave a big grin (which looked more like a grimace, clenching of teeth). He was shaking constantly and at times I struggled because I felt upset. I do wonder how my personal feelings for the participants may be impacting on this study but I suppose it's a question of remaining aware and exploring these feelings throughout the course of this research.

## June 2008 Reflections on Alison and Jim

If I'm going to proceed with my idea of using case studies as my method approach to my PhD then I need to consider those elements which are salient in my choice of participants and my structuring and language when seeking to understand and engage the participants. It strikes me that in identifying Jim and Angela as participants they fulfil a number of different objectives within my research.

Jim has Parkinson's disease, has been an active member of society who has sought to engage and take part in societal discussions and has sought to generate change in society through those organisations which have societal approval and are recognised, legitimated and acknowledged power-holders.

Alison is bi-polar and has been active in seeking change from within recognised and society approved organisations for change such as tenants committees, however she also seeks to 'voice' opposition to societal constraints through mechanisms which are outside the recognised and legitimated power structures. She has sought to impact upon society through methods not always traditionally recognised as legitimate.

How does their language, socialisation, class, gender, educational attainment, position within society etc affect both their vision of what wellbeing means and how they internalises and externalises their inner and outer presentation of what wellbeing means to them. Is this a process which is significantly affected by their health, their upbringing, their local environment, their family structure?

The positionality of the co-researchers' is different both in respect to my PhD and the role they wish to play in my thesis. Jim undertook a role as research advisor on the Long Term Neurological Condition research project and attended steering group meetings. How might this exposure to the process issues and concerns of the project have affected how he thinks and talks about his own experiences? How has this contact and knowledge of the research process affected his relationship to this research? How has it affected his perception of me and my role in this thesis? Jim has elected to take part in the research as a co-researcher but does not want to be part of the data analysis process, to what degree might his former involvement on the Long Term Neurological Condition project have influenced this decision?

Alison undertook a role as co-researcher on the Long Term Neurological Condition research project and conducted solo face to face interviews with participants for this project. How might this exposure to the issues and concerns of the participants' affected how she thinks and talks about her own experiences? What impact did the research training have on her commitment to voicing and challenging community problems? Did her experiences of the training alter her perception of the role of research in society? How has it affected her perception of me and my role in this

thesis? Alison has elected to take part in the research as a co-researcher and wants to be part of the data analysis process, to what degree might her former involvement on the Long Term Neurological Condition project have influenced this decision?

#### **July 2008**

I have just read Ryan and Deci A Review of Research on Hedonic and Eudaimonic Well-being 'this review has been instrumental in facilitating a period of reflection in which the end result was that I need to acknowledging that the significance I place on the environment may influence how I weight conversations around the built and natural environment and its relationship with wellbeing. This relates to my investment in the research and what I want to achieve. I had forgotten that my masters dissertation was concerned with exploring the role of environmental education on the environmental behaviour of secondary school children. Although I am mindful of my 'attachment' to the natural environment when interviewing the co-researchers my relationship with them means that they are probably aware that the natural environment is a particular (emotional) interest of mine. Therefore I will have to try to guard against unintentionally 'leading' the co-researchers to 'present' the environment as being of intrinsic importance to them in order to give responses which they think I want to hear given my interest in this field. This reflective practice is proving very helpful as I am now very mindful of my 'attachment' to the natural environment when interviewing the participants and do not initiate a direct question regarding the role of the environment but did ask them to elucidate if the subject arises organically.

#### August 2008

Newton's review has been instrumental in facilitating a period of reflection in which the end result was that I need to acknowledging that the significance I place on the environment is likely to have influenced how I have weighted conversations around the built and natural environment and its relationship with wellbeing. This relates to my investment in the research and what I want to achieve. I had forgotten that my masters dissertation was concerned with exploring the role of environmental

education on the environmental behaviour of secondary school children. Although I was mindful of my 'attachment' to the natural environment when interviewing the co-researchers my relationship with them meant that they were probably aware that the natural environment was a particular interest of mine. This may have meant I unintentionally 'led' the co-researchers to 'present' the environment as being of intrinsic importance to them in order to give responses which they thought I wanted to hear given my interest in this field.

#### September 2008

Hedonism and its impact on wellbeing is notable and when thinking about the 'politics of happiness' (Edwards and Imrie, 2008; Shah, 2006; Layard, 2005) I have thought about how the modern wellbeing agenda is still influenced by those perceptions that certain socially acceptable behaviour is more beneficial to wellbeing than others and it seems to me that they have a class or socio-economic basis. *Republic* (Plato 1992 [C4 BCE], 582d-583a) talked about the differences between 'higher' and 'lower' pleasures (1998 [1863], ch. 2). There is still an inherent assumption I think in academic wellbeing that some pleasures, by their very nature, are more valuable than others. For example, the pleasure gained from reading a book is more valuable than the pleasure gained from smoking a cigarette.

J.S. Mill, tried to counter the claims that this was parochial by suggesting that it is rationale to assume that for those who have experienced both types, and are 'competent judges', will make their choices on this basis. Competency is a line of argument pursued by Edwards and Imrie against the wellbeing agenda as a barrier to the emerging positive disability agenda.

#### November 2008

In the spirit of reflexivity I reflected at length upon the terminology I would use in my thesis, positioned as I am within an epistemology, which is based upon social constructionism and takes into account the doctrine of historicism and a double hermeneutics approach, and is underpinned by a commitment to the tenets of Participatory Research. I have speculated on how the term 'researcher' promotes a

representation of myself to myself and to others. Within the term is located a position of authority and power particularly in relation to 'participant' which perpetuates an in-balance of power between them and me. This places both myself and the 'participant' in pre determined roles in which I hold the balance of power. Within those research projects, which seek to redress this power imbalance the term collaborator is often used in place of subject, informant or participant in recognition of the centrality and importance, which those involved in the research hold. The word collaborator however also has disagreeable connotations. I therefore reflected at length on how I would choose to portray those people who took part in this thesis. After much deliberation I have decided to discard the term 'collaborators' which at the time I thought would acknowledge their part in the process, despite the fact that it has some unsavoury connotations e.g. collusion with invading enemy forces. However as I have developed my reflective practice I have become increasingly unhappy with this term and have sought to find a term which is less value laden and so after much deliberation I have decided to use the term participants and co-researchers in recognition of their collaboration in this study.

The term of reference for myself is likewise a linguistic flag of where and how I situate myself both in reference to the participants and to the data. I have traditionally been schooled to think of myself as a researcher within which a frame of reference I am overtly positioned as predominant within the process of research. This is to some degree inevitable and perhaps justified in the context of a PhD which will ultimately benefit me more than the co-researchers and participants. However it fails to adequately reflect the epistemological position in which I have chosen to locate my work. I considered 'analyst' but that resonates with empirical, authoritarian overtones and tends to be used in disciplines, which are more conversant with quantitative methodologies. 'Enquirer' is a term, which within the context of my epistemological position resonates with seeking understanding, knowledge and awareness and I will henceforth refer to and think of myself as an enquirer.

## January 2009

Reading the Des Gasper (2004b) paper has really illustrated to me that the linguistic origins of wellbeing have been largely overlooked by wellbeing research. I have always been interested in linguistics and whilst studying for my undergrad and post grad degree's I would often find myself in the linguistic section of the library. I suppose this is not surprising as I had originally intended to study English at university but switched to history at the last moment. After reading Des Gasper (2004b) I became convinced that part of my contribution to new knowledge could come from an exploration of the linguistic development of wellbeing research and the possible implications this has had on how we perceive and conceptualise wellbeing today. Since then I have begun to think about how I might achieve this as there. Reading authors such as Hughes, (1988) 'Words in time'; Lobner, (2002), 'Understanding Semantics' and Owen Barfield (1926), 'History in English Words' has given me a greater appreciation of the need for wellbeing research to considers the lexicography, semantics, etymology and historical legacy of the term wellbeing in order to facilitate an awareness of wellbeing which is located within historical and linguistic frameworks. Furthermore reading these authors has stimulated my latent interest in the association between word use and development and the primacy which we attach to certain words in certain languages.

#### March 2009

I am now aware of linguistic concepts such as amelioration where a word takes on favourable connotations and deterioration where a word takes on derogatory connotations (Hughes, 1988). This has been key in my growing awareness that the way words are used and the meanings associated with them change in accordance with changing societal values. An obvious example which I can identify with and which is happening now is the shift of meaning of 'gay'. As a child I was proud to be the 'child born on the Sabbath day- bonny and blithe and good and gay because gay still meant happy, light-hearted fun to be around. However as I was growing up 'gay' started to mean homosexual and only older people still used its former meaning. Now I struggle with my nieces and nephews as they use 'gay' to mean something bad or derogatory and I associate that with efforts to malign

homosexuality. It is mad to think that within 2 generations the meaning of 'gay' has shifted significantly twice. 'Gay' appears to be on a steep deterioration.

Terms can become intellectually vogue especially those which have roots in latin or Greek. Whilst etyologically wellbeing has its roots in Germanic language conceptually it is more closely rooted in Greek and thus may be seen as an example of agglutination (where large number of English words are needed to translate small number of Greek letters).

Abstractions (such as wellbeing) ...provide the essential ambiguity which modern politics requires in the form of the room to manoeuvre and the breadth and flexibility..to make plausible popular claims".

Wellbeing has a high register status which "endows it with pausability if not respectability" "the higher registers are by no means atrophying...they provide caches of rare words useful to those seeking to mystify or impress a gullible public."

#### **April 2009**

I have been thinking about how I argued previously that wellbeing is primarily located within the higher registers. I hadn't really thought deeply about register before and it has really got me thinking about my historical/linguistic chapter. Hughes reckons register is a "fairly recent linguistic term recorded only from c.1056, refers principally to language variation according to social role or social situation, especially to the degree of formality in the language employed. Whilst degree of formality is traditionally the aspect most stressed in discussion of Register, Register can also be demonstrated via numerous alternatives in word choice e.g. direct or vague, old/new, concrete/abstract etc" (Hughes, 1988:5).

I still think that wellbeing is primarily located within the higher registers. I never hear anyone on the street outside of University talk about it although it is used more and more on tv. My research indicates that wellbeing occurs not in the central/neutral or lower registers (slang, colloquialism) but in the higher registers (literary, academic, technical). I think I will attempt to do a registers of wellbeing to map this out.

#### June 2009

MY PhD rests upon the layers of understanding and awareness built upon the work of Foucault, and Rose. Fullagar, (2002) considers questions of power and freedom in relation to "the formation of contemporary modes of subjectivity through leisure and health practices" (p.2) and offers a cultural analysis of the way discourses of leisure and healthy lifestyles have been produced through the governmental objectives of health policy and promotion aimed at the body (Foucault, 1991; Rose, 1999). However after reading widely Fullagar amongst others and having ruminated upon applying policy analysis as part of my analysis and in reviewing one particular government health paper I do not believe this is an avenue I want to pursue.

#### Aug 2009

I have reflected at length on the difficulties I faced in broaching aspects of wellbeing concerned with sexual self image, sexual identity and sexual intimacy with participants particularly with Jim. My relationship with Jim is more formal than my relationship with Alison and I have found that starting a conversation about sexual dysfunction and sexual intimacy has perhaps been made particularly difficult as a result of my research project 'Exploring health and wellbeing in people with LTNC'. During this project I spoke with younger men about how their condition impacted on their opportunity to engage in sexual intimacy. Whilst I found it easier to broach this with the younger male interviewees than the older male interviewees I still felt that this was an area which required a cautious and sensitive approach. I am aware that men with neurological conditions such as Parkinson's disease, Multiple Sclerosis are likely to suffer from reduced sexual intimacy and that this has profound implications for their wellbeing and their identity as males but I struggle to feel comfortable and confident in raising these issues. I have pondered how I might start these conversations with participants particularly with my co-researcher Jim. Would he prefer me to ask him face to face, via email, not at all? Would he be happy to talk about it these things openly? Would a conversation initiated by me

about sexual self image, sexual intimacy and sexual identity alter his perception of himself or alter his perception of how I 'see' him?

How comfortable might Jim feel about talking openly with me? How might our discussion of the topic alter his perception of himself? Has it altered his relationship with his wife? To what degree might his issues with reduced sexual intimacy lead to compensatory actions or behaviour such as a portrayal of himself as a good husband and good provider?

## October 2009 Sexuality and intimacy

Whilst thinking about sexual intimacy and how to broach this in interviews I remembered how difficult and uncomfortable it was for me and the men I interviewed as part of the Itnc project to explore this aspect of their wellbeing. I had forgotten that this aspect of their wellbeing only emerged after I had interviewed the men on at least 2 previous occasions and established some rapport with them. I also remembered that I did not experience a similar sense of awkwardness or embarrassment when discussing this aspect of personal wellbeing with women!!

I have spoken with women about how their health condition has impacted on their sexuality and sexual intimacy and whilst I found it easier to broach than with men I still let this arise organically and felt that this was an area which demanded sensitivity.

Frank, (1995) observed that sexual desire decreased, especially during periods of worsening illness. Many of the male participants I interviewed during the Long Term Neurological Condition project who had suffered a traumatic brain injury had become withdrawn or reclusive individuals who failed to overtly express their more intimate requirements especially their sexual needs but during the course of follow up interviews when a relationship had been established between us several implied that this aspect of their existence was a source of deep regret and undermined their personal wellbeing. Their current position was often alluded to in respect to their former 'normal' life when they had partners. This retrospective reflection further undermined their present and projected sense of holistic wellbeing. For

them the lack of sexual intimacy undermined their masculine identity and meant that for many the unfulfilled desire to have an intimate relationship with someone was especially detrimental to their emotional wellbeing. other men with an acquired brain injury (ABI) who at times exhibited unrestrained sexualised behaviour were enthusiastic about their sexual proclivity and expressed their sexual requirements explicitly. They also engaged in retrospective reflection regarding their former sexual behaviour but this was more likely to be expressed in unfavourable comparative terms of more restrained sexual interaction. However, as the reflective process deepened this was sometimes qualified by a covert admission that the sexual activity engaged in the present did not fulfil their desire for sexual intimacy. It is difficult to know whether or to what extent gaining this insight has fed into my approach towards the participants in my thesis. However these recollections of my experience interviewing others has brought to the forefront of my mind that this can be a particularly difficult situation for the researcher and researched.

#### December 2009

Re-reading Des Gasper has proved really helpful as its helped me to reflect on and clarify differences in research. Some research seeks to explore those conditions which foster a sense of personal wellbeing whilst some seek to explore the domains of wellbeing. I had not previously considered this but it is a key differentiation and one which is rarely clarified by the researcher. Des Gasper refers to this differentiation as that which is "a motivational force and a universal value or goal" (Des Gasper, 2004:7). I believe that this "inconsistent usage and 'conceptual slippage' (Des Gasper, 2004:7) undermines the usability of wellbeing in practice and may act as a barrier to multi-disciplinary approaches to wellbeing research. I believe that whilst there may be universal domains of wellbeing this thesis is concerned with exploring those conditions which foster a sense of personal wellbeing.

One of the most pertinent points in connection to my work is that he voices concerns regarding usage of objective wellbeing and subjective wellbeing. this is a point which may seem pedantic but underlines the manifold issues which abound in wellbeing research. I had originally been intent on using term subjective wellbeing

but this literature review has highlighted that there are too many widely differing understandings of what is meant by subjective wellbeing and this has led me to review my position and adopt a term more readily understood "self reported accounts of wellbeing".

## February 2010

It is over a year since I began to discuss with Alison and Jim how our language, socialisation, class, gender, educational attainment, and position within society affects the choices we make in the research such as what wellbeing means and how we internalise and externalise our inner and outer representation of what personal wellbeing means to us. Initially we all struggled to with this but Alison's recent email confirms that she is beginning to get to grips with this process. my work on reflexivity for the PGC AP has also helped me enormously, I have read much more widely and appreciate how significantly your views and practices affect how you relate to people and ideas and how these in turn are affected by your upbringing, local environment family structures etc. These all impact on what we choose to discuss in our interviews and how we 'talk' about them. I was reflecting the other day that Alison probably projects a certain image of herself to me because we like each other and thoughts or opinions often need to be aligned with image. Likewise I believe the reason Alison and Jim and I began this collaboration has much to do with our shared sense of identities as w/c, left wing, pro-environmental Mancunians.

#### March 2010 Questions to ask of the data

To what degree does academic official and professionally based concepts of wellbeing occur in lay perspectives?

How much do the expectations of different certain sectors of society vary and how much does this impact on my wellbeing accounts?

Sexual self image of people with long term conditions was primarily identified through the eyes of others (neighbours, friends, children and societal attitudes) and this may have an impact little considered by the individual but never the less is important to their sexual identity and confidence levels. Sexual identity of older

women in particular is something which society portrays through relationships with men. Women are expected to have a man in their life and without one feel that their position in society is undermined. This may partially be as a result of their own perception of needing a man but societal pressure undoubtedly plays an important role in the manifestation of single Alison looking for a man.

Sexual activity and societal expectations need to be addressed, particularly in relation to older people wellbeing as it has traditionally been ill regarded but I don't know to what degree this may surface in this research.

The use of recreational drugs has emerged at points in interviews so to what extent do I consider the use of recreational drugs to alleviate some of the symptoms of poor wellbeing? What about the use of recreational drugs in those sectors of society not traditionally associated as recreational drug users such as older people and the impact this may have on their sense of self and sense of wellbeing?

#### June 2010

Double hermeneutics - I had not thought much about tangible and intangible modifier, it is reading through Jim's words that puts this more to the forefront of my thoughts. It is this circular learning and knowledge construction which is really interesting. Jim and I are shaping and shifting each other's conceptualisation of wellbeing. It is a very active process (other words like gay and celebrity are also going through re-construction by different parts of community. They are terms with contested meanings and may be conceptualised quite different for older people than to say young people). Jim and I have each taken on board aspects of our own interpretations of wellbeing and adding it to what we think about wellbeing or at least we reflect on it (is this me thinking of things from other people's perspective, I can't know how much Jim has taken my meaning into his reckoning he may have just thought briefly about it.

#### September 2010

Jim has been asked to consider 'wellbeing' in depth as a result of his involvement first with the Valuing older people board and then the long-term conditions

project, how much of this has influenced his thoughts and attitude to wellbeing? This is an important part of my position of using social construction. Jim because of his background, position in society has taken an academic and possibly prosaic approach to thinking, discoursing and recording his thoughts about wellbeing. His language is that which you might expect from someone working in a professional capacity (which he was - unpaid 'expert' voice) and he undertook research from around the world about what wellbeing research has thrown up, what it means to people etc (almost totally from westernised English speaking nations – US, Australia, Canada, NZ).

#### October 2010

Jim idea of culture has altered as a result of me asking him what he perceives culture to be. He spent many hours thinking and reflecting upon what culture in the wider sense and culture in a personal sense meant to him. Jim does not perceive class as an important part of cultural identity, in fact he fiercely opposes the notion of class. It will be interesting to see Alison's response as I perceive her position on class is perhaps closer to my own.

I am fiercely w/c and proud of my class heritage and the differences between us and m/c or u/c. In seeking to construct knowledge between us we impact on each other's ideas and constructs. This process continues indefinitely. My idea of jim's cultural identity is very much led by my own cultural identity. E.g. I perceived Jim to be a w/c, Mancunian like myself but he defined himself reluctantly as 'upper w/c' and from 'Bury' whilst these may appear subtle distinctions it imperceptibly altered the way I perceived Jim and perhaps the way he perceived himself through my eyes. These are also perhaps a symptom of his education and era?

Jim said he didn't think wellbeing affected culture but that culture could affect wellbeing. Puts a spin on causality which I had not thought of before. I need to ask Carolyn how much my analysis can include these changes to my thoughts, as I don't want this thesis to become one of those where the author talks more about themselves in relation to the data than the data!! the questions I went on to ask Jim

and Alison were often the result of things the other had said and I thought ooh that's interesting I will ask them.

#### January 2011

I have found few studies which are interviews with older people about wellbeing so I am keen to explore the basis of finding out about wellbeing and whether it is rarely used in lay discourse or thoughts. I would want to try and get a 'picture' of what wellbeing means in every day context and was as in more specific aspects so that the research moves from the general to the specific enabling me to get a picture of wellbeing in terms of their everyday lives and their life-course.

So for example particular dimensions/aspects of health that it made participants think about wellbeing. PERHAPS the difference between their health as a kid when they could run around in the woods and never giving a thought to health and now when getting up out of the chair makes participants think about how health controls a lot of what your able/unable to do and the implications for their wider wellbeing? What did wellbeing then mean in the context of their lives now?

I want to get some 'stories' of their lives as the context within which the state of health makes a difference and so 'wellbeing' becomes concretised alongside their conceptualisation of health. Maybe wellbeing becomes a more concrete 'thing' when it emerges in relation to the life narratives of people? When asked about wellbeing you think of its relativeness, your health now and how it consciously affects your wellbeing and can reflect that as a kid your health was still central (because it allowed you the freedom to roam the woods but you never consciously thought about it.

#### March 2011

Wellbeing must be considered in the context of life course but not in the cut and dried manner which identifies chronological age as the signifier but in the context which takes into account experience. Alison talked frequently about the AA and her immaturity prior to her attendance there she made similar reference twice to her son's attendance at the AA and its impact on his maturity/ growing up (or

developmental age). It is also well accepted that children who have experienced war/famine/displacement are 'old beyond their years' and this is likely to mean their wellbeing is more in keeping with their mental or developmental age rather than chronological age. Similarly people born with health conditions which affect their mental development cannot be considered to have similar requirements for their wellbeing as people with the similar chronological age. Research (nef) continues to identify that young children have different requirements for their wellbeing than older children and that both of these differ in their composition to adults. I think further distinctions have to be made between young adults, mature adults and older people. In which case does it make any sense to talk about wellbeing as if it's a catch all concept which can be placed at the heart of government policies and legislation, particularly when research has already identified that culture must be taken into consideration when addressing wellbeing?

## May 2011

Shortcomings of using interviews - reading through some of the interview transcripts it appears obvious to me that some can be read or interpreted as that which is based on a more traditional researcher/interviewee type of relationship whilst in others there appears to be a much more mutually responsive sharing type of researcher/participant relationship. I Think it's also important to reflect upon how these relationships impact upon the analysis phase as well as the collection phase. Perhaps with someone with whom I have formed a connection and whom I have therefore built up a sense of attachment to may be obscuring or colouring the way I interpret their discourse. There are times when I wonder if I have formed barriers as well as attachments to some of the participants e.g. D and that this emotional interaction means that I may sometimes have read too much or too little into things. Methodological concerns like these have increasingly led to utilisation of a network of qualitative researchers who read through each others analysis to try and offset these concerns.

#### June 2011

Taking into consideration what K said I can see that Sointu's work in which alternative medicine and engagement with the process participants talk deeply about wellbeing and its influence. The work of Newton in showing author's whose research has shown that thinking back to nature is big boon to wellbeing because children formed attachments to nature and wellbeing which feeds them to this day. These 'nature' 'environment' papers will be really important for my analysis of Jim in particular and considering my work in relation to other people's findings because Jim wellbeing story is closely related to 'nature' 'environment'. Could I look at how different Jim and Alison relate 'nature' 'environment' to their wellbeing on the grounds that some research proposes that you have to interact with 'nature' 'environment' as child in order to form 'emotional attachments which make 'nature' 'environment' important to you and your wellbeing?

#### September 2011

It is vital for me to keep on reflecting on all aspects of the research process from the generation of questions to the dissemination of findings. My engagement on the Post Graduate Certificate for Academic Practice has been hugely influential in my reflexive writing and has helped my growth as a researcher and a lecturer. I continue to draw upon Gibbs 'reflective framework' (1988) and Moons' no nonsense 'Guidelines for reflective writing' (1999). I particularly like Moon's directives to "Write in first person, be honest, be flexible, let the words flow and to use your own words -be informal"

One of the most important reflective tools used by Moon (1999) and Brookfield (1994) is trying to learn as much from our shortcomings as from our successes. Engaging in reflective practice is a fundamental aspect of working in qualitative research. I have and continue to critically reflect on my praxis and identified myself and was identified by my colleagues as having a humanistic approach to teaching and learning but as praxis is concerned with adapting both your theoretical and practical understandings and framework. I am striving to develop my humanistic approach into an eco-humanistic approach, that is a collaborative style which is underpinned with a life centred approach rather than a human centred one. This

approach is helping me to evaluate what I have learnt from the research process, particularly the contact with older people in interviews, how listening to others may have changed or entrenched my original views and how this may impact on my personal and professional development of ideas.

I do continue to reflect upon my commitment to environmentalism and how this may have influenced, perhaps at a subconscious level my approach to this aspect of personal wellbeing.

#### December 2011

I am concerned about services which seek to collect wellbeing data generically rather than at individual level. Possible explanation of why some psychological issues were not raised because people relegate these to the back of the queue when they are facing constant challenges to undertake ADL for example the need/urge/desire for sexual contact and or intimacy of a different nature.

Are aspirations for example largely tied to your class, gender, socio-economic position? And like in obesity research is your concurrent class more significant than the class you were born into? Jim like me considers that himself w/c but like me would probably be classified as m/c now.

#### May 2012

It's important to reflect on the difficulties which many participants experienced in trying to formulate and express their conceptualisations of wellbeing and how I was called upon to try and help without 'leading' to facilitate understanding and discussion. It was often hard for people to enunciate and express ideas that they had about wellbeing beyond the superficial, providing guidance concerning how to facilitate participants understanding and engagement in the process was an issue as I retained a notion of impartiality and bias more consistent with positivism. That was symptomatic of the problems which beset this stage of the research because the majority of lay people who have been asked about wellbeing struggle to relate to it conceptually (see Woolrych and Sixsmith, 2008) and to put into words what they perceive when they think about their personal wellbeing and how services can

better reflect the personal wellbeing of older people with complex health needs. The analysis of the data within my theoretical framework for understanding wellbeing has identified some relationships between older people conceptualisations but to what degree can I expand on these to initiate the development of service provision, from which indicators for assessing the impact of service change and transformation can be derived?

#### June 2012

Enrolment on the PGC AP has been hugely influential in helping me develop my critical reflexivity. I have read extensively, written widely and ruminated across different approaches to being critically reflective. My intention in this section is to adopt a critically adoptive stance in relation to the challenges and dilemmas I have faced and continue to face in pursuing a research undertaking (analysis) from a dominant perspective i.e. I am in control of the situation (being the academic overseeing this process). It is also important to reflect that I am also in a less dominant position with regard to satisfying the criteria of my supervisors and exam board. This has a marked influence on my interaction with the research data and research participants, this criterion means I have tended to detach myself from the more personal elements of the process for example I struggled with my intention to use the personal pronoun within my thesis as standard practice frowns upon this. It is also common practice to maintain a professional distance from research participants even in qualitative research. the power invested in me in relation to the participants is overwhelming, at all stages in the process I have primarily directed proceedings, whilst I have tried to mitigate against this by offering Jim and Alison a stake in the data analysis process, e.g. when and where to meet, data collection methods (telephone, face to face, email) by choosing a particular form of data analysis (identified through former iterative training process as most easy for participants to understand and apply), The choices have still emanated from me, because I am located in the process as the professional and it will be me whose status benefits from this undertaking (should my thesis be considered worthy).

## July 2012 Eco-humanistic approach

Eco-humanism has developed from the eco-pedagogy movement, it is concerned with the relationship between humans and the Earth and is critical of the racist, sexist, classist and specieist values which underpin current notions of sustainability and environmental policy (Freire, 2004). Eco-pedagogy challenges the failure of society to articulate the socio-economic and political need for a wider understanding and interpretation of the ways in which modern societies and industrialising cultures promote unsustainable lifestyles (Gadotti, 2000). Michnowski, 2007 defined Eco-humanism as a "partnership-based co-operation for the common good of all people (rich and poor, from countries highly developed and behind in development), their descendants, and the natural environment - commonly supported by science and high technology" (p.35). Eco-humanistic approaches seek common interest solutions to community issues. This is underpinned by rationalisation that community action must encourage development alongside the environment instead of expansion at the cost of the environment Michnowski, 2007.

Eco-philosopher Joanna Macy, created a theoretical framework for personal and social change concentrated on the cultivation of ecological awareness and conceptualisation of an ecological self and the need to extend our eco-logical self by extending our identity to include other forms of life and elements of the life support system (Macy, 1990).

By drawing upon the work of the Eco-pedagogists, Michnowski and Macy I hope to expand my conceptualisation of humanism to incorporate notions of ecology.

However I must also endeavour to reflect upon how these emerging thoughts and perspectives are influencing the direction of my research.

#### August 2012

I have spent enough time to have built up an emotional relationship with J and A and this is increasingly affecting how I relate to/ analyse the data as I do not want to portray J and A in unfavourable light. It is also likely to slow the process down as I send back my analysis to Alison for her to reflect upon my reading of her data.

It is this positioning which has made me reflect upon the places in which J and A had our conversations/ interviews. I always left the choice of venue to them to facilitate their sense of control and input into the process. This meant we met in some very different locations over the 2 years e.g. my workplace (J and A), council chamber room, cafe x2 (prior to J having meetings to attend and once when A was shopping in town) and in their own homes (particularly A).

My relationship is very informal but I know A is not well off so I have tended to give her money (£10) and bus fare for her time as I would if she had been helping on a research project. J however is financially comfortable and declined the money so I have given him petrol money but not participation money I have just paid for his snacks if meeting in cafe. This economic positioning makes my relationship very different and alters what comes out of the collaboration. I think Alison and I have a more intimate relationship because I appreciate her financial position and we have talked about how my mum was in similar financial position at her age (plus the drinking issue they shared). J and I share a more traditional relationship, primarily I think through J desire because he wants to see me as researcher and to keep things within professional boundaries which for example makes me awkward when wanting to ask about sexual intimacy.

I would argue that the in balance of power demeans both the researcher, research participant and the research process. This reflexive process should therefore help me to reflect upon how power can best be negotiated between myself and those engaged in the research process.

## September 2012

A crucial question I must reflect upon is whether my thesis has the critical depth to ensure that participant's voices and understandings are afforded sufficient weight by myself and examiners. In addition to which my reflective process must also seek to take into consideration how the socio-economic, political, cultural and historical positioning of myself and the participants impacts on the research process and findings without placing so much emphasis on this that it overshadows the actual research (a not uncommon problem in research papers which embed reflexivity).

#### November 2012

I think I selected Jim and Alison because they had similar shared identity and political ideology to me and perhaps they agreed to participate because they perceived I had similar shared identity and political ideology to themselves. How does this shared agenda impact on every aspect of our collaboration?

Having Alisontaking part in the analysis of her data has been very important for me. Think it's important that Angela's reading of her data forms part of the findings and it also offers me a different perspective on particular issues or ideas I have formed a connection and or attachment to which may be obscuring or colouring the way I interpret the data. There are times when I think that I have formed barriers as well as attachments to some of the concepts underlying the data and that this emotional interaction may interfere with the conformity of the findings.

#### **December 2012 Agency**

Agency is an important aspect in regulating what we do and helps to shape who we are and what we aspire to. Mead (1934) and Cooley (1902) showed clearly that reflexivity among human beings is rooted in the social process, particularly the process of taking the role of the other and of seeing the self from the other's perspective. As a result of this process, the individual becomes both the knower and the knowledge maker. The second reflexive process refers to agency. Agency refers to the experience of being an active cause in the production of outcome. Reflexive agency is the process whereby I can retrospectively look back on myself in order to perceive effects on myself.

Reflexive processes must I believe incorporate emotional reflexivity so for example it is involved in emotional identification playing an important role in emotional display and coherence. Emotional display involves the self-regulation of emotional exhibition for the purpose of producing intended effects on others' minds Mead (1934). reflexive processes are also involved in stimulating emotional experiences but different emotions may have similar manifestations. According to Tavris (1982, p. 94), anger "shares the physiological symptoms of joy, excitement, fear, anxiety,

jealousy, and the like." In exploring the internal state of physiological arousal, then, it may not be clear which of several possible emotions one may be experiencing Tavris (1982, p. 94). My use of observations of participants including assessments of what their expressions implies or what their vocabulary indicates may be an aspect of this research which I am not entirely well equipped to display. Another reason for my cause for concern is the fact that emotions may be mixed. When people show distaste they may also be feeling other emotions can I truly be confident that what I perceive as a grimace is not actually a grin?! When different emotions are experienced simultaneously, it may be difficult to dissociate them and to identify their respective natures Tavris (1982, p. 94). This ambiguity may also exist because there is no foundation upon which this display of an internal experience can be confidently assessed by me who does not have that experiential understanding to draw upon. Although participants may appear to be referring to the similar experiences when they use certain terms or language it is important to acknowledge that their experiences are unique and may be incommunicable. The great thing I learnt from the PGC AP was that we are all living in our own experiential context. According to Leff (1977), some languages do not have separate terms to designate depression, irritability, and anxiety. For example the German term Schadenfreude, which refers to a kind of malicious sense of glee, may be an emotional state which is not widely recognised by non german speaking peoples (Gordon, 1990). This is very apt for the exploration of wellbeing in relation to culture as oishi and diener have contributed significantly to debates concerning the applicability of the term wellbeing in language groups which do not have a word for this concept.

An important aspect of the double hermeneutic process is that within western cultures I believe, there is the constant effort to establish, displace, or change our fellow citizens perceptions and opinions. This is particularly pertinent I think for Jim who claimed that being able to persuade people to his rational way of thinking was important in that this power reflecting back positively on his sense of wellbeing. perhaps his self image and role in society requires that he perceives himself as a positive and influential force. It is also apparent in A with her great comfort in being

able to influence the opinions and behaviour of her children and grandkids and of course my own efforts to embed thinking around wellbeing in terms of gender, class and the environment (p.8) social situations and social roles dictate emotional responses from people this may go some way towards perhaps explains Alison professed comfort in feeling maternal after 'failing' to conform to her own and others expectations of motherhood.

## January 2013

My work on neo-liberalism has revealed that at its simplest, it can be defined as the sustained attempt, since approximately 1980, to extend the capitalist market to the wider world and previously personal spheres of human experience and endeavour (Rose, 1999). It can be perceived as an ideology; as a politico-economic force of global capitalism; as a source of economic policies and as an ethos and philosophy which has increased power in westernized societies. Carolyn has urged caution in utilising such language in my thesis but it forms a key aspect of the wellbeing agenda and its relationship with the promotion of certain behaviours in the population and has come under increasing criticism in recent years

I attended workshop which brought together Postgraduate Researchers from three of MMU's Research Institutes: the Institute of Humanities and Social Science Research (IHSSR), the Manchester Institute for Research and Innovation in Art and Design (MIRIAD) and the Research Institute for Business and Management (RIBM). The workshop intended to give us the opportunity to hear speakers link their own work on neo-liberalism, across three thematic panels. I found the event illuminating especially the workshop where I was able to talk with others about neoliberalism in relation to our own work.

I particularly enjoyed the session which explored neoliberalism in the context of economic policies implemented by governments and bodies such as the IMF in relation to market capitalism and welfare systems and the session which explored aspects of ethical and epistemological implications and the role of neoliberal ideology and practice for researchers

## March 2013 Reflection on the research process

It's been pretty hard going through the interview transcripts with Alison for the 'sharing and learning experience'. when I was reading through and thinking about my analysis and reading of Alison interviews it was an ethical consideration but more than that I was worried about the conclusion I had made of her data, of her life her dreams and fears and what she would think of the way I had read her life. Would she feel upset, would it make her reflect badly on her life.

The people I've listened to and their stories and analysing them, ordering and coding them into themes has given me a feeling that I have disconnected with the data and this sense of connection with the participants was very important to me and the narrative this PhD seeks to tell. In some way it seems strange, almost like I am exposing the underbelly of their lives however I am happy with what I am producing as I think I am writing this as a result of their confidences. This came to mind when I read back the words of one lady who apologised for ranting and for giving her the opportunity to get some niggles off her chest

Like with lecturing I felt strange sometimes reading through interview because I forged a connection with a few of the people I've interviewed as I had spent time with them and observed them as part of the Long Term Neurological Condition project and I think I make sense of their interviews in the context of the greater time I've spent with them and the knowledge and understanding that has grown between us.

# **April 2013**

Parker (2005) argues that research uncovers how people are positioned by themselves, society and themselves to relate and understand their lives. So in context of my research J as a man is culturally positioned to view/understand his wellbeing within the context of certain socially constructed notions e.g. being seen as a good husband/provider, in terms of power, money, influence, social position and work whilst A projects her wellbeing along familial lines, personal safety, friends, and community,. This is how their realities have been and continue to be

shaped. Parker (2005) also contests that our capitalist (neo-liberal) society forces people to blame themselves e.g. A accepting that she undid herself by believing her wellbeing would be achieved through her relationships with men. In actual fact society positioned her to accept that was effectively her only option as an unqualified working class woman in a post industrial city.

#### June 2013

Marcus (1994) argued that the work of politically motivated researchers who assign their research as giving voice to marginalised people is more likely to put words in their mouths or construct meaning/assign meaning which may not be there. I would agree with this to a degree as it may be that I have inadvertently assigned meaning or constructed meanings which are congruent with my position and my desire to produce a successful PhD. However I am more in the relativist camp in that I believe the research process and analysis and data is socially constructed 'in particular discursive contexts '(Griffin 2000) not discovered and that the work Alison has undertaken in the analysis of her data has helped keep me grounded.

In the spirit of reflexivity I have reflected at length upon the terminology I used I have speculated on how the term 'researcher' promotes a representation of myself to myself and to others. Within the term is located a position of authority and power particularly in relation to 'participants' which perpetuates an in balance of power between them and me. This places both my self and the 'participants' in pre determined roles in which I hold the balance of power. I originally chose to use the term 'enquirer' but have since rejected this because it did not resonate with my interpretation of what I was seeking to do, perhaps I enjoy the power which the term researcher invests in me too much to give it up!!

#### August 2013

Alison and Jim have worked so hard to read and understand more diverse aspects of the research process such as reflexivity and this has contributed so much to this thesis and I suppose it helped them develop into informed and active participants. I still think they should be able to receive some level of acknowledgement close to

mine as they have made an inestimable contribution throughout the research process to this thesis. Their commitment and hard work has enabled us within a collaborative framework to explore in great detail the complexity which underlies ever changing socially constructed notions of wellbeing in an environment which recognises and values the interplay which exists between the 'researcher' and the 'researched'.

#### September 2013

In considering wellbeing within neo-liberalism and as a conceptual lever one only has to view the increasing promotion of workplace wellbeing. The concept of wellbeing has been promulgated as a risk management tool (Harrington, 2009) and a human resources practice (Sointu, 2005). The concept of wellbeing has been lauded for its potential to help reduce absenteeism, enhance staff morale and staff retention, reduce expenditure and increase productivity (Sointu, 2005). (Harrington, 2009) noted that it also has the advantage of being tax deductable and enhancing business reputations within the corporate social responsibility agenda.

My reading indicates that one of the main reasons for the advancement of the wellbeing agenda in the workplace may be the expense associated with the provision of healthcare, particularly psychological wellbeing as stress anxiety and depression are amongst the main causes of poor workplace wellbeing (Lyard, 2005). Investing in the health and wellbeing of employees may simultaneously acknowledge the importance of establishing a satisfied creative workforce, whilst reducing the costs of workplace healthcare provision and raising economic productivity, all of which are perceived as benefiting personal and national interests (Harrington, 2009).

Harrington, (2009) notes that the UK has encouraged companies to harness the 'power of wellbeing' in order to benefit individuals, communities, business and wider society. This has seen the emergence of wellbeing tools, consultation exercises (ONS) and promoted initiatives which are aimed at supporting the corporate world to invest in those aspects of employment practice which contribute to stronger and more efficient companies. Wellbeing practices include

health screening, workshops designed to combat stress, tension and workplace pressures, access to counselling services and the promotion of nutritional advice and resources in order to reflect nutritional practices which have been identified as key in the fight against preventable diseases. The identification of alleviating negative components of wellbeing whilst promoting positive, protective components of wellbeing is perceived as likely to impact on employee health and wellbeing and is now recognised as a legitimate company undertaking which will deliver a wide range of benefits at a relatively small cost (Harrington, 2009).

#### October 2013

How do people decide whether they are in love (or are experiencing positive wellbeing)? According to Averill (1985), society provides them with a set of criteria against which they can match their experiences. Among the criteria are 1) idealization of the loved one (e.g., "the most beautiful girl in the world"), 2) suddenness of onset (love at first sight), 3) the physiological arousal associated with sexual excitement, and 4) "commitment to, and willingness to make sacrifices for, the loved one" (p. 99). Turner (1970: 228-30) identified 11 features of an emotion that is characterised as love. He argued that whether people are in love or not may be based not so much on their actual feelings as on matching their thoughts, feelings, and behaviour with the emotional template provided by society. This example of social conditioning is an important consideration for those who seek to explicate understandings of complex concepts such as wellbeing.

#### November 2013

Reflexivity about the research process which I utilised to produce this thesis is an important aspect of the approach taken to co-create the knowledge and understandings of this PhD. I explored the reflexive understandings of the academic and positional power which I hold in relation to the research participants.

Discussions were held between myself and the research participants with regard to the degree of involvement in the design, production and analysis of the knowledge production process. The positioning, character and depth of collaborative enquiry

for each participant produced very different outlooks and expectations concerning the perceived degree of partnership in the research process.

My personal and professional commitment to more collaborative approaches to knowledge generation is in accordance with a social constructionist / hermeneutic approaches. Engaging research participants at all stages in the research process in such a way which respects their unselfish giving of time, innermost thoughts etc goes some way towards valuing their unique contribution. Reflexivity has facilitated my appreciation of the emotional identification which I have ascribed to my core participants.

It is perhaps very important to reflect upon the relationship between myself and chief participants. Sharing gender and similarities in childhood e.g. alcoholic parent, growing up couple miles apart, class identification, political alignment with A means we have a much more informal relationship. However I share some of these attributes with J and other more formal ones e.g. engagement with council and advisory boards, connection to environment etc. But how have I acted upon, influenced and informed the way this thesis has developed?

## **Dec 2013 Posing challenging questions**

As I have moved through this reflexive process new ideas have emerged such as Why did I adopt my particular research epistemological position - Because it was in agreement with my educational positioning and current ideological perception of the world and meaning making. At this point in my life I still believe knowledge is co-created but having just finished my lecturing degree and witnessed the cynicism in lecturers with years of experience I have become very aware that continued engagement in the learning environment may well lead to a change in my beliefs and behaviour.

Why did I adopt my particular method of data analysis – Thematic analysis has always intrigued me for its simplicity and the depth of understanding which can emerge. The fact that both co-researchers opted for this method of analysis was a vindication of its pragmatism. It has always niggled that it is perceived as a lowly

means of analysis and part of me would like to be in the vanguard of those researchers who in recognition of its versatility are increasingly claiming the right to use thematic analysis.

My research question grew from the circumstances surrounding what I believed might secure funding for a PhD In what could be construed as a circular argument I was aware that 'wellbeing' was a buzz word which ticked the right boxes and it was a strand within the research project I was employed on. It therefore seemed likely that it would secure the funding for a PhD. Perhaps at this juncture reflecting upon why I wanted to do a PhD is appropriate. Whilst I was still a history undergrad I flirted with the idea of doing a PhD. I always loved history and I enjoyed my degree very much. Many of my friends went on to do PhDs but I knew I did not want to work professionally within history so decided not to pursue a PhD at that time. I do have a strong attachment to the learning environment though and went back to university twice to do a PGDIP and MSc, it seemed only a matter of time before I undertook a PhD. Working as a researcher in academia is very different to research work in other fields. University is a very hierarchical environment and I like to challenge myself and grow, I felt my opportunities to grow and progress as a researcher in this environment would be stymied by not having a PhD. My desire to contribute to knowledge and develop into a good researcher also drove my efforts to gain funding for a PhD.

Other questions which I have yet to answer include: Why did I adopt my particular paradigm, Why did I adopt my particular method of data collection, Why did I adopt my particular approach and how these choices may have constrained what resulted from the analysis.

#### January 2014

In terms of what I have revealed in my reflections I think I have attempted to consider the impact of political persuasion, race, gender, class, sexuality on the parameters of this research amongst other things. I would say political positioning is key because of my utilisation of the theoretical framework which may have led me to prompt certain responses or narratives from the participants and co-

researchers. It is also important in terms of how it impacts on my relationship with both co-researchers. It is perhaps important to stress that my thesis has developed along particular ways as a result of the research participants having a similar geographical and cultural background. As individuals we all identify with being (both how we identify ourselves and how we are identified by others as) working class Mancunians, who are politically aware, hold left wing political sentiments and Are community engaged. These similarities are countermanded by other aspects of our socialisation as they are white, from a different generation and have experienced very different educations this impacts on how we relate to each other, our use of language and our shared conceptualisations.

The meanings that Alison and I have generated will always be open to speculation, interpretation, elaboration. Whilst some researchers include full transcripts of the research data so readers can obtain a sense of the researchers active contribution this fails to unearth those unconscious processes which impact on research e.g. choice of language, use of visual cues and how these open up some avenues and narrow off others. German researchers have used process where research team meet to explore the thoughts, feelings and associations which have resulted from the research as a means of considering what has been consciously or subconsciously transmitted to the research participant (Marks, 2001). My interest in the taboo aspects of wellbeing may show through at points when I try to elicit discourse around how research participant experience these aspects when they had no thought of discussing it. However I made the decision to do this from the outset as research indicates that people may not think about the less obvious factors which impact on their wellbeing e.g. socio-economic disparity, political circumstances (Easterlin, 2005, 2009).

By and large though I have employed a narrative style conversation in which the research participant has been encouraged to talk freely and thoughtfully about their experiential understanding of personal wellbeing. this approach however has been challenged as tending to focus more attention on the emotional meanings attach to the research questions rather than their cognitive meanings.

## March 2014

Alison and Jim have worked incredibly hard to read and understand more diverse aspects of the research process such as reflection and they have made an inestimable contribution throughout the research process to this thesis. Their commitment and hard work has enabled us within a collaborative framework to explore in great detail the complexity which underlies ever changing socially constructed notions of wellbeing in an environment which recognises and values the interplay which exists between the 'researcher' and the 'researched'.

## Appendix J: Alison's Reflective Diary (2008–2012)

As part of the participatory research approach Jim and Alison agreed to keep reflective diaries in which they would log reflections, ideas, thoughts and potential discussion points for interviews.

## Email - 20/1/08

Hi Jo

Thanks for the training session I really enjoyed it although it was quite challenging. I'm glad I met up with Jim though I hope he wasn't upset with me?! As his wife is a councillor in his area I found it difficult to speak about my many problems with councillors as it made him defensive. It reminded me of a recent incident which made me very angry. At a council meeting a man was about to complain about large dogs off leads in his local park. A local councillor said "I hope you're not going to say they were rottweillers [sic]??" She then went on about her own 2 rottweillers [sic] and how harmless they were. He never got to finish what he had to say, we made a complaint about it. Its that sort of thing I've been talking about, no respect.

Speak soon Alison

### Email - 30/1/08

Hi Jo hows things? I'm ok just few lines as got dentist appointment soon.

The training sessions have [sic] raised my awareness of how part of me has been closed off and that I'm not as open minded as I thought I was. It's certainly been a learning curve for me and made me aware of the need for my own self-development. I am please to be part of this however small and I hope to develop the skills needed to improve as community researcher.

All the best

Α

#### Email- 6/2/08

Hi Jo

Being part of this research has been great, as its [sic] an opportunity to get more involved with the university, to meet different people, to widen my experiences and to try and look at things from the professionals [sic] side of things. I did struggle at times in the data analysis workshop but by the end I began to understand exploring themes, if I have to choose I would definitely pick the thematic analysis approach. I can see myself being able to get more involved with you in the future now with your PhD now that I have had some training and experience in actually doing it.

All the best. A

#### Email - 15/3/08

Hi Jo

Going to the X centre to do watch you interview and then do some interviews did feel a bit like being thrown in at the deep end but I survived and it was a good experience. Most people using the day centre who were interviewed had cerebral palsy and all though I hadn't really personally known anybody with this disability and didn't know what to expect, I felt quite confident speaking with them regarding services. I did notice that the answers were all the same. Yes all interviewees were happy with services they receive..... unlike anybody I meet, mainly able bodied people!!! I didn't know what to make of this and had suspicions. I realised it wouldn't be right to suggest things that may be a problem but felt that people needed more help to communicate, possibly from a relative or a disabled person with better communication skills voicing their complaints and setting an example of what sort of things could be improved.

Hope these reflections are useful?

Α

### Email - 25/05/08

Been thinking Jo re the interviews and I've been reflecting what if all interviewees say the same things all the way through the project? Would it just be assumed that everyone is happy?

We have similar experiences in our work as community activists, apathy can make people say everything is ok then they attend a public meeting and that same person has loads of complaints and views.

Other than that I feel quite confident now and I'm enjoying the work/practice as it is a totally new area for me. My interest in special needs continues to grow and it felt right for me to be involved in this.

I've been remembering all the names of the people at the x centre especially X who when we asked about his happy attitude in the face of adversity said something like "what's the point complaining getting angry or miserable. You can't change anything...

Look forward to our next wellbeing interview

Speak soon Alison

#### Email - 26/11/08 Respect / Power / Manners /

Hi Jo. Off the top of my head, the things that affect my well being are where I live and general neighbourhood. Even very small things that happen can have a large impact on wellbeing. Being ignored and feeling powerless impacts a lot. The wellbeing of individuals needs to take into account things such as stress and anxiety, sleep or the lack of it and the way people in authority treat you. I will give you an example of those in authority and the disrespect constantly faced in day to day dealings. My next door neighbour has dementia. I have challenged 'carers' who visit. They press the buzzer, give her a few seconds to answer and then keep

pressing. If she has been asleep and is woken like this she gets confused and distressed. Because of this she starts making her way downstairs to answer the main door, rather than let them in with the answer phone. Sometimes she forgets where she's going and comes back up screaming. If I'm in then I let them in myself to avoid the palava. I reported one who sounded very disrespectful, telling her [neighbour] to 'behave yourself' and saying it to her in a nasty voice each time she was leaving so one particular time when I just could not contain myself any longer I went out as she was leaving and said 'NOW YOU BEHAVE YOUR f..G SELF OR I'LL THROW YOU DOWN THE STAIRS!! (Sic).

#### Email - 2/12/08

Hi Jo. Hope you are well. Following on from my last email/rant downstairs there is a lovely old man who has had his operation cancelled for the third time now. This time because they had forgotten to tell him he should stop taking aspirin for 5 days before the operation. I could hear him groaning in pain some nights, a nurse has come recently and he seems better but it makes me feel so powerless that the system isn't looking after him. things like that, feeling powerless and not being able to help the older people who are my neighbours get the respect and care they deserve has a massive impact on my wellbeing.

tHATS ALL FOR NOW jo (sic) .Can you suggest other stuff for me to write about well being?

Alison

#### Email- 30/1/09

Hi Jo first time I've written anything this year and it seems to be another on the theme of respect or disrespect!!! My immediate neighbour moved out, after she had gone her smoke alarm was beeping. After the first weekend of it bleeping every 30 seconds I called housing repairs. No one came. I then reported it to the manager... Now 1 month has gone by, every night when I went to bed, usually feeling relaxed and happy I'd hear it bleep and my heart and stomach would drop

into a miserable feeling, of feeling worthless, that I didn't matter in the scheme of things. Although the housing manager is ok I sense no respect for tenants. This story is typical of the way we are treated by those in authority. They could improve most peoples' wellbeing at a stroke by making sure those professional in the public sector treat people with respect and show them some common decency. This story is typical of the way we are treated by those in authority. They could improve most peoples' wellbeing at a stroke by making sure those professional in the public sector treat people with respect and show them some common decency.

#### Email - 16/03/09

I met a friend from childhood recently they are thin on the ground as I left Withington at 16 and never went back. I am really happy to meet up with her, especially as she is such an intelligent woman who came from a family with as many problems as mine but different, she lives in Hebden Vale now. I met up with her a few months ago. It was great I saw the life I may have had if I had been able to use the opportunity of grammar school and I think about that often and it gets me down about my life now and what the future holds for me. I flunked that opportunity but I can see why now because I was not well I was bi-polar but no-one picked it up!!!!

#### Email - 28/6/09

Hi Jo. I was thinking about what we were last talking about and have some answers for you well more thoughts really. Ill (sic) do more later.

Question - Can you remember when you 1<sup>st</sup> started to think and talk about wellbeing?

Ans. Yes it was over 20 yrs ago, when I was a member of AA FOR 6 YRS, AND BRIEFLY 10 YRS BEFORE THAT WHEN I WAS A MEMBER FOR 2 YRS (sic). Back then it wasn't called wellbeing it was called lack of defensiveness, serenity, stillness, calm, being at peace with myself, happiness but it pretty much amounts to the same thing.

Question – Do you use the term wellbeing?

Ans. No I would probably use QoL, I never say wellbeing, well only when I'm with you.

Question - Do you think that the term wellbeing is a useful concept?

ANS- I NEVER USE IT AND DONT THINK THE PUBLIC ARE AWARE OF THE PHRASE I NEVER HEAR THE PHRASE USED SO NO I DONT THINK ITS A USEFUL CONCEPT, NOT TO NORMAL EVERYDAY PEOPLE (sic).

### Email - 4/8/09

Hi Jo here are some more comments on what we have been working through in terms of my perception of personal wellbeing.

Question - Q - Is wellbeing a meaningful concept to you?

I WOU; DNT(sic) USE THE PHRASE [wellbeing] MYSELF. I WOULD USE OTHER MORE APPROPRIATE PHRASES, DEPENDING ON SITUATION, WHO IM (sic) TALKING TO. FOR ME IT ISNRT (sic) SOMETHING THAT WORKS.

Question - Why is it or isn't it a meaningful concept to you?

I DONT THINK YOU CAN USE A WORD LIKE WELLBEING IN EVERYDAY SITUATIONS, LIKE ASKING PEOPLE IF THEY ARE HAPPY WITH THE SERVICES THEY GET. WELLBEING COVERS SO MANY DIFFERENT STATES AND SITUATIONS. iTS (sic) NOT LIKE 'SOCIAL EXCLUSION' FOR INSTANCE, WHICH WORKS.

### Email 21/10/09

Hi Jo

Sorry I have not been keeping up with my reflections too well, been really busy with with x [grandson] and y [daughter] but heres (sic) some more thoughts on wellbeing.

Question - Do you think wellbeing is significantly different amongst different age groups?

Ans-Yes, my wellbeing has changed a lot over the years, when I was young my image and what people thought of me was important for my wellbeing like it is when you are young but as I have got older things like my grandkids happiness and opportunities for a decent life have become central for my wellbeing.

Question - Do you think wellbeing is significantly different for men and women?

Ans- Mens (sic) wellbeing often depends on status and their position in the pecking order re work. Womens (sic) wellbeing is based more on the emotional/relationships/ friends and family. Worklife is important for some women but I imagine it doesn't lead to much wellbeing.

looking forward to meeting up again to talk through some more of my thoughts and reflections on personal wellbeing

Α

# Email - 14/12/09

Ive (sic) been thinking about what you said about wellbeing and life changing events and it got me thinkinf (sic) if my wellbeing has shifted significantly at various points in your life. The Answer to that is most definitely Yes!! theres (sic) Pre AA and Post AA. Theres pre meeting Carolyn and working with uni and post meeting. They are the biggest shifts in my wellbeing as they are points in my life when I started to explore myself, to reflect and to grow as a person.

For me some of the most important things for my personal wellbeing is my level of awareness, consciousness, personal development of different kinds eg (sic) the AA, the uni the counselling course. They are all things that are about doing something that gives back and that is really important to me.

### Email - 9/02/10

Question - To what extent then do you think a person's early environment and their upbringing plays in their sense of wellbeing?

Ans. I think both play a massive part in your wellbeing. the feelings of security self worth, self esteem, handling disappointment, issues of addiction, lack of education, family support, learning difficulties these all stem from a person's early environment and their upbringing. I think it can be hard almost impossible to put those things behind you without help and then they will always undermine your wellbeing and that of your kids and their kids.

Question - Do you think that defining wellbeing will help or hinders it usability as an outcome measure in how services are provided?

Ans-I think it will hinder things as peoples levels of self awareness are different, everyone would see different things as important for their wellbeing and if that didn't fit with what those in positions of power think is important for wellbeing they would be dismissed. Any definition of wellbeing is likely to rest on what the powerful think people like me should do, act, think and behave.

By for now

Alison

#### Email - 23/04/10

Hi Jo. A bit more re well being. For the last few years my son has helped a lot. He bought my computer a few years ago and has paid 10£ (sic) per month ever since for eastserve. I also have a landphone (sic) and he's paid for calls for last few yrs. I pay rental. He has provided me with a mobile and 500 free minutes per month as I ring X [Grandson's mum] a lot. All this has made well being possible. I even manage to save a bit and still give Y, my grand daughter in her second year at uni 5£ (sic) a week. She has 150 in summer towards her holiday and 100 at Xmas for pressies

[presents]. I always feel good when I hand it over. She works at a taxi office part time and in holidays and never asks for anything so I'm pleased to give this until she finishes uni.

The computer is really important for my well being and my family's well being. I use the computer to keep up with latest news and opinion. It's useful for keeping up with opinion in my own generation which I don't mix with much. I tend to steer away from anyone who is depressed, or constantly negative, so that doesn't leave much engagement with my own generation. Having a good relationship, with the help of free phones with my own kids, though 2 live in other towns is great.

Thats all for now. (sic) If I think of anything else I'll email. Regards Angela

## Email - 08/07/10

Hi Jo.

Things are ok here thanks. I'm looking after baby X a day and a half per week. My g/daugh (sic) is now at Burnage High for boys and doing very well. I made a new friend off a local website. AB has been writing little stories about her chidhood (sic) in Collyhurst. Very funny and original-and we've become friends. I know this will sound mad but I'm actually editing her book now, that she started recently, so its very exciting and inspiring me to write again. Hardly stopped recently. Glad to hear you've had some work. Hope you find something fullfilling after all those years of study. Don't know how you do it.! Have had quite a mental shift recently which feels good.

I'm off to sons (sic) in Middlesex tomorrow night for 10 days. It will be lovely to see you when you have the time. Wellbeing has improved recently as I have less responsibilities and feel more fulfilled writing. Take Care. Love Angelax (sic)

## Email 13/11/10

Hi Jo. Yes Alsation (sic) pups are beautiful. We dont see many alis (sic) [Alsatians] these days. I Remember when every dog was an ali/collie cross, or lab/collie cross

485

(sic) . We had a collie-Panda-when I was a kid. He used to round up the pigeons. We

lived on a green, with a public pathe (sic) across it diagonally. He'd lie in wait, for

any unknowing victim who dared to set foot on our green- then snap at their heels

until they retreated and walked round-so the path was hardly ever used for years.

Thinking back to those days has been a blast, I had forgotton (sic) how much fun

and pleasure I experienced at times during my childhood. It definitely gave my

wellbeing a lift today thinking back and remembering that there were some good

times. Email if you need any housetraining tips if it is a baby. Love Angela. X

Email - 15/02/11

Hi Jo. Thanks for putting that analysis in the post, it arrived few days ago but just

had chance to start reading it and I have really enjoyed reading it. Thought that

your suggestions were sound and that you had interpreted my perspective really

well, some of the things you said I had not considered at all but as I have been

reflecting I think your spot on!! Didnt (sic) feel that I would get as much from this

as I am so I am really looking forward to you coming round so we can discuss it

together.

Thanks

Love Angela

Email – 12/5/11 RE: Meeting

Hi Jo. Thats fine . Where do you fancy? Pub Lunch? I dont mind. If you have

somewhere in mind-maybe we can meet somewhere else? See you at 1.15 at

Palace Theatre.Oxford Rd. If you want to change meeting place email me a new

place. Ok -Look forward to seeing you ive been thinking about my wellbeing and

the kids and what things influence it.

Angela

## Email 6/7/11

Hi Jo. Good to hear from you. I hope you are well. Next Wed would be fine to meet up. The earlier the better for me.

Im still in the choir.loving it, its been great for my wellbeing and weve grown. Had a great gig yesterday in Blackley. Next Tues its our party and gig for various officials. The Tues after some church and gathering of people who work with elderly.

Im in a better placenow and think that as I grow more mature the things that improve my wellbeing change couldn't have seen myself enjoying a choir when I was younger wonder what other things I might get into as I get older!!

Send me a time and place for next Wed. Could you make it Thurs as Im at dentist in Hulme at 9am. I could hang about at mates in Hulme till I meet you. If not Wed will be fine.

Love aNGELA

## Email 21/9/11/

Hi Jo Hows things with you? I hope that you are well and that the family are well too? I am still singing with the choir? Got two Christmas gigs? I know that i was supposed to be in touch about meeting weeks ago but I am so busy it is not true. Thought we could maybe meet in Town next week, maybe wed afternoon/lunch. Be good to catch up and I am enjoying analysing my interviews and discussions i had with you, shame that Jim isn't keen to be part of it but thats life. We could do it through emails and calls if you wanted to so you wouldn't have to set aside loads of time.

Α

#### Email - 19/10/11

Hi Jo. Had a lovely time at G/daughters Graduation at M/c UNI. Be great to meet up again its been ages. Have you gone veggie or anything. I could make us Thai green or red curry? Last choir performance tomorrow at a church in Crumpsall. Gathering of people who work with the elderly. Off for 2 weeks then. Look forward to seeing you Wed Morn when we look through the data analysis.

Love Angela

### Email - 9/1/12

Hi Jo. Yes-As I said it was very wierd (sic) to read your paper. Didnt realize how long it went back. Agree with all youve (sic) written. Re how daughter and I looked after X [grandson] between us. This didnt (sic) work out. I was always out of pocket financially. I ended up doing more of physical care/washing/buying clothes/toys/worrying about school-in fact the more I did -the less she did.

Eventually I put an end to this when he started refusing to go to bed and announcing he was going home-where he stayed up much later. So this phase came to an abrupt end. I said 'Thats it! Get on with it!

I had to be very determined. After a while I started picking him up once a week after school for a few hrs. I think it was the best thing to do-for their relationship. It was difficult for her too. At this point she made a friend of his schoolmates mother and they support one another. Now I let him come around several eves for a few hrs. More because it darker nights and he gets problems if he's out. Bullying sometimes or disappearing and she's had to phone the police. But overall things have improved and had to change as he's getting older. From next week he will have his own support worker in school.

So my wellbeing has improved probably because I don't feel the weight of responsibility for X anymore and its freed up my time so I can do more stuff for me. Didn't realize till I read through your work how much my wellbeing is connected to

the kids and grandkids. So now Im spending more time with the choir. Im still quite surprised at this. That Ive stayed with choir. It costs about 10£ per week -bus fare and subs. But its so worth it.

Take care

Α

### Email - 18/2/12

This week I have given up alcohol and cut down cigs-1 2oz pack per week of tobacco and 5£ weed per week!!. Cost of these 3 had started to cost 30£pw. As I get 93£ and pay 20 bills a week its almost half of what I have to live on. Dont know how Ive managed. So again -wellbeing is affected for the better. It wasnt possible to sense any wllbeing in these areas before. In fact I was quite looking forward to death.

A friend has just moved to a fab part of Cornwall and Im taking 2 grandsons for what I hope will be a fab Cornish adventure hol in June.

So things have looked up. Even though I MOVED HERE 3 YEARS AGO -iM ONLY JUST ABLE TO APPRICIATE MY SURROUNDINGS AS TOO MUCH AGGRO WAS ALWAYS GOING ON.

thats all for now jo. hope to speak soon. anything else i could write about.?

mAYBE IN A FEW WEEKS TIME I could write about my recovery from drink/drugs and tobacco??? I hope so. A happy ending to my wellbeing

Regards Angela

#### Email -27/4/12

Hi Jo. Recieved paper .Thanks. Really enjoyed reading it. Thought you made your points really well. Didnt feel at my best, writing wise at the time-so glad you could use some stuff.

All those involved in gardens here are men. At the last place they were women. I think how they were seen as figures of authority, and respect was not to do with them being male or female, it was more that they could speak to, and work with people on both sides equally. Tenants, even the difficult ones were included and officials, like police, housing. Its different Here it not like the usual tenant groups who end up as stoogies for officials, doing their work and making their job easier, because they don't have the skills to work with tenants!! which is, just like officials-having said that the housing officials here arent much better than most of the other places ive lived in and that has been a constant drain onmy wellbeing and that of most of the people I know to be honest so nothing ever really changes that much.

Anyway I Really enjoyed reading through our work and it has been great to work on something which is bigger than me so Thanks jo

Love Angela

## Appendix K: Jim's Reflective Diary (2008–2011)

As part of the participatory research approach Jim and Alison agreed to keep reflective diaries in which they would log reflections, ideas, thoughts and potential discussion points for interviews.

#### Email - 21/01/08

hi jo

just a few words to say thanks for the other day, I really enjoyed the training sessions. I found the way he related his own experiences helped me to see things from different perspectives and I found them very valuable. I learned a lot and am looking forward to the next session. Jim

### Email - 31/5/08

Hi Jo

Hi Jo sorry I haven't written in a while but I got to thinking the other day about what we have been talking about and I first started to think about well-being when I was asked to write the wellbeing article when I sat on the Valuing older people board. Later I was asked to come in and work with you on the long-term conditions project and I saw that as an opportunity to do something for my well-being. then when you asked me if I wanted to work with you on your PhD I was really happy because It's something I can give back, something important, a lasting legacy really, being part of something that's bigger than just me and my life. Jim

### Email 14/ 10/ 08

I was sitting watching the birds the other day and thinking about wellbeing and you know my perception of what well-being means has changed through the course of my life. Obviously when I was young I wasn't really aware of the term but the underlying sense was that the here and now was the most important thing for my well-being. when I was young I lived in the moment but certainly having kids

changed that and all of a sudden well my well-being was just as much about their happiness and fulfilment as my own.

That got me thinking that there were things important to your wellbeing when I was young that have become much less important as I've got older, for instance my appearance but not what people think of me!! Jim

### Email 26/03/2009

Hi Jo

How are you? I've not been too good recently and it's made me think about being diagnosed with Parkinson's disease and how much that changed what is important to my wellbeing. I never used to really consider death before that and I definitely took my health for granted and I was more....? Well now my health and how I feel on a day to day basis is so important for my wellbeing. remember I told you I don't really drink now because it leaves me feeling grotty and I used to really enjoy a pint down the local. I still enjoy the local but drink no. This is part of how different my life and wellbeing is since being diagnosed with Parkinson's disease.

In response to what we spoke about a few days ago I do feel I've acclimatised to PD, well as much as anyone can, part of that I think is the fact that it has progressed quite slowly and so I suppose I have had chance to come to terms with it, and I've been lucky I know, many people don't. So yes I suppose that my condition and my sense of wellbeing have changed alongside each other. Jim

#### Email 4/04/09

Hi Jo

Wellbeing – my thoughts. If you bought a new car with your own personal number plate both would give a boost to your sense of well being, but in 10 years time the car will give you a down because of its age and condition, whereas the personal number plate will still be capable of lifting your spirits.

Financial improvement would increase your wellbeing and you would 'feel' better in short-term. Financial decline would decrease your wellbeing and you would 'feel' worse for long term. Things like money are linked to feeling good, for example having good job prospects and feeling bad if you have poor prospects.

Thats only part of it though, the research I did indicates that if you have the right philosophy on life and you are resilient your feeling of well-being will remain high.

Things that affect your feeling of wellbeing will vary in areas affected and degree of effect depending upon age, sex and possibly ethnicity. Jim

### Email 23/06/2009

Hi Jo, how are you?

I have been thinking about some discussion points so I have been looking through my old article ['Wellbeing'] and I still believe that the wise men throughout history knew what they were talking about, happiness is not the be all and end all but reflects what we have done with our lives and how we live. I do think personal wellbeing is about being content with who we are and what we have in our lives. Thinking about what I wrote then and what we spoke about I firmly believe we are sold messages about what we should aspire to for our wellbeing. the government and media tell us that we will be happy only if we have more money and more holidays or and more of those things that money can buy, they don't stress things like having parks in your community or access to country walks which can be so much better for your whole wellbeing not just your material wellbeing which is what they target. Government especially should address aspects of human experience and use scientific research to support what they promote as good for our wellbeing.

#### Email 19/01/10

Hi Jo

Following on from what we talked about I do believe that the public is sold a way of thinking about wellbeing, I think the media is mainly responsible for selling people

an idea about wellbeing like getting a new car every 5 years, and taking a foreign holiday every year. I didn't think I was buying into that until we discussed it in our last interview and I have been thinking about that quite a lot since. Its made me think and reflect a I also think the government plays a large part too for example through education which is selling this idea that every child is equal in talent and intelligence etc I mean this new thing where children don't come 1<sup>st,</sup> 2<sup>nd</sup>, or 3<sup>rd</sup> anymore that to me is ridicules [sic] and doesn't reflect real life. Then there is this thing about them all wanting to be a celebrity, well that just isn't realistic and gives them this idea of attainment which really undermines their wellbeing, particularly their future wellbeing. you can see so many young people nowadays becoming disillusioned with their lot in life because they grew up thinking they would be super rich and famous and when that doesn't happen normal life must seem such a let down.

#### Email 08/07/10

Hi Jo I hope that you are ok? I am doing ok and so is X [wife] thanks for asking.

I have been reflecting on what we spoke about together last time we met, I really hadn't thought that I was being influenced about my ideas of wellbeing by media and government but the more I think about it the more I realise that I am! I don't think I'm influenced as much as the majority of people though because I had already established some well grounded and researched beliefs about wellbeing from having written that article but I strongly believe that the way wellbeing is promoted is influencing others, especially young people or the politically niave or those who don't tend to think about deep concepts.

I'm really not sure that it is going to be useful for you to try and define wellbeing perhaps you have a point in that a definition may detract from its usefulness as I think you called it a 'fluid term'. Jim

#### Email 17/09/10

Hi Jo

Well-being does really appear more and more to be the governments buzz word. I have been wondering how much influence these sort of external factors have on wellbeing. I think they probably have a really important influence on many people. I know that some research indicates that personality is important for wellbeing so E.g. do some people naturally have a good sense of wellbeing and are these less influenced than others?

There are also other considerations put forward such as what constitutes national well-being? Are aspirations largely tied to your class? How much might your class channel what you think is important for your well-being? I guess these are all things which you have been thinking about too. How is the PhD coming along? It has been quite a while now hasn't it. Jim

### Email 2/11/10

Hi Jo

I know we have started discussing how much I want to get involved in helping with the analysis. I am not sure that I can commit to getting that involved. In many ways I would love to but I'm not sure that it would work out for us both. I don't think I've really got the inclination necessary to be an active participant and would probably prefer to just carry on with the interviews and diary. All the best Jim

#### Email 12/1/11

Hi Jo

I was thinking about culture and wellbeing and I have decided that I don't think wellbeing affects culture but I think culture could affect wellbeing. So in terms of culture I feel a culture connection to people who live in same region of world, have a similar skin colour, similar earning power, same religion, same traditions, same moral values and treat others as equals. I don't think it makes sense though to try and define wellbeing which is relevant to everyone although of course there are things we all want for our wellbeing regardless of our culture, such as love, health, friendship, family. In some cultures though wellbeing is about the collective good whilst in western cultures its more about the individual and I think things like that

make it tricky if you're trying to define wellbeing. Personality also gives you different ways of looking at things. All I expect from life is a kick in the pants occasionally. Jim

## Email 22/04/11

Hi Jo

I have been reading through my article and I stick by what I said in my article that whilst it is easy to think of well-being as happiness, but it is more than that. It is about having meaning in our lives, developing as a person and feeling that our lives are fulfilling and worthwhile." Its interesting to think of what well-being means to different people. I do think that age plays an important part, your wellbeing ancd what is important to it is different when you are a kid from when your an adult and different again as you get older. I have also been thinking about the possible differences between men and womens wellbeing, there must be some quite different aspects I suppose but I think there are more things in common than different so that relationships and interests, the happiness of your family and having good friends is the same. Perhaps things like work and what leisure activities you enjoy are different though and may have different impact on well-being. I think close personal relationships are of great importance though more than money although of course that is important especially for those people who don't have enough to get by or provide for their children. Jim

# Appendix L: Second Interview with Jim - 27th November 2009

Interview took place in my workplace

Q - Can I start Jim by asking you if you think an operational definition of wellbeing would help or hinder its usability as a concept for improving services for older people with complex health needs?

A - It's impossible to define wellbeing and so I wish you good luck Jo. Everyone has a different slant, it's probably 90% core wellbeing and the rest is individual, unique, but there is probably so much variation within that 10% that you couldn't pin it down if you wanted to.

Q – Jim when did you first started to think about personal wellbeing?

A - I was invited to a seminar on wellbeing for 'Valuing older people' Jo and I used to produce news-sheets for them and then started writing articles too and I decided to write the wellbeing article, so I suppose it's been 4 years now that I've been thinking and talking about wellbeing.

Q - How did the article come about Jim?

A - Through the community health council.

Q – Would you say you started to think about wellbeing through a health perspective then Jim?

A - Oh yes but also through consideration of things like the education system because kids wellbeing is let down by having unrealistic goals.

Q –Do you think that having unrealistic goals undermines personal wellbeing?

A - Yes, it gives them too high expectations and it knocks them back when they don't achieve them. People should be trained to their skill level Jo because otherwise it sets them up for a fall and then depression, [pauses to drink some water] I was happy in my work and I believe if you're happy in your work then its

not work and if you give someone too high an expectation they get depression. A sense of achievement is important for your wellbeing you feel better.

- Q –do you think the education system then is undermining children's personal wellbeing?
- A We're being sold a message about achievement by the government through the education system and yes I think that sets them up for a fall.
- Q Do you think a sense of achievement is important for personal wellbeing then jim?
- A Yes Jo and we're being sold a message about wellbeing by everyone who wants to influence us.
- Q Who is selling you this message about wellbeing J?
- A The government, the media [pause], education system, health services.
- Q Why do you think they're trying to influence you?
- A To sell you something or to try keep you happy politically.
- Q Do you think these messages have influenced your conception of wellbeing?
- A Well they put their own slant on it but I haven't been swayed. I've listened to a variety of speakers on wellbeing who have their own ideas but I think if you're happy and look to the positives then your well on your way to wellbeing. [long pause] You do pick up other people's ideas though, [pause] there have been occasions when I've said "yeah I hadn't thought of it that way" and you take it onboard. [clears throat] When I started looking at wellbeing I probably did it from the social worker side [stammers]. I looked at articles online and I saw differences in wellbeing from country to country, they would place emphasis on different things, I looked mostly at the UK, Australia, US and Canada and was probably more in line with Australia.
- Q Are you aware of the local government act 2000 which made local government responsible for improving citizens' wellbeing?

Jim nods

Q - Do you think governments should be responsible for legislating to enhance citizen's personal wellbeing?

A - Yes I do definitely there are things that people can't do for themselves they need the power and persuasion of government to implement changes for the better, that's why politics are so important. Wellbeing appears to be one of the governments new buzz word and so much of what they do [stammers] affects our wellbeing.

Things like employment laws which can change workplace practice and when you think of how long people spend in the work place that will have an important impact on people's lives, the education system affects children's education, aspirations and our future economy, providing more money for the NHS [clears throat] so we have got a health system that supports people, protecting the environment too or else our kids won't have a world to live in or not one worth living [pauses to drink some water].

When I was a kid I spent all my time playing with my friends in the woods, they were simple times. All that was important to me then was having three or four meals a day, [drinks water] and the freedom to roam. I think being cocooned like that was what made my childhood so special [clears throat]. In the woods we could be anything, we played without a care in the world [gulps]. My thoughts often drift back to memories of playing in the woods, that's why nature and being close to nature is so important to me. I get a high from sitting and reflecting on when I was a kid playing in the woods.

Q – Are you ok Jim, would you like to take a break or end the interview?

A - I'm fine Jo.

Q – Ok but if you want to stop at any point Jim just let me know.

A - Ok

Q – Jim you said before you think there are age differences and gender differences in what is important to peoples personal wellbeing do you think there is a cultural aspect to personal wellbeing?

A - I think there are cultural differences and religious differences too. Those Islamic fundamentalists who strap explosives to themselves feel good just before they push the button. Kamikaze pilots what were they all about [shakes head], it shows the cultural differences. People aspire to different things but whether there are big differences due to culture I don't know. I went to the Ukraine few years ago and they'll aspire to a 10 year old fiat and we'll aspire to a new mini. They'll be just as happy with a 10 year old car as I would with a new car.

Q - You talked before about being sold a message about wellbeing, do you think that perhaps you have been 'swayed by media messages about what's important for your personal wellbeing' and that subconsciously you've been sold a dream that aspiring to new car would enhance your wellbeing?

A - Yes I guess so, [pause] I hadn't really thought of it before but yes of course I know that advertising does impact and influence your wellbeing.

Q - What other messages might we as a society have been sold about wellbeing?

A - Having a strong economy, jobs, owning our own homes, going on foreign holidays and so on things that we would all like.

Q - Do you think these things are important for personal wellbeing?

A - Well I think we've got it the best we've had for a long time with this labour government apart from the greedy bankers. I mean money might be important in the short term but we need to look at ways to improve things for people by other means than money [clears throat]. If you give people more money they'll just spend

it but it won't add to their wellbeing, once you've got to a certain point [coughs] money doesn't add to your wellbeing. I think financial security is more important than money as such. Having financial security and savings means we can help the children out with things before we die [pauses to drink some water]. The pleasure and satisfaction I get from being able to help them is a big thing for my wellbeing. It's nice to be able to take the worry off them before you're dead.

Jo – Well I think we should perhaps wrap things up now Jim, thanks so much for your time.

Jim – No problem Jo I enjoy talking with you and being involved with your PhD.

Jo – Well I couldn't do it without you and Alison so a big thank you to you Jim and we can speak again soon.

Appendix M: Third Interview with Alison - 23rd March 2010

Jo - Hi Alison how are you?

Alison - I'm ok Jo how's things with you?

Jo - Not too bad did you have to wait long for a bus?

Alison - For once no!

Jo - Well that's good, can I get you a coffee or tea?

Alison - No I'm fine Jo

Jo - Ok shall we make a start? Alison nods.

Q- I know we have talked about health and wellbeing before but can I start by asking you if your health important to your sense of personal wellbeing Alison?

A- yes, I go through these cycles of mental ill health which have been a part of my life since I was around 11, I can identify them with all the phases of Bi-polar. In the middle of the cycle I feel most normal but I can experience rapid mood swings then I hit the low when I may go from minor self condemnation to self loathing and by end of week I might be feeling suicidal. But I now know this period is followed by the 'high' and this is the good part of the cycle. I feel like I have gone from experiencing the world in black and white and into colour, during this period my wellbeing is great, I feel confident and productive but I often burn myself out during this period.

Q- Can I ask you Alison about your mental health in relation to your wellbeing?

A- yeah having mental health issues has made me prioritise what's important in my life.

Q- Can you expand on that a little?

A- well it restricts my wellbeing because if you're anxious and depressed you can't have much wellbeing. And it's like I feel compelled to act sometimes in a way which is very in your face but I couldn't stop myself from doing it however hard I try. It's that feeling of compulsion when you're in a different cycle that is quite scary. Let me tell you about my new neighbour upstairs jo. He moved in couple of weeks ago and left loads of bags outside his front door. I was furious so I left him a long shitty note about bags left outside and respect for others. He scurries past me when he sees me now and I have been thinking about why I reacted like that but whether I acted like that because of bi-polar or through my experience of neighbours and rubbish I don't know. See I'd experienced this sort of thing with previous neighbours and this may have led me to expect likewise behaviour from him and that may have been what led me to behave the way I did.

Q- Can you think of what else impacts or influences your sense of personal wellbeing Alison?

A- As regards my wellbeing I feel at my happiest when I've spoken to the kids all on the same day and they are all good, that's my highest feeling of wellbeing. The financial security of having a phone so I can check on X's [grandchild] welfare is paramount to my wellbeing. A lot of my wellbeing is practical about being able to afford stuff, for example, decent warm clothes etc the things that have always been a problem. Car boots and charity shops help my wellbeing. I can get gifts/toys for the children, makeover the house etc. All the things that others take for granted and buy new. I always felt bad about not being able to buy much for the kids so I feel better now it is easy to pick bits up. Definite wellbeing when I find nice winter coats for myself daughters or grandkids for £2. My wellbeing is directly influenced to a large degree by my kids and grandkids and where I live.

Q – Does where you live impact on your wellbeing Alison?

A- well having internal space and not being overlooked by anyone is really important. Living space is massively important to me so that you can have a level of peace and quiet and to have other rooms which means you can take yourself away from a noisy situation so personal freedom is key also. The way the place I live in

now has been designed is really good because it's well lit and all the paths are overlooked by windows. This means that even in winter the route to the shops is overlooked and so you don't think people would chance attacking you. House design is so important for my wellbeing because proximity to your neighbours can really get to you especially when you move from having kids and living in a house to being in a 1 bed flat. You're so much closer to your neighbours and you're surrounded on all sides. It's like I smoke but it's not seen as being acceptable in older people so I have a constant fear of losing my home because someone has smelt the smoke. It's social paranoia which does cut into my general wellbeing and the sense of wellbeing I get from smoking. This place I live now has a sense of civic pride, the gardeners are smart and polite and it's like you're being cared for as well as the environment. Psychologically I feel safer in this place because there is a resident who is big guy and he has made it his job to make his presence known on the green spaces out front

Q- We have talked about wellbeing a lot in the past couple of years but what does wellbeing mean to you?

A- Wellbeing is knowledge! For me its knowledge and understanding, without them I wouldn't have wellbeing. I can help my kids now and if I didn't have wellbeing I'd be like those women who try and keep people and their kids down. Awareness gives me a sense of personal wellbeing but the power to change things I'm not sure I've got. My illness stops me from seeing things and having the power to explain things on the phone things that I want or need without getting angry and frustrated. There is a positive side though in that with bi-polar I can wait for the highs and lows to challenge authority but I can't control this illness.

Q- We have talked before about personal growth being important for your wellbeing can you tell me why personal growth so important to your wellbeing Alison?

A- Because I've experienced the frustration of not experiencing that growth. I went to grammar school then got stuck on Langley with no hopers! I went to a good primary school and then grammar school but was kept down a year at secondary. I

learnt nothing at grammar. I think that going to secondary school was when bipolar kicked in because I couldn't find my way to classes or remember the layout of the school which was really simple and I know that's when it all started.

Q -Do you feel that your early years were important for your current sense of personal wellbeing Alison?

A - I think it has played a large part in my wellbeing. My feelings of insecurity, issues with self worth, self esteem, handling disappointment, issues of addiction, lack of education and learning difficulties have and continue to have an important bearing on my day to day sense of wellbeing and what I expect for my future wellbeing.

The more I think about it the more I think that there were 2 periods in my life which impact most on my wellbeing. The AA [Alcoholic Anonymous] and going to secondary school. My life changed at 11 because I went from having an authoritarian school environment to a school which lacked any of that planned laid out structure. Nobody at school noticed that I wasn't engaged or that I wasn't able to deal with the school situation because I was bi-polar if they had I might have been able to make the most of the opportunity I had going to grammar [school].

Q- We have talked before about the role knowledge, understanding and awareness plays in wellbeing can I ask you Alison why are they so important to your wellbeing?

A- I spent most of life mainly hanging about with uneducated people and I was a big fish in a little pool, I didn't see my peer group from Withington and as I got older I felt more vulnerable and so more and more liked strong people who others were scared of. It was that old thing that I didn't know what I didn't know. The reason for that was that I knew and been taught from childhood that knowledge, understanding and awareness is very important for feeling like your part of something, being accepted for who you are, feeling comfortable in yourself and my time in north Manchester undermined all that. I was with ignorant people who didn't care about that so I began to not care about it.

Q- So during that period of your life when knowledge, understanding and awareness weren't so important for your wellbeing what was important for your wellbeing?

A- Things that coordinated in the house, status, having a nice house, the kids looking well turned out and the fact that I was married to someone big, well a gangster! After that it was going to the AA and gaining acceptance of having people around me who were mature and I had to surrender being a big shot and accept guidance to help me get on the right track.

Now things like having a home that's hospitable is important that I can be hospitable in and have the where withal to put my family up comfortably in, to have enough bedding and beds and the space and peace for them to feel happy here. I personally would not be happy if I didn't have enough space. I know that my son can come and stay here with his wife and feel happy and not worry about his car being stolen or having to go out and face rowdy kids making noise and trouble like in my old place. That's mega important, knowing I have somewhere my kids and grandkids can come and stay because it means you can give back to your family and it's not always them doing the giving. This is good for family dynamics and relationships and the wellbeing of myself and my family.

Q- Alison we have talked before about the importance of self realisation for your wellbeing can you tell me why self realisation has played and continues to play a part of your wellbeing?

A- Yes, I realise now that my personal development and the learning process has always been a driving force behind my wellbeing I just didn't realise it and when I feel like that has stopped I move on like with being active in the community to being on tenancy groups and stuff.

I think getting older has helped me to prioritise better and I don't feel driven by my urges as I did when I was younger, that urge to be sexually driven, stay slim, look good and be attractive to men. I feel very much that I reflect and induce higher thinking to enable me to prioritise about what I need to spend time on whereas

before I was a feather on the wind. And it's all about promoting or maybe promoting isn't the right word, protecting my genetic material, hoping that they in turn will have that knowledge and understanding and protect the next generation. That is a strong driver in my life now but I've invested in other peoples sons in the AA and now people are investing in my son in London. I'm hoping now at the carers association that people will invest in X (daughter) and I carry on investing in the stuff I do with C and yourself. 20 years ago there was no way I would have thought my wellbeing would be so affected by kids and grandkids, I used to hope they would move away so that I could have a life of my own, I think because I had kids so early and had no life before them. My friend used to say 'genes, it's all in the genes, I've seen my sister top herself and my brother would have done too if he hadn't have died first. I used to think she was just making excuses because she was an alcoholic. I've come to understand what my kids need now, what somebody gave to me in those AA rooms; my own children need that now. When my kids are happy and on track my wellbeing is ok.

Q- Is it important to your wellbeing to feel that you are investing or reinvesting? Yeah definitely, I often feel powerless...I want to help older and more vulnerable members of the community to be involved and be listened to and to have their opinion heard. They should be treated with the respect they deserve, the respect we all deserve.

I feel more alive when I'm doing stuff like this, like on the bus coming here I felt I don't belong on this planet but now I'm here talking to you and I'm contributing to the bigger picture even if though I'm only a tiny part of it that's part of my wellbeing and on the way home I will feel like I've done my responsibility.

Jo - Well Alison it's getting late so we should probably call it a day.

Alison - Yeah we can talk by email or phone soon.

Jo - All the best then Alison

Alison - And you Jo.

507

Appendix N: Interview with Tim - 17th August 2008

Details: Interview took place in the X centre for people who have experienced brain

injuries

Hi Tim

My name's Jo and I'm a researcher from MMU undertaking research for my PhD.'

I've come here today to interview people in order to explore personal wellbeing

amongst older people with complex health needs which forms an additional strand

of an ongoing research project entitled 'Changing services for people with long

term neurological conditions: Promoting wellbeing within service provision.' My

research aims to gather the views and experiences of older people with complex

health needs in order to better understand personal wellbeing. The interviews will

take between 30-45 minutes, if at any point you don't understand the question

please just ask me to repeat it, if at any point you no longer want to take part you

can stop the interview, if at any point you want to take a break please tell me. I

have read through the information sheet with you and the informed consent form

which explains that the information you give during the course of this research is

completely confidential.

Q – Are you happy to be interviewed by me today Tim?

A - Ok

A – Could you just sign this consent form for me please Tim?

[signs form]

Q – Tim is it ok if I call you that or would you prefer something else?

A - That's ok.

Q – Can I ask you how old are you Tim?

A – I'm 57

Q – Are you familiar with the term wellbeing?

A – Yes I've heard it about

Q – If you had to define it what would you say it meant?

A – [puffs out cheeks] that's a tough one. Being happy in your own skin maybe.

Q – Where have you heard or seen the term being used?

A – In newspapers, on the tv and radio.

Q – Can you tell me in what context is it used?

A – All sorts, they usually talk about it in relation to health but also heard it in relation to community, exercise,..

Q – Do you think that has influenced your idea of what you think wellbeing means?

A – [long sigh] I suppose it must have done because I hadn't really heard of it before so I had no idea of what it meant. I suppose its all part of the governments' attempts to make us more responsible for our actions which I suppose when you consider how many young people have got liver damage through binge drinking may not be a bad thing

Q – Is wellbeing a term you would use in general conversation/in your social circle?

A – Not really, I guess it would depend on who I was with I would be unlikely to use it down the pub for example.

Q - Why?

A – It's probably a bit too deep.

Q – What things are important for your personal wellbeing?

A – My family, going to the pub, meeting up with people you can pass the time of day with without knowing them or meeting them again. I think I took that social interaction for granted before my accident. The natural environment is really important for my personal wellbeing. I'm lucky I've got a big garden which was the

main reason we bought the house. We even put the conservatory to 'bring the outside in' as they say. It's a haven for me really I sit and watch the birds and the bees because my girlfriend has planted lots of flowers so it's a hive of activity out there. When I'm feeling low and the weather's ok I will go out and do a bit of gardening. Not only does it lift my spirits but being active means that I usually sleep better too. Being able to get out into nature, I love being in the outdoors although I can't walk very well I try to get into the country like Edale as much as possible. I get a sense of peace and perspective when I'm in the country it's very calming and soothing.

Q – Why are these things important for your personal wellbeing?

A – My family is the most important for me. I suppose I grew up thinking a man's role is to look after his family and I've always tried to be a good husband and father. Losing my job undermined that because I was no longer the bread winner, suppose it hurts my pride that I can't help the kids as much financially as I used to do.

A – My speech has made things awkward for me, I lost lot of mates after my accident till then I hadn't realised how important it was for me to mix socially. That's a big reason I come here, everyone's in the same boat, we can all relate to head injuries and how it changes your life. Were like a big family here, we chat, play pool, listen to each other's problems so now

Q – What things negatively affect your personal wellbeing?

A – My wellbeing is negatively affected by feeling anxious about paying the mortgage and financial worries, my divorce hit me hard too and it took me a long time to adapt but once I accepted it I felt able to start thinking about enjoying life again, that I could be happy and satisfied with my life and since my marriage broke down I've had to get used to doing things different and now I've been living with a woman that I can be close with again for over a year.

Q –By close do you mean sexually intimate?

A – "Yeah, that side of things hadn't been there for a long time, my wife no longer saw me in that way and it was hard for me, I felt unwanted and didn't feel comfortable in my own home anymore. The woman I'm with now [long pause] it's hard to explain [long pause] but I feel happier in myself again."

Q – Has your health become more important to your sense of wellbeing since you've been diagnosed with complex health needs?

A – Yeah I never really thought about my health before but now I think about it everyday, several times a day because it limits me so much.

Q – Has your health condition had a direct influenced on your personal wellbeing?

A – Yes, I lost my wife and damn near all my mates too, you're a different person after a head injury and the quicker you accept that and get on with things the better. I lost my job too because of my head injury so it's a good job I got my mortgage before it happened because otherwise I dread to think where I would be now.

Q – Do you still own your home Tim?

A -yeah we own our house and that's very important for my wellbeing.

Q – Why is owning your own home important for your wellbeing Tim?

A – [pause] Well I feel like we have some control over our home environment and neighbourhood, having that control and respect makes life easier because you know you can go ahead and change things without having to get permission or wait till the council can afford to do the work like some people I know. It's important to have respect, it's part of wellbeing.

Q – Have any of the services you use impacted on your personal wellbeing?

A – Yes! This place has made a real impact I feel like this is my 2<sup>nd</sup> family and I can always come back if I've got any problems and they help me to sort them out, its important for your wellbeing I think to be part of something, to be able to share your problems and to get rid of them.

Q - Can I ask in what way using this service has impacted on your wellbeing?

A – it's a way of getting out, being more sociable, getting out to make new friends after you've lost those mates you used to have before you're accident. We've all had damage to our heads here so we all understand each other, if we went to other places they wouldn't understand what we're going through or what we say.

Q – Do you find that people struggle to understand you?

A – Yes and its a common problem for people with head injuries, that's one of the hardest things to adapt to being able to make people understand you its so frustrating and depressing but this place it's like a family unit, it's comfortable and there's trust here. Plus it makes you more independent, getting here on public transport in itself makes you more independent.

Q – The local government act 2000 made local government responsible for improving the wellbeing of its citizens, can you think of any examples of how local government has sought to improve the wellbeing of citizens?

A – The thing is if I was lesbian say or Muslim or a refugee the council would make sure I got the support I need, there are services available to them and that's what's unfair, those in power decide where to spend the money and there is nothing I or anyone else can do about it.

Q – What sort of things could local government do that would have an immediate positive impact on your personal wellbeing?

A – Protect the environment more I love to be outside see trees, birds etc but I have to travel for that because they let companies build on what little green space is left.

Q – What sort of things could local government services do that would have an immediate impact on those things that have a negative impact on your personal wellbeing?

A – "I know how lucky I am to live in an area which still has some centres, it means I get to meet with people and that's important for my wellbeing. Some of those who

live in north Manchester have got it really tough, they don't have half the things we do in south Manchester. Its relative though isn't it because I've got friends who live in Trafford and they are better provided for than us. So I think local government should be trying to keep hold of those places where people of all ages and that come together."

Q - Do you think that the term wellbeing is a useful concept?

A – I'd refer to a harmonious holistic life as 'QoL' I've only used wellbeing because that's what you have used. What was wrong with QoL anyway?

Q - Do you think that defining wellbeing would help or hinders it usability as an indicator in how well services provide for older people with complex health needs?

A – "I think it would hinder its usability as people's levels of self awareness are very different and everyone would see different things as being important for their wellbeing.

Ok well thanks very much for talking with me today Tim and all the best.

Thanks all the best to you too.

# Appendix O: Interview with Ivy - January 9th 2008

Details: Interviews took place in the hillside resource centre

Hi Ivy my name's Jo and I'm a researcher from MMU undertaking research for my PhD which forms an additional strand of an ongoing research project entitled 'Changing services for people with long term neurological conditions: Promoting wellbeing within service provision.' This particular piece of research seeks to explore personal wellbeing amongst older people with complex health needs. The research aims to gather the views and experiences of older people with complex health needs in order to better understand personal wellbeing. The interviews will take between 30-45 minutes, if at any point you don't understand the question please just ask me to repeat it, if at any point you no longer want to take part you can stop the interview, if at any point you want to take a break please tell me. I have read through the information sheet with you and the informed consent form which explains that the information you give during the course of this research is completely confidential.

Q – Ivy is it ok if I call you that or would you prefer something else?

A – Ivy's fine love.

Q – Are you happy to be interviewed by me today Ivy?

A - Yeah

Q - Would you just sign this consent form for me please?

[Ivy signs form]

Q – Are you familiar with the term wellbeing Ivy?

A - Yes I've heard it before Jo but wouldn't say I'm familiar with it.

Q – Where have you heard or seen the term being used Ivy?

A - Well only because they said you were coming in today to talk to people about wellbeing, before you came today they said at coffee break that you were coming and you wanted to talk to some of us about our wellbeing.

Q – Ah somebody else mentioned that, did they say anything else Ivy?

A - Yes they asked us if we knew about wellbeing, asked what we knew about it and then explained what it meant.

Q – What did they say Ivy?

A - They said it was to do with things like independence, managing your condition, empowerment, keeping healthy and active

Q – Would you agree with that or does the term mean something different to you?

A - As far as I understood it it's concerned with your QoL Jo. Actually I've heard it on the radio too thinking about it.

Q – Do you think that what they said here about wellbeing and what you have heard on the radio has influenced your idea of what wellbeing means?

A - To be honest Jo I didn't really listen to what they were saying today and when they were talking about it on the radio I didn't really understand it seemed a bit complicated and no one seemed to agree what it meant.

Q – Is wellbeing a term you would use in general conversation/in your social circle?

A - No I wouldn't

Q – Why?

A - It's just not a word that you hear in every day conversation, I'm not sure people would know what I was talking about, not sure I would know what I was talking about perhaps I will have a better idea after talking to you [ we both laugh]

Q – What is important for your personal wellbeing Ivy?

A - For me wellbeing is knowing that my family is here for me, that they are safe and happy and well like my granddaughter who comes and cooks for me every night.

Q – Why is that important for your personal wellbeing?

A - Well Jo I would be lonely if it wasn't for my family. I'm 60 and live on my own and don't get out much so I feel safer knowing that she will be here in the evenings. I need more in my life than that though, sometimes I feel like my whole life has been about looking after my kids and then helping to look after their kids. Don't get me wrong I love them all to bits but sometimes I think about how different my life and my kids might have been if I'd had waited a bit. I don't think I achieved anything but having kids! See it's different now, but when I was growing up women got married, had children and stayed at home that was it.

Q – How is your health Ivy?

A - I had a stroke in 2005, and I've got angina. I was at home and I fell to the floor had a stroke and crawled into living room because couldn't get up so called doctor and she took ages to come and she said hadn't had a stroke and I said I had my arm and leg wouldn't work so she said she'd call rapid response and I said no I want an ambulance and it finally came and took me to hospital and they said hadn't had a stroke too so sent me to respite place and they called another doctor and he said I'd had a stroke. They sent me back to MRI for a scan and tried to give me Physio but it was too painful and I couldn't walk at all. It was 3 weeks between having the stroke and any one believing me.

Q – How long were you in hospital?

A - About 3 weeks, can't remember what it was like at hospital then after that went to respite.

Q – Were you happy with your treatment?

A - The nursing was fab but if they'd believed me in the 1st place...

Q – Have any of the services you use had a positive impact on your personal wellbeing?

A - Respite was very good it was in levenshulme and it was very good I even had some Physio there, that was good, then went home for few months and they kept on with the Physio but at home. Getting physio is really important I know some in here who didn't really get enough and its messed them up. At first I couldn't move my hand and it was really painful having physio but I'm glad I kept up with it.

Q – Have any of the services you use had a negative impact on your personal wellbeing?

A - Well apart from the physio they've done precious little for me but luckily I've got a very good family, they've done everything for me, and got all the stuff I needed.

Q – How could service improve your personal wellbeing?

A - look at my leg jo I need help with it [she pulls up her skirt to show me her leg which looks very sore and swollen and miss-shapen] I need a wheelchair really but I've rung up aids and adaptations and this worker keeps telling me I've got the lightest walking frame and I haven't but you get sick of asking you just give up. You know their attitude is "well she won't be here for long so don't bother" you're age goes against you. I wouldn't bother to ask my doctor either it's the same thing your age goes against. They shouldn't be able to discriminate against people because of their age. I'm lucky though my granddaughter finds out stuff for me.

Q – Does this service I mean the service at this centre impact on your personal wellbeing?

A - Well I would like to get out and about more. I was always out when I was independent and I miss that. I've put on an awful amount of weight and I'm not eating more but as doctor says I'm not getting the exercise. It's not doing my chest any good so he said the things they do here should help. Things like yoga, exercising on Monday morning, and bingo, though it will be better when we get a proper bingo machine. There's also exercising on Wednesday morning and they've just

started doing arts and crafts, painting and stuff and Friday morn is tai chi but there's not a lot to do in the afternoons. I know they do try but most of that stuff I can't do though. I thought coming here would boost my spirits and give me a reason to get out of bed but they need to do things which the less mobile ones can do too.

Q – Is health important to your sense of wellbeing?

A - It's one of the most important things! Wasn't when I was younger but the older I've got the more important its got [rueful smile]

Q – Has your health become more important to your sense of wellbeing since you've been diagnosed with complex health needs?

A - Yes because I have to pretty much rely on people now for the things I need.

Q – Has your health condition had a direct impact on your personal wellbeing?

A - Yeah, I lost my job and it just really hit me, after twenty odd years of working suddenly I was stuck in the house all the time and to make it worse there were no women in the house and I missed that female companionship. I was lucky that the nurse sussed what was wrong and encouraged and supported me to get out of the house and use the community centre to meet people.

Q – Has your health condition had an indirect impact on your personal wellbeing?

A - [sigh] Yes I became less confident, I lost my independence, my self respect, all of a sudden there was nothing to get up for and I couldn't do much for myself. I've worked hard all my life and I felt like I'd lost part of my identity.

Q – Do you still feel like that now?

A - I've come to terms with it now, Jo well pretty much, but it was a struggle at first because I couldn't afford things and I struggled with the bills. I still get a bit down when I can't buy the grandkids things but as I say to them all the time money doesn't grow on trees. Youngsters nowadays think it's their right to have a

television in their room, and the newest mobile phone and computers. When we were kids we had none of those things but we were happy playing hide and seek in the park, or climbing trees or whatever. I blame the media they've put these ideas in their heads that and the pop stars and footballers who have ridiculous amounts of money. All kids seem to want now is fame and fortune and they can't seem to take pleasure in the small things in life like we do.

Q - Do you think the media and advertising influences what is considered important for wellbeing?

A - Oh yes, Advertising definitely does have an impact on and influence your wellbeing.

Q – Has growing older had an impact on your personal wellbeing?

A - Yeah but it's not all bad by 50 I'd learnt what my strengths and weaknesses were and that's given me more confidence and [pause] it allows me to do stuff I'm good at so I feel like I'm being useful and helpful for my family.

Q – What things could local government services so that would improve your personal wellbeing?

A - In my first year I went on boat trip it was lovely but have had so much trouble here about that. X who organises the trips takes the favoured few and you don't hear about them till they've come back or when it's full and they say they put things on the wall but like the theatre that's 15 quid and I can't afford that. Last year we didn't go out at all.

Q – So what sort of things would improve your personal wellbeing?

A - Holidays, garden centres things like that jo, [ivy leans in towards me and lowers her voice] the committee here is rubbish, Y [chair of the committee] is a lovely man but no use to us. I've got a big mouth and I'm always complaining about this on everyone's behalf. They had school kids in here once and have put plays on and that was very good, everyone enjoyed the kids being here because most in here don't really have much contact with the younger generation. Over Christmas we

had people coming in to play music. They say they put notices up on the wall but they don't and if they did what about people like me with bad eyes. They need more activities in the afternoons but I suppose management gets sick of those who don't join in and there are quite a lot of them and I know a lot have Alzheimer's and so can't because of their condition but we've all complained about the chosen few going on all the trips.

Q – The local government act 2000 made local government responsible for improving the wellbeing of its citizens, can you think of any examples of how local government could improve your wellbeing?

A - Well ring and ride wont cross borders and that's a bone of contention for everyone and they're never on time, it's like older people are second class citizens and you don't get owt you don't pay for you know, the bus that brings you here you pay for, you pay for it all and now the government's trying to take your pension away.

Ok ivy well thanks so much for talking with me and all the best.

That's ok it's been good to get a few things off my chest Jo.

### Appendix P: Well Being - Jim Trotman

Throughout history wise men have said that happiness is not a goal but a consequence of how we live, that it comes from being content with what we have.

Today, we are sold a different message - that we will be happy only if we have more money and more of what money can buy, human experience and scientific research do not support this belief.

Our well-being is shaped by our genes, our upbringing, our personal circumstances and choices, and the social conditions in which we live.

Our collective well-being is improved if we live in a peaceful, flourishing, supportive society, so promoting well-being should be a public as well as a personal task.

We often think of well-being as happiness, but it is more than that. It is about having meaning in our lives, developing as a person and feeling that our lives are fulfilling and worthwhile.

Well-being comes from having a web of relationships and interests. Family and friends, work, leisure activities and spiritual beliefs can all increase our well-being. The intimacy, sense of belonging and support offered by close personal relationships are of greatest value. Material comforts are essential up to a point and there is no doubt that poverty remains a serious problem. For most people, more money would add little to their well-being.

The following suggestions could help increase our national well-being.

## 1. Work

Fulfilling work is essential if we are to flourish. Workplaces that provide secure, rewarding jobs should be encouraged. Workplace flexibility including quality part-time jobs, should operate in the interests of the employee as well as the employer.

Unemployment is more damaging than just the loss of income and disparaging the unemployed only increases their anxiety and sense of exclusion.

### 2. Time

We overestimate the amount of well-being associated with high incomes and long work hours. As a result our families, health and sense of achievement suffer.

Spending more time with our families, friends and communities would make most of us happier.

If we took productivity gains in the form of a shorter working week rather than higher pay we could improve our quality of life and create new job opportunities, all without reduction in pay.

#### 3. The Environment

A healthy, diverse natural environment is essential to human well-being.

Failure to tackle bio-diversity loss, pollution and waste will affect the well-being of future generations.

## 4. Education

It is impossible for all students to come first in their class and our education system should stop pretending they can.

Our schools should be dedicated to creating capable, confident, emotionally mature young people who are equipped to face life's ups and downs.

### 5. Materialism

Buying a particular brand of margarine will not give us a happier family and owning a four-wheel drive will not deliver us from humdrum lives. Advertisers seek to persuade us otherwise.

Advertising makes us more materialistic even though we know that people who are more materialistic are more self-absorbed, less community minded and less happy.

### 6. Communities

A flourishing society is characterised by vibrant, resilient and sustainable communities. Loneliness and isolation cause much unhappiness, especially among single parents, unemployed people, older people living alone and people with disabilities and their carers.

# 7. Future society

Economic growth is treated as the panacea for our ills, but for affluent societies growth in GDP has almost no connection with improvements in national well-being

A flourishing society is not a futile hope. Democracy offers people the opportunity to shed their cynicism and commit themselves to creating a better future.